

Contents lists available at ScienceDirect

# Progress in Materials Science



journal homepage: www.elsevier.com/locate/pmatsci

# Mechanisms of aortic dissection: From pathological changes to experimental and *in silico* models

Malte Rolf-Pissarczyk<sup>a</sup>, Richard Schussnig<sup>b,c</sup>, Thomas-Peter Fries<sup>c</sup>, Dominik Fleischmann<sup>d</sup>, John A. Elefteriades<sup>e</sup>, Jay D. Humphrey<sup>f</sup>, Gerhard A. Holzapfel<sup>a,g,\*</sup>

<sup>a</sup> Institute of Biomechanics, Graz University of Technology, Austria

<sup>b</sup> High-Performance Scientific Computing, University of Augsburg, Germany

<sup>c</sup> Institute of Structural Analysis, Graz University of Technology, Austria

<sup>d</sup> 3D and Quantitative Imaging Laboratory, Department of Radiology, Stanford University, USA

<sup>e</sup> Aortic Institute, Yale University School of Medicine, New Haven, USA

<sup>f</sup> Department of Biomedical Engineering, Yale University, New Haven, USA

<sup>g</sup> Department of Structural Engineering, Norwegian University of Science and Technology (NTNU), Trondheim, Norway

#### ARTICLE INFO

Keywords: Aortic dissection Microstructure Morphology Material characterization Computational modeling Computational modeling Computational mechanics Fluid-structure interaction Damage mechanics Hemodynamics Computational fluid dynamics Patient-specific simulation

# ABSTRACT

Aortic dissection continues to be responsible for significant morbidity and mortality, although recent advances in medical data assimilation and in experimental and in silico models have improved our understanding of the initiation and progression of the accumulation of blood within the aortic wall. Hence, there remains a pressing necessity for innovative and enhanced models to more accurately characterize the associated pathological changes. Early on, experimental models were employed to uncover mechanisms in aortic dissection, such as hemodynamic changes and alterations in wall microstructure, and to assess the efficacy of medical implants. While experimental models were once the only option available, more recently they are also being used to validate in silico models. Based on an improved understanding of the deteriorated microstructure of the aortic wall, numerous multiscale material models have been proposed in recent decades to study the state of stress in dissected aortas, including the changes associated with damage and failure. Furthermore, when integrated with accessible patient-derived medical data, in silico models prove to be an invaluable tool for identifying correlations between hemodynamics, wall stresses, or thrombus formation in the deteriorated aortic wall. They are also advantageous for model-guided design of medical implants with the aim of evaluating the deployment and migration of implants in patients. Nonetheless, the utility of in silico models depends largely on patient-derived medical data, such as chosen boundary conditions or tissue properties. In this review article, our objective is to provide a thorough summary of medical data elucidating the pathological alterations associated with this disease. Concurrently, we aim to assess experimental models, as well as multiscale material and patient data-informed in silico models, that investigate various aspects of aortic dissection. In conclusion, we present a discourse on future perspectives, encompassing aspects of disease modeling, numerical challenges, and clinical applications, with a particular focus on aortic dissection. The aspiration is to inspire future studies, deepen our comprehension of the disease, and ultimately shape clinical care and treatment decisions.

\* Corresponding author at: Institute of Biomechanics, Graz University of Technology, Austria. *E-mail address:* holzapfel@tugraz.at (G.A. Holzapfel).

#### https://doi.org/10.1016/j.pmatsci.2024.101363

Received 24 January 2022; Received in revised form 31 July 2024; Accepted 2 September 2024

Available online 12 September 2024

<sup>0079-6425/© 2024</sup> The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

# 1. Introduction

Patients often describe the onset of aortic dissection as the most severe pain a person can feel. Notably, this pain is frequently characterized by anatomically precise terms such as 'tearing' and 'ripping'. Medical textbooks have taught us the 'classic' nature of the constellation of symptoms and signs of acute aortic dissection. A classic case might be a 72-year-old male patient who has a long history of high blood pressure and suddenly experiences the onset of intense, sharp pain between his shoulder blades. Upon arrival at the hospital, the patient appears sweaty and agitated, his blood pressure is 195 mmHg/110 mmHg. Although many patients report symptoms such as chest, back, or abdominal pain, there is significant overlap with other acute syndromes and signs, making a precise diagnosis particularly difficult. In fact, most patients with aortic dissection do not have these classic symptoms, but only some of the well-known indicators. This has led to aortic dissection being referred to as 'the great masquerader' [1,2].

Aortic dissection is relatively uncommon, with incidences ranging from approximately three to eight cases per 100,000 individuals, yet responsible for significant morbidity and mortality [3–5]. This condition is characterized by delamination of the aortic wall, followed by blood inflow into the medial layer creating a new, false lumen, and eventually resulting in backflow into the actual, true lumen downstream. When left untreated, the mortality rate of acute ascending aortic dissection increases by 1 to 2% per hour within the first 24 hours of arriving at the hospital. As a result, mortality rates of up to 50 to 74% have been observed within the first two weeks [3,4]. Estimates suggest that up to 40% of cases are fatal prior to arrival at the hospital [6]. In contrast, an acute descending aortic dissection is generally less severe, with survival rates of up to 89% in medically treated patients after one month and 80% after five years [6].

Traditionally, aortic dissections involving the ascending aorta were treated with open surgery, while dissections beyond the aortic arch were managed through medical treatment [7,8]. However, over the past two decades, this approach has undergone a paradigm shift due to significant advances in stent-graft technology, a better understanding of the initiation and propagation of the disease, including hemodynamics in the true and false lumina, and improvements in medical treatment, resulting in fewer long-term dissection-related complications [9]. These advances are not solely due to new medical data but are also associated with the progression of *in silico* medicine, which describes the application of computational modeling and simulations to study, diagnose, treat, or prevent diseases, such as aortic dissection. Novel numerical tools enable detailed analyses of specific phenomena in healthy and dissected aortas without the need for costly and time-consuming experiments or invasive measurements. Moreover, these tools have demonstrated their ability to accelerate the development of new medical devices and enable treatment comparisons or studies on virtual cohorts.

Over the last decade, several review articles on aortic dissection have been published, providing various perspectives on what is arguably one of the most challenging and complex vascular diseases. An early review article on this topic was given by Humphrey [10]. Subsequently, Rajagopal et al. [11] published a review article to better understand the mechanics behind aortic dissection. Tsamis et al. [12] characterized the microstructure of the aorta and its relationship to aging and disease, with a particular emphasis on alterations in fibrous components. Sherifova and Holzapfel [13] examined pathological structural changes in aortic dissection, focusing on possible genetic causes. Peterss et al. [14] highlighted the differences in the acute and chronic phases of the disease, discussed changes related to morphology, and provided a comprehensive overview of the current state of research on these changes. Similarly, Spinelli et al. [15] provided an overview of current evidence on predicting dilatation and growth in aortic dissections, while Zilbert et al. [16] discussed the relationship between false lumen hemodynamics and aortic remodeling. In addition, several review articles summarized mechanical experiments and structural investigations on dissected tissue [17–19], as well as experimental models both *in vivo* and *ex vivo* [20], or solely *in vitro* [21]. These experiments and models are critical for the development of multiscale material and *in silico* models for the various pathological phenomena observed in aortic dissection. Finally, several review articles have discussed multiscale material and *in silico* models [19,22–25]. Some have even considered aortic dissection from both clinical and biomechanical perspectives [22,23,26–31]. Moreover, several other review articles focus on imaging techniques and their challenges [32–35], as well as the diagnosis and treatment of aortic dissections [36], among others, which lie beyond the scope of this work.

A common limitation of current studies is their narrow focus. The studies available today often focus on specific pathological aspects, experimental models, or modeling approaches. These may include the microstructure or mechanical behavior of the pathological aortic wall or hemodynamics in patient-specific studies. The limited scope of these studies prevents a more comprehensive understanding of aortic dissection, which is crucial for comprehending the initiation and progression of aortic dissection and for improving treatment and management strategies. Our objective in this review article is to provide a detailed summary of two main aspects. First, this review article provides a detailed presentation of the available medical data describing the pathological changes in aortic dissection, including remodeled anatomy and vessel walls, changed hemodynamics, and potential thrombus formation. These medical data are fundamental for the development and validation of physics-based computational or in silico models. It is important to note that the accuracy of in silico models heavily depends on the available medical data, such as patient-specific geometries, blood inflow and outflow conditions, and information about mechanical behavior and aortic microstructure. Without such data, the obtained results may be subject to significant uncertainty, affecting their reliability and applicability for clinical decision-making. Therefore, advances in disease understanding and the development of in silico models would be unthinkable without the significant impact of modern medical imaging technology and associated data. Second, this review article summarizes experimental, multiscale material, and patient data-informed in silico models that have advanced our understanding of the mechanisms underlying aortic dissection, as well as patient management. These models also underscore the application of the medical and experimental data utilized in their development. Specifically, it is important to emphasize the methods used, applications, and key findings to help the reader understand the available literature and the current state of the art. In conclusion, we offer a discussion on future perspectives, particularly focusing on aortic dissection. This includes considerations of disease-related modeling aspects, numerical challenges, and clinical applications,

all aimed at motivating targeted future research in biomedical engineering and healthcare. Ultimately, these efforts are intended to enhance clinical decision-making and advance personalized medicine.

The structure of this review article is as follows. Section 2 provides background information about aortic dissection, followed by an overview of pathological changes in Section 3, including remodeled anatomy, remodeled wall microstructure and mechanics, changed hemodynamics, and the role of thrombus formation. Section 4 then discusses the available experimental models, followed by an examination of multiscale material models, with emphasis on the damage and failure of the dissected aorta in Section 5. The next three sections discuss *in silico* models that explore various phenomena related to aortic dissection using patient-derived medical data. This encompasses *in silico* hemodynamic models in Section 6, coupled *in silico* models on the interaction between fluid and solid in Section 7, and, subsequently, *in silico* models on selected topics such as wall stress analysis in patient-specific studies, stent-graft deployment and migration, and thrombus formation in Section 8. Finally, Section 9 provides a summary and discusses future perspectives.

# 2. A basic introduction to aortic dissection

Frank Nicholls is believed to have been the first person to describe type A dissection more than 250 years ago, in 1760 [37]. As the personal physician to George II, King of England, he witnessed the King's death due to sustained fatal cardiac tamponade caused by an acute type A dissection that had ruptured into the pericardial sac. In 1819, about 60 years later, René Laennec was probably the first to introduce the term 'dissecting aneurysm'. Laennec did not know at the time that this term would unfortunately lead to ongoing confusion about aortic dissection and thoracic aortic aneurysm. This term is now considered incorrect because dissection without the presence of an aneurysm tends to be more common. The more accurate term 'aortic dissection', proposed by Jean Pierre Maunoir in 1802, has been largely overlooked due to Laennec's greater prominence [38].

A first milestone was achieved in 1954 with the successful surgical resection of a dissecting thoracic aortic aneurysm by the renowned surgeons Michael E. DeBakey, Denton A. Cooley, and Oscar Creech [39]. Subsequent advances in cardiopulmonary bypass and adjunctive techniques for circulatory management opened the door to surgical treatment of ascending aortic dissection. This led to several further milestones in the early and mid-1960s: Spencer and his colleagues at New York University first reported surgical treatment of chronic ascending aortic dissection with aortic insufficiency in 1962 [40]; and Morris and his colleagues in Houston reported the first successful treatment of acute aortic dissection with aortic insufficiency in 1963 [41,42]. Finally, pharmacological treatment of descending aortic dissection was introduced by Wheat and his colleagues in 1965 [43], based on control of systemic arterial pressure and 'global' rate of the left ventricular ejection dp/dt. As early as 1980, DeBakey and his colleagues demonstrated significant clinical and surgical expertise in the management of aortic dissection patients, as evidenced by their 20-year follow-up study of 527 patients who had received surgical treatment [44]. Remarkably, DeBakey himself underwent and survived open surgery for a type A dissection at age 97 [45].

Decades later, our modern understanding and treatment of aortic dissection have been shaped by two significant developments. The first was the establishment of the International Registry of Acute Aortic Dissection (IRAD) in 1996, whose numerous contributions have significantly advanced our knowledge and understanding of aortic dissection over the last decades. The second pivotal event was the publication of two groundbreaking papers in the 1999 issue of the New England Journal of Medicine, which for the first time detailed endovascular stent-graft intervention for acute type B dissection [7,46]. These articles marked the beginning of the endovascular era in the management of aortic dissections.

This fascinating early history of aortic dissection was partly extracted from the documentation by Criado [47] and by Rajagopal et al. [11]. The following section provides a basic introduction to the biomechanics of the normal aorta. This is followed by a concise overview of current guidelines on classification and definition, patient treatment and imaging, and recognized associated aortopathies.

# 2.1. Normal aortic biomechanics

Anatomy. The aorta, the largest and most important artery in the human body, is responsible for transporting oxygen-rich blood from the heart to the distal vessels that distribute it throughout the body and for returning the blood to the heart muscle [48]. As illustrated in Figs. 1(a) and (b), this crucial vessel can be divided into five main anatomical segments, following the most common anatomical description among the numerous segments available.

The first section of the aorta, called the aortic root, arises in the left ventricle, includes the aortic valve and the sinuses of Valsalva, and extends to the sinotubular junction. The ascending thoracic aorta, the second section, is approximately 5 cm long and extends from the sinotubular junction to the innominate or brachiocephalic artery. It then forms the third section, the aortic arch, also referred to as the transverse aorta. The arch stretches from the innominate to the left subclavian artery and is approximately 4 to 5 cm long. First, it ascends diagonally to the left over the anterior portion of the trachea, then descends to the left of the fourth thoracic vertebral body, continuing as the descending segment of the thoracic aorta. There are numerous variations in the anatomy of the thoracic aortic arch, typically involving branching patterns of the large vessels and the location of the descending thoracic aorta in relation to the spine [51]. The most common configuration is the left aortic arch, from which the brachiocephalic artery, the left common carotid artery, and the left subclavian artery branch off. The brachiocephalic artery is further subdivided into the right common carotid artery and the subclavian artery.

Following the arch, the aorta proceeds as a descending thoracic aorta, the fourth section, which extends from the left subclavian artery to the diaphragm and is about 20 cm long. It arises at the level of the intervertebral disc between the fourth and fifth thoracic vertebrae, penetrates the diaphragm, and ends anteriorly between the twelfth thoracic and first lumbar vertebrae. The descending thoracic aorta is initially located on the left side of the spine; however, at the level of the seventh thoracic vertebra, it shifts slightly to



Fig. 1. (a) Schematic representation of the aortic anatomy and its main branch vessels. (b) Computed tomography angiography images of the aorta and peripheral vasculature. Figure (a) is adapted from Shen et al. [49] with permission from Elsevier, while (b) is from Patel et al. [50] (licensed under CC-BY 4.0).

the right, anterior to the vertebral bodies. It is worth noting a small anatomical feature located on the proximal descending aorta, just distal to the left subclavian artery: the ligamentum arteriosum. This remnant is a leftover from the ductus arteriosus, a structure that connects the fetal pulmonary circulation to the aorta. The descending thoracic aorta gives rise to several larger vessels as well as a number of smaller but clinically important arteries. These vessels supply the bronchi, esophagus, intercostal muscles, and spinal cord. The intercostal arteries arise from the posterolateral aspect of the thoracic aorta and extend from the level of the third to the twelfth vertebral bodies.

Finally, the abdominal aorta extends from the diaphragm to the aortic bifurcation and branches into numerous smaller arteries that supply blood to the abdominal organs and lower limbs. Sometimes it is divided into the superior and inferior segments, which are delineated from the aorta by the branching of the renal arteries. From its origin at the aortic valve to the aortic bifurcation in the abdomen, the size of the aorta decreases [52,53]. Directly above the coronary sinuses, the diameter of the ascending aorta is typically between 2.5 and 3.5 cm. The arch and descending thoracic aorta are often only slightly narrower than the ascending aorta. This tapering is particularly important for the so-called 'Windkessel effect'.

*Windkessel effect.* The heart continuously pumps oxygen and nutrient-rich blood throughout the body to sustain life. To achieve this, the heart generates pressure and flow waves in the arteries. As the heart contracts, the pressure in the ventricle rises rapidly at the beginning of systole and soon exceeds the pressure in the aorta, causing the aortic valve to open, blood to be ejected, and aortic pressure to rise. While ventricular pressure falls rapidly as the heart muscle relaxes, aortic pressure falls more slowly due to the elastic properties of the aorta. The aorta and some proximal large vessels act as an elastic buffer chamber behind the heart, as explained by the Windkessel theory [54–56]. In this way, they serve as a reservoir for the stroke volume ejected by the left ventricle during systole. During diastole, the elastic properties of the aortic wall propel this volume forward into the peripheral arteries. This systolic-diastolic interaction, which represents the Windkessel effect, not only aids perfusion of peripheral arteries, but also benefits the heart by reducing left ventricular afterload and improving coronary blood flow and ventricular relaxation.

As one moves away from the aortic valve, the pressure wave changes, in particular there is a sharpening and an increase in amplitude pressure, while the so-called dicrotic notch gradually decreases. This elevation in systolic pressure as it travels through the circulatory system is primarily due to wave reflections, see Fig. 2(a). Wave reflections occur in the arterial system at locations where there is a discontinuity in artery size, elasticity, or a mismatch in vascular impedance. Common reflection sites include branching



**Fig. 2.** (a) Representative curves illustrating the principles of arterial wave reflection theory and the effects on pressure P and flow Q waveforms caused by reflected waves. (b) Selected non-invasive medical imaging methods (phase contrast-magnetic resonance imaging (MRI) and ultrasound imaging) to obtain cross-sectional area or diameter waveforms and flow or mean velocity waveforms of a patient's artery. Figure (a) is adapted from Pagoulatou et al. [57] (licensed under CC-BY 4.0), while (b) is from Mynard et al. [58] (licensed under CC-BY 4.0).

points, areas where diameter and elasticity vary, and high-resistance arterioles, which are significant reflection points. Conversely, for a constant flow in a rigid tube, the pressure would decrease along the flow direction unless there is deceleration. Although the amplitude of the pressure between systole and diastole nearly doubles due to reflections, resulting in a reduced pressure load on the left ventricle, the average pressure over the cardiac cycle decreases with increasing distance.

The speed at which arterial pressure propagates through the circulatory system is called pulse wave velocity and is due to vascular compliance, i.e. the ability of the vessel to expand under the forces exerted by blood pressure. It is typically calculated by dividing the distance between two points in the recording arteries by the pressure wave transit time. This measurement is a standard indicator of central arterial stiffness [59–63]. Higher pulse wave velocity values indicate stiffer arteries, which may be associated with an increased risk of cardiovascular events such as heart attack and stroke. Conversely, lower pulse wave velocity values suggest more elastic and healthier arteries. In addition, increased arterial stiffness due to aging or pathophysiological conditions can lead to decreased distensibility of the aorta and other vessels, resulting in elevated intraluminal pressure.

These characteristic pressure and wave waveforms can be measured using both invasive and non-invasive techniques. While the gold standard for acquiring the blood pressure waveform is invasive measurement, such as with a micromanometer-tipped catheter, various surrogates are often used [64]. These surrogates measure distension (arterial diameter, area, or volume change) to approximate the pressure waveform. However, it is important to note that the relationship between pressure and distension is not strictly linear and exhibits hysteresis due to the viscoelastic properties of the vessel walls. Despite this, and due to their limited spatial and temporal resolution, ultrasound imaging and magnetic resonance imaging (MRI) are widely available and provide acceptable accuracy [58]. Fig. 2(b) shows representative results of MRI, illustrating the changes in cross-sectional area and flow information as well as edge tracking of the artery wall in B-mode and M-mode, and Doppler ultrasound for mean velocity measurements. An introduction to the main imaging modalities used can be found in Section 2.3.1.

*Wall microstructure.* The normal aortic wall is characterized by three primary layers: intima, media, and adventitia, as depicted in the schematic image in Fig. 3(a) and the histological images in Figs. 4(a) and (c) [65]. Each layer exhibits a unique structure and composition required for its particular purpose and function. Apart from that, the intima and media are separated by the internal elastic lamina, while the external elastic lamina separates the media from the adventitia.

The intimal layer consists of a single layer of endothelial cells [65]. For many years, this cell layer was considered relatively inert, a mere physical barrier between the circulating blood and the underlying tissue. However, it is now known that endothelial cells are metabolically active [67,68]. They form a semi-permeable barrier that controls the passage of materials and white blood cells into and out of the bloodstream, while also allowing the selective transfer of nutrients, hormones, and waste products between the bloodstream and tissues. They play a crucial role in vascular homeostasis, which involves the regulation of blood flow and vascular tone through the release of various vasoactive factors that cause vasodilation or vasoconstriction. These may include vasodilatory factors such as nitric oxide (NO), an endothelium-dependent vasodilator of the underlying smooth muscle. In addition, they balance blood clotting, mediate inflammation and immune responses, contribute to the metabolism of lipids and other substances, and maintain vascular integrity by remodeling. Typically, endothelial cells are flat and elongated in the direction of blood flow and are subject to shear stresses caused by blood flow, commonly referred to as wall shear stresses (WSSs). WSSs are tangential or frictional forces exerted by blood flow on the inner surfaces of blood vessels. Endothelial cells can convert these mechanical stimuli into intracellular signals that result into the production of multiple biomolecules, some of which diffuse and affect smooth muscle cells (SMCs) located in the media, impairing their function [52].



**Fig. 3.** Illustration of the microstructure of the aortic wall showing (a) the composition of a healthy yet aged aortic wall exhibiting non-atherosclerotic intimal thickening. This structure consists of three layers, namely the intima ('I'), media ('M'), and adventitia ('A'). (b) Layered collagen architecture of a healthy, aged abdominal aorta, where the top image shows an out-of-plane structure in the circumferential-radial plane. The three lower images present in-plane sections of the intima ('I'), media ('M'), and adventitia ('A'), with white scale bars corresponding to  $100 \mu$ m. (c) 3D microstructure of an aortic media composed of multiple lamellar units. These include circumferentially-oriented, radially-tilted smooth muscle cells ('SMCs') featuring elliptical nuclei ('N'), located between elastic lamellae ('EL'). These are surrounded by a dense network of interlamellar elastin fibers ('IEFs', indicated by black arrows), elastin struts ('ES'), and reinforced elastin pores ('EP'). (d) Schematic depiction of two SMCs and two fenestrated elastic lamellae with their interconnections. Specifically, collagen fibers ('Coll') are closely associated with elastic lamellae. The left SMC's surface ridges are connected to both elastic lamellae via elastin protrusions, while the right SMC is connected to the lower elastic lamellar via an oxytalan fiber (Ox). Larger deposits ('D') containing collagen and heparan sulfate proteoglycan can be found at indentations in the cell surface. Figure is reprinted from Sherifova and Holzapfel [18] with permission from Elsevier.

Adjacent to this monolayer is a thin basal lamina with a thickness of about 80 nm [52], which consists of net-like type IV collagen and adhesion molecules such as laminin. The endothelium not only senses shear stresses, it also plays a key role in hemostasis and also regulates the transport of substances such as nutrients, leukocytes, or lipids into the wall, thereby playing a critical role in the proper functioning of the aorta. In fact, endothelial dysfunction is often associated with the initiation and progression of cardiovascular diseases, including aortic dissection. Moreover, the layer only becomes mechanically significant with age-related changes, when collagen fibers are deposited, resulting in non-atherosclerotic intimal thickening [69].

In contrast, the media consists of many lamellar units, also known as musculo-elastic fascicles [70], with layers of elastic fibers separated by SMCs, and collagen fibers (primarily types III and V), as depicted in Fig. 3 and Fig. 4. These components are all embedded in the ground substance, which primarily consists of water and glycosaminoglycans (GAGs)/proteoglycans (PGs) (mainly versican). This complex network is commonly referred to as the extracellular matrix (ECM), as shown in Fig. 3. Interestingly, the thickness of a single lamellar unit remains the same across mammalian species [71,72], with an average thickness of 13.9 $\pm$ 1.2 µm [73]. Consequently, the thickness of the thoracic aortic media increases proportionally with the number of lamellar units. Overall, the media has 53 to 78 lamellar units in the proximal human aorta, with fewer units observed distally as the aortic diameter tapers [52]. The lamellae appear wavy in most collapsed specimens, but almost straight in distended specimens.



**Fig. 4.** Histological images of the normal aortic wall of a young adult showing (a) a transverse the three layers of an aorta: intima, media, and adventitia from top to bottom (50x, H&E stain). (b) Image of the elastic fibers of the intima, a distinctly paler layer than the media, and the numerous lamellar units of the media delineated by black elastic laminae. There is a marked change at the media-adventitia boundary, with the adventitia comprised mostly of loose fibrous tissue (yellow). The normally thick vasa vasorum can be clearly seen (50x, Movat's pentachrome stain). (c) Higher magnification of the lamellar units of the media with pronounced eosinophilic, refringent elastic laminae (500x, H&E stain). (d) Close-up view of the lamellar units. A single lamellar unit is defined as the components sandwiched between two elastic laminae: (i) elastic lamina, (ii) extracellular matrix (fibrous tissue (yellow) and mucopolysaccharides (green/blue)), (iii) SMCs (red cytoplasm and dark blue/burgundy nuclei), (iv) extracellular matrix, and (v) another elastic lamina (500x, Movat's pentachrome stain). Figure is reprinted from Halushka et al. [66] with permission from Elsevier.

The primary component of the lamellar unit is elastic fibers, which consist mainly of elastin (~90%) and multiple elastin-associated glycoproteins (~10%) such as the fibrillins and fibulins. In contrast to collagen fibers, which undergo constant turnover in the arterial wall [74], elastic fibers are mainly deposited during the perinatal period and, due to their extended half-life, endure millions of loading cycles over the course of an individual's life [75]. These elastic fibers exist in three forms: lamellae, interlamellar fibers, and interlamellar struts. Lamellae are thick, continuous sheets with a fibrous surface, numerous rounded fenestrations and a thickness of 2.1  $\pm 0.6 \,\mu\text{m}$  [73,76]. Gross irregularities are also observed, with the lamellae splitting and branching [77]. The intralamellar space contains dense, intricately organized interlamellar elastic fibers, which are cords or wisps of elastin and elastin-associated microfibrils that protrude obliquely from the top to the bottom of lamellar surfaces. They can form a cage-like structure surrounding and abutting SMCs, and are important for mechanosensing by the SMCs [78]. In contrast to less dense interlamellar elastic fibers, interlamellar struts are more substantial and form connections between adjacent lamellae and separate SMCs. Due to their considerable thickness and mainly radial orientation, they can play a critical role in supporting radial loads, including those originating from Gibbs-Donnan swelling pressures emanating from the aggregating medial PGs [79], which likely contributes to SMC mechanosensation.

Elliptically shaped SMCs are arranged between lamellae with their long axis aligned nearly circumferentially with a radial tilt of 19  $\pm$  3° [73], the details of which remain controversial. SMCs are also indirectly connected to the lamellae via conspicuous oxytalan and elaunin fibers to sense mechanical stress. Cyclic physiological tensile stresses tend to promote what is known as a contractile phenotype of SMCs, which under normal conditions maintains the normal ECM based on actomyosin-mediated mechanosensing and slow turnover (synthesis and degradation). In contrast, non-physiological stresses can lead to a synthetic SMC phenotype characterized by increased production of matrix metalloproteinase (MMP) and increased ECM synthesis. Depending on the relative intensity of these

two processes, fibrosis or atrophy may ultimately occur. Finally, SMC apoptosis tends to increase when the medial layer is stressshielded, and SMC anoikis occurs when the cells lose their connection to the ECM (a cell survival signal; [80]).

Collagen fibers, densely packed in the media and typically of type III, appear as interspersed fibers between elastic lamellae, either as fiber bundles, thin bundles, or individual fibers. These fiber bundles run parallel within each layer, implying greater packing efficiency and allowing more fibers to be packed into a smaller volume, which is advantageous for high loading conditions. Additionally, these fibers are helically woven around the vessel but are primarily circumferentially oriented [70,81], similar to other primary mural constituents, such as interlamellar elastic fibers or SMCs. Because of this alignment, they do not exhibit any radial tilt, which remains controversial. Collagen bundles are independent of elastin and SMCs and have no direct connections. However, fiber bundles within each layer tightly envelop the SMCs and may contribute to a pericellular coat. The fiber bundles often twist under mean physiological pressure. As pressure increases, more and more bundles are recruited and the fibers straighten. In addition to the elastic lamellae, collagen fibers serve as the mechanical basis of the media. More specifically, collagen fibers provide the aorta with high tensile strength, ensuring its structural integrity, while elastin gives the aortic wall its ability to expand and recoil in response to blood pressure, a phenomenon known as the Windkessel effect [82], as previously introduced. Due to the greater distensibility of elastin compared with collagen, this dual stress distribution is responsible for the characteristic nonlinear stress–strain behavior observed in soft tissues during *ex vivo* testing [83].

Finally, the adventitia consists mainly of a dense network of type I collagen fibers, along with scattered fibroblasts and tissueresident macrophages, as well as sparse elastin, nerves, and vasa vasorum, which supply nutrients and oxygen to the aortic wall, as shown by the histological images in Figs. 4(a) and (c). The process of collagen I fibrillogenesis is supported by type III collagen as well as the small leucin-rich PGs decorin and biglycan. The importance of these accessory molecules is highlighted well by biglycan deficiency, which is associated with aortic dissection [84]. Normal collagen fibers in the adventitia appear undulated at physiological pressure but straighten at higher pressure, leading to the hypothesis that they serve as a protective sheath. In other words, the adventitia functions to prevent over-distension and vascular rupture due to acute pressure increases [65]. In contrast to the collagen fibers in the media, the collagen fibers in the adventitia are oriented more randomly, with their mean direction coinciding with the axial direction of the vessel [85].

The primary constituents of the ground substance found in each of the three tissue layers are water and GAGs/PGs, with varying compositions and organizations across the tissue depth [86]. GAGs, negatively charged polysaccharides, comprise repeating disaccharide units and are classified into six families: hyaluronic acid, dermatan sulfate, keratan sulfate, chondroitin sulfate, heparan sulfate, and heparin [87]. With the exception of hyaluronic acid, GAGs are covalently linked to protein cores, forming larger structures, the PGs, which are categorized based on cellular location, protein homology, and protein modules within their cores [88]. The ECM harbors both small and large PGs, with small PGs encompassing decorin, biglycan, and fibromodulin, and large PGs incorporating aggregan, versican, neucoran, and brevican [89]. The aortic wall contains hyaluronan and three types of sulfated GAGs: chondroitin sulfate, heparan sulfate, and dermatan sulfate [90,91]. In the media and adventitia, varying proportions of these GAGs are found, exhibiting a non-homogeneous distribution across the wall, with higher concentrations toward the media and intima [92]. Twenty different PG types have been identified within the ECM of the human thoracic aorta [93]. Versican, a large PG, is the primary component, accounting for 53.5% of total PGs and predominantly residing in the intima [94,95]. In contrast, small PGs such as biglycan and decorin account for 40% of the total PGs in the wall [94,95]. Biglycan is localized in the intima, outer media, and adventitia, while decorin is restricted to the adventitia in normal aortas [96]. In general, most PGs are found in the media, where they can either exist within the interstitial space or associate with elastin and collagen [97]. Although GAGs/PGs account for merely 2 to 5% of the thoracic aortic wall [98], they may play a significant role in the mechanobiological functions of the aortic wall due to their swelling capacity [92,99]. As negatively charged molecules, GAGs maintain electroneutrality by drawing in cations, which results in water influx and the generation of osmotic pressure in GAG-containing regions, a phenomenon known as Gibbs-Donnan swelling [100,101]. Swelling distribution across aortic thickness can vary [99], and establishing a correlation between transmural GAGs/PGs and swelling distributions is essential for better understanding residual stresses. Uniform swelling may put tension on the wall network [102], while localized swelling can produce elevated pressures, potentially leading to delamination and dissection in the aortic wall [103].

Residual stresses in the aorta are defined by the internal, self-equilibrating stresses that exist in the absence of external forces or pressure [86]. These stresses likely arise during vascular development, as revealed by changes in the intricate structure and composition of the aortic wall [104]. Residual stresses play a critical role in preserving the integrity and adaptability of the wall to changes in blood pressure and other physiological and pathological influences and contribute to load distribution across the aortic wall. This load distribution appears to be mechanobiologically favorable; it also minimizes stress concentrations and helps prevent damage. Residual stresses can be accounted for computationally via multiple approaches [52], including constrained mixture-based material models [105] and classic continuum mechanics models [106].

In conclusion, the aortic wall has an impressive microstructure of various interwoven constituents, resulting in an extremely complex architecture. Even minor deviations from this delicate structure can cause significant alterations in integrity and mechanical behavior, potentially leading to the onset of diseases such as aortic dissection. Although intense research efforts have already uncovered some peculiarities of arterial tissue and its composition, many questions remain unanswered to date.

*Modeling aortic mechanics*. Aortic tissues pose significant challenges for modelers because they are complex mixtures of fluids, such as water; fibers such as collagen and elastin; cells; and a myriad of connective tissue proteins [107]. Over the course of the cardiac cycle, the aortic wall undergoes motion and deformation in response to various applied forces or loads, which are independent variables, while motions and deformations are dependent variables. Blood pressure acts perpendicularly and into the aortic intima, while

blood flow along the aorta induces shear forces parallel to the intimal surface. These forces can be described, for instance, by the Cauchy stress tensor, which describes the state of stress at a given point within a material in the deformed configuration. This mathematical object has three normal components and six shear components. In fact, due to the symmetry of the stress tensor, only three of these shear components are independent. Additional complexity arises from the unique geometry of the aorta and its movement due to heart contractions, resulting in axial deformation and torsion. In contrast to intimal surface forces, aortic tethering to adjunct organs results in adventitial surface forces. Taken together, these factors culminate in a fully 3D stress state in the aortic wall.

To establish a relationship between the occurring stresses and deformations, constitutive relations can be formulated, typically based on continuity equations, although aortic tissue may not adhere to these assumptions due to its multiphasic and multicomponent nature. The most common approaches include classic continuum models that act on the organ scale and homogenize the aortic microstructure, multiscale models that incorporate detailed microscale knowledge by what is known as a representative volume element and allow for correlations between micro and macro behavior, and constrained mixture models that interpret biological tissue as a mixture of multiple constituents. For more details, we refer readers to [108-110]. While all constitutive relations have their justification, they always require a detailed understanding of the material composition. As previously demonstrated, this proves to be difficult because the aortic wall is a complex, layered structure composed of a heterogeneous mixture of various solid and fluid components. Additionally, the aortic wall is anisotropic, with varying degrees of anisotropy depending on location, and demonstrates a location-dependent viscoelastic response that has a positive effect on smoothing the pulsatile nature of flow, which is beneficial for perfusion [111–113]. While it is generally assumed to be incompressible, some studies define it as slightly compressible [114]. Most of its constituents exhibit nonlinear responses, and it is also essential to consider both the active and the passive mechanical behavior of the aorta. These material properties of aortic tissue are typically obtained by biopsying small pieces of tissue and measuring their characteristics. These measurements are then extrapolated to represent the entire aorta. Microstructural information can be obtained using a range of techniques, including histology, optical imaging methods such as second harmonic generation microscopy [115], Xray transmission imaging methods such as micro-computed tomography (CT) [116], and non-invasive medical imaging techniques such as diffusion MRI [117,118], among others. Remarkably, only a few studies take into account the heterogeneous nature of the aortic wall – that is, the fact that each material point behaves differently – despite its critical importance.

In addition to that, a particular challenge lies in what is known as the initial stress or strain within the aorta. This is especially relevant for blood vessels like the aorta, as it is not *a priori* clear what the stress-free or strain-free reference configuration is – a baseline from which to measure deformation. In the past, and mainly based on Fung and his colleagues [119,120], researchers have assumed that the stress-free state of the aorta could be determined by making repeated incisions in the tissue. However, there is no certainty that all stresses have been completely relieved after a finite number of cuts. Additionally, while such incisions may lead to a global configuration, they do not guarantee that every local point in the body under study is stress free. Although this approach is a good approximation, the *in vivo* situation might be more complex.

The complexity of understanding aortic behavior lies not only in the aortic tissue itself but also in the intricate nature of blood, which presents its own unique challenges. Blood is a composite fluid made of gel-like cellular matter – including erythrocytes, leukocytes, and platelets – suspended in a plasma solution in which numerous proteins and various ions are dissolved in water as a solvent, adding another level of complexity. While the behavior of plasma can often be approximated using the Navier–Stokes constitutive theory, whole blood resists such simple modeling. For example, its behavior in the vasa vasorum or other small arteries cannot be captured by Navier–Stokes equations because blood shears significantly [121,122] and stress relaxes [123]. This shear-thinning property, resulting from the disassembly of erythrocyte rouleaux aggregates at high shear rates, should be accounted for in any comprehensive model.

Similar to other arteries, accurate mathematical modeling of the aorta presents particular challenges and becomes even more complex in the event of disease. It becomes clear that, with some exceptions, constitutive relationships that treat the arterial wall as a purely elastic body and are based on Navier–Stokes fluid models for blood are generally insufficient to provide reliable modeling results. To conclude, mathematical models that are as simple as possible without compromising accuracy and precision should be used. The underlying mathematical and physical theories must be conceptually rigorous, while accurately modeling real-world phenomena [124]. Although this is easy to state, the biomechanical complexity of the aorta in both health and disease makes modeling challenging, and it is often still unclear how many and which simplifications are still valid.

# 2.2. Definition and classification systems

Aortic dissection is a complex and challenging medical emergency that presents a wide range of clinical features. Over the past several decades, experts from various disciplines have sought to define and classify this condition to enhance communication and understanding based on anatomy, acuity, and those with prognostic impact [125].

*Definition.* In addition to intramural hematoma and penetrating atherosclerotic ulcers, aortic dissection is the most common form of acute aortic syndrome [126]. These life-threatening conditions compromise the integrity of the aortic wall. Intramural hematoma is a contained aortic wall hematoma with bleeding within the media but without the development of an intimal tear. On the other hand, a penetrating atherosclerotic ulcer is an ulceration of an aortic atherosclerotic plaque that penetrates through the internal elastic lamina into the media. Interestingly, both intramural hematoma and aortic dissection primarily affect the media, while a penetrating atherosclerotic ulcer primarily affects the intima [127]. A schematic representation of these acute aortic syndromes, along with representative histology and CT images, is shown in Fig. 5.

M. Rolf-Pissarczyk et al.



Progress in Materials Science 150 (2025) 101363

**Fig. 5.** (a) Schematic overview of acute aortic syndromes and their main morphological characteristics, including (i) (classic) aortic dissection with true lumen ('TL') and false lumen ('FL'), (ii) penetrating atherosclerotic ulcer ('PAU'), and (iii) intramural hematoma ('IMH'). (b) Histological sections of the aorta (Masson's trichrome stain) showing (i) the ascending aorta in a (classic) aortic dissection with visible intima (blue, 'i'), media (red, 'm'), and false lumen ('FL'), where arrows indicate the separation of the media forming the dissection flap and a thrombus in the false lumen (asterisk); (ii) the descending aorta in the case of penetrating atherosclerotic ulcer ('PAU') features an ulcerated crater, which penetrates the media and is characterized by adventitia ('a'), intima ('i'), and media ('m'), with intramedial hemorrhage also detected (asterisk); and (iii) an intramural hematoma ('IMH') with the dissection plane within the outer aortic media (asterisks) showing layers including adventitia ('a'), intima ('i'), and a caute type A dissection with an intimal rupture and a dissection flap (here: dissecting membrane, 'DM'), creating a true lumen ('TL') and a false lumen ('FL'); (ii) a penetrating atherosclerotic ulcer in the descending aorta (red arrows) with a transmural lesion and a localized subadventitial hematoma; and (iii) an intramural hematoma in the ascending aorta (asterisk) with no visible lesion in the inner aortic layers. Figures (a) and (b) are adapted from Vilacosta et al. [128], while (c) is from Carrel et al. [129], with permission from Elsevier, respectively.

According to the classic definition, an aortic dissection occurs when an intimal tear permits blood to flow through the tear and into the aortic media. This process splits the wall and forms a dissection flap that separates the true lumen from a newly created false lumen [130–132]. The dissection flap can progress antegrade or retrograde, potentially leading to several life-threatening complications. These complications may include acute aortic valve regurgitation, myocardial ischemia, acute stroke, malperfusion syndromes, or complete aortic rupture, which can ultimately lead to cardiac tamponade if the dissection occurs in the proximal aorta. Blood flowing through the false lumen may rupture back into the true lumen and create additional connections or fenestrations, the term 'fenestra' originating from the Latin word for 'window'. If instead the blood in the false lumen ruptures through the outer media and adventitia, complete aortic rupture can occur.

In chronic aortic dissection, the development of an aortic pseudoaneurysm is not uncommon. This technical term refers to the dilatation of the false lumen. Because not all three layers of the aortic wall are affected, it is distinguished from a classic aneurysm. However, this terminology is rarely used clinically. Instead, 'aneurysmal degeneration of the false lumen' or simply 'false lumen dilatation' are more commonly employed for several primarily clinical reasons: typically only the entire aortic diameter is measured and used as a size criterion for surgical interventions. The clinical manifestation is more similar to that of a dissected aneurysmal aorta. Therefore, the term pseudoaneurysm alone might not be appropriate in many cases. To ensure consistency throughout this review



**Fig. 6.** (a) Stanford and DeBakey classification systems for aortic dissection. The Stanford classification divides aortic dissections into two groups: type A and type B, while the DeBakey classification divides aortic dissections into three groups: type I (including the ascending and descending aorta), type II (limited to the ascending aorta), and type III (limited to the descending aorta). (b) Society for Vascular Surgery and Society of Thoracic Surgeons aortic dissection classification system of dissection subtype by zone location of primary entry tear [134]. In the example illustrated, the aortic dissection has a primary intimal tear in zone 0, with the dissection process extending distally to zone 9. Therefore, the dissection is fully classified as type A<sub>9</sub>. Figure (a) is adapted from Tadros et al. [135], while (b) is from Vacirca et al. [136], with permission from Elsevier, respectively.

article, we aim to use commonly accepted clinical terminology.

Anatomical classification. Aortic dissections are traditionally classified according to the DeBakey and Stanford systems, as depicted in Fig. 6(a). The DeBakey classification system [133] consists of type I, type II, and type III, determined by the origin of the intimal tear and dissection extent. Type I arises in the ascending aorta, with the dissection propagating distally to encompass the aortic arch and usually the descending aorta. Type II is restricted to the ascending aorta, whereas type III originates distal to the subclavian artery in the descending aorta and typically extends distally. Type III dissections can be further subdivided into IIIa, which is limited to the descending thoracic aorta, and IIIb, which begins in the descending thoracic aorta and extends below the diaphragm.

The more common Stanford classification [137] simplifies the DeBakey system and divides aortic dissections into two categories based on ascending aorta involvement, regardless of the origin site. Type A includes all dissections involving the ascending aorta, irrespective of the location of the intimal tear. Type B encompasses dissections that exclude the ascending aorta, including even those that involve the aortic arch but exclude the ascending aorta.

In particular, no classification addresses aortic dissections that originate in the arch. In 2019, a third category to treat thoracic arch pathology termed 'non-A non-B dissection' was introduced for patients with arch involvement due to the most proximal tear or retrograde extension – in other words, the proximal dissection of the flap begins in the aortic arch [138]. Non-A non-B dissections are rare and among the most challenging cases for surgical or endovascular treatment [139].

In 2020, the Society for Vascular Surgery and the Society of Thoracic Surgeons introduced a comprehensive classification system that provides a more detailed perspective on aortic dissection anatomy [134], see Fig. 6(b). This approach characterizes dissections based on the location of primary intimal tears and the proximal and distal extent of the dissection. The updated classification scheme differentiates between type A and type B dissections based solely on the location of the primary intimal tear. This refinement was critical in accurately describing the involvement of the aortic arch in aortic dissection and in aligning with current treatment approaches. Use of this updated classification system enables clinicians to effectively envision the primary intimal tear location and the proximal and distal dissection extent through a single, concise descriptor.

There is also an alternative classification approach, still based on an anatomical system, from the European Society of Cardiology [4]. This approach comprises five classes and is derived from the initial description by Svensson et al. [140]. The classification system is based on the anatomic presentation of the dissection and includes: (i) classic aortic dissection with true and false lumina separated by

a dissection flap, (ii) intramural hematoma or hemorrhage without tear or flap imaged, (iii) subtle or discrete dissection, (iv) penetrating atherosclerotic ulcers, and (v) iatrogenic or traumatic dissection. While little is known about subtle or discrete dissections due to difficulties in imaging detection, they have been described as a partial stellate or linear tear in the vessel wall without a dissection flap, eccentric bulge, or intramural hematoma.

*Temporal classification.* Historically, aortic dissections have been classified into two groups based on the time of symptom onset: acute, defined as less than 14 days from symptom onset, and chronic, defined as more than 14 days from symptom onset. This 14-day cutoff was originally established due to the prevalence of dissection-related complications during this period and was derived in part from the work of Hirst et al. [141] in the late 1950s. Since then, however, there have been significant advances in imaging techniques and treatment options. The IRAD proposed a more nuanced classification system that includes four distinct time periods: hyperacute (symptom onset to 24 hours), acute (2 to 7 days), subacute (8 to 30 days), and chronic (over 30 days) [142]. This updated classification better reflects the different phases of acute aortic syndromes and associated mortality rates.

The most contemporary temporal classification system, proposed by the Society for Vascular Surgery and the Society of Thoracic Surgeons, similarly divides aortic dissection into four temporal types: the hyperacute phase (first 24 h) within the acute phase (up to 14 days after onset), followed by the subacute (15 to 90 days after onset) and the chronic phase (> 90 days after onset). This refined classification system helps improve prognosis and guide decision-making about the timing and types of potential interventions in aortic dissection patients [34,126,134,143]. From a prognostic perspective, acute aortic dissection involving the ascending aorta typically has a poor prognosis. However, chronic dissections tend to have a relatively mild natural course compared with the acute phase [144].

*Prognostic classification.* In contrast, prognostic classification systems offer clinicians a useful method for predicting the likely outcome or course of a condition. Perhaps the most common differentiation is to distinguish between complicated and uncomplicated dissections, irrespective of the time elapsed since the onset of dissection and the proximity of the dissected segments of the aorta. This is particularly useful for type B dissection [145]. Complicated type B dissections are characterized by the presence of a malperfusion syndrome, potentially leading to visceral, renal, or extremity ischemia. Other distinguishing characteristics include rupture, uncontrolled hypertension, persistent abdominal or chest pain, or signs of rapid expansion, which can be detected through medical imaging. The management of complicated acute type B dissections can be particularly challenging, as discussed later. In contrast, uncomplicated dissections are typically managed only through aggressive blood pressure control.

#### 2.3. Imaging and treatment of patients

Imaging techniques and management strategies for aortic dissection patients play a crucial role in ensuring accurate diagnosis, appropriate risk stratification, and effective treatment. Advanced imaging techniques not only enhance our understanding of the remodeled anatomy and altered hemodynamics but also facilitate the development of *in silico* models, as we will demonstrate in Sections 6 to 8. A synopsis of the strengths and limitations of the subsequently discussed imaging modalities and their variants for clinical use and research is given in Table 1.

#### 2.3.1. Imaging modalities and variants for clinical use and research

A wide variety of imaging techniques are available for anatomic and functional assessment of the aorta, both for routine clinical use and for research applications [34,126]. The main diagnostic imaging modalities are CT imaging, MRI, and ultrasound techniques. It is important to be familiar with the basic principles of these technologies to understand their strengths, their limitations, and how the variety of modern imaging applications based on these technologies can be used to assess cardiovascular diseases, such as aortic dissection, and to inform computational models.

Angiography. Classic angiography, a traditional invasive radiologic technique for imaging blood vessels, is based on the direct injection of an iodinated contrast agent into the bloodstream, either through a hollow needle or, nowadays, through a catheter inserted into the femoral artery in the groin. The contrast-opacified vessels become then visible in an X-ray image on a radiographic film, which is referred to as an angiogram. However, catheter-based angiography is no longer used for diagnostic purposes in patients with aortic disease and has been completely replaced by non-invasive imaging techniques, particularly CT angiography (CTA). Today, angiography is routinely used in catheter-based interventions such as minimally invasive procedures, percutaneous fenestrations, and other endovascular procedures.

*Computed tomography imaging.* CT is an X-ray-based imaging technique commonly used for diagnostics, providing 24/7 availability [146–148]. The produced raw data is reconstructed into thin transverse cross-sectional slices, with these images typically measuring  $512 \times 512$  pixel and having a section thickness as thin as 0.5 mm. A full-body CT scan, including the chest, abdomen, and pelvis, has between 900 and 1500 slices, and its acquisition time is in the order of a few seconds. Disadvantages of CT include ionizing radiation and low contrast resolution for soft tissues. With a few exceptions, all cardiovascular CT applications require the use of an intravenously injected contrast medium to enhance the X-ray attenuation of the blood and enable visualization of the heart and blood vessels.

#### Table 1

A synopsis of the strengths and limitations of imaging modalities and their variants for clinical use and research is provided, where performance is indicated as high (+), medium (•), or low (-) (CT: Computed tomography; CTA: Computed tomography angiography; ECG: Electrocardiogram; PCCT: Photon-counting CT; MRI: Magnetic resonance imaging; PC: Phase-contrast; VENC: Velocity encoding; US: Ultrasound; TTE: Transthoracic ECG; TEE: Transesophageal ECG).

| Parameter            | Angiography  | CT | CTA | ECG-gated CTA | 4D CTA | PCCT | MRI | Cine MRI | MRA | 2D PC-MRI | 4D PC-MRI / 4D-flow MRI | (Dual) VENC 4D PC-MRI | Doppler US | TTE | TEE | Intravascular US |
|----------------------|--------------|----|-----|---------------|--------|------|-----|----------|-----|-----------|-------------------------|-----------------------|------------|-----|-----|------------------|
| Portable             |              |    |     |               |        |      |     |          |     |           |                         |                       | 1          | 1   | 1   |                  |
| Invasive             | 1            |    |     |               |        |      |     |          |     |           |                         |                       |            |     | 1   | 1                |
| Radiation            |              | 1  | 1   | 1             | 1      | 1    |     |          |     |           |                         |                       |            |     |     |                  |
| Flow sensitive       |              |    |     |               |        |      |     |          |     | 1         | 1                       | 1                     | 1          |     |     |                  |
| 3D data set          |              | 1  | 1   | 1             | 1      | 1    |     |          |     |           | 1                       | 1                     |            |     |     |                  |
| Contrast-based       | $\checkmark$ |    | 1   | 1             | 1      | 1    |     |          | 1   |           |                         |                       |            |     |     |                  |
| Availability         | -            | +  | +   | +             | +      | 0    | 0   | 0        | o   | -         | -                       | -                     | +          | +   | 0   | 0                |
| Speed of acquisition | -            | +  | +   | +             | +      | +    | -   | -        | -   | -         | -                       | -                     | 0          | 0   | 0   | -                |
| Spatial resolution   | -            | +  | +   | +             | +      | +    | 0   | 0        | 0   | 0         | 0                       | 0                     | +          | +   | +   | +                |
| Temporal resolution  | -            | -  | -   | -             | -      | -    | +   | +        | +   | +         | +                       | +                     | +          | +   | +   | +                |

By far the most commonly used imaging technology to assess aortic disease is CTA, a CT scanning technique optimized for displaying vascular structures and reminiscent of conventional (catheter-based) angiography. CTA scanning combines optimal synchronization of the contrast medium injection with CT data acquisition to achieve particularly strong contrast medium opacification of the vascular structures. The resulting high-contrast, high-resolution CTA data set can be further reconstructed into specific angiographic representations, such as 3D representations of the vessels, using volume-rendering techniques.

Although modern CT scanners have impressively short 'shutter open' times between 65 and 150 ms, these durations are still not sufficient to capture motion-free images of the beating heart or the thoracic aorta [149]. This results in images of the aortic root and ascending aorta that appear blurry, with 'double contours' due to the motion transmitted from the heart, potentially obscuring or mimicking disease and decreasing measurement accuracy. Consequently, CT and related techniques rely heavily on the synchronization of CT acquisition with the patient's electrocardiogram (ECG) signal. This is commonly referred to as ECG-synchronized CT, gated CTA, gated chest, or cardiac CT, which all refer to the same imaging procedure. ECG gating is increasingly used for thoracic aortic imaging, but is only necessary when the aortic arch or descending thoracoabdominal aorta are of clinical interest.

Another related advanced imaging technique is 4D CT or 4D CTA, which typically requires prolonged exposure to radiation. This technique typically reconstructs ten sequential 'frozen' 3D data sets across the entire cardiac cycle, using the R-peaks of the patient's ECG signal as a temporal reference. The ten data sets, reconstructed with time stamps between 0 and 90% (in 10% increments) of the R-R interval, can be viewed in a cine-loop. Note that the R-R interval is the time period between two consecutive R waves, typically the highest and most prominent upward deflection in an ECG representing the duration of a complete cardiac cycle. This ultimately leads to visualization of a beating heart, the motion transmitted to the aorta, as well as the cyclic diameter change (or 'pulsation') of the aorta and the cyclic motion and deformation of a dissection flap. 4D CTA data are clinically used in many academic or large medical centers and enable quantitative analysis of cardiac phase-dependent deformations with high spatial resolution. In addition, 4D CTA data can be used to measure the distensibility of large vessels, such as the aortic arch and the abdominal aorta, using ECG-gated CTA [150–152]. Distensibility is defined as the maximum change in cross-sectional area relative to the pressure change, scaled by the minimum area [153]. This method can also be applied to smaller vessels, like the carotid arteries [154]. However, measuring distensibility in smaller vessels becomes more challenging because the annotation error increases relative to the change in cross-sectional area.

Recently, photon-counting CT (PCCT) has emerged as an advanced alternative imaging technique and is becoming a standard clinical use in CT imaging [155,156]. Compared to conventional CT imaging, PCCT offers several advantages, including improved spatial and contrast resolution, reduced image noise and artifacts, and lower radiation exposure, resulting in an overall improvement in quantitative imaging. In addition, the high resolution could enable future investigations into the structure of the aortic wall and measurement of wall thickness along its length, which has not been possible previously.

*Magnetic resonance imaging.* MRI is an accepted alternative to CT, but has the disadvantage of a longer acquisition time, lower spatial resolution, and typically limited availability in the immediate vicinity of an emergency room [32,157]. The advantage is that it does not require ionizing radiation and is therefore well suited for monitoring the size of the aorta, especially in young patients and in

#### M. Rolf-Pissarczyk et al.

surveillance imaging. The resulting images typically have a thickness of 1 to 3 mm and a pixel size of only 1 to 2 mm. An MRI study protocol typically comprises several pulse sequences, with each sequence lasting from seconds to minutes, and is optimized for specific purposes. Total study time is generally between 30 and 60 minutes, but may be significantly longer in a research setting. A pulse sequence in MRI involves certain settings and instructions for the MRI device that affect the appearance of the image. It regulates the timing, strength and type of pulses used and can highlight different tissues or disease processes.

Magnetic resonance angiography (MRA) imaging, much like CTA, is most commonly performed by injecting a gadolinium-based agent. It uses pulse sequences to generate optimized vascular images. MRA data sets can be ECG-gated, but usually are not. Since the 3D data acquisition averages over many heartbeats, this results in mild blurriness of the ascending aorta, which is acceptable for clinical use except for dedicated aortic root imaging. In addition, fast MRA techniques allow multiphasic imaging, resulting in a series of MRA images every few seconds, allowing observations of vascular or tissue enhancement patterns over several seconds to a minute. However, this requires the use of specialized pulse sequences and data acquisition methods.

Cine MRI is the basic cardiac MRI technique used to assess both morphology and function. It is based on a steady-state free precession pulse sequence, which allows for the generation of time-resolved images, typically consisting of 20 frames per heartbeat, to assess the beating heart. This imaging technique is considered the gold standard for determining functional parameters such as the ejection fraction of the left ventricle. In addition to cardiac imaging, cine MRI can also be used to detect motions of the dissection flap, flow disturbance at entry and exit tears, and to distinguish flowing blood from thrombus at selected slice locations.

Phase contrast (PC)-MRI, which is based on the phase difference in moving protons, is a standard magnetic resonance sequence available on all cardiac MRI-capable scanners. The PC pulse sequence can be used to visualize and quantify blood flow in the cardiovascular system in 2D [158]. Usually, anatomical images are then overlaid with a color-coded map. PC imaging can also use the patient's ECG signal to map the velocity measurements to the phase of the cardiac cycle, a technique called cine PC imaging. This allows the visualization and measurement of blood flow over time, i.e. over the cardiac cycle. Time-resolved (cine) 2D PC images are typically reviewed and analyzed in a cine loop of 20 frames over the cardiac cycle. This allows quantification of maximum, minimum, and average blood flow velocities, and many other flow parameters used in cardiac and vascular imaging. It can be used to measure blood flow velocities in the true and false lumina.

An extension of cine 2D PC-MRI to a 3D volume is time-resolved 3D PC-MRI, also known as 4D-flow MRI or 4D PC-MRI. With the latest acceleration techniques, such as parallel imaging, a 4D-flow MRI data set can be acquired in about 10 minutes. 4D-flow MRI, predominantly used in academic centers, is a promising technique with prognostic value for patients with aortic dissection [159–162]. It facilitates direct, non-invasive blood flow measurements within an entire anatomical volume, and contains velocities encoded in the three dimensions of space. For example, it enables the measurement of blood flow in both the true and the false lumen, blood flow across the fenestrations, and also facilitates the derivation of 4D-flow-based WSSs. However, it is important to note that the assessment of WSS is highly dependent on the accuracy of the surface reconstruction. Due to the limited spatial resolution of 4D-flow MRI, this accuracy is often quite low, leading to frequent errors [163–165]. Moreover, capturing the wide range of velocities present in aortic dissection can pose a challenge with 4D-flow MRI, as a velocity encoding (VENC) must be preselected based on the expected velocity of the target vessel. If the velocity encoding (VENC) is set too high or too low, this can lead to inaccurate measurements. One recently described potential solution is dual-VENC 4D-flow MRI, which enables more accurate measurements of a wider range of velocities.

Ultrasound imaging. In ultrasound imaging, high-frequency sound waves are emitted, and the 'echo' reflected by tissue is received by the same ultrasound transducer. The temporal resolution of ultrasound is high, and gray-scale anatomic images are generated in 'real time' while the echo probe is moved over the tissue of interest. Ultrasound is a widely used technique to image the heart, commonly referred to as echocardiography (ECG) or, in short, echo. Ultrasound imaging is completely non-invasive and used for many other soft tissue imaging applications, as well as vascular imaging. The frequency of ultrasound waves is also altered when reflected by moving blood (the Doppler effect), which can be exploited to measure blood flow velocities, a technique called Doppler ultrasound. When the Doppler signal is displayed in color, for example, blue for velocities toward the probe and red for velocities away from the probe, and overlaid on a simultaneously acquired gray-scale anatomic image, the technique is referred to as color Doppler or duplex ultrasound. When the ultrasound probe is placed on a patient's thorax to examine the heart, it is called transthoracic ECG (TTE). TTE also allows visualization of the aortic root and parts of the ascending aorta. In contrast, transesophageal ECG (TEE) provides a closer look at the heart and heart valves at higher spatial resolution. TEE is considered an invasive procedure that requires sedation or anesthesia to introduce an ultrasound probe into a patient's esophagus. It is also used intraoperatively during procedures such as open heart or aortic surgery. Additionally, it facilitates the visualization of the aortic root and provides limited views of the ascending aorta and the aortic arch [166]. In selected patients, TTE and TEE may also be useful to evaluate the dissection flap [167], to demonstrate communication between the true and false lumina, and to assess blood flow versus thrombus in the false lumen. Sophisticated variations of the insonation pulses and signal analysis allow the generation of 3D Doppler ultrasound images, mostly used in the setting of ECG. Miniaturized ultrasound probes mounted on a catheter tip can be used to generate ultrasound images and even Doppler signal from inside a blood vessel to examine the vessel wall, known as intravascular ultrasound. This is an invasive procedure and is always used in conjunction with another catheter-based diagnostic (angiography) or interventional procedure, for example, to guide endovascular procedures such as intravascular ultrasound-guided fenestration of the dissection flap.

#### 2.3.2. Current treatment strategies

All patients with acute or chronic dissection are medically managed, with the primary goal to control blood pressure. In addition to medical management, open surgical intervention involving the replacement of a segment of the dissected aorta with a synthetic graft, or endovascular surgery with the placement of a stent-graft, also called an endograft, into the diseased aorta may be necessary. The

current indications are based on the 2022 ACC/AHA guidelines [126], which differ slightly from the 2014 ESC guidelines [168]; some of which are summarized here [169].

*Medical (conservative) management.* Non-invasive therapy of aortic dissection consists of pharmacologic blood pressure control, typically with beta-receptor blockers. The rationale for reducing blood pressure in patients with acute dissection is halting disease propagation, and reducing the risk of aortic rupture. Pain medication is often required in the acute phase as well. In the chronic phase, lowering blood pressure reduces wall tension and may thus reduce risk and delay the formation of a false lumen dilatation and delayed rupture.

*Open surgical repair*. Open surgical repair may be necessary for patients in various situations. Classic type A repair is necessary when patients with acute type A dissection require urgent surgery to replace the dissected ascending aorta with a synthetic graft to prevent rupture. The procedure requires a sternotomy and placing the patient on cardiopulmonary bypass, similar to other open heart surgeries such as a coronary artery bypass graft. In addition to the ascending aorta replacement, the procedure may require concomitant repair or replacement of the aortic root, which includes the aortic valve and the coronary artery ostia. An ascending replacement may also include a distal extension along the undersurface of the aortic arch, a procedure known as a hemi-arch procedure. Open surgical repair may also be necessary when a residual dissection remains after type A repair, meaning when an aortic dissection persists following initial treatment. In the majority of patients with type A dissection (~ 90%), the dissection extends beyond the ascending aorta into the arch and descending aorta. This residual dissection distal to the ascending aortic graft is typically managed conservatively, and requires lifelong surveillance of the dissected aorta. Finally, descending aortic repair, or open surgical repair of the descending thoracic aorta, remains a treatment option for patients with complicated type B dissection when the anatomy is not suitable for endovascular repair. Also, in patients with the Marfan syndrome or other connective tissue disorders, open surgical repair is the preferred option to avoid anchoring of an endograft in potentially fragile aortic tissue.

*Endovascular surgery*. The principle of endovascular therapy for aortic diseases consists in the placement of a stent-graft into the aorta through a small groin incision and through sheath and catheter introduction into the common femoral artery. A stent-graft consists of graft material, typically a fabric like polyester, mounted on a metal frame, known as stent, which can be crimped down



Fig. 7. Key synopsis of treatment strategies for aortic dissection, including the class of recommendation ('COR') and level of evidence ('LOE'), based on the 2022 ACC/AHA guidelines [126]. The figure showing endovascular repair is adapted from Vacirca et al. [136] with permission from Elsevier.

enough to fit through the delivery system and then expand to its desired diameter within the aorta. A stent-graft is placed in an angiography suite, or in a hybrid operating room under fluoroscopic and angiographic guidance. Thoracic endovascular aortic repair (TEVAR) is the first-line therapy for patients with complicated type B dissection. The technical goal is to anchor the endograft in a nondiseased segment of the aorta proximal to the dissection, referred to as the landing zone. Subsequently, the endograft is expanded in the true lumen of the aorta to cover the primary entry tear and re-expand the true lumen. The distal extent usually covers at least half of the descending thoracic aorta or more. Ideally, the false lumen thromboses completely and the dissection heals. In most cases, however, a distal portion of the dissection remains perfused, mandating life-long surveillance and follow-up.

Endovascular technology is rapidly evolving and a growing number of patients have become anatomically suitable even if the proximal landing zone would be considered too short for a descending thoracic aortic endograft. Indications for TEVAR are also evolving, with the main controversy related to the treatment of patients with uncomplicated type B dissection (Fig. 7). While TEVAR has the potential to prevent or delay late adverse events, only a subset of patients with uncomplicated type B dissection are likely to benefit from the procedure, which is not without its own procedural risks.

Despite being minimally invasive, endovascular surgery can involve several complications. One common issue is an endoleak, where blood leaks into the dilated aorta outside the stent-graft, potentially causing persistent or recurrent dilatation and possibly requiring further procedures to correct. Another complication is the migration of the stent-graft from its intended position. Additionally, endografts can cause new intimomedial injuries, a condition known as stent-graft-induced new entry (SINE) or distal SINE (dSINE).

*Hybrid procedures.* Hybrid procedures refers to the combination of open surgical repair with endovascular techniques. Consider a patient with a type A dissection who might undergo open surgical repair of the ascending aorta and the entire aortic arch, while the surgeon simultaneously inserts a stent-graft into the true lumen of the descending thoracic aorta. Another example involves the placement of an open surgical left carotid to left subclavian artery bypass graft. This procedure subsequently allows an aortic endograft to be landed in the aortic arch, covering the subclavian artery ostium. It thus facilitates an endovascular treatment option for an individual with an inadequate landing zone in the descending thoracic aorta.

*Treatment strategy*. All patients with aortic dissection are medically managed. The selection of open and endovascular interventions depends predominantly on the anatomy involved (type A versus type B dissection), and on the presence or absence of acute complications (in type B dissection). All patients who survive an aortic dissection remain at high risk for late adverse events and require lifelong specialized care and surveillance imaging to monitor the size of the dissected aorta. See the key synopsis of treatment strategies for aortic dissection, including the class of recommendation and level of evidence based on the 2022 ACC/AHA guidelines [126].

*Alternative treatment concepts*. In addition to the primary treatment options recommended by the 2014 ESC [168] and the 2022 ACC/AHA [126], there are alternative options for patients who are at extremely high-risk for open surgery, such as elderly patients. According to the key synopsis of treatment options for aortic dissection based on the 2022 ACC/AHA guidelines (shown in Fig. 7), open surgery remains the primary treatment for type A dissections. This preference is due to the complex technical and anatomical challenges posed by endovascular therapy in the ascending aorta [170]. These challenges include the dynamic motion of the ascending aorta, the proximity of vital structures to the TEVAR landing zone such as the aortic valve, coronary arteries and supra-aortic branches, and the short and wide shape of the aorta. Furthermore, the lack of specially designed endografts further complicates endovascular approaches. These factors make endovascular approaches less viable and reinforce the preference for open surgery in these cases. Because TEVAR has been used in a small number of patients with type A dissection, clinical data are limited. In fact, 10 to 30% of



**Fig. 8.** (a) A representative patient case of late complications after TEVAR in ascending aortic dissection with (i) a localized type A dissection, (ii) excellent results after TEVAR with a short stent-graft, and (iii) a stent-induced re-entry tear (erosion) from the proximal contact between the crown of stent-graft and the out-curve of the aorta after six months. (b) A representative patient case for a false lumen intervention to promote remodeling and thrombosis (FLIRT) procedure, showing (i) a depiction of a localized type A dissection with an entry tear at the outer curvature of the aorta just proximal to the innominate artery, (ii) the result of endovascular treatment with coils inserted into the false lumen to promote thrombosis and a patent for a men ovale occluder (arrow) to isolate communication between the true and false lumina, and (iii) a complete remodeling of the aorta without any complications after two years of surgery. Figure is adapted from Yuan et al. [170] (licensed under CC-BY 4.0).

patients are not accepted for surgery, making them potential candidates for TEVAR [171]. Furthermore, it has been shown that up to 50% of patients with type A dissection are technically suitable to TEVAR [172,173]. However, the outcome of TEVAR also depends on the location of the tear. A tear in the center of the ascending aorta is more beneficial for endovascular treatment than a tear near the aortic valve. One of the most common complications after stent-grafting is endoleak, which indicates inadequate sealing of the endograft and can lead to unfavorable aortic remodeling. This is illustrated in Fig. 8(a), where the TEVAR procedure resulted in a so-called SINE followed by significant aortic dilatation.

Another alternative treatment method is false lumen intervention to promote remodeling and thrombosis, abbreviated as FLIRT [170]; the interested reader is also referred to [174]. This treatment option has so far received little attention. FLIRT is based on minimalist, percutaneous interventions with occluder devices, vascular plugs, distal stent-grafts, or coils and glue, applicable to both chronic type A and B dissections. These occluder devices are used to cover the primary intimal tear, which helps create favorable hemodynamic conditions that promote remodeling and subsequent false lumen thrombosis, as shown in a representative patient case in Fig. 8(b).

Finally, the multilayer flow modulator should be briefly introduced, especially due to its recent interest in *in silico* modeling, as shown later in Section 6. The multilayer flow modulator is a device consisting of interconnected wires forming a mesh with a specific porosity of 60 to 70%, tailored for applications such as aneurysm or dissection repair [175]. The design of the multilayer flow modulator directs and laminates blood flow along the vessel wall, aligning it with systemic pressure flow to eliminate turbulence and vortices. Vortices, in particular, are dangerous because they are known to increase peak stresses in the wall and alter viscoelasticity, plasticity, and cellular wall activity in aneurysms, resulting in damage at specific regions in the dilated aortic wall. Unlike traditional stiff stent-grafts, the multilayer flow modulator's compliance closely matches the native aorta, reducing peak stresses in the wall and minimizing complications such as stent-graft migration and aortic neck dilatation due to rapid endothelialization. While there is ongoing debate about its long-term impact on aortic aneurysms and dissections, the multilayer flow modulator has shown its technical feasibility and acceptable survival rates [176]. However, it remains controversial whether treatment with the multilayer flow modulator affects the natural history of aortic aneurysms and dissections [177–180].

# 2.4. Congenital, hereditary, and other related risk factors

The majority (~ 80%) of thoracic aortopathies – dilatation, dissection, and rupture – tend to present sporadically in men and women, with hypertension and vascular aging among the most common risk factors. Hypertension is defined as a sustained elevation of blood pressure. Although hypertension initially increases circumferential wall stress, this stress tends to be restored toward normal via mechanobiological responses by the cells of the wall (endothelial, smooth muscle, fibroblasts, and resident macrophages) that thicken the medial and adventitial layers [181,182]. Because functional elastic fibers are produced primarily during development, this thickening necessarily results largely from an increase in mural collagens, SMC mass, and PGs, which by definition alters the otherwise finely balanced composition and properties of the wall. In this way, the homeostatic tendency to reduce wall stress can nevertheless compromise aortic function and, in some cases, compromise its structural integrity [79,183]. Noting that hypertension results in a structural stiffening along the entire aorta, this global effect increases the aforementioned pulse wave velocity, which, in turn, causes earlier reflections of the pressure wave from distal sites and, thus, an earlier return of the reflected waves that augment proximal pulse pressure. In this way, hypertension not only alters the properties of the proximal thoracic aorta, it also increases the hemodynamic load on the proximal segment, which appears to increase its vulnerability to aortopathy. In this case, therefore, local mechanobiological responses result in global pathological consequences, thus driving a potentially positive feedback loop that can promote disease progression [184,185]. Albeit via different cell mediated mechanisms, vascular aging also results in local changes in composition and mechanical properties [186,187], with increasing central artery stiffness also resulting in proximal pressure augmentation [188].

By contrast, the remaining ( $\sim$  20%) presentations of thoracic aortopathy stem from either congenital or hereditary causes. The primary congenital defect predisposing to thoracic aortic aneurysm is bicuspid aortic valve, in which the valve has two rather than the normal three leaflets. It is widely thought that the associated perturbations in hemodynamic loading on the aortic root and ascending aneurysm contribute to the progressive mural deterioration that characterizes these aortopathies ([189]; also see the discussion of computational fluid mechanics in the following sections). The list of genetic mutations predisposing to hereditary thoracic aortopathies continues to grow (Table 2), with the first mutation discovered in 1988. Advances in genomics, transcriptomics, and proteomics, as well as the development of mouse models of the human mutations, continue to yield increasing insight into the underlying biological and mechanical mechanisms responsible for both the pathogenesis and the progression of disease, but there remains a pressing need to connect molecular mechanism and mechanical failure.

As noted throughout this review article, the aortic wall consists of a complex collection of multiple cell types and myriad ( $\sim 100$ ) intramural proteins, glycoproteins, and PGs. In 1988, the first connective tissue disorder definitively associated with arterial disease, including thoracic aortopathy, implicated mutations to *COL3A1* [192], the gene that encodes the individual alpha helices that constitute the collagen III molecule. Noting that type III collagen is the second most abundant collagen within the aortic wall, found primarily in the medial layer, defects in collagen III render the wall fragile, susceptible to dissection and rupture [213]. The associated syndrome is known as vascular Ehlers-Danlos syndrome, which is rare, with an incidence of one in 50,000 to 200,000 live births. A mere three years later, it was reported in three simultaneous studies that mutations to the gene *FBN1* that encodes fibrillin-1 predispose to thoracic aortic aneurysm and subsequent dissection and rupture [196,214,215]. The associated syndrome is known as Marfan

#### Table 2

Some of the many mutations that predispose to thoracic aortopathy, including dissection. Those at the top have definitively been shown to be disease causing, while those at the bottom have strong to moderate associations. Note the three classes of mutations – those affecting ECM, those affecting the mechanosensitive cytokine TGF- $\beta$  and its signaling, and those affecting SMC actomyosin activity. See Pinard et al. [190] and Creamer et al. [191] for more detailed discussion (SM: smooth muscle; cGMP: cyclic guanosine monophosphate).

| Discovery            | Report    | Gene           | Affected protein                | Effect                                |
|----------------------|-----------|----------------|---------------------------------|---------------------------------------|
| Definite association |           |                |                                 |                                       |
| 1988                 | [192–195] | COL3A1         | Collagen III                    | Reduced ECM (collagen III) integrity  |
| 1991                 | [196]     | FBN1           | Fibrillin-1                     | Reduced ECM (elastic fiber) integrity |
| 2005                 | [197]     | TGFBR1, TGFBR2 | TGF- $\beta$ receptors          | Altered TGF- $\beta$ gene products    |
| 2006                 | [198,199] | MYH11          | SM-myosin heavy chain           | Reduced SMC actomyosin activity       |
| 2007                 | [200]     | ACTA2          | SM-alpha actin                  | Reduced SMC actomyosin activity       |
| 2010                 | [201]     | MYLK           | Myosin light chain kinase       | Reduced SMC actomyosin activity       |
| 2011                 | [202]     | SMAD3          | TGF- $\beta$ signaling molecule | Altered TGF- $\beta$ gene products    |
| 2012                 | [203,204] | TGFB2          | TGF- $\beta$ 2 ligand           | Altered TGF- $\beta$ gene products    |
| Strong to modest as  | sociation |                |                                 |                                       |
| 2011                 | [205]     | SMAD4          | TGF- $\beta$ signaling molecule | Altered TGF- $\beta$ gene products    |
| 2011                 | [206,207] | FLNA           | Filamin A                       | Cytoskeleton/mechanotransduction      |
| 2013                 | [208]     | PRKG1          | cGMP-dependent kinase           | Reduced SMC actomyosin activity       |
| 2015                 | [209]     | TGFB3          | TGF- $\beta$ 3 ligand           | Altered TGF- $\beta$ gene products    |
| 2016                 | [210,211] | LOX            | Lysyl oxidase                   | Reduced ECM (cross-linking) integrity |
| 2019                 | [212]     | SMAD4          | TGF- $\beta$ signaling molecule | Altered TGF- $\beta$ gene products    |

syndrome, which is also rare but not uncommon, presenting in approximately one out of 5,000 live births. Importantly, the ECM glycoprotein fibrillin-1 associates with elastin to form the elastic fibers of the medial lamellar units; it also exists as microfibrils that connect the SMCs to these laminae. Although it was initially thought that fibrillin-1 promoted elastogenesis (i.e. the *de novo* formation of elastic fibers), early mouse models of Marfan syndrome suggested that disrupted fibrillin-1 affects the structural stability of the elastic fibers [216], reducing their otherwise long half-life (which is estimated at 50+ years for normal elastic fibers). Recalling again that functional elastic fibers appear to be produced and organized only during aortic development, reducing the structural stability of these fibers has dire consequences, both accelerating its fatigue considerably and resulting in elastic fragments that promote proteolytic activity. Of course, compromising elastic fiber integrity also diminishes the primary mechanical function of the aorta, to store elastic energy during systolic distension and to use this energy to work on the blood during diastole to augment antegrade and retrograde blood flow [217]. Although fibrillin-1 mutations affect all segments of the aorta, disease first manifests at the aortic root and proximal ascending aorta, the region most susceptible to dissection and rupture in Marfan syndrome, again likely accelerated in part by proximal pulse pressure augmentation [218].

In contrast to these direct connective tissue disorders (*COL3A1* and *FBN1* mutations), well over a decade later it was discovered (in 2005) that mutations to genes (*TGFBR1*, *TGFBR2*) that encode receptors of the transforming growth factor- $\beta$  (TGF- $\beta$ ) system also predispose to thoracic aortopathies [197,219]. The cytokine TGF- $\beta$  has diverse effects on cell function, including synthesis of ECM constituents and smooth muscle contractile proteins, which play critical roles in aortic development, homeostasis, and remodeling. Importantly, subsequent studies further identified other mutations within the TGF- $\beta$  system that predispose to thoracic aortopathies, including mutations to ligands (*TGFB2*; [203,204], and similarly *TGFB3*; [220]) and downstream signaling species (*SMAD3*; [202,221], and similarly *SMAD4*; [205]). TGF- $\beta$  is a highly mechano-regulated cytokine – increases in wall stress can drive both its synthesis [222] and its activation within the ECM from the latent form in which it is secreted [223]. There are also documented interactions between TGF- $\beta$  and integrin signaling [224].

At about the same time that TGF- $\beta$ -opathies were discovered, it was also found (in 2006 and 2007) that mutations to genes that encode intracellular contractile proteins also predispose to thoracic aortic aneurysms and subsequent dissections [198,200]. The first two causative mutations were in *MYH11*, the gene that encodes smooth muscle myosin heavy chain, and *ACTA2*, the gene that encodes smooth muscle alpha-actin. Additional contractile related mutations were discovered soon thereafter [201,208]: loss-of-function mutations to *MYLK*, the gene that codes myosin light chain kinase, and gain-of-function mutations to *PRKG1*, the gene that codes a cyclic guanosine monophosphate-dependent kinase; both predispose to thoracic aortopathies, noting that these two mutations similarly serve to reduce smooth muscle contractility. Although mutations to these four genes first implicated diminished vasoconstriction secondary to reduced smooth muscle contractility, it is important to note the following. First, the primary role of the SMCs of the aorta is to establish, maintain, and remodel the ECM of the medial layer, not to vasoregulate the lumen; second, actomyosin activity is also central to other cellular functions, including migration and mechanotransduction. Interestingly, mutations to the gene (*FLNA*) that encodes the actin cross-linker filamin-A also appear to predispose to thoracic aortopathies [225].

Finally, note that mutations to the gene (LOX) that encodes lysyl oxidase, a potent cross-linker of newly deposited elastic and collagen fibers, have also been found to predispose to thoracic aortopathies [210,211]. Consistent with this discovery, it has been

known since the 1960s that the lathrogen  $\beta$ -aminopropionitril, which blocks lysyl oxidase, also gives rise to aortic dissections and rupture [226,227], thus emphasizing further the overall importance of ECM integrity, which is primarily but not exclusively dictated by competent elastic fibers and fibrillar collagens.

*Dysregulated mechanosensing and mechanoregulation.* Given that many dissections occur secondary to aneurysms, it is instructive to realize that aneurysms arise similarly in cases wherein genetic mutations affect ECM, mechanosensitive signaling pathways, or intracellular contractile proteins (Table 2). Importantly, these three classes of mutations affect components along what is known as the mechanotransduction axis – that is, the axis that enables cells to use actomyosin machinery and transmembrane protein complexes to assess their local mechanical environment. It has been suggested, therefore, that thoracic aortic aneurysms and their sequelae result, in part, from compromised mechanosensing and mechanoregulation of the ECM that otherwise endows the wall with its mechanical functionality and structural integrity [228]. Diverse aspects of this failed mechanobiological process in the context of thoracic aorta aneurysm, dissection, and rupture are discussed in related papers [78,229–233].

Briefly, mechanobiology is the study of biological responses to mechanical stimuli, which ultimately depend on three cell activities – transduction, transcription, and translation. Translation refers to the actual production of gene products based on the associated gene expression (or transcription), but within this context often includes processes that also affect post-translational modifications, including cross-linking. Importantly, it appears that cells do not merely synthesize and secrete the extracellular proteins that contribute to the mechanical integrity of the ECM, they also work on and organize (i.e. mechano-regulate) this matrix to endow it with desirable properties that affect stiffness and strength. Such work requires actomyosin activity. A cell's ability to assess external mechanical stimuli (i.e. mechano-sense) requires both transmembrane protein complexes (i.e. clusters of integrins that are referred to as focal adhesions) to couple the intracellular cytoskeleton to the ECM and actomyosin activity to enable the cell to probe or assess the mechanical state of the ECM [234,235]. Transduction of these mechanical stimuli occurs both directly and via second (chemical) messengers, both typically resulting in altered gene expression (transcription). Although it has been suggested that cells sense continuum concepts of stress or strain, or metrics derived from them, this is unlikely [236]. Rather, the cells likely sense conformational changes in mechanosenstive molecules within the cell membrane, the cytoskeleton, or the nucleus, or a combination of all three. Nevertheless, it has proven convenient to correlate mechanosensitive biological responses, including altered turnover of cells and matrix, with continuum level quantities such as stress, strain, and stiffness, as this facilitates computational modeling [237].

Even under normal quiescent conditions (i.e. homeostatic), the ECM of the aorta is nevertheless undergoing continual turnover, that is, there is continual production of matrix to replace that which is removed consistent with the finite half-life of all biological materials and cells. So-called growth (changes in mass) and remodeling (changes in structure) can thus be modeled mathematically given three classes of key constitutive relations, those for the production, removal, and material properties of individual structurally significant constituents [237], which in the normal aorta includes elastic fibers and fibrillar collagens. That one must track different constituents, having different rates of turnover and different mechanical properties, naturally suggests the use of a continuum theory of mixtures [237]. Indeed, recalling that mutations to genes that affect diverse ECM components or contribute to their integrity, ranging from fibrillin-1 to lysyl oxidase, reminds us that modeling the evolving biomechanical properties increasingly demands greater consideration of the many constituents that contribute to structural integrity, not just the elastic and collagen fibers. Similarly, noting that this integrity ultimately stems from gene products that result from complex cell signaling, there is clearly a need to pursue multiscale modeling that couples transcriptional changes with tissue-level manifestations (cf. Irons et al. [238]).

*Other strong risk factors.* As noted above, two of the strongest risk factors for aortic dissection are vascular aging and hypertension. Much has been written on the effects of both of these risk factors on the aorta, hence the interested reader is referred to related review articles, see, e.g., [185,188]. Of particular note, albeit for different reasons, both of these conditions tend to increase central artery stiffness, which increases pulsatile loading on the proximal thoracic aorta that is most vulnerable to dissection, namely type A dissection. This increase in loading stems from an increase in the pulse wave velocity, which depends on the material stiffness of the wall as well as its thickness. Hypertension tends to drive a thickening of the aortic wall in large part due to an increase in collagen accumulation. Vascular aging similarly tends to include increased accumulations of collagen fibers and GAGs, though also with a progressive loss of elastic fiber integrity. As a result, aging affects aortic distensibility [239] as well as biaxial mechanical properties [186,240].

Another situation of critical importance to understanding the risk of aortic dissection is pregnancy. Again, much has been written and the interested reader is referred to multiple review articles [241–245]. There is, however, a dearth of information on changes in the mechanical properties of the aorta during pregnancy, particularly in cases where a mutation exists (e.g., Marfan or Loeys-Dietz syndromes) that predisposes to aortic dissection. As noted below, mouse models can be useful in studying mechanisms, and this is a pressing need with regard to biomechanical changes during pregnancy. Finally, it is noted that the risk of aortic dissection can also remain high for the mother after birth, with a study in mice suggesting that it is increased oxytocin (which drives uterine contractions and lactation) that increases risk [246]. Much remains to be learned, however.

# 3. Pathological changes associated with aortic dissection

The pathological changes associated with aortic dissection significantly alter both the aortic anatomy and the microstructure of the wall. These modifications lead to changes in mechanical behavior and hemodynamics, potentially resulting in thrombus formation



**Fig. 9.** Image of harvested tissue samples from patients with aortic dissection: (a) remodeled thoracic aorta showing the true and the false lumen and (b) cross-sectional ring sample with a thrombosed false lumen. Figures (a) is adapted from Amabili et al. [250] with permission from Elsevier, while (b) is unpublished data provided by S. Sherifova (Institute of Biomechanics, Graz University of Technology, Austria).

within the false lumen. Some of these changes may not be immediately evident but become more pronounced as the condition becomes chronic.

Two representative tissue samples harvested from the thoracic aorta of a patient with aortic dissection are shown in Fig. 9. During the event of an acute dissection, the dissection flap is visible, separating the true and false lumina and usually comprising the intima and part of the media. The acute angle formed where the dissection flap and the outer wall of the false lumen intersect is typically sharp in acute aortic dissection. However, local thrombus formation and neointimal growth can cause this angle to become less defined, or, in other words, result in a more obtuse configuration. This will also be demonstrated later in Section 4.1 (Fig. 35(b)). The acute angle, also known as 'beak sign' [247–249], appears at the lateral, distal, and proximal ends of the false lumen. This edge, which consists of the borders of the outer wall and flap, may be thrombosed. As the cylindrical ring extracted from an aortic dissection case in Fig. 9(b) shows, the false lumen can be entirely thrombosed. Curiously, the outer wall shows significant pathological changes. Over time, chronic aortic dissection may result in remodeling of the microstructure of the wall and the anatomy, depending on the individual patient case.

# 3.1. Pathophysiology of aortic remodeling

An aortic dissection is associated with a remodeled anatomy that can include various characteristics. One of these features is the formation of a dissection flap, which is typically fenestrated by one or more intimal tears. In addition, the anatomy of the aorta can remodel over the course of the disease, leading primarily to dilatation, but also to axial growth, an altered angulation, and an increased tortuosity. However, the implications and potential impact of these remodeled anatomical features on the initiation and progression of aortic dissection, which may be interrelated, are not always obvious.

## 3.1.1. Extent and configuration of the dissection flap

The dissection flap, a characteristic feature of aortic dissection, separates the true from the false lumen, and is often referred to by various terms in the literature, such as 'dissection septum' [251], 'intimal flap' [252], or 'dissection membrane' [253]. Although these terms share the same meaning, some may not be entirely accurate. As Berguer et al. [254] pointed out, the term intimal flap is misleading because it implies involvement of only the intimal layer. This terminology is inaccurate, as we will demonstrate later. The dissection flap is typically fenestrated by a single or multiple intimal tears, which can vary in location and size.

Location, number, and size of intimal tears. The intimal tear connects the true and false lumina, enabling blood flow between the two lumina. In the literature, various terms are used to describe this characteristic feature, such as 'entry' and 're-entry' tears, 'primary' or 'secondary' tears, 'entry' and 'exit' tears, 'fenestrations', 'connections', or a combination thereof. However, distinguishing between entry and re-entry tears is often challenging. A significant number of aortic dissection patients present multiple intimal tears, adding complexity to the classification. While some of these terms may not be entirely incorrect, they can imply specific pathophysiological sequences, including intimal tear creation, blood entering the media, cleavage plane or false lumen formation within the wall, and intramural hemorrhage decompression back into the true lumen. Recently, Czerny et al. [138] suggested replacing the terms 'multiple entries and re-entries' with 'most proximal tear', 'communications between lumina', and 'most distal tear', in addition to the term

'primary entry tear'. Despite this proposal, the classification of intimal tears remains contentious and unresolved [138,141,255,256]. In this review article, we aim to maintain consistent and clear notation with respect to the intimal tear.

The location of the intimal tear can be identified using various medical imaging methods, see Section 2.3.1, the most common being CT or MRI [257], but in many cases determining the exact location has proven challenging [258–260]. Studies have shown that intimal tears are most commonly located in the initial few centimeters of the ascending aorta and just distal to the left subclavian artery in the descending aorta, also called aortic isthmus [258]. In addition, intimal tears can be located proximally to the brachiocephalic artery in the ascending aorta, within the aortic arch or proximal to the celiac or renal arteries, as well as proximal to the iliac arteries [137,257,261–264]. Cuellar-Calabria et al. [265] reported on the most frequent location of the distal tear in 72 chronic type B and repaired type A dissection patients: the tear was most commonly found in a single iliac artery (56%), followed by the abdominal aorta (22%), with bilateral iliac distal tears observed in 18% of the cases. Others stated that the location of the intimal tear was related to either the convexity or concavity of the aorta [266] and found that 48% of intimal tears occur at the convexity (the outer circumference of the aorta) and at the concavity (the inner circumference of the aorta) in 52%. Nonetheless, Roberts [261] acknowledged numerous exceptions and noted that intimal tears typically involve the right lateral aortic wall along the aortic convexity, including the ascending aorta, the aortic arch, and the descending thoracic aorta. Dziodzio et al. [267] argued that an intimal tear proximal to the ligamentum arteriosum, at the aortic concavity, is relatively rare. In contrast, Peery [268] and Shennen [269] argued that the aortic wall is weaker and more degenerated at the ligamentum arteriosum's insertion point, rendering it more susceptible to tearing. This increased vulnerability is attributed to the hinge-like motion at the junction between the aortic arch and the descending aorta during each pulsation, which is caused by the ligamentum arteriosum's location between the relatively free aortic arch and the relatively fixed descending aorta. These conflicting viewpoints illustrate that the precise location of a potential future intimal tear cannot be predicted with absolute certainty.

The number and sizes of intimal tears, in addition to their locations, play a significant role in the progression of aortic dissection, specifically the extent of the dissection flap and how it is treated medically. Various studies have examined the average number of intimal tears present in each aortic dissection patient. Khoynezhad et al. [263] discovered up to ten intimal tears along a single aorta using CT imaging, with an average of  $2.8 \pm 2.11$  intimal tears per patient. Similarly, Quint et al. [256] identified up to seven intimal tears along a single aorta and an average of  $2.48 \pm 2.11$  intimal tears. In contrast, some studies have identified cases in which no intimal tear was detected. This supports the hypothesis that an intramural hemorrhage may precede aortic dissection, as we will discuss in Section 3.2.1. As an instance, Hayashi et al. [258] reported an absence of intimal tears in 5% of aortic dissection patients. Ergin et al. [257] found similar results while examining type A and B dissections. They consistently identified intimal tears in type A dissection cases, but the intimal tear was absent in 5 to 10% of type B dissection cases.

Each patient is unique, and despite advances in imaging techniques, detecting all fenestrations remains challenging. Besides the intimal tears visible in the dissection flap using, *inter alia*, CT imaging, it is not uncommon for numerous additional tears, known as natural fenestrations, to be present as well. In a normal aorta, numerous intercostal arteries originate from the aorta, although the exact number may vary among individuals. When an aortic dissection initiates, it is inevitable that some intercostal arteries will be affected. This may result in the development of small natural fenestrations within the dissection flap, potentially allowing blood flow from the false lumen, even if such connections are too small to be identified with current CT technology [270]. These natural fenestrations can significantly affect hemodynamics in the false lumen, particularly blood pressure. However, they are often neglected or not taken into account because they are difficult to identify using currently available imaging techniques.

The size and shape of intimal tears can vary among patients, locations, and dissection types. For example, Evangelista et al. [264] reported an average intimal tear size of  $10.4\pm5.2$  mm for patients with type B dissection and  $7.3\pm4.2$  mm for those with type A dissection. The size was quantified as the maximum diameter measured by TEE. In their study, the tear size ranged from 3 to 27 mm across all patients. Quint et al. [256] similarly conducted a study on the size of 129 intimal tears in 52 patients, providing both the width and length of these tears. The width was defined as the maximum distance between the free edges of the flap, a dimension obtained from CT images. Meanwhile, the length was defined as the thickness of images in which each tear was visible, referring thereby to the axial extent of the tear. Their findings revealed that the tear size varied between  $2 \times 1$  and  $21 \times 45$  mm, with most of the tears (68%) measuring <10mm in each dimension. No significant differences were noted in the size of intimal tears in the ascending aorta, the aortic arch, or the descending aorta. However, the tears in the thoracic aorta were significantly larger than those in the abdominal aorta. Additionally, Cuellar-Calabria et al. [265] conducted a study on the intimal tear size in 72 patients with repaired type A and chronic type B dissections. They found that the proximal tear area was significantly larger in the type B dissection group, with an average size of  $1.5 \text{ cm}^2$  (ranging from 0.6 to  $2.4 \text{ cm}^2$ ), compared to the type A group, which had a smaller average tear area of  $0.4 \text{ cm}^2$ (ranging from 0.1 to  $1.35 \text{ cm}^2$ ). In contrast, the distal tear area was larger in the type A group, with an average size of  $1.0 \text{ cm}^2$  (ranging from 0.55 to 1.5 cm<sup>2</sup>), compared to the type B group, which had an average size of 0.4 cm<sup>2</sup> (ranging from 0.2 to 1.1 cm<sup>2</sup>). Furthermore, their analysis revealed that a difference greater than 1.2 cm<sup>2</sup> between the proximal and distal intimal tear areas during the subacute phase of aortic dissection was associated with an increased risk of adverse events.

In contrast to the size of intimal tears, their shape has often been examined to a lesser extent. Hirst et al. [141] reviewed 161 dissecting aneurysm cases and characterized the orientation and shape of intimal tears. They found that the tears were most commonly circumferentially oriented, although some were axially oriented. Moreover, they demonstrated that the shape of intimal tears varied significantly, making it difficult to determine a general shape. They identified various shapes, including round, elliptical, and zigzag forms. Similarly, Hukill [271] reported a single case with an intimal tear measuring 35 mm in length and oriented at a 45° angle to the circumferential axis. Thubrikar et al. [272] used a mathematical model of an aneurysm to explain the circumferential orientation of the tear. In the dilated aorta, they noted that in certain regions, axial stress exceeded circumferential stress. This finding is consistent with the characteristics of aortic dissection, which is often associated with dilatation, too. Since the rupture strength in the axial direction is

generally lower, a rupture caused by axial stress will result in a circumferential tear. Additionally, many of the medical and *in silico* studies discussed in this and subsequent sections typically report the size and shape of intimal tears, as well as the number of tears identified in case reports. However, accurately determining the precise size and shape of intimal tears from medical imaging often proves to be challenging.

*Extent of the dissection flap.* The path of the false lumen or the dissection flap along the true lumen may vary due to multiple factors. However, the most common locations are on the right side and anteriorly in the ascending aorta, superiorly in the aortic arch, and on the left side and posteriorly in the descending and abdominal aorta. Moreover, the false lumen in the descending thoracic and abdominal aorta often spirals posterior and appears on the right side of the true lumen [258,273]. Although the propagation direction of the false lumen is not yet fully understood, it is clear that anatomical barriers significantly affect its course. These barriers may not be immediately apparent, but the aortic branches, including the numerous intercostal arteries or anchor points such as the ligamentum arteriosum, can considerably impede the progression of the false lumen [267,268].

The dissection flap can propagate in either antegrade or retrograde directions, with a helical shape often observed when it propagates in the descending aorta. Additionally, the flap can naturally extend in the circumferential direction. Weiss et al. [266] conducted a study on type B dissection and compared retrograde dissection propagation based on the intimal tear's location at the convexity or concavity. They examined 52 patients and found that the retrograde dissection propagation by the left subclavian artery was obstructed when the intimal tear was located at the convexity and served as a natural barrier. In contrast, intimal tears at the concavity allowed retrograde dissection propagation in the aortic arch and ascending aorta in the absence of obstructing vessels at the concavity, typically leading to a higher risk of complications. These findings align with the study of Segesser et al. [262]. However, Roberts [261] described retrograde dissection propagation as a rare occurrence, asserting that most dissections involve the entire aorta and that dissection propagation is only prevented by extensive obstructions such as atherosclerotic plaques or the ligamentum arteriosum. In such cases, the ligamentum arteriosum acts as a natural barrier that prevents anterograde dissection progression, making retrograde type A dissection more likely, which in turn increases the risk of complications [267].

*Configuration of the dissection flap.* It is known that the dissection flap can exhibit different configurations throughout the cardiac cycle as it demonstrates significant motion, particularly during the acute phase. The configuration of the dissection flap is primarily influenced by the pressure difference between the lumina [4], the remodeled microstructure of the flap, and the circumferential extent of the outer wall of the false lumen. Previous studies have attempted to track the motion of the dissection flap throughout the cardiac cycle, from the acute to chronic phases, and to classify the predominant configurations.

In 1997, William et al. [274] proposed a classification system for the dissection flap, distinguishing between ischemic and benign configurations. The ischemic configuration is characterized by a dissection flap curving toward the true lumen, causing the true lumen to take on a C-shaped envelope. This is mainly attributed to a pressure deficit in the true lumen that occurs during diastole, where the term ischemic configuration is derived from the word 'ischemia', implying that an insufficient amount of blood flows through the true lumen. Conversely, the benign configuration is more commonly observed during systole when the true lumen pressure is hemodynamically adequate or higher compared with the false lumen pressure. In this configuration, the dissection flap curves toward the false lumen. Lee et al. [275] described these configurations as resembling a gibbous and crescent moon for the true and false lumina, respectively. Both configurations can be observed within a single cardiac cycle, demonstrating the dynamic motion of the dissection flap [276]. The motion of the dissection flap is related to the blood pressure in the true and false lumina and correlates with the phase of the cardiac cycle. Yang et al. [277] found that the measured maximum motion was  $12.2 \pm 4.1$  mm, ranging from 2.6 to 17.4 mm. Conversely, the minimum motion measured was  $6.7 \pm 4.1$  mm, ranging from 0 to 15.3 mm. Moreover, the maximum and the minimum dissection flap motion was at 15 and 75% of the R-R interval, respectively. The benign configuration was most frequently observed, followed by the ischemic configuration, with the dissection flap appearing nearly flat in some parts of the R-R interval. LePage et al. [247] examined the flap configuration at three locations along the dissected aorta, comparing acute and chronic dissections, as shown in Fig. 10(a) with exemplary CT images of ischemic and benign configurations. In chronic aortic dissection, the dissection flap appeared flat in all three locations. In acute aortic dissection, there was significant variability among flat, ischemic, and benign configurations at one, two, and three-quarters of the distance along the dissected length of the aorta. This suggests that the configuration also depends on the distance to the intimal tear.

An alternative classification for flap motion in type B dissection was proposed by Weber et al. [152]. This method allowed for a more accurate assessment of the motion of the dissection flap and provided insights into the dynamics of the dissection process. They determined the change in area and diameter of the true lumen at three different locations during the R-R interval of the cardiac cycle, which correlates with the motion of the dissection flap. The relative area changes for the true and false lumina were  $16.\pm 5.9$  and  $10.5\pm 5.7\%$ , respectively. Additionally, the diameter was measured at the major and minor axes of the true lumen throughout the cardiac cycle. The relative diameter changes for the true and false lumina were  $5.9\pm 2.0$  and  $6.1\pm 3.6\%$  at the major axis, and  $12.7\pm 6.3$  and  $6.0\pm 2.2\%$  at the minor axis, respectively. They noted that the area and diameter fluctuated during the R-R interval of the cardiac cycle.

Similarly, Murayama et al. [281] proposed a potential classification system for aortic dissection that distinguishes between pulsating and static types to identify unstable aortic dissection. This classification is based on the area change of the true lumen and aims to predict future events such as dilatation, occurrence of additional fenestrations, and death. They measured the true lumen area at three different locations in the descending aorta and established a threshold value to distinguish between pulsating and static types.

(c)

(a)

(b)



**Fig. 10.** (a) A representative CT image shows a patient with a dissection flap oscillation amplitude of 7.9mm, which is due to different luminal pressures, with (i) a minimal short axis diameter of 8.8mm (ischemic configuration) and (ii) a maximal short axis diameter of 16.7mm (benign configuration) measured at the end of the aortic arch. (b) Another CT image shows (i) a 'triple-barreled' dissection in the descending aorta ('T': true lumen; '1' and '2': false lumina), which later developed into (ii) a 'quadruple-barreled' dissection ('N': new false lumen). (c) In addition, a CT image shows (i) the dissection flap (red arrow) in the ascending and descending aorta with linear hypodense structures within the false lumen (white arrow) resembling the dissection flap, known as 'cobweb sign', and (ii) a type A dissection with intimointimal intussusception, in which the circumferential dissection flaps with oval morphology ('windsock sign') involve the aortic arch (yellow arrow). Figure (a) is adapted from [278] with permission from Elsevier, while (b) and (c) are from Awal et al. [279] (licensed under CC-BY-NC 4.0), and (d) is from Sueyoshi et al. [280] with permission with Elsevier.

According to their classification, the pulsatile type is present when the maximum area of the true lumen in any of the three locations exceeds 125% of the minimum area of the true lumen, and when the intimal tear is not situated at the distal portion of the descending branch with retrograde flow. The results indicated that the pulsatile type occured  $4.3 \pm 5.9$  months after onset, while the static type occured  $42.0 \pm 13.0$  months after onset.

Ganten et al. [278] investigated pulsatile changes in the aorta, which contributed to the highly variable motion of the dissection flap. They found that the true lumen diameter decreased by up to 29% (mean decrease of 4.4%) during certain phases of the cardiac cycle, which is directly linked to flap motion. The area of the true lumen, associated with its diameter, changes not only throughout the cardiac cycle but also with the progression of aortic dissection. After the initial dissection event, the true lumen area undergoes significant alterations but eventually stabilizes as the aortic dissection progresses until the area remains constant.

Lescan et al. [253] studied the impact of pre-operative dissection flap motion on aortic remodeling and development of dSINEs in patients undergoing TEVAR. They assessed pre-operative dissection flap mobility by measuring the true lumen strain from diastole to systole. They found that the true lumen strain was superior in the acute and subacute groups compared with the chronic group. Moreover, a mobile flap with a mean true lumen strain greater than 22.5% was associated with a higher true lumen expansion rate in the thoracic aorta and a comparable distal stent-induced new entry incidence compared with the immobile flap. These findings suggest that pre-operative assessment of dissection flap mobility can be useful in predicting aortic remodeling and post-operative outcomes in patients undergoing TEVAR.

Similarly, Lortz et al. [252] examined the relationship between the mobility of the dissection flap and aortic remodeling in patients with chronic type B dissection after a TEVAR or a frozen elephant trunk procedure in 52 patients with a mean follow-up of 26.6  $\pm$  20.7 months. They classified non-mobility as  $\leq$ 3 mm and considered a real-time dissection flap movement of >3 mm with a high

amplitude as mobile and a low amplitude as non-mobile. They found that patients with a highly mobile dissection flap prior to operation had improved aortic remodeling and a lower re-intervention rate compared with those with impaired dissection flap mobility. Intravascular ultrasound assessment showed that the mobility of the dissection flap led to a greater true luminal loss and shift in the mobile flap group during every cardiac cycle. These findings suggest that intravascular ultrasound assessment of dissection flap mobility can be useful in predicting post-operative outcomes in patients after TEVAR or frozen elephant trunk.

Numerous studies have also reported the collapse of the true lumen. For example, Williams et al. [282] attributed collapse to the loss of transmural pressure across the dissection flap, indicating a decrease in radial tension but not necessarily axial tension. As a result, the elastic fibers recoil and shorten the flap due to the release of transmural pressure. Furthermore, they suggested that an increased circumference of the outer wall of the false lumen could be an indicator of potential collapse of the true lumen.

In conclusion, the motion of the dissection flap typically diminishes during the transition from the acute to the chronic phase of aortic dissection. This reduction is primarily associated with microstructural remodeling of the flap, see Section 3.2. The stiffness of the flap also serves as a decision criterion for surgical intervention. While increasing stability can enhance the safety of TEVAR, it can conversely decrease its likelihood of success [283]. In other words, immediate intervention could offer benefits because the dissection flap is in its most pliable state and provides the greatest possibility for complete remodeling [284]. However, this advantage needs to be balanced against the potential increased risk of injury to the acutely vulnerable and fragile inflamed aorta with the stent-graft, which could make the patient more susceptible to complications [285].

*Rare manifestations of the dissection flap.* Typically, the true and false lumina are separated by a single dissection flap, as previously discussed. In some cases, however, the false lumen is 'multi-barreled', exhibiting multiple false lumina at the initial presentation or during the follow-up period, leading to a 'triple-barreled' or 'quadruple-barreled' aortic dissection [131,132,271,279,286], see Fig. 10 (b). This manifestation of the dissection flap is not widely recognized and has been considered a relatively rare condition; in approximately 9% of acute aortic dissection patients, a multi-barreled false lumen is observed [130]. These cases are typically only documented in individual case reports. Patients with multi-barreled aortic dissections show significantly poorer survival rates than patients with double-barreled aortic dissection.

Other relatively rare observations in clinics include the 'cobweb' and the 'windsock' sign. The cobweb sign, as introduced by Williams et al. [287], appears as slender linear areas of low attenuation that are specific to the false lumen and are caused by residual ribbons of media that have incompletely sheared off during the dissection process. Example medical images can be found in Fig. 10(c) or in the literature [248,288]. The term 'windsock sign' in type A dissection was introduced by Karabulut et al. [289] to describe an intraluminal windsock-like appearance filled with contrast material and surrounded by a contrast column, a curvilinear intimal flap in the aortic root, the absence of a flap in the dilated mid-ascending aorta, and a dissection with intimointimal intussusception, see also [279,290–293]. This involves a circumferential dissection of the aortic lumen, followed by impediment of the origin of the supra-aortic vessels and occlusion of the aortic lumen, as shown in Fig. 10(c).

## 3.1.2. Dilatation and growth of the aorta

In addition to the development of a dissection flap and possible intimal tears, the aortic anatomy undergoes remodeling over time. Some patients experience aortic dilatation, while others may exhibit elongation or axial growth of the aorta, increased tortuosity, pronounced angulation in the aortic arch, or a combination thereof. Some of these remodeling processes can be associated with agerelated growth, making them potential risk factors for aortic dissection. In other words, certain types of remodeling might occur predissection, making the aorta more prone to developing a dissection, while others are caused by the dissection itself. As an illustration, Fig. 11 presents a comparison of various morphological changes resulting from significant aortic remodeling, such as dilatation, increased tortuosities, and altered angulations, in aortic dissection patients, as captured by CT imaging. To gain a clearer understanding of aortic dilatation in dissection, it is beneficial to first revisit Laplace's law.

*Laplace's law of wall tension*. Laplace's law is a fundamental relation in physics that explains the relationship between the tension in an elastic idealized thin tube and the pressure exerted by the wall on the fluid inside [300]. The circumferential tension in the wall of a cylindrical tube is expressed as T = Pa, where P is the internal pressure and a is the inner radius of the tube, with a < r < b, where b is the outer radius. The circumferential stress within the tube wall is given by  $\sigma = Pa/h$ , where h is the thickness of the wall.

This law implies that the inward pressure exerted by the blood on the arterial wall is directly proportional to the tension in the wall and inversely proportional to the radius of the artery. Consequently, larger arteries must have thicker walls to withstand higher tension, while smaller arteries can afford thinner walls. In other words, at the same pressure and wall morphology, arteries with larger diameters experience greater wall tension than those with smaller diameters. Therefore, arteries that carry higher blood pressure have thicker walls compared to veins. When an artery wall develops a weak spot and dilates, the dilatation subjects the weakened area to even more tension, which can lead to a aneurysm (i.e. a dilatation of 50% or more) or a rupture. Although rupture is rare when the diameter is less than 55 mm. A possible protective mechanism against additional dilatation is wall thickening, which will decrease wall stress and be explained in more detail in Section 3.2.2. Wall stress alone does not determine the likelihood of aneurysm rupture. Material failure occurs when the mechanical stress exceeds the strength of the material. The diameter of the aortic aneurysm alone does not fully account for either wall strength or stress [302,303].



**Fig. 11.** (a) to (g) CTA images demonstrate various remodeled anatomies of aortic dissection patients and showcase dilatation and growth of the aorta, including altered elongation, tortuosity, and angulation. Figure (a) is adapted from Chan et al. [294] with permission from Elsevier, while (b) and (c) are from Lopes et al. [295] (licensed under CC-BY-NC-ND 4.0), (d) is from Dai et al. [296] (licensed under CC-BY-NC 4.0), (e) is from Sultan et al. [297] (licensed under CC-BY 4.0), (f) is from Anwar and Hamady [298] (licensed under CC-BY 4.0), and (g) is from Ma et al. [299] with permission from Elsevier.

In conclusion, although Laplace's law helps to understand the fundamental principles of arterial wall tension and pressure, predicting the risk of aortic dilatation and its rupture requires a more sophisticated approach that considers the complex geometry of the aortic wall and the material properties of the vessel. In fact, aortic dilatation is the most commonly observed factor to assess long-term survival of patients with aortic dissection.

*Dilatation*. In type B dissection, aortic dilatation and the rate at which the aorta dilates are crucial factors in assessing long-term survival of patients. False lumen dilatation, as shown in Fig. 11 and in Fleischmann et al. [34], is the primary reason for reintervention in patients with chronic aortic dissection [34,304]. Furthermore, many patients often experience false lumen dilatation following dissection onset, typically observed in the upper descending aorta [2]. According to the IRAD and others, only 65.5 and 26.7% of type B dissection patients are free of false luminal dilatation three and five years after onset, respectively [305,306]. In contrast, type A dissection rarely progresses into the chronic phase due to the high mortality rate of patients. In surveillance imaging, dilatation and annual growth rate are typically determined based on the difference between the initial and most recent maximum diameter measurements. These measurements are most commonly obtained in clinical settings for the total aortic diameter.

When calculating the annual growth rate, it is typically assumed that the aorta grows uniformly and linearly over time, an assumption that can be questioned. Peterss et al. [14] reported that a rapid change in diameter occured early in the post-dissection period, stabilizing after 25 days and reaching a plateau after 88 days, see Fig. 12(b). From this point onward, the increase in dilatation occured much more slowly and there were no longer any further variations over time. The dilatation rate was estimated to have a



**Fig. 12.** (a) Evolving morphology of type B dissection based on follow-up CT imaging in a single representative patient at the same aortic levels, showing notable changes, particularly in the degeneration and dilatation of the false lumen, with important observations including a significant initial increase in aortic diameter (orange arrow), progressive intimal thickening (orange star), decreased movement of the aortic flap over time (orange triangles), gradual straightening of the flap (green star), and increasing false lumen thrombosis lumen over time (yellow star). (b) Growth rate (mm/day) of aortic dilatation following an exponential decay function with rapid growth in the acute phase, deceleration in the subacute phase, and a plateau in the chronic phase. Figure is adapted from Peterss et al. [14] with permission from Elsevier.

mean of 9.31 mm/year in the acute phase, 1.30 mm/year in the subacute phase, and 0.32 mm/year in the chronic phase.

Several important predictors that may influence dilatation and growth rate were summarized and identified by Spinelli et al. [15]. These included the total cross-sectional size of the aorta, the cross-sectional size of the false lumen, the location and size of the proximal intimal tear, the number of intimal tears, the status of the false lumen with respect to thrombosis, axial extension of the false lumen, and branch vessel involvement. This underlines the highly correlated remodeling phenomenon. However, the various studies presented did not always report consistent findings.

In order to measure the maximum diameter of the total aorta, i.e. the cross-sectional size, various methods have been used in multiple studies. For example, it was measured as the largest diameter perpendicular to the center-lumen-line or as the maximum short-axis distance in axial view [307–309]. Others developed a vascular deformation mapping technique to provide a reliable quantitative assessment of 3D aortic growth and growth patterns in patients with thoracic aortic aneurysms and dissections undergoing CT surveillance [310,311]. This technique is one of several methods used to describe the geometric evolution of the aorta; see also the recent work of Khabaz et al. [312]. Despite the diverse imaging techniques employed, a good agreement was found among these studies concerning its prognostic value for adverse events and mortality. In the majority of the examined studies, a large aortic maximum diameter in the acute phase was a positive predictor for late aortic growth or negative outcomes. Intriguingly, certain studies identified a link between initially smaller aortic diameter and unfavorable consequences [313–316], whereas other investigations failed to establish a significant correlation. Additional studies established a threshold of 40 to 41 mm, measured in the acute phase, indicating an increased risk of late adverse events. This benchmark was first introduced by Kato et al. [317] in 1995, who observed no late aortic growth below this limit. Later, in 2011, Miyahara et al. [318] confirmed the threshold value of 40.5 mm, although this finding might only be applicable to patients with a height of less than 180 cm [319]. Most recently, Evangelista et al. [320] showed that an initial aortic diameter of >45 mm in the acute phase identified patients with increased risk for complications.

To assess the false lumen cross-sectional size, various methods have been employed, either as absolute values or relative to the true lumen. For example, the cross-sectional area of the false lumen was evaluated by calculating the ratio of the true lumen to the entire lumen at the maximum diameter level [317] or by determining the maximum area of the false lumen [321]. In 2007, Song et al. [307] discovered that an initial absolute false lumen diameter at the upper descending aorta  $\geq$ 22 mm was predictive of late aneurysmal change, which was defined as a total aortic diameter greater than 60 mm. The ratio of false lumen to total aortic diameter increased over time in patients experiencing aneurysmal changes. Other researchers reported that the circular configuration of the true lumen was significantly associated with a reduced growth rate [322], or that the circumferential extent of the false lumen, which is the angular distance in degrees between the two insertion points of the dissection flap, predicted adverse events [270]. In addition, others found that false lumen volume and the ratio between true and false lumen volume were indicators of delayed aortic intervention [323]. In summary, numerous findings were presented, but the employed measurement techniques vary and statistical significance could not be shown for all outcomes [264,309,324,325]. Also note that information about the presence of false luminal thrombus and its impact on the measurements obtained was usually not provided.

The presence of an entry tear in the thoracic aorta was also identified as a significant predictor of aortic growth by Kato et al. [317]. However, in their study, it was reported merely as the presence or absence of an entry tear in the thoracic aorta. In two studies, Sueyoshi et al. [326,327] examined the presence of an entry tear in the aortic arch as a predictor of aortic growth in two separate studies, finding it insignificant in 2004 and significant in 2009. Similar to others, some studies investigated the role of the distal exit tear or its location on the aortic arch's concavity and its distance from the left subclavian artery [266,328]. The latter was associated with a more complex clinical course. In summary, there is no consensus. It has been hypothesized that a single proximal entry tear is linked to increased false lumen pressurization and a higher risk of growth and adverse events [264]. Results are however inconsistent [315,329–331]. The number of intimal tears was found to be protective against aortic growth in two studies [309,332]. In addition, Sailer et al. [270] discovered that the number of intercostal arteries was also protective against adverse events, possibly due to natural fenestrations. Earlier studies, on the other hand, reported conflicting findings, with the number of branch vessels as predictor for higher mortality and complications [321,333–335].

An alternative methodology for evaluating aortic dilatation was introduced by Kinoshita et al. [336]. They measured the outer media thickness of the false lumen using microscopic images of specimens taken intraoperatively (see Fig. 13) and assessed its influence on secondary dilatation of the downstream aorta, similar to the investigation on intramural hematoma in 2005 [337]. Of the 238 investigated patients, 129 met the inclusion criteria of this study: DeBakey type I dissection with a patent false lumen, histopathological examination of the full-thickness aortic wall, and at least one follow-up CT scan more than three months after the surgical procedure. The aortic wall specimens were stained with hematoxylin and eosin and Elastica van Gieson stains, and the thickness was measured by two physicians at eight equally distributed locations, from which average values were calculated. Overall, they noted two important findings. First, there were considerable variations in the thickness of the outer media ranging from 0.04 to 0.51 mm, with the distribution shown in Fig. 13(b), in a ortic wall specimens obtained during surgery for acute a ortic dissection. Second, the growth rate of the distal remaining dissecting aorta was inversely correlated with the thickness of the outer media (see Fig. 13(c)). In particular, patients in the lowest tertile of outer media thickness (0.04 to 0.15 mm) exhibited remarkable dilatation within one year in the early postoperative period, with about 5 to 8 mm/year growth observed in the first year, while the change after that was only 1 to 2 mm/ year. It was also found that the risk of aortic-related events is significantly higher in patients with an outer media thickness of 0.04 to 0.15 mm. They later compared the location of the false lumen within the medial layer between an acute intramural hematoma and an acute aortic dissection and showed that the acute intramural hematoma group had significantly thinner outer media thickness than the acute aortic dissection group [336].



**Fig. 13.** (a) Representative examples of a thin and a thick outer media (original magnification 40x, Elastica van Gieson stain). (b) A histogram of measured outer media thickness showing a right-skewed distribution normalized by square-root transformation. (c) Scatter plots illustrating the inverse relationship between outer media thickness and total annular growth rate in the middle descending aorta. Figure is adapted from Kinoshita et al. [336] with permission from Elsevier.

Recently, Houben et al. [338] have expanded the analysis of aortic conditions by going beyond measuring maximum aortic diameter. They used available CTA scan images to correlate 3D aortic growth and cyclic transmural wall stress with the location of intimal tear formation. This study retrospectively identified three patients with type B dissection and connective tissue disorder, each with two available ECG-gated pre-dissection CTA scans and no surgical repair during the pre-dissection interval. 3D aortic growth between two pre-dissection clinical CTA studies was quantified using vascular deformation mapping, a method previously introduced by [310,311]. Vascular deformation mapping involves three primary steps: segmentation, image registration, and deformation quantification. This technique may be more suitable for investigating the relationship between aortic dimensions and the risk of aortic dissection because it is less prone to measurement variability and can represent the 3D nature of growth. While it works well for small changes in aortic dimension, the predictability for large changes is not yet reliable, which is why short time intervals between followup examinations are required. In addition, finite element models of the structural domain were developed to estimate the cyclic transmural wall stress in patient-specific baseline CTA geometries (Section 8.1). The findings showed that focal areas of growth and transmural wall stress co-localized with the site of intimal tear formation. This supports the hypothesis that the risk of dissection is highest at sites where inadequate wall integrity (e.g., growth) and abnormal wall stress intersect. These hypothesis-generating results suggest a new analytic avenue for a more sophisticated assessment of the factors leading to initiation of dissection in patients with connective tissue disease. In particular, the application of vascular deformation mapping emerges as a promising approach that can not only enhance existing risk-stratification techniques but also deepen our understanding of aortic dilatation throughout disease progression, as recently demonstrated in Marway et al. [339].

In summary, the initial aortic size in the acute phase – typically measured either as the total aortic diameter or the false lumen diameter – is the most extensively used morphological parameter to monitor false lumen dilatation. Interestingly, a large initial false lumen diameter consistently predicts the risk of adverse events and mortality in both early and late stages. Notably, multiple studies support a 40 mm threshold value, with initial aortic diameters above this value independently predicting adverse events. While less studied than the initial aortic diameter, the initial false lumen size also appears to be a significant prognostic factor, with reasonably consistent results across studies. Various methods have been employed to measure the initial false lumen size, yielding partly contradictory results, as seen in Fig. 14. Established threshold values for the initial false lumen include a diameter of 22 mm and an area of 922 mm<sup>2</sup>. Due to their predictive capabilities, diameter plots have already been partially integrated into clinical practice [340,341]. They are used for long-term surveillance of aortic dissection patients to visually compare complex features and assess the associated risk of aortic dissection, as exemplified by Bäumler et al. [342], particularly due to their simplicity. In addition to the aortic diameter, they can be augmented to show features such as blood supply to the aortic branches or previous treatments. Moreover, compression of



**Fig. 14.** Illustration highlighting the difficulties in measuring the size of the false lumen for predicting patient risk, and the resulting variations in assessing potential risks. The false lumen size is considered as a predictor of lower risk according to Song et al. [307] ((a); initial false lumen diameter of <22 mm), Evangelista et al. [264] ((a); true lumen diameter is greater than one-fourth of the total diameter), Tolenaar et al. [322] ((b); true lumen circular shaped), and Chang et al. [321] ((c); initial false lumen area of <922 cm<sup>2</sup>). Conversely, it is an indicator of higher risk according to Sailer et al. [270] ((d); angular extent of initial false lumen of >249%). Figure is reprinted from Spinelli et al. [15] with permission from Elsevier.

the true lumen by the false lumen indicates false lumen pressurization, which is thought to promote aortic growth and complications. However, the significance of other factors remains controversial, especially because measuring these factors consistently is challenging. In the following sections, the impact of the flow pattern and thrombus formation in the false lumen on the onset of dilatation post-dissection onset will be explored. It is important to note that both depend heavily on the number, size, and location of intimal tears.

*Elongation, tortuosity, and angulation.* In contrast to previously discussed aortic dilatation, the roles of elongation, tortuosity, and angulation in aortic dissection remain largely uncertain. However, these factors have the potential to be risk factors. It is still unclear to which extent geometric remodeling and growth prior to dissection should be considered risk factors, especially when compared with the remodeling that occurs post-dissection. This type of geometric remodeling is frequently observed in clinical settings but has not been studied adequately, partly due to challenges in measurement. It is widely believed that the aorta, like other large elastic arteries, grows in the axial direction, meaning that it elongates, developing increased tortuosity and a pronounced angulation with age. Studies have reported an axial growth of 0.07 to 0.09 mm/year [168,343–345]. Aortic elongation, similar to dilatation, may also contribute to increased tortuosity and changed angulation, likely resulting from a sustained loss of elastic fibers and, consequently, a decrease in arterial wall elasticity. This phenomenon may be attributed to the aortic tissue experiencing 'material fatigue' with advancing age, leading to fragmentation and disorganization of elastin fibers and elastic lamellae [343] or adverse mechanobiological responses [346]. Hence, many of the subsequently presented studies discuss geometrical remodeling with respect to aging effects. This is also because the role of elongation, tortuosity, and angulation in aortic dissection has still gained little attention as a risk factor for dissection onset or mortality.

Due to the non-intuitive nature of the measurements, a variety of techniques exist for studying these anatomical features [347]. Therefore, it can be difficult to compare absolute values from different studies. Aortic length is often defined as the centerline distance between two corresponding planes. The axial length of the true and false lumina in the descending aorta are determined by establishing their respective centerlines. In contrast, aortic tortuosity can be calculated by dividing the centerline length by the linear distance

between the first and last points in each aortic segment. Finally, aortic angulation can be defined as the angle formed by tangent lines drawn along the first and last points of the aortic centerline in each segment.

In the 1950s, Dotter et al. [348] conducted one of the first studies on the length of the thoracic aorta, as there had been no previous reports regarding this measurement. The study aimed to investigate the relationship between aortic length and factors such as age, arteriosclerosis, syphilis, and hypertension. The authors concluded that the normal length of the thoracic aorta spans a relatively wide range, and despite the influence of age, no significant correlations were observed. Sugawara et al. [343] conducted a study on aging with 256 healthy adults aged 19 to 79 years. They discovered that the length of the ascending aorta increased with age, while the length of the descending aorta exhibited no correlation with age. Moreover, elongation of the ascending aorta was associated with increased pulse wave velocity in the aorta, suggesting that elastic properties and local pulsatile pressure might play a causal role. These findings were also observed in adults without apparent cardiovascular disease. Morrison et al. [239] also investigated age-related effects by conducting a study on 14 patients aged 35 to 80 years who had no visible aortic pathology. They measured the cyclic strain of the arch and descending thoracic aorta. Cyclic axial strain was found to decrease by 50% with increasing age, from  $2.0 \pm 0.4\%$  in the younger group to  $1.0 \pm 1\%$  in the older group. However, arch length increased by 14% from the younger to the older group. Therefore, the thoracic aorta elongates with increasing age and has a significantly lower axial strain.

A series of studies by Krüger and his colleagues examined various aspects of aortic length in patients with aortic dissection, predissection, and healthy controls. In the first study [349], they identified a positive correlation between age and aortic length in healthy controls. Dissected aortas and aortas prior to dissection were significantly elongated compared with healthy counterparts, suggesting that aortic length parameters are related to morphological changes before or during aortic dissection. The midline length of the healthy ascending aortas was 71 mm, while the same values for thoracic aortas pre- and post-dissection were 81 and 92 mm, respectively. The second study [350] revealed that the ascending aorta and the aortic arch were significantly longer in both thoracic aortas pre- and post-dissection compared with control aortas. Aortic morphology correlated with age but not with parameters of body size. Similarly, in the control aortas, the central line distance from the aortic valve to the brachiocephalic trunk was 93 mm. In thoracic aortas pre-dissection, it was 111 mm, and in thoracic aortas post-dissection groups. It demonstrated that the ascending aorta was significantly longer in all groups compared with the control group. However, the correlation between diameter and length of the ascending aorta indicated that both parameters must be examined separately. Finally, in Lescan et al. [352], they retrospectively compared a healthy control group and patients with type B dissection. The segment with the most considerable difference in elongation between the study groups was the non-dissected aortic arch of patients with type B dissection. Together, these studies highlight the importance of considering aortic length as a relevant parameter in the context of thoracic aortic dissection and related conditions.

Similar findings were reported in two subsequent studies, which first investigated the role of aging with respect to aortic elongation and then compared healthy control patients with aortic dissection patients. Adriaans et al. [353] conducted a cross-sectional case study with 210 consecutive patients. They discovered that the length of the thoracic aorta was significantly related to age, increasing by 59 and 66 mm in males and females, respectively, over the age range from 20 to 80. The lengthening was most pronounced in the proximal descending aorta, which displayed an almost 2.5-fold length increase during a person's lifetime. The lengthening of the thoracic aorta was accompanied by marked changes in geometry, with the aortic apex shifting to a more distal position in elderly patients. Later, in a study by Heuts et al. [354], 250 patients were included to investigate the relationship between acute type A dissection and aortic dimensions. The ascending aorta was found to be longer in acute type A dissection patients compared with healthy controls. However, no differences were observed in the lengths of the aortic arch and descending aorta. Moreover, the results revealed that the ascending aortic length and diameter were independent predictors for acute type A dissection.

Analogously, Wu et al. [355] discovered that an ascending aortic length of  $\geq$ 13 cm was associated with an almost fivefold higher average yearly rate of aortic adverse events compared with an ascending aortic length of <9 cm. They suggested an aortic elongation of 11 cm as a potential intervention criterion for acute type A dissection. The mean estimated annual aortic elongation rate was 0.18 cm/ year, and aortic elongation was found to be age-dependent. Eliathamby et al. [356] demonstrated that the ascending aorta was longer in patients immediately following acute type A dissection, in the pre-dissection group, and in the aneurysm group compared with controls. Furthermore, Sun et al. [347] found that the length of the ascending aorta and the total aorta was significantly longer in the acute type B dissection group, while in the acute type B dissection group the length of the total descending aortic lumen was significantly longer than in the control group. In a follow-up study, Sun et al. [357] then divided 112 patients into three groups: with normal aortic diameter, with ascending aortic aneurysm, and with type A dissection group was 14.5 mm longer than that in the aneurysm group. Interestingly, in the dissection group, there was no correlation between the length of the aorta and age, and the median age was younger than that in the aneurysm group. Cao et al. [358] showed that the ascending aorta and the aortic arch lengths in acute type B dissection increased significantly compared with controls. Shirali et al. [359] compared hypertensive patients with and without aortic dissection and found that the length of the proximal and entire aorta was greater in the aortic dissection group.

Aortic tortuosity and angulation have been the focus of numerous studies, investigating their significance in the context of aortic dissection in various patient populations. Gode et al. [360] found that among patients undergoing ascending aortic surgery due to aneurysm or aortic dissection, ascending aortic angulation was significantly less in aortic dissection patients compared with aneurysm patients. In a separate study of patients with Marfan syndrome, Franken et al. [361] observed that increased aortic tortuosity was an independent predictor of clinical events, including aortic dissection. Focusing on patients with acute type B dissection, Cao et al. [358] found that aortic arch tortuosity was increased and angulation was more pronounced in this group compared with controls. Aortic arch tortuosity was also identified as a significant factor in bicuspid aortic valve patients by Alhafez et al. [362], as it may help identify those

at increased risk for aortic dissection. Other studies have demonstrated correlations between aortic tortuosity and various patient groups. Jie et al. [363] found a moderate correlation with ascending aortic tortuosity in acute type A dissection patients, while Shirali et al. [359] noted increased aortic tortuosity in hypertensive patients with acute type B dissection compared with normotensive and hypertensive patients without aortic dissection. Rylski et al. [364] discovered that acute aortic dissection led to increased tortuosity in the originally non-dissected ascending aorta, with this increase observed only in the ascending aorta and not the proximal descending aorta or arch. Finally, Sun et al. [347] compared acute type B dissection patients with a control group and found smaller angulations in the ascending aorta and aortic arch, along with significantly increased tortuosities in the aortic arch and total aorta. Although the mechanisms of tortuosity remain to be determined, complex mechanical and mechanobiological factors likely contribute [346]. Regardless, these studies underscore the relevance of examining aortic tortuosity and angulation as relevant factors in the context of aortic dissection and related conditions [365].

In addition to diameter, other factors such as an elongated aorta, increased tortuosity, and a smaller angulation angle may significantly contribute to the pathophysiology of aortic dissection. These factors could potentially be age-related risk factors for aortic dissection, as demonstrated by numerous studies, particularly in the ascending aorta and the aortic arch, which have been observed to elongate. As discussed by O'Rourke et al. [366], it is noteworthy that the aortic arch is anchored to surrounding structures by three branches, while the proximal descending aorta is tethered to the spine by intercostal arteries. In contrast, the proximal aorta connects to the heart, which moves within the pericardial sac during the cardiac cycle. As the ventricle contracts, the aortic valve annulus moves downward, leading to axial movement of the most proximal aorta. This suggests that not all parts of the aorta grow similarly. Due to the tethered descending aorta, the aorta might be more prone to developing a lower angulation angle and increased tortuosity to balance elongation. In fact, ascending aortic tears are typically transverse rather than axial, suggesting greater disruptive strain in the axial direction than in the radial direction. Additionally, as reiterated by numerous studies, changes in aortic dimensions might be associated with increased aortic pulse wave velocity. Current guidelines recommend treatment based on hinge points in aortic diameter as the sole criterion, which might be insufficient [367,368]. Recent results also suggest that volumetric distensibility, which is correlated with pulse wave velocity, can serve as a new metric in the risk-stratification process, as it strongly correlates with aneurysm failure risk [63]. However, the development of a comprehensive global risk score with threshold values is unlikely to encompass all aspects of this complex disease. Moreover, there is no consensus on how to accurately measure these quantities, particularly tortuosity and angulation, making it difficult to compare absolute values between studies.

# 3.2. Pathophysiology of mural remodeling

# 3.2.1. Degenerative aortic histopathology

Idiopathic cystic medial necrosis according to Erdheim-Gsell. In 1929, Jakob Erdheim reported an unusual disease in his analysis of a patient's case [369]. This specific case involved a patient with a dissecting aneurysm without any noticeable atherosclerosis. Due to the non-atherosclerotic nature of the intima, it was delicate, smooth, and translucent, allowing detailed study of the aortic media. Intriguingly, Erdheim initially misinterpreted the described symptoms as signs of syphilis.

Erdheim, an Austrian pathologist, documented two primary histological findings predominantly found in the outer media. First, he noted focal areas of necrosis. Second, he identified a degradation of the medial tissue, located primarily in the outer half of the media, while the remainder of the wall remained relatively normal. He reported that the latter observation was related to degenerative changes characterized by sharply defined patches of different appearance. These patches often appeared in a linear pattern. Moreover, the typically clearly visible elastic lamellae were either completely missing or barely visible, as demonstrated in Fig. 15(a). The loss of collagenous tissue occurred gradually, possibly resulting in the absence all three main components of the media. Interestingly, in certain cases the SMCs persisted until the final stages of the disease. It has also been observed that the fluid contents of these cavities seep into the surrounding tissue crevices. The second observation, necrosis, resulted in a conspicuous absence of nuclei. In areas of advanced changes, necrosis played a significant role, resulting in widespread cell apoptosis and tissue degradation. Beyond gross changes, necrosis gave the impression of a uniform reduction in the number of cell nuclei, suggesting a slow, progressive process of SMC loss that eventually led to a complete absence of nuclei.

The relationship between necrosis and tissue degradation seemed complex, and was not entirely clear to Erdheim. Nevertheless, he proposed two possible scenarios. In the first case, necrosis could precede and lead to subsequent tissue degradation. This hypothesis was supported by the fact that the process of necrosis could be traced microscopically. However, there were many cases without visible signs of necrosis. In several cases, tissue degradation without preceding necrosis was observed. The localized loss of non-necrotic media supports the concept that in a given area, elastic fibers and collagenous tissues often disappeared first, leading to cavity formation, while SMCs remain preserved. In conclusion, tissue degradation can occur either secondary to necrosis or primarily parallel to it.

Erdheim also examined the remaining inner media, intima, adventitia, the presence of vasa vasorum, and the area of dissection. In fact, necrosis played a less prominent role in the inner media. Areas of media loss were replaced by collagen deposits; these zones were smaller but more common. Elastic lamellae maintained a parallel orientation at specific intervals, but they thinned significantly due to subtle tissue degradation. Elastic fibers disappeared first without preceding necrosis, followed by SMCs. Concurrently, the connective tissue became denser, indicating an increase in collagen deposit. What remained was a connective tissue scant in nuclei, devoid of a layered arrangement. The partially restored media, rebuilt through regenerative processes, contains the same components as the original media but in an irregular composition or ratio. In contrast, the adventitia and intima were not primarily affected by the above-mentioned pathological changes, but rather secondarily. The proliferation of SMCs, which typically occur in diseases such as syphilis,



**Fig. 15.** Two histological images showcase the main findings of Erdheim, denoted as cystic medial necrosis. (a) A histological image of medial necrosis features the inner elastic intimal layer ('a'), the markedly thin outer muscular intimal layer ('b'), and the inner and outer medial layers, ('c') and ('d'). Notably, the inner medial layer ('c') is abundant with cell nuclei, while ('d'), the outer medial layer, is necrotic, extending close to the adventitial layer ('f') and is replaced by degenerated tissue devoid of cell nuclei. Finally, ('e') represents the adventitia. (b) The distribution and condition of tissue in the medial layers, where ('M1') signifies the inner two-thirds of the original media, marked by several focal areas of cell apoptosis, ('a') and ('b'). ('M2') represents the outer third of the media, exhibiting regenerated tissue with mucoid accumulation ('c') and small cysts ('C'). The adventitia, labeled ('A'), is seen thickened and poor in nuclei. Figure (a) is adapted from Erdheim [369], while (b) is from Erdheim [370], with permission from Springer Nature, respectively.

was less pronounced. In the intima, hypertrophic SMCs were observed, arranged in an irregular pattern, indicating a regenerative process and the development of new or altered intima. Interestingly, the vasa vasorum was never found in the focal areas and was predominantly found in the outer medial layer, suggesting that it does not play a significant role. Finally, Erdheim studied the area of rupture or delamination. The tear zigzags through all wall layers, including zones of strong nuclear staining but significant medial degeneration, and through areas with necrotic foci. Examination revealed that the dissection was not immediate. In fact, the adventitia partially resisted and allowed the tear to penetrate easily.

In summary, the observed pathological findings results in reduced resilience of the aortic wall and a clear decline in its ability to perform the Windkessel function, as the degeneration of its two crucial components – elastic fibers and SMCs – is integral to the is integral to the recoil and contraction process. Elastic fibers recoil elastically due to their inherent restorative force when stretched – a purely mechanical process – while the actively stimulated SMCs perform the majority of the work driven in part by the nervous system. Moreover, Erdheim repeatedly emphasized that the described medial changes that lead to necrosis occur without an inflammatory reaction. This enigma was one of his main points of interest; he could not explain why there were no inflammatory reactions and associated cells despite significant changes. He was unsure of the causes of this novel disease and called it 'idiopathic medial necrosis'.

One year later, in 1930, Erdheim examined a second patient case [370]. Unlike the previous case, medial necrosis was less dominant. Instead, the presence of scattered mucoid accumulation was highly significant, leading to a degeneration with cystic changes. The appearance of distinct cysts, predominantly filled with mucoid, gave this case a unique characteristic, as shown in Fig. 15 (b). These cysts were mainly found in the medial layer of the tissue. As a result, he extended the previously introduced term to 'idiopathic cystic medial necrosis'. As a side note, this term was originally coined by Babes and Mironescu [371] in 1910, but received little interest [271], until after 1928 it was Gsell [372] and then Erdheim who made this term public.

While Erdheim's terms garnered significant interest and acceptance in the decades that followed, there were later considerable doubts as to whether this term accurately described the lesions or if cystic medial necrosis is in fact the most common lesion found in dissecting aneurysm [373]. The term 'cystic', proposed by Schlatmann and Becker [374,375] in 1977, seemed inappropriate because the lesions do not form true cysts with a distinct lining. Instead, they represent structural gaps or voids in the media, which become filled with semi-fluid ground substance due to intramural tension. In addition, the term 'medial necrosis' might not be entirely justified because, as Erdheim himself conceded, necrosis is rarely encountered in the lesions. Unfortunately, no specific terms have as yet gained full acceptance. Meanwhile, the term is often replaced by 'medial degeneration'. Because cystic medial degeneration or necrosis indicates disparate histopathologies in different studies, it is often not possible to distinguish subtle histopathological differences between diseases [376]. Despite the different opinions based on the above observations, the term 'cystic medial necrosis' is still overwhelmingly used today.

While cystic medial necrosis has been identified in histopathological studies of tissue from patients with aortic dissection, it is certainly not unique to this condition. For example, a clinicopathological study of 513 consecutive ascending aortic surgical specimens revealed the presence of cystic medial necrosis in 40.7% of patients and in 56 (or 51.4%) of the 109 patients diagnosed with aortic

dissection [377]. It was associated with connective tissue disorders, such as Marfan syndrome [378,379] and Ehlers-Danlos syndrome [380], or was found in patients with annuloaortic ectasia [381]. Furthermore, in patients with aortic stenosis [382] and, in addition, in patients without Marfan syndrome, cystic medial necrosis is more commonly found in geriatric and hypertensive individuals [383]. Several authors reported it in congenital disorders, instances of coarctation of the aorta and bicuspid aortic valve associated with cystic medial necrosis [379,384–386]. Other reported that in patients receiving surgical treatment for aortic dilatation with aortic valve disease, the incidence of cystic medial necrosis was 100% [387]. It is notable that patients with bicuspid aortic valve and aortic aneurysm exhibit cystic medial necrosis unrelated to age [388]. In summary, cystic medial necrosis is typically due to the fragility of the aortic wall due to degenerative changes, which may also be related to the normal aging process [376]. Nevertheless, it remains controversial whether cystic medial necrosis directly leads to aortic dissection, is merely a coincidental finding in the diseased aortic wall, or develops as a consequence of the dissection [389]. For a more detailed exploration of cystic medial necrosis, the reader is referred to [390], among others.

*Medial degeneration.* In 2016, Halushka and his colleagues, all members of the Society for Cardiovascular Pathology and the Association for European Cardiovascular Pathology, made efforts to generalize the aforementioned histopathological terms [66]. Their use is now recommended for general surgical pathology reports and for categorizing histopathologies. Moreover, since the degree of medial degeneration can vary widely, ranging from minor to severe destructive alterations of the lamellar units in the aortic wall, a grading classification was introduced. The grading classification of these lesions is crucial as medial degeneration can vary significantly between patients. Overall, medial degeneration is a consequence of individual histopathological degenerative lesions.

First, alterations in elastic fibers stand out as a prominent histopathological finding [93,391–401]. Specifically, reports indicated fragmentation or loss of elastic fibers. This implies the creation of increasingly extended translamellar spaces due to the absence of elastic fibers, along with an increase in gaps within the elastic fiber lamellae. This degradation of elastic fibers was often reported to be patchy in nature. In addition, thinning of the elastic fibers and noticeable disorganization was observed, the latter being due to the locally non-parallel arrangement of the elastic fibers. Examples of these findings are shown in Figs. 16(a) to (e) and Fig. 17, and were often observed in the context of SMC disorganization or local accumulation of GAGs.



**Fig. 16.** Demonstration of (a) fragmentation, loss, and disorganization of elastic fibers that no longer follow a strict circumferential course and may be oriented perpendicular to the lumen wall (400 x, Movat pentachrome strain); (b) patchy-like loss of elastic fiber areas with possible complete loss of elastic fibers (400 x, Movat pentachrome strain); and (c) thinning of elastic fibers (20 x, Verhoeff-van Gieson strain). 3D architecture of the elastic fibers in (d) a normal subject and (e) an aortic dissection patient in which the aorta was treated with formic acid to remove components other than the elastic fibers and then examined using scanning electron microscopy. The top image represents the intimal side, the bottom the adventitial side (scale bar:  $20 \mu$ m). Figures (a) to (c) are adapted from Halushka et al. [66] with permission from Elsevier, while (d) and (e) are from Nakashima [400] (licensed under CC-BY-NC-SA 4.0).



**Fig. 17.** Scanning electron microscopy-based evaluation of elastic lamellar sheets and other extracellular matrix components in control tissue and tissue from dissection patients, with native tissue in the top row and decellularized tissue in the bottom row. (a) A representative scanning electron microscope image of control ('NA') tissue showing elastic fibers (white arrows), cellular material (white asterisks), and interlaminar elastin fibers (white arrowheads). Two representative scanning electron microscope images of dissection tissue illustrate (b) the dissection lamellar sheet (white arrow) before and after decellularization and (c) the extracellular structure adjacent to the dissection lamellae with white arrows pointing to melded thin fibers (bottom row). In the dissected media, a completely destroyed structure can be seen in the scanning electron microscope images. Thin individual extracellular matrix components cannot be clearly distinguishable and appear fused together, even with the elastic lamellae. Removal of the cell leads to the complete collapse of the elastic fiber structure. Figure is adapted from Mimler et al. [401] (licensed under CC-BY 4.0).

The second finding is associated with apoptosis and dysfunction of SMCs. This loss of cell nuclei can manifest in patchy (Fig. 18(a)) or band-like patterns (Fig. 18(b)), a finding also known as 'laminar medial necrosis', see, e.g., [402–404]. In fact, this condition was not only observed in aortic dissection patients, but also in non-dissected and normal aortas [403–405]. It was strongly associated with aging and ischemic injury responses following aortic dissection [404,406]. This finding is similar to, but not identical with, 'laminar medial collapse', where the lamellar units collapse together, as shown in Fig. 18(c). In addition, a disorganization of SMCs was found, which is characterized by the non-parallel arrangement of the SMCs in the media, leading to focal disarray.

Third, medial degeneration is also associated with remodeling of fibrillar collagen, possibly due to SMC dysfunction. In particular, a focal increase in collagen fibers was observed, leading to areas of substitutive fibrosis or a widening of intralamellar spaces in the media, a condition known as medial fibrosis, see Figs. 19(a) and (b). Additionally, collagen fibers often appeared thinner and disorganized. For more details, the interested reader is referred to the literature [399,407–411].

Finally, the accumulation of GAGs, or mucoid extracellular matrix, has been frequently observed in histopathological investigations; a unique feature that continues to be recognized as singularly characteristic of aortic aneurysms and dissections, see [374,375,412]. However, this occurrence was seen across a wide range of ages and in diverse disease origins [376,413–423]. For example, Cikach et al. [93] reported PG accumulation, including aggrecan and versican, in thoracic aortic aneurysms and dissections (Fig. 20(b)).

In 2013, Humphrey described for the first time a possible biomechanical consequence of pooled GAGs/PGs [103] in the context of aortic aneurysms and dissections. Accumulation of GAGs/PGs in the media, particularly in compromised regions of elastic fibers and SMCs, is frequently associated with elevated TGF- $\beta$  activity. As previously mentioned, dysregulated TGF- $\beta$  activity can lead to increased production of GAGs/PGs in the arterial wall, similar to what occurs in other tissues. Furthermore, the enzymes MMP-3 and -7, known for their role in degrading elastin and collagen III, have been associated with these accumulations and may play a role in medial degeneration. It was hypothesized by Humphrey [103] that the accumulation of GAGs, along with the associated Donnan swelling pressure that arises from negatively charged GAGs – a phenomenon well-documented in cartilage tissue [424] – could modify the intramural mechanical stress field, potentially causing aortic medial delaminations. For more details, the interested reader is referred to [103]. The swelling pressure, possibly between 16 and 170 kPa [425,426], can be comparable to or even significantly higher than standard compressive stresses, which are typically in the range of -10 to -16 kPa. Both experimental models of dissecting thoracic aneurysms and human studies indicated that regions with accumulated GAGs/PGs had larger separation distances between



**Fig. 18.** Histological images of (a) patchy and (b) band-like loss of SMC nuclei (100 x, Movat's pentachrome stain), and (c) laminar medial collapse in conjunction with a loss of SMCs in the lamellar units (200 x and 160 x, H&E stain). Figure is adapted from Halushka et al. [66] with permission from Elsevier.



**Fig. 19.** Histological images showing medial fibrosis: (a) intralamellar accumulation of collagen (royal blue) preserves the integrity of the lamellar unit and (b) translamellar fibrosis disrupts the lamellar unit and may be associated with disorganization of SMCs (100 x, Masson's trichrome stain). Figure is adapted from Halushka et al. [66] with permission from Elsevier.

elastic lamellae and were associated with marked pooling of GAGs/PGs [420,427], as depicted in Fig. 20(a). This swelling could initiate the separation of elastic lamellae [428], potentially initiating dissections and disrupting matrix turnover and wall homeostasis by affecting cell signaling pathways. The structural integrity of the wall might further deteriorate due to elastin degradation caused by swelling. Furthermore, swelling-induced disruptions of smooth muscle interactions with 'radially-directed' elastic fibers, caused by swelling both before and after local dissection, could have negative effects on cellular mechanotransduction [394] and potentially lead to harmful alterations in the signaling pathways of the cells as well as the matrix turnover and the wall homeostasis [429]. Similar concerns were raised regarding disrupted mechanotransduction in dysregulated smooth muscle contractility [416]. Additionally, the production of degradation products from swelling-induced damage to inter-lamellar elastic fibers could destabilize wall homeostasis and exacerbate the loss of wall structural integrity [430].

Evidently, the histopathological findings associated with medial degeneration are not solely linked to aortic dissection. This suggests that there might not be a single histopathological finding that leads to aortic dissection in all cases. Instead, the specific severity and distribution of focal lesions might determine the initiation of dissection. However, the accumulation of GAGs and the associated mechanical impact could play a significant role. Some emphasized that it remains unclear whether PGs/GAGs contribute to disease onset or whether they react to early disease-related changes after the initial dissection event [431]. Notably, early studies, such as those by Erdheim, highlighted the non-inflammatory nature of this condition. However, assessing the role of inflammation prior to dissected, thus providing insight into the causes of disease. On the other hand, it is more common to acquire human tissue after dissection, which is likely to exhibit substantial inflammatory processes as part of the injury response. This topic will be discussed in more detail in Section 3.2.3.

Initiation of dissection. The classic definition of the initiation of aortic dissection is as follows: aortic dissection is initiated as a small tear in the intimal (inner) layer of the aortic wall before it extends to the medial layer, causing a wall separation. Subsequently, the tear propagates along the axial direction of the medial layer resulting in a false lumen. The tear can be triggered by focal defects in the aortic wall, as previously described in this section, making it prone to delamination. Once dissection is initiated, it tends to propagate more easily within elastic lamellae than across the lamellae [432,433], a phenomenon clearly governed by the microstructure. However, there is a second scenario that has received significant attention in recent years. Instead, an intraluminal hematoma might cause the delamination of the medial layer, potentially initiating aortic dissection, which subsequently triggers an intimal tear at a later stage of



Fig. 20. (a) Schematic representation of (i) a normal aortic medial lamellar unit composed of paired elastic laminae (top and bottom) encasing a SMC, and collagen fibers, adhesion molecules (e.g., fibronectin), and GAGs (not to scale). Also illustrated are thin 'radially oriented' elastic fibers (composed primarily of elastin and fibrillin-1 associated microfibrils) that may provide direct mechanical connections between the elastic laminae and the SMC, thus extending mechanosensory capabilities beyond the typical cytoskeletal ('CSK') - integrin - extracellular matrix (fibronectin/ collagen) axis. Negatively charged GAGs may attract water, potentially contributing to normal intralamellar pressure that could help maintain tension in the thin elastic fibers. (ii) Previous image is accompanied by a histological image of the normal medial layer of the human ascending aorta with GAGs (blue) (Alcian blue stain), where the black arrow points to a normal structure, primarily consisting of SMCs, collagen, and modestly distributed GAGs. (iii) Schematic representation conveys the central hypothesis: localized accumulation of GAGs, on the right side of the medial lamellar unit, leads to increased Donnan swelling pressure. This may separate the elastic laminae and possibly disrupt connections between the SMCs and either thin elastic fibers or the collagenous matrix. Such effects could initiate local delamination and/or an altered mechanosensitive cellular response, resulting in dysregulated wall homeostasis. (iv) Medial layer of a human ascending aorta, stained with Alcian blue (which colors GAGs blue) for an aneurysmal wall. Note the significant accumulation of GAGs in the latter, indicated by the black star, and the absence of SMCs or elastic fibers within these GAG pools. (b) Areas of medial degeneration in cases of (i, iii) idiopathic aortic aneurysm and dissection, as well as (ii, iv) Marfan syndrome, are shown with and without intense aggrecan and versican staining (red). These areas are identified here as regions with fragmented or absent elastic fibers (stained green), and few SMCs (nuclei stained blue) in aortic aneurysms and dissections (scale bars: 200 µm). Figure (a) is adapted from (i, iii) Humphrey [103] with permission from Karger Publisher and (ii, iv) Borges et al. [417] with permission from Elsevier, while (b) is from Cikach et al. [93] with permission from American Society for Clinical Investigation.

the disease [103]. The aforementioned accumulation of GAGs, along with the related delamination process, may be of significant importance for this hypothesis. This controversy demonstrates that although we know much, this remains an unresolved question. Is the initial event a primary tear of the intima leading to secondary dissection of the media, or is the initiation of dissection due to an initial defect or delamination, leaving the wall vulnerable to subsequent dissection and possible rupture? Histological examination can provide evidence for both scenarios, suggesting that both cases could potentially occur. The wall appears to be riddled with various material and geometrical discontinuities, which could lead to the delamination of the media or even a fatal rupture of the aortic wall, depending on the size, distribution, and severity of the focal lesions. Material or geometric discontinuities can result in stress concentrations, which can typically increase local stress two- or threefold, initiating aortic dissection, as demonstrated later in Section 5.3.

Aside from the two primary dogmas determining the onset of dissection, another hypothesis has been notably overlooked. This hypothesis originates from the observation that vasa vasorum dysfunction due to malperfusion or leakage can accelerate atheroma formation, referred to as 'outside-in progression of atherosclerosis' [434,435]. Obstruction of the vasa vasorum is believed to result from viruses, bacteria, and tiny dust particles. This is likely influenced by risk factors such as hypertension, smoking, and age. It is crucial to understand that the rupture of the vasa vasorum and subsequent bleeding into the media play a role in the development of intramural hematomas, which could potentially lead to a subsequent dissection. Generally, malperfusion of vasa vasorum, exacerbated by the compressive forces of hypertension, can induce hypoxic conditions in the outer media. Such hypoxia is thought to stimulate angiogenesis of the vasa vasorum, a mechanism that yields new, potentially fragile and leaky neovessels, especially in the nascent stage. If the outer media remains in its hypoxic state, it can quickly become ischemic and promote necrosis and medial degeneration, which could be different phases of the same condition: a vasa vasorum disease. This hypothesis rests on five key observations: (i) both conditions manifest at identical anatomical sites (ascending aorta and aortic arch); (ii) vasa vasorum dysfunction has been identified in both diseases; (iii) there have been instances where an intramural hematoma evolves into an aortic dissection, suggesting a two-stage disease; (iv) both conditions are characterized by medial necrosis and inflammatory processes; and (v) they share several common risk factors.

Although the precise initiation event remains elusive, it appears improbable that the majority of aortic dissection cases are triggered by vasa vasorum dysfunction. Undoubtedly, a rupture in the vasa vasorum can lead to intramural hematoma, which eventually results in an aortic dissection if an intimal tear develops. However, a rupture in a vessel with a radius in the micrometer range is mechanically unlikely because the wall tension is miniscule; it should also be taken into account that the vessel is surrounded by a high pressure environment. Although this can occur, it is probably not the cause for the majority of a dissections. Moreover, while it is undeniable that aortic dissection and intramural hematoma are similar diseases, they are in general not related to the vasa vasorum. Nevertheless, the underlying condition points to medial degeneration with obvious loss of cohesive forces between the medial layers. However, the arguments postulating vasa vasorum as the primary cause are often weak and rarely discussed in the literature; one of the few reports on this was published by Osada et al. [439], and such arguments have never been conclusively proven. Moreover, if bacteria invaded the aortic wall, the likelihood of aortitis [440] or mycotic aneurysms [441] would be greater. Only a few studies, mainly related to coronary arteries, provide interpretable results in this context [442,443].

# 3.2.2. Layer-specific wall remodeling

*Dissection flap.* The dissection flap separates the true from the false lumen and balances the pressure differences between them. Especially in acuity, it moves over the cardiac cycle driven by the pressure difference, as described above. Typically, the dissection flap consists of the intima and part of the media. Immediately following dissection, one might anticipate that the dissection flap and the outer wall of the false lumen would be roughly equal in thickness to the outer wall of the true lumen. However, CT imaging often reveals a thicker dissection flap, especially in chronic aortic dissection. Due to remodeling processes, the dissection flap gradually becomes thicker and loses mobility as the aortic dissection progresses [14].

Studies have reported that the thickness of the dissection flap can sometimes exceed the thickness of the outer wall of the true lumen. This phenomenon may be attributed to the recoil of elastic fibers [275,282] due to the disappearance of transmural pressure in the dissection flap, leading to a loss of circumferential tension. However, axial tension may not necessarily be affected. It has also been shown that the thickneed dissection flap results from the formation of a neointima, see, e.g., [258]. The neointima is a newly formed intimal layer in the false lumen that develops in response to the absence of an intima in the false lumen, similar to the reaction to injuries [444]. Neointima formation begins within a few hours of the dissection onset and involves the replication of SMCs in the media and their migration through the internal elastic lamina to form the neointima alongside an ECM. This process can significantly affect the thickness of the flap. An example of neointimal formation in the false lumen is presented in Section 4.1 (Fig. 35(b)).

Peterss et al. [14] characterized the increase in dissection flap thickness as an exponential decay over time, with the change in thickness decelerating at approximately 83 days and reaching a plateau at approximately 235 days after the onset of the aortic dissection (Figs. 21(a) and (b)). They estimated the average change in dissection flap thickness per year, which amounted to 1.2mm during the acute stage, 0.41 mm in the subacute stage, and 0.02 mm in the chronic stage. A thickened dissection flap is strongly associated with altered elasticity and mobility (Fig. 21(c)). The change in stiffness can be attributed to fibrosis, which is associated with decreased mobility. Furthermore, studies using CT and MRI have shown that, over time, the dissection flap loses its curvature and waviness, appearing more straightened [14,278,445,446]. While the dissection flap remains elastic and flexible during the initial days following the onset of aortic dissection, it becomes stiffer and thicker within the first few weeks and months. Karmonik et al. [447] demonstrated that this decreased motion is observed in both short-term and mid-term examinations, with high variability between patients in the short term and low variability in the mid-term. The short-term variability among patients is high but decreases during mid-term examinations. The pathological changes in the dissection flap are crucial indicators of chronic aortic dissection, but their systematic examination is still inadequate. Tang and Dake [283] discuss how the success of stent-graft deployment depends on an understanding of the dissection flap configuration. As time progresses, the dissection flap matures, becoming thicker and more stable. This increased stability facilitates the deployment of medical devices and reduces the risk of requiring surgical re-intervention. Bäumler et al. [448] point out that it is clinically important to determine whether the stiffness of the dissection flap and its reduced motion correlate with an increase in intraluminal traction. The pressure difference between the true and false lumina causes intraluminal traction, which can increase as the dissection flap becomes stiffer and can no longer balance the pressure difference. However, the detailed microstructure of the dissection flap remains largely unknown and has rarely been investigated.

*Outer wall of the false lumen.* The outer wall of the false lumen serves as a load-bearing structure and arises from the intact aortic adventitia after the initial delamination of the intima and variable portions of the medial layer within the aortic wall. The hemodynamics and resulting loading conditions can vary depending on the size of the false lumen, and the location, number, and size of intimal tears. In most cases, the maximum pressures in the false lumen are lower than in the true lumen. Post-dissection, it is often observed that the outer wall of the false lumen is thin and expanded. For example, Roberts [261] reported that the thickness of the outer wall of the false lumen is approximately one-quarter of the outer wall of the true lumen. However, this changes over time. An early response is neointimal formation in the false lumen, leading to thickening over time.

Due to altered mechanical stresses within the false lumen, the thin wall may respond to the changed loading conditions by undergoing remodeling, potentially becoming thicker. A chronic case example was shown in Section 3 (Fig. 9(a)). The restoration of false lumen wall strength and homeostasis is maintained by adventitial fibroblasts, which are sensitive to changes in wall tension. These mechanosensing fibroblasts respond with proliferation, increased synthesis, and cross-linking of collagen. This was observed at late surgical intervention as a strong, leathery thickening of the adventitial layer. Even a marginally inadequate tissue response to wall tension can initiate a vicious cycle in which any increase in false lumen diameter proportionally translates into ever-increasing wall tension, resulting in continuous growth of the aortic diameter over time. Once restorative capabilities are exhausted and wall tension


**Fig. 21.** (a) Absolute thickness and (b) thickening rate (mm/day) of the dissection flap, which follows an exponential decay function with a rapid decline in the initial thickening rate, a deceleration starting at 83 days, and a plateau after 235 days after start. (c) Dissection flap movement (mm) in type A and B dissection, with a gradual decrease in flap movement over time. Figure is adapted from Peterss et al. [14] with permission from Elsevier.

exceeds tensile strength, structural failure and aortic rupture follow. This occurs with increasing frequency when the maximum aortic diameters exceed 55 mm in men and 60 mm in women, respectively [34,301]. However, the effects of this remodeling on the microstructure remain largely unclear.

Recent biomechanical investigations on the outer wall of the false lumen showed an increase in collagen and a decrease in elastin due to fragmentation, see Fig. 22. When mechanically loaded, this results in an initially very compliant behavior due to the lack of elastin, which is responsible for the load-bearing at lower stretches. At higher stretches, the outer wall of the false lumen exhibits then a significant increased stiffness. Despite the lack of extensive mechanical and microstructural studies, the results of Amabili et al. [250] confirm this finding.

*Mechanical behavior*. Mechanical tests are required to determine the layer-specific mechanical behavior of tissue samples from aortic dissection patients. The currently available experimental studies on such tissue are summarized in Table 3. These studies involve uniaxial and biaxial extension tests and may selectively examine specific layers, including the outer wall of the true lumen, the dissection flap, and the outer wall of the false lumen, either individually or a combination thereof. This layer-specific approach is essential, as each layer may undergo distinct remodeling processes during the progression of an aortic dissection, influenced by the loads they experience and their unique microstructures, which may be altered by inflammatory responses triggered by associated wound healing. Notably, these studies fail to compare mechanical behaviors between tissue samples from acute and chronic patients, whose mechanical properties can differ significantly due to remodeling. Currently, the data primarily reflect passive behavior; active behavior has not yet been characterized. Given the rarity of data on the active behavior of the normal human aorta [450] and considering that tissue cells die within hours of death, it is unlikely that such data will be obtained from patients with aortic dissections in the near future. For more details on active behavior in healthy or normal human tissue, see, e.g., [451].

Although some review articles also included histological examinations, comprehensive information on fiber alignment and dispersion in dissected tissues is lacking. It should also be noted that some studies in Table 3 focus primarily on material or computational modeling and use a limited number of tissue samples to determine relevant material parameters [450,458]. Additionally, follow-up studies used published experimental data on dissected tissue to identify material parameters for specific constitutive models, as shown by [461,462]. Note also that we omitted a summary of delamination and failure tests on human and animal tissues, as these topics have been extensively summarized and reviewed in recent literature, see, e.g., [17–19]. It is important to recognize that these



**Fig. 22.** Biochemical analysis of the dissection flap ('FP'), outer wall of the true lumen ('TL'), and outer wall of the false lumen ('FL'): (a) overall percentages of elastin fragmentation, graded by age for each layer, were determined using criteria modified from Schlatmann and Becker [374], and, additionally, boxplots illustrate the statistical significance of (b) collagen levels, (c) GAG levels, (d) elastin levels, and (e) the ratio between collagen and elastin across patient groups and sample regions: descending ('Desc') and ascending ('Asc') regions in aortic dissection patients with and without Marfan syndrome. Figure is adapted from Panpho et al. [449] (licensed under CC-BY 4.0).

tests did not involve tissues from aortic dissection patients, who may exhibit different delamination characteristics due to potentially reduced cohesive forces between layers. Nevertheless, such tests offer valuable insights into the general behavior of delamination and its underlying mechanisms.

*Role of mouse models.* Although the ultimate goal is to understand dissection of the human aorta, mouse models have emerged as important, which is due primarily to the availability of relevant genetically modified mice. Motivated by mutations that predispose humans to thoracic aortic aneurysms and dissections, mouse models are available for Marfan syndrome, Loeys-Dietz syndrome, and vascular Ehlers-Danlos syndrome, among others [191]. Moreover, these mice can be rendered hypertensive using different approaches (e.g., chronic infusion of angiotensin II, high salt combined with endothelial nitric oxide inhibition, and transverse aortic constriction), which increases hemodynamic loading and thereby risk for dissection. For studies on mouse models, we refer to the work of Trachet and his colleagues, which offer exemplary research on angiotensin II-infused  $Apoe^{-/-}$  mice for preclinical investigations of aneurysms and dissections [463–467]. Importantly, mouse models enable much greater control of the altered hemodynamics and more convenient studies of the natural history. As in humans, lesions tend to present in the thoracic aorta, but have also been found in the suprarenal abdominal aorta in the angiotensin II model. Noting that mutations to lysyl oxidase also predispose to thoracic aortopathy, an inhibitor of lysyl oxidase ( $\beta$ -aminopropionitrile) has also been used to increase disease risk [226].

Given the small size of the murine aorta, the emergence of mouse models of thoracic aortopathy has driven the development of new multimodality, multiscale methods for studying the biomechanics of the aorta. Such methods have employed or combined various techniques, including micro-CT, high-resolution ultrasound, optical coherence tomography, and panoramic digital image correlation [468–472]. Fig. 23 shows an optical coherence image from a study of *Tgfbr2* null mice, depicting significant intramural delamination of the wall under physiological loading. Importantly, this delamination initiated once the aorta was rendered passive, suggesting that active stresses associated with SMC contraction can be partially protective by reducing overall stress in the aortic wall. Compromised actomyosin activity is a characteristic feature of many genetic mutations that give rise to thoracic aortopathy, see Table 2, including Marfan syndrome and the familial conditions resulting from *Myh11* and *Acta2* mutations, with associated mouse model (*Myh11* and *Acta2*) mutants available. See for example a biomechanical comparison across multiple mouse models that revealed increased circumferential material stiffness as most highly correlated with increasing diameter [473]. Continued studies in mouse models are thus warranted.

### 3.2.3. Roles of inflammation and proteases

As noted above, diverse molecular mechanisms are thought to contribute to degeneration of the thoracic aortic wall, ultimately resulting in aneurysms and dissections [475]. These mechanisms lead to histologic disruption, including the appearance of 'cystic spaces' among the lamellar layers (see Section 3.2) and wall thinning (see Fig. 24). Such spaces may provide a substrate for blood to enter and disrupt the lamellar structure of the aortic wall, resulting in propagating aortic dissection. In this section, we will concentrate on two additional important processes that disrupt the aortic wall: inflammation and proteolytic activity.

Inflammation. It has long been recognized that inflammation plays an important role in diverse aspects of degeneration of the aortic wall as well as in the body's response to injury, including aortic dissection [476]. This evidence of the importance of inflammation has

#### Table 3

Publication index of reports on mechanical experiments and microstructural investigations on tissues extracted from aortic dissection patients, ordered by year of publication (TL: true lumen; FL: false lumen).

| Publication                  | Year | No. of donors                        | Age   | Туре           | Objective(s)   |
|------------------------------|------|--------------------------------------|-------|----------------|--|
| Yamada et al. [452]          | 2015 | Human ascending (9)                  | 52–85 | Acute type A   | Histology (collagen, elastin); uniaxial tests  |
| Babu et al. [453]            | 2015 | Human ascending (16)                 | 7-71  | Type A         | Biaxial tests  |
| Sommer et al. [454]          | 2016 | Human ascending (3) & descending (1) | 43–73 | n.a.           | Shear tests (in-plane, out-of-plane)   |
| Ahuja et al. [455]           | 2018 | Porcine descending (1) <sup>a</sup>  | n.a.  | n.a.           | Biaxial tests; layer-specific (outer wall<br>of the TL, dissection flap, outer wall of<br>the FL)  |
| Manopoulos et al. [456]      | 2018 | Human ascending (12)                 | 35–77 | Acute type A   | Histology; uniaxial tests; layer-specific<br>(outer wall of the TL, outer wall of the<br>FL)   |
| Deplano et al. [457]         | 2019 | Human ascending (3)                  | 53–57 | Acute type A   | Histology (collagen, elastin); biaxial<br>tests; layer-specific (outer wall of the<br>TL, dissection flap, outer wall of the<br>FL)  |
| Amabili et al. [250]         | 2020 | Human descending (1)                 | 73    | Chronic type A | Uniaxial tests (quasi-static, dynamic);<br>layer-specific (outer wall of the TL,<br>dissection flap, outer wall of the FL)   |
| Rolf-Pissarczyk et al. [450] | 2021 | Human ascending (1)                  | 51    | n.a.           | Uniaxial tests; layer-specific (media, adventitia)   |
| Kan et al. [458]             | 2021 | Human (5)                            | n.a.  | Туре В         | Uniaxial tests   |
| Panpho et al. [449]          | 2022 | Human descending & ascending (14)    | 37–68 | Chronic        | Histology (elastin); biochemical<br>analysis (elastin, collagen, GAGs);<br>custom-indentation test; layer-specific<br>(outer wall of the TL, dissection flap,<br>outer wall of the FL) |
| Li et al. [459]              | 2022 | Human (35)                           | 64–38 | Type A         | Uniaxial tests; layer-specific<br>(dissection flap)  |
| Guo et al. [460]             | 2024 | Human (13)                           | 41–61 | Туре А         | Biaxial tests; histology (collagen,<br>elastin); layer-specific (outer wall of<br>the TL, dissection flap, outer wall of<br>the FL)  |

<sup>a</sup>Artificially created aortic dissection (see Table 5)

accumulated over decades, especially regarding abdominal aortic aneurysm, dating from an initial key publication by Tilson and his colleagues who found evidence for both humoral and cellular influences [477]. Elastin degradation products appeared to be at least one important target for an excessive immune reaction. Whereas inflammation clearly plays a significant role in disease progression in abdominal aortic aneurysms [478,479], which tend not to dissect, the role of inflammation in the pre-dissection diseased thoracic aorta is less definitive but emerging [480]. Tellides and his colleagues implicated a role of IFN- $\gamma$  producing T-cells in outward remodeling of the thoracic aorta [481], while Milewicz and her colleagues demonstrated that T-lymphocytes and macrophages contribute to the pathologic process, with interleukins and interferon subtypes playing major roles [482,483]. Wang et al. [484] also found interleukin signaling and T-cell activation pathways to be upregulated in patients with ascending aortic aneurysms, see Fig. 25.

Heightened inflammation and the corresponding immune responses have also been implicated as a possible instigating factor for acute aortic dissection [485]. Infiltrating immune cells can have myriad effects, however, including promotion of fibrosis in the adventitial layer [486]. An exuberant accumulation of collagen in the adventitia reduces aortic function (compliance and resilience) but can serve to protect against transmural rupture, perhaps contributing to non-lethal dissection by leading to contained rupture and false lumen dilatation. By contrast, inflammation necessarily plays a central role in the post-dissection thoracic aorta – accumulation of blood within the medial layer provides an unencumbered inlet for monocytes and leukocytes, which are expected to play a role in the attempted 'wound healing' response. A large study from China suggested that, given the inflammatory nature of aortic dissection itself and the vigorous inflammatory response to open heart surgery for treatment of acute type A dissection, treatment with the inflammation inhibitor ulinastatin might be beneficial post-operatively [487]. Some supportive evidence was amassed in a large multicenter clinical trial. There is, however, a need for caution – to balance immuno-suppression that reduces disease progression versus that which attenuates inflammation-mediated tissue repair. Much remains to be learned regarding the myriad roles of inflammation in thoracic aortopathy and its treatment.

*Proteases.* The accumulation of blood within the medial layer not only allows an influx of inflammatory cells, it also promotes an intramural thrombotic response [488,489]. The serine protease plasmin plays a central role in subsequent fibrinolysis, that is, resolution of the fibrin mesh within the thrombus. Plasmin plays many other roles as well, including loss of vascular SMCs, proteolysis of adhesive proteins, release of sequestered TGF- $\beta$  from the extracellular matrix, and activation of MMPs, all of which have been implicated in thoracic aortic aneurysms and dissections [490]. There exist over 20 different MMPs, many of which are produced by the different cells that are involved in aortic remodeling, repair, and disease progression, including SMCs, fibroblasts, and of course



**Fig. 23.** Optical coherence tomographic images display circumferential and axial (here: longitudinal) sections of a representative cannulated artery of a mouse model during biaxial mechanical tests on a rapamycin-treated, *Tgfbr2* disrupted descending thoracic aorta subjected to 70 mmHg luminal pressure at *in vivo* axial stretch. The artery underwent vasoconstriction (top row) followed by a 40 min vasodilation using multiple washouts (middle row) and finally experienced cyclical pressurization from 10 to 140 mmHg under passive conditions (bottom row). Intramural delaminations appeared locally upon SMC relaxation and propagated along the length and circumference, occurring most frequently in rapamycin-treated, *Tgfbr2* disrupted samples. This phenomenon was absent in control samples and occurred in only one non-treated *Tgfbr2* disrupted sample. White arrows indicate the initiation and progression of delamination, while the dark arrow on the left side shows the temporal progression of the experiment. Scale bars (top row) represent 400  $\mu$ m. Figure is reprinted from Ferruzzi et al. [474] with permission from Wolters Kluwer Health, Inc.

macrophages [491]. The ultimate effects of MMPs on the extracellular matrix of the aortic wall depend on many factors, including their increased production in a latent form (precursors of MMPs), activation by blood pressure-induced mechanical stresses within the wall, cell-mediated tractions, and other proteases, and inhibition by tissue inhibitors of MMPs (i.e. TIMPs).

Nearly two decades ago, Koullias et al. [398] evaluated MMP and TIMP levels in aortic tissue specimens from patients with ascending aortic aneurysms and dissections compared with control aortas. They found a significant increase in expression of MMP-1 (a collagenase) and MMP-9 (a gelatinase) in aneurysm and dissection patients compared with controls. Aortic dissection patients manifested even higher MMP-2 (a gelatinase) and MMP-9 levels that non-dissected aneurysm patients. The ratio of MMP-9 (proteo-lytic) to TIMP (proteo-protective) was elevated in both the aneurysm and dissection patients, thus indicating a net excess tendency toward proteolysis. Since that early study, overwhelming evidence has accumulated that excess MMP activity, coupled with decreased activity of TIMPs, results in net degeneration of the extracellular matrix, with accompanying loss of mechanical strength of the wall [492–494]. This is vividly evident in the extraordinarily thin aortic wall seen in Fig. 24.

Finally, the aforementioned marked accumulation of mucoid material within the degenerated thoracic aorta, historically associated with cystic spaces, can play multiple adverse biomechanical and biological roles [93,103]. This accumulation of mucoid material, that is, aggregating GAGs/PGs, implies either heightened production or reduced removal, or both. The ADAMTS (a disintegrin and metalloproteinase with thrombospondin motifs) family of proteases play key roles in the turnover of GAGs and PGs [495], and some have been implicated in thoracic aortic aneurysm and dissection [496,497].

In summary, extracellular matrix turnover plays a key role in aortic homeostasis and its loss [498] and there is a pressing need to understand better the effects of diverse proteases (e.g., serine, MMPs/TIMPs, or ADAMTS family) and how their regulation or dysregulation contributes to the structural integrity of the wall as well as the extracellular matrix ligands that are presented to the resident vascular cells and influence their gene expression. Much has been learned, but much remains to be discovered regarding roles of both inflammation and proteolysis.

### 3.3. Changed hemodynamics

Hemodynamics in aortic dissection has a significant impact on patient outcomes and influences aortic dilatation, organ malperfusion, thrombus formation, remodeling processes, and more. Therefore, it plays a crucial role in the progression of the disease and possible complications and should also be considered in treatment strategies. Despite recent advances in medical imaging, hemodynamics is usually not considered in everyday clinical practice; rather, it is performed exclusively for research purposes. In general, blood flow velocity in the true and false lumina can be determined using advanced imaging methods such as 4D-flow MRI [499,500], as introduced in Section 2.3.1. These mostly non-invasive techniques enable the assessment of anatomical and hemodynamic features of the dissected aorta. Furthermore, they allow evaluation of hemodynamics in true and false lumina using calculated parameters such as WSS, kinetic energy, flow helicity, and stasis. However, note that, for instance, the assessment of WSS depends heavily on the accuracy of the surface reconstruction, which is still very low and therefore often error-prone due to the spatial resolution of 4D-flow MRI [163–165]. Accurate hemodynamic data can, in turn, be used to validate *in silico* models, which will be discussed in the following



Fig. 24. Aortic wall from an ascending aneurysm that has been so strongly degraded that one can read the markings on a ruler through the aortic wall. It is frightening to envision this thin aortic wall restraining the bloodstream and the systolic blood pressure.

sections. Readers are also referred to the review article by Sherrah et al. [32] for a systematic overview.

In 1988, after an earlier study by Mohri et al. [501], Bogren et al. [502] investigated flow rates in three patients with chronic aortic dissection and observed that antegrade flow rates in the false lumen were low and remained constant throughout the cardiac cycle. In contrast, they observed high flow rates during systole and normal flow rates during diastole in the true lumen. In the same year, Mitchell et al. [503] reported similar observations in a study analyzing blood flow patterns in a single patient with chronic aortic dissection. They found antegrade flow in both lumina during peak systole, albeit at a reduced rate in the false lumen. During late systole, antegrade flow persisted in the true lumen, while retrograde flow occurred in the false lumen. In diastole, flow nearly stopped in both lumina. A delay in the maximum flow rate in the false lumen compared with the true lumen was observed, possibly due to the presence of an intimal tear. Mohr-Kahaly et al. [504] subsequently performed imaging in a larger patient cohort, comparing blood flow patterns in 18 patients with and without prior aortic dissection surgery, focusing on type I, II, and III dissections. They detected systolic laminar flow in the true lumen, while in the false lumen, blood flow was observed only during late systole and diastole. Two distinct blood flow patterns in the false lumen were identified: laminar biphasic flow and slowly circulating flow. Some patients exhibited delayed laminar flow in the false lumen during systole, which moved distally and reversed during diastole. Others displayed a slowly circulating flow in the false lumen, characterized by a 'swirling pattern'. Blood flow was found to be influenced by the location, number, and size of intimal tears, which impacted the pressure difference between lumina. Unidirectional flow was observed in proximal intimal tears, while bidirectional flow occurred in more distal intimal tears. Flow from the true to the false lumen was detected during systole and reversed during diastole. Chang et al. [499] measured mean flow velocity and peak systolic velocity in the true and false lumina. They recorded a mean velocity of  $134 \pm 14.0$  mm/s in the true lumen and  $31 \pm 8.4$  mm/s in the false lumen, with peak systolic velocities of  $436 \pm 72.0$  and  $143 \pm 23.0$  mm/s for the true and false lumina, respectively. However, no significant differences were observed in the average flow rate between the true and false lumina, which could be attributed to the larger size of the false lumen compared with the true lumen.

The finding of the studies discussed were then confirmed by other publications in the twenty-first century. In the study by Strotzer et al. [505], 14 patients were examined to determine the mean flow rate and average peak velocities of the true and false lumina. The mean flow volume per minute was found to be 1982ml in the true lumen and 1052ml in the false lumen. Similarly, the average peak velocities were significantly higher in the true compared with the false lumen, with values of 980 and 470 mm/s, respectively. Additionally, ten patients exhibited bidirectional flow in the false lumen, with a retrograde flow ranging from 6.8 to 98%. The false lumen displayed an early systolic peak and a delayed systolic negative peak, while the blood flow in the true lumen was unidirectional.

Subsequently, Markl et al. [506] analyzed the blood flow in the thoracic aorta for both healthy subjects and those with a diseased aorta, one of which exhibited an aortic dissection. The quantitative results revealed that there was almost no blood flow in the false lumen throughout the cardiac cycle. In contrast, the blood flow velocity in the true lumen was higher than in the aortic outflow region due to the larger area. Additionally, the authors observed a retrograde flow in the entire ascending aorta during diastole.

François et al. [159] investigated flow patterns in 12 patients. They found higher total flow and peak flow in the true lumen compared with the false lumen. Furthermore, they characterized the flow pattern in the false lumen as complex, abnormal, and non-laminar, with a higher retrograde flow than in the true lumen. In contrast, the true lumen primarily exhibited laminar flow. As a representative patient case, Figs. 26(a) and (b) show a visualization of flow acceleration in the proximal descending aorta, while Figs. 26(c) and (d) illustrate low flow velocities in the false lumen.

In the study by Sherrah et al. [507], ten patients with chronic type B dissection were examined after undergoing surgical repair of ascending aortic dissection. The findings showed that, at peak systole, the overall flow profile in the false lumen was lower than in the



**Fig. 25.** Hierarchical clustering of 61 whole blood samples analyzed using the 1,199 differentially expressed genes determined by significance analysis of microarray. The level of expression of each gene in each sample, relative to the mean level of expression of that gene across all the samples, is represented using a red–black-green color scale as shown in the key (green: below mean; black: equal to mean; red: above mean). (a) Scaled down representation of the entire cluster of the 1,199 signature genes and 61 whole blood samples. (b) Experimental dendrogram displaying the clustering of the samples into two main branches: the thoracic aortic aneurysm branch (red) and the control branch (blue) with a few exceptions. (c) Gene expression pattern of representative genes within biological pathways that are statistically significantly over-represented (random over-lapping *p*-value, 0.05) by the up-regulated (red bars) or the down-regulated (blue bars) signature genes of thoracic aortic aneurysm. Note departure from control levels in interleukin signaling and T-cell activation pathways. Figure is reprinted from Wang et al. [484] (licensed under CC-BY 4.0).

true lumen and normal aortas (Figs. 27(a) and (b)). Additionally, the peak systolic flow rate per aortic lumen area was lower in the false lumen compared with the true lumen, and both rates were lower than those observed in control aortas. Patients experienced higher retrograde flow in the false lumen than in the true lumen throughout the descending aorta. Moreover, the derived pulsatility index was higher in the true compared with the false lumen among all patients. Briefly, the pulsatility index equals the difference between the peak systolic flow velocity and the end diastolic flow velocity, divided by the mean flow velocity. Thus, it is a dimensionless parameter that provides insight into the resistance to blood flow and the pulsatility of the blood flow waveform. The pathline images generated in the study provided detailed views of the flow patterns in both lumina, including the fenestration sites between the true and false lumina, as shown in Fig. 27(c).

Liu et al. [508] analyzed the hemodynamic characteristics within the true and false lumina in 16 patients with aortic dissection of different types. They found that peak flow velocity and peak flow rate, among other parameters, were substantially higher in the true lumen. Furthermore, they identified a negative correlation between the size of the intimal tear and both the flow velocity and flow rate in the true lumen, as well as a positive correlation with the average through-plane velocity, average net flow, and peak flow rate in the false lumen. These results indicate a direct relationship with the number and size of intimal tears. Additionally, the blood flow indices



**Fig. 26.** A representative case of acute type B dissection shows (a) a color-coded streamline visualization from 4D-flow MRI during mid-systole, highlighting flow acceleration (open arrow) in the proximal descending aorta directed into the false lumen through a large intimal tear ((b), black arrow), seen on axial CTA at the indicated level (dashed line), with false lumen flow characterized by a large vortex ((a), white arrow), resulting in predominantly retrograde flow. A representative case of chronic repaired type A dissection shows (c) a color-coded streamline visualization during mid-systole revealing slow flow through the false lumen ('FL') compared to the true lumen ('TL'). This can also be seen in the subtracted maximum intensity projection images (d) acquired with time-resolved contrast-enhanced MRA. Figure is adapted from François et al. [159] with permission from Elsevier.

in the true lumen were enhanced with an increase in the number of intimal tears, and the peak flow in the false lumen was reduced. In cases where false luminal thrombi were present, the average through-plane velocity and peak velocity magnitude in the true lumen were found to be significantly higher. To conclude, they posited that blood flow velocity, flow rate, flow pattern, and pressure differences between the true and false lumina are crucial factors affecting the collapse of the true lumen and dilatation of the false lumen.

Allen et al. [509] conducted a study on 19 patients with type B dissection in which active fenestrations were assessed hemodynamically. They examined the number and location of fenestrations, as well as the direction of flow. The results indicated that different methods detected different numbers of fenestrations. More fenestrations resulted in biphasic flow over the cardiac cycle, with flow entering the false lumen during systole and exiting during diastole. Antegrade flow was most apparent in large fenestrations, while retrograde flow from the false to true lumen (diastolic flow) was often useful in identifying smaller fenestrations.

In a longitudinal case study by Takei [510], which focused on patients with chronic type B dissection, changes in flow direction and rate within the aortic lumen, as well as at entry and fenestration sites, were analyzed. They conducted this analysis following entry closure using thoracic endovascular aortic repair. The study assessed one entry tear and four fenestrations. Before the operation, the entry site produced a significant antegrade flow toward the four fenestration sites. However, shortly after the closure of the entry site, fenestration sites one to three transformed into a new entry site, exhibiting a retrograde flow pattern. This change led to an increase in the false lumen rate during the acute phase. As a result, the flow from the true lumen to the false lumen at the previous re-entry sites gradually decreased. This process ultimately led to aortic remodeling, which was characterized by a reduction in the size of the false lumen.

In the study conducted by Jarvis et al. [162], 20 chronic descending dissection patients were examined, along with 21 age-matched controls. Of the patients, six were medically managed, while 14 had undergone repair procedures. Hemodynamics were assessed in these individuals, and repaired dissection patients had received either an ascending aortic repair or an elephant trunk stage one procedure. The results showed that patients with repaired type A dissection had higher levels of retrograde flow and increased kinetic energy in the true lumen compared with the control group. They also exhibited higher true lumen kinetic energy than patients with type B dissection. Additionally, repaired type A dissection patients were associated with higher false lumen kinetic energy and lower

M. Rolf-Pissarczyk et al.



**Fig. 27.** Visualization of sagittal pathline views in the isolated thoracic aorta during peak systole within one cardiac cycle: (a) from a control patient, (b) from a chronic type B dissection patient with isolated pathlines for both the true and false lumina, and (c) showing the pathline isolated for the true lumen in a representative patient with chronic type B dissection, specifically highlighting blood flow through the fenestrations connecting the true and false lumina. Figure is adapted from Sherrah et al. [507] (licensed under CC-BY 4.0).

false lumen stasis compared with type B dissection patients. Imaging demonstrated both global and regional hemodynamic differences between patients with descending aortic dissection and the control group. Repaired type A and type B dissection patients exhibited significantly changed regional true and false lumina hemodynamics. A patient with ascending aortic repair showed elevated antegrade flow, retrograde flow, and kinetic energy in the true lumen when compared with a patient with type B dissection and a control subject. On the other hand, the type B dissection patient had elevated levels of stasis. This information suggests that patients with repaired type A dissection may encounter complex flow situations that could potentially lead to late complications. One possible explanation for the difference in kinetic energy and retrograde flow after ascending aortic repair could be the use of a noncompliant graft replacing the ascending aorta, although this hypothesis is still speculative. Overall, they proposed a data-driven classification of aortic dissection patients for risk stratification, which also takes hemodynamic parameters into account.

Then, Evangelista et al. [320] presented a prospective study of 131 consecutive patients, from which 78 were surgically treated type A dissections and 53 were medically treated type B dissections. In the acute phase, these patients initially showed a persistent patent false lumen in the descending aorta. They reported a false lumen systolic antegrade flow of  $\geq$ 30% relative to total systolic antegrade flow in combination with retrograde diastolic flow of  $\geq$ 80% relative to total diastolic false lumen flow, which were identified as predictors of a higher risk of complications.

Cosset et al. [511] investigated the influence of TEVAR on blood flow in the aorta and its branches in patients with type B dissection using 4D-flow MRI. Seven patients scheduled for TEVAR underwent 4D-flow MRI before and after the procedure. The results showed that TEVAR significantly increased the antegrade flow in the true lumen from 59.9 to 81.6% and significantly decreased both antegrade flow (from 15.9 to 3.3%) and retrograde flow (from 10.3 to 4.6%) in the false lumen. Interestingly, the retrograde flow in the true lumen increased from 4.36 to 10.8% after TEVAR, which may be due to the increased wall rigidity caused by the metallic stent. Additionally, user-independent helicity quantification revealed elevated helicity at the level of the secondary entry tears, which had been missed by streamline visualization.

In earlier discussions, we emphasized the critical role of aortic or false lumen dilatation in dissection as a significant risk factor (Section 3.1.2). The surgical treatment of a progressively dilating false lumen is vital to reduce the risk of rupture. Numerous studies have investigated the relationship between dilatation during the chronic phase and hemodynamics, underscoring the importance of understanding these dynamics in managing aortic dissection.

In two subsequent studies, Inoue et al. [512] investigated the relationship between blood flow characteristics and aortic dilatation. They examined six patients with chronic type B dissection and an existing intimal tear, which allowed communication between the two lumina. In five of these patients, they observed bidirectional flow in the false lumen. They provided a more detailed analysis of two patients' results. One patient showed a peak average velocity and flow rate in the true lumen of 700 and 100 ml/s, respectively. In contrast, the flow rate in the false lumen was extremely low, with retrograde flow noted during diastole. During systole, the output volume in the true and false lumina was 34 and 4ml, respectively, resulting in a ratio of 0.12. Another patient exhibited a dilated false lumen with a flow rate exceeding that of the true lumen during systole. The authors hypothesized that the dilated false lumen compressed the true lumen, impeding its circulation. In this instance, the ratio between the output volume of the true and false lumina during systole was 1.68. Moreover, they investigated the relationship between the false lumen diameter and the blood flow ratio in the true and false lumina. They discovered an inverse correlation between the maximum diameter of the false lumen and the peak average velocity of the true lumen. As the maximum diameter of the false lumen increased, the peak average velocity of the true lumen decreased. In a subsequent study, Inoue et al. [513] analyzed 21 patients and discovered that flow patterns in the false lumen could be divided into three categories: primarily antegrade flow, primarily retrograde flow, and bidirectional flow. They found that the flow rate in the false lumen, when compared with the total flow rate in the true and false lumina, was significantly higher in patients with a primary antegrade flow pattern. Additionally, the average yearly growth rate of the maximum diameter of the dissected aorta was also higher in these patients. There was a notable correlation between the flow rate in the false lumen relative to the total flow rate in the true and false lumina, and the average yearly growth rate of the maximum diameter of the dissected aorta. The blood flow velocities in the true and false lumina, as well as the direction of the flow were highly dependent on the location and size of the dominant intimal tears.

In a later study, Clough et al. [158] aimed to visualize and quantify flow characteristics in patients with type B dissection, exploring



 HELICAL FLOW
 HELICAL FLOW

 PROXIMAL AORTA
 MID-THORACIC AORTA

**Fig. 28.** Visualization and quantification of helical flow in a patient with type B dissection assessed with 4D PC-MRI: (a) entire aorta examined to identify regions exhibiting helical flow, (b) plane placed perpendicular to the primary flow direction, and (c) helicity (in degrees per second) calculated by evaluating the amount of in-plane rotation and the start and end times of the cardiac cycle. Figure is reprinted from Clough et al. [158] with permission from Elsevier.



Fig. 29. 3D PC-MRI results of two representative patients with a history of a stable and dilating false lumen ('FL') ('TL': true lumen). Figure is reprinted from Burris et al. [514] with permission from Elsevier.

their relationship to the rate of aortic dilatation. They discovered that an increased false lumen stroke volume and the velocity were associated with faster aortic dilatation. Helical flow, as demonstrated in Fig. 28, was observed in the false lumen in eight out of 12 patients and was also linked to the rate of aortic dilatation. In five patients, helical flow occurred at a single location, while it was present at two locations in one patient and throughout the false lumen in two patients. The helical flow typically developed in early systole before disappearing in early diastole. Furthermore, the rate of rotation of the helical flow was found to correlate with the rate of aortic dilatation.

In a series of studies, Burris and his colleagues investigated the role of hemodynamic stress as a primary factor in the progressive dilatation of the false lumen in chronic type B dissection. In their first study [514], the authors examined 12 patients and determined and evaluated the entry tear regurgitant fraction, defined by the ratio of retrograde flow rate at the dominant entry tear during diastole divided by the antegrade systolic flow rate. For their assessment, they first located the primary entry tear, placed a 2D flow analysis plane into the 3D dataset of the 4D-flow MRI, and then calculated the antegrade and retrograde flow. They found that entry tear regurgitant fraction was significantly higher in patients with a history of false lumen dilatation, which was likely due to inadequate distal outflow pathways, see Fig. 29. As previously mentioned, these results support the 2007 statement by the IRAD, which suggests that reduced outflow in partial false lumen thrombosis leads to increased false lumen pressures, i.e. to increased hemodynamic stress in the false lumen [515]. However, the association between elevated regurgitant flow and false lumen dilatation is likely not just a result of aneurysmal degeneration. Instead, retrograde flow is more likely a direct consequence of decreased false lumen outflow, as shown by computerized blood flow simulations on nonaneurysmal models [270,516]. False luminal outflow tells only half of the hemodynamic story and its direct measurement is almost impossible because of the often numerous and small outflow pathways. Therefore, alternatively, direct measurement of the entry tear regurgitant faction using 4D-flow MRI is an intuitive and simplified approach that aims to quantify the false lumen hemodynamic stress that occurs as a result of inadequate outflow pathways in the false lumen.

In a subsequent study [161], an analysis of 18 patients with chronic type B dissection was performed to further evaluate the ability of various hemodynamic variables to independently predict aortic growth rate during follow-up, with particular emphasis on the false lumen ejection fraction, previously termed entry tear regurgitant fraction. Retrograde flow in the false lumen was particularly common, with 16 patients showing measurable retrograde flow during diastole, while the flow in the true lumen was predominantly antegrade. The mean false lumen ejection fraction ( $28.9 \pm 24.4\%$ ) did not correlate well with the false lumen retrograde flow fraction measured 3cm distal to the primary tear. However, it was significantly higher in patients with increased aortic dimension than in patients with stable dimension ( $43.8 \pm 22.1$  versus  $10.3 \pm 10.1\%$ ). In patients with a dilated aorta, the entry tear peak velocity was lower. Key observations include that peak velocity at the entry tear jet is lower in patients with a history of a dilating false lumen, likely due to obstructed or insufficient false lumen outflow. Furthermore, false lumen ejection fraction demonstrates a moderate to strong



Increasing FL pressure

**Fig. 30.** (a) Representative patient examples for false lumen ('FL') hemodynamic evaluation in patients with slow aortic growth: (i) extraction of false lumen ejection fraction ('FLEF'), (ii) maximum systolic deceleration rate ('MSDR'), and (iii) false lumen maximum relative pressure ('FL  $\Delta P_{max}$ ') is shown along with the associated output ('VF': virtual field). (b) A conceptual model illustrates the relationships between aortic growth and 4D-flow-derived markers of false lumen pressurization and shows that increased false lumen ejection fraction is correlated with increased retrograde flow (light blue), maximum systolic deceleration rate reflects a faster flow deceleration due to higher pressurization, and a false lumen maximal relative pressure indicates a reduced pressure difference with increased aortic growth rate. Figure is adapted from Marlevi et al. [517] (licensed under CC-BY 4.0).

correlation with aortic growth rate. Finally, baseline maximal diameter and the distance of the entry tear from the left subclavian artery are independent predictors of aortic growth rate. Despite the limitation of investigating a relatively small and specific cohort composed only of chronic dissections and considering the timing of 4D-flow MRI, which raises the possibility that elevated false lumen ejection fraction is a consequence of aortic growth rather than a cause, their findings provide further evidence for the importance of false lumen hemodynamics in the risk assessment of patients with type B dissection.

In a third study [517] with 12 patients, they investigated the relationship between aortic growth and three different methods for assessing hemodynamic stress in the false lumen: (i) false lumen ejection fraction, analogousto their previous studies; (ii) maximum systolic deceleration rate; and (iii) false lumen relative pressure. The maximum deceleration rate was calculated as the difference between maximum and minimum acceleration during systole divided by the corresponding time interval between these two points, while the false lumen relative pressure was based on a non-invasive measurement of intravascular pressure drop [518]. The results shown for a patient with slow aortic growth and a conceptual overview depicting the proposed relationship are presented in Fig. 30. These findings emphasize the variable nature of aortic hemodynamics in patients with type B dissection and suggest that these three hemodynamic metrics may be helpful in identifying patients at highest and lowest risk for progressive aortic growth and complications.

In a related study, the aim of Chu et al. [519] was to evaluate the predictive value of the hemodynamics in the true and false lumina using 4D-flow MRI compared to aortic morphology measurements for adverse aorta-related outcomes and aortic growth in patients with type B dissection. The study included 51 patients, including 26 with a type B dissection and 25 with a repaired type A dissection, who had residual type B dissection, all with at clinical follow-up of at least six months. Patients were categorized based on the presence or absence of adverse aorta-related outcomes and an aortic growth rate of  $\geq$ 3 mm/year. Their findings indicated that patients with *de* 



**Fig. 31.** Four representative false lumen flow patterns observed in aortic dissection patients at the diaphragmatic level. These patterns are classified based on the directionality of flow during systole and diastole, with positive values representing antegrade flow and negative values denoting retrograde flow: (a) biphasic systolic flow with predominantly antegrade diastolic flow, (b) biphasic systolic flow with predominantly antegrade diastolic flow, and (d) monophasic systolic flow with predominantly retrograde diastolic flow, and (d) monophasic systolic flow with predominantly retrograde diastolic flow, Figure is adapted from Rudenick et al. [516] (licensed under CC-BY 4.0).

*novo* type B dissection and adverse aorta-related outcomes had larger baseline diameters, lower false lumen stasis, and lower true lumen peak velocity. Additionally, a higher kinetic energy ratio was observed in patients with aortic growth of  $\geq$ 3 mm/year across both the overall cohort and the *de novo* type B dissection subgroup.

A combined clinical and computational study was developed by Rudenick et al. [516] to investigate the aortic wall stiffness, the size and location of the tear, and the presence of abdominal side branches arising from the false lumen. These factors, among others, influence the dilatation of the false lumen, and are often associated with chronic aortic dissection. For this purpose, they monitored the false lumen flow pattern of 33 patients with chronic descending aortic dissection from three to 12 months after the acute phase in a cross-section by PC-MRI. After quantifying the measurements, four characteristic false lumen flow profiles, as shown in Fig. 31, were identified in 94% of all cases: (i) systolic biphasic flow and primarily diastole antegrade flow (18%), (ii) systolic biphasic flow and primarily diastole retrograde flow (42%), (iii) systolic monophasic and primarily diastole antegrade flow (27%), and (iv) systolic monophasic flow and primarily diastole retrograde flow (6%). Results from the *in silico* model are presented in Section 6.

The study by Ruiz-Muñoz et al. [520] aimed to understand the relationship between false lumen anatomy (diameter, entry tear location, and size), fluid dynamics (inflow, rotational flow, WSS, kinetic energy, and flow acceleration and stasis), and biomechanics (pulse wave velocity) with the presence and extent of a thrombus in the false lumen. The study included 68 patients with chronic non-thrombosed or partially thrombosed false lumen in the descending aorta after aortic dissection using CTA, cardiovascular MRA, and 4D-flow MRI. The study found that aortic growth rates in patients were higher with partial thrombosis than with a patent false lumen. False lumen kinetic energy was the primary indicator for distinguishing between patients with a patent false lumen and partial thrombosis and was independently associated with the extent of the thrombus.

Numerous studies have examined the hemodynamics of aortic dissection, focusing on kinetic flow patterns, flow rates, flow velocity, helicity, kinetic energy, and pressure in patients with varying disease manifestations and outcomes. Some studies also highlight local phenomena and associated hemodynamic changes, such as flow acceleration due to narrowing in the true lumen or the flow jet impacting on the outer wall of the false lumen, which is primarily influenced by the location and size of the intimal tear. More specifically, the flow patterns in the true and false lumina differ significantly during both the systolic and diastolic phases of the cardiac cycle. During systole, the laminar flow is predominantly antegrade in both true and false lumina. However, the maximum flow rate in the false lumen is slightly delayed and often considerably lower. In diastole, a primary retrograde flow in the false lumen is typically observed, leading to a bidirectional flow in the lumina throughout the cardiac cycle. Several studies by Burris and his colleagues also aimed to find a correlation between hemodynamic markers, such as the introduced false lumen ejection fraction, and late adverse events like aortic dilatation, yielding promising results. These findings are particularly appealing due to the simplicity of the markers used. In conclusion, hemodynamic studies repeatedly confirmed that blood flow between the lumina is consistently determined by the location, number, and size of intimal tears, which subsequently affect the flow pattern. However, the detailed hemodynamic conditions are always patient-specific.

Moreover, the distribution and role of WSS in aortic dissection *in vivo* have not been extensively studied using techniques such as 4D-flow MRI, which may be due to the challenging assessment requiring high spatial resolution. However, similar studies have been performed for other diseases. As an example, Guzzardi et al. [422] observed in bicuspid aortic valve patients that regions with higher WSS experienced greater elastin degradation, which is associated with aortic wall remodeling, compared to adjacent regions with normal WSS. For a more extensive discussion, interested readers are referred to Section 9.2. Similar imaging techniques could potentially be used to examine aortic dissection and provide further detailed insights. However, due to errors related to the low spatial resolution of 4D-flow MRI measurement, the results should generally be interpreted with caution. In conclusion, despite many important findings that have improved our understanding of aortic dissection, the available data are too sparse to generalize many of the findings due to the complexity of the disease and the individuality of the patients. To improve statistical significance, larger patient cohorts are required. Table 4 provides an overview of the presented studies on hemodynamic changes in aortic dissections, along with additional studies that have not been previously mentioned.

## Table 4

Publication index of reports on hemodynamics in aortic dissection patients assessed with medical imaging, ordered by year of publication (TL: true lumen; FL: false lumen; DSA: digital subtraction angiography; CS: compressed-sensing; PVM: phase velocity mapping).

| Report                      | Year | Flow imaging method(s)    | No. of patients | Туре                         | Objective(s)   |
|-----------------------------|------|---------------------------|-----------------|------------------------------|--|
| Mohri et al. [501]          | 1985 | pulsed Doppler ECG        | 1               | Type A                       | Hemodynamics in patient (flow pattern)   |
| Bogren et al [502]          | 1988 | MRI PVM <sup>b</sup>      | 1               | Type B                       | Hemodynamics in patient (flow rate)  |
| Mitchell et al [503]        | 1088 | MBI DVM <sup>b</sup>      | 1               | Type B                       | Hemodynamics in patient (flow rate)  |
| Mohr-Kahaly et al [504]     | 1080 | 2D pulsed color-coded     | 18              | Type A & B                   | Hemodynamics in patients (flow pattern)  |
|                             | 1001 | Doppler ECG               | 10              | Type A & D                   |  |
| Chang et al. [499]          | 1991 | cine VENC MRI             | 6               | Туре В                       | Hemodynamics in patients (average flow velocity, flow rate)  |
| Inoue et al. [512]          | 1996 | PC-MRI                    | 6               | Туре В                       | Hemodynamics in patients (flow pattern, peak average velocity)   |
| Inoue et al. [513]          | 2000 | cine VENC MRI             | 21              | Type A <sup>a</sup> , A, & B | Hemodynamics in patients (flow volume, flow pattern)   |
| Strotzer et al. [505]       | 2000 | MRI PVM                   | 14              | Type A <sup>a</sup> & B      | Hemodynamics in patients (peak flow velocity, antegrade/retrograde flow rate)  |
| Amano et al. [521]          | 2011 | cine 3D MRI PVM           | 16              | Type A & B                   | Hemodynamics in patients (flow pattern, flow velcity)  |
| Müller-Eschner et al. [522] | 2011 | VENC 4D-flow MRI          | 1               | Type B                       | Hemodynamics in patient (flow velocity)  |
| Clough et al. [158]         | 2012 | VENC 2D PC-MRI & VENC 4D- | 12              | Type B                       | Comparison of hemodynamics in patients (stroke   |
|                             | 2012 | flow MRI                  | 12              | 1990 2                       | volume, flow velocity, antegrade/retrograde flow   |
|                             | 0004 | VENC 4D G MDI             | 1               | T 1 <sup>a</sup>             | Have demonstration in matient (haliaal flam, flam  |
| Marki et al. [506]          | 2004 | VENC 4D-flow MRI          | 1               | Type A                       | pattern, flow velocity, antegrade/retrograde flow  |
|                             |      |                           |                 |                              | rate)  |
| François et al. [159]       | 2013 | VENC 4D-flow MRI          | 12              | Type A/B                     | Comparison of hemodynamics in patients with/   |
|                             |      |                           |                 |                              | without intervention (flow pattern, peak flow rate,  |
|                             |      |                           |                 |                              | flow velocity, retrograde/antegrade flow rate)   |
| Sherrah et al. [507]        | 2017 | multiVENC 4D-flow MRI     | 10              | Type A <sup>a</sup>          | Hemodynamics in patients (flow rate, flow velocity, pulsatility index)   |
| Rudenick et al. [516]       | 2017 | VENC PC-MRI               | 33              | Type A <sup>a</sup> & B      | Hemodynamics in patients (flow velocity, flow<br>profile in FL at diaphragm level)   |
| Liu et al. [508]            | 2018 | VENC 4D-flow MRI          | 16              | Type A & B                   | Hemodynamics in patients with diverse<br>characteristics (flow velocity, peak flow, regurgitant  |
|                             |      |                           |                 |                              | fraction in TL (EL)  |
| Guo et al. [523]            | 2018 | VENC 4D-flow MRI          | 1               | Туре В                       | Hemodynamics in patient (flow velocity, flow   |
| Allen et al. [509]          | 2019 | VENC 4D-flow MRI          | 19              | Туре В                       | pattern)<br>Hemodynamics in patients (flow velocity); detection  |
| Takei et al. [510]          | 2019 | 4D-flow MRI               | 1               | Туре В                       | of tear<br>Comparison of hemodynamics in patient with pre-/  |
| Burris et al. [514]         | 2019 | 4D-flow MRI               | 12              | Type A <sup>a</sup> & B      | post TEVAR (flow rate, flow velocity)<br>Comparison of hemodynamics in patients (FL  |
|                             |      |                           |                 |                              | ejection fraction, peak flow velocity) with growth rates   |
| Burris et al. [161]         | 2020 | VENC 4D-flow MRI          | 18              | Type A <sup>a</sup> & B      | Comparison of hemodynamics in patients (peak flow velocity, FL ejection fraction) with growth rates  |
| Jarvis et al. [162]         | 2020 | VENC 4D-flow MRI          | 20              | Type A <sup>a</sup> & B      | Hemodynamics in patients (antegrade/retrograde<br>flow rate stasis kinetic energy)   |
| Marlevi et al. [517]        | 2021 | VENC 4D-flow MRI          | 12              | Type A <sup>a</sup> & B      | Comparison of hemodynamics in patients (FL   |
|                             |      |                           |                 |                              | ejection fraction, max. systolic deceleration rate, FL<br>relative pressure) with growth rates   |
| Chen et al. [524]           | 2021 | VENC 4D-flow MRI          | 32              | Type A <sup>a</sup> & B      | Comparison of hemodynamics in patients<br>(required the fraction in TL/EL) with (without TEVAR   |
|                             |      |                           |                 |                              | in descending aorta  |
| Fang et al. [525]           | 2021 | DSA with syngo iFlow      | 51              | Туре В                       | Impact of TEVAR in descending aorta on renal blood flow (pre-/post-TEVAR)  |
| Takahashi et al. [526]      | 2021 | VENC 4D-flow MRI          | 33              | Type A <sup>a</sup> & B      | Comparison of hemodynamics in patients (gross<br>flow, average flow velocity, regurgitant fraction in  |
| Evangelista et al. [320]    | 2022 | VENC 4D-flow MRI          | 131             | Type A <sup>a</sup> & B      | Impact of hemodynamics in patients (flow velocity,<br>antegrade/retrograde flow rate) on grow rates/<br>outcome  |
| Chu et al. [519]            | 2022 | VENC 4D-flow MRI          | 51              | Type A <sup>a</sup> & B      | Comparison of hemodynamics in patients (kinetic  |
| Cosset et al. [511]         | 2022 | VENC 4D-flow MRI          | 7               | Туре В                       | energy, stasis, peak now velocity, antegrade/<br>retrograde flow rate) with growth rates<br>Comparison of hemodynamics in patients (pre-/<br>post-TEVAR) (antegrade/retrograde flow rate,<br>statis, belical flow) |
|                             |      |                           |                 |                              | statis, helical flow)  |

(continued on next page)

Table 4 (continued)

| Report                      | Year | Flow imaging method(s)          | No. of<br>patients | Туре                         | Objective(s)   |
|-----------------------------|------|---------------------------------|--------------------|------------------------------|--|
| Urmeneta Ulloa et al. [527] | 2022 | VENC 4D-flow MRI                | 3                  | Type A <sup>a</sup> , A, & B | Comparison of hemodynamics in patients (with/<br>without endograft in descending aorta) (flow<br>velocity)   |
| Ruiz-Muñoz et al. [528]     | 2022 | VENC 4D-flow MRI                | 54                 | Type A <sup>a</sup> & B      | Comparison of hemodynamics in patients (helical flow, pulse wave velocity, WSS) with growth rates  |
| Liu et al. [529]            | 2023 | 4D-flow MRI                     | 1                  | Туре В                       | Hemodynamics in patient (flow velocity, flow rate)   |
| Kilinc et al. [530]         | 2023 | Accelerated VENC<br>4D-flow MRI | 12                 | Туре В                       | Hemodynamics in patients (kinetic energy, peak<br>flow velocity, antegrade/retrograde flow rate,<br>stasis); comparison between GRAPPA- and CS-<br>accelerated 4D-flow MRI |
| Ruiz-Muñoz et al. [520]     | 2024 | VENC 4D-flow MRI                | 68                 | Type A <sup>a</sup> & B      | Hemodynamic in patients with patent/partially<br>thrombosed FL in descending aorta (inflow, helical<br>flow, WSS, kinetic energy, flow acceleration, stasis)               |
| Bellala et al. [531]        | 2024 | VENC 4D-flow MRI                | 15                 | Туре В                       | Comparison of hemodynamics in patients<br>(regurgitation fraction in FL, energy loss) with<br>growth rates   |

<sup>a</sup>Repaired type A with residual type B dissection; <sup>b</sup>Based on [532]

## 3.4. Role of thrombus formation

Aortic dissection can lead to the formation of false lumen thrombosis [126,533]. A thrombus usually forms due to slow or stagnant blood flow, endothelial damage, inflammation in the aortic wall, or changes in the hemodynamics, such as favorable WSSs and pressures [534]. These factors are influenced by pathological changes in the aortic structure. However, the specific factors that contribute to or prevent thrombus formation in aortic dissection are not fully understood. Understanding these mechanisms is crucial for developing effective treatments and surgical interventions, as the level of thrombus formation is tightly linked to the mortality rate. Thrombus formation within the false lumen in chronic aortic dissection might lead to several complications, including false lumen dilatation, occlusion of aortic branches, or embolization.

*Thrombosis.* Blood primarily consists of plasma, its liquid component, and formed elements, which include red blood cells, white blood cells, and platelets [535]. Platelets, also known as thrombocytes, play a critical role in hemostasis, which is the body's natural mechanism to stop bleeding and promote wound healing through the formation of blood clots [536]. The clots are the result of the coagulation cascade and consist mainly of platelets and red and white blood cells in a fibrin network. Thrombin, an enzyme that catalyzes the conversion of fibrinogen to fibrin and activates platelets and other coagulation factors, plays a key role in coagulation. In addition to its role in coagulation, thrombin is involved in the activation of platelets and other coagulation factors, and functions as an antithrombotic agent by regulating the coagulation process through the production of inhibitors.

The formation of blood clots is a complicated process that involves a combination of biochemical mechanisms, as well as transport mechanisms [535]. During hemostasis, platelets are sensitized and activated, causing them to adhere to injured blood vessels, form aggregates, and release additional agonists. This leads to a feedback mechanism that promotes further platelet activation. The coagulation cascade, which is the final step in hemostasis, involves intricate biochemical processes that ultimately lead to the formation of blood clots. This complex mechanism can be divided into three major phases: platelet adhesion and activation, aggregation, and coagulation cascade.

Initially, when the body experiences an injury, platelets, which can exist in both resting and active states, undergo a conformational change and become activated [537], adopting a rounded shape that enables them to adhere to damaged tissue and form a platelet plug. Platelets become activated when they encounter injured areas of the vascular wall, as specific glycoprotein receptors interact with components of the ECM, such as von Willebrand factor or collagen. This interaction initiates the process of platelet activation. A feedback mechanism then stimulates the release of other chemicals, such as adenosine diphosphate, thromboxane  $A_2$ , and thrombin, promoting further platelet activation. Shear stress is considered an important factor in thrombus formation because it can impact the function and phenotype of endothelial cells, which can influence platelet activation. Further information regarding the impact of shear stress on platelet activation can be found in relevant literature [538,539].

In the second phase, aggregation is triggered by the release of chemical agonists, resulting in platelet-to-platelet adhesion and activation of receptors. This process leads to the formation of the primary platelet plug, also known as primary hemostasis, which temporarily seals the injured area and reduces minor bleeding. Finally, in the third phase, known as the coagulation process or secondary hemostasis, fibrin is produced, which cross-links with the platelet plug to form stable clots. This mechanism is regulated by coagulation factors and cofactors, along with calcium and platelets. The coagulation process consists of three pathways: extrinsic, intrinsic, and common pathways, which are described in more detail in relevant literature [536].

M. Rolf-Pissarczyk et al.

Progress in Materials Science 150 (2025) 101363



**Fig. 32.** Representative scanning electron microscope images of coronary arterial thrombi: (a) arterial thrombus consisting mainly of fibrin fiber bundles (1) and fibrin sponge (2), and (b, c) dense, contracted thrombi with platelet aggregates (3) surrounded by fibrin and red blood cell balloons (4) trapped within the fibrin mesh; white blood cells (5) are also present. Figure is reprinted from Chernysh et al. [540] (licensed under CC-BY 4.0).

Yet there are remarkably few detailed analyses of thrombus structure that enhance our understanding of its relationship to vascular origin and duration *in vivo*, as well as the composition or internal structural features of thrombi most closely associated with embolization risk [540]. In other words, the composition, physical properties, and evolution are likely to differ based on factors such as their location (e.g., arterial and venous thrombi or pulmonary embolism), local conditions, and the time since formation. As they age, thrombi form a dense, stiff fibrin network that becomes stiffer over time [541]. For example, Fig. 32 illustrates the microstructure of coronary artery thrombi, which primarily consist of fibrin and platelets. The dense fibrin structure appeared inhomogeneous, with prominent thick bundles and a highly branched 'fibrin sponge'. Most red blood cells were found in compressed forms, with smaller amounts of normal biconcave and balloon-like red blood cells. Cellular microvesicles and leukocytes were also present in the thrombi.

As previously mentioned, several factors may contribute to the formation of thrombi in patients with aortic dissection. Local pathological changes in morphology can cause hemodynamic alterations that result in complex and convoluted geometry and result in areas of stasis, turbulence, or local flow disturbances, which collectively increase the risk of thrombus formation [542–544].

Turbulent flows enhance mass transfer and local mixing of coagulation factors, influence endothelial cell function, and bring platelets into close proximity to vessel walls [545]. In contrast, areas of recirculation and stasis are associated with low shear rates (below  $100 \text{ s}^{-1}$ ) and long residence times – the time a platelet remains in contact with a wall. These conditions promote deposition through the action of fibrinogen, increasing local blood viscosity and thereby enhancing the adhesion of platelets to the vessel walls [534,546]. This effect is further amplified by the tendency of cells to accumulate in recirculation areas due to transport mechanisms that move them from regions of high shear rates to regions of low shear rates.

In addition to shear rates, platelet activation can also be triggered by shear stress [543], as both can act as physical agonists to promote platelet activation [544] and affect endothelial function, which leads to hemolysis of red blood cells and the resulting release of adenosine diphosphate [547]. While lower shear stresses prolong the interaction time between platelets and coagulation factors, high shear stresses affect platelet activation and adhesion. Therefore, the effect of shear stress on platelet activation largely depends on the duration of exposure, i.e. the residence time, with longer exposure times being required for low shear stress levels and *vice versa* [548]. Interestingly, exposure to pulsatile shear stress can elicit a much stronger response and a significantly higher level of platelet activation, highlighting the importance of shear gradients in thrombosis [549–552]. Finally, platelet activation can be triggered by the tearing of the aortic wall when blood cells come into contact with different and potentially thrombogenic surfaces between the first two wall layers, analogous to an injury response.

*Correlation between thrombosis, dilatation, and mortality rates.* The progression of false luminal thrombi has been found to have a major impact on patient survival and therapeutic outcomes. Several studies have investigated the correlation between the status of false luminal thrombus and long-term outcome in patients with aortic dissection. These studies typically involve patients with type B dissection, type A dissection following surgical repair, or a combination thereof. Study cohorts are often categorized according to false lumen patency, distinguishing between patent, partially thrombosed, or completely thrombosed false lumina, as shown in Fig. 33. However, some studies did not distinguish between the latter two categories, making an interpretation of the results difficult. According to the definition of Tsai et al. [515], a partially thrombosed false lumen is defined as the concurrent presence of both blood flow and thrombus. Interestingly, the outcome of these studies can be contradictory at times. Additionally, the terminology used to describe the false lumen status is not consistent in the literature, with some studies referring to incomplete or complete thrombus formation [4] or other terms. We have adapted the most commonly used and unambiguous terminology of the false lumen status.

Tsai et al. [515] determined that the risk of death is 2.7 times higher for partially thrombosed false lumina compared with patent false lumina. However, for completely thrombosed false lumina, the risk is only 1.9 times greater. Following their study, they found that the mortality rates after three years were  $13.7\pm7.1$ ,  $31.6\pm12.4$ , and  $22.6\pm22.6\%$  for patent, partially thrombosed, and

Progress in Materials Science 150 (2025) 101363



Patent

Partial thrombosis

Complete thrombosis



completely thrombosed false lumina, respectively. They also linked the occurrence of partially thrombosed false lumen to an increased risk of false lumen dilatation caused by diminished outflow and thus increased mean pressure in the false lumen, which is consistent with other studies [554–556]. Particularly, areas featuring new or increased partial false luminal thrombi are strongly associated with late adverse outcome, following Higashigaito et al. [556]. In Trimarchi et al. [554], mortality rates of patent (37.5%), partially thrombosed (56.3%), and completely thrombosed (6.3%) false lumina were observed. They stated that a thrombus might lead to a diameter stabilization, but also to a higher incidence of malperfusion, i.e. the loss of blood supply to vital organs. Consequently, these studies suggested that a partially thrombosed false lumen is an independent risk factor for surgical re-intervention and overall survival rates. It is noteworthy, however, that in contrast to Tsai et al. [515], Trimarchi et al. [554] reported that a persistent patent false lumen carries the highest risk of complications, which is consistent with most other studies.

By contrast, the remaining studies did not associate partial thrombosis of the false lumen with worst late outcomes, such as a higher mortality rate or faster aortic growth, compared with a patent false lumen. Although some studies did not observe a correlation between the mortality rate and the thrombosis status [553,557,558], most studies have reported that a completely thrombosed false lumen is beneficial for late outcomes, whereas a patent false lumen increases the mortality rate [264,308,327,329,559–564]. In the study by Fattori et al. [562], for instance, it was found that the ten-year survival rate for patients with a patent false lumen was 58.8  $\pm$  3.5%, whereas for those with a completely thrombosed false lumen, the rate was significantly higher at 89.8  $\pm$  2.1%. Note however, they did not distinguish between a partially and completely thrombosed false lumen. These findings were also confirmed by De Leon et al. [564].

Other studies have reported that a partially thrombosed false lumen predicted a higher operation rate but does not affect long-time survival rates [558]. Tanaka et al. [553] found that the probability of requiring surgical treatment was 0% for completely thrombosed, 16% for partially thrombosed, and 26% for patent false lumina, with no observed differences in the long-term mortality. Late mortality was instead associated with patient age (>70 years old) and cerebrovascular accidents, coma, or renal failure [557]. Similarly, Bernard et al. [560] found that in patients older than 70 years, the presence of a patent false lumen after surgery increased the relative late mortality risk by a factor of 3.4. While the mortality rate tended to increase with an age of 70 years, patients with a patent false lumen were usually younger. Mean ages per group as reported by Tanaka et al. [553] were:  $58.3 \pm 16$  years old for patent,  $72.4 \pm 10.3$  years old for partially thrombosed, and  $68.4 \pm 11.1$  years old for completely thrombosed false lumen. Larsen et al. [557] reported that patients with a patent false lumen were on average three years younger than those with a partially thrombosed and six years younger than those with complete false lumen thrombosis. These findings suggest that the probability of thrombus formation tends to increase with age, which could be related to the increased likelihood of blood coagulation with age. The increased late mortality associated with partially thrombosed false lumina might be explained by the possible occurrence of local hypoxia in the aortic wall adjacent to a formed thrombus [515]. Similar to aortic aneurysms, hypoxia could increase local inflammations, neovascularization, and localized wall weakening [565,566], thus making the aortic wall more prone to fail. To explain why partially thrombosed false lumina are often associated with an increased late mortality, Suevoshi et al. [327] subdivided partially thrombosed false lumina into a sac formation type and a non-sac formation type. Of the patients with a partially thrombosed false lumen, 15% showed a sac formation and 85% did not. The sac formation type was defined in this study as a partially closed false lumen at the distal site of the false lumen, forming a blind pouch with a persistent intimal tear, as shown in Fig. 34. A sac formation was also associated with higher pressures in the false lumen [515], which was in turn associated with a worse outcome. Due to the low incidence of 15%, these results did not support the hypothesis that the formation of a partial thrombus per se leads to an increased risk of death. In summary, it might not be correct to state that partially thrombosed false lumina increase the mortality rate of patients. Specifically, some authors classify cases with a marginal amount of thrombus in the patent group, while others categorize them into the partially thrombosed cohort. This discrepancy does not allow formulating a clear statement.



**Fig. 34.** Two phases of imaging after contrast injection, (a) and (b), of enhanced CT images showing a sac-like formation of partially thrombosed false lumen, defined as a blind pouch with a persistent entry tear (arrows). Figure is reproduced from Sueyoshi et al. [327] with permission from Elsevier.

Although some findings have been controversial, a patent false lumen appears to be associated with high risks, while complete false lumen thrombosis has been linked to improved outcomes for patients. According to [567], surgical intervention can result in complete false lumen thrombosis in up to 91.3% of patients, along with morphological evidence of remodeling over time. Therefore, it is essential to understand the conditions that enable complete false lumen thrombosis. In addition, studies suggested that reabsorption of the area occupied by the false lumen contributes to patient recovery after complete thrombosis [308,560]. Although early case reports on self-healing of dissection did not specifically refer to complete false lumen thrombosis [271,568], it is reasonable to assume that thrombus formation could potentially stop aortic dissection or even facilitate healing in some patients.

As previously mentioned, the occurrence of late adverse events is associated with aortic dilatation, which, according to various studies, can also be linked to the status of the false lumen. While some studies focused solely on false lumen dilatation, this condition typically occurs alongside overall aortic dilatation, and the two are often interconnected and inseparable. In particular, the absence of a thrombus in the false lumen is often associated with dilatation [313,315,316,326,327,554,560,562]. In other words, a patent false lumen can accelerate growth of the aorta whereas a completely thrombosed false lumen can potentially stop it. Thus, a formed thrombus might act to stabilize the diameter of the aortic or the false lumen. For example, Fattouch et al. [562] reported the lowest growth rates for a completely thrombosed false lumen. Based on their clinical data, a patent false lumen grows at  $2.8 \pm 0.4$  mm/year and a completely thrombosed false lumen at  $1.1 \pm 0.2$  mm/year in patients with and without Marfan sydrome. Another related study was published by Sueyoshi et al. [326] assessing 62 patients. Over a follow-up period of  $\sim$  40.1 months, they observed growth rates in five aortic segments: the arch, the suprarenal abdominal aorta, the infrarenal abdomonial aorta, and the iliac arteries. The mean growth rates were  $3.3 \pm 4.2$  and  $-1.4 \pm 8.6$  mm for a patent or partially thrombosed and a complete thrombosed false lumen, respectively. Another interesting observation pertains to the difference in the mean growth rate between the thoracic and the abdominal aorta, which are 4.1 and 1.2mm, respectively. Overall, the fastest growth rates were found in the descending aorta. After Tsai et al. [515] identified partial false lumen thrombosis as a predictor of mortality in 2007, this variable was further investigated, also with respect to dilatation. Trimarchi et al. [308] reported growth rates in 84 patients of  $2.10 \pm 5.56$ ,  $4.25 \pm 10.18$ , and  $1.51 \pm 5.65$  mm for a patent, partially thrombosed, and completely thrombosed false lumen, respectively. Hence, aortic growth rates were highest in those segments having a false lumen with partial thrombosis. Song et al. [555] made similar observations in the ascending aorta, the arch, and the descending aorta. In contrast, Sueyoshi et al. [327] reported that partial thrombosis was generally not a significant risk factor for false luminal growth. They reported that the highest growth rates were  $4.9 \pm 4.5$ ,  $4.0 \pm 4.3$ , and  $-0.2 \pm 0.6$  mm, for a patent, a partially thrombosed, and a completely thrombosed false lumen, respectively. Different results were obtained when classifying cases with partial thrombosis into saccular and non-saccular formations. The sac-type formation had annual growth rates of  $12.7 \pm 1.1$  mm, while the non-saccular type achieved  $2.6 \pm 2.7$  mm. It is also worth noting that inconsistencies were present in terms of definitions and inclusion criteria. Some studies included intramural hematoma and others did not make a distinction between a completely thrombosed false lumen and an intramural hematoma. Hence, the results are not completely consistent, as discussed before. Albeit in the minority, some studies did not associate the status of the false lumen with its growth [569].

### 4. Experimental in vivo, ex vivo, and in vitro models

Early experiments on aortic dissection used animal, human, and phantom models to study disease onset and progression, microstructural changes in the aortic wall, and to evaluate the safety and efficacy of medical devices such as endovascular implants. At that time there were few or no computational models, so the results of experimental models were crucial and the only viable option. However, with the rise of *in silico* models over recent decades, experimental models have taken on an additional role in validating these models. This development helps to investigate various pathological phenomena without relying on often costly experimental models. Moreover, animal models can help in understanding the microstructural and mechanical changes in the newly formed layers from the acute to the chronic phase. Specifically, creating an artificial dissection in porcine aortas can provide valuable insights into the microstructure of the dissected aorta [455]. For this purpose, mock circulation loops play a crucial role [570], as they enable the controlled evaluation of animal, human, or phantom models with artificial dissections.

This section offers a overview of *in vivo*, *ex vivo*, and *in vitro* experiments on animal, human, and phantom models. *In vivo* experiments are carried out on a living organism of a human or animal, while *ex vivo* experiments test tissue samples taken outside the body in a more artificial environment, such as a mock loop. Note that in the presented *in vivo* models, the aortic dissection was artificially created or modified, typically through surgical intervention. *In vitro* experiments, on the other hand, take place in an artificial environment with phantom models, some of which may be 3D printed.

# 4.1. Replicating aortic dissection to evaluate implant deployment

Quite early on, a number of *in vivo* experimental models were developed to reproduce aortic dissection with the aim of evaluating potential treatments and management strategies. In 1959, one of the first models in this regard was published by Blanton et al. [571]. The aortas of dogs were slit circumferentially to create a cleavage plane within the media and then sutured at the interfaces to maintain the opening, or tear, in a funnel-like position. Of the 26 dogs, 13 developed a dissection extending distally from the intimal opening, while the proximal segment remained collapsed. The time required for dissection was between two and three minutes. Other studies used similar techniques to create artificial dissections [572–579], which were then used to test the effectiveness of the fenestration procedures [580,581] and the efficacy of the use of vascular implant deployment [582–588] or other medical devices [589]. For example, Marty-Ané et al. [586] showed the continued existence of a false lumen in an aortic segment beneath the stent, as shown in Fig. 35(a), while others demonstrated the presence of neointimal proliferation around the inner surface of the false lumen [583], see Fig. 35(b). Another experimental study was published by Moore et al. [590]. They developed an experimental flow loop that included the true and false lumina and tested 24 shear-thinning biomaterials made from blends of gelatin, silicate nanoparticles, and silk fibroin. These biomaterials were able to occlude 99% of blood flow from the true to the false lumen and represent a novel, minimally invasive approach to the treatment of type B dissection by false lumen embolization.









**Fig. 35.** (a) Photograph showing destruction of the dissection after placement of stents along the entire length of the dissection, with a false lumen still visible in the aorta segment beneath the stent. (b) The true lumen appears histologically normal but is compressed, with neointimal proliferation present around the entire false lumen. Figure (a) is adapted from Marty-Ane et al. [586], while (b) is from Moon et al. [583], with permission from Elsevier, respectively.

#### 4.2. Dissection propagation with a single entry tear

In the past, researchers have created artificial dissections by inserting a needle and injecting a fluid, such as India ink, into the aortic media under constant or pulsating pressure. By increasing the pressure gradually, they obtained volume-pressure curves that provided data on (i) the peak pressure, which is indicative of the strength of the media; (ii) the distensibility of the media defined by the initial linear portion when the pressure rises but the volume changes only very little; and (iii) the pressure drop representing the propagation of the dissection. However, long-term effects could not be investigated because the exposure times were only in the range of minutes. In addition, many others have investigated the propagation of dissection and the pressure necessary to achieve it by constructing models that have only one entry tear.

In the first documented needle experiments done by Halliday and Robertson [591], a constant pressure of 600 mmHg was used to induce artificial dissection, but the number of aortas was not enough for meaningful results. Robertson and Viner Smith [592] later conducted consecutive experiments with 42 human aortas, all of which were flat and unstretched. They found a wide range of pressure needed to initiate dissection, the lowest being 230 mmHg and the highest 975 mmHg. No gender differences were observed, but regional differences were seen with the greatest resistance in the ascending aorta, followed by the aortic arch and the lowest in the descending aorta. Additionally, a weak negative correlation between resistance and age was noticed.

Contrary to that, Hirst and Johns [593] employed the water-injection method, a modification of the technique of Milazzo [594], resulting in a range of outcomes. A total of 63 human aortas were tested with a needle, which was inserted either in the center or outer third of the media, as later ascertained by histology. No distinctions was detected between segments, genders, and races. In terms of age-related effects, the least pressure needed to initiate dissection was between 320 and 520 mmHg for aortas from individuals in their first four decades of life, with no significant variations in aortas from older donors. The findings were then validated with two dog aortas with similar results.

Some years later, in 1987, Roach and his colleagues conducted three consecutive needle experiments, one on dog and two on porcine aortas. First, van Baardwijk and Roach [432] conducted 19 successful needle experiments on thoracic aortas from dogs. However, an additional 17 attempts were unsuccessful. They inserted a needle into the aorta and an oscillating pressure was applied until the dissection stopped propagating. The circumference length of the tears was measured after the experiment and ranged between 36 and 81% of the circumference, without taking into account the needle penetration depth. The least important factor in the propagation of dissection was the peak pressure that was measured. On the other hand, a larger tear was linked to a slower rate of dissection, and it was found that the rate of dissection was proportional to the rate of change of pressure (dp/dt) in a significant manner. More specifically, the rate of dissection was determined to be proportional to the square of the rate of change of pressure, i.e. (dp/dt)<sup>2</sup>. An intriguing observation was that the dissection front always advanced during the upstroke of the pressure vave. Second, Carson and Roach [595] conducted experiments with 34 upper descending thoracic porcine aortas to acquire pressure–volume curves. However, their attempts were not always successful. From the obtained pressure–volume curves, the peak pressure for tearing was estimated to be 579 mmHg, the initial slope to be  $3.02 \pm 0.28 \text{ MPa}^{-1}$ , and the work per unit area of tissue required to tear to be 15.9

 $\pm 0.9$  mJcm<sup>-2</sup>. The work per unit area was calculated from the area under the pressure–volume curve when the dissection occurred. Notably, there was no correlation between tear depth and peak pressure. In the third study of their series, Roach and Song [596] harvested 17 porcine aortas and applied their experimental protocol to compare results from five locations between the heart and the lower abdominal aorta. Interestingly, the lower abdominal aorta was found to tear at a lower pressure, but required more energy to propagate the dissection. Peak pressures of  $634 \pm 204$  and  $816 \pm 816$  mmHg were reported, and the dissection work was determined as  $1.88\pm0.89$  and  $11.34\pm4.05$  mJcm<sup>-2</sup> in the lower abdominal and lower thoracic aorta, respectively. They also identified two different types of dissection propagation, as shown in Fig. 36, with mixed types existing. The first type, which was found in the thoracic aorta in 50% of the cases and less than 25% in the abdominal aorta, propagated readily once the tissue tore. The second type, however, required a constant increase of pressure to propagate the dissection. This could be attributed to the structural differences between the two aortas, as observed in previous studies using scanning electron microscopy [597–599]. The thoracic aorta had primarily parallel sheets of elastin, appearing attached to each other with thin fibers, whereas the abdominal aorta had a honeycomb structure. Finally, the experiments did not include results based on direction, meaning that differences between circumferential and axial propagation were not examined. Nonetheless, it was suggested that these differences could be similar. In their last study of this type, Roach and Kratky [600] conducted needle experiments on 48 pig specimens. They applied a constant pressure of 130mmHg to bubbles of different depths; however, the tearing and propagation processes were not affected by the depth of the bleb. They reported an average tearing pressure of 547 mmHg with a range of 208 to 995 mmHg and an average propagation pressure of 54 mmHg, which represent the pressures required to initiate the tear and to further propagate the dissection, respectively.

Differently, Prokop et al. [601] developed an experimental approach to explore the impact of pulsating blood pressure on the extension and rupture of dissections. They used two models, a phantom and a dog model of the aorta, to examine the effects of nonpulsating and pulsating blood flow at different pressures. The phantom model consisted of an outer layer of Tygon tubing and an inner layer of rubber cement, which were connected through holes representing intimal tears. When non-pulsating blood flow was applied up to 400 mmHg, no dissection was observed in the phantom model. When a pulsating blood flow was used, in contrast, a dissection occurred at a pressure of 120 mmHg. Based on the evidence, they determined that the primary sources of the dissecting pressure are the pulsating blood flow and the form of the pulse wave. The results were subsequently validated with descending aortas from 15 dogs. It was proposed that the media surrounding the defect in the aortic wall does not necessarily expand at the same rate or to the same extent as the remaining intact aortic wall. This discrepancy in expansion could create a pressure difference, which then generates a force that causes the dissection. This explanation was supported by the fact that no dissection occurred until the threshold value for the



**Fig. 36.** Pressure–volume curves obtained from needle experiments by creating a bleb on thoracic and abdominal aortas, based on experimental data from Roach and Song [596]. (a) In the thoracic aorta, the slope of the first portion of the curve represents the distensibility, while the tearing pressure corresponds to the maximum pressure. The portion to the right of the peak pressure is associated with dissection propagation. (b) In the abdominal aorta, the initial tear occurs at a relatively low pressure, but a higher pressure is required for dissection to occur. The area under the respective curves reflects the work required for dissection.

rate of change of pressure (dp/dt) was reached, here 790 mmHg/s. Additionally, they concluded that shear forces acting in the direction of the flow did not have a mechanical impact on the dissection propagation, as a dissection was not observed even when the Reynolds number was high. Furthermore, they noted that shear forces could not explain a dissection propagation opposite to the flow direction.

Then, *in vivo* experiments on 30 dogs with three distinct groups were carried out by Carney et al. [602]. They studied the effect of myocardial contractility, or the rate of change of pressure, i.e. (dp/dt), and systolic pressure on the development of aortic dissection. The results revealed that a decreased myocardial contractility by itself did not inhibit the advancement of aortic dissection. Moreover, when hypotension was added to a decreased myocardial contractility, they found that aortic dissection was completely blocked.

In another in vivo experimental study, Senoo et al. [603] divided 49 dogs into two groups, acute and chronic aortic dissections, and monitored them for one year starting three days after the disease onset to study the predisposing factors and the propagation mechanisms in dissection. To initiate the condition, they used a modified version of Blanton's method [571], which involved the surgical creation of a transversely oriented entry tear above the right coronary cusp. They found that the length of the dissection was correlated with the pressure amplitude in systole and diastole and, in addition, that the extent of the dissected layer of the pocket was not a factor in dissection progression. The depth of the entry tear was, however, a major factor influencing the propagation of the aortic dissection. Histological investigations of the aortic wall revealed that the aortic wall was fragile in the acute phase but organized itself two weeks after the onset of the aortic dissection, and that ischemic lesions in the medial layer were observed. They speculated that these changes were secondary to the aortic dissection and caused by the disrupted vasa vasorum. Upon realizing that the inner media, located between the true and the false lumen, was nourished by the diffusion of nutrients from the lumina, they noted the effects of a damaged vasa vasorum. Specifically, the central zone was no longer supplied with nutrients, leading to ischemic changes occurring within three days of disease onset. Moreover, they identified a pseudointima (or neointima) on the inner aortic wall of the false lumen two weeks after the onset of the aortic dissection, consisting of a proliferation of collagen and elastic fibers and a loss of SMCs. However, the elastic lamellae were not disrupted, but instead were stretched. Finally, they identified five factors responsible for dissection propagation: blood viscosity, blood pressure, shearing force, turbulence, and the pressure gradient. They concluded that the pressure gradient and the blood pressure were the two most important factors.

Then, several years later, Mitsui et al. [604] conducted a study on aortas isolated from 99 adult mongrel dogs to investigate the relationship between the depth of an intimal tear and the progression of aortic dissection. Using the Blanton method [571], they created an aortic pocket on the aortic wall and connected the aorta in series with a closed circuit at pressures of 150 mmHg/100 mmHg and a rate of 60 beats per minute. The results showed that the progression of aortic dissection occurred most prominently in the aortic pocket prepared in the first one-third of the external media, with an occurrence rate of 87.5%. In fact, they found no correlation between the pocket width and the progression of the dissection.

In 1998, Tam et al. [605] conducted *in vitro* experiments using 20 porcine thoracic aortas to investigate the influence of entry tear depth on propagation pressure. A saline solution was injected into the medial layer to create a bleb, which was then connected to the true lumen through a circumferential incision. The aorta was pressurized under no-flow conditions until an antegrade or retrograde propagation occurred. Histology was then used to measure the depth of the entry tear, which was between 0.44 and 0.89 mm. The results of the experiment showed that the depth of the intimal tear correlated with propagation pressure, with deeper tears associated with lower propagation pressure.

Peelukhana et al. [606] studied how the geometry of entry tears and hemodynamics affected the propagation of a dissection in *ex vivo* models of 36 porcine aortas, each with a single entry tear. The circumference of the tear was varied between 15 and 65% of the aortic circumference and its axial length between 5 and 32 mm. The depth of the tear was either one third or two thirds of the wall thickness. A pulse duplicator was used to apply a mean pressure of 100 and 180 mmHg and a pulse pressure amplitude ranging from 40 to 200 mmHg. The results were summarized into three main findings: first, the length of the dissection was greater for a deep entry tear,



**Fig. 37.** Flap motion at a given mean pressure of 100 mmHg with changing pulse pressure over the cardiac cycle. At the beginning of the cycle, the flap was curved toward the false lumen and as the vessel diameter increased, it was pushed toward the true lumen. At peak systole, the vessel diameter was at its maximum and the flap was further pushed toward the true lumen. As the diameter decreased, the flap was pushed toward the true lumen resulting in the lowest point during the cycle. The flap returned to its original position as the cycle progressed. Figure is adapted from Peelukhana et al. [606] (licensed under CC-BY 4.0).

but only when the circumference of the tear was >36% and its axial length >12mm. Second, pulse pressure, rather than mean pressure, was the primary factor in flap motion and propagation, and, third, the flap was usually curved toward the true lumen before propagation. Therefore, it can be inferred that the size of the entry tear significantly influences the propagation of dissection. In addition to that, the energy acting on the flap before propagation was calculated by summing the elastic energy stored in the flap, the energy required for the flap and outer wall of the false lumen to move, and the energy needed to break the aortic wall. These findings were consistent with previous studies. As shown in Fig. 37, they also visualized the flap motion at a given mean pressure of 100 mmHg with changing pulse pressure over the cardiac cycle.

In a more recent *in vivo* study, Guo et al. [607] evaluated the correlation between dissection propagation and the depth of the entry tear in 12 porcine aortas. For this purpose, they created models with either a thick or a thin dissection flap, see Fig. 38, and found that the axial propagation length was significantly greater in the thick flap model than in the model with a superficial tear. Additionally, 24 months after the onset of the aortic dissection, the antegrade propagation length was significantly longer in the thick flap model. However, the retrograde propagation length did not differ between the two models. Doppler wire measurements revealed that the average peak blood velocity was higher close to the entry tear in the thick flap model. In addition, they observed an exit tear at the ostium of the aortic branch vessels after 24 months, which they suggested was caused by aortic pressure and shear forces. Histology then showed that the collagen fibers in the outer wall of the false lumen had reached their straightened length and formed a stiff, 'jacket like' tube that prevented the wall from rupturing.

To investigate the progression of dissection, Chi et al. [608] created two silicone models with varying pocket sizes that included a torn area and a primary tear feature on the inner layer. To assess changes in the diameter and volume of the lumina during the study, CT imaging and laser lightening were applied. Pressurization was made possible by a pipe circulation system that did not have pulsating characteristics. Pressure was increased and CT imaging was carried out alternately until the tear development process was initiated. After that, only CT imaging was conducted. Their findings indicated that the model with larger interlayer adhesion damage required lower pressure to initiate dissection. As part of the investigation, they also attempted to present an analytical framework for the general mechanical behavior of dissection initiation based on the prevailing forces.

In a recent exceptional study by Brunet et al. [609], aortic dissection was artificially induced in *ex vivo* aortic segments from 11 New Zealand white rabbits, after which the propagation was visualized. For this purpose, they combined tension-inflation tests with both conventional and synchrotron CT imaging. This synergy made it possible to capture vivid images of macro- and microstructural changes in the aortic wall during dissection, and achieve real-time visualizations with impressive clarity, as depicted in Figs. 39(a) and (b). Utilizing a custom-made tension-inflation device, samples of aortic segments were placed within a 50 mm cylindrical PMMA tube, designed for reduced X-ray interference. An axial pre-stretch of 1.5 was maintained to mimic physiological states. Meanwhile, 3D imaging captured the evolution of the dissection under varying static pressures ranging from 100 to 300 mmHg. During imaging, the pressure was maintained constant. Additionally, tears were induced in both circumferential and axial directions. The scans not only provided detailed contrasts of aortic layers but also highlighted the intricate interplay between tear morphology and pressure during dissection. It is noteworthy that the dissections often covered up to three quarters of the aortic circumference. Mean critical pressures were recorded at  $595 \pm 189$  and  $685 \pm 88$  mmHg for tears in circumferential and axial directions, respectively. It is worth noting that three samples dissected at pressures below 300 mmHg. In most specimens, after the dissection, the external wall of the false lumen failed, leading to vessel rupture. Interestingly, they observed a directional shift of the circumferential tear at 100 mmHg, which



**Fig. 38.** Destruction of the elastic lamellar architecture in the medial layer of the aortic wall was observed through microscopic examination of elastin staining. The elastic fibers on the adventitial side of the false lumen were more stretched than those on the intimal side in both models, with model A having a thin flap and model B having a thick flap. In addition, it was found that the entry tear in model B was deeper in the proximal dissection than in model A. In contrast, in the distal dissection, the exit tears developed superficially, and the flap thickness appeared similar in both models: (a) elastin staining, (b) proximal dissection, and (c) distal dissection. Figure is adapted from Guo et al. [607] with permission from Elsevier.

triggered a dissection between the medial lamellae in the axial direction. This dissection amplified axially at 200 mmHg and advanced to the distal segment at 251 mmHg. Confirmations of these observations based on synchrotron CT images were supported by histological findings, see Fig. 39(c). The study underscored the potential to quantify tear and aortic morphologies at different stages of aortic dissection and establishes a clear association between tear geometry and critical pressure thresholds.

#### 4.3. Effect of intimal tears on hemodynamics and dissection flap

Several research studies have analyzed how the location, size, and depth of intimal tears, mainly with a single entry and exit tear, impact hemodynamics. This includes the pressures and flow velocities within the true and false lumina, and the motion of the dissection flap. These studies utilized animal, human, and phantom models, some of which are illustrated in Fig. 40, to investigate various phenomena. In order to replicate realistic flow and pressure conditions in their respective models, many of these studies developed a mock loop or similar flow circuits.

In 1991, Iwai et al. [614] created experimental phantom models made of styrene–butadiene–styrene rubber, a flexible and compliant material, to analyze the flow patterns in aortic dissection. This study used three phantom models that varied in the location of the intimal tear, which was approximately 5 mm in diameter. The models had either a proximal tear, both a proximal and a distal tear, or only a distal tear. To visualize and analyze the blood flow in the lumina, the flap motion, and the wave propagation, cine PC-MRI and standard magnitude MRI were used. The blood flow in the false lumen was similar with either a proximal tear alone or both proximal and distal tears, but was lowest when there was only a distal tear. The model with only a proximal tear had the highest overall flow rate in the false lumen and the greatest changes in the size of false lumen during the cardiac cycle. They also monitored the characteristic flow jet from the true into the false lumen and *vice versa*. When only a proximal tear was present, a flow jet was observed into the false lumen during liate systole and into the true lumen during diastole and early systole. For the model with both proximal and distal tears, a flow jet into the false lumen was present in mid-systole, and a reversed jet into the true lumen was present in mid-diastole at the upstream entry tear. At the distal entry, the flow jet into the true lumen was observed entering the false lumen during late systole. When the lumina were connected only by a distal tear, the flow jet was observed entering the false lumen during late systole and entring the true lumen during end-diastole and late systole. The results showed that bidirectional flow in the false lumen only occurs when multiple tears were present. When bidirectional flow was present, the upstream and downstream flow waves collided and dispersed in the false lumen.

At the start of the twenty-first century, Chung et al. [615,616] used two *in vitro* models with a pulsating mock-flow loop to investigate the causes of true lumen collapse in aortic dissection. The first model consisted of a compliant, opaque bicycle tire inner



**Fig. 39.** 3D segmentation of a specimen showing a circumferential notch post-aortic dissection: (a) 3D perspective of the specimen at 0mmHg, highlighting the initial notch, and (b) 3D depiction of the dissected sample at 0mmHg showing three cross-sections at different positions along the axial axis with the dissected flap colored in yellow. (c) Synchrotron CT ('sCT') images ('TL': true lumen; 'FL': false lumen) and histology images of a dissected descending aorta, imaged after complete dissection at a pressure of 0mmHg with a critical pressure of 251 mmHg, show a cross-section of the dissection flap indicated by a white arrow. The histology section, which approximately aligns with the synchrotron CT image locations, exhibits morphological variations due to transportation and histological processes (section thickness:  $5\mu$ m thick; original magnification 4 x, Picro-Sirius red staining). Figure is adapted from Brunet et al. [609] with permission from Elsevier.

tube combined with a PTFE graft material, while the second model was a rigid, transparent model made of PTFE. Through these models, they studied the effects of anatomic factors such as the location, number, and size of intimal tears, as well as the branch-vessel flow distribution. They also examined the effects of physiological factors such as the peripheral resistance in the branch vessels, the pump rate, and its output. Results showed that the collapse of the true lumen was associated with an increase in the size of the proximal tear. Furthermore, a decrease in the false lumen outflow rate was attributed to an occlusion of the false lumen branch vessels, while an increase in the true lumen outflow rate could be achieved by lowering the peripheral resistance in the true lumen branch vessels. Thus, both anatomical and physiological factors can potentially affect the true lumen collapse. They hypothesized that the differing inflow and outflow rate ratios between the true and false lumina create a pressure difference between the lumina, which subsequently moves the flap. In fact, it was more difficult to relieve the true lumen collapse than to prevent it. Repair of a proximal tear was the most effective treatment for preventing true lumen collapse in aortic dissection because it decreased the blood flow into the false lumen.

The impact of morphological parameters of the intimal tear (e.g., location, number, and size) on the pressure in the false lumen was also investigated by Tsai et al. [610]. They used three phantom models simulating the most common configurations in aortic dissection patients, see Fig. 40(a). These models included a proximal and a distal tear, only a proximal tear, and only a distal tear. The tear size varied between 3.2 and 6.4 mm. Each model was connected to a pulsating pump to mimic the blood flow conditions in the lumina. The results showed that with a proximal and a distal tear, the systolic blood pressure was slightly lower and the diastolic pressure was slightly higher in the false lumen when compared with the true lumen. When the distal tear was absent, the diastolic pressure was elevated in the true lumen and the systolic pressure was slightly higher. Conversely, in the absence of a proximal tear, the diastolic pressure in the false lumen was observed with a smaller proximal tear size and the absence of a distal tear. These *in vitro* models suggest that distal tears are associated with significant pressures in the false lumen that may not lead to thrombosis and remodeling.

In 2011, Dziodzio et al. [267] conducted an *ex vivo* study of the entire thoracic aorta of 26 pigs, including the supra-aortic branches. In the experiment, pulsating blood pressure was applied to the aortic annulus and intimal tears occurred in two distinct locations. The first location was 15 mm downstream from the origin of the left subclavian artery, specifically at the concavity. This location typically represents a clinical manifestation that is considered rare. The second location was at the convexity site of the aorta. Analysis of the dissections revealed that 65% had a retrograde component, with the median antegrade and retrograde propagation lengths being 65 and 20 mm, respectively, when the intimal tear was located at the convexity site. The retrograde dissection usually stopped at the left subclavian artery. For an intimal tear at the concavity site, the median retrograde propagation was 21 mm, and in 16% of cases the dissection extended into the ascending aorta. However, in 31% of cases, antegrade propagation was stopped by the ligamentum arteriosum, which acted as an anatomical barrier. Additionally, retrograde propagation occurred at a later stage of the experiment, while antegrade propagation occurred in the initial stage.

One year later, Qing et al. [617] used 15 porcine aortas as *ex vivo* models to evaluate the hemodynamics in three different model types, which were distinguished by the location and number of intimal tears. They also developed models with only a distal tear, only a proximal tear, and tears present in both the distal and proximal areas. A pulsating pump was used to replicate the cardiac cycle, causing the dissection flap to move back and forth due to the pulsating blood pressure. According to the results, the flap played a role in controlling the blood pressure between the lumina. For instance, during systole, the false lumen began to expand, leading to compression of the true lumen. Additionally, in the model with only a proximal tear, the higher pressure in the false lumen caused



**Fig. 40.** Four representative experimental models, each embedded into a flow circuit to simulate realistic flow and pressure conditions. These conditions are achieved through the use of either a pulsatile pump and control system or a simple pump to apply pressure. The models are as follows: (a) flow circuit featuring an aortic dissection model made of polymeric silicone ('TL': true lumen; 'FL': false lumen); (b) flow circuit utilizing a porcine model with an aortic dissection, where intimal tears were created using plastic tweezers (i) and scalpels (ii); (c) MRI-compatible flow circuit with a 3D printed aortic dissection model that has varying tear sizes; and (d) inverted porcine aorta with an imposed dissection (i), accompanied by the bench test setup (ii). Figure (a) is adapted from Tsai et al. [610] with permission from Elsevier, while (b) is from Liang et al. [611] with permission from Sage Publications, (c) is from Zimmermann et al. [612] (licensed under CC-BY 4.0), and (d) is from Ahuja et al. [613] (licensed under CC-BY 4.0).

more significant compression of the true lumen.

Rudenick et al. [618] conducted an *in vitro* study where they used a phantom model made of latex with synthetic geometry to create three different models. These models had either a distal tear, a proximal tear, or both to analyze the pressure, velocity, and flow pattern in the true and false lumina. The diameter of the intimal tears varied between 4 and 10mm, corresponding to 10 and 25% of the dissected segment. Larger tears caused pressure equalization in both the true and false lumina, leading to slower velocities and a complex flow pattern in the false lumen. In contrast, smaller tears not only resulted in lower pressure but also in higher tear velocities and a well-defined flow pattern within the false lumen. This suggested that the hemodynamics of the false lumen are highly dependent on the cumulative size of the tears. The study also revealed that peak systolic pressure in the false lumen was lower and occurred later in the cardiac cycle than in the true lumen. Furthermore, blood flow across the intimal tear was bidirectional and independent of the model used. Interestingly, there was minimal blood flow between a distal and proximal tear or *vice versa*.

In their *ex vivo* experimental study, Faure et al. [619] investigated whether the location of an intimal tear in human aortas is related to the pattern of dissection propagation and also evaluated the suitability of endovascular implants for human use. To accomplish this, they harvested 20 human aortas and created a 20 mm intimal tear below the left subclavian artery at four different locations: lateral, medial, anterior, and posterior. They then used a bench-top pulsating flow model to induce an antegrade dissection propagation. The results showed that the antegrade propagation always reached at least the celiac trunk. In 80% of cases, the dissection propagated to the infrarenal aorta. Notably, when the intimal tear was located at the lateral or medial site, the false lumen typically involved both the left and right renal arteries.

In a series of studies, Veger and his colleagues then studied different *ex vivo* porcine aortas. In one study, Veger et al. [620] created artificial dissections in two *ex vivo* porcine aortas to examine the role of branch vessels in type B dissection. For this purpose, a thin cannula was inserted into the false lumen, mimicking a branch vessel originating from it. The aorta was then placed in a validated circulatory system, which replicated physiological pulsatile flow and pressure characteristics of 130 mmHg/70 mmHg. This cannula was connected to a small silicone tube equipped with an adjustable valve mechanism. Three distinct valve settings were applied to create outflow from the false lumen: fully closed, half-open (50%), and fully open. Lumen areas and flow rates were measured using 4D-flow MRI with periodic triggering. They found that an increase in antegrade outflow through the branch vessel of the false lumen significantly expanded the mean false lumen area at both proximal and distal locations in both models. In conclusion, when no distal entry tear was present, increasing antegrade outflow through a branch vessel originating from the false lumen resulted in an expansion of the cross-sectional area of the false lumen. Utilizing the same validated *in vitro* circulatory system, Veger et al. [621] first conducted

measurements of the true and the false lumen volume via CT imaging across six porcine models. In these models, a dissection flap was created in an aortic segment and then axially divided to induce fenestrations at various locations. Subsequently, in their third consecutive study, Veger et al. [622] employed three *ex vivo* porcine aortas and performed pressure measurements in both the true and the false lumen at several locations. After conducting baseline experiments, the aortic wall elasticity was adjusted with silicone, and the experiments were repeated to examine the effects of arterial stiffening and the resultant aortic dilatation. Finally, Veger et al. [623] conducted a study using a single dissected porcine model to investigate the impact of heart rate on hemodynamics, utilizing 4D-flow MRI data. Notably, also calculated WSS in the false lumen from the imaging data. For validation purposes, a healthy porcine aorta was also tested.

A silicone model of the aorta with a single proximal tear was developed by Birjiniuk et al. [624], utilizing CT imaging data. Flow velocity was measured and flow rates were calculated using a 4D-flow MRI. The study aimed to observe the properties of aortic flow and suggested that the model could be applied to more complex dissection geometries in the future. As a result of this baseline study, three consecutive studies were published, all of which were based on the same model. In the first study, Birjiniuk et al. [625] developed two models: a normal aorta model and a dissected aorta model, none of which included branches. The dissected model had a proximal and a distal tear, each of which was incised circumferentially along the flap on the lateral wall. The study found pockets of reverse flow and vortices primarily in the false lumen of the dissected aorta, with flow reversal being significantly greater in the proximal to middle dissection area compared with normal. The pulsatility caused unsteady vortices and a pumping motion of the distal dissection flap that corresponded to a flow reversal. The time-averaged velocity near stagnation in false luminal flow reversal could induce future thrombosis. Focal vortices could identify the location of tears that can potentially be covered with endovascular implants. The shape of the dissection flap changed over the extent of the dissection, mainly assuming a sinusoidal cross-sectional profile, with the greatest displacement of the flap observed at the proximal and distal tears. During peak systole, the distal portion of the flap remained collapsed onto the false lumen near the distal tear, achieving maximum true lumen distension. In their second study [626], they extended the model by adding linear, transverse fenestrations, approximately 25 mm in length at locations 50 and 75% of the distance from proximal to distal tear, creating models with two, three, and four fenestrations. The study revealed the presence of pulsatile vortices and jet-like structures at fenestrations immediately distal to the proximal tear. The addition of fenestrations significantly reduced false lumen flow reversal, from  $19.2\pm3.3\%$  in dissections with two tears to  $4.67\pm1.5$  and  $4.87\pm1.7\%$  with each subsequent fenestration. In contrast, increasing pressure did not affect flow rates, flow reversal, or vortex formation. Moreover, increasing the number of additional fenestrations decreased flow reversal compared with two-tear dissection, which may prevent false lumen thrombosis, promoting persistent false lumen flow. Finally, in their third consecutive study, Birjiniuk et al. [626] extended the model to study the effect of endovascular stent-graft deployment in a model with multiple fenestrations, where fenestrations were placed similar to the previous study [626]. An initial 60 mm graft was deployed to cover the proximal tear and to provide coverage of 60% of the initial dissection. Each subsequent graft was inset in the previous one to provide continuous coverage of the dissection. The study found that complete obliteration of the false lumen was achieved in the grafted aorta, resulting in normal parabolic flow profiles in the true lumen. A blind false lumen pouch was created distal to the grafted region with low velocity and highly reversed flows. In the distal free false lumen segments, flows were comparable to those without grafting. Visualization studies revealed antegrade flow in these regions with lefthanded vortices of true-to-false lumen. The study suggests that coverage of the proximal tear alone is insufficient to restore true lumen area and flow along the entire extent of the dissection, suggesting inadequate collapse of the false lumen distal to the grafted region.

Marconi et al. [627] presented an *in vitro* study on the effect of location and size of the intimal tear on the hemodynamics in dissection using a pulsatile mock loop. A simplified geometry was 3D printed using a transparent rigid resin to form a stiff case. The silicone elastomer used in the experiment imitated the dissection flap separating the two lumina. The true lumen had a wall thickness of 3 mm, while the false lumen and the flap had thicknesses of 1 and 2 mm, respectively. In total, six access points were placed to measure pressure at different positions. Circular tears were created with three different diameters: 3, 6, and 10 mm, and in three variations: proximal, distal, and both proximal and distal. In addition, one reference configuration without tears was created. According to the study, a single intimal tear leads to higher relative false lumen pressure than two tears, particularly for the configuration with a proximal tear. The difference in relative false lumen pressures between a proximal and distal tear was minimal. Furthermore, the diameter of the tear affected the false lumen pressure, with a 3 mm diameter resulting in a higher relative false lumen pressure than a 10 mm diameter. Interestingly, the intermediate diameter of 6 mm consistently produced lower relative false lumen pressure compared with the effects of proximal tears within physiological ranges. The study concluded that tears have a considerable impact on the connection between true and false lumina, but in different cases other factors such as the vessel wall elasticity, the presence of false luminal thrombus, and aortic geometry may can also play a role in various behaviors observed in clinical practice.

Following their initial study on layer-specific material properties in aortic dissection [455], Ahuja et al. [613] employed five *ex vivo* porcine models. These models were created with both entry and exit tears, designed for bench testing under static pressure, as shown in Fig. 40(d). In the experiment, a balloon was inflated to apply bending forces to either the middle or distal region of the dissection flap within the static aortic pressure environment. The main objective was to identify the radial pressures required to reassemble the middle and distal ends of the dissected aorta. To ensure the validity of their findings, they compared the experimental data with computational models, as described in Section 5.2.

Canchi et al. [628] conducted an *ex vivo* study using 16 porcine aortas to assess the difference in luminal pressure and configuration of the dissection flap following dissection. Similar to Peelukhana et al. [606], they created a single proximal tear in each aorta, and used a pulse duplicator setting to apply an initial pulse pressure of 120 mmHg/80 mmHg, which increased to 140/80 and 160 mmHg/



**Fig. 41.** The boxplots illustrate (a) the spatial variation and mean lumen pressure difference between the true lumen ('TL') and the false lumen ('FL') compared for conditions both with and without a distal re-entry tear, and (b) the distribution and the mean percentage of true lumen crosssectional area at peak systole and diastole, compared under conditions with and without re-entry tear development. Mean values are denoted by a diamond symbol within each boxplot. Figure is adapted from Canchi et al. [628] (licensed under CC-BY 4.0).

80 mmHg during the experiment. The pressure difference was then measured at three locations, namely proximal, middle, and distal, along the dissection flap, both with and without an additional distal tear. The results showed a mean pressure difference of 4.6 mmHg across all locations when only a proximal tear was present, with higher pressures in the false lumen and a decreasing pressure difference along the length of the dissection flap, reaching a maximum difference at the distal end. However, the pressure difference disappeared with a distal tear, see Fig. 41(a). The study also evaluated the configuration of the dissection flap and found that it curved toward the false lumen during systole in 37.5% of cases, which increased to 75% with a distal tear. A dynamic movement of the dissection flap over the cardiac cycle was also observed, attributed to the pulsatile pressure. This, combined with the higher pressure in the false lumen, caused compression of the true lumen, as shown in Fig. 41(b), through the cross-sectional area of the true lumen. A notable finding was that increasing pulse pressure did not result in significant changes in the dissection flap.

In their research, De Beaufort et al. [160] examined the precision of 4D-flow MRI scans for evaluating true and false lumina flow rates in *ex vivo* porcine dissection models. This involved the inversion of a porcine aorta, the creation of intimal tears, and the expansion of the false lumen using a balloon catheter. After these procedures, the porcine model was integrated into a pulsatile flow loop. Flow rates were measured by two procedures, and the averages of these measurements were used for comparisons. The 4D-flow MRI and 2D PC-MRI were compared against each other, as well as against a reference standard of sonotransducer flow measurements. The conclusion drawn from this study was that the 4D-flow MRI method is a reliable tool for assessing flow within the true and false lumina of the aorta.

Salameh et al. [629] created an *in vitro* model using eight plexiglass models to study the hemodynamics of patent and non-patent aortic dissection. The model mimicked only the dissection present in the aortic arch and was attached to a continuous flow circulatory system with water acting as a blood surrogate. Particle image velocimetry was utilized to measure the velocity field, while laser-induced fluorescence was used to visualize the flow and quantify the perfusion of injected dye, similar to contrast-enhanced CT. The model was separated into two categories: four with a patent false lumen and four with a non-patent false lumen. Each group was then divided into four cases, with one of the tears having a different size than the other model and a rigid flutter feature appearing at either the proximal or distal tear. The location of the initial tear in the aortic arch in a patent false lumen had a significant impact on the velocity of blood flow in the true and false lumina. The velocity of blood flow was determined by the size of the tear, with the smaller tear providing the maximum velocity in the false lumen. As the true and false lumina merged at the distal tear, the velocity difference created vortices that moved downstream along the wall of the non-dissected aorta. In a non-patent false lumen, there was no flow and all the blood flowed through the true lumen. However, in a patent false lumen, the perfusion rate of the dye was comparable to that in the true lumen and occurred simultaneously.

A combined experimental and computational study was conducted by Zadrazil et al. [630] to analyze the flow characteristics in the true and the false lumen over a range of intimal tear sizes. To do this, a simplified model made of perfluoroalkoxy was created, which included the ascending aorta, the aortic arch, and the descending aorta. The model was designed to be a 1:2 scale of an idealized aortic geometry, with a constant internal diameter of 16 mm. To obtain realistic flow conditions, deionized water, seeded uniformly with inert particles, was pumped through the simplified model with a pulsatile blood pump. Four different tear sizes and size ratios were tested, and the phase-averaged velocity vector maps at mid-acceleration, mid-deceleration, relaminarization, and peak-systole were measured. The experimental results showed that an increase in the distal tear size increased the flow rate, decreased the WSS, and decreased the true-to-false lumen pressure difference. Conversely, an increase in the proximal tear size increased the flow rate, the WSS, and the pressure difference. Notably, the highest WSS was located at the edge of the intimal tear. The experimental results were then used to validate an *in silico* model, which is further discussed in Section 6.1.

Liang et al. [611] conducted a study using 24 porcine aortas to investigate the hemodynamic effects of various *ex vivo* aortic dissection models. The aortas were randomly divided into four groups, with group A serving as the control and groups B to D representing the aortic dissection group. To create the aortic dissection models, as shown in Fig. 40(b), an intimal tear was made 30 mm below the left brachiocephalic artery using a scalpel. This tear was then enlarged to 18 to 20 mm using plastic tweezers. A spatula was used to separate the intima and media until the false lumen reached a length of 150 (group B) or 200 mm (group C). Group D had a 200 mm false lumen with a proximal and a distal tear. The groups were placed in a mock circulation loop to simulate realistic flow and pressure conditions, with the flow distribution rate of the aortic branches calculated, and Doppler ultrasound used to visualize the aortic dissection group compared with the control group. The dissection flap showed pulsatile movement, particularly in groups B and C without a distal tear, suggesting that the flap motion was associated with comparatively higher pressures. In contrast, the pressure in the true and false lumina of model D was almost the same due to the lower flap oscillation. The study also revealed that the flow velocity at the distal end of the false lumen in the aortic dissection group was lower than that in the true lumen, which may lead to stagnant blood flow and platelet aggregation.

In the study by Chen et al. [631], patient-specific silicone models of both a normal aorta and an aortic dissection were incorporated into a mock circulation loop that enabled precise measurements of flow rates and pressure conditions in the aortic branches. Attached to each of the 12 branching vessels was a three-element Windkessel model. Significant differences were observed when comparing the normal and dissected aortic models under resting conditions. In particular, the dissected aortic model exhibited elevated systolic and diastolic pressures. Additionally, the time-average velocity was higher in the true lumen than in the false lumen. The direction of blood transport between the true and false lumina varied in different fenestrations, providing further insights into the intricate dynamics of aortic dissection.

Based on a previous study on a normal aorta [632], Zimmermann et al. [612] performed an *in vitro* study using 4D-flow MRI and coupled *in silico* simulation through FSI to examine how changing the size of proximal and distal tears affects hemodynamics. They utilized these *in vitro* results to confirm the accuracy of the coupled *in silico* simulations. The study used a patient-specific 3D printed aortic dissection model, which was created in three versions: one original model based on CTA, and two others with smaller proximal and distal tears, respectively. The model was placed in a flow-controlled and pressure-controlled setup to perform MRI imaging (Fig. 40 (c)). The luminal pressure data were recorded at different locations in the true and false lumina, while an MRI-compatible flow circuit with glycerol-water as a blood-mimicking fluid was used for MRI signal enhancement.

The study found that each of the three models had unique flow patterns and velocity distributions, as shown in Figs. 42(a) to (c). The proximal tear region exhibited highly complex flow patterns, while helical flow patterns were observed in the proximal true lumen near the proximal tear. The modified models in the distal part of the dissected aorta showed increased flow velocities in the true lumen, consistent with higher true lumen net flow volumes in these models. The model with the smaller distal tear also showed a recirculation zone distal to the distal tear, while the other models showed unidirectional laminar flow in this region. During diastole, additional flow oscillations were observed through the proximal tear and helical flow in the false lumen was retained. The size of the tears had a significant impact on the inter-luminal pressure differences and flow splits, with the difference between peak and systolic pressure increasing in the reduced proximal tear model but becoming negative in the reduced distal tear model. The inlet to outlet pressure drops were lowest in the original model and larger for the modified models, while both modified models showed a significant reduction in false lumen flow ratio compared with the original model. This suggested that the flow is determined by the total resistance and not by the location of the narrowing. The model with the smaller distal tear had increased outflow resistance and false lumen flow ratio, which contributed to false lumen pressurization. In addition, they observed that in the model with a smaller proximal tear, the flow jet impinged on the opposite wall of the false lumen through the entry tear. They hypothesized that this could potentially result in tissue degradation and destructive remodeling via mechanobiological pathways, leading to false lumen dilatation due to the presence of local pressure gradients at the inner aortic wall. Readers are referred to their follow-up study for further investigation on this matter [342], see Section 7.2.

The main limitation of this study is related to the comparison between 4D-flow MRI and FSI simulations, which requires further attention. When comparing the visualized streamlines from 4D-flow MRI and FSI simulations, they appear remarkably similar, including recirculation zones, flow helices, and the flow jet through the tears. However, FSI simulations generally showed higher flow velocities compared to 4D-flow MRI, except during diastole where 4D-flow MRI displays higher velocities. They attributed these differences to two factors: first, the lower resolution of 4D-flow MRI compared to 2D PC-MRI may reduce peak flow velocities. Although 4D-flow MRI often underestimates peak velocities compared to 2D PC-MRI, this difference depends generally on several factors [633]. Second, FSI simulations experience lower damping due to differences between the computational model and the 3D printed model, as the latter has larger cross-sectional areas, leading to greater compliance, possibly due to underestimated resin stiffness. This is evident from the initial well-matched flow rates at the landmark inlet, but higher systolic and lower diastolic flow rates in FSI simulations at all downstream landmarks, as shown in their study. Overall, the well-matched streamlines between 4D-flow MRI and FSI simulations are particularly noteworthy.

In another study series, Franzetti et al. [634] developed a mock loop in which a computer-controlled pulsatile pump system combined with a tunable three-element Windkessel simulator was used to reproduce *in vitro* vascular physiological conditions to study complex patient-specific vascular pathologies. In this initial study, this mock loop was tested to verify its overall mechanical and hydraulic behavior. Subsequently, a follow-up study was published by Bonfanti et al. [635], which was based on the previously developed mock loop. Using 3D printing technology, a rigid and transparent 3D model of a patient-specific chronic aortic dissection was fabricated from TuskXC2700T. The model featured inlet and outlets such as the brachiocephalic trunk, left common carotid, left subclavian artery, and descending aorta, where it was connected to the remaining experimental setup. However, for simplicity,



**Fig. 42.** Comparison between 4D-flow MRI images and FSI simulation results showing the streamlines at peak systole for three models: (a) original model ('TL': true lumen; 'FL': false lumen), (b) model with a smaller proximal tear, and (c) model with a smaller distal tear. Each model has distinct local flow characteristics. First, the region of the proximal tear experiences increased flow velocities, particularly in model (b), and there is localized helical flow in the proximal true lumen and the false lumen near the proximal tear. Second, in models (b) and (c), the flow velocity in the true lumen is increased (arrow), and finally, the measurements in model (c) show a flow jet through a small distal tear with recirculating true lumen flow distal to the distal tear. Figure is adapted from Zimmermann et al. [612] (licensed under CC-BY 4.0).

abdominal aortic branches were excluded. According to the CT scans, there was a proximal tear approximately 10mm distal to the proximal end of the dissection. No further communication between the lumina was evident from the CT data. In vivo waveform data were obtained from 2D PC-MRI scans acquired at multiple locations along the aorta. The study found that during the acceleration phase and the peak systolic phase, the velocity profiles in the aortic arch were shifted toward the inner curvature of the arch. Moreover, a highly organized motion with parallel streamlines was observed during the systolic phase, as shown in Fig. 43. Flow separation and bidirectional velocity profiles were observed during the deceleration phase and early diastole. Flow in the aortic arch during diastole was characterized by disorganized streamlines. Particle image velocimetry-derived phase-averaged velocities showed organized and high-velocity flow in the narrowed true lumen during the systolic phase, whereas the false lumen was dominated by vortical and complex flow structures during both systole and diastole. The absence of distal tears in the patient led to a zero net-flow entering the false lumen via the proximal tear, with recirculating flow patterns throughout the cardiac cycle. The flow structure and velocity magnitude in the false lumen remained relatively constant throughout the cardiac cycle, with slightly higher velocities observed during diastole. The results were then used to validate in silico hemodynamics simulations. In their study, Franzetti et al. [636] reexamined and expanded previous findings and investigated the evolution of vorticity, Reynolds shear stress distribution, and turbulent kinetic energy. Vortex formation with low shear rate values occurred in the false lumen throughout the cardiac cycle. This pattern has been linked to false lumen thrombosis due to the presence of small vortices and stagnant flow. During acceleration, deceleration, and diastole, the flow fields became disorganized and flow reversal occurred, leading to enhanced flow mixing, increased velocities and velocity gradients. Reynolds shear stress and turbulent kinetic energy also exhibited higher values during deceleration and diastole,



**Fig. 43.** Phase-averaged velocity magnitude fields  $\langle u \rangle$  were analyzed in the proximal part of the true ('TL') and the false lumen ('FL') at (a) peak systole and (b) diastole. The velocity magnitude is color-coded, while the flow direction is indicated by overlapping streamlines. Figure is reprinted from Franzetti et al. [636] (licensed under CC-BY 4.0).

while the latter showed low values during acceleration and peak systole due to the stabilizing effect of flow acceleration. Overall, hemodynamic indices have been associated with platelet activation, hemolysis, and blood coagulation.

Building on previous studies and the developed experimental model, a reduced-order model was subsequently developed to encapsulate the primary flow features within a patient-specific aortic dissection [637]. A proper orthogonal decomposition and robust principle component analysis was applied on *in vitro* hemodynamic data acquired by particle image velocimetry. The decomposed flows were then compared with those derived from hemodynamic simulations for the same geometry and flow conditions. This approach significantly simplifies the problem and promotes computationally more efficient flow simulations. Such advances could pave the way for translating these models into clinical practice.

Lastly, Aghilinejad et al. [638] developed a framework for 3D printing compliant, patient-specific aortic dissection phantom models. This framework was based on CT imaging and utilized a novel deep-learning-based segmentation method. Notably, data from 19 patients were used in the process. The deep-learning architecture was trained on a data set of 15 unique CT imaging scans from dissection subjects and then blind-tested on four additional sets of scans intended for fabrication. After segmentation, 3D models were created and printed using polyvinyl alcohol coated with latex. These models were then coated with latex to create compliant, patient-specific phantom models and implemented in a systemic circulation system. The *in vitro* experiments showed that the fabricated phantoms provided physiologically accurate pressure results that were comparable to MRI scans.

## 4.4. Key findings and publication index

To summarize this section, we present the publication index of experimental models with a focus on *in vivo*, *ex vivo*, and *in vitro* approaches in Table 5. Essential information about the modeling approaches and the objectives of each study is included. Finally, the main findings are listed conclusively.

- Needle experiments yielded pressure–volume curves necessary for dissection propagation, which were classified into three distinct categories: distensibility, peak pressure, and dissection propagation. The pressure required for dissection propagation may vary, depending on the site and intrinsic microstructure of the aorta. An increase in pressure is needed for propagation in the abdominal aorta, while the required pressure for dissection propagation in the thoracic aorta decreases as the volume of injected fluid increases.
- Blood pressure and pressure gradient were the most important factors in dissection propagation. In fact, the pressure gradient, referring to the pressure change over time, i.e. (dp/dt), appears to have the greatest impact.
- The dynamic motion of the dissection flap over the cardiac cycle was driven primarily by the pressure difference between the true and false lumina. This difference, typically in the range of 5 mmHg, was determined by the location, number, and size of the intimal tears. In particular, a larger size of intimal tears size usually resulted in lower pressure differences.
- Anatomical barriers such as the ligamentum arteriosum or plaque in atherosclerotic aortas could stop or impede dissection propagation. Additionally, antegrade propagation typically occurred in the early stage of aortic dissection, while retrograde propagation was observed at a later stage.
- Luminal pressures in both the true and false lumina were influenced by the location, number, and size of intimal tears. During diastole, the false lumen exhibited higher pressures, whereas during systole, the pressures in the false lumen was lower. A large entry tear led to equalized pressure between the lumina, while a smaller entry tear created a higher pressure difference.

## Table 5

Publication index of experimental *in vivo, ex vivo,* and *in vitro* models related to aortic dissection, ordered by publication year (FL: false lumen).

| Report                                  | Year          | No. of models                        | Model<br>type | Main objective(s)   |
|---|---------------|--------------------------------------|---------------|---|
| Aartic dissection replication           |               |                                      | <b>91</b>     |   |
| Blanton et al [571]                     | 1050          | Dog (26)                             | In vivo       | Replication of a rtic dissection  |
| Trent et al. [582]                      | 1990          | Dog(12)                              | In vivo       | Efficacy of endovascular implants   |
| Moon et al. [583]                       | 1993          | Dog(12)                              | In vivo       | Efficacy of endovascular implants   |
| Kato et al. [584]                       | 1994          | Dog(12)                              | In vivo       | Efficacy of endovascular implants   |
| Kato et al. [585]                       | 1995          | Dog(10)                              | In vivo       | Efficacy of endovascular implants   |
| Marty-Ané et al. [586]                  | 1995          | Dog (14)                             | In vivo       | Efficacy of endovascular implants   |
| Razavil et al. [572]                    | 1998          | Pig (15)                             | In vivo       | Efficacy of endovascular implants   |
| Morales et al. [580]                    | 1998          | Dog (7)                              | In vivo       | Effectiveness of fenestration   |
| Zannoli et al. [589]                    | 2007          | Silicon (1)                          | In vitro      | Efficacy of medical device (intravascular passive                                   |
|   |               |                                      |               | counterpulsating damper)  |
| Cui et al. [575]                        | 2009          | Dog (12)                             | In vivo       | Replication of aortic dissection  |
| Tang et al. [576]                       | 2010          | Dog (20)                             | In vivo       | Replication of aortic dissection  |
| Fujii et al. [573]                      | 2000          | Pig (12)                             | In vivo       | Replication of aortic dissection  |
| Terai et al. [574]                      | 2005          | Dog (15)                             | In vivo       | Replication of aortic dissection  |
| Okuno et al. [577]                      | 2012          | Pig (14)                             | In vivo       | Replication of aortic dissection  |
| Faure et al. [587]                      | 2015          | Human (15)                           | Ex vivo       | Efficacy of endovascular implants   |
| El Batti et al. [581]                   | 2018          | Sheep (17)                           | In vivo       | Effectiveness of fenestration   |
| Wu et al. [578]                         | 2021          | 3D printed materials (4)             | In vitro      | Replication of aortic dissection with 3D printed materials                          |
| Liu et al. [579]                        | 2023          | Dog (18)                             | Ex vivo       | Efficacy of endovascular implants   |
| Mohl et al. [588]                       | 2024          | Transparent flexible resin (1)       | In vitro      | Efficacy of endovascular implants   |
| Dissection propagation with a single er | try tear      |                                      |               |   |
| Halliday & Robertson [591]              | 1946          | Human (n.a.)                         | Ex vivo       | Impact of blood pressure on dissection propagation<br>with needle-based experiments |
| Robertson & Viner Smith [592]           | 1948          | Human (42)                           | Ex vivo       | Impact of blood pressure on dissection propagation                                  |
|   | 1910          |                                      |               | with needle-based experiments   |
| Hirst & Johns [593]                     | 1962          | Human (63) & Dog (2)                 | Ex vivo       | Impact of blood pressure on dissection propagation                                  |
|   |               |                                      |               | with needle-based experiments   |
| Prokop et al. [601]                     | 1970          | Tygon – rubber cement (1) & Dog      | In vitro &    | Impact of pulsating blood pressure on dissection                                    |
|   |               | (15)                                 | ex vivo       | propagation   |
| Carney et al. [602]                     | 1975          | Dog (30)                             | In vivo       | Impact of rate of pressure change on dissection                                     |
|   |               |                                      |               | propagation   |
| Roach et al. [432,595,596,600]          | 1987–99       | Dog (36)/Pig (34)/Pig (17)/Pig (48)  | Ex vivo       | Impact of rate of pressure change on dissection                                     |
|   |               |                                      |               | propagation with needle-based experiments   |
| Senoo et al. [603]                      | 1989          | Dog (49)                             | In vivo       | Impact of intimal tear (location, size, depth) on                                   |
|   |               |                                      |               | propagation pressure  |
| Mitsui al. [604]                        | 1994          | Dog (99)                             | Ex vivo       | Impact of intimal tear (depth) on propagation                                       |
|   |               |                                      |               | pressure  |
| Tam et al. [605]                        | 1998          | Pig (20)                             | In vitro      | Impact of intimal tear (location, depth) on   |
| Desivities at al. [COC]                 | 0017          | <b>D</b> '- (0()                     | <b>F</b>      | propagation pressure  |
| Peelukiialia et al. [600]               | 2017          | Pig (36)                             | EX VIVO       | homodynamics (flan mation   |
| Cup at al [607]                         | 2010          | $\operatorname{Bia}(12)$             | In vino       | Import of intimal toor (location, donth) on   |
|   | 2019          | Pig (12)                             | 111 1110      | hemodynamics /dissection_propagation  |
| Brunet et al [609]                      | 2023          | Rabbit (11)                          | Fr vivo       | Visualization of intimal tear/propagation/flap                                      |
|   | 2020          |                                      | LA VIVO       | configuration with CT imaging/histology: impact of                                  |
|   |               |                                      |               | blood pressure on propagation   |
| Wang et al. [639]                       | 2023          | Pig (70)                             | Ex vivo       | Impact of intimal tear (location, size, depth) on                                   |
| 0                                       |               |                                      |               | propagation pressure  |
| Effects of intimal tears on hemodynamic | cs and dissec | tion flap                            |               | 1 1 0 1   |
| Iwai et al. [614]                       | 1991          | Styrene-butadiene-styrene rubber     | In vitro      | Impact of intimal tear (location) on  |
|   |               | (3)                                  |               | hemodynamics/flap motion  |
| Chung et al. [615,616]                  | 2000          | Bicycle tire inner tube – PTFE graft | In vitro      | Impact of intimal tear (location, size) on  |
|   |               | (1) & PTFE (1)                       |               | hemodynamics/flap motion  |
| Tsai et al. [610]                       | 2008          | Polydimethylsiloxane (1)             | In vitro      | Impact of intimal tears (location, size) on   |
|   |               |                                      |               | hemodynamics  |
| Dziodzio et al. [267]                   | 2011          | Pig (26)                             | Ex vivo       | Impact of intimal tears (location) on dissection                                    |
|   |               |                                      |               | propagation   |
| Qing et al. [617]                       | 2012          | Pig (15)                             | Ex vivo       | Impact of intimal tear (location) on  |
|   |               |                                      |               | hemodynamics/flap motion  |
| Rudenick et al. [618]                   | 2013          | Latex (8)                            | In vitro      | Impact of intimal tear (location, size) on  |
|   |               |                                      | _             | hemodynamics  |
| Faure et al. [619]                      | 2014          | Human (20)                           | Ex vivo       | Impact of intimal tear (location) on dissection                                     |
| . 1 FC001                               | 001-          | D: (0)                               |               | propagation   |
| veger et al. [620]                      | 2015          | P1g (2)                              | Ex vivo       | Impact of branch arteries on false lumen dilatation                                 |

(continued on next page)

# M. Rolf-Pissarczyk et al.

#### Table 5 (continued)

| Report                         | Year    | No. of models  | Model<br>type | Main objective(s)  |
|--------------------------------|---------|--|---------------|--|
| Birjiniuk et al. [624–626,640] | 2015–20 | Silicone (1)/Silicone (2)/Silicone<br>(3)/Silicone (4) | In vitro      | Impact of intimal tear (location) on<br>hemodynamics/flap motion   |
| Veger et al. [621]             | 2017    | Pig (6)  | Ex vivo       | Impact of tear (location) on hemodynamics  |
| Marconi et al. [627]           | 2017    | Transparent rigid resin/silicone<br>elastomer (10)     | In vitro      | Impact of intimal tear (location, size) on<br>hemodynamics   |
| Ahuja et al. [455,613]         | 2018    | Pig (5) & (5)  | Ex vivo       | Impact of balloon catheter in parallel channel on<br>radial pressure; calibration to layer-specific<br>material behavior |
| Canchi et al. [628]            | 2018    | Pig (10)   | Ex vivo       | Impact of intimal tear (location, size) on<br>hemodynamics/flap motion   |
| Moore et al. [590]             | 2019    | PLA (2)  | In vitro      | Development of biomaterials for vascular embolic treatment   |
| Salameh et al. [629]           | 2019    | Plexiglass (8)   | In vitro      | Impact of intimal tear (location, size) on<br>hemodynamics   |
| De Beaufort et al. [160]       | 2019    | Pig (13) & Human (14)                                  | Ex vivo       | Comparison of hemodynamics with <i>in vivo</i> data<br>(4D-flow MRL 2D PC-MRI)   |
| Zadrazil et al. [630]          | 2020    | PFA (4)  | In vitro      | Impact of intimal tear (location, size) on<br>hemodynamics   |
| Bonfanti et al. [635–637]      | 2020-23 | TuskXC2700T (1)  | In vitro      | Impact of intimal tear on hemodynamics   |
| Veger et al. [622]             | 2020    | Pig (3)  | Ex vivo       | Impact of outer wall elasticity on FL dilatation   |
| Veger et al. [623]             | 2021    | Pig (2)  | Ex vivo       | Impact of heart rate on hemodynamics; validation to <i>in vivo</i> data (4D-flow MRI)                                    |
| Chi et al. [608]               | 2022    | Silicone (2)   | In vitro      | Impact of intimal tear (size)/layer adhesion on dissection propagation   |
| Liang et al. [611]             | 2022    | Pig (24)   | Ex vivo       | Impact of intimal tear (location) on<br>hemodynamics/flap motion   |
| Chen et al. [631]              | 2022    | Silicone (2)   | In vitro      | Impact of heart rate/flow rate on hemodynamics<br>with healthy and dissection models                                     |
| Zimmermann et al. [612]        | 2023    | Photopolymer resin coating film (3)                    | In vitro      | Impact of intimal tear (size) on hemodynamics  |
| Aghilinejad et al. [638]       | 2023    | Polyvinyl alcohol coated with latex (4)                | In vitro      | Development of fully automated patient-specific<br>aortic dissection model fabrication                                   |

• A correlation was found between the depth of the intimal tear and the pressure required for dissection propagation. Specifically, a deeper tear required less pressure to initiate dissection propagation. This phenomenon could be attributed to the increased stress in the outer wall of the false lumen, which leads to a greater circumference of the outer wall of the false lumen.

- The flow was characterized by high-velocity flow in the narrowed true lumen during the systolic phase, while the false lumen was dominated by vortical and complex flow structures during both systole and diastole. Moreover, the flow throughput was mainly determined by the total resistance of the tears. Because there was no exit tear, there was almost zero net flow entering the false lumen. Bidirectional flow was regularly observed in multiple tears, particularly during the deceleration phase and early diastole.
- Smaller entry tears led to a stronger flow jet through the tear, which impinged on the outer wall of the false lumen. This could potentially activate specific pathways of SMCs and subsequently lead to wall remodeling.

# 5. Multiscale material models on damage and failure

Mechanical models in continuum mechanics lead to initial and boundary value problems in 2D or 3D. As outlined above, the involved aortic wall is characterized by spanning multiple scales, from macromolecules, to fibers and layers up to the organ level. Given the complexity of the mechanical organization and behavior of the aortic wall, (nonlinear) multiscale models with nonlinear characteristics become a natural and crucial component for providing meaningful descriptions of the mechanics in both healthy and pathological aortic states. For an introduction to the various applied models in this section, readers are referred to the relevant literature on fibrous tissue modeling [83,110,641], damage and failure modeling [642,643], and crack propagation [25,644–646], as these topics are beyond the scope of this review article.

In this section, we review multiscale material models that have been developed to investigate the pathological wall of the aorta from a solid mechanics perspective. By demonstrating the outcomes of different damage and failure processes, these models offer valuable insights into the initiation and propagation process of aortic dissection. They account for several key factors, including the delamination behavior of dissected tissue based on experimental data, the direction of dissection propagation, the degradation of specific constituents, the impact of focal areas of medial degeneration such as the accumulation of GAGs, and the growth and remodeling of the different tissue layers. In conjunction with other computational techniques, most importantly, finite element methods (FEMs) and some of their variants are essential tools to approximate solutions to boundary value problems derived under specific model assumptions. While modeling the dissected aorta at different scales might not immediately affect the clinical outcomes for patients with aortic dissection, it can foster a better understanding of pathological processes. Furthermore, when implemented in patient-specific studies, it may contribute to long-term improvements in clinical treatments.

#### 5.1. Delamination of the wall

While there may be other studies that have applied multiscale material models to aortic delamination, the first substantive discussion on material models in relation to the pathology of aortic dissection was by Rajagopal et al. [11]. They introduced a continuum model that accounted for the orthotropic and viscoelastic behavior of the aortic wall to study aortic dissection delamination. An initiation criterion was proposed based on stress in the delamination area, with the assumption that prolonged exposure to high stress could cause delamination. Although this study did not apply the proposed model, it discussed key hemodynamic and mechanical factors contributing to aortic dissection initiation and propagation. They posited that the two primary hemodynamic factors were mean and maximum systolic aortic pressure, pulse pressure, and cycle frequency, all of which relate to the cyclic, pulsating load on the aorta. The main mechanical factors identified were aortic wall anisotropy, causing high axial and circumferential stresses, significant shear stresses from aortic torsion, and the composite structure of the aortic wall, which is held together by adhesion. Dissection propagation is thus promoted by cyclic loading on the anisotropic aorta. However, they noted that tears between aortic layers can only occur due to microcracks or inherent weaknesses, both acting as local stress concentrations. They also questioned whether maximum systolic ejection  $\max(dp/dt)$ , a putative hemodynamic factor for dissection progression introduced by Prokop et al. [601] and later validated experimentally [43,647], is an appropriate independent hemodynamic measure for dissection propagation. They argued that  $\max(dp/dt)$  may be a helpful predictor but oversimplifies the complex hemodynamics and aortic mechanics. Pressure is positiondependent, and using a global, position-averaged pressure over time is inaccurate, especially since dissection is spatially inhomogeneous. Additionally, no experimental studies have yet demonstrated that dissection propagation is a function of dp/dt, where dp/dt is an independent hemodynamic variable. In fact, dp/dt is more related to pulse pressure and heart rate and influences dissection propagation indirectly.

It appears that the computational model published by Gasser and Holzapfel [648] was one of the first to investigate the delamination process in aortic dissection. To simulate aortic delamination, they used the partition of unity FEM (PUFEM), in which a constitutive relationship, based on a transversely isotropic traction separation law of exponential type with isotropic damage, was employed to characterize the cohesive zone. The resulting model was then applied to examine balloon angioplasty coupled with aortic dissection using a thin aortic slice obtained from MRI. For this analysis, the aortic wall was separated into two layers: the media combined with the fibrous intima, and the adventitia. Assuming a stress-free configuration, residual stresses were not taken into account. Material parameters of the aortic wall were derived from previous experiments [649] using the model by Holzapfel et al. [65], referred to as the Holzapfel-Gasser-Ogden (HGO) model, while the cohesive zone constitutive parameters were based on peeling experiments [433]. Interestingly, the results showed that dissection propagation protected other parts of the aortic wall from damage. Furthermore, they discovered that mode I fracture plays a prominent role in dissections, making it perhaps the most important mechanism for dissection progression.

In a subsequent study, in 2006, Gasser and Holzapfel [650] utilized the computational framework they had developed earlier [648] to conduct a peeling test on aortic tissue. In this study, they chose a simple stress-based initialization criterion as described in Gasser and Holzapfel [651]. The peeling test of a medial strip, which utilized the so-called Gasser-Holzapfel-Ogden (GHO) model [652] incorporating an isotropic ground substance and dispersed collagen fibers, was validated by experiments involving direct tension and peeling tests on aortic tissue [433]. Fig. 44 shows the strong correlation between numerical outcomes and experimental data.

Following a different approach, Ferrara and Pandolfi [653] employed a 3D finite element model to examine dissection progression by performing a peeling test on a medial strip. They used the HGO model for aortic tissue modeling and an anisotropic cohesive model with a linearly decreasing cohesive law, adapted from a previous study [654], to describe crack propagation. This model includes an anisotropic failure criterion and a cohesive law, allowing a distinction between cohesive responses along various directions on the cohesive surface. The failure criterion relies on the normal and shear traction resistance of the surface. The geometry of the medial strip, material parameters, and cohesive model parameters were based on Gasser and Holzapfel [650]. The numerical results demonstrated good agreement with experiments. Dissection in the axial direction exhibited oscillating behavior, while it appeared more regular in the circumferential direction. Moreover, they found that reduced shear strength of the aortic tissue resulted in a smoother response and decreased average pulling force. Then, experimental observations by Tsamis et al. [655] inspired the development of a microstructurebased peeling test. These observations revealed that elastic and collagen fibers form bridges between the layers of the media, which often appear disrupted and broken. Building on this, Pal et al. [656] modeled radially-oriented collagen fibers as bridges between layers that rupture when a specific dissection energy threshold is reached, using a nonlinear exponential force-separation law with a linear drop. The failure energy of each fiber bridge is based on the individual fiber energy, which is represented by the area under the force-separation diagram. A 2D peeling test was conducted with constitutive parameters derived from previous experimental studies [657,658]. When analyzing the computational results, the average peeling tension remained constant and corresponded to the experimentally determined delamination strength. Overall, they achieved strong qualitative agreement with the experimental findings.

Later, Shah et al. [659] proposed a multiscale model to characterize both pre-failure and failure responses of the porcine ascending aortic media. This multiscale model consists of three levels: the finite element domain at the millimeter-scale, the representative volume element domain at the micrometer-scale, and the fiber domain with radii at the nanometer-scale. The representative volume element, defined by an idealized network of discrete fibers with uniform diameter embedded in an isotropic matrix material, employed the GHO model to describe its mechanical behavior. Failure was modeled at the fiber domain, assuming that individual fibers fail when stretched beyond a critical threshold. To consider a wide range of failure lengths, a stochastic failure model was applied. The multiscale model was successfully validated against uniaxial failure experiments in both circumferential and axial directions as well as equibiaxial extension experiments as part of the study. Witzenburg et al. [660] built on and improved the computational model developed by Shah et al. [659], while also conducting peeling and shear tests on porcine tissue. Driven by the initially inaccurate out-of-plane behavior of



**Fig. 44.** (a) Distribution and evolution of radial Cauchy stress during propagation of aortic media dissection driven by displacements on two attached rigid bodies. (b) Photomicrograph of a dissected human aortic media cross-section from a peeling experiment showing distinct cleavage between elastic lamellae and fiber bridging within the cohesive zone – cohesive length approximately 0.5 mm. (c) Comparison of the average experimental load with the calculated load per width for the propagation of human aortic media dissection, showing very good agreement. Figure is adapted from Gasser and Holzapfel [650] with permission from Elsevier.

the model, they incorporated the layered structure of the lamellae and their interlamellar connections. After validating the model against their experimental results and the experiments of Shah et al. [659], the model successfully captured the clear difference between the high tissue strength in the lamellar plane and the low strength of the interlamellar connections.

The delamination of aortic plaque was modeled by Leng et al. [661] using a cohesive zone model. They employed the HGO model for material representation and conducted delamination experiments. The applied traction-separation law, which included damage initiation, evolution, and element removal, featured failure criteria based on the energy release rate. Cohesive model constitutive parameters were fitted to experimental data [410,650], although some origins remained unspecified. The finite element model, comprising medial and adventitial layers, was consistent with experiments on delamination of atherosclerotic plaques in abdominal aortas of mice [410]. Later, Leng et al. [662] utilized the cohesive zone model to simulate aortic delamination, incorporating the HGO model. They developed a detailed aortic strip consisting of intimal, medial, and adventitial layers, and meshed the geometry with hybrid elements to ensure incompressibility. To determine the anisotropy of the aortic wall, mixed-mode and T-shaped peeling tests were performed in axial and circumferential directions. The energy release rate was calculated based on peeling tests on porcine tissue as described in their study and incorporated the experimental results of [661]. The computational approach demonstrated strong agreement with the experimental findings.

In 2016, Thunes et al. [663] then introduced a 3D finite element model of the lamellar unit, a representative volume element for the ascending thoracic aorta, which directly incorporated the collagen fiber microstructure. Several network features, such as orientation distribution, intersection density, and areal concentration, were used to model the fiber architecture. The data were obtained from multiphoton microscopy images taken from human aneurysmal ascending aortic media specimens with a bicuspid aortic valve phenotype. The 1D fibers were embedded directly into a non-fibrous matrix. A bilinear constitutive relation with a tension–compression switch was employed for the collagen fibers, while the constitutive behavior of elastic fibers was modeled using a neo–Hookean model. The model was applied to a uniaxial extension test, with data from aneurysmal tissue obtained from a previous study [664]. The model was generally able to reproduce the typical J-shaped constitutive response. Notably, not all fibers began bearing load simultaneously, leading to regions of higher stress that SMCs could potentially be exposed to, triggering growth and remodeling. The degree of heterogeneity resulted from the local network architecture. Moreover, the stress distribution of the non-collagenous matrix after fiber recruitment was homogeneous at low stretches and highly heterogeneous at higher stretches. Later, Thunes et al. [665] extended this model to assess the role of the implemented structural features in determining in-plane tissue strength, which governs dissection initiation, i.e. the formation of an intimal tear. A damage parameter was introduced to monotonically decrease the stress upon reaching a peak stretch, thereby enabling the modeling of localized collagen fiber breakage. The results also indicated that the colocalized elastic lamellar stress exceeded the elastic strength.

Conversely, Nobel et al. [666] investigated dissection propagation using a rigid, planar wedge penetrating the delamination zone to study catheter-induced dissection in coronary arteries. They employed an incompressible Ogden model [667] with a traction-separation law and determined the critical release rate associated with tissue dissection from experimental studies [433,668,669].



**Fig. 45.** (a) Finite element model of two medial strips connected by discrete interlamellar collagen fibers and removed upon failure. (b) Engineering strain in fibers during dissection – most stretched fibers at the separation front. (c, d) Peeling force versus displacement in the finite element simulation reveals avalanche-like aortic failure during delamination ('MC': model outcome for circumferential direction; 'ML': model outcome for axial direction), where  $\beta$  is defined as the negative slope of a straight line fit. Figure is reprinted from Yu et al. [671] (licensed under CC-BY-NC 4.0).

Compared with the wedge-driven dissection experiments on porcine aortas [669], the finite element analysis demonstrated qualitative agreement with experimental data, although with less variation and size between samples.

Brunet et al. [461] proposed an analytical approach to replicate observations from X-ray imaging during uniaxial extension tests [670]. A 1D model was created, consisting of multiple layers representing the media of the aortic wall, each possessing distinct elastic and damage properties. The material behavior was modeled adopting a Mooney-Rivlin model [667]. Additionally, a stress criterion in the normal direction of the crack was defined with a linear damage evolution. However, this approach could only predict mode I failure. The results were subsequently validated against a 2D finite element model using a cohesive zone model, which showed good quantitative agreement. The model demonstrated that a crack initially forms due to mode I failure, and then propagated in mode I in the transverse direction. When the elastic recoil stress surpasses the mode II strength of the axial plane, the crack propagated in the direction requiring less energy, leading to the formation of a delamination plane. Therefore, mode II failure might play a crucial role in keeping layers together (due to earlier rupture in a parallel delamination plane) and has a significant impact on crack propagation.

Inspired by experimental findings, Yu et al. [671] conducted a peeling test involving two lamellae to demonstrate that the delamination process exhibits an avalanche behavior, described by a power law, as shown in Fig. 45. To achieve this, they modeled two lamellae, characterized by an isotropic neo–Hookean model, interconnected by discrete and randomly oriented collagen fibers. These fibers were represented by Timoshenko beam elements and a specific stress–strain relationship [672]. The connecting fibers were removed upon reaching an experimentally specified failure strain [673]. The random structural arrangement of interlamellar fibers, interpreted as the source of energy-releasing avalanches on a wide scale, was based on multiphoton imaging presented in their study. These multiphoton images revealed that the two elastic lamellar layers were connected by both elastic and collagen fibers, which were sparsely arranged and non-uniformly distributed in density and orientation. Additionally, peeling tests were performed in circumferential and axial directions using aortic media samples to validate the computational model. These experiments showed a sharp initial increase in delamination force before reaching a plateau region, followed by a sudden drop. The probability density distribution of these force drops suggested a power law distribution across nearly two orders. A comparison between the experiments and the computational model led to satisfying results. Furthermore, the finite element analysis showed that the more interlamellar fibers engaged before failure, the higher the failure strain. Consequently, the probability of higher force drops increases, resulting in larger avalanches and a smaller power law exponent.

Building on previous research, Wang et al. [674] explored the contribution of elastin and collagen fibers to intralamellar bonding through mechanical testing, multiphoton imaging, and finite element modeling. Mechanical testing involved peeling tests and biaxial extension tests in both circumferential and axial directions with a porcine aortic media and a purified elastin network, respectively. A comparison between the media and the purified elastin network revealed that the peeling force and energy release rate associated with mode I failure were significantly higher for the media. It should be noted that the peeling force oscillated considerably due to discrete microstructural events. Furthermore, axial peeling consistently exhibited higher energy release rates and strength. Multiphoton

imaging revealed the recruitment of both elastin and collagen fibers, while also providing evidence that in-plane anisotropy of fiber directions could potentially explain the direction-dependent phenomena observed in peeling tests. Experimental results informed the development of a 3D finite element model of a peeling test, which included two initially separated tongues at the top and one layer of cohesive elements. The cohesive zone model featured a bilinear traction-separation law and focused on mode I failure due to the chosen testing methods. The constitutive model of the aortic media was represented by the GHO model. Biaxial extension test simulations using shell elements with an inherent plane stress assumption were also performed to determine constitutive parameters. In contrast, the constitutive stress versus stretch relation of the purified elastin network was obtained from a previous study [675]. The findings demonstrated that both elastic and collagen fibers play a crucial role in interlamellar bonding, strength, and toughness. They also emphasized that the contribution of elastin fibers to interlamellar bonding should not be overlooked, as pre-stretched elastin exerts an intrinsic compressive stress on collagen fibers, resulting in their waviness. A loss of interlamellar elastin also leads to stress concentration around the remaining fibers.

In summary, numerous multiscale models have been developed over the past decade using various approaches. These models serve as valuable tools for examining the failure behavior of the aortic wall during delamination and revealed, for example, a power law behavior in dissection propagation. However, it remains uncertain to what extent this behavior is related to the false lumen propagation in aortic dissection. As it has already been demonstrated that delamination strength decreases with age [676], it is likely that the pathologically altered aortic wall significantly influences delamination behavior is crucial to enhance our understanding of the delamination process in aortic dissection. Additionally, some of the models discussed here have the potential to be incorporated into a more comprehensive patient-specific, organ-scale analysis to model dissection propagation, which will be addressed later in this review article, see Section 9.2.

## 5.2. Dissection propagation and flap configuration

Modeling dissection propagation in a patient-specific context remains numerically challenging and seemingly infeasible. Consequently, recent developments in constitutive modeling have also focused on creating simplified or idealized geometries to study dissection propagation. These models effectively investigate dissection propagation and its direction by incorporating the healthy or pathological wall microstructure. Moreover, they allow for a detailed examination of the flap configuration throughout the cardiac cycle and possible changes during dissection propagation.

Wang et al. [675] conducted a study on tear propagation in soft tissues, focusing on the assessment of the energy release rate through an effective numerical method that employs an adaptive tear length. They utilized the HGO model to describe the uniform, fiber-reinforced material. To explore the impact of fiber orientation on dissection progression, they implemented a 2D strip featuring a single axial tear under internal pressure. The surrounding tissue was represented using a linear elastic model. The constitutive parameters of the HGO model were derived from studies on rabbit carotid arteries [65]. The findings indicated that fiber orientation, tear length, and surrounding tissue could influence the critical pressure for dissection. Notably, as fibers became more parallel to the tear, the energy release rate decreased due to the fibers bearing a greater load to resist tear opening. In contrast, the energy release rate increased consistently with tear length. The stiffness of the surrounding tissue was found to be a key factor in the propensity for tear propagation, with lower stiffness promoting dissection progression.

In the subsequent study, Wang et al. [677] expanded on their earlier approach by employing the extended FEM (XFEM) to simulate dissection propagation in a 2D cylindrical tube with a predefined tear. To ensure incompressibility, they utilized plane-strain hybrid elements to discretize the 2D geometry. In addition to applying a linear cohesive traction-separation law, residual stresses were incorporated into the model. Residual stresses in arteries can be determined experimentally using so-called opening angle experiments. In such experiments, a ring-shaped section of the artery is cut, causing it to spring open. The angle at which the ring opens, the so-called opening angle, directly measures the circumferential residual stress, which can then be taken into account in a constitutive framework [678]. Consistent with their previous research, the constitutive parameters were obtained from the literature [65]. By altering the tear length, they discovered that a longer tear is more prone to propagation, with the shortest and longest tears being the most stable. They also observed that all tears tended to expand radially outward (Fig. 46(b)). The dissection flap was found to buckle at certain tear lengths (Fig. 46(a)), which was subsequently confirmed by CT imaging of aortic dissection patients. Furthermore, residual stresses appeared to offer protection against tear propagation, as shown in Fig. 46(c). In summary, the study concluded that tear propagation is governed by an inherent interplay of tear length, wall buckling, fiber orientation, and residual stresses.

In their third study, Wang et al. [679] further applied the computational model introduced in their 2017 study to both stable displacement-driven peeling and unstable pressure-driven tear propagation. They demonstrated that the thickness of the flap might play a crucial role in predicting dissection evolution. Several key observations were made. First, for a long and shallow dissection flap, the flap would collapse, potentially slowing down tear propagation. Second, the critical propagation pressure was higher for a deeper and shorter tear. Third, a deep tear without buckling tended to propagate at a lower critical pressure, implying that the buckled state of the dissection flap redistributes energy and stresses within the wall, potentially hindering tear propagation. Finally, tear propagation typically followed the direction of the material axis with the highest stiffness, which corresponds to the fiber orientation. These findings shed light on the complex interplay of factors that influence the progression of dissection in soft tissues.

Building on the framework established by Raina and Miehe [680], Gültekin et al. [681] employed a phase-field approach to model damage and failure initiation in biological tissues. They adapted the Ginzburg–Landau type phase-field method, originally designed for brittle fracture in isotropic solids, to model the anisotropic aortic wall. The study incorporated the HGO model to represent the anisotropy arising from the collagen fibers in the aortic wall. To simulate the fracture of arterial walls, they identified and implemented



**Fig. 46.** (a) Deformed configuration displays the maximum principal stress with varying tear lengths subjected to the critical pressure. Simulation results indicate that the dissection flap tends to buckle beyond a specific propagation length. (b) Steady deformed configuration with different tear length (between 30° and 210°), where all the tears propagate radially. (c) Dimensionless critical pressure  $p'_c$  versus the opening angle  $\alpha$ , for four different tear lengths, where the letter 'B' indicates that the inner wall is buckled. Figure is adapted from Wang et al. [677] (licensed under CC-BY 4.0).

an energy-based anisotropic failure criterion. The model was then applied to experimental data from uniaxial extension and simple shear failure tests [454]. This analysis demonstrated that the finite element simulations aligned with the experimental findings, validating the effectiveness of the adapted phase-field approach for modeling anisotropic biological tissues. In two consecutive studies, Gültekin and Holzapfel [25] first conducted a review of computational models for rupture in soft biological tissues. They then introduced a ratedependent phase-field evolution and compared various failure criteria, including energy-based, Tsai-Hu, principal stress, and stressbased Hill criteria, see Gültekin et al. [682]. They found that the energy-based criteria resulted in more stable and physically significant crack growth. However, it is important to note that due to numerical issues in this study, they did not use experimental data to calibrate their constitutive parameters. In the latest research, Gültekin et al. [683] employed the phase-field model to study aortic dissection. They used a symmetric cylinder with a predefined tear, subjecting it to axial stretch, end-systolic twisting, and internal pressure. The aortic wall was represented using multiple layers: healthy media, degenerated media, and adventitia, which were based on experimental data [454]. It is important to note that the material parameters for the degenerated media were simply reduced by 20% compared with the healthy media (Fig. 47(a)). To simulate physiological and supra-physiological loading conditions, they adjusted the end-systolic twisting and internal pressure (Fig. 47(b)). In fact, their numerical analysis revealed a helical crack pattern in the damage zone surrounding the tear, which aligned with the fiber orientation, see Fig. 47(c). Additionally, they identified potential damage and fracture mechanisms that could occur in the aorta under mode I, mode II, and mixed-mode fracture conditions, such as collagen fiber pullout, collagen fiber bridging, collagen fiber-matrix debonding, and matrix cracking. Upon examining the findings, they concluded that dissection progression may be more likely to occur due to mode II fractures rather than mode I.

Inspired by earlier experiments, Ahuja et al. [613] validated a computational model of flap reapposition by comparing it with *in vitro* experiments using porcine aortas. For details, see Section 4.3 and their previous study [455]. These experiments ascertained radial pressures when a balloon catheter was inserted into a parallel channel model under a static inflation pressure. This experimental analysis was subsequently mirrored in a finite element model. They applied an anisotropic model, namely the HGO model, and assigned different material properties to the middle and distal flap segments, where the parameters were calibrated to biaxial extension tests [455]. The finite element models successfully represented the variations in radial pressures necessary for the reapposition of the dissection flap in correlation with pressure.

Mousavi et al. [684] introduced a layer-specific damage model based on constrained mixture theory to estimate the risk of tear formation by considering the internal stresses within the aortic wall. In this model, they utilized a damage evolution approach with a linear softening law derived from continuum damage theory. The mechanical behavior of collagen and elastic fibers was integrated using the HGO model, with distinct material parameters for tension and compression. The constitutive parameters of the material model were adjusted to fit experimental data [104,105] before it was applied to three representative examples to predict damage evolution. These examples included an inflation test of a perfectly cylindrical artery, a uniaxial extension test of a human ascending aortic aneurysm strip, and a bulge inflation test. The computational results demonstrated that the onset of damage begins at the intimal layer and extends through the media, but does not penetrate into the adventitia.
M. Rolf-Pissarczyk et al.



**Fig. 47.** (a) Idealized geometry of an extracted 3D segment, consisting of four associated medial sub-layers and two represented adventitial sub-layers. Note that media 3 refers to a degenerated layer (with lower strength), where the sliced view of the geometry represents the prescribed representative initial tear size. All dimensions are in millimeters. (b) Physiological and supra-physiological loading conditions were applied to the idealized geometry, encompassing end-systolic twist, internal pressure, and constant pressure. (c) 3D evolution of the crack phase-field at the three load points A, B, and C, showing the influence of the applied loads. This progression reveals a helical crack pattern in the degraded media, distinguished by tear sizes: small (i) and large (ii). Isosurfaces are employed to illustrate the extent of the damage zone. Figure is adapted from Gültekin et al. [683] (licensed under CC-BY 4.0).

Rolf-Pissarczyk et al. [685] later introduced a constitutive model that accounts for the pathological degeneration of inter-lamellar elastic fibers during aortic dissection. These fibers, aligned radially, contribute to the cohesion of lamellar units in the aortic media. The model incorporates a degradation parameter to exclude damaged or degraded elastic fibers from the strain-energy function by modifying the specific strain-energy term of the dispersed elastic fibers. For this purpose, they applied and extended a discrete fiber dispersion model, previously introduced for modeling collagen fiber dispersion [686]. They presented two computational models. First, the authors fitted the constitutive model of elastic fibers to experimental data from a single elastic lamella [687] and validated the model using two representative numerical examples: uniaxial extension and simple shear. Second, they developed an aortic dissection geometry with two distinct tissue layers, motivated by patient data [14,278], to investigate the effect of degraded radiallydirected elastic fibers on stress distribution in aortic dissection. They adjusted the mechanical and structural parameters to match experimental data from their laboratory, existing literature [687-689], and thickness measurements [690]. The findings revealed that the stiffness contribution of elastic fibers might have limited impact on stress distribution at higher pressures. Additionally, they proposed a possible mechanism for dissection propagation: medial delamination caused by the stretch applied to the dissection flap. Specifically, the flap is stretched due to the cyclic dilatation of the true and false lumina, see Fig. 48, leading to its separation from the intact wall. In subsequent studies, Rolf-Pissarczyk et al. [450] utilized the discrete fiber dispersion method in a layer-specific model of the aortic wall, which accounted for passive constituents and the active contribution of SMCs, though only the mechanical behavior was modeled. Both fiber constituents and SMCs were represented using the discrete fiber dispersion method, enabling the modeling of pathological phenomena associated with aortic dissection such as lamellar loss of SMCs in dissected tissue.

The geometry of an intimal tear has a significant influence on the hemodynamics of the aortic dissection and thus on the dissection propagation. To quantify this effect on aortic dissection initiation and propagation, Brunet et al. [691] developed a model that compares various tear orientations to identify primary factors contributing to aortic dissection. This model featured a two-layered aortic wall, consisting of medial and adventitial layers, and employed the GHO model to represent collagen fibers. Residual stresses were incorporated using a uniform opening angle, while dissection propagation was modeled using an XFEM approach with a linear traction-separation law. Here, the stress-based anisotropic Hashin failure criterion was employed. Material parameters were calibrated using experimental data, including tensile extension tests from ascending aortas [692], radial and shear strength measurements from aneurysmal and dissected human thoracic aortas [433], and experimental data on the fracture energy [433]. The model was then applied to a tension-inflation test of an idealized aortic segment. Three distinct tear orientations were evaluated and a numerical design of experiments was employed to identify the most sensitive parameters affecting the aortic pressure required for intimal tear propagation. Seven key parameters were considered: crack length, depth, width, and position; tensile and shear strength; opening angle; and initial axial stretch. The tensile and shear strength direction components were varied equally. The outcomes of the analysis included the critical pressure and the plane of propagation. The average critical pressure ranged between 206 and 251 mmHg. While shear strength did not demonstrate a significant influence, tensile strength was identified as the most influential factor due to the lack



**Fig. 48.** Computational results of the finite element analysis illustrate the maximal principal stresses (in kPa) in the true and false lumina at (a) diastole and (b) systole compared with evaluated CT images at the end of the aortic arch and show the minimum and maximum dimensions of the true and false lumina over the cardiac cycle. Figure is reprinted from Rolf-Pissarczyk et al. [685] (licensed under CC-BY 4.0).

of consideration of the lamellar structure of the media. Additionally, the length and depth of the crack played a crucial role in the dissection propagation. For the aortic dissection to propagate closer to the adventitia, lower pressure was required, and the initial axial stretch contributed to a reduction in the critical pressure. In particular, the residual stress appeared to provide a protection against the dissection propagation in the aorta.

Early needle-based experiments [592,593], followed by later studies of Roach and his colleagues [595,596], see Section 4.2, uncovered varying tear propensities to aortic dissection along the aorta, which can be explained by the inherent microstructure. In brief, the thoracic aorta displays thin elastic fibers that are oriented radially, whereas the abdominal aorta exhibits radial interlamellar connections resembling a honeycomb structure. As a result of this microstructural disparity, the abdominal aorta might be more prone to tearing, while the thoracic aorta is more prone to dissection propagation. Ban et al. [693] incorporated these failure characteristics into a phase-field model of the medial layer. Two models were developed: a homogenized model exhibiting nonlinear, anisotropic material behavior for the thoracic aorta, and a microstructure-based model for the abdominal aorta that explicitly included interlamellar struts. They utilized the neo-Hookean model for elastin and the HGO model with four fiber families to represent the passive behavior of SMCs and collagen fibers. Separate symmetric boundary value problems were established for the thoracic and abdominal aortas, each incorporating a damage zone for needle insertion to inject fluids. They also explored a second scenario that examined the behavior of multiple intramural mucoid fluid pools. The constitutive parameters for both models were derived from the literature [694,695]. Additionally, they conducted multiphoton imaging on mouse aortas to identify the elastin structure in the descending and abdominal aortas. As shown in Fig. 49(a), the descending aorta displayed large circular fenestrations, which were less common in the suprarenal abdominal aorta. Instead, the abdominal aorta exhibited numerous elongated and relatively thick radial elastin structures. These struts traversed the intralamellar space between SMCs, connecting adjacent elastic lamellae, and were absent in the descending aorta. The multiphoton imaging results informed their subsequent computational analyses. The key computational findings are as follows. First, in a single intramural injection into the thoracic aorta, increased injection volume resulted in two deformation regimes: wall elevation above the pool, followed by tearing that began at a critical pressure. The tear expanded only in-plane, initially in the axial direction and later in both axial and circumferential directions, which is consistent with experimental observations. Notably, they discovered that the square of tearing pressure was directly proportional to tissue stiffness and critical tearing energy and inversely proportional to the square root of the torn area. Second, during a single intramural injection into the abdominal aorta, the elastic lamellae stretched and the radial struts initially bent without tearing. However, at a critical pressure, a strut tore, causing a sudden pressure drop and subsequent pressure build-up. The pressure-volume curve displayed S-shaped steps, analogous to experimental findings. In addition, they conducted simulations comparing a weaker and stronger intraluminal media without struts. Finally, they modeled a case with multiple fluid injections. As the pools grew closer together, they began to coalesce after an initial period of growth, with tears occurring at lower pressures.

Motivated by Ban et al. [693], who demonstrated that the spatial distribution of structurally significant interlamellar struts can affect the likelihood of dissection, Yin et al. [697] developed a data-driven surrogate model to predict the delamination process for various strut distributions. This model utilized an operator-regression neural network [698] to enhance predictive accuracy. The neural network successfully predicted the pressure–volume curves of injected fluids and the damage progression field of the wall across



**Fig. 49.** (a) Mechanism for the progressive, step-wise tearing of the abdominal aortic wall as observed in needle experiments by Roach and his colleagues: (i) snapshot of the microstructural model depicting the progressive tearing of the abdominal aorta; (ii) graph showing the injection pressure versus the injection volume normalized by the initial pool volume. Three cases are considered: a microstructural model, a model without struts and a weak, less stiff intralamellar material, and a model without struts but where the intralamellar material is represented by a tangential stiffness and  $G_c$  equal to those of the surrounding medium. The peaks marked I, II, and III correspond to the pressure build-up and eventual tearing of the radial struts marked with the same numbers in (iii) and (iv). (b) Graph showing injection pressure–volume curves and the morphology of the injected volume, which are qualitatively and quantitatively consistent with the mentioned needle experiments. Figure (a) is adapted from Ban et al. [693], while (b) is from Ban et al. [696], with permission from Springer Nature, respectively.

diverse strut distributions. The results suggested that this composite branch-trunk neural network effectively captured the underlying functional relationship between unique microstructures and their mechanical properties. Furthermore, the incorporation of the current damage field allowed accurate characterization of characteristic pressure drops. To generate *in silico* data, a total of 2100 phase-field solutions were created, with 1900 used for training and 200 for testing. According to them, this was the first attempt to predict dissection progression and mechanical behavior in a heterogeneous aortic wall using scientific machine learning.

In a later study, Ban et al. [696] applied the previously developed phase-field model to simulate intra-lamellar tearing induced by intramural ink injections into the medial layer. They implemented two models: a phase-field model and a simplified model. For the constitutive model, they employed a Fung-type model with four collagen fiber families, using material parameters from the literature for both healthy human descending thoracic aorta [699] and diseased aorta [683]. Two main findings were reported. First, a powerlaw relationship was observed between pressure and key geometric and mechanical factors such as surface area, wall stiffness, and tearing energy. Second, it was found that focal regions of weakening or stiffening can influence the direction of tear propagation, such that the tear moves toward either the lumen or adventitia. To validate their findings, they replicated experiments from Roach et al. [600] with two needle sizes on 18 descending thoracic aortas of similar length ( $56 \pm 3$  mm) harvested from pigs. The results were consistent both qualitatively and quantitatively with those of Roach et al. [600], as shown in Fig. 49(b). Specifically, they discovered that the pressure at which tearing initiates depends nonlinearly on the pressure in the true lumen, wall stiffness, critical tearing energy, tear area, and axial stretch. A power-law relation emerged for the pressure at which tear propagation begins, with respect to wall stiffness, critical tearing energy, and the inverse of the initial area. Additionally, axial stretch often diminishes in vessels prone to dissection. At tear nucleation sites, normal static blood pressure is insufficient to separate layers and advance the tear, but the Donnan swelling pressure might be adequate [700]. As volume increases, the influence of Donnan swelling pressure diminishes and blood pressure becomes more significant, enabling tear propagation. This is due to the inverse relationship between pressure of tear resistance and increase in torn area. Finally, the study found that tear propagation follows oblique weak regions, meaning that localized areas of weakness or strength can affect the propagation direction. This can cause a tear to shift toward the lumen or adventitia, resulting in a fatal rupture or additional fenestration. A strong oblique inhomogeneity could also deflect the tear propagation path.

In 2022, FitzGibbon et al. [701] analyzed the mixed-mode behavior of exponentially softening path-dependent cohesive zone formulations. A new non-potential path-dependent cohesive zone model was used to simulate aortic dissection in two examples. The artery was modeled using a hyperelastic, anisotropic bilinear fiber model [702], with cohesive zone material parameters obtained from previous experimental-computational studies of mode II dissection [703]. In the first example, the risk of tear propagation under luminal pressure was investigated. To this end, dissection propagation was simulated in an artery with a pre-existing radial notch tear. A cohesive zone was prescribed in the circumferential plane around the tear, and a supra-physiological load of 500 mmHg and 10% axial stretch were applied. The crack initiated at 275 mmHg and continued to propagate, with final crack growth reaching 5.75 mm at a pressure of 500 mmHg; no crack propagation occurred below 275 mmHg. Initiation and propagation occurred in pure mode II due to

compressive tractions at the medial interface caused by hypertensive blood pressure. A 50% reduction in interface strength led to tear propagation under typical hypertensive blood pressure at 190 mmHg. In the second example, the risk of further false lumen propagation in an artery with a pre-existing patent false lumen was investigated by coupling hemodynamics with solid mechanics. Contrary to other studies, the recent study suggested that the eventual formation of a false lumen, following mode II initiation, will result in mixed-mode conditions at the crack tip.

In their research, Li et al. [704] developed an idealized model of an artery, described as a single-layered, thick-walled axisymmetric tube with an axisymmetric tear. They accounted for residual stresses by employing the opening angle method. Both the lumen of the artery and the interior of the dissection were assumed to be under uniform, constant pressure, simulating conditions analogous to blood pressure. The HGO model was utilized to describe the material behavior. The progression of the tear was interpreted as an incremental deformation within the tube. The study found that changes such as a decrease in the fiber angle, a decrease in axial prestretch, and an increase in the opening angle contributed to the dilatation of the dissection. Furthermore, a gradual increase in pressure in the lumen and the dissection was identified as a factor leading to the dilatation of the dissection.

By applying a combination of experimental procedures, constitutive modeling, and computer simulations, Zhang et al. [705] examined the 3D residual stress field of the aortic wall. The authors performed biaxial tests and opening angle experiments [706] on healthy porcine aortic samples. A key facet of their research was the development of a stress-driven tissue growth model [707], which was built upon the decomposition of the total deformation gradient into two distinct parts: the elastic deformation and the volumetric growth. This model was developed with a particular focus on accounting for the stress deviation between the current and target stress in the *in vivo* stress distribution. After verifying their model, they pursued further validation of their methodology and examined the buckling behavior of the dissection flap using experimental and computational approaches. For this purpose, circular tears of varying sizes were manually induced on three ring-shaped samples of the same porcine aortas. They observed an intriguing inward buckling of the dissection flap. Their numerical simulations were successful in recreating the buckling behavior of the dissection flap of the existes. This study was pioneering in its inclusion of residual stresses in relation to aortic dissection, and it presented a correlation between the buckling phenomenon and the configuration of the flap. However, the extent of these residual stresses remains difficult to determine accurately without considering *in vivo* boundary conditions. As discussed earlier, the configuration of the dissection flap might be related to the pressure difference between the true and false lumina. Further investigations are necessary to ascertain the degree to which residual stresses influence the configuration, particularly as these stresses could potentially be modified in pathological tissue.

Subsequently, Han et al. [708] investigated tear propagation in aortic dissections using a model of an idealized cylindrical, tubeshaped, bilayer thick aortic wall. Their study focused on the influences of geometrical parameters, loading conditions, and residual stresses on tear propagation. They employed a cohesive zone method with a bilinear traction-separation law to model in-plane dissection propagation between concentrically distributed elastic lamellae in the aorta. The study used the incompressible HGO model to describe the material behavior of the aortic wall, which was modeled as three concentric layers and calibrated using experimental data [707]. Additionally, the model was calibrated for three modes of interfacial damage in the media, based on experimental findings [433,454]. A critical aspect of this model is the use of cohesive elements, with the force-displacement relationship characterized as a linear function of relative separation, and a power law defining the failure criteria. Two primary computational models were used in this study. The first model focused on calibration, incorporating delamination and shear tests to identify material parameters. In the second model, an idealized cylindrical, tube-shaped, bilayer thick wall of the aorta was subjected to axial stretch and inner pressure. This model featured a cohesive zone in the media and an initial tear characterized by its angle, axial length, and depth ratio. They investigated the influence of the initial tear, axial stretch, and residual stress on the critical pressure for in-plane dissection propagation. The outcomes of the model demonstrated that deeper and larger cracks tend to correspond to a lower critical pressure. Key discoveries from the study include, inter alia: (i) greater axial stretch results in a reduction in critical pressure, (ii) higher residual stress is accompanied by an increase in critical pressure, (iii) variations in pressure between the true and false lumina significantly affect the critical pressure, and (vi) the altered geometry of the tear influences the critical pressure.

Gheysen et al. [709] conducted an uncertainty quantification study on the wall thickness and stiffness of an idealized dissected aorta. They developed a model that assumed a cylindrical geometry with two layers, the media and the adventitia, and two tears, using length and diameter values from the literature of the normal descending aorta [710–712]. The circumferential false lumen size was determined from pre-operative CT scans of a patient-specific case with type B dissection [713], with the healthy pre-dissection diameter estimated based on the curvature of the true lumen wall. Similar to their previous study [714], they employed the GOH model with parameters fitted to uniaxial and biaxial experimental data of the descending aorta [240,715]. They ensured physiological behavior by coupling the parameters of the media and adventitia to pulse wave velocity and applying an axial stretch with a prestressing algorithm [716]. Uncertainties associated with unknown total wall thickness, medial thickness, flap thickness, and pulse wave velocity were analyzed. The resulting 4D input space was sampled using Latin hypercube sampling, assuming a uniform probability distribution for the four independent input parameters. A total of 300 simulations were performed and a surrogate model was developed based on these simulations. Due to the limited output of 300 simulations, a Gaussian process regression model was trained to account for uncertainty in the surrogate model itself rather than using a regular neural network. Based on the output distribution of 10,000 surrogate model simulations, different displacement measures and the maximum principal Cauchy stress at a specific flap position were evaluated. The parameter-specific uncertainty analysis indicated that the modeled material stiffness significantly influences the deformation of the dissected wall, while the relative thickness of the dissected flap is the most critical factor affecting the wall stress. The large ranges of uncertainty and the varying impact of often unknown wall stiffness and thickness parameters highlight the need for caution when interpreting the outcomes of dissected wall models.

In conclusion, the cohesive zone model, along with the phase-field approach and the XFEM, has been widely used to investigate

dissection propagation. Key findings involve the detection of helically-shaped propagation patterns, with the tear generally not extending through the adventitial layer. Consequently, the findings suggested that the propagation remains predominantly in-plane. Moreover, the studies indicated that factors such as dissection flap stretching due to luminal expansion or motion can significantly influence delamination, with flap stiffness playing a crucial role in flap deformation. Additionally, flap thickness is identified as the most critical factor affecting wall stress, which may also impacts delamination. Given the challenges in accurately measuring flap thickness and its variable stiffness, caution is needed when interpreting the outcomes. Moreover, the size and depth of the intimal tear have a significant influence on its propagation. Observations showed that tear propagation tends to follow oblique regions of wall weakness, suggesting that localized areas of strength or vulnerability may influence the propagation direction. This underscores the need to consider inhomogeneities within the aortic wall when modeling, which will be discussed in the subsequent section.

# 5.3. Focal areas of medial degeneration

Modeling the focal areas of medial degeneration or material inhomogeneities in the pathological wall is essential, as recently emphasized [717]. Yet it is frequently overlooked. To date, research has primarily focused on modeling the accumulation of intraluminal GAGs and their effects on local stress distribution, among a few other studies.

The accumulation of GAGs in the aortic wall can result in considerable stress concentrations and an intra-lamellar pressure known as Donnan swelling pressure. This hypothesis, proposed by Humphrey et al. [103], served as the motivation for the computational study conducted by Roccabianca et al. [700]. For demonstration purposes, they modeled a rectangular section of the aortic media containing a central GAG inclusion and applied various stress ratios in the axial, circumferential, and radial directions. Notably, the radial direction consistently experienced compressive stress. The aortic media was modeled as a homogenized material, utilizing the neo–Hookean model to represent the matrix material and the HGO model with four fiber families to account for axial and circumferential alignment of collagen fibers. The GAG inclusion was represented using a modified neo–Hookean model, which exhibited minimal tensile stiffness, and incorporated osmotic loading of both the media and the inclusion. The osmotic loading in the inclusion was assumed to be significantly higher. To ensure accurate computational outcomes, the constitutive parameters were fitted to experimental data [699]. The finite element analysis revealed that both circumferential and axial stresses around the periphery of the inclusions were locally increased. Furthermore, radial stress exhibited a sharp increase directly in front of the tip of the inclusion along its major axis. The study thereby confirmed that spatial heterogeneities could contribute to the propagation of aortic dissection.

In a subsequent study, Roccabianca et al. [79] employed two distinct models to address a similar boundary value issue: a mechanistic, mixture-based finite element model and a phenomenological, continuum-based semi-analytical model. In this investigation, they represented the aortic wall as a two-layered cylinder with specified osmotic pressures for each layer, demonstrating that typical GAG accumulations within the aortic media might be equally crucial in both supporting and detecting mechanical loads. Analyzing the computational results, they proposed that a pathological increase in these GAG aggregates could potentially cause excessive pressure in the intralamellar units, which could lead to tension in the radially-oriented elastic fibers that connect SMCs to the elastic laminae. This tension could then trigger mechanosensing in SMCs.

Based on the same hypothesis but using a different approach, Rausch et al. [718] integrated the isotropic continuum damage theory with the smooth particle hydrodynamics (SPH) approach to simulate damage and failure in soft tissues. In short, unlike in mesh-based methods such as the FEM, SPH is a meshfree Lagrangian computational method used to simulate continuum media such as solid mechanics and fluid flows. It solves differential equations of motion using particles, smoothing quantities of interest. SPH can handle large deformations without remeshing and, as shown in this study, can account for macroscopic damage and gross failure. However, it is computationally expensive and defining boundary conditions is often a challenge. To model failure, the developed algorithm automatically identified damaged particles and separated the spatial domain. Constitutive relations for the aortic wall were chosen according to the HGO model, as it accounted for the orientation of collagen fibers. The computational findings of the SPH approach were then compared with predictions from uniaxial and biaxial extension tests. To conclude, the SPH approach was effectively validated against experimental data from a peeling test and failure under clamped uniaxial extension [433,668].

Building on the previously developed SPH approach, Ahmadzadeh et al. [719] later investigated the potential role of accumulated GAGs in initiating and propagating intra-lamellar delamination. To accomplish this, they expanded the SPH approach to model growth, coalescence, and propagation of GAGs. Additionally, they introduced the active contribution of SMCs, modeled using a purely mechanistic approach, and incorporated osmotic swelling pressure into the model. Consistent with previous studies [79,700], a modified neo–Hookean model was employed to capture the ability of GAGs to absorb compressive stresses. To assess the stress distribution surrounding GAG pools, an idealized 2D cylindrical aorta was created, and four different GAG pool sizes were compared. For the constitutive parameters, experimental data from murine aortas were used [720,721]. The computational findings revealed substantial intramural stress concentrations and a shift from typically compressive to tensile stress in the radial direction near the tip of pooled GAGs, which aligns with earlier studies [79,700]. Furthermore, the activation of SMCs in the media can partially shield the wall from swelling-induced damage and reduce the circumferential stress in the adventitia.

Ahmadzadeh et al. [722] subsequently refined their approach by developing a multilayered model, as shown in Fig. 50(a), encompassing both the media and the adventitia, to investigate the stress field resulting from the disruption of individual elastic lamellae, loss of SMC contractility, and GAG production within an intra-lamellar space. The material response of collagen fibers was incorporated using four families of collagen fibers, while elastic fibers were explicitly modeled with an isotropic neo–Hookean model. The boundary value problem comprised approximately six nearly concentric elastic lamellae and five distinct layers of SMCs subject to pre-stretch. The constitutive parameters of the model were determined based on literature data [720–723]. Through computational analysis, they confirmed that GAGs could elevate radial stresses beyond a certain threshold, potentially leading to intra-lamellar



**Fig. 50.** (a) Analysis using an SPH aortic model illustrating stress field variations based on different configurations: (i) a single pool of apoptotic cells; (ii) two separate pools; and (iii) two interconnected pools of apoptotic cells replaced by GAGs. The combined stress impact on elastic lamellae and intra-lamellar cells is most pronounced near the merged pools, suggesting an elevated risk of lamellar rupture and SMC impairment in this scenario. It is noteworthy that the symbol ( $\geq$ ) illustrates how the stress concentrations around a single pool and two separate pools are comparatively close to each other at lower pressures (102mmHg), but the separate pools induce greater stress in their vicinity at elevated pressures (184mmHg). (b) (i) Visualization of the deformation in both the global model and its submodel (in  $\mu$ m) and (ii) observed results after the swelling phase with maximum principal Cauchy stresses and those in radial, circumferential, and axial directions (in kPa). Noticeable stress concentrations can be seen at the boundaries between the different materials. Consequently, the strust of the elastic fibers limit the radial expansion of the GAGs/PGs. Conversely, in the circumferential and axial directions, the primary resistance to swelling arises from the adjacent, relatively softer interlamellar material, which includes both the GAGs/PGs and the SMCs. Figure (a) is adapted from Ahmadzadeh et al. [722] (licensed under CC-BY 4.0), while (b) is from Liu et al. [724] (licensed under CC-BY 4.0).

delamination. Interestingly, as shown in Fig. 50(a), the stress effect on the elastic lamellae and intra-lamellar cells is most pronounced near the merged pools, suggesting an elevated risk of lamellar rupture. In addition, GAGs can grow circumferentially and even coalesce with other pools.

Continuing to model the effects of GAG accumulation, Liu et al. [724] adopted a distinct approach to model the swelling pressure of GAGs in the aortic media. They employed a strain-energy function with weighted fractions of individual constituents, including SMCs, collagen fibers, elastic fibers, and GAGs/PGs. The swelling pressure was modeled using swelling polymer theory. The constitutive behavior of collagen and elastic fibers was represented using the HGO model, while GAGs/PGs were modeled with an isotropic contribution, incorporating the osmotic or Donnan swelling pressure. Constitutive parameters were drawn from various sources [700,719,725–727] and three numerical examples were established. First, a thick-walled cylinder with two layers was generated, with inner pressure applied. The results indicated that the current fixed charge density is inversely proportional to the volume ratio and proportional to the reference fixed charge density. Second, seven lamellar units were modeled with fiber struts orthogonal to the lamellae. This submodel (Fig. 50(b)) was employed to represent the detailed microstructure using a finer mesh. The swelling pressure of GAGs generated failure stresses comparable in magnitude to the failure stress observed in tensile struts. Additionally, stress concentrations were identified between interlamellar struts, see Fig. 50(b). Finally, a comparison between pressurization with and without preconditioning was conducted, revealing that residual stresses contribute to increased load-bearing capacity with respect to aortic dissection.

A general method for modeling local inhomogeneities in the aortic wall was presented by Ranftl et al. [717], who illustrated its application by simulating the random distribution of degraded elastic fibers in aortic dissection. They applied a previously developed constitutive model [685] to a simple uniaxial extension test example. A beta random field was used to model the distribution of elastic fiber degradation, and uncertainties were determined using a Bayesian encoder-decoder network. Although the approach was demonstrated using a relatively simple numerical example and a detailed discussion of the theoretical framework, the computational method can also be employed for 3D simulations and patient-specific settings to model local inhomogeneities in both healthy and pathological aortic walls. This approach can be particularly effective because pathological changes in the microstructure typically occur on a local scale.

The outside-in progression hypothesis (see Section 3.2.1) suggests that dysfunction of the vasa vasorum may significantly reduce nutrient supply to the arterial wall, leading to a hypoxic state [438]. This hypoxic condition could initiate ischemic events, potentially catalyzing inflammatory responses and ultimately triggering arterial dissection. To address this, Soleimani et al. [728] introduced a mathematical framework for modeling the progression of atherosclerosis and arterial wall dissection, which accounts for potential tears or ruptures in the arterial wall due to microinjuries within the vasa vasorum. The proposed coupled multiphysical framework was described using finite strain theory and formulated within a 3D continuum framework. From a methodological standpoint, it postulates that nutrient transport within the arterial wall can be described by the classic diffusion–reaction equation. Additionally, the model applied a phase-field approach to represent inflammation, using a binary indicator to depict the inflammation status, as introduced in a

previous study [729]. Subsequently, it connected the inflammatory state to the mechanics to model tissue overgrowth. To then model the rupture of the arterial wall, the model again used the phase-field approach. The developed framework was finally applied to selected numerical examples. For this purpose, material and geometrical parameters were collected from the literature. The first numerical example considered only atherosclerosis and illustrated the thickening of the arterial walls due to vasa vasorum malfunction. The inflammatory processes were governed by the nutrient gradient. Another numerical example, similar to the first but incorporating the damage mechanism, showed that a rupture occurs when a specific fracture threshold is violated. In this context, the intramural hematoma arised mainly from blood leakage from the vasa vasorum, which subsequently catalyzed the progression of the rupture. The results indicated pronounced stress concentrations at the rupture sites, with dissections progressing in both circumferential and axial directions.

In summary, most of the studies presented have examined pooled GAGs in depth. The growth and coalescence of GAGs can result in considerable stress concentration surrounding the pooling areas. In contrast, Ranftl et al. [717] proposed a general approach for modeling local material inhomogeneities, illustrated through the example of local elastic fiber degradation, which has the potential to be applied to patient-specific geometries. In addition, the role of vasa vasorum dysfunction in focal medial degeneration and subsequent arterial dissection has recently been investigated [728]. It is particularly important to consider local material or geometrical inhomogeneities resulting from medial degeneration, especially as they are likely to result in aortic dissection or fatal rupture of the wall. However, they can be difficult to identify in each patient's aorta and may vary depending on the aortic region. For example, regions susceptible to abnormal WSSs or wall stresses could be susceptible to the development of medial foci of degeneration. A random field approach could thus be a viable solution to address this issue. Other researchers have employed a similar approach to patient-specific geometries, as demonstrated by its application to aortic aneurysms [730] or normal aortas [731].

## 5.4. Growth and remodeling

To date, only two *in silico* studies have delved into modeling growth and remodeling in aortic dissection, each addressing several critical aspects detailed in Sections 3.1 and 3.2. These approaches are particularly noteworthy because growth and remodeling are of significant interest given the distinctive and complex nature of aortic dissection. While it shares similarities with diseases like aortic aneurysms, notably in terms of dilatation, aortic dissection distinctively results in two lumina separated by a dissection flap, along with a newly formed outer wall of the false lumen. These characteristics are particularly relevant during the transition from the acute to the chronic phase and can only be properly understood when considered in the context of a growth and remodeling framework. To this end, the studies discussed below use the theory of homogenized constrained mixture [732], a special case of the constrained mixture theory [733].

In their pioneering computational study, Zhang et al. [734] focused on the role of growth and remodeling in false lumen dilatation following chronic type B dissection. To explore this, they developed a 3D finite element model of growth and remodeling for aortic dissection. This model integrates principles from mechanobiology of the aorta, formulated based on the homogenized constrained mixture theory, see [732.735]. In brief, the homogenized mixture comprises a matrix with a network of elastic fibers, passive reinforcements represented by collagen fiber families in various directions, and active reinforcements accounting for the contractility of SMCs oriented in the circumferential direction. Subsequent model applications represented the aortic wall, including the adventitia and media, connected by elastic springs with high stiffness to represent a healthy aortic wall in a homeostatic state. The false lumen was created by breaking interfacial elastic springs in a predefined region between the media and adventitia. Initially, the model was applied to idealized aortic geometries with cylindrical and toric shapes to validate its feasibility and efficiency. It was then applied to a patient-specific geometry of a thoracic aorta, obtained from [736,737], and demonstrated its potential for more complex and clinically relevant applications, as shown in Fig. 51(a). In each circumferential segment of the respective model, the length of interfacial springs to be broken was predefined, as this study did not primarily focus on the initial tear formation or the subsequent tear progression. The findings showed that growth and remodeling naturally trigger false lumen dilatation after dissection with the aim of restoring stress equilibrium. The so-called gain parameter, which is related to collagen fiber growth and remodeling, plays a pivotal role in the stability of aortic dilatation. A small gain parameter might provoke excessive false lumen dilatation, while a large gain parameter helps restore a stable state of the artery during the chronic phase. Other important mechanobiology-related parameters include the circumferential length of the dissection and the pressure in the false lumen, both of which can significantly impact the stability of the false lumen dilatation. The results suggested that both a wide tear and elevated false lumen pressure tend to promote an unstable development of false lumen dilatation, as depicted in Fig. 51(b).

A second recent study that focused on the growth and remodeling of a dissected aortic wall and integrated the framework of homogenized constrained mixture theory [732] was published by Gheysen et al. [714]. Their study was motivated by the finding that the presence of inflammatory cells in the dissection suggests a role of inflammation in the thickening process of the dissected flap, see Section 3.2.3 and [483,738,739]. Intriguingly, this remodeling occurs even when the stress state matches the homeostatic level, a concept introduced by Latorre and Humphrey [721] in the context of severe inflammatory responses that cause maladaptation of arterial walls. To this end, they created an idealized model of a dissected wall to simulate the production and degradation of elastin and collagen under stress and inflammation-related conditions, see Fig. 52. The model assumed that the aortic wall remodels collagen to restore homeostasis by sensing differences between the current stress state and a homeostatic one [733]. The study used an idealized geometry of a dissected aorta section derived from CT imaging [609], applying an axial stretch, a pre-stressing algorithm, and an inner pressure of 80 mmHg in both lumina. The acute material behavior of the medial and adventitial layers of the dissected wall was modeled with an HGO model that encompassed two non-dispersed fiber families. Notably, parameters were chosen based on pulse wave velocity, a method presented in [740]. The growth and remodeling framework used to model the transition from acuity to the chronic phase considered both transient and permanent effects of inflammation on the wall (Fig. 52(a)). Moreover, they applied this



**Fig. 51.** (a) Illustration of patient-specific results of an aortic dissection demonstrating both the geometric alterations and the corresponding von Mises stress changes at three selected time points over a six-year period following a chronic type B dissection. It also shows the dilatation of the circumferential cross section at the site of maximum dissection. These changes are depicted at three distinct time points: (i) at the onset, (ii) after three years, and (iii) after six years of growth and remodeling. (b) Impact of the luminal pressure ratio on aortic dilatation, depicting the temporal evolution of the maximum outer diameter of the patient-specific dissected artery over a period of seven years. Figure is adapted from Zhang et al. [734] with permission from Springer Nature.

framework to all layers (full model) and to only the dissection flap and the outer wall of the false lumen (local model). They compared diameter dilatation, flap thickening, and microstructural shifts over a 90-day period. For the growth and remodeling framework, they implemented 30 growth steps, each spanning three days, to reflect the 90-day period of growth and remodeling. Due to the lack of established optimal growth and remodeling parameters for aortic dissections, they utilized a Latin hypercube sampling with 1000 parameter combinations, evaluating four inflammatory patterns: transient and permanent inflammation, and a full and a local model. The results indicated a decrease in elastin and in most cases an increase in collagen in the dissection flap microstructure (Fig. 52(b)). The transient inflammatory pattern was particularly effective in replicating clinically observed rates of dissection flap thickening, diameter expansion, and microstructural adaptations (Fig. 52(c)). It is important to highlight that this study did not aim to reveal the ground-truth growth and remodeling mechanisms and corresponding parameters in aortic dissections. It simply supported the assumptions that transient inflammation potentially plays an essential role in tissue growth and remodeling of aortic dissections. Furthermore, it is important to note that despite the limitations mentioned in their study, they did not consider the pressure difference between the lumina and the effect of locally altered WSSs on endothelial mechanosensing. For example, as we will show later in Sections 6 and 7, locally altered WSSs were associated with the characteristic flow jet. These aspects are likely to have a significant impact on the process leading to growth and remodeling.

In summary, it is important to emphasize the significance of studying growth and remodeling particularly in chronic type B dissection, as demonstrated by these preliminary studies. Although there is room for improvement in various areas, these studies establish a promising foundation for future investigations into false lumen dilatation and other growth and remodeling-related processes.

## 5.5. Key findings and publication index

To summarize this section, we present the publication index of multiscale material models for damage and failure in Table 6. Due to the large number of studies available, only a selected range was discussed; however, this table covers an even broader range of studies. Essential information about the modeling approaches and the objectives of each study is included. Finally, the most important findings are listed conclusively.

- The direction of dissection progression generally follows the microstructure of the aorta and the direction of maximum stiffness, which coincides with the alignment of collagen fibers. This often results in a helical pattern.
- Both radially-directed collagen and elastic fibers appear to contribute to the delamination strength of the aortic media. The
  delamination process follows an avalanche-like failure pattern that results from the local buildup of strain energy and then leads to
  cascade-like failure of the distributed interlamellar fibers. Additionally, the delamination strength is influenced by the inherent
  microstructure of the aorta, which varies depending on its location along the vessel.
- Accumulation of GAGs through growth and coalescence can lead to significant stress concentration around the areas of pooling. Moreover, local inhomogeneities such as pooled GAGs or focal areas of medial degradation can influence the direction of



**Fig. 52.** (a) Schematic overview of the inflammatory pattern (permanent or transient) and the location (local or full) indicated in red color. (b) Representative evolution of the elastin and collagen content over time with a local and transient inflammatory pattern, with content expressed as a fraction of the initial volume. (c) Predicted geometrical changes over time for samples with a transient inflammatory pattern applied over different locations (local and full) that resulted in thickening rates outside and within the extended clinical range. The deformed configurations at day 0,15, 42, and 90 are shown in an increasingly bright shade of red color. Figure is adapted from Gheysen et al. [714] (licensed under CC-BY 4.0).

propagation to either the lumen or the adventitia, regardless of whether they represent weaknesses or strengths. The propagation could then follow the oblique region of these local inhomogeneities.

- In vessels prone to dissection, axial stretch often decreases. Therefore, the presence of axial stretch could provide potential protection against dissection. Additionally, residual stress could play a role in both dissection propagation and the configuration and extent of the dissection flap. However, the exact nature of this role is not yet fully understood.
- The mean and maximum systolic aortic pressure along with the pulsating frequency are the most important factors for dissection propagation. In this context, Donnan swelling pressure plays a critical role in the initiation phase.
- Tear nucleation sites typically experience normal static blood pressure, which is usually too low to separate the layers and extend the tear. In these instances, Donnan swelling pressure might reach the threshold required to initiate separation. However, as the volume of the tear increases, the significance of Donnan swelling pressure diminishes, making blood pressure a more dominant factor. If it becomes sufficiently high, it can cause the tear to propagate further.
- The deposition of collagen fibers, the circumferential length of the outer wall of the false lumen, and the pressure within the false lumen may play a crucial role in the stability of aortic dilatation. Conditions such as low fiber deposition, a wide tear, and elevated pressure in the false lumen tend to favor false lumen dilatation.
- Comparative analysis between transient and permanent inflammation in aortic dissection layers shows that computational results favor a transient inflammatory pattern, especially when comparing the growth rates of dissection flap thickness and aortic diameter with those observed in clinical studies.

## Table 6

Publication index of multiscale material models on damage and failure related to aortic dissection, ordered by year of publication (FL: false lumen).

| Report                             | Year         | Method(s): | Aethod(s):                                |                                 | Main objective(s)   |  |  |
|------------------------------------|--------------|------------|---|---------------------------------|---|--|--|
|                                    |              | Numerical  | Damage and failure                        |                                 |   |  |  |
| Delamination of the wall           |              |            |   |                                 |   |  |  |
| Gasser & Holzapfel [648,650]       | 2005/06      | PUFEM      | -   | FEAP                            | Modeling balloon angioplasty in aortic<br>dissection; validation to peeling tests   |  |  |
| Rajagopal et al. [11]              | 2007         | FEM        | Constrained mixture model                 | -                               | -   |  |  |
| Ferrara & Pandolfi [653,654]       | 2008/10      | FEM        | Cohesive zone model                       | Custom code                     | Validation to peeling tests   |  |  |
| Pal et al. [656]                   | 2014         | FEM        | Cohesive zone model                       | Custom code                     | Validation to peeling tests; impact of<br>radially-directed collagen fibers on<br>delamination  |  |  |
| Shah et al. [659]                  | 2014         | FEM        | Individual fiber failure                  | Custom code                     | Validation to uniaxial and biaxial extension tests  |  |  |
| Leng et al. [661]                  | 2015         | FEM        | Cohesive zone model                       | Abaqus <sup>a</sup>             | Validation to atherosclerotic plaque<br>delamination tests  |  |  |
| Witzenburg et al. [660]            | 2016         | FEM        | Individual fiber failure                  | Custom code                     | Validation to uniaxial extension, biaxial extension, peeling, and lab shear tests   |  |  |
| Thunes et al. [663,665]            | 2016/18      | FEM        | Individual fiber failure                  | Custom code                     | Modeling medial lamellar unit with microstructural-based model  |  |  |
| Noble et al. [666]                 | 2017         | FEM        | Cohesive zone model                       | MARC                            | Validation of peeling tests to model<br>catheter-induced dissection   |  |  |
| Leng et al. [662]                  | 2018         | FEM        | Cohesive zone model                       | Abaqus <sup>a</sup>             | Validation of atherosclerotic plaque  |  |  |
| Brunet et al. [461]                | 2019         | FEM        | Cohesive zone model                       | Abaqus &                        | Validation to tensile rupture tests with  |  |  |
| Yu et al. [671]                    | 2020         | FEM        | Individual fiber failure                  | Abaqus <sup>a</sup>             | Validation of peeling tests; role of radially-<br>directed collagen fibers in avalanche-like  |  |  |
| Wang et al. [674]                  | 2020         | FEM        | Individual fiber failure                  | Abaqus <sup>a</sup>             | Validation of peeling test; role of radially-<br>directed elastic/collagen fibers in oscillating<br>delamination                          |  |  |
| Dissection propagation and flap co | onfiguration |            |   |                                 |   |  |  |
| Wang et al. [675]                  | 2015         | FEM        | Cohesive zone model                       | FEAP                            | Impact of fiber-reinforced tissue on tear<br>propagation with 2D medial strip   |  |  |
| Gültekin et al. [681–683]          | 2016–19      | FEM        | Phase-field approach                      | FEAP                            | Impact of pressure on tear propagation in<br>symmetric 3D tube  |  |  |
| Wang et al. [677,679]              | 2017/18      | XFEM       | _   | Abaqus                          | Impact of residual stress/tear length/<br>pressure/fiber alignment on flap<br>configuration/propagation direction in<br>summetric 2D tube |  |  |
| Ahuja et al. [613]                 | 2018         | FEM        | -   | Abaqus                          | Validation of flap configuration to <i>in vitro</i>   |  |  |
| Mousavi et al. [684]               | 2018         | FEM        | Continuum damage<br>theory & constrained  | Abaqus <sup>a</sup>             | Impact of pressure on location of initiation/<br>direction of dissection propagation in aortic  |  |  |
| Li et al. [704]                    | 2019         | FEM        | Incremental deformation<br>model          | FEAP                            | Impact of residual stress/axial stretch/fiber<br>alignment/pressure on FL volume with 2D<br>medial layer                                  |  |  |
| Rolf-Pissarczyk et al. [450,685]   | 2021         | FEM        | Exclusion of degraded fibers              | FEAP &<br>Abaqus <sup>a</sup>   | Modeling degradation of radially-directed   |  |  |
| Brunet et al. [691]                | 2021         | XFEM       | _   | Abaqus <sup>a</sup>             | Impact of tear geometry/pressure/axial<br>stretch/material parameters on tear<br>pronagation in symmetric 3D tube                         |  |  |
| Ban et al. [693,696,697]           | 2021/22      | FEM        | Phase-field approach                      | FEniCS                          | Modeling needle experiments to uncover<br>microstructural-based tear propensities in<br>descending (abdominal porta                       |  |  |
| FitzGibbon et al. [701,703]        | 2021/22      | FEM        | Cohesive zone model                       | Abaqus                          | Impact of pressure on tear propagation with<br>symmetric 3D tube; development of novel  |  |  |
| Zhang et al. [705]                 | 2022         | FEM        | -   | Abaqus <sup>a</sup>             | Impact of residual stress on flap   |  |  |
| Han et al. [708]                   | 2023         | FEM        | Cohesive zone model                       | Abaqus <sup>a</sup>             | Impact of tear geometry/residual stress/<br>luminal pressure difference on tear   |  |  |
| Yeerella et al. [741]              | 2024         | FEM        | Fracture mechanics theory (energy release | Abaqus <sup>a</sup> &<br>Semi-  | propagation in symmetric 3D tube<br>Impact of material parameters/tear depth/<br>wall thickness/pressure on propagation in                |  |  |
| Gheysen et al. [709]               | 2024         | FEM        | rate)<br>-                                | analytic<br>Abaqus <sup>a</sup> | symmetric 2D tube<br>Impact of material parameters/flap<br>thickness on stress/deformation  |  |  |

#### Table 6 (continued)

| Report  | Year    | Method(s):<br>Numerical | od(s): S<br>erical Damage and failure        |                              | Main objective(s)   |
|---|---------|-------------------------|--|------------------------------|---|
| Focal areas of medial degeneration<br>Roccabianca et al. [79,700] | 2014    | FEM                     | -  | FEBio                        | Impact of pooled GAGs in media on wall<br>stress/stress concentrations with 3D medial   |
| Ahmadzadeh et al. [719,722]                                       | 2018/19 | SPH                     | Continuum damage<br>theory                   | FEBio &<br>Custom code       | Impact of microstructural changes in media<br>on wall stress; modeling growth and<br>coalescence of pooled GAGs                         |
| Liu et al. [724]  | 2022    | FEM                     | Material volume fraction                     | Abaqus <sup>a</sup>          | Impact of swelling pressure of pooled GAGs<br>in media on stress in elastic fiber struts  |
| Ranftl et al. [717]   | 2022    | FEM                     | Exclusion of degraded fibers                 | FEAP                         | Stochastic modeling of degradation of<br>radially-directed elastic fibers with 3D<br>medial strip                                       |
| Soleimani et al. [728]  | 2023    | FEM                     | Phase-field approach                         | AceGen &<br>Ansys            | Modeling tearing (rupture) in the artery<br>wall following microinjuries in vasa<br>vasorum   |
| Growth and remodeling   |         |                         |  |                              |   |
| Zhang et al. [734]  | 2022    | FEM                     | Homogenized<br>constrained mixture<br>theory | Open-<br>Source <sup>b</sup> | Modeling FL dilatation with growth and<br>remodeling framework  |
| Gheysen et al. [714]  | 2024    | FEM                     | Homogenized<br>constrained mixture<br>theory | Abaqus <sup>a</sup>          | Comparison between dilatation/layer<br>thickness growth rates obtained with<br>different inflammatory patterns/clinical<br>observations |

<sup>a</sup>Standard; <sup>b</sup>[742,743]

# 6. In silico hemodynamic models

Due to its complex nature, the vascular system in health and disease is initially modeled using core components that focus on single functions such as arterial fluid dynamics, wall mechanics, and even thrombus formation. Each core model requires thorough mathematical analysis and efficient numerical methods. These core models are then integrated into comprehensive, coupled models to represent significant parts or the entire system, which requires appropriate coupling conditions and innovative numerical strategies for stable and efficient solutions.

Clinical data, essential yet challenging for vascular system models, have advanced alongside computational modeling, largely due to improvements in medical imaging. Radiological images such as CT and MRI are crucial for creating computational domains, but geometric reconstruction, especially when modeling aortic dissections, is complex and requires advanced tools. Beyond image acquisition, segmentation, and geometric reconstruction, meshes must be generated, realistic boundary conditions obtained, ideally from *in vivo* data and biological vascular data acquired for various components, such as rheological parameters for the blood or stiffness parameters for the arterial wall. The governing equations for fluid and solid problems must be defined, coupled, and solved efficiently. Each step brings its own challenges, particularly in aortic dissection. Significant inter- and intra-patient variability and uncertainty further complicate model calibration and validation and underline the need for (patho-) physiological modeling of *in vivo* situations. Particularly noteworthy in this context is the pioneering work of Quarteroni and his collegues in the last decades, see, e.g., [744–746], as it provided the basis for the accurate modeling of real physiological conditions of the cardiovascular system.

Despite these challenges, numerous models have successfully addressed aortic dissection using computational fluid dynamics (CFD). These models aim to enhance the understanding of pathological processes and open new avenues for therapeutic planning and the design of implantable devices such as TEVAR or graft deployment. Although the elasticity of the arterial wall and dissection flap are neglected, unlike fluid–structure interaction (FSI) simulations, which are discussed in Section 7, CFD simulations can still provide valuable insights under certain conditions.

CFD simulations can be utilized, for instance, to estimate flow splits between true and false lumina, to predict stresses acting on the vessel interior or constituents of the fluid, or to provide hemodynamic indicators relevant for the quantitative comparison of alternative treatment options or in the context of parameter studies on virtual cohorts. With a particular focus on CFD applied to aortic dissection, the following selected works have significantly contributed to our current understanding of this complex disease, its genesis, possible influencing factors, and treatment options. The following studies can be categorized into four groups: (i) CFD studies that employed simplified geometries; (ii) cross-sectional case studies on patient-specific geometries; (iii) virtual case studies in which the geometry was artificially modified to investigate specific phenomena; and (iv) longitudinal case studies.

#### 6.1. CFD studies that used simplified geometries

Initially, CFD studies were conducted to examine the hemodynamics of aortic dissection using simplified geometries, which was often necessitated by limited computational resources, lack of software infrastructure or medical data and suitable tools to process them. However, today, the state-of-the-art approach incorporates patient-specific geometries when necessary, while simplified studies

on idealized geometries are employed for specific reasons. A key aspect in this regard is the development of computational technology, motivated by advances in the broad research area of biomedical engineering and computational mechanics, so that complex patient-specific geometries are now within reach.

In their first attempt to quantitatively evaluate the accuracy of a CFD model in predicting intra-luminal pressures in various clinical scenarios for aortic dissections in 2015, Soudah et al. [747] demonstrated the ability of the CFD model to replicate the primary characteristics of an *in vitro* experimental setup accurately [618], as illustrated in Section 4.3. This study was subsequently followed by a series of additional studies. Then, in the work of Rudenick et al. [748], the influence of wall elasticity, also called compliance, on intraluminal hemodynamics in a dissecting aorta was investigated by developing and validating a lumped-parameter model based on the in vitro data obtained from a pulsatile hydraulic circuit [618]. The results indicated that setting wall elasticity to physiological values resulted in a decrease in systolic pressure by up to 33% and an increase in diastolic pressure of up to 63%, which consequently resulted in a reduction in pressure wave amplitude by up to 86%. Furthermore, a decrease in stiffness was associated with an increase in multidirectional intraluminal flows and a transition in behavior, with the vessel transforming from two parallel vessels to a vessel with a side-chamber configuration. This study highlighted the critical role that wall elasticity plays in determining intraluminal pressures and flow patterns in aortic dissection. Subsequently, Rudenick et al. [516] investigated how temporal variations in size and spatial distribution of tears, as well as the presence of abdominal side branches emanating from the false lumen, influence the flow pattern. A smaller tear resulted in a more damped and shifted flow profile in the false lumen. In contrast, a large tear increased the retrograde flow proximally and antegrade flow distally. In the true lumen, blood pressure was more likely to be hypertensive with a small tear or hypotensive with a large tear. As shown in Fig. 53, when more than 50% of the total tear area was located proximal, systolic false lumen flow became more monophasic with a decrease in the reverse flow and an increase in diastolic retrograde flow. Conversely, with more than 50% of the tear area located distally, the model showed a clear systolic biphasic pattern with an increase in systolic false lumen reverse flow and predominantly antegrade diastolic flow. Although the aortic wall stiffness did not change the temporal variation of the flow, a stiffer aortic wall increased pulse pressure in the true and false lumina, respectively. In systole, the pressure was then increased, while the maximum pressure decreased in diastole. The results were finally compared with patient flow profiles in the false lumen at the diaphragm level, obtained using PC-MRI. This comparison gave a satisfactory match with maximum



**Fig. 53.** Illustration of a simplified aortic dissection geometry showing the true lumen ('TL') and false lumen ('FL') connected via two intimal tears. The baseline case has two large intimal tears without visceral side branches, demonstrating the spatial variability of the flow pattern in (a) the true and (b) the false lumen. Positive values indicate antegrade flows, and negative values indicate retrograde flows. The variability in the false lumen flow pattern is illustrated for scenarios with (c) a larger proximal tear, (d) equally sized tears, and (e) a larger distal tear. The yellow dashed lines represent the diaphragm level. Figure is adapted from Rudenick et al. [516] (licensed under CC-BY 4.0).

flow rate errors below 10%.

Additionally, several studies were published using simplified CFD models, as outlined in the following. In their study, Fan et al. [749] used a simplified CFD model to study the effects of a TEVAR in type B dissection, particularly with regard to the coverage of the proximal entry tear during the post-repair process. The variations in the model also involved the ratio of the area of the false to the true lumen as well as the location and size of the exit tear. The results indicated that a smaller false lumen likely poses less risk because the region of stagnant fluid or complete thrombosis would be larger. Moreover, it has been found that a larger exit tear generally causes blood to move to a larger region. Guan et al. [750] investigated bypass treatment for type B dissection using 2D idealized geometric models and observed decreased blood flow velocity and pressure in the false lumen after treatment. Chitsaz et al. [751] developed five 2D models of aortic dissection to examine the role of length, antegrade propagation, and retrograde propagation. Of the five models, four had a single entry tear and one had a distal exit tear. They discovered that the tearing force was proportional to dp/dt and the squared dissection length, with pulsatile flow conditions resulting in higher tearing force as steady flow conditions. In fact, a second distal tear significantly reduced the tearing force. Tang et al. [752] developed a simplified 3D model of aortic type B dissection and investigated the effect of false lumen dilatation by creating five models with increasing diameters of the false lumen. Among their findings, they demonstrated that a flow jet through the intimal tear impinges on the outer wall of the false lumen, where the resulting force acting on the false lumen wall increases with increasing diameter. However, it might be difficult to generalize the observations from the developed model because their significance might be limited by the simplified geometric modeling that occludes all supraaortic vessels, and the choice of outflow boundary conditions that directly determine the ratio of true-to-false lumen flow at the outlet. Ben Ahmed et al. [753] created 14 simplified chronic type B dissection models with straight and curved vessel shapes, varying numbers of proximal and distal tears, and a false lumen location in the inner and outer arch to virtually compare different surgical treatment options. The curvature had no influence on the pressure and flow ratios between true and false lumina. The authors concluded that the combination of proximal and distal tear sizes influences hemodynamics, with larger proximal tears increasing the false lumen pulsatile pressure and larger distal tears decreasing it. Occlusion of the proximal tear resulted in a 54% reduction in pulsatile pressure. Zadrazil et al. [630] investigated type B dissection using particle image velocimetry and CFD simulations in a simplified geometry and used the particle image velocimetry results as validation. The experimental and numerical results using the simplified geometry agreed well, even though the in silico model considered different tear sizes and size ratios. It was observed that increasing the exit tear size considerably increased the flow rate in the false lumen, significantly decreased the WSS, and reduced the pressure difference between the false and true lumina. Increasing the entry tear size increased the flow rate through the false lumen, slightly increased the WSS, and raised the pressure difference between the false and the true lumen. The study by Zorilla [754] is also worth mentioning in this context. This study validated the proposed CFD methodology using an in vitro reference test case and explored various configurations of dissection flap tears in type B dissection, thereby mimicking blood flow in real-life scenarios.

In addition to the previous studies, Peng et al. [755] conducted a study on acute type B dissection in which a total of 163 patients were examined and compared with a control group of 184 patients. The aim was to understand the morphological differences and the associated changes in blood flow. For this purpose, various parameters were statistically analyzed and idealized CFD models were established for hemodynamic simulations. The results of the study highlighted that the diameters at landmarks of the proximal aortic segment and the lumen volume were significantly larger in the dissection group, suggesting a positive correlation between volume and diameter. Furthermore, the study found that with increasing volume, the kinetic energy, time-averaged WSS (TAWSS), and oscillatory shear index (OSI) at the distal area of the left subclavian artery also increased. This suggested that these factors may contribute to disease progression. On top of that, Qiao et al. [756] developed idealized models of diseased aortas to simulate conditions such as aortic aneurysm, aortic coarctation, and aortic dissection. These models were then employed to virtually deploy a stent-graft introducer sheath commonly used in TEVAR to assess its hemodynamic effects. Comparative analyses from this study indicated that the obstruction caused by the introducer sheath led to an elevated blood flow to the supra-aortic branches. Furthermore, the sheath leads to significantly altered blood viscosity in the aortic arch and the descending aorta. This underscored the importance of using non-Newtonian models for more accurate simulations in scenarios where medical devices might cause reduced circulation, stagnant flow zones, or areas of high shear rates that could lead to shear-thinning.

Lastly, two consecutive studies are presented. In the first, Li et al. [757] constructed population-based 3D models of post-operative thoracic aortic dissection to investigate the hemodynamic consequences of varying tear characteristics. Factors such as the number of tears, the maximum distance between tears, and tear size were taken into account. Their findings revealed that the area of high OSI and activation potential of endothelial cells increased with increasing number and size of tears, although a consistent tear distance lead to a stable low relative residence time (RRT) area. Additionally, they noticed a significantly increased false lumen blood flow. Keeping the number and size of tears constant, increasing the maximum tear distance by 10 mm resulted in an enlarged area of low RRT in the false lumen and a higher average pressure difference. Subsequently, Li et al. [758] applied this model to explore the influence of tear parameters on the formation of false lumen thrombosis after TEVAR. Simultaneously, they conducted a retrospective study of acute type B dissection, focusing on the prediction of false lumen thrombosis based on morphological parameters. Their computational findings reflected previous research. In particular, they found that as the number of tears increased, the flow field became more regular and stable, and flow jet was more evident at the most distal exit tear.

In summary, simplified geometries have proven effective to validate methods against experimental models, develop numerical methods, or investigate specific hemodynamic mechanisms in aortic dissection. For instance, a broad understanding of the effects of multiple tears on hemodynamics, which facilitate communication between the true and false lumina, could be developed. While some of these findings, such as the general presence of a pressure difference between the lumina or the occurrence of antegrade and retrograde flow, could be further confirmed using patient-specific models, many must first be adapted to patient-specific geometries to establish a practice-relevant proof of concept.

#### 6.2. Cross-sectional case studies

After providing an overview of studies investigating simplified setups, we will now focus on patient-specific geometries. In the context of *in silico* hemodynamic models, this encompasses cross-sectional, virtual, and longitudinal case studies. First, an overview of cross-sectional case studies is provided, which involves the analysis of medical data collected from a representative case at a specific point in time. It therefore provides a snapshot of the subject under investigation with the aim of drawing conclusions or deriving connections from the currently observed patterns.

In 2010, Cheng et al. [759] investigated flow phenomena in a pre-surgical type B dissection using a shear stress transport (SST) turbulence model and neglecting vessel wall motion. A time-scaled, flat flow profile at the inlet and a zero reference pressure were specified as boundary conditions. In the presented case, 80% of the volumetric flow entered the false lumen, causing a high WSS near the flap in the true lumen wall and a locally highly perturbed, possibly turbulent flow with strong recirculations within the false lumen. TAWSS, OSI, and turbulence intensity were measured and a comparison of Newtonian and non-Newtonian fluid models showed rather small differences in the results obtained in terms of absolute value. However, the comparison of laminar and turbulent models showed a significantly increased WSS in the latter case. The highest WSS was reported at the coarctation of the aorta, which is due to the reduced diameter, which may in turn influence the subsequent dilatation and progressive weakening of the affected aorta.

The computational models were then validated in a follow-up study in 2014. In this study, Cheng et al. [760] compared the computational results with *in vivo* velocity data obtained via PC-MRI (see Figs. 54(a) and (b)). Regions with complex flow patterns such as reversal and recirculation were also qualitatively captured, assuming rigid walls, excluding arch branches, and using simple reference pressures at the outlets. In the CFD simulations, the false lumen flow rate was over-predicted by less than 25%, however, the overall good agreement confirmed the physiological relevance and validity of the computational model for type B dissection with a completely immobile dissection flap.

In their third study, one year later, Cheng et al. [761] then employed these developed numerical tools to compare two groups of four patients treated either medically or surgically via TEVAR. Hemodynamic factors were successfully linked to likely late outcome. High RRT of fluid particles in the false lumen was a strong predictor of false lumen thrombosis and high true-to-false lumen pressure



**Fig. 54.** (a) Comparison of the velocity profiles in three directions: inferior–superior ('IS'), anterior–posterior ('AP'), and right–left ('RL') during the systolic deceleration phase of the cardiac cycle at imaging plane 3. The data includes PC-MRI velocity measurements (in m/s) and corresponding CFD simulation results. Red represents positive velocities, while blue indicates negative velocities. Note that each direction and time point may have different color scales, reflecting the range of velocity values. Secondary velocity vectors are also projected onto the plane for further detail. (b) The diagram shows the flow rate waveforms in the true and false lumina, derived from both CFD simulation results and PC-MRI velocity data at imaging plane 3. Figure is adapted from Cheng et al. [760] with permission from Elsevier.

differences, and its location might be associated with aortic dilatation. Furthermore, the entry tear size was positively correlated with the false lumen flow rate, whereas the distance from the arch top to the entry tear was negatively correlated. False lumen patency was promoted by high flow rates in the false lumen and high true-to-false lumen pressure differences, a large entry tear size, and an intimal tear near the aortic arch. Although some simplifications were made in the computational modeling (rigid walls, simplified outlet conditions, etc.) and the small patient cohort limits the statistical relevance, the results presented are of great value and consistent with clinical observations.

Djorovic et al. [762] investigated the hemodynamics in type A dissection using patient-specific geometries derived from CT imaging of two patients. The boundary conditions for the study were consistent with those used by Shang et al. [763], see Section 6.4, including zero outlet pressures, prescribed inflow at the aortic root, and no-slip conditions on the stationary vessel wall. Laminar flow of an incompressible Newtonian fluid was considered using physiological parameters. Similar observations were made: during systole, a high-velocity flow jet forms through the intimal tear, leading to flow disturbances and high WSSs. In addition, the authors reported on the true-to-false lumen pressure differences, from which no further relations could be derived because they were not found to be particularly insightful. Moreover, high shear stresses were observed where velocities occurred, i.e. in the aortic arch, the brachiocephalic trunk, and the exit tear regions. The latter could therefore represent potential zones of further ruptures, according to them. This study highlights the potential of CFD simulations for use in clinical support and personalized medicine. However, to increase significance, larger study cohorts than the two patients considered here are required.

Abazari et al. [764] further extended a previous framework [765,766] and incorporated a generalized Newtonian fluid adopting the Carreau-Yasuda law. Three-element Windkessel models were taken from a similar patient-specific case and the geometry was reconstructed from CT data. The main goal was to verify the effectiveness of anti-hypertensive treatment of type B dissection, modeled by adapting the stroke volume and heart rate accordingly. This *in silico* study investigated the effects of changing the prescribed inflow on TAWSS, OSI, and highly oscillatory low magnitude shear (HOLMES). The clinically observed effectiveness of the drug was confirmed and showed numerically: (i) reduction of high velocity flow jets and high shear rates, (ii) reduction of oos of the drug was confirmed and thereby reduction of the stress endothelial cells were exposed to, (v) that HOLMES indicated regions prone to calcification, and (vi) that the areas that have high OSI and low TAWSS were reduced, which indicated a high risk of rupture, calcifications, or wall thickening. Overall, these hemodynamic indicators, correlating with clinical observations, bear high potential for future use in surgical planning and optimal treatment design.

Chen et al. [767] investigated hemodynamics and the effects of laminar or turbulent flow assumptions and highlighted the importance of proper mesh construction including boundary layers. For this purpose, a patient-specific type B dissection geometry with multiple tears connecting true and false lumina was derived from CTA images. Based on the observation of a stiff dissection flap in the present scenario, the study was limited to pure CFD computations, i.e. rigid aortic walls. No patient flow data were collected, so a representative scale was computed from MRI measurements in 13 individuals, which provided realistic temporal scaling but did not necessarily fit to the specific case. In order to deal with missing data, severe simplifications were made to the boundary conditions – a scenario that has high practical relevance, but also has a strong influence on the computational results. Flow rates were prescribed in the supra-aortic vessels that scaled the inlet profile at the aortic root and produced zero pressure at the three remaining outlets. Since the outlet on the celiac trunk was located at the distal end of the false lumen, a prescribed zero pressure had a direct influence on flow splits and pressure gradients. The relevance of this study lies in the detailed analysis of the flow fields and WSS, the resulting volumetric flow rates, and the true-to-false lumen pressure differences. Great care was taken to investigate numerical aspects such as the influence of the choice of k-w SST turbulence model or laminar flow assumptions and grid sensitivity. According to their study, constructing an improper mesh without boundary layers could dramatically affect the results, leading to a deviation of up to 30% of the solution components at critical points. In fact, the assumption of laminar flow led to reasonable results, with only a 1% difference compared with the solution obtained with a turbulence model, provided that boundary layers are properly accounted for. These two aspects are of particular importance for clinical use.

Building on previous studies [768,769], which we will discuss later in Section 7, Bonfanti et al. [770] applied a lumped parameter model [771] to account for the distal vascular system and arterial compliance. This approach allowed them to omit FSI simulations or even flow simulations in instationary domains, while still retaining some compliance in the model. In three cases of type B dissections, only routinely acquired and commonly available clinical data sets were used for parameter estimation and showed good agreement with *in vivo* blood pressure measurements. However, for configurations with highly mobile dissection flaps affecting volumetric flow rates and pressures in true and false lumina, the necessary assumptions may not apply because vessel motion does not only cause a phase shift of the pressure wave but also leads to a strong coupled problem. Of particular note are the pressure field and differences between true and false lumina, as shown in Figs. 55(a) and (b). Substantial true-to-false lumen pressure differences lead to increased dynamic loading on the dissection flap. Hemodynamic parameters such as TAWSS, OSI, displayed in Figs. 55(c) and (d), and RRT computed during post-processing were related to the onset and progression of aortic dissection. These time averages represent vessel WSSs (TAWSS), fluctuation in disturbed flow (OSI), and fluid particle residence time (RRT) correlating with thrombus formation.

In a cross-sectional study by Wang et al. [772], a strong correlation between blood flow measurements from medical imaging and CFD simulations was demonstrated. This study not only validated the accuracy of these methods but also provided visually appealing insights into the hemodynamic characteristics associated with aortic dissection. The CFD model was based on a patient's geometry post-surgery for type A dissection, which included various repairs such as ascending aorta replacement and aortic root repair, leaving a residual dissection extending from just before the left subclavian artery to the bifurcation of the right common iliac artery. By integrating CTA imaging, 4D-flow MRI, and routine blood pressure measurements, the study found consistent results between the CFD simulations and 4D-flow MRI measurements, with notable differences in diastole and the false lumen, where accurate flow estimations



**Fig. 55.** Results show (a) the pressure field at peak systole; (b) the true-to-false lumen pressure difference calculated during three phases of the cardiac cycle: mid-acceleration, peak, and mid-declining phases in systole; (c) TAWSS; and (d) OSI. The hemodynamic indicators show low TAWSS and high OSI in the false lumen, as well as high TAWSS at entry and exit tear positions and in regions of high flow rate in curved lumen regions. Figure is adapted from Bonfanti et al. [770] with permission from Elsevier.

from 4D-flow MRI were challenging to obtain. Fig. 56(a) from the study illustrated the characteristics of six intimal tears in the aorta and how blood flows through these tears during mid-systole. It was observed that blood flowed from the true into the false lumen at the entry and the first two re-entry tears, and then returned to the true lumen at the more distal tears. These flow patterns were visually confirmed using 4D-flow MRI. Additionally, Fig. 56(b) showcased the flow path within the thoracic aorta at various cardiac phases, highlighting the formation of vortices and recirculation zones, particularly in the false lumen and around the graft site. At mid-systole, for instance, recirculation regions developed around the jet flow in the false lumen, particularly at the locations of the tears. In the thoracic aorta, two slower-moving zones were found after flow separation points at the inner corner of the aortic graft and also after the luminal narrowing in the aortic arch at the site of the distal anastomosis of the graft. This detailed visualization underscored the complex and dynamic nature of blood flow in the presence of aortic dissection, providing valuable information for future patientspecific CFD simulations.

Armour et al. [773] adopted a similar computational setup, as introduced by Xu et al. [774], which is described later in Section 6.4. A total of five type B dissection patients were included in this cross-sectional case study. CTA images, 4D-flow MRI data, and invasive Doppler-wire pressure measurements were collected. Inflow profiles were constructed from the 4D-flow MRI data and the pressure measurements taken at multiple locations during TEVAR were used to fit three-element Windkessel models. To account for the laminar, incompressible flow of a non-Newtonian fluid, the Quemada model was considered with parameters taken from the literature. They conducted grid convergence studies for all five cases, which featured a large variety in the morphology of true and false lumina. In all cases considered, the CFD models predicted velocity fields and thus flow splits well. The computational approach led to higher average velocities due to the neglected compliance, while the flow jet in the entry tear region was still underestimated. The Doppler-wire pressure measurements were in reasonable agreement with the numerical results, but errors were noted in both the absolute systolic and diastolic values. These deviations could be due to the assumption of a fixed grid, which typically results in increased systolic pressure and decreased diastolic pressure. The model assumptions therefore influenced the simulation results in the expected manner and motivate investigations considering vessel deformation. In general, there was good agreement between the measurement data and the CFD simulation results.

Another study on aortic dissection using CFD was performed by Kimura et al. [775]. A total of 19 patients were selected, seven of whom had type B dissection and 12 patients were classified as type non-A non-B dissection. In brief again, type non-A non-B dissection is a condition where an intimal tear is located beyond the ascending aorta. This type of dissection is confined to the aortic arch or can be described as a retrograde dissection originating from the descending aorta, extending into the arch, and stopping before reaching the ascending aorta. CTA data from each patient were used to construct computational models that prescribed non-patient-specific but



**Fig. 56.** (a) Illustrations depicting the size and locations of six intimal tears, along with 3D velocity streamlines in the tear regions during midsystole. The locations of the tears are marked with dashed arrows, while the direction of flow at each tear is indicated with solid arrows. (b) Visualization of the flow path within the cut plane of the thoracic aorta, shown using velocity vectors and line integral convolution contours at four time points: early systole, mid-systole, late systole, and early diastole. Blood recirculation and vortices are highlighted with black ovals, and regions with relatively low velocity are outlined with orange ovals. The direction of flow during early diastole is indicated by a white arrow. Figure is adapted from Wang et al. [772] with permission from Elsevier.

realistic inflow rates, and Windkessel models tuned to achieve physiological pressure levels. They focused on the hemodynamic and morphological differences between type B and type non-A non-B dissection and compared true-to-false lumen pressure differences, intimal entry tear size, and WSSs between the groups. They reported that the median entry tear size was larger in the non-A non-B group, leading to increased false lumen perfusion. Furthermore, the proportion of cases in which the false lumen pressure was higher than the true lumen pressure increased significantly in early, middle, and late systole, while no relevant differences occurred in WSS. Therefore, they concluded that the increased systolic pressure in the false lumen, possibly caused by the increased entry tear size, could be associated with a complicated course and promote unstable conditions. Hemodynamic analysis may provide additional guidance for treatment planning in this regard, but larger study cohorts are required to further validate these results.

In their recent study, Moretti et al. [776] investigated flow patterns in the false lumen of three aortic dissections in the chronic phase: one fully perfused, one partially thrombosed, and one fully thrombosed. These flow patterns were compared with those observed in a healthy aorta. They reconstructed 3D geometries of the aorta using CTA and developed CFD models for analysis. The study revealed that partially thrombosed dissection and fully perfused dissection had the highest pressure and TAWSS values, which are significant risk factors for aortic degeneration and potential wall rupture. Furthermore, all cases demonstrated poor perfusion in the branch vessels connected to the false lumen. Additionally, both the true and false lumina showed elevated turbulence accompanied by critical stagnation points compared with the healthy aorta. However, a generalization of these results is difficult due to the distinct anatomical differences observed in the selected dissection cases, particularly regarding the location and extent of the false lumen. In addition, the determination of the parameters for the three-element Windkessel models, which were first determined for the healthy case and then applied to the dissection cases, could have a significant impact on the results.

The influence of inflow velocity profiles on hemodynamics was examined by Stokes et al. [777], specifically focusing on the relationships between flow helicity, oscillatory WSS, and false lumen dilatation. These relationships were demonstrated using 4D-flow MRI. A patient-specific geometry of a type B dissection was employed and CFD simulations were conducted. Blood was modeled as an incompressible, non-Newtonian fluid using the Tomaiuolo formulation [778] of the Carreau-Yasuda viscosity model. Additionally, a kw SST Reynolds-averaged turbulence model was utilized, setting a low turbulence intensity (1%) at non-wall boundaries. To evaluate different inflow profiles, the study compared simulations using the gold-standard 3D three-component inlet velocity profile derived from 4D-flow MRI data with flow-matched flat and through-plane profiles. These alternative profiles are commonly used when 4D-flow MRI is not available, see Fig. 57(a). To estimate the effects of imaging errors in 4D-flow MRI on flow solutions, the 3D three-component inlet velocity profile was scaled by a 25% increase, reflecting the extent of imaging error observed in previous studies. The outlets were modeled using three-element Windkessel models and the mean outlet flow rates for each major branch were obtained from 4D-flow MRI data. It is noteworthy that the measurement uncertainties in 4D-flow MRI were higher in these smaller branches than in the aorta. The study compared velocity, pressure, helicity, and WSS distributions. Helicity was used to identify streamwise vortex structures by quantifying the local orientation of velocity and vorticity vectors. Interestingly, since helical structures in the aorta have a larger scale than the fluid boundary layer, these features can be measured more reliably with 4D-flow MRI than with WSS measurements. Moreover, the study included a proper orthogonal decomposition analysis of the velocity field to assess how each inlet condition affects the coherent structures within the flow and the energy distribution. The results indicated that secondary inlet flows significantly impact the distribution of oscillatory shear and helicity throughout the false lumen. Even the 3D three-component inlet velocity profile showed notable differences in helicity and oscillatory shear when adjusted to account for imaging errors. The locations of the most significant false luminal growth are shown in Fig. 57(b) and marked by the sections  $\alpha$  and  $\beta$  in the 3D energy contribution to the



**Fig. 57.** (a) Computational domain along with schematic representations of the inlet and outlet boundary conditions. The lower right corner shows the velocity magnitude contours and vectors captured at peak systole for each patient-specific inlet velocity profile: 3D three-component inlet velocity profile ('3D') derived from 4D-flow MRI data ('4DMR') with flow-matched flat ('F'), through-plane profiles ('TP') and scaled inlet velocity profile ('+25%'). Differently colored regions within the aortic geometry indicate the areas targeted for helicity analysis. (b) The left side shows contours for TAWSS, OSI, endothelial cell activation potential ('ECAP'), and RRT in the thoracic aorta for the 3D case. The right side shows different contours compared with other cases. It is important to note that the contour ranges for TAWSS, endothelial cell activation potential, and RRT have been clipped to improve visual clarity. Contours vary between 25 and 50% of the boundaries set by the 3D contours. Figure is adapted from Stokes et al. [777] (licensed under CC-BY 4.0).

average pressure plot. In the thoracic region, TAWSS was highest and was located immediately distal to the primary entry tear, near section  $\alpha$ . Minimal variations in TAWSS occured in the remainder of the false lumen. TAWSS was elevated in the +25% case, particularly in the aortic branches due to the increased flow volume. Regions with low TAWSS, which were directly related to false luminal growth in aortic dissections, generally coincided with areas experiencing higher oscillatory shear. There is general consensus that a 3D three-component inlet velocity profile should be used where available, with through-plane profiles serving as the next best option. While a flat inlet velocity profile adequately captured the magnitude and trends of the circumferentially-averaged WSS metrics, the local values of OSI, energy contribution to average pressure, and RRT in the most dilated regions showed substantial differences when using the 3D three-component inlet velocity profile. Increasing the inlet velocity components by 25% to account for the underestimation by 4D-flow MRI affected the average magnitude of the differences in the WSS and helicity metrics. Overall, the quality of inlet velocity conditions in simulations of type B dissection can profoundly influence their clinical relevance.

A final study to summarize is by Black et al. [779], who introduced a novel reduced-order computational framework for the iterative flow-based calibration of three-element Windkessel model parameters. This development has significant implications for the creation of patient-specific boundary conditions, especially in healthy individuals and those with type B dissection. The study utilizes boundary conditions identified from retrospective VENC 4D-flow MRI data at both the inlet and outlet of an incompressible Reynolds-averaged Navier–Stokes approach in conjunction with a standard  $k-\omega$  SST turbulence model. Through a fully-integrated 0D-3D numerical framework, the calibration process aimed to identify a combination of parameters at each aortic outlet that best matches the clinical 4D-flow MRI data, which required an iterative process, see Fig. 58(a). Initial estimates for this process came from 1D modeling, which provided rudimentary pressure and flow waveforms at each outlet of the thoracic aorta; these initial Windkessel parameters then formed the starting point for the iterative calibration process. Remarkably, this automated calibration only required about 3.5 minutes per outlet and yielded physiologically relevant and highly accurate flow waveforms. A comparison between the estimated, calibrated, and *in vivo* data is shown in Fig. 58(b) for the percentage of inlet flow at the outlets and in Fig. 58(c) for the flow rate at a



**Fig. 58.** (a) Flowchart illustrating the calibration process for impedance ('Z'), resistance ('R'), and compliance ('C') within the boundary conditions of the three-element Windkessel model ('3EWM'). To configure a 0D-3D CFD model, each branch (or outlet) is integrated with a three-element Windkessel model. At the inlet, a flow waveform generated by 4D-flow MRI is converted into a parabolic velocity profile. The equation for the discretized three-element Windkessel model describes the relationship between pressure ('P') and flow ('Q') at each branch, where 'n' represents the current iteration. (b) Blood flow perfusion distribution in the entire 0D–3D CFD model before and after calibration compared with *in vivo* 4D-flow MRI data and (c) a representative flow waveform at the aortic dissection outlet of the true lumen, calculated via the 0D three-element Windkessel model before (dashed colored curves) and after (solid colored curves) calibration compared with *in vivo* 4D-flow MRI data (error bars represent mean standard deviation). (d) Boolean difference (calibrated – estimated) in TAWSS, OSI, and time-averaged pressure within the dissected aorta determined by 0D-3D CFD simulation. Figure is adapted from Black et al. [779] (licensed under CC-BY 4.0).

representative outlet. Moreover, Fig. 58(c) illustrates the Boolean difference between the calibrated and estimated hemodynamic parameters. When these calibrated boundary conditions were prescribed, the computed metrics related to near-wall hemodynamics and perfusion distribution were found to be consistent with existing clinical measurements and the literature, confirming the physiological relevance of the model. Therefore, this calibration technique can be applied in clinical settings where outlet flow rates are measured using methods such as 4D-flow MRI or ultrasound. It enables efficient determination of patient-specific boundary conditions for CFD simulations, provided the reduced-order model accurately captures the key flow characteristics.

The presented works and the summarized findings demonstrate that cross-sectional case studies can provide valuable insights into the hemodynamics of various patient cases. For example, they compared cases with stent and without stents as well as cases with varying numbers, locations, and sizes of intimal tears. These comparisons allowed them to observe the significant effect of fenestrations on hemodynamics. Furthermore, case studies facilitated the investigation of the influence of boundary conditions such as inflow and outflow conditions and emphasized their considerable impact on the validation results. Even small changes can lead to substantial deviations in the computational outcome. Overall, the model validation results demonstrated that the results of the *in silico* hemodynamic models agree well with *in vivo* data. Virtual case studies in which patient-specific geometries are virtually altered to examine the effects of certain morphological phenomena are a valuable extension of cross-sectional case studies, particularly when longitudinal data are not available.

#### 6.3. Virtual case studies

Based on cross-sectional case studies, invaluable insights into the underlying principles of cardiovascular diseases can be obtained, taking into account different geometric configurations, hemodynamic scenarios, or different problem parameters. In the context of an individual patient, virtual case studies have proven to be an excellent tool for identifying underlying mechanisms. They have demonstrated their applicability in clinical support and treatment planning within personalized medicine. For example, the patient-specific geometry can be modified by adding, occluding, or altering the size of intimal tears. This approach can be used to examine specific geometric features or explore possibilities for surgical interventions, even without longitudinal data.

In this sense, Karmonik et al. [780] studied the influence of entry and exit tear occlusion on hemodynamic parameters using fixed-grid CFD simulations, virtually adapting the geometry to model TEVAR, false lumen thrombosis, or fenestrations. In this particular case, changes in flow patterns and blood pressure were reported. Covering the exit tear resulted in a 300Pa increase in false lumen pressure. In contrast, virtual TEVAR resulted in negligible differences in true-to-false lumen pressure. Additionally, complete removal of the dissection flap reduced the pressure in the combined lumen from an initial value of 2400 to 800Pa. Despite the limitations of this study due to zero reference pressure applied to various branching vessels, resulting in flow splitting and possibly non-physiological pressure levels, the general observations still transfer to more realistic setups with correct pressure scaling and a deformable dissection flap.

The *in-silico* setup of Chen et al. [767], which was further improved in their follow-up study [781], has physiologically meaningful (instationary) outlet pressures. The latter study compared a four-year follow-up study modeling reduced stroke volume for the treatment of hypertension with multiple virtual endovascular repair variants. Two main findings from this analysis confirmed that (i) treatment of hypertension significantly reduces both the pressure drop between true and false lumina and the WSS and (ii) TEVAR stabilizes the true lumen, which can significantly reduce false lumen perfusion, which in turn leads to stagnant flow in the false lumen. TEVAR can thus trigger thrombosis, but this further depends on the size and number of tears in the dissection flap, with complete closure of all tears resulting in fully stagnant flow, giving the highest probability of false lumen thrombosis and the greatest potential for full aortic remodeling.

Wan Ab Naim et al. [782] investigated the effect of introducing additional fenestrations in a patient-specific model of type B dissection. Flow distribution, pressures in the true and false lumina, and WSS distributions were investigated in three configurations: the original geometry with a single entry and exit tear, and two variants where a fenestration with a diameter of either 10 or 16 mm was virtually added to the CTA-based geometry. The laminar flow of an incompressible Newtonian fluid was considered, driven by a flat inflow profile with temporal scaling from the literature. Prescribed flow fractions through the supra-aortic outlets were used to prescribe velocity profiles and pressure scaling was achieved via pulsatile pressure at the distal outlet. The artificially introduced entry tear connecting true and false lumina led to increased fluid exchange between the lumina, effectively reducing the true-to-false lumen pressure difference while only increasing TAWSS locally around the additional fenestration. They argue that reducing pressure in the false lumen, particularly in the distal region, reduces the risk of false lumen dilatation. The TAWSS was found not to differ significantly at the model scale. However, locally increased values could trigger the onset of dissection and should therefore be avoided. Therefore, hemodynamic investigations on patient-specific models might be used to determine optimal stent placement to reduce the true-to-false lumen pressure difference and to minimize the risk for further rupture.

Alimohammadi et al. [765] investigated a case of type B dissection in which a patient-specific geometry was reconstructed from CT images. Blood was modeled as an incompressible Newtonian fluid under laminar flow conditions in a fixed lumen. The main contribution of this work is an algorithm for tuning parameters of three-element Windkessel models to clinical data, which, depending on the number of outlets involved, tremendously facilitates tuning in patient-specific scenarios. The vessel wall and dissection flap were assumed to be fixed, but within this limitation the pressure wave propagation was accurately captured. The flow characteristics in the areas surrounding the entry and exit tears at peak systole and mid-diastole were rigorously analyzed. Fig. 59 shows the streamlines, the TAWSS, and the OSI. The developed methodology was further employed in patient-specific virtual stent implantation [766], comparing single and double stent-graft strategies based on velocity, pressure, energy loss, and WSS distribution. The computational model identified significant differences in the surgical procedures: a double stent-graft strategy further reduced the true-to-false lumen pressure difference and the energy loss was reduced by 40% compared with the single stent-graft variant.

Dillon-Murphy et al. [783] studied the effects of dissection flap morphology, i.e. the number and size of intimal tears or even the complete removal of the dissection flap on the hemodynamics. A detailed model of type B dissection was reconstructed from CTA data, imposing boundary conditions and optimizing Windkessel models based on 2D PC-MRI measurements. As for the numerical treatment, the flow of an incompressible Newtonian fluid was controlled either by a prescribed parabolic inflow profile informed by 2D PC-MRI data or by a lumped parameter model that indirectly enforces a target inflow waveform. Using the latter approach, the authors demonstrated that the required left ventricular stroke work was increased by 14% for the present configuration. The reduced true lumen cross-section in the dissected case resulted in increased velocities in the true lumen and a drastic increase in the TAWSS from 5 to 100 dynes/cm<sup>2</sup> in the entry tear region. The computational model was validated using 4D PC-MRI, which demonstrated good agreement of mean flows. The maximum error observed was 7.6% across three measurement planes, with an in vivo flow fraction of 79% compared with an in silico flow fraction of 76% in the dissected aorta. This method was also used to identify additional fenestrations connecting the true and false lumina. They demonstrated that neglecting additional fenestrations can lead to severe alterations of the true-to-false lumen pressure difference (18%) and false luminal flow (increased by 200%), which are important indicators for late complications and have a major impact on false lumen thrombosis. Although wall motion was not considered in this study and patient-specific pressure data were not available, the comprehensive analysis presented here has contributed significantly to our understanding of aortic dissection. This research emphasizes the importance of accurately capturing the size and location of entry and exit tears together with possible additional fenestrations.

Shi et al. [784] compared various configurations of type A dissection by virtually placing a dissection flap in an aorta reconstructed from CTA data. Each dissection flap had a single entry and exit tear, while the ratio of the entry-to-exit tear area was set to 1.0 and 2.0, respectively. The location of the tear also varied from the proximal end of the false lumen in the ascending arch to the apex of the arch, resulting in a total of four different configurations. The average pressure and flow data were considered to account for physiological flow conditions and drive the laminar flow of an incompressible Newtonian fluid. They reported that flow rates and TAWSS in the false lumen were higher with a larger exit tear and that the entry tear closer to the aortic root also increased TAWSS and pressure in the false lumen. Additionally, in cases where the proximal end of the false lumen did not have an entry tear, high residence times were observed, potentially triggering thrombosis in the region of stagnant flow in the false lumen. Due to the virtual nature of this study, it was able to



**Fig. 59.** Flow characteristics around the entry (top row) and exit (bottom row) tears during two points of the cardiac cycle: (a, e) streamlines at peak systole, (b, f) streamlines at mid-diastole, (c, g) orientation of TAWSS vectors, and (d, h) OSI related to the preferential flow direction. Figure is adapted from Alimohammadi et al. [765] with permission from Elsevier.

determine isolated effects of the tear size and position on hemodynamics. The insights gained from this small virtual cohort appear promising for translation to clinics and patient-specific scenarios. However, it remains unclear to what extent the results of this study hold when further variations in geometry, such as tortuosity, radius, and curvature, or different inflow conditions and pressure levels, are taken into account.

In a similar study, Yu et al. [785] created a patient-specific model of type B dissection. They derived a computational model from CTA data that included an entry tear and three fenestrations. However, they did not distinguish between a dominant distal exit tear and fenestrations. An incompressible, laminar flow of a Newtonian fluid was considered and boundary conditions generated from flow data from the literature. Three additional models were constructed by altering the number, size, and location of tears, namely, (i) occluding two fenestrations, (ii) introducing one additional fenestration, and (iii) increasing the entry tear size compared with the original model. They found that pressure and flow rates were different in the true and false lumina for each of the cases, while occluding the entry tear resulted in stagnant flow in the cut-off portion of the false lumen. Additionally, occluding the original entry tear resulted in flow reversal at the re-entry tears and introduction of additional tears resulted in increased false lumen perfusion. They concluded that CFD modeling may be a useful tool for predicting hemodynamic changes in aortic dissection, providing additional guidance for treatment planning and for non-invasive serial monitoring.

When branch vessels were located near the tears, careful attention to blood supply is crucial. In these cases, multiple overlapping uncovered stents are strategically employed to address the challenge. Dai et al. [786] performed a virtual case study on a CTA-based geometry in which they compared various treatment strategies and evaluated their effectiveness in false lumen thrombosis. Simplified stent geometries were considered in the CFD mesh, with the stents having different mesh densities, resulting in different porosity. Flow rates adopted from the literature were used to drive the flow of an incompressible Newtonian fluid with SST turbulence model. Supraaortic vessels were neglected and the corresponding flow fraction was subtracted from the inflow. This was a reasonable strategy when focusing on the regions distal to the aortic arch. Notably, the use of multiple overlapping uncovered stents and the coverage of branch vessels complicated the application and calibration of Windkessel models. They reported that multiple overlapping of uncovered stents reduced blood flow to the false lumen and resulted in lower WSS. Furthermore, this strategy expanded the true lumen and also stabilized the false lumen, while keeping the blood flow to the branch vessels intact. The combination of standard covered stent-grafts with multiple overlapping uncovered stents where necessary was found to be the most effective option and resulted in the largest

region of stagnant flow. In the present scenario, the blood flow in the latter case was reduced to a small region near the right renal artery, which continued to be supplied. This virtual case study thus highlighted how favorable thrombus formation in the false lumen can be achieved by analyzing the hemodynamics using CFD.

In Xiong et al. [787], CFD was used to investigate the hemodynamic effects of what is known as type I hybrid arch repair in a patient suffering from non-A non-B dissection. This procedure consists of repositioning aortic arch vessels on the healthy ascending aorta in addition to TEVAR. To quantify the effectiveness of this alternative procedure, patient-specific geometries were reconstructed from medical images pre-operatively, one week, one month, and one year after the intervention. In addition, the dissection flap was virtually removed to obtain the pre-dissection geometry. In a longitudinal study, true-to-false lumen pressure difference, WSS, and volumetric flow rates were then compared with normal blood flow. Morphological changes led to the return of hemodynamic indicators to baseline, which indicates the effectiveness of the procedure or remodeling of the aorta. Incompressible flow simulations based on a typical volumetric flow rate and tuned Windkessel models showed that entry tear closure regularized the flow pattern in the true lumen, while the flow field in the false lumen became more complicated and even resulted in retrograde flow during diastole in the proximal false lumen. Of course, covering the entry tear led to a large reduction in the volumetric flow rate in the false lumen, so that the overall flow splits had a strongly positive effect. TAWSS was more evenly distributed in the normal range of 10 to  $50 \, \text{dyne/cm}^2$ s at the one-year follow-up, with less pronounced peak values. Moreover, the region of abnormal WSS in the true lumen progressively decreased, while the WSS in most areas of the false lumen remained below 4 dyne/cm<sup>2</sup>. Elevated pressure and WSS were related to the onset of dissection, whereas low magnitude and large variations of the WSS triggered thrombosis, so the risk of further rupture was considered low. Moreover, the patent false lumen exhibited gradual thrombosis at the one-year follow-up, as suggested by the CFD analysis. Hybrid arch repair resulted in smaller true-to-false lumen pressure differences, such that at the one-year follow-up the pressure between the two lumina was nearly equal. The surgical procedure was therefore verified using CFD analysis, whereby the influence of the specific choice of the inflow profile could have a significant impact on the computational results. The artificial vessels were created very close to the aortic root, so that inlet and leaflet motion and the specific form of inflow profile and temporal scale can strongly influence the flow field. Nevertheless, this work demonstrated how longitudinal case studies on hemodynamics can be used to analyze, compare, and potentially optimize surgical interventions on digital twins.

Li et al. [788] also adopted a similar numerical approach to investigate the effects of intimal tears on hemodynamics in five post-TEVAR aortic dissection cases. Subjects were grouped according to the false lumen dilatation that occurred within the initial post-TEVAR and the first follow-up CTA image. For comparison, additional models were generated based on the first post-TEVAR CTA scans by artificially occluding intimal tears. The vessel wall and dissection flap were assumed to be rigid, and a flat inflow profile along with a single identical flow waveform and Windkessel parameters were adopted from the literature for all models. The individual models, including the first post-TEVAR and corresponding variants that virtually occlude the intimal tears, were then compared based on hemodynamic indicators. The true-to-false lumen pressure difference, the related first point of equal cross-sectional and timeaveraged pressures in true and false lumina, the first balance point, as it is known, the flow split between true and false lumina, and the RRT were calculated. They found that the shift in the first balance point was more pronounced in the group with enlarged false lumen than in the stable lumen group and therefore could serve as a predictor in clinical practice.

Li et al. [789] investigated the potential triggering mechanisms underlying aortic dissection and conducted a CFD study on 15 type B dissection cases by virtually repairing them, i.e. removing the dissection flap. These repaired cases were analyzed using CFD and compared with 12 control patients using geometric features, flow helicity, and WSS-based indicators. They considered laminar flow of an incompressible Newtonian fluid driven by the same flow waveform. Windkessel models were used for each of the outlets, fitting the parameters to the patient's admission blood pressure levels via a tuning algorithm. They reported no significant differences in the geometric properties of the groups, while the healthy control cases showed increased contralateral helical blood flow and helicity. The mean normal transverse WSS was significantly (25%) higher in the virtually repaired type B patient cases, while the locations of primary entry tears matched well with regions of high mean normal transverse WSS. Hence, given limited patient data, CFD analysis could therefore already be a useful tool for early detection of high-risk patients. The strategy used to develop virtually repaired geometries assumed that the pre-dissected cross-section can be restored by removing only the dissection flap without changing the radius. In many cases, such a strategy is useful. However, this procedure required a pre-selection of the dissection cases that corresponded to these assumptions. Therefore, it remains unclear whether the study's conclusions are applicable to cases in which dissection resulted in a significant increase in the volume of the entire vessel.

Wen et al. [790] conducted a comparative study on healthy, dissected, and virtually repaired aortas to identify hemodynamic parameters effective in predicting type B dissection. They considered four indicators based on the WSS, namely the TAWSS, the OSI, the cross flow index (CFI) [791], and the topological shear variation index (TSVI). TAWSS and OSI signal areas of high and high oscillatory shear stress on the vessel wall, while CFI only captures the direction of the WSS but not its magnitude. The TSVI is higher in regions with large variations in endothelial cell contraction and expansion [792,793]. Another three helicity-based indicators were used, namely the locally normalized helicity (LNH), the averaged helicity, and the absolute intensity of the spiral flow [794,795]. CTA images were obtained and corresponding meshes were created for all 16 patients. The flat inflow profile was scaled according to the literature [768], Windkessel models were chosen based on a previous study [796], and blood was modeled as an incompressible, non-Newtonian fluid. The hemodynamic indices between the groups were compared in the descending aortic arch region, where the intimal tear originated in the dissected cases. Statistical analysis showed that OSI and CFI were significantly increased in patients with aortic dissection and that the average helicity also differed significantly between dissected and healthy patient groups. This study suggests that hemodynamic indicators may be more accurate than anatomical characterization in predicting aortic dissection.

The virtual case studies presented have provided valuable insights into the hemodynamics of patients with aortic dissection. This is achieved by virtually altering the number, location, or size of the intimal tears or by removing the dissection flap to obtain a virtual

pre-dissection geometry. This has proven beneficial not only in virtually modeling specific treatment scenarios, but also in gaining a better overall understanding of the complex hemodynamics. Moreover, because pre-dissection geometries are rarely available and estimates of pre-dissection geometry from CT imaging are limited [712], such investigations are invaluable for enhancing our understanding. Longitudinal case studies, discussed below, introduce another level of complexity as they necessitate the creation and simulation of multiple patient-specific geometries over time. Because all geometries are based on patient data, they can yield more reliable results.

#### 6.4. Longitudinal case studies

Cross-sectional case studies cannot prove causality or determine the timing of events. However, virtual and longitudinal case studies that address these important aspects can provide valuable insights. In the following section, we will present longitudinal case studies that follow individual patients or cohorts over time to investigate the effects of specific factors on various outcomes. Longitudinal studies can be either retrospective, involving the collection of data about past events, or prospective, starting with a baseline assessment and tracking individuals over time, with *in silico* longitudinal case studies generally being retrospective to date.

In 2011, Karmonik et al. [797] presented pre- and post-treatment CFD simulations of a type B dissection case, investigating changes in WSS and pressure to compare the initial configuration with a completely remodeled one-year follow-up. They emphasized that to achieve accurate results, patient-specific geometries and measured boundary conditions have to be provided. Then, the relevant quantities such as WSS and true-to-false lumen pressure differences could be accurately captured in both scenarios. In a follow-up study by Karmonik et al. [798], the validation of the proposed framework for patient-specific aortic dissection was carried out based on the clinical examination of a 45-year-old patient, including a ten-month follow-up examination. Axial through-plane velocity PC-MRI data from sites in the thoracic and abdominal aorta were compared with the obtained results and showed good correlation of major flow patterns in true and false lumina. A comparison of hemodynamic parameters obtained by CFD of a healthy aorta and a chronic aortic dissection case in [446] demonstrated the serious effects of false lumen dilatation. In the dissected case, a reduction of 77% in WSS, a 69% reduction in pressure, and an increased true-to-false lumen pressure difference were noted, along with a reduction in WSS by a factor of three.

In related work, Cheng et al. [799] investigated the effect of aortic and entry tear morphology on flow characteristics and corresponding clinical outcomes in four patients with acute type B dissection. Blood flow in the true and false lumina varied drastically among the subjects considered. The individual geometries included the aorta from the arch to the iliac bifurcation and excluded all secondary vessels. Based on these studies, significant morphological parameters were identified. According to the results obtained, the maximum height and width of the entry tear, as well as the distance between the peak of the aortic arch and the upper edge of the tear, were important parameters that governed the flow field. The size and location of the tear dominated the flow rate in the false lumen and have a major impact on the progression of false lumen thrombosis or the onset of dissection. These findings suggested that patients can be divided into groups based on CFD results. Furthermore, the prediction of flow rates and the risk for onset of dissection appears promising.

Xu et al. [774] explored two patient cases with type B dissection treated by TEVAR, denoted as PI and PII and shown in Fig. 60. Only one case showed an enlarged false lumen at follow-up, while the other case remained stable. Patient-specific follow-up models were then developed based on Doppler ultrasound velocimetry and CTA images. Flow rates computed using Doppler ultrasound velocimetry were prescribed assuming flow profiles at the aortic root and supra-aortic vessels. For the remaining outlets, time-dependent mean pressures were adopted from another patient and additionally mapped between the individual stages within the longitudinal case



**Fig. 60.** (a, b) Distribution and variation of normalized RRT to its maximum value in two patients with type B dissection post-TEVAR (PI and PII) at multiple follow-ups. (c) The absolute RRT differences at comparable times are similar. (d) Short- to middle-term differences in the normalized maximum RRT of PI-2 and PII-2, indicating significantly elevated values (above one) for several months in PI. This biomarker could therefore be correlated with positive false lumen remodeling. Figure is adapted from Xu et al. [774] with permission from Elsevier.

study whenever the computational domain had to be shortened due to missing CTA data. It should be noted that the choice of pressure scaling can have a significant impact on the flow split. Choosing neither patient-specific pressure curves nor resistance-based models with tuned parameters (both of which provide the desired flow splits and pressure values) can lead to unpredictable alterations in the results obtained. However, when comparing the computational results and the Doppler ultrasound velocimetry measurements, a difference of less than 5% in the mean velocities over a cardiac cycle was found at selected points. Therefore, the aforementioned possible consequences of imposing pressure values in different patients did not impact the results excessively within this study. However, in the second case, the false lumen was shortened, causing problems with outflow conditions and potentially drastically affecting the results. Temporal and spatial grid convergence analyses and comparisons with a simulation using the k-w SST turbulence model showed that for an average Reynolds number over a cardiac cycle of about 2100 and about 2900 at the ascending aorta assuming laminar flow, a grid with boundary layers and 45 time steps per cardiac cycle resulted in sufficiently small errors compared with the baseline. The flow fields at multiple follow-ups are compared by analyzing the true-to-false lumen pressure differences, RRT, WSS, and OSI. They showed that the true-to-false lumen pressure difference was highly dependent on the size and location of the entry tear, while the largest true-to-false lumen pressure differences correlated with regions of maximum false lumen dilatation. The patient with a patent false lumen still showed a significant true-to-false lumen pressure difference at the final follow-up. Furthermore, TEVAR initially caused an increased true-to-false lumen pressure difference, resulting in dilatation of the false lumen distal to the stent-graft. Thrombus formation in the false lumen was related with TAWSS, OSI, and RRT. The WSS for the patient with patent false lumen remained elevated, while the RRT was higher at the proximal end of the false lumen because the distal ends had additional fenestrations in one case and even an outlet in the second case, as shown in Figs. 60(a) and (b). The RRT distributions differed directly post-TEVAR but are similar at follow-up, leading them to suggest that the difference in RRT at short- to middle-term follow-up could be a potential predictor for false lumen thrombosis, as shown in Fig. 60(d). An increase in RRT after TEVAR signals stagnant flow in the false lumen, a condition that must be maintained for an extended period of time. This was evident from the increasing relative maximum RRT, normalized by the maximum RRT at the first follow-up, required to induce positive false lumen remodeling. In conclusion, this hemodynamic parameter might be clinically applied to predict false lumen thrombosis. This could suggest early re-intervention, long before routinely acquired morphological changes trigger similar measures. However, similar computational studies on larger cohorts are needed to confirm these initial findings and increase the statistical relevance.

Another longitudinal case study was carried out by Shang et al. [763], who performed CFD-based comparisons of uncomplicated type B dissection with cases featuring subacute or chronic false lumen dilatation. Groups of seven patients each were considered, reconstructing patient-specific geometries from CTA data and computing hemodynamic indicators such as flow rates and TAWSS. They found that patients with rapid false lumen dilatation had growth rates of  $5.3 \pm 2.7$  mm/month, while the stable control group had growth rates of  $0.2 \pm 0.02$  mm/month. Although neither the initial aortic diameters, false lumen size, nor peak velocities differed between groups, the false lumen flow rate, entry tear size, shear stress, area of high shear stress, and TAWSS (ignoring the dissection flap) were significantly elevated in the group suffering rapid false lumen dilatation. As shown in Fig. 61, patients with aortic dilatation experienced strong recirculation in the intimal tear region and the proximal false lumen. Despite the greatly simplified boundary data, i.e. prototypical inflow conditions from the literature that are equal for each case and zero pressure boundary conditions, they were able to show a remarkably strong link between the hemodynamic quantities and rapid dilatation in type B dissection cases. Because cases may be advanced at the time of data collection, their application for early detection or even prognosis during the acute phase may be limited. Nonetheless, this study also showed the great potential of CFD in personalized medicine, even if only limited data are available.



**Fig. 61.** (a) False lumen flow rate and maximum TAWSS for type B dissection cases with stable and rapid growth rates. (b) Typical velocity fields in the intimal tear region: (i) uncomplicated type B dissection and (ii) patient case with rapid growth. Figure is adapted from Shang et al. [763] with permission from Elsevier.

The case study by Costache et al. [800] reported on the outcomes of a multilayer flow modulator implantation for treating type B dissection, demonstrating its effects on aortic remodeling and associated hemodynamics. Briefly, this device is an aortic stent with multiple layers and varying porosities. Its aim is to reduce volumetric flow rates and promote vessel reconstruction in the false lumen, rather than completely obstructing the false lumen by blocking entry and exit tears along with additional fenestrations. After treatment, the velocity fields showed immediate laminarization of flow in the false lumen, resulting in complete aortic remodeling. In this case, the false lumen gradually disappeared between 12 and 24 months after treatment, and a stable configuration was reached thereafter, as predicted by the CFD analysis. CFD analysis showed decreasing false lumen flow, supporting the argument. The most striking changes were observed in the volumetric flow rate and laminarization in the false lumen. Quantification of these indicators and detailed analysis of hemodynamic indicators such as TAWSS or true-to-false lumen pressure difference would have further supported their conclusions. However, this work is an excellent example of how computational models can be used in medical device design, demonstrating their effectiveness or optimizing their function. In the following years, this work was further extended in subsequent studies aimed at understanding aortic remodeling and hemodynamics both before and after deployment of such multilayer stents [801,802].

The longitudinal case study by Costache et al. [802] investigates the effectiveness of a multilayer flow modulator by virtually analyzing pre-operative, immediate post-operative, and annual follow-up data from 23 patients over a 36-month period. The treated cases were complicated acute, subacute, and chronic type B dissection, and CFD analysis demonstrated effective relaminarization of aortic flow following a previously defined protocol [800]. The presented study, supported by the CFD results, showed that the implanted devices modulate the flow field, leading to laminar flow and a reduction in peak WSSs. This might promote positive aortic remodeling and false lumen thrombosis, while over-stented vessels remain patent due to locally reduced stent porosity. Additionally, no device migration, kinking, or fracture was observed, making this device a particularly viable option. Regarding the contribution of patient-specific CFD modeling, the numerical results were used to support the quantitative measurements through visualization.

In the study of Pirola et al. [803], the focus is on combining patient-specific geometries with a tuning algorithm to obtain patientspecific boundary conditions that are compatible with pre-operative 4D-flow MRI data and pre- and post-operative Doppler-wire pressure measurements acquired from a type B dissection patient undergoing TEVAR. The medical data were then used further for validation, comparing the predicted flow patterns in a stationary lumen, showing good qualitative agreement of the instantaneous velocity streamlines. In addition, the predicted fluid pressure deviated from the measured data by a maximum of 11 mmHg (9.7%). Furthermore, it was demonstrated that the presented methodology allowed the derivation of post-operative patient-specific boundary conditions for hemodynamics, which further enhanced the predictability of virtual surgery. The drastic impact of patient-specific boundary conditions on overall system behavior – such as flow splits, shear rates and stresses, pressure distributions, regions of recirculatory or stagnant flow, and the related biomarkers derived from them – makes this contribution highly relevant. Adjusting boundary conditions is a costly process, both in terms of time and computational resources. Therefore, tailored algorithms such as the one presented are crucial for real clinical application.

Xu et al. [804] focused on the prognosis of aortic dissection based on CFD simulations and proposed a functional index based on the difference in luminal pressure as an indicator of the hemodynamic status of type B dissection to predict luminal remodeling after surgical intervention. A total of 51 cases of type B dissection were considered, each of which via an initial pre-TEVAR geometry and three follow-ups (at three month, six month, and 12 month). In terms of computational modeling, the blood was considered to be Newtonian and incompressible, imposing flow ratios determined using Doppler ultrasound and pressure waveforms from the literature. Especially for such large study cohorts, collecting more comprehensive data sets for individual patients is challenging, so the limited data per patient is acceptable. Boundary layers were included in each of the models and a temporal and spatial grid-sensitivity analysis for representative cases revealed that the grid and time step sizes employed were appropriate. The main contribution of this work was statistical investigations using a linear mixed-effects model. This analysis showed that the more distal the first balance point, the better the functional status of aortic dissection after TEVAR repair. Patients with the first balance point still present in the abdominal region showed clinically unsatisfactory results. On the other hand, it was shown that cases in which the first balance point was moved from the dissected region after TEVAR showed improved long-term results, meaning reduced false lumen diameter and true lumen expansion. This study thus showed, using a significant number of cases, that (i) the effectiveness of TEVAR could be quantified using CFD by shifting the first balance point distally and (ii) that the same indicator could be used to predict subsequent false lumen dilatation. The lack of a first balance point indicated stable configurations since the true and the false lumen pressures might be instationary, but the sign of the true-to-false lumen difference does not change over time. This did not lead to a change in the hydrodynamic forces on the dissection flaps, which was the crucial aspect in this context.

Armour et al. [805] investigated the influence of the choice of inflow velocity profile and waveform, targeting two cases of type B dissection, the latter including pre- and post-TEVAR configurations. Because patient-specific 4D-flow MRI is not routinely acquired and the need for immediate surgical intervention may make data acquisition impossible, establishing inlet flow profiles based on the limited data available is crucial. This study compares 3D flow profiles reconstructed from 4D-flow MRI with through-plane profiles and flat inlet velocity profiles with non-patient-specific flow rates, the latter mimicking missing data scenarios. The geometries are derived from CTA images and three-element Windkessel models were applied at the outlets. For the first case, the parameters were tuned based on Doppler-wire pressure readings combined with flow splits calculated from the 4D-flow MRI data. For the second case, however, missing patient-specific data had to be artificially supplemented. For the post-TEVAR configuration of the second case, the left subclavian artery occluded by TEVAR was blocked in the CFD model, redirecting flow. Grid convergence studies were performed based on the velocity and TAWSS in selected planes and the computational model was validated with the 4D-flow MRI data, while the changes in the solution and hemodynamic indicators are reported using the 3D flow profile as a reference. Differences in the TAWSS and velocity fields were most distinct in the ascending arch, which was expected but also of great importance for simulations of type A dissection. In type B dissection, the overall flow patterns were relatively independent of the specific inflow profile chosen. However, discrepancies in TAWSS and the true-to-false lumen pressure difference of up to 6% were observed when changing the inlet condition, especially near

the entry tears and additional fenestrations. Regions with low WSS were more affected, especially when the entry tear was near the left subclavian artery. Using non-patient specific waveforms and stroke volume, there was a 30% reduction in WSS and maximum velocity, resulting in a 25% lower flow rate compared with 4D-flow MRI measurements. Furthermore, the chosen waveform had a substantial impact on the predicted pressure. They concluded that 3D, through plane, or flat inlet profiles can be used, resulting in similar flow patterns and TAWSS distributions, while the second yields a better approximation of the true-to-false lumen pressure difference than the flat profile. If no patient-specific inflow data can be obtained, the stroke volume and heart rate, i.e. the volumetric flow rate over the cardiac cycle, must be carefully selected. Due to the changes in TAWSS depending on inflow condition, thrombus formation models that include a TAWSS threshold are also severely affected, so incomplete inflow data can lead to increased uncertainties in predicted thrombus growth. Doppler-wire pressure measurements gave a difference between true and false lumina of 2.3 mmHg, while the CFD simulations resulted in 4.6 mmHg, most likely due to the assumption of a fixed grid and neglected vessel compliance. At the same time, the flat inflow profiles produced errors of up to 6% and the effects of prescribing non-patient-specific flow rates resulted in errors of 25% in terms of this hemodynamic indicator. Therefore, it is likely that both neglect of vessel compliance and the introduction of artificial inflow ratios can lead to drastic errors in the simulation. This effect is expected to be even more relevant when flow in the ascending aorta is of central interest.

The study presented by Zhu et al. [806] focused on a comparison of geometrical and hemodynamic parameters in a total of 17 cases of type A dissection. Using a false luminal growth rate of 2.9 mm/year as the threshold, nine cases were categorized as stable, while the remaining eight cases were categorized as unstable. Each patient's pathology was assessed with CTA imaging and further processed to allow for CFD analysis. Multiple follow-up CTA image sets were obtained for each patient, allowing the identification of potential factors indicative of dilatation risk based on the false luminal growth rate in vivo. The entry tear and possible fenestrations were identified along with the radii of the true and false lumina. Doppler ECG was employed to determine the maximum flow velocity, which was combined with patient-specific heart rates and physiological flow transients to prescribe suitable volumetric flow rates at the aortic root. Three-element Windkessel models were optimized assuming that 21% of the volumetric flow exits through the arch vessels and the flow ratios depend linearly on the outlet surface areas. Despite the limited sample size, this study had significant implications due to the following four conclusions. First, the stable and unstable configurations of type A dissection did not significantly affect either the size of the entry tear or the volume ratio between the true and false lumina. They argue that this is due to the complex geometries and false lumen flow rate and pressure, which are not solely dependent on tear size and volume ratio. Second, the stable cases had a larger number of fenestrations. Third, the stable cases showed a more even flow distribution between true and false lumina, while the true lumen velocity was increased in the unstable cases. Therefore, more even TAWSS distributions were seen in the stable cases, whereas the unstable aortic dissections showed increased heterogeneity and maximal WSS. In all cases of stable and unstable aortic dissections, recirculatory and helical flow patterns as well as strong flow separation were observed. Finally, in conjunction with the lower number of fenestrations, the true-to-false lumen pressure difference increased in the unstable cases, reaching 6.7 mmHg, while in the stable cases they were 0.9 mmHg. Therefore, the geometrical analysis along with hemodynamic investigations following replacement of the ascending aorta seem promising for predicting late false lumen dilatation.

In a follow-up work, Zhu et al. [807] then focused on the true-to-false lumen pressure difference as the main hemodynamic indicator for predicting increasing aortic diameters, as shown for a selected patient in Fig. 62. Maintaining the numerical setup unchanged from [806], longitudinal case studies were performed using CFD analysis on four cases after type A dissection repair, tracking



**Fig. 62.** Patient's aortic diameter changes at the maximum diameter slice, along with true-to-false lumen pressure difference variations during a cardiac cycle. The aortic growth rate (mm/year) is evaluated and compared at each follow-up scan, along with the maximum luminal pressure difference over a cardiac cycle. Both the true and false lumina have outlets where it is difficult to obtain flow data. Imposing boundary conditions inevitably affects the results and causes the pressure difference and flow rates to be influenced by the enforced flow rates or pressures chosen. In clinical practice, such situations are often unavoidable but create a modeling conundrum. The presence of associated true and false lumina outlets or extensive dissection flap fenestration would mitigate this effect. Figure is reprinted from Zhu et al. [807] (licensed under CC-BY 4.0).

the true-to-false lumen pressure difference and maximum aortic diameter in 12 cross-sections along the aorta. The resulting data were then assessed using a linear mixed-effects model to identify the statistical relevance of an increased true-to-false lumen pressure difference. They showed that the true-to-false lumen pressure difference can predict unstable growth (above 2.9mm/year) in the current, relatively small, and heterogeneous patient group. Using a receiver operating characteristic curve, values of >5 mmHg true-tofalse lumen pressure difference were identified as promoting unstable aortic growth. They noted that the small sample size of only four patients and the highly different medical histories among them - including one patient with mechanical aortic valve replacement, another with Marfan syndrome and hypertension, a third with a bicuspid aortic valve, and a fourth with unstable aortic growth - are limitations of this study. Classifying patients into groups according to their medical history in larger cohorts is required, but the strategy used has high potential for clinical application. As far as computational modeling is concerned, limiting CFD to a fixed grid can have drastic effects on the pressure difference, as a highly mobile dissection flap can significantly reduce the true-to-false lumen pressure difference. In addition, the three-element Windkessel models were not fitted to patient data but were selected based on the approach described in [806]. Specifically, the maximum velocity and heart rate were measured and then combined with an assumed typical flow waveform and a flat inflow profile to obtain a realistic, partially patient-specific inflow condition. This has a direct impact on the absolute pressure values, while the effect on the true-to-false lumen pressure difference may be smaller. The true-to-false lumen pressure difference could actually be influenced by the number and location of the Windkessel outlets relative to the fenestrations and the false lumen geometry. In this present study, the computational domain was shortened so that the false lumen has a Windkessel boundary, as shown in Fig. 62, which leads to a drastic influence of the Windkessel parameters on the false lumen pressure, and hence the true-to-false lumen pressure difference. Consequently, obtaining credible boundary data is particularly difficult. However, such configurations can not be avoided in clinical practice, which motivated their discussion. Despite these limitations, promising results were presented, highlighting that luminal pressure was a particularly relevant hemodynamic indicator that could be used in clinical practice in the future.

Parker et al. [808] investigated partial false lumen thrombosis and quantified the relation between hemodynamic pressure, thrombus morphology, and clinical outcome in type B dissection. CTA data were collected from 69 patients, each correspondingly divided into groups based on thrombus morphology. The true-to-false lumen pressure difference was determined, a survival analysis for adverse aortic events at one year was carried out, and a morphology classification were performed. This study showed that patients with proximal thrombosis had a 10.1 mmHg decreased false lumen pressure compared with patients with minimal thrombosis. This reduces the risk of acute complications, resulting in fewer late adverse aortic events within a year. Note again that they observed no differences in WSS, low and oscillatory shear, or OSI between thrombosis groups. They identified proximal false lumen thrombosis at initial presentation as the 'strongest clinical predictor of both reduced perfusion pressure in the false lumen and favorable short-term and one-year outcomes'. However, it remains unclear whether the proximal thrombosis results from an overall favorable condition, thus being merely a positive side effect, or whether it initiates a cascade of beneficial events.

Armour et al. [809] investigated the role of multiple fenestrations in type B dissection in a longitudinal case study. For this purpose, a controlled swine model was analyzed using CT imaging and 4D-flow MRI as well as computational modeling to assess aortic hemodynamics. In a previous study by Guo et al. [607], as detailed in Section 4.2, a type B dissection swine model with a distal tear 7 cm beyond the left subclavian artery was developed. In the current study, a re-intervention was carried out to introduce two additional fenestrations, one of which was nearly occluded. This resulted in three configurations: one with a single natural exit tear, one with a natural exit tear and one additional fenestration, and one with a natural exit tear and two additional fenestrations. Image and flow data were collected before and after the intervention, including multiple follow-ups. 3D inlet profiles were constructed from the 4D-flow MRI data and Windkessel models were fitted to flow rates or Doppler-wire pressure measurements when available. Instationary flow simulations were performed to collect flow fields, reverse flow index, TAWSS, and pressures in the true and false lumina. In the combined study, false luminal growth stabilized with the introduction of additional fenestrations, and false lumen by 22%. This, of course, altered the observed velocities; TAWSS was reduced at the tears, and the pressure difference between the true and false lumina also decreased. They showed that the introduction of additional fenestrations could be beneficial when the pressure in the false lumina is high and the false lumen was fully perfused. This strategy could be used to stabilize false lumen dilatation, making dissection flap fenestration a key aspect in terms of hemodynamics and control of false lumen dilatation.

Another longitudinal case study using CFD simulation techniques was conducted by Fatma et al. [810]. The hemodynamics of two cases with residual type B dissection following surgical repair of type A dissection were explored. Only one of the two considered patient-specific geometries showed a favorable evolution, potentially allowing the identification of hemodynamic features that lead to further disease progression over time. The aortic wall and dissection flap were assumed to be rigid, blood flow was considered laminar based on the maximum observed Reynolds number and the minimum critical Reynolds number, a generalized Newtonian fluid was considered via the Carreau-Yasuda law, and the Windkessel parameters for regulating flow rates at the outlets were fitted to reproduce the flow rates and pressure as under hypotensive treatment. It should be noted that these parameters were not adjusted to each specific patient's data. A mesh sensitivity study was also carried out and further assumptions were introduced. First, identical volumetric flow rates from the literature were prescribed for both patients. This is a common obstacle in scenarios where flow data are not available, but can potentially alter the observed hemodynamics, particularly in the cases presented where the radii of the ascending aortas differ by 15%. Second, and more strikingly, the computational domain for the favorable outcome case was shortened by trimming the true and false lumina. If volumetric flow data were available for true and false lumina, one might fit the Windkessel parameters for these outlets as well, but the present study appears to avoid this complication by combining the true and false lumina into a single lumen where relative flow rates and pressure levels are easier to determine. This chosen strategy, which neglected the computational domain at the celiac trunk and combines the true and false lumina, could have a dramatic impact on the results obtained and once again

highlights the exceptionally challenging aspects of modeling aortic dissection. Nevertheless, they identified three intimal tears when analyzing the two cases and the respective follow-up examinations. The case with a subsequent unfavorable outcome had larger tear diameters, which increased by 72 and 59%, respectively, during follow-up. The large entry tear was located in the descending aorta. In contrast, the case that resulted in a favorable outcome featured a smaller tear just distal to the implanted Dacron prosthesis that decreased in area by 41% over time. Overall, significant remodeling was observed in both cases, leading to a pronounced volume increase (27%) in the unfavorable case with partial thrombosis, compared with a 19% volume increase in the favorable case. Using CFD, Fatma et al. [810] then identified four possible markers of late complications, namely (i) a sustained high volumetric flow ratio into the patent false lumen above 50%; (ii) a high WSS at the entry tear (>18Pa) and on the false lumen wall (regional >10Pa), caused by the flow jet through the entry tear hitting the false lumen wall and creating vortex structures; (iii) low TAWSS into the distal false lumen (<0.5 Pa); and (iv) pronounced helical flow patterns and recirculation. Interestingly, they report that a comparison of the vortex structures in the two cases presented is much more pronounced than the differences in the areas showing combined low TAWSS and high OSI values. This observation motivates further investigation on the vortex structures within the false lumen, as these could also lead to promising hemodynamic markers in aortic dissection.

Osswald et al. [811] investigated the relationships between WSS and the development of new entry tears caused by distal stentgrafts using what is known as the frozen elephant trunk procedure, see Section 2.3.2. It combines replacement of the aortic arch with a prosthesis, while dissection of the descending aorta is treated with a stent-graft. Five patients with diagnosed with dSINE were retrospectively analyzed using CFD both pre- and post-complication. They examined hemodynamic alterations with a focus on WSSs in CTA image-based geometries. To reduce the computational effort, steady-state simulations of the geometries were carried out, assuming a laminar incompressible flow of a Newtonian fluid, a constant inflow taking 0.4 m/s as the average flow velocity, and an outlet pressure of zero. This avoids the time-consuming tuning procedures related with Windkessel models and fills in missing data with only a few, but fairly strong assumptions. Despite these assumptions, they were able to detect a significantly increased average WSS within the stent-graft and in its landing zone, while only a small increase was observed distal to the stent-graft. Additionally, the WSS was found to be higher at the convexity of the aorta than at the concavity, which is due to the redirection of flow and the creation of Dean vortices. They concluded that elevated WSSs after stent-graft deployment may contribute to weakening of the aortic wall and potentially trigger aortic rupture. However, enforcing zero pressure outflow conditions could be viewed as an over-simplification since setting stationary outlet pressures (relative to the outlet area or flow splits) is just as simple as assuming zero pressure, resulting in a more realistic flow split but requiring further pre-processing. Nonetheless, this study showed that CFD studies can be employed to identify cases susceptible to developing new entry tears caused by distal stent-graft and that WSSs could be used as an indicator even with these simplifications.

In summary, these findings suggest that CFD simulations of aortic dissection can indeed support clinical decision-making in both cross-sectional and longitudinal case studies as well as in virtual case studies. However, predictions about the stress distribution within the vessel wall and thus the risk of rupture remain unattainable. The flow field can actually be recovered in silico and provides important insights into connections between topological features such as location, number, and size of tears, resulting flow split between true and false lumina, pressure gradients between in- and outlets and through the dissection flaps, luminal pressure and false lumen dilatation as well as the effect of flow jets on local pressure. These quantities can further be related to allow virtual comparison of different surgical treatment options. CFD alone can therefore provide valuable insights, but depends heavily on the patient data supplied. Therefore, the clinical application of CFD must combine powerful numerical schemes and parameter fitting methods to recover realistic flow splits between outlets. A second fundamental assumption concerns the elasticity of the tissue. In contrast to the healthy case, where neglecting any deformation in the vessel might be acceptable given the vessel does not experience large motions, in aortic dissection the dissection flap could be generally mobile, at least in the acute and subacute phases. Such effects can only be taken into account by using FSI, which drastically increases the numerical effort. From this perspective, CFD modeling still has its place in aortic dissection due to the reduced number of required parameters, simpler discretization in terms of mesh construction and solver design, and, most importantly, shorter run times, which are crucial in clinical applications. Depending on the specific case, fixed-grid flow simulations can be an adequate tool, while in the case of finite deformations or if the stresses in the tissue are to be examined, the coupling of fluid and structure must be taken into account.

# 6.5. Key findings and publication index

To summarize this section, we present the publication index of hemodynamic *in silico* models in Table 7, with a focus on CFD simulations of aortic dissection using both simplified and patient-specific geometries. Due to the large number of studies available, only a selected range was discussed; however, this table covers an even broader range of studies. Essential information about the modeling approaches and patient cases as well as the objectives of each study is included. Finally, the most important findings are listed conclusively.

- Depending on the pathology in question, modeling via fixed-grid CFD may be possible, while other configurations may require the use of more advanced FSI schemes. For healthy aortas with little radial motion, consideration of the coupled problem may not be essential depending on the aim of the study and available resources.
- Studies have shown a strong correlation between *in silico* models and *in vivo* or *in vitro* data. However, these studies also highlight the sensitivity of the results to the *in silico* model itself. More specifically, CFD simulations require several advanced methods to handle the immense complexity of this application, including: (i) a grid resolving the domain and the solution accurately, (ii)

#### Table 7

Publication index of *in silico* hemodynamic models related to aortic dissection, ordered by publication year (FL: false lumen; Lam.: laminar; Turb.: turbulent).

| turbulent):                     |         |                     |                    |                    |                  |  |
|---------------------------------|---------|---------------------|--------------------|--------------------|------------------|--|
| Report                          | Year    | Solver              | Flow<br>assumption | No. of<br>patients | Туре             | Main objective(s)  |
| Simplified studies              |         |                     |                    |                    |                  |  |
|                                 | 0000    | A                   | T                  |                    |                  |  |
| Guan et al. [750]               | 2009    | Ansys               | Lam.               | -                  | Туре в           | Comparison of nemodynamics in simplified models with/          |
|                                 |         |                     |                    |                    |                  | without bypassing treatment (from left subclavian artery       |
|                                 |         |                     |                    |                    |                  | to abdominal aorta)  |
| Fan et al. [749]                | 2010    | Ansvs <sup>a</sup>  | Lam.               | _                  | _                | Impact of intimal tear (location, size)/luminal area ratio     |
|                                 |         | ) -                 |                    |                    |                  | on hemodynamics in simplified models                           |
|                                 |         |                     |                    |                    |                  | on nemodynamics in simplified models                           |
| Chitsaz et al. [751]            | 2012    | OpenFOAM            | Lam.               | -                  | -                | Impact of intimal tears (number) on tearing force in           |
|                                 |         |                     |                    |                    |                  | longitudinal direction/( $dp/dt$ ) in symmetric 2D tubes       |
| Tang et al. [752]               | 2012    | Ansvs <sup>a</sup>  | Turb.              | _                  | Type B           | Impact of intimal tear (number, location)/FL diameter on       |
| 0                               |         | - 5 -               |                    |                    | J I -            | force acting on the wall in simplified dissection models       |
| Durdeniels et al. [E16 747 740] | 2015/17 | Matlah              |                    |                    |                  | Immed dissection inducts                                       |
| Rudellick et al. [510,747,746]  | 2015/17 | Mattab              | 11.d <b>.</b>      | -                  | -                | inipact of wan summess/mumar tear (size, iocation) on          |
|                                 |         |                     |                    |                    |                  | hemodynamics with lumped-parameter model;                      |
|                                 |         |                     |                    |                    |                  | validation to in vitro experiments                             |
| Ben Ahmed et al. [753]          | 2016    | CRIMSON             | Lam.               | _                  | _                | Impact of intimal tear (number, size, location)/vessel         |
| ben finned et an [/ ob]         | 2010    | Grandoort           | Luin               |                    |                  | autivature on homodynamics in simplified models                |
|                                 |         |                     |                    |                    |                  | curvature on hemodynamics in simplified models                 |
| Zadrazil et al. [630]           | 2020    | Ansys"              | Turb.              | -                  | Туре В           | Validation of hemodynamics to in vitro experiments             |
| Zorilla et al. [754]            | 2020    | Kratos <sup>d</sup> | n.a.               | -                  | Type B           | Validation of hemodynamics to in vitro experiments             |
| Li et al. [757,758]             | 2021    | Ansvs <sup>a</sup>  | Lam.               | _                  | Type B           | Morphological parameters                                       |
| Oine et al [756]                | 2022    | Anovo               | Lom                |                    | Tupo P           | Import of stort graft introducer shouth during TEVAD on        |
|                                 | 2022    | Allsys              | Lalli,             | -                  | туре в           | impact of stellt-graft infoducer sheath during TEVAK on        |
|                                 |         |                     |                    |                    |                  | hemodynamics in simplified aneurysm/coarctation/               |
|                                 |         |                     |                    |                    |                  | dissection models  |
| Peng et al. [755]               | 2022    | Ansvs <sup>a</sup>  | Lam.               | _                  | _                | Impact of morphological parameters on hemodynamics             |
|                                 | 2022    | 111090              | 20110              |                    |                  | in simplified healthy model                                    |
|                                 |         |                     |                    |                    |                  | in simplified healthy model                                    |
| Cross-sectional case studies    |         |                     |                    |                    |                  |  |
| Karmonik et al. [812]           | 2008    | Ansys <sup>a</sup>  | n.a.               | 3                  | Multiple         | Comparison of hemodynamics in patients with mobile             |
|                                 |         |                     |                    |                    |                  | thrombus in aortic arch/acute type II/abdominal aortic         |
|                                 |         |                     |                    |                    |                  | onouriem ronair  |
|                                 |         | . b                 | - 1                | -                  |                  |  |
| Cheng et al. [759]              | 2010    | Ansys               | Turb.              | 1                  | Туре В           | Hemodynamic simulations with hybrid SST k- $\epsilon/k-\omega$ |
|                                 |         |                     |                    |                    |                  | turbulence model   |
| Chen et al. [767]               | 2013    | CFD-ACE+            | Turb.              | 1                  | Type B           | Hemodynamic simulations with SST k- $\omega$ turbulence        |
|                                 |         |                     |                    |                    | J I -            | model  |
|                                 | 0014    | • b                 | m 1                |                    |                  |  |
| Cheng et al. [760]              | 2014    | Ansys-              | Turb.              | 1                  | туре в           | validation of nemodynamics to in vivo data                     |
| D'Ancona et al. [813]           | 2014    | n.a.                | n.a.               | 4                  | Туре В           | Hemodynamics in patients                                       |
| Djorovic et al. [762]           | 2015    | PAK-F               | Lam.               | 2                  | Type A           | Hemodynamics in patients                                       |
| Cheng et al [761]               | 2015    | Ansys <sup>b</sup>  | Turb               | 8                  | Type B           | Comparison of hemodynamics/morphology in patients              |
| Glieng et al. [/ 01]            | 2010    | 7111595             | Turb.              | 0                  | Type D           | with an direct methodynamics/morphology in patients            |
|                                 |         |                     | _                  |                    |                  | with medical treatment/TEVAR                                   |
| Wan Ab Naim et al. [814]        | 2016    | Ansys <sup>a</sup>  | Lam.               | 1                  | Туре В           | Impact of intimal tear (number) on hemodynamics in             |
|                                 |         |                     |                    |                    |                  | patients   |
| Osswald et al. [815]            | 2017    | Star-CCM+           | Lam.               | 20                 | Type B           | Comparison of hemodynamics in patients with/without            |
|                                 |         |                     |                    |                    | - <i>J</i> P = - | subsequent retrograde type A dissection                        |
| 1 VI - 1 FO1 (1                 | 0015    | • b                 | m 1                |                    |                  | Subsequent renograde type A dissection                         |
| Long Ko et al. [816]            | 2017    | Ansys               | Turb.              | 3                  | Туре В           | Hemodynamic simulations with SST $k \in turbulence$            |
|                                 |         |                     |                    |                    |                  | model  |
| Iida et al. [817]               | 2017    | n.a.                | n.a.               | 1                  | Type B           | Hemodynamics in patient with ulcer-like projection             |
| Chietal [818]                   | 2017    | Answea              | Turb               | 7                  | Type A           | Hemodynamics in patients (pre-dissection)                      |
|                                 | 2017    | Ab                  | Turb.              | 1                  | Турс Л           | Development of all with a few all hasting of flow              |
| Bonfanti et al. [770]           | 2019    | Ansys-              | Lam.               | 1                  | туре в           | Development of algorithm for calibration of flow               |
|                                 |         |                     |                    |                    |                  | boundary conditions  |
| Peng et al. [819]               | 2019    | Ansys <sup>a</sup>  | n.a.               | 1                  | -                | Hemodynamics in patient with interrupted aortic arch           |
| 0                               |         |                     |                    |                    |                  | (implications for dissection)                                  |
| Renfonti et al [625]            | 2020    | Americab            | Treat              | 1                  | True D           | Validation to in vitro evacuimente (in vive date               |
| bollialiti et al. [035]         | 2020    | Allsys              | Turb.              | 1                  | туре в           | validation to <i>in vitro</i> experiments/ in vivo data        |
| Tomasi et al. [820]             | 2020    | Ansys               | Turb.              | 2                  | Туре В           | Comparison between CFD and reduced-order models in             |
|                                 |         |                     |                    |                    |                  | healthy/dissection patients                                    |
| Li et al. [821]                 | 2020    | Ansvs <sup>a</sup>  | na                 | 120                | Type A           | Comparison of hemodynamics in patients with different          |
|                                 |         | ) -                 |                    |                    | - <i>J</i> P =   | provimal to distal toor size ratios                            |
|                                 |         |                     | -                  |                    |                  | proximal-to-distance is size ratios                            |
| Qiao et al. [822]               | 2020    | Ansys               | Lam.               | 2                  | Туре В           | Hemodynamics in patients with TEVAR/in situ double             |
|                                 |         |                     |                    |                    |                  | fenestrations of left carotid and subclavian artery post-      |
|                                 |         |                     |                    |                    |                  | operation  |
| Marrocco-Trischitta et al [823] | 2021    | LifeV               | na                 | 5                  |                  | Hemodynamics in natients with type I/II/II arch                |
|                                 | 2021    | LICY                |                    | 5                  |                  | (implications for dissoction)                                  |
|                                 |         |                     |                    |                    |                  | (implications for dissection)                                  |
| Li et al. [824]                 | 2021    | CFD-ACE+            | Lam.               | 2                  | Type B           | Impact of inlet velocity profiles on validation to in vivo     |
|                                 |         |                     |                    |                    |                  | data in healthy/dissection patients                            |
| Hohri et al. [825]              | 2021    | Ansys               | Turb               | 6                  | _                | Comparison of hemodynamics in healthy/pre-dissection           |
| Little of an [ono]              |         |                     |                    | 5                  |                  | notionto   |
|                                 |         | . b                 |                    | -                  |                  | patients   |
| Abazari et al. [764]            | 2021    | Ansys               | Lam.               | 1                  | Type B           | Impact of beta-blocker medication on hemodynamics              |
| Armour et al. [805]             | 2021    | Ansys <sup>b</sup>  | Lam.               | 2                  | Type B           | Impact of inlet velocity profile on hemodynamics on            |
|                                 |         | -                   |                    |                    |                  | validation to in vivo data                                     |
| Armour et al [773]              | 2022    | Ansys <sup>b</sup>  | Lam                | 5                  | Type B           | Validation of hemodynamics to in vivo data                     |
|                                 | 2022    | 1 113y 5            |                    | 5                  | турсы            | valuation of nemouynamics to in nivo data                      |

# Table 7 (continued)

| Report                        | Year | Solver               | Flow<br>assumption | No. of patients | Туре        | Main objective(s)  |
|-------------------------------|------|----------------------|--------------------|-----------------|-------------|--|
| Wen et al. [826]              | 2022 | Ansys <sup>a</sup>   | n.a.               | 30              | Healthy     | Impact of arch type I/II/III on hemodynamics   |
| Kimura et al. [775]           | 2022 | scFLOW               | n.a.               | 19              | Type B &    | (implications for dissection)<br>Comparison of hemodynamics/morphology in type B/  |
| Takeda et al. [827]           | 2022 | OpenFOAM             | Turb.              | 11              | Type A & B  | Comparison of hemodynamic simulations with SST k- $\in$<br>turbulence model in healthy/type A/type B patients  |
| Wang et al. [772]             | 2022 | OpenFOAM             | Lam.               | 1               | Type A      | Validation of hemodynamics to <i>in vivo</i> data post-  |
| Moretti et al. [776]          | 2023 | OpenFOAM             | Turb.              | 4               | Multiple    | Operation with residual dissection<br>Comparison of hemodynamics in healthy patient and<br>patients with patent/partially thrombosed/completely<br>thrombosed El |
| Stokes et al. [777]           | 2023 | Ansys <sup>b</sup>   | Turb.              | 1               | Туре В      | Impact of different <i>in vivo</i> data for inlet conditions on<br>hemodynamics in aneurysmal dissection   |
| Black et al. [779]            | 2023 | Ansys <sup>a</sup>   | Turb.              | 1               | Туре В      | Development of reduced-order framework to calibrate<br>Windkessel model parameters with <i>in vivo</i> data in<br>healthy/dissection patients                    |
| Chatpattanasiri et al. [637]  | 2023 | Ansys <sup>b</sup>   | Turb.              | 1               | Туре В      | Validation of hemodynamics in reduced-order model to<br>in vitro data  |
| Tsai et al. [828]             | 2024 | n.a.                 | n.a.               | 1               | Туре А      | Hemodynamics in patient with iatrogenic aortic dissection  |
| Messou et al. [829]           | 2024 | SimVascular          | n.a.               | 3               | Туре В      | Comparison of hemodynamics in patients with FL<br>thrombosis, a large fenestration, and the orbital  |
| Wang et al. [830]             | 2024 | Ansys <sup>b</sup>   | Lam.               | 5               | Туре В      | Impact of different <i>in vivo</i> data (only ECG, ECG with<br>stroke volume, 4D-flow MRI) for inlet conditions on<br>hemodynamics                               |
| Zorilla & Soudah [831]        | 2024 | Kratos <sup>d</sup>  | n.a.               | 4               | Туре В      | Development of two-step segmentation procedure for<br>thin-walled flap; hemodynamics in patients   |
| Wang et al. [832]             | 2024 | Ansys <sup>b</sup>   | n.a.               | 5               | Туре В      | Development of novel framework to tune inlet boundary<br>conditions to <i>in vivo</i> data in dissection patients (post-<br>TEVAR)                               |
| Virtual case studies          | 2011 | A marva <sup>a</sup> | -                  | 1               | True P      | Lange of signally coluded toos (someword flor or   |
| Karmonik et al. [780]         | 2011 | Allsys               | II.a.              | 1               | туре в      | hemodynamics   |
| Chen et al. [781]             | 2013 | CFD-ACE+             | Lam.               | 1               | Туре В      | Comparison of hemodynamics in patient by virtually occluding different tears (pre-/post-medical treatment)   |
| Alimohammadi et al. [765]     | 2014 | Ansys <sup>b</sup>   | Lam.               | 1               | Туре В      | Impact of virtually occluded tears on hemodynamics in<br>patients with coarctation   |
| Alimohammadi et al. [766]     | 2014 | Ansys <sup>b</sup>   | Lam.               | 1               | Туре В      | Impact of virtually occluded tears on hemodynamics in patients   |
| Wan Ab Naim et al. [782]      | 2014 | Ansys <sup>a</sup>   | Lam.               | 1               | Type B      | Impact of virtually added tears on hemodynamics  |
| Dillon-Murphy et al. [783]    | 2016 | CRIMSON              | n.a.               | 1               | Туре В      | Comparison of virtually added tears/occluded tears/<br>removed flap on hemodynamics  |
| Shi et al. [784]              | 2016 | Ansys <sup>a</sup>   | Lam.               | 4               | Туре А      | Impact of virtually altered tears (size, location, number)<br>on hemodynamics  |
| Yu et al. [785]               | 2016 | Ansys <sup>a</sup>   | n.a.               | 4               | Туре В      | Impact of virtually altered tears (size, number) on hemodynamics   |
| Rikhtegar Nezami et al. [833] | 2018 | Ansys <sup>b</sup>   | n.a.               | 1               | Туре В      | Impact of multilayer flow modulator on end-organ<br>perfusion (pre-/post-operation and virtual pre-  |
| Jiang et al. [834]            | 2019 | Ansys <sup>a</sup>   | Lam.               | 1               | Туре В      | Comparison of hemodynamics by virtually occluding<br>different branch arteries (nost-TEVAR)  |
| Qiao et al. [835]             | 2019 | Ansys                | Lam.               | 1               | Туре В      | Comparison of hemodynamics between occluded (pre-<br>TEVAR)/virtually added left subclavian artery (post-  |
| Dai et al. [786]              | 2020 | Ansys <sup>2</sup>   | Turb.              | 1               | Туре В      | TEVAR)<br>Comparison of hemodynamics in patient with multiple<br>overlapping uncovered stents by virtually occluded/<br>partially occluded tears                 |
| Qiu et al. [836]              | 2020 | Ansys <sup>a</sup>   | Lam.               | 1               | Туре В      | Comparison of hemodynamics post-TEVAR with virtual<br>pre-TEVAR /virtually removing graft material   |
| Xiong et al. [787]            | 2020 | Ansys <sup>a</sup>   | n.a.               | 1               | Non-A non-B | Comparison of hemodynamics pre-/post-hybrid arch   |
| Polanczyk et al. [837]        | 2021 | Ansys <sup>a</sup>   | Lam.               | 12              | Туре В      | Comparison of hemodynamics/hematocrit value pre-/<br>nost.TEVAR and by virtually removed flap  |
| Sengupta et al. [838]         | 2022 | Ansys <sup>b</sup>   | Turb.              | 2               | Туре А      | Comparison of hemodynamics post-TEVAR (double-<br>branched endograft in aortic arch) by virtually altering<br>tunnel branch diameters                            |

## Table 7 (continued)

| Report  | Year    | Solver             | Flow<br>assumption | No. of patients | Туре   | Main objective(s)   |
|---|---------|--------------------|--------------------|-----------------|--------|---|
| Li et al. [788]                               | 2022    | Ansys <sup>b</sup> | n.a.               | 5               | Туре В | Comparison of hemodynamics in pre-TEVAR/post-   |
| Li et al. [839]                               | 2022    | Ansys <sup>a</sup> | n.a.               | 2               | Туре В | Comparison of hemodynamics in different bypass<br>strategies (left subclavian artery) of healthy patients and   |
| Stokes et al. [840]                           | 2023    | Ansys <sup>b</sup> | Turb.              | 1               | Туре В | patients with/without dissection<br>Validation of hemodynamics with minor/major branch<br>arteries and virtually occluded minor branch vessels to <i>in</i> |
| Li et al. [789]                               | 2023    | Ansys <sup>a</sup> | Lam.               | 27              | Туре В | vivo data<br>Comparison of hemodynamics between in healthy/<br>dissection patients and virtually removed flap   |
| Motoki et al. [841]                           | 2023    | Ansys <sup>b</sup> | Turb.              | 2               | Type A | Comparison of hemodynamics by virtually altered tears<br>(size, location.number) (post-operation)   |
| Wen et al. [790]                              | 2023    | Ansys <sup>a</sup> | n.a.               | 16              | Туре В | Comparison of hemodynamics/morphology in healthy/<br>dissection/virtual pre-dissection patients   |
| Zhu et al. [842]                              | 2023    | Ansys <sup>b</sup> | Turb.              | 1               | Type A | Comparison of hemodynamics in patients with frozen<br>elephant trunk procedure (virtual) (pre-/post-operation);<br>validation to <i>in vivo</i> data        |
| Chen et al. [843]                             | 2023    | n.a.               | n.a.               | 1               | Type B | Comparison of hemodynamics in double/single (virtual)<br>FL abdominal aortic dissection aneurysm, and saccular  |
| Liu et al. [844]                              | 2024    | Ansys <sup>a</sup> | n.a.               | 1               | Туре В | Comparison of hemodynamics by virtually altering tears<br>(number) (post-dissection) and virtual pre-dissection   |
| Girardi et al. [845]                          | 2024    | Ansys <sup>a</sup> | Turb.              | 1               | Туре В | Comparison of hemodynamics in three different grafting strategies (pre-/post-operation)   |
| Longitudinal case studies<br>Tse et al. [846] | 2011    | ADINA              | Lam.               | 1               | Туре В | Hemodynamics in patients with aortic valve/proximal ascending aortic replacement (pre-/post-operation)  |
| Karmonik et al. [797]                         | 2011    | Ansvs <sup>a</sup> | n.a.               | 1               | Type B | Hemodynamics in patients (pre-/post-operation)  |
| Karmonik et al. [446,798]                     | 2012/13 | Ansys <sup>a</sup> | n.a.               | 1               | Туре В | Comparison of hemodynamics in healthy patients and in<br>patients with FL dilatation  |
| Cheng et al. [799]                            | 2013    | Ansys <sup>b</sup> | Turb.              | 5               | Туре В | Comparison of hemodynamics/tear/FL diameter in<br>uncomplicated (complicated cases  |
| Mori et al. [847]                             | 2013    | Ansys <sup>a</sup> | Lam.               | 2               | Туре В | Hemodynamics in patient with ulcer-like projections in completely thrombosed FI   |
| Rinaudo et al. [848]                          | 2014    | Ansys <sup>a</sup> | Lam.               | 25              | Туре В | Impact of tear/luminal area/hemodynamics in patients  |
| Shang et al. [763]                            | 2015    | Abaqus/CFD         | n.a.               | 14              | Туре В | Comparison of hemodynamics/growth rates in patients   |
| Stefanov et al. [849]                         | 2017    | Ansys <sup>b</sup> | Lam.               | 12              | Туре В | Comparison of morphology in follow-up and impact of   |
| Xu et al. [774]                               | 2017    | CFD-ACE+           | Turb.              | 2               | Туре В | Streaminer multilayer now modulator on nemodynamics<br>Comparison of hemodynamics/tear in patients with<br>stable FL/FL dilatation (post-TEVAR)             |
| Xu et al. [850]                               | 2018    | LifeV              | Lam.               | 1               | Туре В | Comparison of hemodynamics/morphology in patients   |
| Wan Ab Naim et al. [851]                      | 2018    | Ansys <sup>a</sup> | Turb.              | 5               | Туре В | Comparison of hemodynamics/morphology in patients   |
| Costache et al. [800]                         | 2018    | Ansys              | n.a                | 1               | Туре В | with completely/partially thrombosed FL (post-TEVAR)<br>Comparison of hemodynamics in patients with multilayer  |
| Pirola et al [803]                            | 2019    | Ansys <sup>b</sup> | na                 | 1               | Type B | flow modulator (pre-/post-operation)  |
|   | 2010    | A                  |                    | 1               | Type D | TEVAR)  |
| Autanasiou et al. [852]                       | 2019    | Allsys             | II.a.              | 2               | туре в | flow modulator (pre-/post-operation)  |
| Costache et al. [801]                         | 2020    | Ansys <sup>a</sup> | n.a.               | 16              | Туре В | Impact of multilayer flow modulator on aortic<br>remodeling/hemodynamics (pre-/post-operation)  |
| Xu et al. [804]                               | 2021    | CFD-ACE+           | n.a.               | 51              | Туре В | Comparison of hemodynamics/morphology in patients<br>(pre-/post-TEVAR)  |
| Armour et al. [805]                           | 2021    | Ansys <sup>b</sup> | Lam.               | 2               | Туре В | Impact of inflow profile on hemodynamics (pre-/post-<br>TEVAR): validation to <i>in vivo</i> data   |
| Zhu et al. [806]                              | 2021    | Ansys <sup>b</sup> | Turb.              | 17              | Type A | Impact of hemodynamics/morphology on FL dilatation in<br>patients (post operation)  |
| Costache et al. [802]                         | 2021    | Ansys              | n.a.               | 23              | Туре В | Comparison of hemodynamics/morphology in patient<br>with multilower flow modulator (pre /poet operation)  |
| Wang et al. [853]                             | 2021    | Ansys <sup>a</sup> | Lam.               | 15              | Type A | Impact of ascending aortic diameter/hemodynamics in<br>dissection chealthy patients on outcome  |
| Shad et al. [854]                             | 2022    | SimVascular        | n.a.               | 9               | Type A | Impact of hemodynamics/morphology on FL dilatation in<br>patient (next operation)   |
| He et al. [855]                               | 2022    | Ansys <sup>a</sup> | n.a.               | 53              | Туре В | Impact of hemodynamics in patients pre-/post-residual dissection and in patients with closed FL (post-TEVAR)  |

| М. | Rolf-Pissarczvk     | et  | al. |  |
|----|---------------------|-----|-----|--|
|    | reeg r would object | UL. | uu  |  |

Table 7 (continued)

#### Solver Flow Main objective(s) Report Year No. of Type patients assumption Liu et al. [856] 2022 Ansys<sup>a</sup> Lam 2 Type B Impact of inlet/outlet boundary conditions on hemodynamics pre-/post-TEVAR with/without involvement of the abdominal aorta; validation to in vivo data Impact of hemodynamics/morphology on FL dilatation in Zhu et al. [807] 2022 Ansys<sup>b</sup> Turb 4 Type A patients (post-operation) Comparison of hemodynamics in patient with/without Fatma et al. [810] 2022 2 Type A Ansvs<sup>a</sup> Lam FL dilatation (post-operation) Parker et al. [808] 2022 69 Type B Impact of FL thrombosis on FL pressure/complications n.a. n.a. Armour et al. [809] 2022 Ansys<sup>b</sup> 1 (Pig) Type B Impact of tears (number) on hemodynamics/FL Lam. dilatation Zhang et al. [857] 2022 CFD-ACE+ 1 Type B Comparison of hemodynamics in patient with modular n.a. single-branched stent-graft with aberrant right subclavian artery (pre-/post-operation) Wang et al. [858] 2023 n.a. n.a. 1 Type B Hemodynamics in patient with residual dissection (pre-/ post-TEVAR) Osswald et al. [811] 2023 5 Star CCM+ Type B Role of hemodynamics in patient with frozen elephant Lam. trunk procedure in dSINE (pre-/post-operation) 2023 Yoon et al. [859] 16 Type B Impact of specific stent-graft on dSINE (post-TEVAR) n.a. n.a. Pan et al. [860] 2023 FOAMExtend Validation of hemodynamics to in vivo data (pre-/postn.a. 1 Type B TEVAR) Bäumler et al. [861] 2023 SimVascular n.a. 4 (Mouse) Type B Impact of FL thrombus formation on hemodynamics in mice models with angiotensin II-induced dissection Ansys<sup>b</sup> Wang et al. [862] 2023 n.a. Type B Comparison of hemodynamics with/without dSINE 4 (post-TEVAR) Jiang et al. [863] 2023 Ansys<sup>a</sup> Lam 6 Type A Impact of physician-modified endograft in patient with complicated dissection on outcome (pre-/post-operation) Ritter et al. [864] 2024 Star CCM+ Lam. 5 Type A, B, & Impact of stent-assisted balloon-induced intimal non-A non-B disruption and relamination technique on hemodynamics (pre-/post-operation) Kimura et al. [865] 2024 Comparison of hemodynamics in patient with mesenteric n.a. n.a. 1 Type B artery malperfusion due to dynamic obstruction (pre-/ post-TEVAR) Forneris et al. [866] 2024 22 Comparison of hemodynamics/aortic dilatation in Ansvs<sup>a</sup> Lam Type B complicated/uncomplicated cases 2024 Mei et al. [867] Ansvs<sup>b</sup> 55 Impact of hemodynamics/morphology on FL dilatation in n.a. patient with superior mesenteric artery dissection

<sup>a</sup>Fluent; <sup>b</sup>CFX; <sup>c</sup>Mechanics; <sup>d</sup>Multiphysics; <sup>e</sup>Explicit; <sup>f</sup>Standard

suitable inflow and outflow conditions, and (iii) realistic rheological parameters. Even minor changes in these parameters can lead to significant deviations in the results.

- The characteristic flow jets can cause a local increase in pressure on the outer wall of the false lumen and, depending on the entry tear location, trigger a disturbed, oscillatory flow in the proximal false lumen. The volumetric flow through the false lumen heavily depends on the location and size of the primary entry and exit tears, as well as on additional fenestrations. The same parameters also affect residence times in the false lumen and thus possible false lumen thrombosis. The pressure in the true and the false lumen can be elevated in the arch region and is generally higher than in FSI models when CFD is taken into account, i.e. if the cushioning effect of a deformable vessel and a dissection flap is neglected.
- By virtually adding or occluding fenestrations, the entry tear, or the exit tear, CFD models enable the study of possible implications of stent-grafts or other medical implants for hemodynamics in a patient-specific setting. Specifically, the major influence of tear occlusion on flow in the false lumen was demonstrated and its relation with thrombosis in the false lumen was established.
- WSS concentrations are observed locally in the aortic arch region as the ejected fluid is reoriented and Dean vortices form. The region in which the dissection flap separates from the remaining vessel wall can also be exposed to increased stress, which largely depends on the vicinity to the intimal tear and aortic arch as well as the geometry and location of the intimal tear.
- The true-to-false lumen pressure difference in aortic dissection is highly patient-specific and is influenced by morphological factors such as the number, size, and location of intimal tears, as well as the volume of the false lumen. Typically, the dissection flap maintains a pressure differential of 2 to 3 mmHg, but in some cases, it can exceed 5 mmHg. This pressure difference can be positive, meaning that the true lumen experiences higher pressure, especially during systole. However, it can also be negative, where the false lumen has higher pressure because it cannot release the buildup pressure as quickly as the true lumen. Moreover, the gradient direction of the pressure difference can also vary along the length of the dissection flap. This dynamic can vary significantly based on the individual's unique morphology, ultimately influencing the direction of flow within the false lumen and through the respective tears.
- Despite the necessary and sometimes major assumptions, the derivation of hemodynamic parameters now finally allows for the prediction of complications where purely geometric indicators might fail. True-to-false lumen flow splits and pressure differences,

WSS-based indicators, and localized stress concentrations are promising examples in this regard. However, it is still unclear how reliably CFD models describe the clinical environment in general. The numerous studies presented highlight their potential for clinical application, even if only routinely acquired medical data are available. One of the biggest remaining problems is the complex pre- and post-processing and simulation turnaround times, which represent a major bottleneck for personalized medicine. This last aspect in particular motivates further research and development, not only in the application and combination of numerical methods but also in the development of the underlying numerical methods themselves.

# 7. In silico fluid-structure interaction models

The simulation of aortic dissection and all its essential elements is a particularly multifaceted and challenging task in terms of biomechanical modeling and from the perspectives of mathematical solutions and computer implementation. The basic components are largely the same as in hemodynamic simulations of the cardiovascular system, namely the reconstruction of the computational domain, the selection of appropriate inflow and outflow boundary conditions, and the definition of applicable fluid and tissue models. Incorporating the motion of the flap and the aorta as a whole has been shown to significantly alter the computational results and possibly the conclusions drawn [448,868,869]. When performing FSI simulations, additional complexities emerge, especially in the selection and customization of boundary conditions and constitutive models for the vessel wall to address patient-specific characteristics. In this discourse, the emphasis is on contributions and work that specifically target aortic dissection. Other aspects have been extensively covered and discussed in the literature; see [22,24,870,871] and others.

For various reasons, it is extremely difficult to account for vessel wall compliance by solving an FSI problem. From a mathematical perspective, challenges lie in the parameter combinations that occur in hemodynamic applications. As the density ratio of solid to fluid approaches one, high added-mass effects come into play, hindering the convergence of standard partitioned coupling schemes but also impacting monolithic solution approaches. In addition, incompressible fluid and solid phases have to be considered, further limiting the admissible formulations and solvers for each tissue and blood model. The difficulties encountered in CFD modeling of the circulatory system and numerical treatment are transferred to the FSI case because the FSI problem naturally involves a fluid flow problem. However, in addition to the fluid and solid subproblems, interface tracking or interface capturing techniques must be employed to consider the domain and interface motions. The coupled problem typically involves fluid and solid subproblems, mesh deformation, and strategies for decomposition or level-set updates. It must also ensure effective coupling of the involved fields to achieve temporal stability. Specialized modeling approaches for viscoelastic support, pre-stress in the tissue at the time of image acquisition, and the connection of inlets and outlets significantly increase complexity and computational effort for both the fluid and solid phases. However, compared with fixed-grid CFD simulations, it is not clear to what extent or under what circumstances domain motions or interaction of blood and tissue can be neglected. It has to be evaluated on a case-by-case basis if the additional computational effort, time to solution, and the introduction of additional parameters to adapt to clinical data are well invested.

#### 7.1. Simplified studies

Similar to CFD approaches, useing idealized geometries in an FSI setting can enhance the basic understanding of the mechanisms underlying aortic dissection. Especially due to the complexity of FSI models, this may be a good first step toward patient-specific applications. In this overview, the available FSI models based on simplified geometries are briefly presented and their most important findings are summarized below.

In this manner, Qiao et al. [872] investigated the effects of bypassing type B dissection using two alternative virtual bypass strategies: (i) bypass between the ascending aorta and abdominal aorta and (ii) bypass between the left subclavian artery and the abdominal aorta. Due to the choice of parameters, the maximum flap displacement at peak systole in a blind (or patent) lumen model before bypassing was approximately 1.3 mm. However, even in this scenario, comparing a rigid wall simulation with the full FSI model revealed significant differences in the observed fluid pressure and velocities, highlighting the potential significance of FSI modeling in the context of aortic dissection. Furthermore, they demonstrated *in silico* that the bypass strategies positively influenced the vessel walls by reducing the mean pressure amplitudes, both in the case of a false lumen with an exit tear and in a virtually created blind lumen.

Their study was subsequently followed by numerous others. Chen et al. [873] investigated blood flow and flap wall motion using a simple approach consisting of a tube with an integrated flap. They identified regions of stress concentration at the leading edge of the flap. Jayendiran et al. [874] developed a simplified composite structure composed of a human aortic segment. The segment was modeled with three aortic layers using both isotropic and anisotropic material models in conjunction with a Dacron graft. This setup made it possible to examine the interaction between fluid and structure. In a related study, Ryzhakov et al. [875] used a monolithic FSI approach to simulate fluid flow in a flexible tube with a parallel channel connected via entry and exit tears formed on the basis of the *in vitro* study of Rudenick et al. [618]. Comparisons between rigid and elastic flaps revealed that the elasticity of the silicon flap had minimal impact on fluid flow. Keramati et al. [876] concentrated on studying blood flow characteristics within an idealized aortic dissection geometry that included 5 and 10 mm diameter tears. Their findings suggested that flow resistance was lower and less flutter of the dissection flap occurred when both entry and exit tears were 10 mm in diameter, resulting in increased stability. They later compared this model with a reduced-order model [877] and, subsequently, analyzed the sensitivity of hemodynamic quantities to the lumped parameters of the 0D, or reduced-order, model using a Monte Carlo simulation [878]. A rudimentary model, which included a healthy aorta and three distinct types of aortic dissection models, was developed by Saveljica and Filipovic [879] to study the interaction between fluid and solid. Building on previous studies [756], see Section 6.1, the findings by Qiao et al. [880] provided

valuable insights into understanding the energy loss mechanisms in aortic conditions, with viscous friction in particular contributing the most to energy loss, followed by deformation of the aortic wall. Aghilinejad et al. [881] adopted immersed boundary and lattice–Boltzmann methods (IB-LBM) to explore the relationship between endograft length and hemodynamic variables within the true and false lumina. The findings revealed a nonlinear trend, indicating a progressively increasing reversal of false lumen flow as endograft length increased. Recently, Kim et al. [882] conducted a study of dynamic obstruction of the true lumen in type B dissection, and Lee et al. [883] examined the effect of cannulation on hemodynamics with two simulated methods: (i) axillary cannulation through the branchiocephalic trunk and (ii) combined axillary and femoral cannulation through the branchiocephalic trunk and the right common iliac artery. The first method resulted in a pressure difference between the lumina, causing the true lumen to collapse. In contrast, the second method maintained similar pressures, preventing the collapse of the true lumen. Additionally, they found that the stiffness of the flap influenced the collapse of the true lumen.

Subsequently, two studies by Chong et al. [884,885] established an FSI model that took thrombus formation into account. To date, this is the only *in silico* model on aortic dissection that combines FSI and thrombus formation. First, Chong et al. [884] developed a simplified geometry of a type B dissection, including an entry and exit tear. They investigated flap dynamics and its correlation with WSS, pressure, and wall stresses. This model was subsequently extended to account for the gradual influence of thrombus formation on the flow field, as described in Section 8.

In summary, simplified models have provided important contributions to modeling FSI in aortic dissection. However, due to recent advances in FSI and the availability of patient-specific models, these are becoming increasingly outdated. However, when incorporating thrombus formation modeling into FSI simulations, simplified models remain the only feasible option to date due to the high computational effort required. Nonetheless, the majority of these studies have made valuable contributions by enhancing our understanding of aortic dissection and encouraging the use of FSI modeling through targeted comparisons.

#### 7.2. Patient-specific studies

In contrast to the extensive research on CFD simulations, there are only a limited number of patient-specific FSI models of aortic dissection, largely due to the complexity of the challenges involved. Below is a summary of the basic components and the main findings of these published works.

Alimohammadi et al. [868] published one of the first studies that captured the rheological behavior of blood using a non-Newtonian model, considering the aortic tissue as a 3D continuum in the context of aortic dissection. There, a patient-specific lumen geometry was constructed from CT images and expanded by a fixed margin of 2.5 mm to obtain a geometric representation of the vessel wall and a flap thickness of  $2.45\pm0.34$  mm. For the fluid, the Carreau-Yasuda model was considered along with an SST turbulence model. Data on the inlet boundary were obtained from the literature, while Windkessel models were used to match volumetric flow data and pressure of patients at the outlets obtained from invasive pressure measurements. The constitutive law governing the isotropic, incompressible, hyperelastic response of the tissue was a specific form of generalized power law of the neo–Hookean model [886], and the outlets were not fixed but could expand and contract in the radial direction of the vessel. Furthermore, a uniform external pressure of 52.5 mmHg was applied, neglecting the elasticity of the surrounding tissue. Due to the specific geometry and tissue parameters considered, small displacements of the vessel wall and interface were reported, resulting in deviations in the cross-sectional areas of the true and false lumina in the range of  $\pm5\%$ . It is all the more remarkable that when comparing fixed-grid CFD and FSI simulations, clear differences in flow distribution and OSI were found even under these conditions. Additionally, it was observed that standalone CFD computations might suffice when the primary focus is on TAWSS. They argue that critical differences and key features may not be accurately captured when wall and flap motion were neglected in the modeling of aortic dissection, otherwise particularly regions of high OSI combined with low TAWSS were not well captured.

In an attempt to include vessel wall motion at a lower cost, Bonfanti et al. [768] introduced what is known as the moving boundary method (MBM). The displacement of the vessel wall was linearly related to the pressure of the fluid, while the dissection flap remained rigid. For the fluid, a Carreau-Yasuda model was considered along with flow profiles and Windkessel models to account for realistic boundary conditions. The approach was tested on a patient-specific case of type B dissection compared with standard fixed-grid CFD, and an improved match with target patient data in terms of flow rates and pressure waveforms was documented. In [769] the method was then further improved and a deformable dissection flap was also introduced. When comparing FSI, fixed-wall CFD, and the newly introduced method, equal differences in TAWSS were observed, while pressure and flow rates of the MBM were closer to FSI than the fixed-grid CFD approach. Further simplifications were subsequently introduced in [770] using lumped parameter models, as discussed in Section 6. In summary, the aim of these studies was to capture the phase-lag of blood flow and pressure waves due to vessel compliance, which cannot be captured using standard fixed-grid CFD methods. In addition, the parameters in the MBM were tuned using an FSI solution. However, in clinical practice, non-invasive measurements could be employed to perform parameter fitting, which is easier due to the limited number of model parameters compared with complex tissue models that consider hyperelastic, fiber-reinforced 3D continua. Therefore, the main advantage of the developed methods is that they significantly reduce the computational costs, while capturing the relevant effects of vessel motion to some extent, making them increasingly attractive for application in clinical scenarios. For a brief breakdown of potential precision medicine pipelines based on these concepts, see [887].

Qiao et al. [869] presented the first study on aortic dissection considering FSI in combination with two-phase non-Newtonian fluid flow, where red blood cells (the main particulate phase) are suspended in plasma (the continuous phase). A patient-specific geometry was created from CT images, which specified a uniform wall thickness of 2 mm. For the boundary conditions of the solid subproblem, a fixed external pressure corresponding to the diastolic pressure in the false lumen [868] was applied. Additionally, the center points of the vessel inlets and outlets were fixed, effectively stabilizing the position of the aortic tree. It should be noted that the influence of the surrounding tissue was not considered. Three-element Windkessel models with parameters from the literature were applied at the outlets and a physiological flow rate was enforced at the inlet. An isotropic, linear elastic material behavior was assumed and the tissue was assigned an increased density of  $2000 \text{ kg/m}^3$  and a Young's modulus of 2.7 MPa. The resulting displacement of the dissection flap in the range of 0.68 ± 0.2 mm was found to coincide well with the MRI data from [447]. The observed small wall displacements resulted from the aforementioned supra-physiological stiffness and density, which significantly simplified the numerical treatment. Nonetheless, important conclusions can be drawn from this study, as it accounted for the multiphase nature of blood. The main objective of the study was to compare the results obtained from the FSI and rigid wall models, as well as from single- and two-phase non-Newtonian models, by evaluating OSI and TAWSS, in addition to assessing the deformation of the vessel wall during a cardiac cycle. They demonstrated that the relative differences in WSS between single- and two-phase non-Newtonian models could be up to  $\pm 50\%$  during the cardiac cycle, suggesting the potential to improve blood flow modeling in cases of aortic dissection. However, it should be noted that the study did not report absolute differences in WSS. Although the relative differences might seem significant, the absolute values typically fall below 1 Pa, suggesting that small absolute differences can result in large relative variations. Consequently, comparing relative values may yield misleading interpretations. According to them, accounting for the complex rheology of blood using multiphase non-Newtonian models involving individual constituents has dramatic effects on hemodynamic parameters. However, these models typically require detailed patient-specific data, which are often not available. Therefore, single-phase generalized Newtonian models that account for shear-thinning and plug flow effects can be considered a worthwhile compromise between predictive power and complexity. Although this study provided initial insights into the significance of non-Newtonian models, the interaction between larger motions of the dissection flap and the resulting significantly altered flow rates in true and false lumina remains unclear. Similar computational tools that attribute the rheological behavior to single-phase non-Newtonian behavior were then applied in the clinical scenario in Qiao et al. [835]. Herein, pre- and post-TEVAR configurations in acute type B dissection, with and without coverage of the left subclavian artery, were compared in terms of hemodynamic indicators based on TAWSS to quantitatively assess the effectiveness of the surgical procedure.

Whether FSI modeling could provide additional insights into the potential evolution of aortic dissection was investigated by Khannous et al. [888] by comparing the flow behavior within a patient-specific model, both with and without considering FSI. For the analysis, a residual type B dissection was reconstructed from the patient's CT scans. Blood was modeled as an incompressible fluid with shear-thinning behavior using the Carreau–Yasuda model. Flow rate and unsteady velocity profiles were set at the inlet, and a three-element Windkessel model was used for the outlet. The thickness of the aortic wall was assumed to be constant at 2.5 mm, while that of the flap varied around 1.85 mm. A linear elastic material model was applied for the aortic wall. The results of the study indicated that the rigid model tends to overestimate the velocity across the domain on average. However, it was observed that the velocities at the entry tear were underestimated. Moreover, a flow jet was detected in the FSI case, which did not appear in the rigid model analysis. This underlines that FSI may provide more comprehensive insights into aortic dissection behavior.

One of the most notable FSI studies, published by Bäumler et al. [448], involved patient-specific FSI simulations of type B dissection and was acclaimed for its highly realistic approach at the time. Blood was considered as an incompressible Newtonian fluid and the aortic tissue was modeled as an isotropic, hyperelastic and nearly incompressible continuum. Patient-specific 4D-flow MRI data were utilized to prescribe an inflow profile matching the measured volumetric flow rate, and three-element Windkessel models were tuned to account for the downstream vasculature. Furthermore, Robin boundary conditions were considered, see, e.g., [889], to account for external tissue support, with values adopted from the literature to account for elastic support of the aorta. A pre-stressing algorithm [890] was applied to compute the stress present at the time of image acquisition in the solid domain with the exception of the dissection flap, which was assumed to have no stress in the diastolic phase. The anatomical model was developed from high-resolution CTA data, the entry and exit tears as well as multiple fenestrations. The model was further processed manually to achieve a consistent wall and flap thickness of 2mm. This setup enabled investigations of flow rates, true-to-false lumen pressure difference, TAWSS, and geometric quantities such as cross-sectional area or diameter variations. For example, Fig. 63(a) shows the TAWSS, while Fig. 63(b) displays two snapshots of the WSS at peak systole and end-diastole. This contribution highlighted the importance of flap stiffness by varying the elastic modulus of the dissection flap in and around the physiological range in a validated computational environment. Good agreement was found, including remarkably large flap motion within the realistic range of 1.4 to 14.3mm depending on the stiffness of the dissection flap. The combination of an adequate geometric approximation, a monolithic FSI solver including realistic boundary conditions for both fluid and solid sub-problems with patient-specific data leads to one of the most holistic approaches presented in the literature. On the other hand, it remains unclear to what extent the assumption of an initially stress-free dissection flap influences the results achieved or how the inclusion of pre-stress in the dissection flap affects the convergence behavior of the prestressing algorithm.

The simultaneous analysis of blood flow and tissue within the framework of FSI models is a challenging task not only due to the complexity of the individual fluid and structural models, but above all due to the coupling of these fields. Coupled problems in continuum mechanics occur in various contexts and have their own methods of analysis, mainly distinguishing between monolithic and partitioned approaches. In the vascular context, including aortic dissection, the works by Schussnig et al. [891,892] can be considered key technologies for efficient and stable analysis using partitioned algorithms. They combined (i) Robin transmission conditions, a mixed-type interface condition that simultaneously enforces continuity of velocities and tractions, (ii) semi-implicit FSI schemes that merely iterate the structural displacements and fluid pressure resulting from a consistent split-step fluid solver [893], and (iii) quasi-Newton methods to accelerate the FSI coupling algorithm. This approach proved to be robust and efficient in practical application to a patient-specific case of aortic dissection in [894], taking into account Windkessel models, exterior tissue support, and pre-stress. In terms of constitutive modeling, the shear-thinning behavior of blood was captured via generalized Newtonian models, while the non-symmetric distribution of collagen fibers was captured by the model proposed by Holzapfel et al. [895] with realistic





fiber orientations generated by a custom algorithm [896,897], and different tissue parameters in the dissection flap as well as in the medial and adventitial layers of the aorta were taken into account. The results of the algorithm proposed to construct material orientation in aortic dissection geometries are shown in Fig. 64. These works show that partitioned schemes can indeed be applied in a clinical scenario. This is of particular relevance because monolithic coupling was long considered the only reliable FSI coupling strategy in hemodynamics, a viewpoint that is no longer valid today. As [898] shows, partitioned FSI has the potential to outperform monolithic FSI in terms of throughput by several orders of magnitude (factor 100 with parameters in the vascular regime), thereby significantly reducing simulation turnaround times. The adaptation of modern high-performance software for use in a medical context is indeed a promising topic, as it bears great potential to significantly enhance the clinical relevance of both CFD and FSI methods.



**Fig. 64.** Construction of material orientation basis vectors in a patient-specific aortic dissection model (only aortic arch region shown): (a) axial direction, (b) circumferential direction based on two subsequent Laplace solutions, and (c) circumferential direction as obtained via the approach by Schussnig et al. [892], which uses an extrapolated interface normal. The latter approach delivers better results, especially in the flap region of the aortic dissection. Figure is adapted from Schussnig et al. [892] (licensed under CC-BY 4.0).
Schussnig et al. [899] then extended the framework previously established by Bäumler et al. [448] while maintaining the original geometry and flow boundary conditions. In this study, they introduced a numerical framework that considered, among other things, a combination of: (i) patient-specific geometry with three layers comprising the combined intima and media, the adventitia, and the dissection flap; (ii) layer-specific tissue models such as the anisotropic constitutive model for the aortic wall layers that takes wall microstructure into account; (iii) a non-Newtonian rheological model; (iv) patient-specific boundary conditions derived from 4D-flow MRI; (v) viscoelastic exterior tissue support; (vi) pre-stressed tissue that counteracts diastolic fluid load; and (vii) prescribed aortic root motion at the inlet. This framework also required the introduction of a local coordinate system, as previously developed and shown in Fig. 64. In summary, the study compared hemodynamic indicators and stress measurements in computational models with varying complexity of vessel tissue material models, from rigid walls to anisotropic hyperelastic materials, making the presented framework one of the most comprehensive for aortic dissection studies to date. The Central new aspects rooted in the detailed analysis of tissue stress measures in a layer-specific and direction-dependent manner, utilizing the material coordinate system embedded within tissue layers. They evaluated and compared dissection flap displacements, flow rates in both true and false lumina, true-to-false lumen pressure differences, and shear stress-based indicators such as TAWSS, OSI, HOLMES, and endothelial cell activation potential. The results closely matched those of previous studies [448]. Most results showed significant differences between rigid wall flow simulations and those using linear elastic and hyperelastic models in an FSI approach. However, some hemodynamic indicators exhibited only small variations. A detailed analysis of stress measurements, as exemplified in Fig. 65, revealed that a significant portion of the pressure load is transferred to the viscoelastic support. In particular, the maximum positive principal stress depends on the tissue-tosupport stiffness ratio. In fact, all hyperelastic models with the current parameter set produced mostly similar results. This suggests that existing methods that use relatively stiff viscoelastic support may not adequately replicate the load-bearing behavior in vivo. If the viscoelastic support in vivo is significantly lower than assumed in the literature, the relative importance of ring forces increases and thus the influence of constitutive modeling, anisotropy, and fiber reinforcement. It is noteworthy that this latter conclusion has received little attention in the past.

Zimmermann et al. [612] then adopted the computational framework by Bäumler et al. [448] and qualitatively and quantitatively compared the hemodynamics in a type B dissection between the FSI analysis and a 3D-printed model, as discussed in Section 4.2 (see Figs. 42(a) to (c)). Both are based on CTA data and assume a dissection flap thickness of 2mm. As previously introduced, the experimental setup allowed tuning of inflow and outflow conditions to achieve physiological inflow waveforms and luminal pressure levels. In the *in silico* model, blood was assumed to be Newtonian and an isotropic neo–Hookean model was fitted to the tensile test results of the polymer used for the 3D printed model. The boundary conditions for the computer simulation were then fitted to the measured *in vitro* data, including the inflow and tuning of the Windkessel parameters. This allowed a comparison of flow rates, changes



**Fig. 65.** Close-up of the layer-specific finite element mesh in the aortic arch region, showing the cut adventitia (dark red) and media (peach) layers and exposing the dissection flap (light blue) originating at the left subclavian artery: (a) to (c) box plots display the maximum positive principal stress at the time step with the highest value, considering only the top 40% of values and removing the top 1% to exclude outliers. The load distribution between the tissue layers (a) to (c) differs significantly when comparing the E90 with hyperelastic models. The gHGO model with base parameters also differs from the NH, gHGO90, and gHGO120 models in the adventitial layer, indicating different load-bearing behavior. Large differences can be observed in the dissection flap. Figure is adapted from Schussnig et al. [899] (licensed under CC-BY 4.0).

in cross-sectional area and pressure using 4D-flow MRI, cine 2D MRI, 2D PC-MRI, and pressure catheter measurements at selected points. The flow splits of 78/22 *in silico* and 73%/27% *in vitro* agreed well, but despite adjusting velocities and pressures, both were overestimated in the computational model. The mean pressure in the computational model differed by 15.8%. The expansion of the false luminal area was 11% in the FSI analysis, while it was measured at 5% in the cine 2D MRI, which may be caused by the different mean pressure levels and possible differences in the true-to-false lumen pressure. As already discussed in Section 4.3, discrepancies in the results between *in vitro* experiments and *in silico* models could be due to different resolutions between 4D-flow MRI and 2D PC-MRI. Additionally, higher damping in the 3D printed model, possibly due to underestimated resin stiffness, and neglect of the *in vitro* support from the embedding gel in the FSI simuations may also contribute to these discrepancies. This work highlights two aspects of computational methods for clinical support: (i) FSI modeling can actually be used to predict hemodynamics even in complex pathologies such as aortic dissection, but (ii) fitting all model parameters and obtaining all necessary data is a formidable challenge. Nevertheless, the validation study presented shows that computational biomechanics is indeed a promising tool for application in the cardiovascular system and can provide new insights as long as the model parameters are sufficiently calibrated.

The first longitudinal case study using FSI simulations that captures the evolution of a patient's aortic dissection, from predissection through the subacute phase to five years after disease onset was recently presented by Bäumler et al. [342]. For this purpose, they applied monolithically coupled FSI models with tissue pre-stress, external tissue support, and an anisotropic tissue model to study potential growth-related hemodynamic markers under realistic conditions. Moreover, the extracted wall geometries differentiate between the tissue layers to allow for layer-specific material properties. Aortic dissection geometries, in vitro 4D-flow MRI [612], and the patient's blood pressure were used to apply in vivo boundary conditions at each follow-up. This allowed to study FSI simulation results and anatomical changes, at each available follow-up, which provided the unique opportunity to qualitatively correlate growthrelated hemodynamic markers and aortic growth over time to advance our understanding of the evolution and interplay of hemodynamics, biomechanics, and aortic remodeling that contributes to the most common late adverse event in aortic dissections, namely aortic dilatation. Computational results and aortic growth over time were analyzed, revealing a decrease in flow jet velocity at the entry tear as the disease progressed (Fig. 66). This flow jet caused locally increased WSS and blood pressure in the impingement zone on the outer wall of the false lumen, correlating with significant aortic dilatation in these areas. Similar effects were noted in the true lumen distal to the entry tear. Furthermore, the comparable pressure levels in the true and false lumina indicate a hypertensive-like state in the false lumen, particularly during acute phases when the outer wall is abnormally thin. This disruption of homeostasis characterized by altered WSS, increased pressure, and hypertension - has been associated with changes in mechanosensing and growth and remodeling mechanisms in the aortic wall, highlighting the need for future research in this area.

Expanding on their previous work that neglected vessel wall compliance [806,807], Zhu et al. [900] investigated repaired type A dissection adopting an FSI algorithm and compared the results with pure flow simulations. Two patient-specific models were derived from CTA data, in which the structural domain was generated by uniformly expanding the luminal surface by 1.4 mm in the normal



**Fig. 66.** Displayed at peak systole, the flow velocity for the pre-dissection ( $\mathcal{M}_0$ ) and the three post-dissection models ( $\mathcal{M}_1$  to  $\mathcal{M}_3$ ) highlights the flow jet and the increase in blood velocity in the distal part of the false lumen. Additionally, a close-up view of the region where the flow jet impinges on the outer wall of the false lumen (solid arrow) is provided. This view also emphasizes a localized dilatation in the true lumen (dotted arrow) and the flow separation occurring in the true lumen (dashed arrow). Figure is reprinted from Bäumler et al. [342] with permission from IEEE.

direction. The boundary conditions for the fluid flow problem were adopted from their previous study [806] and enforce a flat inflow profile in combination with prototypical flow waveforms fitted to patient data that took into account patient-specific heart rates and maximum velocities. Windkessel models were fitted assuming flow splits at the outlets, with the computational model shortened depending on the case and available imaging data. Therefore, the outflow conditions had to be tuned to achieve the desired volumetric flow rates when exiting the true and false lumina. Although not available, these conditions strongly influence the flow split between the true and false lumina and thus the true-to-false lumen pressure difference. An incompressible Newtonian fluid was considered together with the transitional SST turbulence model. Both the stent and the aortic tissue were considered linear elastic, with Young's moduli of 7.8 and 1.3MPa, respectively. External tissue support was represented by Rayleigh damping, see, e.g., [901], where volumetric strong damping was used, which should not be confused with conventional spring and dashpot supports as adopted by, e.g., [448,894]. The inclusion of pre-stress was based on the stress state resulting from the diastolic pressure load, which is a reasonable approach.

They reported vessel displacements of about 1 mm, which naturally depended on the chosen tissue stiffness. A change in stiffness also led to changes in volumetric flow rates and velocities, as higher true lumen pressure caused displacements of the dissection flaps and expanded the true lumen cross-section. Depending on the specific case and whether the dissection flap has already remodeled or not, these results may be in good agreement with *in vivo* observations. In this study, regions with higher TAWSS are significantly increased when vessel compliance is taken into account, as the relative motion of fluid and solid can potentially contribute to shear in the boundary layers. Furthermore, the turbulence intensity was increased in the presented configurations, but the true-to-false lumen pressure differences were only slightly affected.

Depending on the subject of investigation, they also argue that considering FSI in the context of surgically repaired type A dissection does not result in more insights in the clinical setting. In fact, the high costs and the additional model assumptions required do not allow any further conclusions to be drawn than can be obtained from (much faster) CFD computations alone. We would like to emphasize that the issue of data availability is crucial, especially in the patient-specific context. However, as data sets become more available and parameter estimation methods become more popular in biomechanics and medicine, the increased computational and modeling effort is actually justified if simulation turnaround times can be kept reasonable. Tissue properties can then be selected to represent the actual anisotropic behavior, although in this study they had a relatively small impact, apart from changes in TAWSS. However, as shown in Fig. 67(b), wall stiffness also affects the pressure difference between true and false lumina. Even though pressure differences on this scale, i.e. between 0 and 4Pa, are not clinically measurable, they reveal the impact of tissue models and parameters on the results. Such differences can also lead to flap movement and associated risks, as discussed in Section 3.1.



**Fig. 67.** (a) Spatial distribution of von Mises stress for patient 1 and patient 2 in a study by Zhu et al. [900]. Black arrows indicate the highest von Mises stress values at the wall-graft interface, while red arrows highlight isolated areas of elevated von Mises stress in regions not in direct contact with the graft. (b) Quantitative assessment of the pressure differences between the true and false lumina across selected cross-sectional planes of the aorta of patient 1. Pressure differences were analyzed across FSI models that differed in Young's modulus values. Figure is adapted from Zhu et al. [900] (licensed under CC-BY 4.0).

The high stiffness considered corresponds to a remodeled dissection flap, but the external tissue support was not considered via springs and dashpots, but via Rayleigh damping, which can drastically alter overall system behavior, see, e.g., [894,901]. As a result, dissection flap displacements were hindered in multiple ways, so that a vast change in hemodynamics was not to be expected. However, Fig. 67(a) demonstrates tissue stresses in the region of the aortic arch and highlights the typical locations for stress concentrations by plotting the von Mises stress resulting from the deployment of a stent-graft. This highlighted that stent placement induced local stress peaks and thus potential future dissection sites. Since this region was also strongly influenced by inflow and aortic root motion, future studies investigating such scenarios might incorporate the mentioned aspects and anisotropic tissue models to investigate the stress and strain concentrations in more detail.

Wang et al. [902] explored the ability of a biomechanical model to predict risks associated with a non-dilated thoracic aorta in type A dissection. They focused on the correlation between von Mises stress, WSS, and blood flow in a patient with type A dissection. The implemented model included a patient-specific 3D aortic dissection geometry and two aortic layers with constant thickness: intima and adventitia. The anisotropic properties of the aortic wall were modeled using the HGO model. Furthermore, *in vivo*-based physiological time-varying blood velocity profiles from the literature, non-Newtonian rheology (Quemada model), and a turbulence model were adopted. The sizes of the entry and exit tears were derived from the existing literature data. Aortic stress values have been found to be primarily determined by a mixture of blood flow dynamics, wall properties, and arterial tortuosity. The maximum von Mises stress was identified in the region of the aortic arch adjacent to the brachiocephalic artery. Notably, the intimal tear was subjected to less stress compared with the true lumen wall. Moreover, the von Mises stress in the intimal layer exceeded that in the adventitia. However, the study had certain limitations. The lack of a pre-stressing algorithm reduces the significance of the obtained wall stresses. Although the study used patient-specific geometry and *in vivo* boundary conditions – many of which were obtained from the literature from different patients with varying geometries – this may have limited the validity of the results. However, despite these limitations, the use of an anisotropic material model, a non-Newtonian model to describe blood flow, and a turbulence model in an FSI framework represents significant improvements compared with previous works.

In summary, considering the coupling of solid and fluid phases when modeling aortic dissection using FSI can have a tremendous effect on the results obtained in terms of true and false lumina flow rates and true-to-false lumen pressure differences, and derived quantities and clinical markers such as OSI or TAWSS associated with initialization and onset of dissection. Taking vessel compliance into account, these quantities may vary significantly depending on the specific case. By increasing aortic wall stiffness, WSS, and related quantities such as RRT, OSI, and HOLMES undergo potentially large changes. Modeling the tissue allows for studying stresses and strains, which are particularly important in aortic dissection. The dissection flap is dynamically loaded by blood flow, making it subject to significant strains during the cardiac cycle. This dynamic loading may be a key factor in delamination. FSI modeling allows investigations on stress concentrations in the vessel wall due to stent-graft placement or intimal tears of different sizes, locations, and numbers with resulting dissection flap configurations. Likewise, vessel joints or bifurcations, in which there is a relative motion of individual segments, as well as stent-graft ends, where unavoidable misfits lead to a contact pressure being exerted on the vessel from the interior, can lead to localized strains. In all of these cases, the dynamic loading and coupling of solid and fluid can have drastic effects on the stress and strain distribution in the tissue.

Due to the increased complexity and the high computational cost, the numerical treatment and the clinical application derived from it remain a challenge. The need for medical data is even greater compared with fixed-grid CFD approaches, as not only flow rates, pressure measurements, and rheological parameters require parameter tuning. At this stage, critical questions remain about parameter selection for arterial tissue and other relevant models, such as viscoelastic support. Additionally, the extent to which individual constitutive models impact overall system behavior and the best methods for modeling different disease stages are still uncertain. In this regard, uncertainty quantification and sensitivity analysis tools need to be considered to address remaining doubts about the importance of FSI modeling given the high costs involved. In FSI, the relative importance of the subproblems involved and specific modeling techniques as well as the related parameters remain unclear, especially when targeting patient-specific simulations. Therefore, many open questions remain, motivating further research toward modeling aortic dissection using FSI.

# 7.3. Key findings and publication index

To summarize this section, we present the publication index of *in silico* models on FSI in Table 8, with a focus on FSI simulations of aortic dissection using both simplified and patient-specific geometries. Due to the large number of studies available, only a selected range was discussed; however, this table covers an even broader range of studies. Essential information on the modeling approaches, study size, and main objectives of each contribution are summarized. Finally, the most important findings are listed conclusively.

- Overall, there is a favorable correlation between *in vivo* wall deformation and results derived from FSI models; however, the accuracy still depends heavily on boundary conditions and material parameters, the tuning of which requires extensive data sets that are hardly available from today's clinical practice.
- Aortic dissection configurations with a highly mobile dissection flap require modeling using FSI because the flow field is heavily
  influenced by tissue deformation. Consequently, alterations in the stiffness of the dissection flap affect both the volumetric flow rate
  and velocity. If the stress distributions within the tissue are to be investigated throughout the entire cardiac cycle, the dynamic
  coupling of tissue and blood has to be accounted for.
- Under physiological conditions, pre-stress plays an important role and needs to be rigorously considered, primarily capturing the diastolic pressure acting on the tissue at the time of image acquisition. However, the question of whether the dissection flap is pre-stressed and the magnitude of this effect remains uncertain.

### Table 8

Publication index of *in silico* models of FSI related to aortic dissection, ordered by publication year (TL: true lumen; Lam.: laminar; Turb.: turbulent; Iso.: isotropic; Aniso.: anisotropic).

| Report   | Year    | Method(s) | Solver                                      | Flow<br>assumption | Wall<br>model | No. of patients | Туре       | Main objective(s)  |
|--|---------|-----------|---|--------------------|---------------|-----------------|------------|--|
| Simplified studies   |         |           |   |                    |               |                 |            |  |
| Qiao et al. [872]  | 2015    | FSI       | Ansys                                       | Lam.               | Iso.          | -               | Туре В     | Comparison between two bypass<br>strategies (from ascending to abdominal<br>acts) in simplified models   |
| Chen et al. [873]  | 2016    | FSI       | Abaqus                                      | Turb.              | Iso.          | -               | -          | Validation of simplified model (tubular<br>segment with flap) to <i>in vitro</i> data  |
| Jayendiran et al. [874]                                      | 2018    | FSI       | Abaqus                                      | Turb.              | Aniso.        | -               | -          | Comparison between different material<br>models in simplified models (tubular  |
| Ryzhakov et al. [875]  | 2019    | FSI       | Kratos <sup>d</sup>                         | Lam.               | Iso.          | -               | -          | Validation of simplified dissection model<br>to <i>in vitro</i> data   |
| Keramati et al. [876–878]                                    | 2020–24 | FSI       | COMSOL                                      | Lam.               | Iso.          | -               | -          | Impact of entry/exit tears in simplified<br>model (tubular segment with flap);<br>comparison between FSI and reduced.                          |
| Chong et al. [884]   | 2020    | FSI       | COMSOL <sup>c</sup>                         | Lam.               | Iso.          | -               | Туре В     | order model; sensitivity analysis<br>Impact of flap motion on hemodynamics<br>in simplified models; comparison<br>between CED and EEI models   |
| Saveljica & Filipovic [879]                                  | 2021    | FSI       | PAK-SF                                      | Lam.               | Iso.          | -               | Type A & B | Comparison of hemodynamics/wall<br>stress in simplified healthy/dissection   |
| Aghilinejad et al. [881]                                     | 2022    | IB-LBM    | n.a.  | Lam.               | Iso.          | -               | -          | models<br>Impact of stent-graft length on<br>hemodynamics in simplified models   |
| Qiao et al. [880,903]  | 2022/23 | FSI       | Ansys                                       | Turb.              | Iso.          | -               | Туре В     | (tubular segment with flap/tears)<br>Comparison of hemodynamics/heat<br>transfer in simplified healthy/<br>aneurysmatic/coarctation/dissection |
| Lee et al. [883]   | 2023    | FSI       | Ansys <sup>c</sup>                          | Turb.              | Iso.          | -               | Туре В     | models<br>Impact of different cannulation methods<br>in simplified models  |
| Kim et al. [882]   | 2023    | FSI       | n.a.  | n.a.               | n.a.          | -               | Туре В     | Impact of hemodynamics/mechanics on<br>dynamic obstruction of TL with<br>simplified models   |
| <u>Patient-specific studies</u><br>Alimohammadi et al. [868] | 2015    | FSI       | Ansys                                       | Turb.              | Iso.          | 1               | Туре В     | Comparison between CFD and FSI<br>models; validation of flap motion to <i>in</i><br>wine dote (literature)                                     |
| Malvindi et al. [904]  | 2016    | FSI       | Abaqus <sup>f</sup> &<br>Ansys <sup>a</sup> | n.a.               | n.a.          | 1               | _          | Modeling hemodynamics/wall stress pre-<br>dissection with ascending aortic<br>aneurysm   |
| Bonfanti et al. [768]  | 2017    | MBM       | Ansys <sup>b</sup>                          | Lam.               | Iso.          | 1               | Туре В     | Validation of hemodynamics to <i>in vivo</i> data  |
| Bonfanti et al. [769]  | 2018    | FSI & MBM | Ansys <sup>b</sup>                          | Turb.              | Iso.          | 1               | Туре В     | Comparison between MBM, FSI, and CFD models  |
| Qiao et al. [869]  | 2019    | FSI       | Ansys                                       | Lam.               | Iso.          | 1               | Туре В     | Comparison between FSI and CFD models<br>with two-phase non-Newtonian blood<br>flow  |
| Khannous et al. [888]  | 2020    | FSI       | Ansys                                       | Lam.               | Iso.          | 1               | Type B     | Modeling residual aortic dissection  |
| Bäumler et al. [448]   | 2020    | FSI       | SimVascular                                 | Lam.               | Iso.          | 1               | Туре В     | Impact of flap motion/stiffness on<br>hemodynamics; validation of<br>hemodynamics to <i>in vivo</i> data                                       |
| Deng et al. [905]  | 2021    | FSI       | ADINA                                       | Lam.               | Iso.          | 1               | Type A     | Impact of different cannulation methods  |
| Zhu et al. [900]   | 2022    | FSI       | Ansys <sup>b,c</sup>                        | Turb.              | Iso.          | 2               | Туре А     | Impact of wall stiffness on wall stress<br>post-graft replacement; comparison<br>between CED and FEI models                                    |
| Qiao et al. [906]  | 2022    | FSI       | Ansys                                       | Turb.              | Iso           | 6               | Туре В     | Hemodynamics/wall stress in patients<br>with distal dSINE (post-TEVAR follow-  |
| Schussnig [894]  | 2023    | FSI       | Abaqus <sup>f</sup>                         | Lam.               | Aniso.        | 1               | Type B     | Development of local fiber orientation<br>algorithm for layer-specific dissection<br>models  |
| Wang et al. [902]  | 2023    | FSI       | Ansys                                       | Turb               | Aniso         | 1               | Type A     | Hemodynamics in dissection nationt   |
| Zimmermann et al. [612]                                      | 2023    | FSI       | SimVascular                                 | Lam.               | Iso.          | 1               | Туре В     | Validation of hemodynamics to <i>in vitro</i><br>data; impact of tear size on<br>hemodynamics/flow jet   |

(continued on next page)

#### Table 8 (continued)

| Report                 | Year | Method(s) | Solver               | Flow<br>assumption | Wall<br>model | No. of<br>patients | Туре   | Main objective(s)  |
|------------------------|------|-----------|----------------------|--------------------|---------------|--------------------|--------|--|
| Luan et al. [907]      | 2023 | FSI       | Ansys                | Lam.               | Iso.          | 23                 | Туре В | Comparison of morphology/wall stress in<br>patients with dSINE (pre-dSINE and pre-/<br>post-operation follow-ups)                        |
| Deplano et al. [908]   | 2024 | FSI       | Ansys <sup>b,c</sup> | Lam.               | Iso.          | 1                  | Туре А | Comparison between FSI and CFD (with/<br>without rigid flap) models in patient with<br>residual dissection (post-operation<br>follow up) |
| Girardin et al. [909]  | 2024 | MBM       | Ansys <sup>b</sup>   | Lam.               | Iso.          | 1                  | Туре В | Impact of wall stiffness/graft length/graft<br>stiffness on hemodynamics post-<br>operation  |
| Schussnig et al. [899] | 2024 | FSI       | Deal.II              | Turb.              | Aniso.        | 1                  | Туре В | Impact of material models/non-<br>Newtonian flow on hemodynamics/wall<br>stress  |
| Bäumler et al. [342]   | 2024 | FSI       | SimVascular          | Lam.               | Aniso.        | 1                  | Туре В | Hemodynamics/wall stress in patient<br>(pre-/post-dissection follow-ups); impact<br>of flow jet on wall remodeling                       |

<sup>a</sup>Fluent; <sup>b</sup>CFX; <sup>c</sup>Mechanics; <sup>d</sup>Multiphysics; <sup>e</sup>Explicit; <sup>f</sup>Standard

- In addition to the requirements of CFD simulations, namely (i) a grid that resolves the domain and the solution with sufficient accuracy, (ii) suitable inflow and outflow conditions, and (iii) realistic rheological parameters, the computational modeling of aortic dissection using FSI requires (iv) reconstruction of the tissue domain and the pre-stress present in the tissue at time of model construction, (v) appropriate tissue models and related parameters, and (vi) consideration of external (visco-)elastic support.
- In the patient-specific context, it remains unclear whether FSI simulations are required or whether vessel compliance can be neglected. The often tremendously increased runtimes for FSI simulations compared with fixed-grid CFD, the difficult mesh construction and the large number of model parameters continue to make their use in clinical practice difficult. Current literature suggests that the difference in terms of biomarkers may be significant in some cases and negligible in others.
- Without accurately accounting for pre-stress, the anisotropic material behavior of the vessel wall across different tissue layers, and the *in vivo* external tissue support—factors that remain challenging to quantify the absolute stress magnitudes computed in FSI simulations have only limited value. However, these simulations can still provide useful insights by highlighting regions of stress concentration and indicating the relative stress distribution within the vessel wall.

## 8. In silico models on selected topics

In the previous sections, we examined the various *in silico* approaches for aortic dissection, grouped into primary categories such as multiscale tissue modeling, fluid dynamics approaches, and coupled FSI methods. Our objective in this section is to highlight studies with a specific research focus rather than the methodology employed. First, we discuss approaches based on computational solid mechanics (CSM) that investigate wall stress in patient-specific geometries and assess its connection to the tear location. Second, we provide a detailed analysis of stent-graft deployment and migration using an exclusively CSM-based method to study medical implant development and treatment improvement. We then explore thrombus formation models in aortic dissection in both CFD and FSI frameworks, which have the potential to examine, for example, the effects associated with deployed stent-grafts. This section provides a succinct overview of all existing techniques for modeling aortic dissection and highlights the likely impact of these emerging methods on clinical outcomes.

## 8.1. Patient-specific studies on wall mechanics

Several studies specifically targeting aortic dissection are based on CSM, which will be briefly discussed in the following section. In these investigations, patient-specific geometries of individuals affected by aortic dissection were employed to calculate inherent wall stress without incorporating the fluid domain adjacent to the solid domain. Simplified boundary conditions were utilized to account for hemodynamic forces.

In 2004, Beller et al. [910] conducted one of the first *in silico* studies aimed at a better understanding of aortic dissection. They measured the axial displacement of the aortic root in 40 cardiac patients with coronary artery heart diseases (excluding patients with aortic dissection) to investigate whether downward movement of the aortic root during the cardiac cycle could be responsible for the development of the circumferential tear observed in aortic dissection. In addition to the results from imaging, a finite element model of the human aortic root, aortic arch, and supra-aortic vessels was developed, with geometry comparable to a 3D MRI reconstruction. They analyzed the downward movement of the aortic root during the cardiac cycle, along with the resulting torsion and pressure exerted on the aortic wall. In the control model, a luminal pressure of 120 mmHg was applied, followed by an additional axial translation of 8.9 mm and a twist of 6°. To examine the impact of root motion versus hypertension on aortic wall stress, the analysis was also performed with a luminal pressure of 180 mmHg. The results revealed a critical increase in axial stress in the ascending aorta above the sinotubular junction, which may be responsible for the higher occurrence of circumferential intimal tears and aortic

dissections at this location. Of note, the right lateral aspect of the ascending aorta, several centimeters above the sinotubular junction, demonstrated increased axial stress with increased axial displacement, pressure, and aortic wall stiffness.

Doyle et al. [911] conducted a study on *in vivo* wall stresses in aortic dissection. Fifty patients and corresponding models were considered. The aim of the study was to estimate the stresses in the tissue more accurately using a patient-specific finite element analysis. To identify significant correlations, geometric parameters were measured and statistical analyses were performed. They revealed that the highest wall stresses were predominantly located in the ascending aorta and the aortic arch. Notably, these regions coincided with areas in which more than 80% of intimal entry tears occur in acute aortic dissection. This research provided valuable insight into the potential factors contributing to the development of aortic dissection and may serve as a basis for future preventive measures. Similar results were reported by Plonek et al. [912]. They studied the aorta both before and after dissection, focusing on predissection wall stress and the location of entry tears. After an initial study with a simplified geometry [913], they summarized two key findings: (i) a strong correlation between high wall stress and the areas where intimal tears occurred, and (ii) the peak wall stress predominantly concentrated in regions of abrupt geometrical changes rather than in the most dilated segments.

Menichini et al. [914] conducted a case study in which they examined two patients. Both patients had undergone TEVAR for type B dissection, with one patient in the chronic phase (patient 1) and the other in the subacute phase (patient 2). Interestingly, a dSINE was observed in patient 1, while the dissected aorta in patient 2 remained stable. Computational analysis revealed that the patient who experienced a dSINE had significantly higher stent-graft tortuosity compared with the patient who experienced this complication, and considerably elevated wall stress in the area next to the dSINE. This case study suggests that increased stent-graft tortuosity may be associated with higher wall stress, potentially leading to complications such as a dSINE in patients undergoing TEVAR for type B dissection.

In a study by Subramaniam et al. [915], they analyzed the distribution of systolic wall stress in two cases of proximal aortic dissection, employing finite element analysis to determine how the stress distribution was affected by the choice of anisotropic material models and aortic root motion. Specifically, they used the Fung-type and the HGO model. In the case of spiral dissection, they discovered that the peak stress correlated with the origin of the tear at the sinotubular junction. However, in another case involving root dissection, the highest stress was found at the distal end of the tear. Interestingly, they found a 9 to 11% increase in predicted tear pressure for root motions of up to 10 mm. Comparing the two material models, they noted that predicted tear pressure determined by the HGO model was 8 to 15% lower than the estimate of the Fung-type model. This study indicated that the selection of anisotropic material models and root motion could significantly influence both stress distribution and predicted tear pressure in aortic dissections. However, it should be noted that the stress distribution strongly depends on the selected material parameters, which makes further generalizations difficult. Such statements must therefore be based on carefully tuned tissue parameters.

Hu et al. [916] evaluated the wall stress distribution, particularly at the sites of proximal tears, in patient-specific type B dissection. The study utilized 30 patient-specific CTA images to reconstruct 3D models. The elastic behavior of the aortic wall was described by a third-order hyperelastic Ogden model, where the material parameters were obtained with data from [912]. Furthermore, inner blood pressure and aortic morphology were included in the simulations. Anatomical variables such as aortic angulation, aortic diameters, curvatures, and tortuosity were measured for each patient and compared with the computed wall stress. The results indicated that tears were associated with either locally high or low wall stress regions. Interestingly, the study also identified abnormal aortic angulation as a potentially crucial risk factor that could be monitored to predict aortic dissection.

In conclusion, the previously summarized *in silico* studies have demonstrated that local stress concentrations in patients with aortic dissection, located primarily in the ascending aorta and aortic arch, may be associated with the localization of entry tears [911,912]. Furthermore, these studies suggested that changes in anatomy due to the use of stent-graft deployment could lead to additional stress localization [914]. However, the presented studies appear to be limited due to the small number of patients involved, resulting in restricted geometric variations. Moreover, assessing the applicability of these findings becomes challenging when fluid flow is omitted, and instead, simplified boundary conditions are applied to the exterior or fluid loading is substituted without adequately investigating the implications of such simplifications.

# 8.2. Stent-graft deployment and migration

Simulations of stent-graft deployment have recently become significantly more important, particularly in the treatment of aortic dissection. The FEM including contact formulations is commonly used to replicate the biomechanical interactions between the stent-graft and the aorta during TEVAR procedures. Although virtual stent-graft deployment, possibly in combination with FSI, poses computational challenges, it has attracted increasing interest in recent years. The accuracy and potential of virtual stent-graft simulation models as interventional planning tools have already been demonstrated in the non-dissected aorta, see, e.g., [917–919]. Nonetheless, success in this area remains a challenging endeavor due to the complex anatomical structure of the dissected aorta and the nonlinear contact relationship between the stent-graft and the dissected aorta.

The first known study to assess aortic injuries caused by stent-grafts in aortic dissection was conducted by Ma et al. [920]. They examined the interaction between the aorta and a stent-graft, and identified risk factors for injury using CSM simulations. The study involved a patient with type B dissection who underwent TEVAR treatment with two different types of stent-grafts. The stent wire was modeled as an isotropic superelastic Nitinol, the graft was made from polyethylene terephthalate with a Young's modulus of 1840 MPa, and the aorta was modeled as an isotropic hyperelastic neo–Hookean material with a thickness of 1 mm and a Young's modulus of 5 MPa. The TEVAR simulations included compression, bending, and release of the stent-graft. The simulations were performed with or without coverage of the left subclavian artery, and each stent-graft deployment was performed seven times within the landing zone area for statistical analysis. The study found that in all TEVAR models, the maximum aortic stress was exerted on the proximal bare

stent caused by the angled contact between the proximal bare stent and the stent-graft.

Stent-graft deployment in a tortuous aorta, such as in aortic zones one to three, which refers to the arch and proximal descending aorta, resulted in significantly increased aortic stress compared with a straight artery. Kinking of the arch and proximal descending aorta resulted in incomplete stent-graft apposition at the concavity, resulting in a so-called bird-beak configuration [917]. The proximal bare stent in the stent-grafts generated higher stress toward the aortic wall than other parts of the stent-graft, and inadequate apposition between the stent-graft and aorta led to a characteristic maximum aortic stress distribution at the convexity. Higher oversizing ratios and the presence of a connecting bar also significantly increased the stress on the aortic wall after stent-graft implantation.

In a follow-up study, Meng et al. [921] then utilized the same patient-specific geometry to examine the biomechanical mechanisms behind new lesions induced by stent-graft implantation. They analyzed the relationship between the radial force and the spring-back force of the stent-graft, especially during virtual implantation with various oversizing ratios between 0 and 15%. The peak wall stress of the aorta was identified at the point where the proximal bare stent interacted with the aortic wall, with its value notably increasing by 62.2%. The simplified computational solid mechanics approach used in these studies did not account for variable blood pressure or viscous stresses, anisotropic material behavior of the aortic wall, variations in wall thicknesses, or outer wall boundary conditions, which might result in misleading and incorrect absolute stress values. However, given the complexity, the chosen approach still seems reasonable and the comparison of the relative values between different stent-grafts provided valuable insights.

Simultaneously, Yuan et al. [922] developed a framework for a patient-specific virtual stent-graft deployment model using FEM to analyze TEVAR-induced changes under physiological conditions. This framework was later extended by several follow-up studies. Preinterventional CTA images were acquired from an 80-year-old patient. The model incorporated a uniform wall thickness of 1.5 mm and a non-uniform thickness of the dissection flap as extracted from ECG-gated CTA. A second-order hyperelastic Ogden model fitted to circumferential tensile tests of aortic tissue [457] was used to model the aortic wall, while a linear elastic material model was used for the dissection flap. For the simulation, both the proximal and distal ends of the aorta were fixed and constant pressure was applied. The deployed TEVAR device consisted of two components: a metal stent and a fabric. For the quasi-static simulations, a general penalty contact approach was chosen and two different landing positions were simulated. Scenario I corresponded to the position adopted in the actual procedure. To understand the influence of the proximal landing position of the stent-graft on its biomechanical performance, scenario II was simulated in which the landing position was shifted 10mm toward the aortic root. The virtual stent-graft was crimped from its stress-free state onto the diameter of the delivery catheter. Both the proximal landing position and the assumed curvature of the virtual stent-graft were simulated before removing the virtual catheter. The simulation results were validated by comparing the simulated with the actual stent-graft position. Simulation of the real procedure showed that the proximal bare metal stent pushed the flap into the false lumen, leading to further stent-graft migration during the insertion phase. As shown in Fig. 68, an alternative landing position reduced local deformation of the dissection flap and prevented migration of the stent-graft. When deployed in the actual position, a higher maximum principal stress was found on the flap, while the alternative strategy reduced the stress. Note, however, they this study did not take into account the zero-stress configuration of the aortic tissue, the pre-stress of the aorta, pulsatile pressure, or viscous stresses, all of which could influence the final landing position and stent-graft migration. Nevertheless, the applicability in clinical support could be demonstrated using a comparatively inexpensive computational approach.

Building on previous work, Kan et al. [458] refined the model by incorporating a pre-stressing algorithm for the aortic wall, as described in [923]. CTA scans were obtained from a 44-year-old patient with type B dissection before TEVAR and three months after the procedure. Similar to the previous study, a uniform thickness of 1.5 mm and a dissection flap thickness between 0.6 and 1.6 mm were chosen. To determine the material parameters of the dissected aorta, 12 tissue samples were collected from five patients with type B dissection, as briefly summarized in Section 3.2.2. The experimental results were used to calibrate the material parameters of a Yeoh material model. A stent-graft was implanted with a proximal bare metal stent consisting of a Nitinol stent scaffold and a PET fabric graft. The mechanical behavior of the Nitinol stent-graft was reproduced using superelastic material properties, while the PET fabric was simplified by considering it as an isotropic elastic material. For illustration, the TEVAR deployment workflow is depicted in Fig. 69. When the stent-graft was deployed into the true lumen, it successfully sealed the proximal intimal tear. Coverage of the luminal surface by the stent-graft resulted in a change in loading conditions, with blood pressure at the aortic internal surface decreasing from 80 to 0 mmHg, and graft internal pressure increasing from 0 to 80 mmHg. The simulated post-TEVAR configuration was compared with follow-up CT scans and revealed a strong correlation. The mean deviations were 5.8% in the local open area, which was defined as the enclosed area of the unfolded stent-graft, and 4.6 mm in the stent-strut position, which referred to the position of the individual struts of the stent. The deployment of the stent-graft increased the maximum principal stress in the narrowed true lumen but reduced the stress in the entry tear region where false lumen dilatation occurred. Comparisons of simulation results with varying model complexities suggested that the pre-stress of the aortic wall and the blood pressure inside the stent-graft should be included for accurate prediction of the deformation of the deployed stent-graft.

In the follow-up study, Kan et al. [924] applied their numerical framework to a patient with type B dissection who underwent TEVAR during which a tapered self-expandable stent-graft was deployed into the true lumen, sealing the entry tear. Follow-up CTA scans were conducted at three months and 12 months post-TEVAR. The wall and dissection flap thicknesses were assumed to be constant at 1.45 mm, with model generation based on [448]. The outer wall was modeled as an isotropic hyperelastic material, while the flap was treated as a linear elastic material, similar to previous studies. Three stent-grafts of varying lengths were included in the virtual deployment simulation. The stent-graft consisted of stainless steel stent struts and a PET fabric graft, both modeled as linear elastic materials. Pre-stress was computed before simulations were carried out and contact between the stent-graft and virtual sheath was modeled using a general contact algorithm via a penalty formulation. The stent-graft was delivered to the target landing position determined by the centerline, as seen in the three-month follow-up scan. The model was able to predict the post-TEVAR stent-graft



**Fig. 68.** (a) Different stages of virtual stent-graft deployment from left to right: (b) post-interventional digital subtraction angiogram showing stent-graft migration; (c) *in silico* stent-graft model of the actual outcome (scenario I); and (d) landing position following placement of the stent-graft 10mm proximal (scenario II). The final configurations of the stent-grafts in both scenarios are projected onto the post-interventional digital subtraction angiogram. Figure is reprinted form Yuan et al. [922] with permission from Elsevier.

configuration and wall stress. Comparisons of simulation results for different stent-graft lengths revealed that the maximum stress location varied with stent-graft length. With the short stent-graft, which was actually deployed in the patient, the maximum von Mises stress was observed on the dissection flap at the distal landing zone, where a stent-induced new fenestration was identified at the three-month follow-up. Increasing the stent-graft length caused the maximum von Mises stress to shift away from the distal landing zone, reducing stress values by approximately 17% with the medium-length stent-graft and by 60% with the long stent-graft.

In their third consecutive study, Kan et al. [925] applied their developed computational model to investigate the impact of various stent-graft designs and lengths on wall stress and aortic morphology, particularly in relation to the 'spring-back' effect and its connection to dSINEs. Spring-back force measures the tendency of a curved stent-graft to return to its straight configuration [926,927]. They noted that specific design features of stent-graft products, such as the connecting bar, may increase bending stiffness and potentially increase the risk of tears at the dSINE. However, quantifying this effect remains a challenge. For their study, a patient-specific model of type B dissection (pre-TEVAR) was developed, which featured a single entry and exit tear with a uniform wall thickness of 1.45 mm in the outer wall of the aorta and the dissection flap. The aortic wall was modeled as a hyperelastic material using a second-order polynomial strain-energy function. Moreover, parametric models were developed for three different stent-graft products, each available in two lengths. These models incorporated various stent and graft materials, strut patterns, and assembly methods, each modeled individually. The study revealed that areas of high stress coincided with regions where tears of the dSINE



**Fig. 69.** Virtual stent-graft deployment: (a) valiant (Medtronic) stent-graft for TEVAR covered by a virtual sheath; (b) compressing the stent-graft into its crimped state by the virtual sheath; (c) curved tube employed to open the local narrowing in the compressed true lumen; (d) virtual delivery and deployment of the stent-graft at the intended location; and (e) overview of the deployment of stent-grafts in *in silico* with model variants highlighted. Figure is reprinted from Kan et al. [458] (licensed under CC-BY 4.0).

maximum principal stress on the dissection flap were analyzed and compared across five zones for short stent-grafts and seven zones for long stent-grafts. Increased tortuosity was observed at the proximal and distal ends, with reduced tortuosity in the intermediate zones. Of particular note was the third stent-graft model, which consisted of a single helically twisted ring and a tubular expanded synthetic graft, especially in its longer variant. This design effectively minimized the spring-back effect, led to a more uniform expansion of the true lumen, and significantly reduced stress in the distal landing zone. This model demonstrated significant improvements: it reduced the spring-back effect by 65%, achieved a more uniform dilatation of the true lumen by 70%, and lowered peak stress in the distal landing zone by 78%. Overall, these results, with less than 10% stent strut diameter accuracy in validation, underscore the potential of using virtual stent-graft deployment models in surgical planning. They also underlined that their modeling approach is recognized as a potential new technique for achieving optimal outcomes in endovascular procedures [126]. To show how the virtual stent-graft deployment model can serve as a pre-procedural planning tool, they proposed a timeline (49 hours) for the management of type B dissection with this *in silico* tool. For further information the reader is referred to [928,929].

In conclusion, all studies discussed simulate the effects of stent-graft deployment, virtual TEVAR procedures, and surgical planning on digital twins, which appear to have enormous potential. In addition, a related study by Zhou et al. [930], which was not discussed in depth but is worth mentioning, focused exclusively on the development of a stent-graft model for deployment in aortic dissection patients. Rapid technological advances combined with increasingly shorter product cycles lead to a substantial demand for reliable *in silico* validation methods that can be executed repeatedly and efficiently. Although the presented approach can still be substantially enhanced, particularly with regard to the refinement of boundary conditions, its integration with technological advances and clinical applications indicates a promising future.

## 8.3. Thrombus formation

In Section 3.4, we discussed the role of thrombus formation in aortic dissection and the relationship between false lumen status and patient mortality, dilatation, and hemodynamics. In brief, thrombus formation is a complex multistage biological process involving many species and biochemical reactions and encompasses the recruitment, activation, and adhesion of platelets to form a blood clot. Therefore, modeling such a process presents significant challenges. Various studies, mainly kinetic-based, have addressed specific stages with a range of assumptions, see [931] for a brief overview. While these studies provide valuable insight into thrombus formation, their focus on the micro or cellular level makes it difficult to apply these methods to an anatomically realistic model in which macro-scale hemodynamics strongly influence the coagulation process. To realistically simulate thrombus formation in combination with the flow field, most aortic dissection models to date are based on hemodynamic approaches, which simplify the complex process and make it applicable to patient-specific models based on the original work of Menichini and Xu [932].

The groundbreaking work of Menichini and Xu [932] introduces a hemodynamics-driven model that predicts thrombus formation in aortic dissection for the first time. This model was originally applied to a simplified aortic dissection geometry and uses shear rates, fluid residence time, and platelet distribution to assess the likelihood for thrombosis and its effect on blood flow. The primary features of the model and the key species transported have been established in a number of previous studies. A computational model for blood flow and thrombosis in patient-specific aortic dissection was first presented in [488] and then validated in [933]. Under physiological flow conditions, they demonstrated how the model utilizes shear stress, fluid residence time, and platelet distribution to predict thrombus formation in a non-moving lumen. Three type B dissection patients were considered in this study, in which predicted thrombus growth was compared with follow-up CT scans. Good agreement was found between the *in vivo* and *in silico* results, with the maximum difference between the predicted and measured false lumen reduction being less that 8%. To reduce the computational cost, accelerated kinetics are introduced into the model, which consequently led to a loss of connection between physical and computational time. Therefore, the need for correlation in time is required to fully bridge the time scales of thrombus formation and fluid flow. Nevertheless, the results were consistent with clinical studies indicating that exit tears and branching vessels are risk factors for incomplete thrombosis. Moreover, the distance between tears was found to be an important factor in determining the likelihood of false lumen thrombosis, which is of high clinical relevance.

The framework was then further utilized by Armour et al. [934] to investigate effects of post-TEVAR morphological features on false lumen thrombosis. Three patient-specific cases were considered where CT image data were employed to create models immediately after surgery and then annually for up to three years. The CFD model including thrombus formation was then applied to predict thrombosed regions based on the initial configurations and compared with the actual outcome at follow-up. For a representative patient case, Fig. 70(a) shows the flow velocity at peak systole, while Fig. 70(b) depicts the evolution of the lumen boundary during the modeling of thrombus growth. The high flow velocities in the branch vessels of the arch are noteworthy, appearing to be higher than in comparable studies; however, this is not further discussed in the study. With this follow-up data, the model was validated once again, yielding satisfactory agreement with discrepancies stemming from virtual occlusion of minor branch vessels only. The main goal of this study was to shed some light on the role of the distance from the entry tear to the first, more distal fenestration in false lumen thrombosis after TEVAR. Artificially altering the reconstructed flow domain, this distance was then adapted, yielding two additional



**Fig. 70.** Patient-specific (a) hemodynamics in aortic dissection post-TEVAR with streamlines at peak systole and (b) lumen boundary evolution during modeling of thrombus growth: (i) initial reconstructed geometry, (ii) to (iv) computational approximations, and (v) one-year follow-up showing good agreement between predicted and actual false lumen thrombosis. Figure is reprinted from Armour et al. [934] (licensed under CC-BY 4.0).

virtual models. Comparing all these configurations, Armour et al. [934] concluded that a greater distance between the entry tear and the first, more distal fenestration led to lower false lumen flow rates, reduced TAWSS, and thus ultimately resulted in increased thrombosis of the false lumen, which was found to be consistent with clinical observations. This virtual case study therefore suggested a TEVAR design that took into account the local false lumen morphology and reduces false lumen perfusion by enlarging the region of zero or slow flow in the proximal false lumen. Consequently, this computational study also supported the current understanding that the introduction of additional fenestrations during stent-graft placement can prevent false lumen thrombosis.

Armour et al. [805] aimed to enhance the understanding of the role of branch vessels in aortic dissection by studying two patients with type B dissection from the ADSORB trial [935]. One patient was post-TEVAR, while the other was pre-TEVAR. Branching vessels have been demonstrated to reduce false lumen thrombosis and decrease the likelihood of false lumen dilatation, as discussed in Section 3.4. Their investigation encompassed all branching vessels in the original model, along with additional fenestrations feeding the false lumen in the modified model. One model included the inferior mesenteric artery, while the other featured intercostal arteries. The results revealed increased flow and flow disturbance around branch vessels, leading to a rise in WSS in local branch areas. Depending on the model, there was a 2 to 10% increase in WSS, with intercostal vessels experiencing higher WSS due to their larger diameters. This, in turn, reduced the likelihood of thrombus formation. Overall, small areas of thrombus formation deviated from follow-up observations, which may be due to non-patient-specific boundary conditions as inlet and outlet values were derived from the literature. Inclusion of branched vessels in the model also resulted in reduced pressure in the false lumen.

Based on these investigations, Armour et al. [931] continued to investigate the effect of false lumen perfused branch vessels. Two patient-specific models were considered including the inferior mesenteric artery or intercostal arteries, which were meshed in a semiautomatic fashion based on CT data due to their small radii of 1.6 and 2.0 mm, respectively. In this study, boundary conditions were in part derived from the literature, since medical data were limited. Note here that the volumetric flow rates through the intercostal arteries were fixed at 5% of the inflow according to the literature. Then, a comparison of models considering or neglecting the false lumen perfused branch vessels was performed. Including the minor branch vessels, the authors revealed that the peak systolic flow velocity in the intimal tear increased by 13%, which leads to an increase of the volume-averaged TAWSS in the abdominal false lumen by 25.1%. Simultaneously, the true-to-false lumen pressure difference was not altered significantly. Including the inferior mesenteric artery in one of the cases led to a reduction in abdominal false lumen thrombosis of 15%, reducing the discrepancies between the simulation results and the one-year follow-up. This work thus confirmed anatomical studies on post-TEVAR false lumen thrombosis, which suggested that false lumen perfused side branches inhibit thrombus formation. This finding motivates further investigations on larger study cohorts.

In an initial study, Jafarinia et al. [936] examined the impact of hematocrit values on false lumen thrombosis using a 2D idealized geometry of a type B dissection, including an entry and an exit tear. The considered geometry was parameterized based on literature data, and a non-Newtonian fluid model incorporating hematocrit values was considered. Their findings indicated that elevated hematocrit levels hinder thrombus formation due to rheological factors, suggesting that patients with higher hematocrit values may have a reduced likelihood of experiencing complete false lumen thrombosis. They explicitly underlined a limitation of their study: it only considered a single geometry. Hemodynamics, defined by the interaction of flow and rheology and a primary driver of thrombosis, is significantly influenced by geometry. Thus, a different geometry might yield different results. The same consideration applies to hematocrit levels, or, to put it another way, the concentration of red blood cells. Subsequently, Jafarinia et al. [937] utilized the previously established geometry to explore the impact of morphological parameters on false lumen thrombosis, providing further insight into the factors influencing false lumen thrombosis. In doing so, the major limitation of the previous study, which was its use of only a single geometry, was addressed. With a fully automatic pipeline from geometry generation and CFD simulation including the thrombosis model and evaluation, a global sensitivity analysis of input parameters was performed. This sensitivity analysis included selected morphological parameters such as the size of the tears, the diameter and length of the false lumen, the diameter of the true lumen, and the location of the entry tear. The study also introduced dimensionless morphological parameters to generalize the results. To ensure sufficient representation of the input variability of the chosen geometry, the input sample space of the global sensitivity analysis was set to a dimension of 4000. The sensitivity analysis revealed that the false lumen diameter and the size and location of intimal tears were the most sensitive parameters influencing false lumen thrombosis. A higher risk of partial thrombosis was observed when the false lumen diameter was larger than the true lumen diameter. Reducing the ratio of entry to exit tear size increased the risk of false lumen patency. These parameters played a dominant role in classifying morphologies into patent, partially thrombosed, and fully thrombosed false lumen. Although simplified morphologies of the dissected aorta were used, the results provide promising insight into the mechanisms underlying false lumen thrombosis in type B dissection. This study therefore highlighted the predictive role of morphological parameters for false lumen thrombosis in type B dissections, which is consistent with clinical observations.

Due to the time-consuming nature of the hemodynamics-driven model by Menichini and Xu [932], Jafarinia et al. [938] sought to simplify the model. They significantly reduced the number of components involved, making thrombus growth predictions 65% more efficient. This proposed model is based on a previous study by Melito et al. [939], who performed a sensitivity analysis of the original model [932], and revealed that not all parameters influenced thrombus formation equally, suggesting model could be simplified. In patients with type B dissection, thrombus formation was predicted with this simplified model in patient-specific geometries obtained one month after TEVAR and compared with the *in vivo* configuration observed three years later. The *in silico* predictions closely matched the actual clinical observations in terms of location and overall thrombosed volume, validating the model. Although the direct link to physical time is lost, as seen in related models, future work could potentially assist in estimating time correlations.

As discussed in Section 7.1, Chong et al. [885] developed the first FSI model that incorporated thrombus formation in a simplified geometry of type B dissection, which was based on the thrombus formation model of Menichini and Xu [932]. Despite knowledge of fluid–solid interaction, reducing flap mobility by increasing the Young's modulus of the flap slows thrombus growth. In fact, when

compared with the rigid model, the predicted thrombus volume is 25% larger using the FSI-thrombosis model with a relatively mobile flap, which highlighted the importance of incorporating thrombus formation in FSI modeling. Moreover, they reported vortices near the tears caused by drastic flap motion, which is an example of hemodynamic behavior only seen when considering FSI.

In the recent research by Wang et al. [940], the significance of low WSS in the initiation and growth of thrombus was explored. In the past, this aspect has been highlighted in many studies, but the consequence of thrombus breakdown due to increased WSS has been widely overlooked. To fill this gap, they proposed a novel WSS-induced thrombus breakdown function, which was integrated into the previously developed hemodynamics-based thrombosis model. The performance of the enhanced model was assessed by quantitative comparison with experimental *in vitro* data on thrombus formation [941] in a backward-facing step geometry. Moreover, a qualitative comparison with *in vivo* data from a type B dissection patient with post-TEVAR follow-up was performed. Remarkably, incorporating the aspect of thrombus breakdown considerably improved the accuracy in predicting thrombus volume. In instances like the backward-facing step geometry, this inclusion revealed that thrombus breakdown hinders its expansion over the step and downstream, thereby enabling a stable thrombus formation at a quicker rate. When compared with the original model that omits thrombus breakdown, the refined model showed a better alignment with experimental measurements in terms of thrombus volume, height, and length. Conclusively, their study underscores that the effect of thrombus breakdown warrants its inclusion in computational thrombosis models.

Despite the computational framework proposed by Menichini [533], and its recent simplification by Jafarinia et al. [938] and further extension to model thrombus breakdown by Wang et al. [940], several other frameworks have been employed to model thrombus formation in aortic dissection. One of which was proposed by Yazdani et al. [489]. They aimed to find out why some false lumina thrombose entirely, while others do so only partially or not at all. Using a data-driven particle-continuum model [942], they examined thrombus formation in a murine model of aortic dissection. This dissection was induced in adult male  $Apoe^{-/-}$  mice by a standard angiotensin II subcutaneous infusion at a rate of 1,000 ng/kg/min. The 3D geometric model was derived from multimodal imaging that integrated 3D ultrasound with optical coherence tomography. Moreover, the average flow velocities were determined using pulsed wave Doppler and anatomic 3D ultrasound [469]. They analyzed three types of aortic dissection that did not rupture: a large false lumen with minimal thrombus, a moderate false lumen with significant thrombus, and one filled primarily with thrombus (Fig. 71(a)). Their computational model incorporated the two-way coupling of Lagrangian particle transport with blood flow. They studied the motion of platelets within the flow and their adherence to the damaged dissection surface by merging the spectral/hp element method (SEM) [943] with a force coupling method, similar to the previous work of Pivkin et al. [944]. Drawing inspiration from previous research, platelets in their model were perceived to exist in one of three states: passive, triggered, or activated [944], see Fig. 71(b). Both passive and triggered platelets were classified as non-adhesive. However, when a passive platelet interacts with an activated platelet or a damaged luminal boundary, it was triggered and activated after a short delay. This activated state allows the



**Fig. 71.** (a) Reconstructed geometries from 3D ultrasound and coherence tomography image stacks show dissections of the suprarenal aorta in  $Apoe^{-/-}$  mice: (i) large; (ii) medium; and (iii) small false lumen [469]. The red-shaded volume indicates the thrombus, the pink volume represents the false lumen with flowing blood, and the gray represents the patent vessel. Blood flow is directed from the proximal end (right side) to the distal end (left side). (b) Illustration of results showing three stages of particle distribution within the large false lumen at specific moments during the simulation: (i) initial particle arrangement; (ii) intermediate phase, which concludes diastole; and (iii) final phase during peak systole. Selected streamlines are shown, with their colors indicating velocity magnitude. The particle color codes are as follows: passive particles (green); activated or triggered particles (red); and active, adhered particles (yellow). For better visualization, not all particles are presented and the flow direction runs from left to right. Figure is adapted from Yazdani et al. [489] (licensed under CC-BY 4.0).

platelets and associated fibrin to grow in size. This approach allowed them to use fewer platelets than were normally observed at physiological concentrations, while still simulating the transformation of the clot in size and shape upon interaction with injured walls or other platelets. Moreover, in their methodology they divided the thrombus formation process into two stages. Initially, a thrombus formed and developed in the false lumen, occurring minutes to hours after dissection. This was followed by the expansion and remodeling phase of the clot, which was primarily driven by fibrin degradation and collagen production, and could last from days to weeks. They focused on the acute phase, in which they thoroughly analyzed the platelet activation, aggregation, and formation of the primary fibrin network. The resulting numerical data emphasized the significance of both geometry and local hemodynamics in determining the acute progression of a thrombus. Even considering the inherent geometrical differences between murine and human dissections, mouse models are invaluable due to their consistency. Their results showed that false lumina with different geometries resulted in greater thrombus formation in areas with lower bulk shear rates. In contrast, regions with high shear or intense vortices were less favorable for the formation of dense thrombi.

Aiming to better understand the role of proteins in the coagulation cascade for thrombosis, Wang et al. [945], in their study, specifically focus on the activation and stabilization of platelets in type B dissection. They introduced a reduced-order fluid-chemical model to simulate the coagulation cascade in a simplified geometry of a two-year aortic dissection and investigate the driving mechanisms of false lumen thrombosis. The coagulation cascade model built on the work of Bodnár and Sequeira [946] and is an extension of the model originally proposed by Anand et al. [947,948]. The model demonstrated a consistently high level of fibrin at the top of the false lumen and in some time-varying areas between the two tears, suggesting a high probability of thrombus formation at these locations. Furthermore, the model showed that the temporal evolution of coagulation factors is strongly influenced by local hemodynamics, with periodic changes in the high disturbance zone aligning with the flow field.

A novel approach to modeling thrombus formation was proposed using a computational model for blood stasis, see the work of Jiang et al. [949]. This model employed the two-fluid principle to monitor the spatial distribution of residual blood over time. The focus of the study was four patients with type B dissection who underwent TEVAR and had persistent false lumina. Two years post-TEVAR, follow-up *in vivo* observations were conducted. In contrast to previous modeling approaches [932], in this model a specific point in time for a fully developed hemodynamic flow field was identified. 'New blood' was then defined as the inlet substance that had similar physical properties to the existing 'old blood', such as density and viscosity, which corresponds to the classic two-fluid or phase model. The volume of fluid surface tracking technology was used to solve the hemodynamic field with two fluids. This made it possible to track the location and shape of the new and old blood interfaces over time by resolving the volume fraction equation in each computational cell, thereby identifying the location of remaining old blood and the area of blood stasis. Their findings revealed that the remaining false lumen was located below the highest tears and its span was closely associated with the distance between tears. Due to the size and location of the tears, a high probability of thrombosis was suggested if a majority of the blood remained in the false lumen. Importantly, the model-predicted positions and topologies of residual blood in the false lumen showed a strong correlation with the *in vivo* observations of thrombus.

On the basis of the theory of porous media, Gupta and Schanz [950] recently presented an approach to modeling thrombus growth and formation in aortic dissections using a multiphasic framework. This macroscopic continuum mechanical approach provides a tool to describe the complex microstructure of thrombi. This was achieved by defining a representative elementary volume in which individual constituents – solid, liquid, and nutrient-rich phases – are considered to be in a state of ideal disarrangement. Through averaging processes over this representative elementary volume, the microscale information of the overall aggregate and its constituents was homogenized into macro-scale quantities. Generally, the thrombus growth process is driven by chemical, mechanical, and metabolic factors. Due to the multiphasic nature of the thrombus, they presented a triphasic model that integrated a solid phase saturated by fluid, with the fluid itself comprising both liquid and nutrients. Since the effects of blood velocity and nutrients on thrombus growth were well described in the literature, they presented a velocity and nutrient concentration-induced thrombus growth model based on the theory of porous media. After outlining the theoretical framework and its implementation, they applied the developed model to a 2D cross-section of a false lumen derived from patient-specific geometry and used simplified boundary conditions. This model emphasizes that thrombus formation occurred in regions of reduced blood velocities. Material parameters have been proven to play a central role, as several parameter studies have shown. In conclusion, although their findings are promising, further advances are of pressing need, especially with regard to material parameters and boundary conditions, to gain clearer insights into thrombus formation and growth based on this multiphasic approach. Nevertheless, their work represented a novel and intriguing method for modeling thrombus growth and formation in aortic dissection through the theory of porous media.

In summary, several studies over the past decade have modeled thrombus formation using both simplified and patient-specific geometries. Most of these studies employ a hemodynamics-based approach [932], which has demonstrated simulation results that agree well with medical data. However, this method remains computationally demanding. Notably, a recent model reduction study further optimized this shear-driven thrombosis model and significantly improved computational efficiency by reducing computation time by 65% in a representative example [938]. The currently available thrombus formation models exhibit good agreement with medical data, but the computational requirements are typically higher than for fixed-grid CFD approaches.

#### 8.4. Key findings and publication index

To summarize this section, we present the publication index of *in silico* models on selected topics in Table 9, with a focus on CSM simulations of arterial tissue stress, the deployment and migration of stent-grafts, and simulations of thrombus formation in aortic dissections. Due to the large number of studies available, only a selected range was discussed; however, this table covers an even broader range of studies. Essential information about the modeling approaches and patient cases as well as the objectives of each study

## Table 9

Publication index of patient-specific *in silico* models on selected topics relating to aortic dissection, more precisely CSM models on wall mechanics, deployment and migration of stent-grafts, and thrombus formation, ordered by year of publication (Lam.: laminar; Iso.: isotropic; Aniso.: anisotropic).

| Report                                     | Year      | Method(s) | Solver              | Flow<br>assumption | Wall<br>model | No. of patients | Туре   | Main objective(s)   |
|--|-----------|-----------|---------------------|--------------------|---------------|-----------------|--------|---|
| Patient-specific studies on wall mechanics |           |           |                     |                    |               |                 |        |   |
| Beller et al. [910]                        | 2004      | CSM       | Ansys <sup>c</sup>  | -                  | Iso.          | -               | -      | Impact of aortic root motion on wall<br>stress in simplified healthy aorta  |
| Doyle et al. [911]                         | 2013      | CSM       | Abaqus <sup>f</sup> | -                  | Iso.          | 50              | -      | Impact of aortic geometry on wall stress<br>in healthy aorta (implications for<br>discretion)   |
| Plonek et al. [913]                        | 2017      | CSM       | Ansys <sup>c</sup>  | -                  | Iso.          | -               | -      | Impact of blood pressure/axial stretch on<br>wall stress in simplified aortic aneurysm<br>(implications for dissection)                     |
| Plonek et al. [912]                        | 2018      | CSM       | Ansys <sup>c</sup>  | _                  | Iso.          | 4               | Туре А | Comparison between wall stress pre-<br>dissection and intimal tear (location)   |
| Menichini et al. [914]                     | 2018      | CSM       | Ansys <sup>c</sup>  | -                  | Iso.          | 2               | Туре В | Comparison between wall stress post-<br>TEVAR and formation of dSINEs   |
| Houben et al. [338]                        | 2020      | CSM       | CRIMSON             | -                  | Iso.          | 3               | -      | Comparison between wall stress pre-<br>dissection and focal area of growth  |
| Subramaniam et al. [915]                   | 2021      | CSM       | FEBio               | -                  | Aniso.        | 2               | Туре А | Impact of aortic root motion/material models on wall stress   |
| Hu et al. [916]                            | 2021      | CSM       | Abaqus <sup>f</sup> | -                  | Iso.          | 30              | Туре В | Comparison between morphological parameters and wall stress   |
| Stent-graft deployment and                 | migration |           |                     |                    |               |                 |        | I   |
| Ma et al. [920]                            | 2018      | CSM       | Abaqus <sup>e</sup> | -                  | Iso.          | 1               | Туре В | Impact of stent-graft with/without<br>connecting bar on wall stress   |
| Zhou et al. [930]                          | 2019      | CSM       | Abaqus              | -                  | -             | -               | -      | Development of stent-graft for TEVAR; validation to <i>in vitro</i> experiments   |
| Meng et al. [921]                          | 2020      | CSM       | Abaqus <sup>e</sup> | -                  | Iso.          | 1               | Туре В | Impact of stent-graft oversizing ratio on wall stress   |
| Yuan et al. [922]                          | 2020      | CSM       | Abaqus <sup>e</sup> | -                  | Iso.          | 1               | Туре А | Impact of stent-graft landing zone on<br>wall stress/stent-graft migration  |
| Kan et al. [924]                           | 2021      | CSM       | Abaqus <sup>e</sup> | -                  | Iso.          | 1               | Туре В | Validation of stent-graft deployment and<br>migration framework to <i>in vivo</i> data<br>(post-TEVAR follow-ups)                           |
| Kan et al. [458]                           | 2021      | CSM       | Abaqus <sup>e</sup> | -                  | Iso.          | 1               | Type B | Impact of stent-graft length on wall stress   |
| Kan et al. [925]                           | 2024      | CSM       | Abaqus <sup>e</sup> | -                  | Iso.          | 1               | Туре В | Impact of stent-graft length/model on<br>wall stress/stent-graft migration  |
| Li et al. [951]                            | 2024      | CSM       | Abaqus <sup>e</sup> | -                  | Iso.          | 2               | Туре В | Comparison between wall stress in<br>patients with/without distal stent-graft<br>induced new tears  |
| Thrombus formation                         |           |           | h                   |                    |               |                 |        |   |
| Menichini & Xu [932]                       | 2016      | CFD       | Ansys               | Lam.               | -             | -               | Туре В | Development of hemodynamics-based<br>thrombus model; application to<br>simplified models  |
| Menichini et al. [488]                     | 2016      | CFD       | Ansvs <sup>b</sup>  | Lam.               | _             | 14              | Type B | Validation to in vivo data (follow-ups)   |
| Menichini et al. [933]                     | 2017      | CFD       | Ansys <sup>b</sup>  | Lam.               | -             | 3               | Туре В | Validation to <i>in vivo</i> data (post-TEVAR follow-ups)   |
| Yazdani et al. [489]                       | 2018      | SEM       | n.a.                | n.a.               | -             | 3 (Mouse)       | Туре В | Development of data-driven particle-<br>continuum thrombosis model; validation<br>to <i>in vivo</i> data (follow-ups)                       |
| Armour et al. [934]                        | 2020      | CFD       | Ansys <sup>b</sup>  | Lam.               | -             | 3               | Туре В | Impact of stent-graft end-to-distal tear<br>distance on thrombus formation;<br>validation to <i>in vivo</i> data (post-TEVAR<br>follow-ups) |
| Jafarinia et al. [936,937]                 | 2020/23   | CFD       | OpenFOAM            | Lam.               | -             | -               | Туре В | Impact of blood rheology/morphological<br>parameters on thrombus formation in<br>simplified models  |
| Wang et al. [945]                          | 2021      | CFD       | OpenFOAM            | Lam.               | -             | -               | Туре В | Development of fluid-chemical thrombus model; application to simplified models  |
| Jiang et al. [949]                         | 2022      | CFD       | Ansys <sup>a</sup>  | Lam.               | -             | 4               | Туре В | Development of two-fluid blood stasis<br>thrombosis model; validation to <i>in vivo</i><br>data (post-TEVAR follow-ups)                     |
| Chong et al. [885]                         | 2022      | FSI       | COMSOL <sup>d</sup> | Lam.               | Iso.          | -               | Туре В | Comparison of thrombus formation<br>between simplified CFD and FSI models   |
| Armour et al. [931]                        | 2022      | CFD       | Ansys <sup>b</sup>  | Lam.               | -             | 2               | Туре В | Impact of false lumen perfused side<br>branches on thrombus formation in pre-/<br>post-TEVAR patients                                       |

(continued on next page)

#### Table 9 (continued)

| Report                 | Year | Method(s) | Solver             | Flow<br>assumption | Wall<br>model | No. of patients | Туре               | Main objective(s)   |
|------------------------|------|-----------|--------------------|--------------------|---------------|-----------------|--------------------|---|
| Jafarinia et al. [938] | 2022 | CFD       | Ansys <sup>b</sup> | Lam.               | -             | 1               | Туре В             | Development of simplified<br>hemodynamics-based thrombosis model;<br>validation to <i>in vivo</i> data (post-TEVAR<br>follow-ups) |
| Wang et al. [952]      | 2023 | CFD       | OpenFOAM           | Lam.               | -             | 1               | Туре В             | Development of multiconstituent<br>thrombosis model; validation to<br>experiments/ <i>in vivo</i> data (post-TEVAR<br>follow-ups) |
| Wang et al. [940]      | 2023 | CFD       | Ansys <sup>b</sup> | Lam.               | -             | 1               | Туре В             | Development of WSS-induced thrombus<br>breakdown model; validation to <i>in vivo</i><br>data (post-TEVAR follow-ups)              |
| Gupta & Schanz [950]   | 2023 | CSM       | PANDAS             | -                  | -             | -               | Туре В             | Development of multiphasic approach<br>for thrombus modeling based on theory<br>of porous media                                   |
| Pei et al. [953]       | 2024 | CFD       | Ansys <sup>a</sup> | Lam.               | -             | 8               | Coronary<br>artery | Impact of tear direction on<br>hemodynamics in iatrogenic coronary<br>artery dissection   |

<sup>a</sup>Fluent; <sup>b</sup>CFX; <sup>c</sup>Mechanics; <sup>d</sup>Multiphysics; <sup>e</sup>Explicit; <sup>f</sup>Standard

is included. Finally, the most important findings are listed conclusively.

- In patient-specific aortic dissection, localized stress concentrations, particularly in the ascending aorta and aortic arch, may be associated with the locations of intimal tears.
- TEVAR procedures increase wall stress in the narrowed true lumen, but at the same time reduce it in the entry tear area, where false lumen dilatation typically occurs. Moreover, an analysis of varying stent-graft lengths indicated that the location of maximum wall stress shifts with the length of the stent-graft. In general, CSM models are effective in capturing stent-graft positioning in patients undergoing TEVAR.
- Current thrombus formation models can provide good predictions and high accuracy in predicting thrombus formation for patients. The likelihood of thrombus formation is elevated due to factors such as the absence of an exit tear, a larger distance between the entry tear and other fenestrations, a larger false lumen diameter, and smaller intimal tears. However, the interplay between these factors and possibly many more associated with thrombus formation is complex and not yet fully understood.

# 9. Conclusion and future perspectives

## 9.1. Conclusion

The previous sections highlighted the extensive research efforts undertaken to uncover the pathological changes of aortic dissection using various approaches. As demonstrated, well-constructed multiscale material and *in silico* models can serve as versatile tools for bioengineers and clinicians, allowing them to explore unsolved problems. This contributes to a deeper understanding of aortic dissection, particularly with regard to specific phenomena.

In Section 2, following a brief overview of the biomechanics of the normal aorta – including details of anatomy and wall microstructure - an introduction to aortic dissection was provided. This involved a definition of the disease, commonly used classification systems, a summary of modern imaging techniques and treatment methods for aortic dissection patients, and aortopathies related to aortic dissection. The purpose of this section was to familiarize the reader with the fundamental aspects of aortic dissection. We then presented medical data focusing on the pathological remodeling of aortic dissection, including both the anatomy and the wall microstructure of the aorta in Section 3. This included an examination of factors such as the extent and configuration of the dissection flap and the dilatation and growth of the aorta. Furthermore, the microstructure and mechanical behavior of the diseased aortic wall were described. The section concluded with a discussion of altered hemodynamics in the true and false lumina, including blood flow patterns and local pressure throughout the cardiac cycle, as well as the role of thrombus formation. Subsequently, experimental models for aortic dissection were discussed in Section 4, including in vivo, ex vivo, and in vitro studies on animal, human, and phantom models. These models have played an important role in uncovering biomechanical mechanisms, particularly in understanding the hemodynamics during deterioration of aortic anatomy, microstructural changes, and evaluating the effectiveness of medical implants. Section 5 then focused on multiscale material models for damage and failure. These models were developed to model wall delamination, dissection propagation and flap configuration, the effects of focal areas of medial degeneration, and growth and remodeling in the newly formed tissue layers. The following three sections provided an overview of in silico models of hemodynamics (Section 6), FSI (Section 7), and selected topics (Section 8). To the best of the authors' knowledge, this is an almost complete synthesis of the literature in this area. Each section contains key findings and the publication indices for the available models. To provide readers with a comprehensive overview of publication trends in recent years, the number of reports on multiscale material models on damage and failure and on in silico models using patient data on aortic dissection is presented in Fig. 72. The trend underscores a significant increase



Fig. 72. Annual number of reports on multiscale materials and *in silico* models incorporating patient data related to aortic dissection (total 227 reports). The data for the year 2024 (hatched bar) remains incomplete.

in the total number of models, especially in the last decade. Particularly in recent years, the number of reports on *in silico* models, particularly those addressing hemodynamics in patient-specific models of aortic dissection, has increased rapidly.

## 9.2. Future perspective

## 9.2.1. Disease-related modeling aspects

Considering future developments, it is clear that there are numerous challenges and unresolved problems in the field of diseaserelated modeling of aortic dissection, particularly with regard to the key factors that contribute to the onset and progression of aortic dissection. It is noteworthy that these factors often dominate in different stages of aortic dissection, from the acute to the chronic phase, as shown in Fig. 73.

As discussed in previous sections, given the remarkable robustness of a normal, healthy aorta, an abnormal aortic wall is likely a major factor in triggering aortic dissection. Multiscale material models have been developed to improve our understanding of specific pathological changes in aortic dissection. In fact, pathological changes in aortic dissection, as in other diseases, cannot be attributed to a single alteration, but rather are correlated [717]. For example, the accumulation of GAGs can cause swelling pressure that can lead to the degradation of interlamellar, radially-oriented elastic fibers. Consequently, SMCs connected to elastic lamellae by elastic fibers could lose their microstructural integrity, which could lead to apoptosis and dysfunction [49]. Although the exact mechanism is not yet fully understood, it could be crucial to couple and investigate these and similar mechanisms. Against this background, the inclusion of mechanobiology and related pathological changes in mechanical models could also be of great importance, especially since none of the material models presented and discussed so far have taken this aspect into account. In this context, central areas of medial degeneration, in other words local material or geometrical inhomogeneities that arise or develop due to pathological changes, should be further investigated for their possible contribution to the initiation and progression of aortic dissection. It is truly fascinating to study the influence of local GAG accumulation [103] or the degradation of elastic fibers [717] on the material behavior at the macroscopic scale in patient-specific geometries. The raises the question of whether these localized pathological changes may accumulate over time, ultimately and triggering aortic dissection, potentially explaining why certain pathological events, such as the formation of intimal tears, tend to recur in specific region. Moreover, as recently highlighted by Myneni and Rajagopal [114], even the normal, healthy aortic wall exhibit significant inhomogeneities that must be considered. While it may be difficult to obtain this information from the entire aorta through traditional microstructural investigations such as second harmonic generation microscopy, future advances may enable non-invasive acquisition using, e.g., MRI imaging [117,118].

From the initial onset to the acute phase and possibly the chronic phase, aortic dissection typically propagates over time. However, most of this propagation usually occurs early or immediately after the hyperacute phase. The progression of an aortic dissection can vary depending on the specific geometry of the aorta and the location of the intimal tear, with antegrade movement being more common, but retrograde movement also occurring in some cases. In certain cases a pronounced helical pattern may be observed, but in others it is not. Using *in silico* models, these variations can be comprehensively studied, and in the near future, patient-specific geometries could also be examined, initially through the use of CSM models. Different geometries could be explored, such as variations in the branch vessel, tortuosity, or unusual geometrical features. Furthermore, the role of the microstructure, particularly the alignment of collagen fibers along the aorta, can be further investigated.

In addition to axial propagation, the aortic dissection or false lumen also extends circumferentially. The circumferential propagation of the false lumen occurs primarily at the onset of the disease and typically progresses only slowly thereafter. In fact, the circumferential extent of the false lumen varies among patients. We hypothesize that the elasticity of the aortic wall, particularly within the intima and media, plays a pivotal role in circumferential propagation. Localized defects in the aortic wall, such as small



**Fig. 73.** Visualization of future perspectives in disease-related modeling with respect to aortic dissection. The monochromatic coloring scheme illustrates the importance of the specific aspects during the acute, subacute, and chronic phases, and their progression over time. For visualization reasons, the hyperacute and subacute phases are neglected.

initial tears or local separations due to accumulated GAGs, may result in delamination of the aortic wall if the interlamellar strength between layers is compromised. Subsequently, the wall undergoes elastic recoil, releasing the stored elastic energy in the process. The segment of the aortic wall that becomes the dissection flap may tend to return to its stress-free configuration to relieve tension and allow blood flow into the false lumen. Consequently, the elasticity of the aortic wall could determine the circumferential extent of the false lumen, with a small newly formed false lumen in very stiff aortic walls, as in older adults, or the false lumen covers (almost) the entire circumference of the true lumen in very elastic aortic walls, as in young adults. Nevertheless, further research is necessary to evaluate this hypothesis and to fully understand the correlation between aortic wall elasticity and the extent and associated configuration of the dissection flap.

At dissection initiation, one or more intimal tears typically form. With each cardiac cycle, blood is pushed through these tears into the false lumen and flows back. This process creates a flow jet and a pressure difference between the lumina, both depending on the size, location, and number of tears. The role of this flow jet and the pressure difference remains a controversial topic, particularly regarding how both factors evolve from the onset to the chronic phases. While the flow jet is often considered a factor in dissection progression, it is uncertain whether the local increase in pressure can further delaminate the aortic wall. A recent computational study [612] demonstrated that the flow jet induce a local increase in WSS and pressure on the aortic wall. However, there is no definitive evidence to suggest that the flow jet is sufficiently strong to initiate aortic wall delamination. Instead, local wall remodeling may occur as the disease progresses. In contrast, the pressure difference between the lumina can have a significant impact on dissection, particularly the pressure in the false lumen, which can pull the outer wall away from the aorta and promote delamination. In addition, the cyclic pressure difference between the lumina may further aggravate progression of the dissection. Further investigations are required to better understand these phenomena.

Next, we would like to emphasize the importance of considering the growth and remodeling of the dissected aortic wall throughout the progression of the disease from onset to the chronic phase. This aspect has received little attention in the context of aortic dissection, as shown in Section 5.4. Growth and remodeling frameworks have recently received considerable attention in the field of cardiovascular modeling in general [954]. As already mentioned, in aortic dissection, the aortic wall can be divided into three distinct layers, namely the outer wall of the true lumen, the dissection flap, and the outer wall of the false lumen. However, the microstructural composition of these layers varies depending on the individual patient case. In the event of a delamination between the media and adventitia, it may happen that the outer wall of the false lumen consists only of the adventitia, but can also include parts of the media or even the entire media. Other geometrical factors may also play a crucial role, but these remain to be determined. Furthermore, these layers may remodel over the course of disease and during the transition from the acute to the chronic phase. This remodeling may be due to changes in the mechanical stress of the respective tissue layers, which trigger remodeling processes, or due to inflammatory processes resulting from the delamination-induced injury response. In fact, regions of altered, usually increased WSSs are generally associated with remodeling of the aortic wall, leading to altered mechanical behavior, particularly with greater elastin degradation. Ultimately, this often results in local wall dilatation, which increases the risk of rupture. Evidence for this has been shown in both in clinical studies using advanced medical image techniques and in computational studies of patients with aortic aneurysms [422,736,955–957], bicuspid aortic valves [958–960], aortic stenosis [961], or intracranial aneurysms [962–966], among others.

As the disease progresses, the dissection flap stiffens, and the outer wall of the false lumen thickens, as discussed in Section 3.2. Moreover, the outer wall of the false lumen becomes more compliant under the prevailing luminal pressure, leading to dilatation of the

false lumen. However, it remains largely unknown how these different regions remodel and what microstructural changes occur. Nevertheless, understanding these changes could potentially enhance patient treatment, as the timing of surgical intervention often depends on remodeling [283–285]. In short, while increased stability could make TEVAR safer, it is important to note that a dissection flap in its most pliable state offers the greatest opportunity for complete remodeling, see Sections 3.1 and 3.2. Growth and remodeling models have been applied to many aspects of vascular disease [237], but there remains an urgent need to consider these effects in dissection.

To date, aortic elongation and its relationship to aortic dissection have not been extensively studied. The connection between an elongated aorta – a condition often associated with aging – and the onset and progression of the disease remains however unclear [967]. This is especially true considering that elongation can also result in pronounced angulation and increased tortuosity. It remains to be determined whether the disease causes aortic elongation or if elongation itself contributes to the onset and progression of the disease, or, in other words, whether it develops during the chronic phase. Computational models can be a versatile tool to study the influence of axial growth on dissection progression. Studies have indicated that the ascending aorta and the aortic arch elongate, affecting not only hemodynamics but also wall stresses. In addition, as highlighted by Poullis et al. [968] and later discussed in [969], aortic elongation can occur even without dilatation. Their study, which was based on a mathematical model, suggested that aortic curvature may be a more critical factor leading to aortic dissection than aortic diameter, blood pressure, and other factors. The greater aortic curvature may explain why normal diameter aortas can dissect and also suggests that the location of the entry tear may be predictable, although this has not yet received much attention. Moreover, with respect to vessel tortuosity, it has been shown to be related to other disease, e.g., tortuosity is a well-documented challenge in coronary arteries [967,970]. There is strong evidence of a causal relationship between the tortuosity of carotid arteries and stroke [971] or connective tissue disease [972]. In addition, computational studies have been developed to study vessel tortuosity, but there are only a few, see, e.g., [346,973].

For the development of novel material models, access to experimental data from mechanical tests and microstructural examinations on both healthy and diseased tissue is crucial. Although some experimental data are available, they are still insufficient, as briefly summarized in Section 4. In particular, there are insufficient data on the mechanical behavior, dispersion, cross-linking, and alignment of fibrous constituents in the different tissue layers: outer wall of the true lumen, dissection flap, and outer wall of the false lumen. As discussed, these regions undergo remodeling as the disease progresses. Therefore, it is crucial to understand and characterize the changes in individual tissue layers that accompany the disease phases from the acute phase to the chronic phase. However, it may be challenging to collect representative samples for specific phases because the parameters are different for each patient and it may be difficult to obtain a sufficient number of donor samples. Animal models could be of great use here and provide valuable insights into the progression of the disease. Therefore, it might be more appropriate to compare the normal aorta with the different tissue layers in the chronic phase when the aortic wall has remodeled and is mostly stable. Additionally, conducting experiments at various locations along the aorta could be valuable because the aortic wall near the aortic arch differs from more distal areas, causing dissections to often occur at this location, as discussed in Section 3.2. In the future, such information could be obtained using MRI imaging [117,118]. In fact, it would be advantageous to derive mechanical and microstructural data from experiments on diseased tissue and integrate this information into advanced material models. Moreover, novel advanced experimental methods can improve the characterization of aortic mechanical properties in health and disease [974]. Full-field optical measurements [470], including digital image correlation (DIC), digital volume correlation (DVC), and the combined use of optical coherence tomography with digital volume correlation (OCT-DVC), to visualize and quantify intramural strain distributions under similar conditions to those in aortic dissection, can significantly enhance our understanding of combined mechanics and microstructure.

In the chronic phase, significant false lumen dilatation often occurs, which increases the risk of rupture. Therefore, stent-grafts are frequently deployed in complicated patient cases. Not only do stent-grafts change the local stress distribution, stent-grafts that are oversized, undersized, or incorrectly placed can also migrate over time. Recent studies have shown that stent-grafts can even affect the microstructure of the aortic wall due to the substantial local pressures, leading to aortic wall remodeling [975,976]. Furthermore, as outlined in two subsequent studies by Suh and his colleagues [977,978], graft-to-arch angulation increased due to cardiac pulsation and respiration, likely resulting in additional local stress concentrations caused by the deployed stent-grafts can be integrated into *in silico* platforms, allowing various conditions to be tested without the need for costly animal and human trials typically required for regulatory approval. Therefore, this particular area is considered a promising application of computational biomechanics with potentially great impact on the clinical practice of patients with aortic dissection.

Finally, we would like to address the role of thrombus formation in aortic dissections, which predominantly occurs in the chronic phase. The currently available models examine thrombus formation in the false lumen exclusively from a fluid dynamics perspective – in particular, they examine the effect of thrombus on blood flow. Typically, these models are validated by individual follow-up cases. However, they only capture part of the overall picture because the important mechanical role of the thrombus in aortic dissection is largely overlooked. Consideration of thrombus formation within the FSI modeling in patient-specific cases is crucial for further validation of these models. This examination could provide valuable insights across various patient profiles, particularly with regard to the use of stent-grafts. Although deployment should theoretically enhance thrombus formation, movement of the flap could potentially impact thrombosis. As mentioned previously, the formation of a thrombus could seal the false lumen, thereby contributing to the patient's recovery. This situation often leads to wall remodeling, a process that requires further investigation due to its complexity. From a mechanical perspective, the intraluminal thrombus appears to play a protective role by limiting biomechanical wall stress due to increased wall thickness [979]. However, this also creates a highly proteolytic and oxidative environment. In the aorta, where blood pressure is high, radial convection pushes the proteolytic and oxidative components of the intraluminal thrombus into the wall, potentially contributing to progressive dilation and rupture risk. The combination of low velocities and oscillatory flow within the

aneurysm leads to flow stagnation and promotes deposition of fibrinogen and circulating cellular elements such as leukocytes, platelets, and red blood cells in the intraluminal thrombus. Enzymes and other components released by these aggregated cells are transported outward to the wall, contributing to extracellular matrix degradation [980,981]. Conversely, if diffusion from the lumen into the wall is hindered by a thrombus, important questions arise about the nutrient supply to the wall. This scenario could lead to possible hypoxia, similar to that observed in aortic aneurysms [566]. To study these effects, the application of multiscale material models and the inclusion of mechanobiology could prove useful. When considering patient-specific aspects, the exclusive use of solid mechanics or multiphase approaches can provide insightful information. For example, the bio-chemo-mechanical impact of intraluminal thrombus deposition on arterial tissue growth and remodeling was investigated using the constraint mixture theory in aortic aneurysms, as reported by Virag et al. [982]. Readers are also referred to the related studies mentioned therein. A multiphasic approach to aortic dissection based on the theory of porous media [950] considers both solid and fluid phases. This approach has already been discussed in Section 8.3. However, their work mainly focuses on the theoretical framework and demonstrates limited applicability. Because these approaches are based on CSM models, they do not explicitly consider blood flow, which significantly reduces complexity. However, for some research questions, considering blood flow can be extremely useful.

## 9.2.2. Numerical challenges and clinical applications

Computational modeling and simulation can be utilized in medicine to simulate and enhance understanding of pathological processes, assist in the diagnosis and treatment of patients, support decision-making in complex clinical cases, and evaluate the safety and efficacy of medical products, commonly known as '*in silico* medicine' [983,984]. Additionally, the term '*in silico* trial' was coined to describe the use of computer modeling and simulations to assess the safety and efficacy of various medical products, including drugs, medical devices, and diagnostic tools. Typically, these are obtained through controlled experiments carried out *in vitro, ex vivo* or *in vivo* on animals or humans, with multiple clinical trials being carried out with increasingly larger numbers of participants [985–990]. *In silico* trials aim to reduce, refine, or replace these experiments. For this purpose, digital twins are being developed [991,992] – a digital replica or virtual model of a patient that faithfully mirrors reality in real-time – and the creation of synthetic data and virtual cohorts [993–996] are of great importance, with a recent publication presenting the first synthetic aortic dissection surface meshes containing two flow channels [997]. These models provide predictive insights and also leverage machine learning techniques and reasoning to aid decision making. In recent years there has been great interest in such predictive *in silico* models, which are emerging as new methodologies for the development and regulatory evaluation of medical products, as evidenced by the numerous published studies and initiatives [984].

In general, there is broad consensus on the potential advantages of *in silico* models. In the United States, the first guidelines for conducting and reporting verification and validation [998] and for assessing the credibility of *in silico* models used in medical device submissions [999], have been published by the American Society of Mechanical Engineers (ASME) and the US Food and Drug Administration (FDA). There are also initial initiatives in Europe that advocate and promote the use of *in silico* modeling to close the gap between biomedical research, industry, and regulatory bodies. These initiatives play a crucial role in shaping policy and regulatory approaches to *in silico* methods in healthcare and highlight the value of computer modeling and simulation in reducing the time, costs, and risks associated with the development and regulatory approval of new devices. Despite consensus on the potential benefits and the existence of initial methodological frameworks, there is still a lack of general acceptance and guidance for researchers, manufacturers, or notified bodies on how to use these approaches to generate evidence. Such evidence is necessary to obtain market authorization, known as certification or qualification, from competent authorities. The first step in certifying a new methodology is to precisely define the so-called 'Context of Use' [998]. It is a clear and concise description of the specific conditions and purposes for which a particular methodology, technology or product is intended to be used. In general, each new certification must be granted for each specific Context of Use for all *in silico* trial methodologies is still lacking, representing a challenge that must be overcome [984,987].

While these problems are discussed in the literature, this review article highlighted the existing models and numerical approaches specifically applied to disease-related aspects of aortic dissection. In particular, it addressed the inherent complexity of aortic dissection and the numerous numerical challenges that must be overcome to utilize such models for patient care. Therefore, we focused on the numerical challenges associated with modeling aortic dissection, particularly when incorporating patient-specific medical data, and their potential for clinical application. Although the disease-related modeling aspects discussed previously are specific to aortic dissection, the numerical aspects examined in the following are not only applicable to modeling of aortic dissection but also relevant to related biomedical applications. Nevertheless, it is crucial to recognize the current challenges from a numerical perspective and to raise clinicians and engineers generally do not speak a common language and consistent terminology is not always established [1000]. Therefore, knowledge transfer can be challenging and lead to mutual issues going undetected.

There are numerous computational models in the literature that simulate blood flow or FSI in dissected aortas. Many of these presented models incorporated patient-specific data to reconstruct 3D geometry or obtained realistic boundary conditions, but most were still based on simplified constitutive relations such as hyperelastic solids or Newtonian fluids. There is currently no clear agreement on the importance and concrete influence of the involved (sub-)models and the simplifications made: while some studies concentrated on increasingly advanced fluid models for blood flow, others focus on boundary conditions, geometric aspects, or tissue models. Sensitivity analyses may help to identify and characterize a hierarchy of influencing factors when modeling and simulating flow phenomena in aortic dissection, also in the context of FSI and coupled tissue models. Moreover, modeling of the essential aspects of thrombus formation in aortic dissection remains rare, and detailed comparisons of available approaches are scarce in the literature due to the almost overwhelming computational demand despite advanced numerical tools. Although there were some comparisons in

terms of model assumptions, combining multiple subjects into a single computational framework, such as FSI simulations with thrombus growth, remains a challenge. It is also worth noting that the integration of initial tissue rupture, FSI, and patient-specific geometries in one framework is still beyond the current state of the art. This remains a challenging topic given the inherent numerical complexity and the difficulties in transferring traditional high-performance numerical methods to clinically relevant scenarios.

When generating spatial discretizations, i.e. computational grids for FSI or CSM simulations in patient-specific cases, vessel wall thickness is an essential component because it is directly related to the calculated wall stress. However, this aspect has received only limited attention, typically assuming constant thickness, see, e.g., [448,612,868,869,894], and many others. One of the challenges in determining wall thickness is that conventional CT and other imaging techniques available for clinical use are often unable to extract this information, despite being the primary source of medical data for computational models. Not only can the wall thickness of different regions vary, it also undergoes remodeling and may become thicker as the disease progresses, as previously discussed. Wall thickness also varies locally and the overall aortic wall thickness of the aorta decreases from the proximal to the distal part [52]. Nevertheless, the inherent complexity of creating patient-specific models with varying wall thicknesses represents a significant challenge, especially given the geometry of aortic dissection, which is characterized by multiple lumina and a dissection flap. Currently, no computational model of aortic dissection exists that does not assume a uniform distribution of wall thickness. However, recently a model has been established that has a different wall thickness for different layers: the outer wall of the true lumen, the dissection flap, and the outer wall of the false lumen [342]. This simplification leads to substantial uncertainty with *in silico* models, which must be taken into account when evaluating their reliability.

The selection of appropriate rheological or turbulence models is a topic frequently discussed in the literature when conducting fluid flow simulations. The choice depends heavily on the specific topology and existing flow conditions. A recent example demonstrates the influence of the choice of rheological model on computed WSS values in aortic hemodynamics [1001]. In particular, the uncertainty of the rheological parameters was considered and how unavoidable errors in the model parameters lead to errors in the observed quantities. The suitability and differences between laminar and turbulent flow as well as rheological modeling were discussed on a case-by-case basis, see, e.g., [759,869,936], which precluded a general statement at this point. Typically, the computationally expensive options are excluded because the computational effort in patient-specific scenarios already reaches the limits of modern computing systems.

The influence of surrounding structures is often underestimated, but can have a significant influence on the predictability of computational results. The aorta is connected to other organs and supported by tissue, as described in Section 2.1, but this is typically ignored or integrated by exterior pressure, as in [868,869], embedding elastic continuum, or condensed zero-dimensional models with uniform or spatially varying parameters, see, e.g., [448,889,892,1002], among others. From a mechanical point of view, it is then modeled as a system of springs and dashpots acting on the exterior of the vessel. Additionally, pressure in the abdominal cavity is poorly understood, let alone described as a function of the respiratory cycle, or considered to vary spatially. In fact, locally different resistances and their dependence on time could have an enormous effect on the observed stresses and the deformation state in the dissected aorta. As a result, the tissue response can change significantly, if used improperly. Therefore, one cannot simply match the measured *in vivo* displacements with tissue stiffness without taking external tissue support into account. Studies have also identified heterogeneous circumferential strains *in vivo* [1003], suggesting that certain regions of the aorta may exhibit restricted expansion, while others demonstrate greater flexibility. This demonstrates that the surrounding tissue is not homogeneous; it varies both along the length and around the circumference of the aorta. Therefore, a uniform external tissue support does not accurately represent the *in vivo* conditions.

Similar challenges exist in translating the stresses present at the time of imaging into patient-specific geometries, where one may distinguish between approaches that recover a stress-free configuration [1004,1005] or integrate pre-stress into the balance equations [890,1006,1007]. The loads occurring during the diastolic phase, commonly referred to as pre-stress, should be included in the computational model, which makes the FSI approach more complex and often requires significantly more numerical effort. Without strict consideration of pre-stress, using more advanced tissue models may not produce more realistic results compared with linear elasticity.

When creating patient-specific models of aortic dissection, spatial discretization using finite element or finite volume grids plays a critical role in the subsequent approximation of the solid or fluid phases, or both. Especially when the dissection exhibits an intricate topology, for example, when the false lumen includes not only the aorta but also several aortic branches, it becomes increasingly difficult to create a computational grid with sufficient quality and good approximation to the real geometry. Undesirably small finite elements or unfavorable aspect ratios are possible consequences if the complex geometries are even suitable for automatic mesh generation. This can affect approximation quality and simulation time in multiple ways: (i) poor finite element quality reduces the solution quality and performance of the linear solver; (ii) under normal stability conditions, small finite elements may require more time steps; and (iii) finite elements that are too small can introduce a large number of undesirable degrees of freedom. Particularly if a layer-specific model is desired, the existing thin tissue layers make mesh construction even more difficult. Semi-automatic algorithms that combine manual segmentation and unstructured meshing are usually used for mesh construction, see, e.g., [1008,1009]. Alternatively, structured meshing approaches can be utilized [1010–1012]. For example, Bošnjak et al. [1013] introduced a fully automated approach for generating structured higher-order hexahedral meshes of healthy blood vessels, which was recently adapted for application to dissected geometries [1014]. This method converts CT images into high-quality finite element grids of arbitrary order including boundary layers. The spatial resolution can be adjusted as coarsely as desired, which significantly shortens computing time. However, an extension to the solid domain is not yet available.

In the context of the finite element mesh, it is necessary to use a local coordinate system, for example, for each finite element, to take into account the anisotropic material properties of the aortic wall. This is crucial for describing the alignment of collagen fibers,

which, as previously mentioned, serve as primary load-bearing constituents in the aortic wall. Within the framework of a local coordinate system, changing fiber alignment during tissue remodeling can be characterized. However, the presence of two lumina and the dissection flap in between pose challenges that require consideration of rule-based methodologies [1015], methods based solely on structured meshing, or hybrid approaches that additionally incorporate wall normals [892]. Furthermore, when calculating a local coordinate system, due to the discontinuity of the intimal tears in the flap, there may be unphysical local fiber orientations that must be taken into account.

Another limitation of currently published studies is that they are not applicable to the general population, as most studies focus on a single patient-specific case, mainly due to the still significant numerical effort involved in patient-specific simulations. Closely related to model selection and the associated complexity is simulation processing time, which ranges from seconds when reconstructing pressures from a specific velocity field in a specific geometry, to hours or days when performing patient-specific simulations with complex constitutive models and accounting for FSI. In order to increase clinical relevance, these times must be significantly reduced. In this context, high-performance tools are the key technology to enable parametric studies on virtual cohorts, virtual surgery, digital twins, and the construction of large data sets, thus creating a solid basis for machine-learning methods. Only when a clinically relevant runtime is reached can computational models be effectively used in clinical practice, which motivates the development of tailored numerical tools. The practitioner is then no longer limited to simplified model assumptions, such as simple material models, neglect of vessel compliance, or even the consideration of 1D wireframe models, which enormously limit the predictive power of the simulation.

The majority of calculation models are often insufficiently validated or not validated at all. Nevertheless, validation of the combined modeling assumptions for highly complex coupled multiphysics or multiscale models remains a central topic. In particular, Sections 7 to 9 illustrate various *in silico* models that have undergone validation, often using 2D or 4D PC-MRI data as ground-truth for flow simulations. The reader interested in further information is referred to Wang et al. [29], who provided a concise overview of validation results by comparing computational results with *in vivo* data from 4D-flow MRI in relation to aortic dissection models. A representative validation result is presented in Fig. 74. At this point, the importance of the inflow and outflow boundary conditions must be pointed out again. In order to achieve sound model validation, high-quality *in vivo* data for inflow and outflow must be considered. While 2D and 4D PC-MRI are most commonly used to determine inlet boundary conditions at the beginning of the ascending aorta or just above the arch, they are also typically used for outlet conditions [805]. However, because PC-MRI cannot provide the traditionally required pressure data, studies often rely on data from the literature. When PC-MRI data are available at the outlet, they can provide flow rates for fine-tuning individual Windkessel parameters [772], see, e.g., [448,773,803,820]. The clinical value of these models could be substantially enhanced by placing more emphasis on validation. Unvalidated models significantly reduce the predictive value of time-consuming simulations.

Conversely, in solid-domain validation, it is often difficult to validate the wall and flap motions obtained derived from the *in vivo* displacement, which is typically obtained from cine MRI. In some cases, displacements of 8 mm were reported [448], while in other



(a) 4D-flow MRI data

(b) Aorta CFD model

(c) True lumen validation

(d) False lumen validation

**Fig. 74.** Comparison and validation of 4D-flow MRI data and CFD simulations in a 74-year-old male patient with residual type B dissection. (a) Pulsatile flow waveform across cardiac cycles acquired by 4D-flow MRI. (b) CFD model with emphasis on visualizing flow patterns through streamlines and location of entry tears. (c, d) Flow data derived from specific locations within the true and false lumina, indicated by arrows. Dotted curves represent CFD simulation results, while solid curves represent 4D-flow MRI results. Overall, the correlation between the two results is stronger for the true lumen, with the deviations being more pronounced in early diastole. The highest observed peak flow differences are approximately 12.5 and 83% for the true and false lumina, respectively. Figure is reprinted from Wang et al. [29] with permission from Elsevier.



**Fig. 75.** Possible workflow for personalized risk assessment and pre-operative planning supported by (validated) *in silico* models, based on Zhu et al. [1024] and Wang et al. [29]. The figure showing virtual stent-graft deployment and prediction for postoperative hemodymamics is adapted from Romarowski et al. [1025] with permission from Elsevier.

cases only  $\sim 1$  mm were detected [447,869,872,900]. Depending on the selection of material parameters, external tissue support, or pre-stress, some simulations reflect relatively large displacements, while others reflect smaller displacements. There is little emphasis on relating these measurements to the actual stage of tissue remodeling. For instance, in chronic aortic dissection, small displacements of the dissection flap are often measured *in vivo*, rendering a stiffer material or CFD on a fixed grid applicable. Conversely, in an acute aortic dissection, the tissue may not yet be remodeled and elastic materials therefore produce results closer to the deformations *in vivo*. Therefore, the choice of the material model that reflects the inherent microstructure in the current state can be crucial. To date, the anisotropic material behavior of the aortic wall, especially the dissected aortic wall, has usually not been taken into account in patientspecific studies.

Moreover, incorporating MRI into patient-specific CFD or FSI simulations is still an evolving field for which no standardized method is currently available to ensure a seamless transition to clinical use [29]. Notably, artificial intelligence, including deep learning techniques, holds immense potential to facilitate the acquisition and segmentation of MRI [1016] and the pre-processing of CFD, including 3D reconstruction and meshing, as well as prediction of CFD results [1017]. Custom geometric models could be automatically created from medical images [1018,1019], paving the way for automatic mesh creation and quality evaluation [1020–1022], as well as simulation setups that includ boundary condition settings and post-processing [1023].

Once these techniques are fully validated across a large patient cohort, image-based CFD simulations could assist physicians in identifying patients at high risk of late complications and enable personalized pre-operative planning that may help prevent post-operative complications such as stent-graft migration. A possible workflow, presented in Fig. 75, could provide clinicians with additional quantitative data and support them in diagnosis and risk stratification. However, since acute aortic dissection requires immediate intervention and decision-making, this work could be particularly applicable to cases of chronic aortic dissection where clinicians have more time for evaluation. In these cases, *in silico* models, whether during the risk assessment phase or in pre-operative planning, could prove invaluable for future personalized medicine. This is particularly important since patient management has so far been primarily driven by clinical factors and specific morphological parameters, as discussed throughout Section 3. While clinical guidelines undoubtedly add value at the population level, implementing patient-specific strategies could potentially lead to more optimal care.

As we integrate mathematical modeling and subsequent numerical simulations into clinical practice, we are confronted with a plethora of possible options, ranging from real-time 1D pipe networks to patient-specific FSI that require substantial computational resources. These decisions obviously affect the quality of our predictions. In traditional engineering science, the materials and related parameters may be known or even thoroughly tested. However, such comprehensive parameter studies are not available for the human body because the data are limited and expensive. Consequently, larger margins must be incorporated, further limiting the often simplistic assumptions of uniform thickness and stiffness of the tissue, uniform exterior support of the vessel, or parabolic or flat inflow profiles in combination with a population-mean volumetric flow rate. The development of complex models has to be accompanied by the capture of the variance of the model parameters involved and a non-invasive parameter estimation in the patient-specific scenario. A balance must be found between the richness of the model, the associated numerical effort and data availability. Therefore, the development and scalable implementation of efficient, optimized, and accurate reduced-order models, as well as the resulting accuracy and reliability of these models, have become a focus of modeling and simulation in recent years [1026]. Due to the approximation error, reduced-order models usually do not have the same qualitative properties and accuracy as full-order models. Therefore, finding a

compromise between model size and approximation is an essential task. Predictive power in practical scenarios can only be truly improved if the overall accuracy is sufficiently high.

# 9.3. Closure

In summary, many key questions remain unanswered and numerous obstacles persist in developing accurate, reliable, and fast multiscale, multiphysics, and data-informed *in silico* models of aortic dissection, including the complexity of the pathological microstructure and the multiconstituent nature of blood and the challenges of numerical simulation of patient-specific models with FSI or CFD tools. Of course, there is no one-size-fits-all model, and each model must be tailored to the specific research question, with appropriate assumptions, simplifications, and the necessary verification of the same. To facilitate progress and accelerate the accumulation of knowledge, it could be beneficial to group research activities in the above directions. This approach would be advantageous from both a computational engineering and a medical or clinical perspective, as it would yield new medical data that would serve as the basis for material and computational models. Crucial to the success of future developments is the collaboration between engineers and clinicians, which is currently often lacking. As a community, we bear the responsibility of promoting knowledge transfer by incorporating current advances and novel technologies from all involved fields. This approach aims to enhance predictive accuracy in clinical settings, improve and accelerate medical device design, facilitate virtual surgeries, and conduct studies on virtual cohorts in clinical practice, ultimately advancing personalized medicine and healthcare technology.

### **Declaration of Competing Interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

### Data availability

No data was used for the research described in the article.

#### Acknowledgments

The authors acknowledge the support of Graz University of Technology, Austria through the LEAD Project (2018–2024) 'Mechanics, Modeling, and Simulation of AD' to MRP; the US National Institutes of Health (U01 HL142518) to JDH; and the Leducq Foundation (ERAADICATE, D. Milewicz (PI)) to JDH.

#### References

- Isselbacher EM. The symptoms and signs of acute aortic dissection: clinical diagnosis of "the great masquerader". In: Elefteriades JA, editor. Acute aortic disease. New York, NY: Informa Healthcare USA; 2007. p. 29–38.
- [2] Ziganshin BA, Dumfarth J, Elefteriades JA. Natural history of type B aortic dissection: ten tips. Ann Cardiothorac Surg 2014;3:247–54.
- [3] Hagan PG, Nienaber CA, Isselbacher EM, Bruckman D, Karavite DJ, Russman PL, et al. The International Registry of Acute Aortic Dissection (IRAD): new insights into an old disease. JAMA 2000;283(7):897–903.
- [4] Erbel R, Alfonso F, Boileau C, Dirsch O, Eber B, Haverich A, Rakowski H, Struyven J, Radegran K, Sechtem U, Taylor J, Zollikofer C, Klein WW, Mulder B, Providencia LA. Diagnosis and management of aortic dissection. Eur Heart J 2001;22(18):1642–81.
- [5] Clouse W, Hallett J, Schaff H, Spittell P, Rowland C, Ilstrup D, Melton L. Acute aortic dissection: population-based incidence compared with degenerative aortic aneurysm rupture. Mayo Clin Proc 2004;79(2):176–80.
- [6] Hiratzka LF, Bakris GL, Beckman J, Bersin RM, Carr VF, Casey DEJ, Eagle KA, Hermann LK, Isselbacher EM, Kazerooni EA, Kouchoukos NT, Lytle BW, Milewicz DM, Reich DL, Sen S, Shinn JA, Svensson LG, Williams DM. ACCF/AHA/AATS/ACR/ASA/SCA/SCA/SIR/STS/SVM guidelines for the diagnosis and management of patients with thoracic aortic disease: a report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines, American Association for Thoracic Surgery, American College of Radiology, American Stroke Association, Society of Cardiovascular Anesthesiologists, Society for Cardiovascular Angiography and Interventions, Society of Interventional Radiology, Society of Thoracic Surgeons, and Society for Vascular Medicine. J Am Coll Cardiol 2010;55(14):e27–129.
- [7] Dake MD, Kato N, Mitchell RS, Semba CP, Razavi MK, Shimono T, Hirano T, Takeda K, Yada I, Miller DC. Endovascular stent–graft placement for the treatment of acute aortic dissection. N Engl J Med 1999;340(20):1546–52.
- [8] Sachs T, Pomposelli F, Hagberg R, Hamdan A, Wyers M, Giles K, Schermerhorn M. Open and endovascular repair of type B aortic dissection in the nationwide inpatient sample. J Vasc Surg 2010;52(4):860–6,discussion 866. https://doi.org/10.1016/j.jvs.2010.05.008.
- Khayat M, Cooper KJ, Khaja MS, Gandhi R, Bryce YC, Williams DM. Endovascular management of acute aortic dissection, Cardiovasc Diagn Ther 2018;8(1): S97–S107. https://doi.org/10.21037/cdt.2017.10.07.
- [10] Humphrey JD. Mechanics of the arterial wall: review and directions. Crit Rev Biomed Eng 1995;23:1–162.
- [11] Rajagopal K, Bridges C, Rajagopal KR. Towards an understanding of the mechanics underlying aortic dissection. Biomech Model Mechanobiol 2007;6(5): 345–59.
- [12] Tsamis A, Krawiec JT, Vorp DA. Elastin and collagen fibre microstructure of the human aorta in ageing and disease: a review. J R Soc Interface 2013;10(83): 20121004. https://doi.org/10.1098/rsif.2012.1004.
- [13] Sherifova S, Holzapfel GA. Biochemomechanics of the thoracic aorta in health and disease. Prog Biomed Eng 2020;2:032002.
- [14] Peterss S, Mansour AM, Ross JA, Vaitkeviciute I, Charilaou P, Dumfarth J, Fang H, Ziganshin BA, Rizzo JA, Adeniran AJ, Elefteriades JA. Changing pathology of the thoracic aorta from acute to chronic dissection: literature review and insights. J Am Coll Cardiol 2016;68(10):1054–65.
- [15] Spinelli D, Benedetto F, Donato R, Piffaretti G, Massimiliano MM-T, Himanshu JP, Eagle KA, Trimarchi S. Current evidence in predictors of aortic growth and events in acute type B aortic dissection. J Vasc Surg 2018;68(6):1925–1938.e8.
- [16] Zilber ZA, Boddu A, Malaisrie SC, Hoel AW, Mehta CK, Vassallo P, Burris NS, Roldán-Alzate A, Collins JD, François CJ, Allen BD. Noninvasive morphologic and hemodynamic evaluation of type B aortic dissection: state of the art and future perspectives. Radiol Cardiothorac Imaging 2021;3(3):e200456.

- [17] Tong J, Cheng Y, Holzapfel GA. Mechanical assessment of arterial dissection in health and disease: advancements and challenges. J Biomech 2016;49(12): 2366–73.
- [18] Sherifova S, Holzapfel GA. Biomechanics of aortic wall failure with a focus on dissection and aneurysm: a review. Acta Biomater 2019;99:1–17.
- [19] Brunet J, Pierrat B, Badel P. Review of current advances in the mechanical description and quantification of aortic dissection mechanisms. IEEE Rev Biomed Eng 2021;14:240–55.
- [20] Jiang D-S, Yi X, Zhu XH, Wei X. Experimental in vivo and ex vivo models for the study of human aortic dissection: promises and challenges. Am J Transl Res 2016;8(12):5125–40.
- [21] Wu M, Liu L, Xiong J. In vitro studies on hemodynamics of type B aortic dissection: accuracy and reliability. J Endovasc Ther 2023. 15266028231182229.
- [22] Wan Ab Naim WN, Ganesan PB, Sun Z, Chee KH, Hashim SA, Einly L. A perspective review on numerical simulations of hemodynamics in aortic dissection, Sci World J 2014;2014(1):652520. https://doi.org/10.1155/2014/652520.
- [23] Doyle BJ, Norman PE. Computational biomechanics in thoracic aortic dissection: today's approaches and tomorrow's opportunities. Ann Biomed Eng 2016;44 (1):71–83.
- [24] Sun Z, Chaichana T. A systematic review of computational fluid dynamics in type B aortic dissection. Int J Cardiol 2016;210:26–31.
- [25] Gültekin O, Holzapfel GA. A brief review on computational modeling of rupture in soft biological tissues. In: O. Oñate, D. Peric, E. de Souza Neto, E. Chiumenti (Eds.), Advances in Computational Plasticity. A Book in Honour of D. Roger J. Owen, Vol. 46, Springer Nature; 2018, p. 113–44. https://doi.org/10.1007/978-3-319-60885-3\_6.
- [26] Karmonik C, Partovi S, Davies MG, Bismuth J, Shah DJ, Bilecen D, Staub D, Noon GP, Loebe M, Bongratz G, Lumsden AB. Integration of the computational fluid dynamics technique with MRI in aortic dissections. Magn Reson Med 2013;69:1438–42.
- [27] Munshi B, Parker LP, Norman PE, Doyle BJ. The application of computational modeling for risk prediction in type B aortic dissection. J Vasc Surg 2020;71(5): 1789–801.
- [28] Mourato A, Valente R, Xavier J, Brito M, Avril S, de Sá JC, Tomás A, Fragata J. Computational modelling and simulation of fluid structure interaction in aortic aneurysms: a systematic review and discussion of the clinical potential. Appl Sci 2022;12(16):8049.
- [29] Wang Q, Guo X, Brooks M, Chuen J, Poon EKW, Ooi A, Lim RP. MRI in CFD for chronic type B aortic dissection: ready for prime time? Comput Biol Med 2022; 150:106138.
- [30] Kamada H, Nakamura M, Ota H, Higuchi S, Takase K. Blood flow analysis with computational fluid dynamics and 4D-flow MRI for vascular diseases. J Cardiol 2022;80(5):386–96.
- [31] Hohri Y, Chung MM, Kandula V, Kim I, Leb J, Hayashi H, Elmously A, O'Donnell TFX, Patel V, Vedula V, Takayama H. Blood flow assessment technology in aortic surgery: a narrative review. J Thorac Dis 2024;16(4). 2623–2336.
- [32] Sherrah AG, Grieve SM, Jeremy RW, Bannon PG, Vallely MP, Puranik R. MRI in chronic aortic dissection: a systematic review and future directions. Front Cardiovasc Med 2015;2:5.
- [33] Pepe A, Li J, Rolf-Pissarczyk M, Gsaxner C, Chen X, Holzapfel GA, Egger J. Detection, segmentation, simulation and visualization of aortic dissections: a review. Med Image Anal 2020;65:101773.
- [34] Fleischmann D, Afifi RO, Casanegra AI, Elefteriades JA, Gleason TG, Hanneman K, Roselli EE, Willemink MJ, Fischbein MP. Imaging and surveillance of chronic aortic dissection: a scientific statement from the American Heart Association. Circ Cardiovasc Imaging 2022;15(3):e000075.
- [35] Shen J, Mastrodocasa D, Tse JR, Fleischmann D. Imaging challenges in chronic dissection. Semin Roentgenol 2022;57(4):345-56.
- [36] Mussa FF, Horton JD, Moridzadeh R, Nicholson J, Trimarchi S, Eagle KA. Acute aortic dissection and intramural hematoma: a systematic review. JAMA 2016; 316(7):754–63.
- [37] Nicholls F. Observations concerning the body of his late Majesty. Philos Trans London 1761;52:265–75.
- [38] Leonard JC. Thomas Bevill Peacock and the early history of dissecting aneurysm. Br Med J 1979;2(6184):260-2.
- [39] De Bakey ME, Cooley D, Creech O. Surgical considerations of dissecting aneurysm of the aorta. Ann Surg 1955;142(4):586–610;discussion,611–2. https://doi. org/10.1097/00000658-195510000-00005.
- [40] Spencer F, Blake H. A report of the successful surgical treatment of aortic regurgitation from a dissecting aortic aneurysm in a patient with the Marfan syndrome. J Thorac Cardiovasc Surg 1962;44:238–45.
- [41] DeBakey ME, Henly WS, Cooley DA, Morris Jr GC, Crawford ES, Beall AC. Surgical management of dissecting aneurysms of the aorta. Cardiovasc Surg (Torino) 1964;58:200–11.
- [42] Morris Jr GC, Cooley DA, DeBakey ME. Correction of dissecting aneurysm of ascending thoracic aorta. Bull Soc Int Chir 1964;23:308–14.
- [43] Wheat Jr M, Palmer R, Bartley T, Seelman R. Treatment of dissecting aneurysms of the aorta without surgery. J Thorac Cardiovasc Surg 1965;50(3):364–73.
  [44] DeBakey ME, McCollum CH, Crawford ES, Morris Jr GC, Howell J, Noon GP, Lawrie G. Dissection and dissecting aneurysms of the aorta: twenty-year follow-up of five hundred twenty-seven patients treated surgically. Surgery 1982;92(6):1118–34.
- [45] Altman LK. The man on the table devised the surgery. The New York Times 07–24-2023 (2006). http://www.nytimes.com/2006/12/25/health/25surgeon. html?pagewanted=all.
- [46] Nienaber CA, Fattori R, Lund G, Dieckmann C, Wolf W, von Kodolitsch Y, Nicolas V, Pierangeli A. Nonsurgical reconstruction of thoracic aortic dissection by stent-graft placement. N Engl J Med 1999;340(20):1539–45.
- [47] Criado FJ. Aortic dissection: a 250-year perspective. Tex Heart Inst J 2011;38(6):694-700.
- [48] Tortora GJ, Nielsen MT. Principles of Human Anatomy. 12th Edition. Wiley; 2010.
- [49] Shen Y, LeMaire S. Molecular pathogenesis of genetic and sporadic aortic aneurysms and dissections. Curr Probl Surg 2017;54(3):95–155.
- [50] Patel KP, Vandermolen S, Herrey AS, Cheasty E, Menezes L, Moon JC, Pugliese F, Treibel TA. Cardiac computed tomography: application in valvular heart disease. Front Cardiovasc Med 2022;9:849540.
- [51] Kaufmann JA, Lee MJ. Thoracic aorta. In: Kaufmann JA, Lee MJ, editors. Vascular and interventional radiology. Philadelphia: W.B. Saunders; 2004. p. 219–45.
   [52] Humphrey JD. Cardiovascular Solid Mechanics: Cells, Tissues, and Organs. Springer Verlag; 2002.
- [53] Erbel R, Eggebrecht H. Aortic dimensions and the risk of dissection. Heart 2006;92(1):137–42.
- [54] Belz GG. Elastic properties and windkessel function of the human aorta. Cardiovasc Drugs Ther 1995;9(1):73-83.
- [55] Fung YC. Biomechanics: Circulation. 2nd Edition. Springer; 1997.
- [56] Westerhof N, Lankhaar JW, Westerhof BE. The arterial Windkessel. Med Biol Eng Comput 2009;47(2):131-41.
- [57] Pagoulatou S, Adamopoulos D, Rovas G, Bikia V, Müller H, Giannakopoulos G, Mauler-Wittwer S, Licker MJ, Stergiopulos N, Noble S. Arterial wave reflection and aortic valve stenosis: diagnostic challenges and prognostic significance. Front Cardiovasc Med 2022;9:863968.
- [58] Mynard JP, Kondiboyina A, Kowalski R, Cheung MMH, Smolich JJ. Measurement, analysis and interpretation of pressure/flow waves in blood vessels. Front Physiol 2020;11:1085.
- [59] Sutton-Tyrrell K, Najjar SS, Boudreau RM, Venkitachalam L, Kupelian V, Simonsick EM, Havlik R, Lakatta EG, Spurgeon H, Kritchevsky S, Pahor M, Bauer D, Newman A. Health ABC Study, Elevated aortic pulse wave velocity, a marker of arterial stiffness, predicts cardiovascular events in well-functioning older adults. Circulation 2005;111(25):3384–90.
- [60] Reference Values for Arterial Stiffness' Collaboration Determinants of pulse wave velocity in healthy people and in the presence of cardiovascular risk factors: 'establishing normal and reference values. Eur Heart 2010;31(19):2338–50.
- [61] Cavalcante JL, Lima JA, Redheuil A, Al-Mallah MH. Aortic stiffness: current understanding and future directions. J Am Coll Cardiol 2011;57(14):1511–22.
- [62] Krüger T, Grigoraviciute A, Veseli K, Schibilsky D, Wendel HP, Schneider W, Schlensak C. Elastic properties of the young aorta: ex vivo perfusion experiments in a porcine model. Eur J Cardiothorac Surg 2015;48(2):221–7.
- [63] Vander Linden K, Vanderveken E, Van Hoof L, Maes L, Fehervary H, Dreesen S, et al. Stiffness matters: improved failure risk assessment of ascending thoracic aortic aneurysms. JTCVS Open 2023;16:66–83.

- [64] Papaioannou TG, Protogerou AD, Stamatelopoulos KS, Vavuranakis M, Stefanadis C. Non-invasive methods and techniques for central blood pressure estimation: procedures, validation, reproducibility and limitations. Curr Pharm Des 2009;15(3):245–53.
- [65] Holzapfel GA, Gasser TC, Ogden RW. A new constitutive framework for arterial wall mechanics and a comparative study of material models. J Elast Phys Sci Solids 2000;61(1):1–48.
- [66] Halushka MK, Angelini A, Bartoloni G, Basso C, Batoroeva L, Bruneval P, et al. Consensus statement on surgical pathology of the aorta from the Society for Cardiovascular Pathology and the Association For European Cardiovascular Pathology: II. Noninflammatory degenerative diseases - nomenclature and diagnostic criteria. Cardiovasc Pathol 2016;25(3):247–57.
- [67] Galley HF, Webster NR. Physiology of the endothelium. Br J Anaesth 2004;93(1):105–13.
- [68] Sandoo A, van Zanten JJ, Metsios GS, Carroll D, Kitas GD. The endothelium and its role in regulating vascular tone. Open Cardiovasc Med J 2010;4:302–12.[69] Canham PB, Finlay HM, Dixon JG, Boughner DR, Chen A. Measurements from light and polarised light microscopy of human coronary arteries fixed at
- distending pressure. Cardiovasc Res 1989;23(11):973–82.
- [70] Clark JM, Glagov S. Transmural organization of the arterial media. The lamellar unit revisited, Arteriosclerosis 1985;5(1):19–34.
  [71] Wolinsky H, Glagov S. A lamellar unit of aortic medial structure and function in mammals. Circ Res 1967;20(1):99–111.
- (72) Wolinsky H, Glagov S. Comparison of abdominal and thoracic aortic medial structure in mammals. Deviation of man from the usual pattern. Circ Res 1969;25: 677–86.
- [73] O'Connell MK, Murthy S, Phan S, Xu C, Buchanan J, Spilker R, Dalman RL, Zarins CK, Denk W, Taylor CA. The three-dimensional micro- and nanostructure of the aortic medial lamellar unit measured using 3D confocal & electron microscopy imaging. Matrix Biol 2008;27(3):171–81.
- [74] Rodriguez-Feo JA, Sluijter JP, Kleijn DP, Pasterkamp G. Modulation of collagen turnover in cardiovascular disease. Curr Pharm Des 2005;11(19):2501–14.
   [75] Arribas SM, Hinek A, González MC. Elastic fibres and vascular structure in hypertension. Pharmacol Ther 2006;111(3):771–91.
- [76] Ushiki T. Collagen fibers, reticular fibers and elastic fibers. A comprehensive understanding from a morphological viewpoint. Arch Histol Cytol 2002;65(2): 109–26.
- [77] Berry CL, Looker T, Germain J. The growth and development of the rat aorta. J Anat 1972;113(1):1-16.
- [78] Humphrey JD, Schwartz MA, Tellides G, Milewicz DM. Role of mechanotransduction in vascular biology. Circ Res 2015;116(8):1448-61.
- [79] Roccabianca S, Bellini C, Humphrey JD. Computational modelling suggests good, bad and ugly roles of glycosaminoglycans in arterial wall mechanics and mechanobiology. J R Soc Interface 2014;11(97):20140397.
- [80] Katsumi A, Orr AW, Tzima E, Schwartz MA. Integrins in mechanotransduction. J Biol Chem 2004;279(13):12001-4.
- [81] Walker-Caprioglio HM, Trotter JA, Mercure J, Little S, McGuffee L. Organization of rat mesenteric artery after removal of cells of extracellular matrix components. Cell Tissue Res 1991;264:63–77.
- [82] Wagenseil JE, Mecham RP. Vascular extracellular matrix and arterial mechanics. Physiol Rev 2009;89(3):957-89.
- [83] Holzapfel GA, Ogden RW, Sherifova S. On fibre dispersion modelling of soft biological tissues: a review. Proc Math Phys Eng Sci 2019;475(2224):20180736.
   [84] Heegaard A-M, Corsi A, Danielsen CC, Nielsen KL, Jorgensen HL, Riminucci M, Young MF, Bianco P. Biglycan deficiency causes spontaneous aortic dissection and rupture in mice. Circulation 2007;115(21):2731–8.
- [85] Pukaluk A, Wolinski H, Viertler C, Regitnig P, Holzapfel GA, Sommer G. Changes in the microstructure of the human aortic adventitia under biaxial loading investigated by multi-photon microscopy. Acta Biomater 2023;161:154–69.
- [86] Ghadie NM, St-Pierre J-P, Labrosse MR. The contribution of glycosaminoglycans/proteoglycans to aortic mechanics in health and disease: a critical review. IEEE Trans Biomed Eng 2021;68(12):3491–500.
- [87] Balagurunathan K, Nakato H, Desai UR. Glycosaminoglycans: Chemistry and Biology, Humana New York, NY. 2015.
- [88] Iozzo RV, Schaefer L. Proteoglycan form and function: a comprehensive nomenclature of proteoglycans. Matrix Biol 2015;42:11-55.
- [89] Bode-Lesniewska B, Dours-Zimmermann MT, Odermatt BF, Briner J, Heitz PU, Zimmermann DR. Distribution of the large aggregating proteoglycan versican in adult human tissues. J Histochem Cytochem 1996;44(4):303–12.
- [90] Salisbury BG, Wagner WD. Isolation and preliminary characterization of proteoglycans dissociatively extracted from human aorta. J Biol Chem 1981;256(15): 8050–7.
- [91] Lark MW, Yeo TK, Mar H, Lara S, Hellström I, Hellström KE, Wight TN. Arterial chondroitin sulfate proteoglycan: localization with a monoclonal antibody. J Histochem Cytochem 1988;36(10):1211–21.
- [92] Azeloglu EU, Albro MB, Thimmappa VA, Ateshian GA, Costa KD. Heterogeneous transmural proteoglycan distribution provides a mechanism for regulating residual stresses in the aorta. Am J Physiol Heart Circ Physiol 2008;294(3):H1197–205.
- [93] Cikach FS, Koch CD, Mead TJ, Galatioto J, Willard BB, Emerton KB, Eagleton MJ, Blackstone EH, Ramirez F, Roselli EE, Apte SS. Massive aggrecan and versican accumulation in thoracic aortic aneurysm and dissection. JCI Insight 2018;3(5):e97167. https://doi.org/10.1172/jci.insight.97167.
- [94] Theocharis AD, Tsolakis I, Hjerpe A, Karamanos NK. Human abdominal aortic aneurysm is characterized by decreased versican concentration and specific downregulation of versican isoform V(o). Atherosclerosis 2001;154(2):367–76.
- [95] Didangelos A, Yin X, Mandal K, Saje A, Smith A, Xu Q, et al. Extracellular matrix composition and remodeling in human abdominal aortic aneurysms: a proteomics approach. Mol Cell Proteomics 2011;10(8):M111.008128.
- [96] Theocharis AD, Karamanos NK. Decreased biglycan expression and differential decorin localization in human abdominal aortic aneurysms. Atherosclerosis 2002;165(2):221–30.
- [97] Halper J. Proteoglycans and diseases of soft tissues. Adv Exp Med Biol 2014;802:49-58.
- [98] Wight TN. Cell biology of arterial proteoglycans. Arteriosclerosis 1989;9(1):1-20.
- [99] Sorrentino TA, Fourman L, Ferruzzi J, Miller KS, Humphrey JD, Roccabianca S. Local versus global mechanical effects of intramural swelling in carotid arteries. J Biomech Eng 2015;137(4):041008.
- [100] Lai WM, Hou JS, Mow VC. A triphasic theory for the swelling and deformation behaviors of articular cartilage. J Biomech Eng 1991;113(3):245-58.
- [101] Mattson YM, Wang Y, Zhang Y. Contributions of glycosaminoglycans to collagen fiber recruitment in constitutive modeling of arterial mechanics. J Biomech 2019;82:211–9.
- [102] Beenakker J-WM, Ashcroft BA, Lindeman JHN, Oosterkamp TH. Mechanical properties of the extracellular matrix of the aorta studied by enzymatic treatments. Biophys J 2012;102(8):1731–7.
- [103] Humphrey JD. Possible mechanical roles of glycosaminoglycans in thoracic aortic dissection and associations with dysregulated transforming growth factor-β. J Vasc Res 2013;50(3):198–9.
- [104] Cardamone L, Valentín A, Eberth JF, Humphrey JD. Origin of axial prestretch and residual stress in arteries. Biomech Model Mechanobiol 2009;8(6):431-46.
- [105] Bellini C, Ferruzzi J, Roccabianca S, Di Martino ES, Humphrey JD. A microstructurally motivated model of arterial wall mechanics with mechanobiological implications. Ann Biomed Eng 2014;42:488–502.
- [106] Pierce DM, Fastl TE, Rodriguez-Vila B, Verbrugghe P, Fourneau I, Maleux G, Herijgers P, Gomez EJ, Holzapfel GA. A method for incorporating threedimensional residual stretches/stresses into patient-specific finite element simulations of arteries. J Mech Behav Biomed Mater 2015;47:147–64.
- [107] Rajagopal KR, Rajagopal K. Modeling of the aorta: complexities and inadequacies. Aorta (Stamford) 2020;8(4):91-7.
- [108] Humphrey JD. Review paper: continuum biomechanics of soft biological tissue. Proc R Soc A: Math Phys Eng Sci 2003;459(2029):3-46.
- [109] Kalita P, Schaefer R. Mechanical models of artery walls. Arch Comput Methods Eng 2008;15:1–36.
- [110] Payan Y, Ohayon J. Biomechanics of Living Organs: Hyperelastic Constitutive Laws for Finite Element Modeling. Elsevier Science; 2017.
- [111] Amabili M, Balasubramanian P, Breslavsky I. Anisotropic fractional viscoelastic constitutive models for human descending thoracic aortas. J Mech Behav Biomed Mater 2019;99:186–97.
- [112] Franchini G, Breslavsky ID, Holzapfel GA, Amabili M. Viscoelastic characterization of human descending thoracic aortas under cyclic load. Acta Biomater 2021;130:291–307.

- [113] Franchini G, Giovanniello F, Amabili M. Viscoelasticity of human descending thoracic aorta in a mock circulatory loop. J Mech Behav Biomed Mater 2022;130: 105205.
- [114] Myneni M, Rajagopal KR. Constitutive modeling of the mechanical response of arterial tissues. Appl Eng Sci 2022;11:100111.
- [115] Pukaluk A, Sommer G, Holzapfel GA. Multimodal experimental studies of the passive mechanical behavior of human aortas: current approaches and future directions. Acta Biomater 2024;178:1–12.
- [116] Leyssens L, Balcaen T, Pétré M, Béjar Ayllón N, El Aazmani W, de Pierpont A, Pyka G, Lacroix V, Kerckhofs G. Non-destructive 3D characterization of the blood vessel wall microstructure in different species and blood vessel types using contrast-enhanced microCT and comparison with synthetic vascular grafts. Acta Biomater 2023;164:303–16.
- [117] Nielles-Vallespin S, Khalique Z, Ferreira PF, de Silva R, Scott AD, Kilner P, McGill LA, Giannakidis A, Gatehouse PD, Ennis D, Aliotta E, Al-Khalil M, Kellman P, Mazilu D, Balaban RS, Firmin DN, Arai AE, Pennell DJ. Assessment of myocardial microstructural dynamics by in vivo diffusion tensor cardiac magnetic resonance. J Am Coll Cardiol 2017;69(6):661–76.
- [118] Linka K, Thüring J, Rieppo L, Aydin RC, Cyron CJ, Kuhl C, Merhof D, Truhn D, Nebelung S. Machine learning-augmented and microspectroscopy-informed multiparametric MRI for the non-invasive prediction of articular cartilage composition. Osteoarthritis Cartilage 2021;29(4):592–602.
- [119] Liu SQ, Fung YC. Zero-stress states of arteries. J Biomech Eng 1988;110(1):82-4.
- [120] Fung YC. What are the residual stresses doing in our blood vessels? Ann Biomed Eng 1991;19(3):237–49.
- [121] Charm S, Kurland G. Viscometry of human blood for shear rates of 0-100,000 sec<sup>-1</sup>. Nature 1965;206:617–8.
- [122] Chien S, Usami S, Taylor HM, Lundberg JL, Gregersen MI. Effects of hematocrit and plasma proteins on human blood rheology at low shear rates. J Appl Physiol 1966;21(1):81–7.
- [123] Thurston GB. Viscoelasticity of human blood. Biophys J 1972;12(9):1205-17.
- [124] Rajagopal K. Commentary: simple enough, but not oversimplified. J Thorac Cardiovasc Surg 2022;S0022–5223(22). 01238–7.
- [125] Nienaber CA, Ince H. Classification of aortic dissection. In: Elefteriades JA, editor. Acute Aortic Disease. New York, NY: Informa Healthcare USA; 2007. p. 1–28.
- [126] Isselbacher EM, Preventza O, Hamilton Black III J, Augoustides JG, Beck AW, Bolen MA, Braverman AC, Bray BE, Brown-Zimmerman MM, Chen EP, Collins TJ, DeAnda A, Fanola CL, Girardi LN, Hicks CW, Hui DS, Schuyler Jones W, Kalahasti V, Kim KM, Milewicz DM, Oderich GS, Ogbechie L, Promes SB, Gyang Ross E, Schermerhorn ML, Singleton Times S, Tseng EE, Wang GJ, Woo YJ. 2022 ACC/AHA Guideline for the diagnosis and management of aortic disease: a report of the American Heart Association/American College of Cardiology joint committee on clinical practice guidelines. Circulation 2022;146(24):e334–e482. https://doi.org/10.1161/CIR.000000000001106.
- [127] Sundt TM. Intramural hematoma and penetrating atherosclerotic ulcer of the aorta. Ann Thorac Surg 2007;83(2):S835–41;discussion S846–50. https://doi. org/10.1016/j.athoracsur.2006.11.019.
- [128] Vilacosta I, San Román JA, di Bartolomeo R, Eagle K, Estrera AL, Ferrera C, Kaji S, Nienaber CA, Riambau V, Schäfers H-J, Serrano FJ, Song J-K, Maroto L. Acute aortic syndrome revisited: JACC state-of-the-art review. J Am Coll Cardiol 2021;78(21):2106–25.
- [129] Carrel T, Sundt 3rd TM, von Kodolitsch Y, Czerny M. Acute aortic dissection. Lancet 2023;401(10378):773-88.
- [130] Sueyoshi E, Nagayama H, Hayashida T, Sakamoto I, Uetani M. Comparison of outcome in aortic dissection with single false lumen versus multiple false lumens: CT assessment. Radiology 2013;267(2):368–75.
- [131] Lie JT. Triple-barrel aorta: dissection of a healed aortic dissection. Hum Pathol 1982;13(764-6):764-6.
- [132] Ohte N, Takase R, Hashimoto T, Suzuki K, Narita H, Fujinami T. Four-barrel aortic dissection involving the ascending aorta: a case report. Angiology 1995;46 (11):1059–62.
- [133] DeBakey ME, Henly WS, Cooley DA, Morris Jr GC, Crawford ES, Beall AC. Surgical management of dissecting aneurysms of the aorta. J Thorac Cardiovasc Surg 1965;49:130–49.
- [134] Lombardi JV, Hughes GC, Appoo JJ, Bavaria JE, Beck AW, Cambria RP, Charlton-Ouw K, Eslami MH, Kim KM, Leshnower BG, Maldonado T, Reece TB, Wang GJ. Society for Vascular Surgery (SVS) and Society of Thoracic Surgeons (STS) reporting standards for type B aortic dissections. J Vasc Surg 2020;71(3): 723–47.
- [135] Tadros RO, Tang GHL, Barnes HJ, Mousavi I, Kovacic JC, Faries P, Olin JW, Marin ML, Adams DH. Optimal treatment of uncomplicated type B aortic dissection: JACC review topic of the week. J Am Coll Cardiol 2019;74(11):1494–504.
- [136] Vacirca A, Dias Neto M, Baghbani-Oskouei A, Huang Y, Tenorio ER, Estrera A, Oderich GS. Timing of intervention for aortic intramural hematoma. Ann Vasc Surg 2023;94:14–21.
- [137] Daily PO, Trueblood HW, Stinson EB, Wuerflein RD, Shumway NE. Management of acute aortic dissections. Ann Thorac Surg 1970;10(3):237-47.
- [138] Czerny M, Schmidli J, Adler S, van den Berg JC, Bertoglio L, Carrel T, Chiesa R, Clough RE, Eberle B, Etz C, Grabenwöger M, Haulon S, Jakob H, Kari FA, Mestres CA, Pacini D, Resch T, Rylski B, Schoenhoff F, Shrestha M, von Tengg-Kobligk H, Tsagakis K, Wyss TR. EACTS/ESVS scientific document group, Current options and recommendations for the treatment of thoracic aortic pathologies involving the aortic arch: an expert consensus document of the European Association for Cardio-Thoracic Surgery (EACTS) and the European Society for Vascular Surgery (ESVS). Eur J Cardiothorac Surg 2019;55(1):133–62.
- [139] Carino D, Singh M, Molardi A, Agostinelli A, Goldoni M, Pacini D, Nicolini F. Non-A non-B aortic dissection: a systematic review and meta-analysis. Eur J Cardiothorac Surg 2019;55(4):653–9.
- [140] Svensson LG, Labib SB, Eisenhauer AC, Butterly JR. Intimal tear without hematoma: an important variant of aortic dissection that can elude current imaging techniques. Circulation 1999;99(10):1331–6.
- [141] Hirst Jr A, Johns Jr V, Kime Jr S. Dissecting aneurysm of the aorta: a review of 505 cases. Medicine (Baltimore) 1958;37(3):217-79.
- [142] Booher AM, Isselbacher EM, Nienaber CA, Trimarchi S, Evangelista A, Montgomery DG, Froehlich JB, Ehrlich MP, Oh JK, Januzzi JL, O'Gara P, Sundt TM, Harris KM, Bossone E, Pyeritz RE, Eagle KA, Investigators IRAD. The IRAD classification system for characterizing survival after aortic dissection. Am J Med 2013;126(8):730.e19–24.
- [143] Clough RE, Nienaber CA. Management of acute aortic syndrome. Nat Rev Cardiol 2015;12(2):103-14.
- [144] Dake MD, Thompson M, van Sambeek M, Vermassen F, Morales JP. DEFINE Investigators, DISSECT: a new mnemonic-based approach to the categorization of aortic dissection. Eur J Vasc Endovasc Surg 2013;46(2):175–90.
- [145] Moulakakis KG, Mylonas SN, Dalainas I, Kakisis J, Kotsis T, Liapis CD. Management of complicated and uncomplicated acute type B dissection. A systematic review and meta-analysis. Ann Cardiothorac Surg 2014;3(3):234–46.
- [146] Blount KJ, Hagspiel KD. Aortic diameter, true lumen, and false lumen growth rates in chronic type B aortic dissection. AJR Am J Roentgenol 2009;192(5): W222-9.
- [147] Patterson BO, Vidal-Diez A, Karthikesalingam A, Holt PJ, Loftus IM, Thompson MM. Comparison of aortic diameter and area after endovascular treatment of aortic dissection. Ann Thorac Surg 2015;99(1):95–102.
- [148] Kicska GA, Hurwitz Koweek LM, Ghoshhajra BB, Beache GM, Brown RKJ, Davis AM, Hsu JY, Khosa F, Kligerman SJ, Litmanovich D, Lo BM, Maroules CD, Meyersohn NM, Rajpal S, Villines TC, Wann S, Abbara S. ACR appropriateness criteria® suspected acute aortic syndrome. J Am Coll Radiol 2021;99(1): S474–S481. https://doi.org/10.1016/j.jacr.2021.09.004.
- [149] Roos JE, Willmann JK, Weishaupt D, Lachat M, Marincek B, Hilfiker PR. Thoracic aorta: motion artifact reduction with retrospective and prospective electrocardiography-assisted multi-detector row CT. Radiology 2002;222(1):271–7.
- [150] Zhang J, Fletcher JG, Vrtiska TJ, Manduca A, Thompson JL, Raghavan ML, Wentz RJ, McCollough CH. Large-vessel distensibility measurement with electrocardiographically gated multidetector CT: phantom study and initial experience. Radiology 2007;245(1):258–66.
- [151] Ganten MK, Krautter U, von Tengg-Kobligk H, Böckler D, Schumacher H, Stiller W, Delorme S, Kauczor HU, Kauffmann GW, Bock M. Quantification of aortic distensibility in abdominal aortic aneurysm using ECG-gated multi-detector computed tomography. Eur Radiol 2008;18(5):966–73.
- [152] Weber TF, Ganten M, Böckler D, Geisbüsch P, Kopp-Schneider A, Kauczor HU, von Tengg-Kobligk H. Assessment of thoracic aortic conformational changes by four-dimensional computed tomography angiography in patients with chronic aortic dissection type B. Eur Radiol 2009;19(1):245–53.

- [153] Laurent S, Cockcroft J, Van Bortel L, Boutouyrie P, Giannattasio C, Hayoz D, Pannier B, Vlachopoulos C, Wilkinson I, Struijker-Boudier H. European Network for Non-invasive Investigation of Large Arteries. Expert consensus document on arterial stiffness: methodological issues and clinical applications. Eur Heart J 2006;27(21):2588–605.
- [154] Hameeteman K, Rozie S, Metz CT, Manniesing R, van Walsum T, van der Lugt A, Niessen WJ, Klein S. Automatic carotid artery distensibility measurements from CTA using nonrigid registration. Med Image Anal 2013;17(5):515–24.
- [155] Meloni A, Frijia F, Panetta D, Degiorgi G, De Gori C, Maffei E, Clemente A, Positano V, Cademartiri F. Photon-counting computed tomography (PCCT): technical background and cardio-vascular applications. Diagnostics (Basel) 2023;13(4):645.
- [156] van der Bie J, van Straten M, Booij R, Bos D, Dijkshoorn ML, Hirsch A, Sharma SP, Oei EHG, Budde RPJ. Photon-counting CT: review of initial clinical results. Eur J Radiol 2023;163:110829.
- [157] Sherrah AG, Vallely MP, Grieve SM, Jeremy RW, Hendel PN, Puranik R. Clinical utility of magnetic resonance imaging in the follow-up of chronic aortic type B dissection. Heart Lung Circ 2014;23(7):e157–9.
- [158] Clough RE, Waltham M, Giese D, Taylor PR, Schaeffter T. A new imaging method for assessment of aortic dissection using four-dimensional phase contrast magnetic resonance imaging. J Vasc Surg 2012;55(4):914–23.
- [159] François CJ, Markl M, Schiebler ML, Niespodzany E, Landgraf BR, Schlensak C, Frydrychowicz A. Four-dimensional, flow-sensitive magnetic resonance imaging of blood flow patterns in thoracic aortic dissections. J Thorac Cardiovasc Surg 2013;145(5):1359–66.
- [160] de Beaufort HW, Shah DJ, Patel AP, Jackson MS, Spinelli D, Yang EY, Ghosn MG, Autry K, Igo SR, Lumsden AB, Little SH, Trimarchi S, Bismuth J. Fourdimensional flow cardiovascular magnetic resonance in aortic dissection: assessment in an ex vivo model and preliminary clinical experience. J Thorac Cardiovasc Surg 2018;157(2):467–476.e1.
- [161] Burris NS, Nordsletten DA, Sotelo JA, Grogan-Kaylor R, Houben IB, Figueroa CA, Uribe S, Patel HJ. False lumen ejection fraction predicts growth in type B aortic dissection: preliminary results. Eur J Cardiothorac Surg 2020;57(5):898–903.
- [162] Jarvis K, Pruijssen JT, Son AY, Allen BD, Soulat G, Vali A, Barker AJ, Hoel AW, Eskandari MK, Malaisrie SC, Carr JC, Collins JD, Markl M. Parametric hemodynamic 4D flow MRI maps for the characterization of chronic thoracic descending aortic dissection. J Magn Reson Imaging 2020;51(5):1357–68.
- [163] Boussel L, Rayz VL, Martin A, Acevedo-Bolton G, Lawton MT, Higashida R, Smith WS, Young WL, Saloner D. Phase-contrast magnetic resonance imaging measurements in intracranial aneurysms in vivo of flow patterns, velocity fields, and wall shear stress: comparison with computational fluid dynamics. Magn Reson Med 2009;61:409–17.
- [164] Corso P, Walheim J, Dillinger H, Giannakopoulos G, Gülan U, Frouzakis CE, Kozerke S, Holzner M. Toward an accurate estimation of wall shear stress from 4D flow magnetic resonance downstream of a severe stenosis. Magn Reson Med 2021;86(3):1531–43.
- [165] Zhang J, Rothenberger SM, Brindise MC, Markl M, Rayz VL, Vlachos P. Wall shear stress estimation for 4D flow MRI using Navier-Stokes equation correction. Ann Biomed Eng 2023;50(12):1810–25.
- [166] Evangelista A, Flachskampf FA, Erbel R, Antonini-Canterin F, Vlachopoulos C, Rocchi G, Sicari R, Nihoyannopoulos P, Zamorano J. European Association of Echocardiography, Echocardiography in aortic diseases: EAE recommendations for clinical practice. Eur J Echocardiogr 2010;11(8):645–58.
- [167] Evangelista A, Aguilar R, Cuellar H, Thomas M, Laynez A, Rodríguez-Palomares J, Mahía P, Gonzàlez-Alujas T, García-Dorado D. Usefulness of real-time threedimensional transoesophageal echocardiography in the assessment of chronic aortic dissection. Eur J Echocardiogr 2011;12(4):272–7.
- [168] Erbel R, Aboyans V, Boileau C, Bossone E, Bartolomeo RD, Eggebrecht H, Evangelista A, Falk V, Frank H, Gaemperli O, Grabenwöger M, Haverich A, Iung B, Manolis AJ, Meijboom F, Nienaber CA, Roffi M, Rousseau H, Sechtem U, Sirnes PA, Allmen RS, Vrints CJ. ESC Committee for Practice Guidelines, 2014 ESC guidelines on the diagnosis and treatment of aortic diseases: document covering acute and chronic aortic diseases of the thoracic and abdominal aorta of the adult. The Task Force for the Diagnosis and Treatment of Aortic Diseases of the European Society of Cardiology (ESC). Eur Heart J 2014;35(41):2873–926.
- [169] Erwin III JP, ACC Solution Set Oversight Committee, Cibotti-Sun M, Elma M. 2022 Aortic disease guideline-at-a-glance. J Am Coll Cardiol 2022;80(24): 2348–2352. https://doi.org/10.1016/j.jacc.2022.10.001.
- [170] Yuan X, Mitsis A, Mozalbat D, Nienaber CA. Alternative management of proximal aortic dissection: concept and application. Indian J Thorac Cardiovasc Surg 2022;38(Suppl):183–92.
- [171] Pape LA, Awais M, Woznicki EM, Suzuki T, Trimarchi S, Evangelista A, Myrmel T, Larsen M, Harris KM, Greason K, Di Eusanio M, Bossone E, Montgomery DG, Eagle KA, Nienaber CA, Isselbacher EM, O'Gara P. Presentation, diagnosis, and outcomes of acute aortic dissection: 17-year trends from the International Registry of Acute Aortic Dissection. J Am Coll Cardiol 2015;66(4):350–8.
- [172] Ihnken K, Sze D, Dake MD, Fleischmann D, Van der Starre P, Robbins R. Successful treatment of a Stanford type A dissection by percutaneous placement of a covered stent graft in the ascending aorta. J Thorac Cardiovasc Surg 2004;127(6):1808–10.
- [173] Sobocinski J, O'Brien N, Maurel B, Bartoli M, Goueffic Y, Sassard T, Midulla M, Koussa M, Vincentelli A, Haulon S. Endovascular approaches to acute aortic type A dissection: a CT-based feasibility study. Eur J Vasc Endovasc Surg 2011;42(4):442–7.
- [174] Balà N, Aranda A, Teixidó P, Molhoek C, Moreno-Jiménez I, Febas G, López-Guimet J, Groothuis A, Edelman ER, Balcells M, Borrós S, Martorell J, Riambau V. In vivo efficacy of an adhesive bioresorbable patch to treat aortic dissections. JACC Basic Transl Sci 2023;9(1):65–77. https://doi.org/10.1016/j. jacbts.2023.08.002.
- [175] Sultan S, Hynes N, Kavanagh EP, Diethrich EB. How does the multilayer flow modulator work? The science behind the technical innovation. J Endovasc Ther 2014;21(6):814–21.
- [176] Ibrahim W, Spanos K, Gussmann A, Nienaber CA, Tessarek J, Walter H, Thalwitzer J, Debus SE, Tsilimparis N, Kölbel T. Early and midterm outcome of multilayer flow modulator stent for complex aortic aneurysm treatment in Germany. J Vasc Surg 2018;68(4):956–64.
- [177] Lazaris AM, Maheras AN, Vasdekis SN. A multilayer stent in the aorta may not seal the aneurysm, thereby leading to rupture. J Vasc Surg 2012;3(3):829–31.[178] Sultan S, Hynes N, Kavanagh EP, Diethrich EB. Early mid-term results of the first 103 cases of multilayer flow modulator stent done under indication for use in
- the management of thoracoabdominal aortic pathology from the independent global MFM registry. J Cardiovasc Surg (Torino) 2014;55(1):21–32. [179] Hynes N, Sultan S, Elhelali A, Diethrich EB, Kavanagh EP, Sultan M, Stefanov F, Delassus P, Morris L. Systematic review and patient-level meta-analysis of the
- streamliner multilayer flow modulator in the management of complex thoracoabdominal aortic pathology. J Endovasc Ther 2016;23(3):501–12. [180] Oderich GS. Evidence of use of multilayer flow modulator stents in treatment of thoracoabdominal aortic aneurysms and dissections. J Vasc Surg 2017;65(4):
- 935-7.
- [181] Humphrey JD. Mechanisms of arterial remodeling in hypertension coupled roles of wall shear and intramural stress. Hypertension 2008;52(2):195–200.
- [182] Hayashi K, Naiki T. Adaptation and remodeling of vascular wall; biomechanical response to hypertension. J Mech Behav Biomed Mater 2009;2(1):3–19.
   [183] Bersi MR, Khosravi R, Wujciak AJ, Harrison DG, Humphrey JD. Differential cell-matrix mechanoadaptations and inflammation drive regional propensities to
- [183] Bersi MR, Knosravi R, Wujciak AJ, Harrison DG, Humphrey JD. Differential cell-matrix mechanoadapta aortic fibrosis, aneurysm or dissection in hypertension. J R Soc Interface 2017;14(136):20170327.
- [184] Humphrey JD, Harrison DG, Figueroa CA, Lacolley P, Laurent S. Central artery stiffness in hypertension and aging: a problem with cause and consequence. Circ Res 2016;118(3):379–81.
- [185] Boutouvrie P, Chowienczyk P, Humphrey JD, Mitchell GF. Arterial stiffness and cardiovascular risk in hypertension. Circ Res 2021;128:864–86.
- [186] Roccabianca S, Figueroa CA, Tellides G, Humphrey JD. Quantification of regional differences in aortic stiffness in the aging human. J Mech Behav Biomed Mater 2014;29. 648–34.
- [187] Ferruzzi J, Madziva D, Caulk A, Tellides G, Humphrey J. Compromised mechanical homeostasis in arterial aging and associated cardiovascular consequences. Biomech Model Mechanobiol 2018;17(5):1281–95.
- [188] Safar ME. Arterial aging-hemodynamic changes and therapeutic options. Nat Rev Cardiol 2010;7(8):442-9.
- [189] Stock S, Mohamed SA, Sievers H-H. Bicuspid aortic valve related aortopathy. Gen Thorac Cardiovasc Surg 2019;67:93–101.
- [190] Pinard A, Jones GT, Milewicz DM. Genetics of thoracic and abdominal aortic diseases. Circ Res 2019;124(4):588–606.
- [191] Creamer TJ, Bramel EE, MacFarlane EG. Insights on the pathogenesis of aneurysm through the study of hereditary aortopathies. Genes (Basel) 2021;12(2):183.[192] Superti-Furga A, Gugler E, Gitzelmann R, Steinmann B. Ehlers-Danlos syndrome type IV: a multi-exon deletion in one of the two COL3A1 alleles affecting
- structure, stability, and processing of type III procollagen. J Biol Chem 1988;263(13):6226–32.

- [193] Schwarze U, Schievink WI, Petty E, Jaff MR, Babovic-Vuksanovic D, Cherry KJ, Pepin M, Byers PH. Haploinsufficiency for one COL3A1 allele of type III procollagen results in a phenotype similar to the vascular form of Ehlers-Danlos syndrome, Ehlers-Danlos syndrome type IV. Am J Hum Genet 2001;69(5): 989–1001.
- [194] Pepin M, Schwarze U, Superti-Furga A, Byers PH. Clinical and genetic features of Ehlers-Danlos syndrome type IV, the vascular type. N Engl J Med 2000;342 (10):673–80.
- [195] Kontusaari S, Tromp G, Kuivaniemi H, Ladda RL, Prockop DJ. Inheritance of an RNA splicing mutation (G+ 1 IVS20) in the type III procollagen gene (COL3A1) in a family having aortic aneurysms and easy bruisability: phenotypic overlap between familial arterial aneurysms and Ehlers-Danlos syndrome type IV. Am J Hum Genet 1990;47(1):112–20.
- [196] Dietz HC, Cutting GR, Pyeritz RE, Maslen CL, Sakai LY, Corson GM, Puffenberger EG, Hamosh A, Nanthakumar EJ, Curristin SM. Marfan syndrome caused by a recurrent de novo missense mutation in the fibrillin gene. Nature 1991;352(6333):337–9.
- [197] Loeys BL, Chen J, Neptune ER, Judge DP, Podowski M, Holm T, Meyers J, Leitch CC, Katsanis N, Sharifi N, Xu FL, Myers LA, Spevak PJ, Cameron DE, De Backer J, Hellemans J, Chen Y, Davis EC, Webb CL, Kress W, Coucke P, Rifkin DB, De Paepe AM, Dietz HC. A syndrome of altered cardiovascular, craniofacial, neurocognitive and skeletal development caused by mutations in TGFBR1 or TGFBR2. Nat Genet 2005;37(3):275–81.
- [198] Zhu L, Vranckx R, Khau Van Kien P, Lalande A, Boisset N, Mathieu F, Wegman M, Glancy L, Gasc JM, Brunotte F, Bruneval P, Wolf JE, Michel JB, Jeunemaitre X. Mutations in myosin heavy chain 11 cause a syndrome associating thoracic aortic aneurysm/aortic dissection and patent ductus arteriosus. Nat Genet 2006;38(3):343–9.
- [199] Pannu H, Tran-Fadulu V, Papke CL, Scherer S, Liu Y, Presley C, Guo D, Estrera AL, Safi HJ, Brasier AR, Vick GW, Marian AJ, Raman CS, Buja LM, Milewicz DM. MYH11 mutations result in a distinct vascular pathology driven by insulin-like growth factor 1 and angiotensin II. Hum Mol Genet 2007;16(20):2453–62.
- [200] Guo D-C, Pannu H, Tran-Fadulu V, Papke CL, Yu RK, Avidan N, Bourgeois S, Estrera AL, Safi HJ, Sparks E, Amor D, Ades L, McConnell V, Willoughby CE, Abuelo D, Willing M, Lewis RA, Kim DH, Scherer S, Tung PP, Ahn C, Buja LM, Raman CS, Shete SS. Mutations in smooth muscle a-actin (ACTA2) lead to thoracic aortic aneurysms and dissections. Nat Genet 2007;39(12):1488–93.
- [201] Wang L, Guo D-C, Cao J, Gong L, Kamm KE, Regalado E, Li L, Shete S, He WQ, Zhu MS, Offermanns S, Gilchrist D, Elefteriades JA, Stull JT, Milewicz DM. Mutations in myosin light chain kinase cause familial aortic dissections. Am J Hum Genet 2010;87(5):701–7.
- [202] van de Laar IMBH, Oldenburg RA, Pals G, Roos-Hesselink JW, de Graaf BM, Verhagen JM, Hoedemaekers YM, Willemsen R, Severijnen LA, Venselaar H, Vriend G, Pattynama PM, Collée M, Majoor-Krakauer D, Poldermans D, Frohn-Mulder IM, Micha D, Timmermans J, Hilhorst-Hofstee Y, Bierma-Zeinstra SM, Willems PJ, Kros JM, Oei EHG, Oostra BA, Wessels MW, Bertoli-Avella AM. Mutations in SMAD3 cause a syndromic form of aortic aneurysms and dissections with early-onset osteoarthritis. Nat Genet 2011;43(2):121–6.
- [203] Lindsay ME, Schepers D, Bolar NA, Doyle JJ, Gallo E, Fert-Bober J, Kempers MJ, Fishman EK, Chen Y, Myers L, Bjeda D, Oswald G, Elias AF, Levy HP, Anderlid BM, Yang MH, Bongers EM, Timmermans J, Braverman AC, Canham N, Mortier GR, Brunner HG, Byers PH, Van Eyk J, Van Laer L, Dietz HC, Loeys BL. Loss-of-function mutations in TGFB2 cause a syndromic presentation of thoracic aortic aneurysm, Nat Genet 2012;44(8):922–7. https://doi.org/10.1038/ng.2349.
- [204] Boileau C, Guo DC, Hanna N, Regalado ES, Detaint D, Gong L, Varret M, Prakash SK, Li AH, d'Indy H, Braverman AC, Grandchamp B, Kwartler CS, Gouya L, Santos-Cortez RL, Abifadel M, Leal SM, Muti C, Shendure J, Gross MS, Rieder MJ, Vahanian A, Nickerson DA, Michel JB, National Heart, Lung, and Blood Institute (NHLBI) Go Exome Sequencing Project, Jondeau G, Milewicz DM. TGFB2 mutations cause familial thoracic aortic aneurysms and dissections associated with mild systemic features of Marfan syndrome. Nat Genet 2012;44(8):916–21. https://doi.org/10.1038/ng.2348.
- [205] Andrabi S, Bekheirnia MR, Robbins-Furman P, Lewis RA, Prior TW, Potocki L. SMAD4 mutation segregating in a family with juvenile polyposis, aortopathy, and mitral valve dysfunction. Am J Med Genet A 2011;155A(5):1165–9.
- [206] de Wit MC, de Coo IF, Lequin MH, Halley DJ, Roos-Hesselink JW, Mancini GM. Combined cardiological and neurological abnormalities due to filamin A gene mutation. Clin Res Cardiol 2011;100(1):45–50.
- [207] Reinstein E, Frentz S, Morgan T, García-Miñaúr S, Leventer RJ, McGillivray G, Pariani M, van der Steen A, Pope M, Holder-Espinasse M, Scott R, Thompson EM, Robertson T, Coppin B, Siegel R, Bret Zurita M, Rodríguez JJ, Morales C, Rodrígues Y, Arcas J, Saggar A, Horton M, Zackai E, Graham JM, Rimoin DL, Robertson SP. Vascular and connective tissue anomalies associated with X-linked periventricular heterotopia due to mutations in filamin A. Eur J Hum Genet 2013;21(5):494–502.
- [208] Guo D-C, Regalado ES, Casteel DE, Santos-Cortez RL, Gong L, Kim JJ, Dyack S, Horne SG, Chang G, Jondeau G, Boileau C, Coselli JS, Li Z, Leal SM, Shendure J, Rieder MJ, Bamshad MJ, Nickerson DA, GenTAC Registry Consortium, National Heart, Lung, and Blood Institute Grand Opportunity Exome Sequencing Project, Kim C, Milewicz DM. Recurrent gain-of-function mutation in PRKG1 causes thoracic aortic aneurysms and acute aortic dissections. Am J Hum Genet 2013;93(2):398–404. https://doi.org/10.1016/j.ajhg.2013.06.019.
- [209] Bertoli-Avella AM, Gillis E, Morisaki H, Verhagen JMA, de Graaf BM, van de Beek G, Gallo E, Kruithof BPT, Venselaar H, Myers LA, Laga S, Doyle AJ, Oswald G, van Cappellen GWA, Yamanaka I, van der Helm RM, Beverloo B, de Klein A, Pardo L, Lammens M, Evers C, Devriendt K, Dumoulein M, Timmermans J, Bruggenwirth HT, Verheijen F, Rodrigus I, Baynam G, Kempers M, Saenen J, Van Craenenbroeck EM, Minatoya K, Matsukawa R, Tsukube T, Kubo N, Hofstra R, Goumans MJ, Bekkers JA, Roos-Hesselink JW, van de Laar IMBH, Dietz HC, Van Laer L, Morisaki T, Wessels MW, Loeys BL. Mutations in a TGF-β ligand, TGFE3, cause syndromic aortic aneurysms and dissections. J Am Coll Cardiol 2015;65(13):1324–36.
- [210] Lee VS, Halabi CM, Hoffman EP, Stitziel NO. Loss of function mutation in LOX causes thoracic aortic aneurysm and dissection in humans. Proc Natl Acad Sci 2016;113(31):8759–64.
- [211] Guo D-C, Regalado ES, Gong L, Duan X, Santos-Cortez RLP, Arnaud P, Ren Z, Cai B, Hostetler EM, Moran R, Liang D, Estrera AL, Safi HJ, University of Washington Center for Mendelian Genomics, Leal SM, Bamshad MJ, Shendure J, Nickerson DA, Jondeau G, Boileau C, Milewicz DM. LOX mutations predispose to thoracic aortic aneurysms and dissections. Circ Res 2016:118(6):928–34. https://doi.org/10.1161/CIRCRESAHA.115.307130.
- [212] Duan XY, Guo D-C, Regalado ES, Shen H, University of Washington Center for Mendelian Genomics, Coselli JS, Estrera AL, Safi HJ, Bamshad MJ, Nickerson DA, LeMaire SA, De Backer J, Milewicz DM. SMAD4 rare variants in individuals and families with thoracic aortic aneurysms and dissections. Eur J Hum Genet 2019;27(7):1054–1060. https://doi.org/10.1038/s41431-019-0357-x.
- [213] Dubacher N, Münger J, Gorosabel MC, Crabb J, Ksiazek AA, Caspar SM, Bakker ENTP, van Bavel E, Ziegler U, Carrel T, Steinmann B, Zeisberger S, Meienberg J, Matyas G. Celiprolol but not losartan improves the biomechanical integrity of the aorta in a mouse model of vascular Ehlers-Danlos syndrome. Cardiovasc Res 2020;116(2):457–65.
- [214] Lee B, Godfrey M, Vitale E, Hori H, Mattei MG, Sarfarazi M, Tsipouras P, Ramirez F, Hollister DW. Linkage of Marfan syndrome and a phenotypically related disorder to two different fibrillin genes. Nature 1991;352(6333):330–4.
- [215] Maslen CL, Corson GM, Maddox BK, Glanville RW, Sakai LY. Partial sequence of a candidate gene for the Marfan syndrome. Nature 1991;352(6333):334–7.
   [216] Pereira L, Andrikopoulos K, Tian J, Lee SY, Keene DR, Ono R, Reinhardt DP, Sakai LY, Biery NJ, Bunton T, Dietz HC, Ramirez F. Targetting of the gene
- encoding fibrillin–1 recapitulates the vascular aspect of Marfan syndrome. Nat Genet 1997;17(2):218–22.
- [217] Humphrey JD, Tellides G. Central artery stiffness and thoracic aortopathy. Am J Physiol Heart Circ Physiol 2019;316(1):H169–82.[218] Jondeau G, Boutouyrie P, Lacolley P, Laloux B, Dubourg O, Bourdarias JP, Laurent S. Central pulse pressure is a major determinant of ascending aorta dilation
- in Marfan syndrome. Circulation 1999;99(20):2677–81.
   [219] Pannu H, Fadulu VT, Chang J, Lafont A, Hasham SN, Sparks E, Giampietro PF, Zaleski C, Estrera AL, Safi HJ, Shete S, Willing MC, Raman CS, Milewicz DM. Mutations in transforming growth factor-*β* receptor type II cause familial thoracic aortic aneurysms and dissections. Circulation 2005;112(4):513–20.
- [220] Kuechler A, Altmüller J, Nürnberg P, Kotthoff S, Kubisch C, Borck G. Exome sequencing identifies a novel heterozygous TGFB3 mutation in a disorder overlapping with Marfan and Loeys-Dietz syndrome. Mol Cell Probes 2015;29(5):330–4.
- [221] Regalado ES, Guo D-C, Villamizar C, Avidan N, Gilchrist D, McGillivray B, Clarke L, Bernier F, Santos-Cortez RL, Leal SM, Bertoli-Avella AM, Shendure J, Rieder MJ, Nickerson DA, National Heart, Lung, and Blood Institute (NHLBI) Go Exome Sequencing Project, Milewicz DM. Exome sequencing identifies SMAD3 mutations as a cause of familial thoracic aortic aneurysm and dissection with intracranial and other arterial aneurysms. Circ Res 2011:109(6):680–6. https:// doi.org/10.1161/CIRCRESAHA.111.248161.

- [222] Li Q, Muragaki Y, Hatamura I, Ueno H, Ooshima A. Stretch-induced collagen synthesis in cultured smooth muscle cells from rabbit aortic media and a possible involvement of angiotensin II and transforming growth factor-β. J Vasc Res 1998;35(2):93–103.
- [223] Wipff P-J, Hinz B. Integrins and the activation of latent transforming growth factor  $\beta$ 1 an intimate relationship. Eur J Cell Biol 2008;87(8–9):601–15.
- [224] Henderson NC, Sheppard D. Integrin-mediated regulation of  $TGF\beta$  in fibrosis. Biochim Biophys Acta 2013;1832(7):891–6.

[225] Chen MH, Choudhury S, Hirata M, Khalsa S, Chang B, Walsh CA. Thoracic aortic aneurysm in patients with loss of function filamin A mutations: clinical characterization, genetics, and recommendations. Am J Med Genet A 2018;176(2):337–50.

- [226] Zheng HQ, Rong JB, Ye FM, Xu YC, Lu HS, Wang JA. Induction of thoracic aortic dissection: a mini-review of β-aminopropionitrile-related mouse models. J Zhejiang Univ Sci B 2020;21(8):603–10.
- [227] Sawada H, Beckner ZA, Ito S, Daugherty A, Lu HS. & Aminopropionitrile-induced aortic aneurysm and dissection in mice. JVS Vasc Sci 2022;3:64–72.
- [228] Humphrey JD, Milewicz DM, Tellides G, Schwartz MA. Dysfunctional mechanosensing in aneurysms. Science 2014;344(6183):477–9.
- [229] Sengle G, Sakai LY. The fibrillin microfibril scaffold: a niche for growth factors and mechanosensation? Matrix Biol 2015;47:3–12.
- [230] Milewicz DM, Trybus KM, Guo D-C, Sweeney HL, Regalado E, Kamm K, Stull JT. Altered smooth muscle cell force generation as a driver of thoracic aortic aneurysms and dissections. Arterioscler Thromb Vasc Biol 2017;37(1):26–34.
- [231] Milewicz DM, Prakash SK, Ramirez F. Therapeutics targeting drivers of thoracic aortic aneurysms and acute aortic dissections: insights from predisposing genes and mouse models. Annu Rev Med 2017;68(1):51–67.
- [232] Ramirez F, Caescu C, Wondimu E, Galatioto J. Marfan syndrome; a connective tissue disease at the crossroads of mechanotransduction, TGF// signaling and cell stemness. Matrix Biol 2018;71–72:82–9.
- [233] Yamashiro Y, Yanagisawa H. Crossing bridges between extra- and intra-cellular events in thoracic aortic aneurysms. J Atheroscler Thromb 2018;25(2):99–110.
- [234] Humphrey JD, Dufresne E, Schwartz MA. Mechanotransduction and extracellular matrix homeostasis. Nat Rev Mol Cell Biol 2014;15(12):802–12.
- [235] Yamashiro Y, Yanagisawa H. The molecular mechanism of mechanotransduction in vascular homeostasis and disease. Clin Sci (Lond) 2020;134(17):2399–418.
   [236] Humphrey JD. Stress, strain, and mechanotransduction in cells. J Biomech Eng 2001;123(6):638–41.
- [237] Humphrey JD. Constrained mixture models of soft tissue growth and remodeling twenty years after. J Elast 2021;145(1-2):49-75.
- [238] Irons L, Latorre M, Humphrey JD. From transcript to tissue: multiscale modeling from cell signaling to matrix remodeling. Ann Biomed Eng 2021;49(7): 1701–15.
- [239] Morrison TM, Choi G, Zarins CK, Taylor CA. Circumferential and longitudinal cyclic strain of the human thoracic aorta: age-related changes. J Vasc Surg 2009; 49(4):1029–36.
- [240] Jadidi M, Habibnezhad M, Anttila E, Maleckis K, Desyatova A, MacTaggart J, Kamenskiy A. Mechanical and structural changes in human thoracic aortas with age. Acta Biomater 2020;103:172–88.
- [241] Immer FF, Bansi AG, Immer-Bansi AS, McDougall J, Zehr KJ, Schaff HV, Carrel TP. Aortic dissection in pregnancy: analysis of risk factors and outcome. Ann Thorac Surg 2003;76(1):309–14.
- [242] Rajagopalan S, Nwazota N, Chandrasekhar S. Outcomes in pregnant women with acute aortic dissections: a review of the literature from 2003 to 2013. Int J Obstet Anesth 2014;23(4):348–56.
- [243] Kamel H, Roman MJ, Pitcher A, Devereux RB. Pregnancy and the risk of aortic dissection or rupture: a cohort-crossover analysis. Circulation 2016;134(7): 527–33.
- [244] Zhu JM, Ma WG, Peterss S, Wang LF, Qiao ZY, Ziganshin BA, Zheng J, Liu YM, Elefteriades JA, Sun LZ. Aortic dissection in pregnancy: management strategy and outcomes. Ann Thorac Surg 2017;103(4):1199–206.
- [245] De Martino A, Morganti R, Falcetta G, Scioti G, Milano AD, Pucci A, Bortolotti U. Acute aortic dissection and pregnancy: review and meta-analysis of incidence, presentation, and pathologic substrates. J Card Surg 2019;34(12):1591–7.
- [246] Habashi JP, MacFarlane EG, Bagirzadeh R, Bowen C, Huso N, Chen Y, Bedja D, Creamer TJ, Rykiel G, Manning M, Huso D, Dietz HC. Oxytocin antagonism prevents pregnancy-associated aortic dissection in a mouse model of Marfan syndrome. Sci Transl Med 2019;11(490):eaat4822.
- [247] LePage MA, Quint LE, Sonnad SS, Deeb GM, Williams DM. Aortic dissection: CT features that distinguish true lumen from false lumen. AJR Am J Roentgenol 2001;1(177):207–11.
- [248] McMahon MA, Squirrel CA. Multidetector CT of aortic dissection: a pictorial review. Radiographics 2010;30(2):445-60.
- [249] Moullet P, Mann H. Beak sign. J Thorac Imaging 2014;29(6):W90.
- [250] Amabili M, Arena GO, Balasubramanian P, Breslavsky ID, Cartier R, Ferrari G, Holzapfel GA, Kassab A, Mongrain R. Biomechanical characterization of a chronic type A dissected human aorta. J Biomech 2020;110:109978.
- [251] Williams DM, Brothers TW, Messina LM. Relief of mesenteric ischemia in type III aortic dissection with percutaneous fenestration of the aortic septum. Radiology 1990;174(2):450–2.
- [252] Lortz J, Papathanasiou M, Rammos C, Steinmetz M, Lind A, Tsagakis K, Schlosser T, Jakob H, Rassaf T, Jániso RA. High intimal flap mobility assessed by intravascular ultrasound is associated with better short-term results after TEVAR in chronic aortic dissection. Sci Rep 2019;9:7267.
- [253] Lescan M, Mustafi M, Wilhelm V, Keller M, Schlensak C, Rosenberger P, Magunia H. The impact of dissection membrane motility on mid-term aortic remodelling after thoracic endovascular repair. Eur J Cardiothorac Surg 2022;61(4):869–76.
- [254] Berguer R, Parodi JC, Schlicht M, Khanafer K. Experimental and clinical evidence supporting septectomy in the primary treatment of acute type B thoracic aortic dissection. Ann Vasc Surg 2015;29(2):167–73.
- [255] Hamada S, Takamiya M, Kimura K, Imakita S, Nakajima N, Naito H. Type A aortic dissection: evaluation with ultrafast CT. Radiology 1992;183(1):155–8.
- [256] Quint LE, Platt JF, Sonnad SS, Deeb GM, Williams DM. Aortic intimal tears: detection with spiral computed tomography. J Endovasc Ther 2003;10(3):505–10.
- [257] Ergin MA, Galla JD, Lansman S, Griepp RB. Acute dissections of the aorta. Current surgical treatment, Surg Clin North Am 1985;65(3):721–41.
- [258] Hayashi K, Meaney TF, Zelch JV, Tarar R. Aortographic analysis of aortic dissection. Am J Roentgenol Radium Ther Nucl Med 1974;122(4):769–82.
   [259] Itzchak Y, Rosenthal T, Adar R, Rubenstein ZJ, Lieberman Y, Deutsch V. Dissecting aneurysm of thoracic aorta: reappraisal of radiologic diagnosis. Am J
- Roentgenol Radium Ther Nucl Med 1975;125(3):559–70.
- [260] Dinsmore RE, Wedeen VJ, Miller SW, Rosen BR, Fifer M, Vlahakes GJ, Edelman RR, Brady TJ. MRI of dissection of the aorta: recognition of the intimal tear and differential flow velocities. AJR Am J Roentgenol 1986;146(6):1286–8.
- [261] Roberts WC. Aortic dissection: anatomy, consequences, and causes. Am Heart J 1981;101(2):195-214.
- [262] Segesser L, Killer I, Ziswiler M, Linka A, Ritter M, Jenni R, Baumann P, Turina M. Dissection of the descending thoracic aorta extending into the ascending aorta. J Thorac Cardiovasc Surg 1994;108:755–61.
- [263] Khoynezhad A, Walot I, Kruse MJ, Rapae T, Donayre CE, White RA. Distribution of intimomedial tears in patients with type B aortic dissection. J Vasc Surg 2010;52(3):562–8.
- [264] Evangelista A, Salas A, Ribera A, Ferreira-González I, Cuellar H, Pineda V, González-Alujas T, Bijnens B, Permanyer-Miralda G, Garcia-Dorado D. Long-term outcome of aortic dissection with patent false lumen: predictive role of entry tear size and location. Circulation 2012;125(25):3133–41.
- [265] Cuellar-Calabria H, Burcet G, Roque A, Rodríguez-Palomares J, Teixidó G, Rodríguez R, Bellmunt S, Zebdi N, Reyes-Juárez J, Sao-Avilés A, Escobar M, Evangelista A. Differences in the area of proximal and distal entry tears at CT angiography predict long-term clinical outcomes in aortic dissection. Radiol Cardiothorac Imaging 2021;3(6):e210029.
- [266] Weiss G, Wolner I, Folkmann S, Sodeck G, Schmidli J, Grabenwöger M, Carrel T, Czerny M. The location of the primary entry tear in acute type B aortic dissection affects early outcome. Eur J Cardiothorac Surg 2012;42(3):571–6.
- [267] Dziodzio T, Juraszek A, Reineke D, Jenni H, Zermatten E, Zimpfer D, Stoiber M, Scheikl V, Schima H, Grimm M, Czerny M. Experimental acute type B aortic dissection: different sites of primary entry tears cause different ways of propagation. Ann Thorac Surg 2011;91(3):724–7.
- [268] Peery TM. Dissecting aneurysm: with a report of five cases. Am Hear J 1936;12(6):650–65.
- [269] Shennan T. Dissecting aneurysms. Med Res Cncl Special Report Series 1934;193:138.

- [270] Sailer AM, van Kuijk SM, Nelemans PJ, Chin AS, Kino A, Huininga M, Schmidt J, Mistelbauer G, Bäumler K, Chiu P, Fischbein MP, Dake MD, Miller DC, Schurink GW, Fleischmann D. Computed tomography imaging features in acute uncomplicated Stanford type-B aortic dissection predict late adverse events. Circ Cardiovasc Imaging 2017;10(4):e005709.
- [271] Hukill P-B. Healed dissecting aneurysm in cystic medial necrosis of the aorta. Circulation 1957;15(4):540-6.
- [272] Thubrikar M, Agali P, Robicsek F. Wall stress as a possible mechanism for the development of transverse intimal tears in aortic dissections. J Med Eng Technol 1999;23(4):127–34.
- [273] Baron MG. Dissecting aneurysm of the aorta. Circulation 1971;43(6):933-43.
- [274] Williams DM, LePage M, Lee DY. The dissected aorta: part I. Early anatomic changes in an in vitro model. Radiology 1997;203(1):23-31.
- [275] Lee DY, Williams DM, Abrams GD. The dissected aorta: part II. Differentiation of the true from the false lumen with intravascular US. Radiology 1997;32(1): 32-6.
- [276] Weintraub AR, Erbel R, Görge G, Schwartz SL, Ge J, Gerber T, Meyer J, Hsu TL, Bojar R, Iliceto S, et al. Intravascular ultrasound imaging in acute aortic dissection. J Am Coll Cardiol 1994;24(2):495–503.
- [277] Yang S, Li X, Chao B, Wu L, Cheng Z, Duan Y, Wu D, Zhan Y, Chen J, Liu B, Ji X, Nie P, Wang X. Abdominal aortic intimal flap motion characterization in acute aortic dissection: assessed with retrospective ECG-gated thoracoabdominal aorta dual-source CT angiography. PloS One 2014;9(2):e87664.
- [278] Ganten M-K, Weber TF, von Tengg-Kobligk H, Böckler D, Stiller W, Geisbüsch P, Kauffmann GW, Delorme S, Bock M, Kauczor H-U. Motion characterization of aortic wall and intimal flap by ECG-gated CT in patients with chronic B-dissection. Eur J Radiol 2009;72(1):146–53.
- [279] Awal SS, Prasad N, Biswas S. CT evaluation of aortic dissection and other acute aortic syndromes: an update. Int J Radiol Radiat Ther 2022;9(5):159–65.
  [280] Sueyoshi E, Nagayama H, Sakamoto I, Ariyoshi T, Hashizume K, Eishi K, Uetani M. Triple-barreled aortic dissection developing into quadruple-barreled dissection. Ann Thorac Surg 2012;93(4):1320.
- [281] Murayama T, Funabashi N, Uehara M, Takaoka H, Komuro I. New classification of aortic dissection during the cardiac cycle as pulsating type and static type evaluated by electrocardiogram-gated multislice CT. Int J Cardiol 2010;142(2):177–86. https://doi.org/10.1016/j.ijcard.2008.12.194.
- [282] Williams DM, Lee DY, Hamilton BH, Marx MV, Narasimham DL, Kazanjian SN, Prince MR, Andrews JC, Cho KJ, Deeb GM. The dissected aorta: part III. Anatomy and radiologic diagnosis of branch-vessel compromise. Radiology 1997;203(1):37–44.
- [283] Tang DG, Dake MD. TEVAR for acute uncomplicated aortic dissection: immediate repair versus medical therapy. Semin Vasc Surg 2009;22(3):145–51.
   [284] Lou X, Chen EP, Duwayri YM, Veeraswamy RK, Jordan Jr WD, Zehner CA, Leshnower BG. The impact of thoracic endovascular aortic repair on long-term survival in type B aortic dissection. Ann Thorac Surg 2018;105(1):31–8.
- [285] Xiang D, Wu F, Chen L, Liang H, Xiong B, Liang B, Yang F, Zheng C. Timing of endovascular repair impacts long-term outcomes of uncomplicated acute type B aortic dissection. J Vasc Surg 2022;75(3):851–860.e3.
- [286] Cramer GG, Amplatz K. Catheter position: an aid in diagnosis of dissecting aneurysm of the thoracic aorta. Am J Roentgenol Radium Ther Nucl Med 1966;98 (4):836–9.
- [287] Williams DM, Joshi A, Dake MD, Deeb GM, Miller DC, Abrams GD. Aortic cobwebs: an anatomic marker identifying the false lumen in aortic dissection-imaging and pathologic correlation. Radiology 1994;190(1):167–74.
- [288] Chen D, Matsunaga F. Aortic cobweb sign. Abdom Radiol (NY) 2022;47(4):1505-6.
- [289] Karabulut N, Goodman LR, Olinger GN. CT diagnosis of an unusual aortic dissection with intimointimal intussusception: the wind sock sign. J Comput Assist Tomogr 1998;22(5):692–3.
- [290] Touati G, Carmi D, Trojette F, Jarry G. Intimo-intimal intussusception: a rare clinical form of aortic dissection. Eur J Cardiothorac Surg 2003;23(1):119–21.
   [291] Gomes de Farias LP, Santos JMMM, Teles GBDS, Baptista LPS. Intimointimal intussusception in acute aortic dissection. Radiol Cardiothorac Imaging 2020;2(4): e200023
- [292] Yamashita Y, Nakagawa S, Kitamoto S, Sakamoto K, Horii T. A case of circumferential type A aortic dissection with intimo-intimal intussusception. Ann Vasc Dis 2020;13(4):426–9.
- [293] Daou R, Khater DA, Khattar R, Helou M. Aortic dissection presenting as a stroke: a case report. Pan Afr Med J 2023;44:91.
- [294] Chan SM, Brahmandam A, Cardella JA, Elefteriades JA, Setaro JF, Mangi AA, Chaar CIO. Aortic remodeling and competitive flow after surgical treatment of aortic dissection. J Vasc Surg Cases Innov Tech 2021;7(3):404–7.
- [295] Lopes A, Gouveia Melo R, Gomes ML, Garrido P, Junqueira N, Sobrinho G, Fernandes e Fernandes R, Leitão J, Nobre A, Pedro LM. Aortic dissection repair using the STABILISE technique associated with arch procedures: report of two cases. EJVES Short Rep 2019;42:26–30.
- [296] Dai M, Liu T, Luo Y, Zhou F, Fan H, Zhu J, Zhang Y, Hu F, Dai X. Treatment of aortic dissecting aneurysm involving visceral arteries with multi-layer bare stents. Rev Assoc Med Bras (1992) 2019;65(2):216–21.
- [297] Sultan S, Acharya Y, Chua V, Long K, Hatem M, Hezima M, Veerasingham D, Soliman O, Hynes N. Management of acute aortic syndrome with evolving individualized precision medicine solutions: lessons learned over two decades and literature review. Front Surg 2023;10:1157457.
- [298] Anwar MA, Hamady M. Various endoluminal approaches available for treating pathologies of the aortic arch. Cardiovasc Intervent Radiol 2020;43:1756-69.
   [299] Ma T, Dong ZH, Fu WG, Guo DQ, Xu X, Chen B, Jiang JH, Yang J, Shi ZY, Zhu T, Shi Y, Jiang BH, Xu XY. Incidence and risk factors for retrograde type A dissection and stent graft-induced new entry after thoracic endovascular aortic repair. J Vasc Surg 2018;67(4):1026-33.
- [300] Fitridge R, editor. Mechanisms of Vascular Disease: A Textbook for Vascular Specialists. Springer Cham; 2020.
- [301] Pape LA, Tsai TT, Isselbacher EM, Oh JK, O'Gara PT, Evangelista A, Fattori R, Meinhardt G, Trimarchi S, Bossone E, Suzuki T, Cooper JV, Froehlich JB, Nienaber CA, Eagle KA. Aortic diameter >or= 5.5 cm is not a good predictor of type A aortic dissection. Circulation 2007;116(10):1120–7. https://doi.org/10. 1161/CIRCULATIONAHA.107.702720.
- [302] Vorp DA, Raghavan ML, Webster MW. Mechanical wall stress in abdominal aortic aneurysm: influence of diameter and asymmetry. J Vasc Surg 1998;27(4): 632–9.
- [303] Vande Geest JP, Wang DH, Wisniewski SR, Makaroun MS, Vorp DA. Towards a noninvasive method for determination of patient-specific wall strength distribution in abdominal aortic aneurysms. Ann Biomed Eng 2006;34(7):1098–106. https://doi.org/10.1007/s10439-006-9132-6.
- [304] Durham CA, Cambria RP, Wang LJ, Ergul EA, Aranson NJ, Patel VI, Conrad MF. The natural history of medically managed acute type B aortic dissection. J Vasc Surg 2015;61. 1192–1189.
- [305] Davies RR, Goldstein LJ, Coady MA, Tittle LS, Rizzo JA, Kopf GS, Elefteriades JA. Yearly rupture or dissection rates for thoracic aortic aneurysms: simple prediction based on size. Ann Thorac Surg 2002;73(1):17–27;discussions 27–8. https://doi.org/10.1016/S0003-4975(01)03236-2.
- [306] Fattori R, Cao P, De Rango P, Czerny M, Evangelista A, Nienaber C, Rousseau H, Schepens M. Interdisciplinary expert consensus document on management of type B aortic dissection. J Am Coll Cardiol 2013;61(16):1661–78.
- [307] Song J-M, Kim S-D, Kim J-H, Kim M-J, Kang D-H, Seo JB, Lim T-H, Lee JW, Song M-G, Song J-K. Long-term predictors of descending aorta aneurysmal change in patients with aortic dissection. J Am Coll Cardiol 2007;50(8):799–804.
- [308] Trimarchi S, Tolenaar JL, Jonker FHW, Murray B, Tsai TT, Eagle KA, Rampoldi V, Verhagen HJM, van Herwaarden JA, Moll FL, Muhs BE, Elefteriades JA. Importance of false lumen thrombosis in type B aortic dissection prognosis. J Thorac Cardiovasc Surg 2013;145(3 Suppl):S208–12.
- [309] Tolenaar JL, van Keulen JW, Jonker FHW, van Herwaarden JA, Verhagen HJ, Moll FL, Muhs BE, Trimarchi S. Morphologic predictors of aortic dilatation in type B aortic dissection. J Vasc Surg 2013;58(5):1220–5.
- [310] Burris NS, Hoff BA, Kazerooni EA, Ross BD. Vascular deformation mapping (VDM) of thoracic aortic enlargement in aneurysmal disease and dissection. Tomography 2017;3(3):163–73.
- [311] Burris NS, Bian Z, Dominic J, Zhong J, Houben IB, van Bakel TMJ, Patel HJ, Ross BD, Christensen GE, Hatt CR. Vascular deformation mapping for CT surveillance of thoracic aortic aneurysm growth. J Vasc Interv Radiol 2022;302(1):218–225. https://doi.org/10.1148/radiol.2021210658.
- [312] Khabaz K, Yuan K, Pugar J, Jiang D, Sankary S, Dhara S, Kim J, Kang J, Nguyen N, Cao K, Washburn N, Bohr N, Lee CJ, Kindlmann G, Milner R, Pocivavsek L. The geometric evolution of aortic dissections: predicting surgical success using fluctuations in integrated Gaussian curvature. PLoS 2024;20(2):e1011815.

- [313] Marui A, Mochizuki T, Mitsui N, Koyama T, Kimura F, Horibe M. Toward the best treatment for uncomplicated patients with type B acute aortic dissection: a consideration for sound surgical indication. Circulation 1999;100(19 Suppl):II275–80. https://doi.org/10.1161/01.cir.100.suppl\_2.ii-275.
- [314] Jonker FHW, Trimarchi S, Rampoldi V, Patel HJ, O'Gara P, Peterson MD, Fattori R, Moll FL, Voehringer M, Pyeritz RE, Hutchison S, Montgomery D, Isselbacher EM, Nienaber CA, Eagle KA. Aortic expansion after acute type B aortic dissection. Ann Thorac Surg 2012;94:1223–9.
- [315] Kotelis D, Grebe G, Kraus P, Müller-Eschner M, Bischoff M, von Tengg-Kobligk H, Böckler D. Morphologic predictors of aortic expansion in chronic type B aortic dissection. Vascular 2016;24(2):187–93.
- [316] Kamman AV, Jonker FHW, Sechtem U, Harris KM, Evangelista A, Montgomery DG, Patel HJ, Eagle KA, Trimarchi S. IRAD investigators, Predictors of stable aortic dimensions in medically managed acute aortic syndromes. Ann Vasc Surg 2017;42:143–9.
- [317] M. Kato, H.-Z. Bai, K. Sato, S. Kawamoto, M. Kaneko, T. Ueda, D. Kishi, K. Ohnishi, Determining surgical indications for acute type B dissection based on enlargement of aortic diameter during the chronic phase. Circulation 1995;92(9 Suppl):II107–12. https://doi.org/10.1161/01.cir.92.9.107.
- [318] S. Miyahara, N. Mukohara, M. Fukuzumi, N. Morimoto, H. Murakami, K. Nakagiri, M. Yoshida, Long-term follow-up of acute type B aortic dissection: ulcer-like projections in thrombosed false lumen play a role in late aortic events. J Thorac Cardiovas Surg 2011;142(2):e25–31. https://doi.org/10.1016/j.jtcvs.2011.02. 015.
- [319] Miller III CC, Sandhu HK, Charlton-Ouw KM, Azizzadeh A, Estrera AL, Leake S, Safi HJ. Patient height as a risk factor for poor outcome in acute type B aortic dissection. J Cardiovasc Surg (Torino) 2017;58(4):551–6.
- [320] Evangelista A, Pineda V, Guala A, Bijnens B, Cuellar H, Rudenick P, Sao-Aviles A, Ruiz A, Teixido-Tura G, Rodriguez-Lecoq R, Bellmunt S, Ferreira I, Rodríguez-Palomares J. False lumen flow assessment by magnetic resonance imaging and long-term outcomes in uncomplicated aortic dissection. J Am Coll Cardiol 2022; 79(24):2415–27.
- [321] Chang CP, Liu JC, Liou YM, Chang SS, Chen JY. The role of false lumen size in prediction of in-hospital complications after acute type B aortic dissection. J Am Coll Cardiol 2008;52(14):1170–6.
- [322] Kamman AV, Brunkwall J, Verhoeven EL, Heijmen RH, Trimarchi S. ADSORB trialists, Predictors of aortic growth in uncomplicated type B aortic dissection from the Acute Dissection Stent Grafting or Best Medical Treatment (ADSORB) database. J Vasc Surg 2017;65(4):964–971.e3.
- [323] Lavingia KS, Larion S, Ahanchi SS, Ammar CP, Bhasin M, Mirza AK, Dexter DJ, Panneton JM. Volumetric analysis of the initial index computed tomography scan can predict the natural history of acute uncomplicated type B dissections. J Vasc Surg 2015;62(4):893–9.
- [324] Delsart P, Beregi JP, Devos P, Haulon S, Midulla M, Mounier-Vehier C. Thrombocytopenia: an early marker of late mortality in type B aortic dissection. Heart Vessels 2014;29(2):220–30.
- [325] Ray HM, Durham CA, Ocazionez D, Charlton-Ouw KM, Estrera AL, Miller 3rd CC, Safi HJ, Azizzadeh A. Predictors of intervention and mortality in patients with uncomplicated acute type B aortic dissection. J Vasc Surg 2016;64(6):1560–8.
- [326] Sueyoshi E, Sakamoto I, Hayashi K, Yamaguchi T, Imada T. Growth rate of aortic diameter in patients with type B aortic dissection during the chronic phase. Circulation 2004;110(11 Suppl 1):II256–261. https://doi.org/10.1161/01.CIR.0000138386.48852.b6.
- [327] Sueyoshi E, Sakamoto I, Uetani M. Growth rate of affected aorta in patients with type B partially closed aortic dissection. Ann Thorac Surg 2009;88(4):1251-7.
- [328] Loewe C, Czerny M, Sodeck GH, Ta J, Schoder M, Funovics M, Dumfarth J, Ehrlich M, Grimm M, Lammer J. A new mechanism by which an acute type B aortic dissection is primarily complicated, becomes complicated, or remains uncomplicated. Ann Thorac Surg 2012;93(4):1215–22.
- [329] Krähenbühl E, Maksimovic S, Sodeck G, Reineke D, Schoenhoff F, Schmidli J, Carrel T, Czerny M. What makes the difference between the natural course of a remaining type B dissection after type A repair and a primary type B aortic dissection? Eur J Cardiothorac Surg 2012;41(5):e110–5.
   [330] Ueki C, Sakaguchi G, Shimamoto T, Komiya T. Prognostic factors in patients with uncomplicated acute type B aortic dissection. Ann Thorac Surg 2014;97(3):
- [330] Ueki C, Sakaguchi G, Shimamoto T, Komiya T. Prognostic factors in patients with uncomplicated acute type B aortic dissection. Ann Thorae Surg 2014;97(3): 767–73.
- [331] Kitamura T, Torii S, Oka N, Horai T, Itatani K, Yoshii T, Nakamura Y, Shibata M, Tamura T, Araki H, Matsunaga Y, Sato H, Miyaji K. Impact of the entry site on late outcome in acute Stanford type B aortic dissection. Eur J Cardiothorac Surg 2015;48(5):655–61.
- [332] Tolenaar JL, van Keulen JW, Trimarchi S, Jonker FHW, van Herwaarden JA, Verhagen HJM, Moll FL, Muhs BE. Number of entry tears is associated with aortic growth in type B dissections. Ann Thorac Surg 2013;96(1):39–42.
- [333] Suzuki T, Mehta RH, Ince H, Nagai R, Sakomura Y, Weber F, Sumiyoshi T, Bossone E, Trimarchi S, Cooper JV, Smith DE, Isselbacher EM, Eagle KA, Nienaber CA. Clinical profiles and outcomes of acute type B aortic dissection in the current era: lessons from the International Registry of Aortic Dissection (IRAD). Circulation 2003;108(Suppl 1):II312–7. https://doi.org/10.1161/01.cir.0000087386.07204.09.
- [334] Trimarchi S, Eagle KA, Nienaber CA, Pyeritz RE, Jonker FH, Suzuki T, O'Gara PT, Hutchinson SJ, Rampoldi V, Grassi V, Bossone E, Muhs BE, Evangelista A, Tsai TT, Froehlich JB, Cooper JV, Montgomery D, Meinhardt G, Myrmel T, Upchurch GR, Sundt TM, Isselbahcer EM. Importance of refractory pain and hypertension in acute type B aortic dissection: insights from the International Registry of Acute Aortic Dissection (IRAD). Circulation 2010;122(13):1283–9.
- [335] Tolenaar JL, Froehlich W, Jonker FH, Upchurch Jr GR, Rampoldi V, Tsai TT, Bossone E, Evangelista A, O'Gara P, Pape L, Montgomery D, Isselbacher EM, Nienaber CA, Eagle KA, Trimarchi S. Predicting in-hospital mortality in acute type B aortic dissection: evidence from International Registry of Acute Aortic Dissection. Circulation 2014;130(11 Suppl 1):S45–50.
- [336] Kinoshita T, Asai T, Suzuki T. Outer media thickness at false lumen and secondary aortic dilatation after acute aortic dissection. Ann Thorac Surg 2022;114(5): 1646–54.
- [337] Uchida K, Imoto K, Takahashi M, Suzuki S, Isoda S, Sugiyama M, Kondo J, Takanashi Y. Pathologic characteristics and surgical indications of superacute type A intramural hematoma. Ann Thorac Surg 2005;79(5):1518–21.
- [338] Houben IB, Nama N, Moll FL, van Herwaarden JA, Nordsletten DA, Williams DM, Patel HJ, Figueroa CA, Burris NS. Mapping pre-dissection aortic wall abnormalities: a multiparametric assessment. Eur J Cardiothorac Surg 2020;57:1061–7.
- [339] Marway PS, Campello Jorge CA, Tjahjadi N, Baker TJ, Mistelbauer G, Baeumler K, Hinostroza V, Higashigaito K, Mastrodicasa D, Masotti M, Nordsletten D, Patel HJ, Fleischmann D, Burris NS. Early 3D growth in uncomplicated type B aortic dissection is associated with long-term outcomes. J Vasc Surg 2024;S0741-5214(24)01819-6.
- [340] Mistelbauer G, Schmidt J, Sailer A-M, Bäumler K, Walters S, Fleischmann D. Aortic dissection maps: comprehensive visualization of aortic dissections for risk assessment. In: Eurographics Workshop on Visual Computing for Biology and Medicine, 2016. https://doi.org/10.2312/vcbm.20161282.
- [341] Mistelbauer G, Schmidt J, Sailer A-M, Walters S, Fleischmann D. Visualization of aortic diameter changes over time in an aortic surveillance program. In: Proceedings of the ECR Book of Abstracts – C - Scientific and Educational Exhibits, 2017, pp. C–1905. https://doi.org/10.1594/ecr2017/C-1905.
- [342] Bäumler K, Rolf-Pissarczyk M, Schussnig R, Mistelbauer G, Fries T-P, Marsden AL, Fleischmann D, Holzapfel GA. Assessment of aortic dissection remodeling with patient-specific fluid-structure interaction models. IEEE Trans Biomed Eng 2024.
- [343] Sugawara J, Hayashi K, Yokoi T, Tanaka H. Age-associated elongation of the ascending aorta in adults. JACC Cardiovasc Imaging 2008;1(6):739–48.
- [344] Vriz O, Driussi C, Bettio M, Ferrara F, D'Andrea A, Bossone E. Aortic root dimensions and stiffness in healthy subjects. Am J Cardiol 2013;112:1224–9.
   [345] Kamenskiy A, Miserlis D, Adamson P, Adamson M, Knowles T, Neme J, Koutakis P, Phillips N, Pipinos I, MacTaggart J. Patient demographics and
- cardiovascular risk factors differentially influence geometric remodeling of the aorta compared with the peripheral arteries. Surgery 2015;158(6):1617–27. [346] Weiss D, Cavinato C, Gray A, Ramachandra AB, Avril S, Humphrey JD, Latorre M. Mechanics-driven mechanobiological mechanisms of arterial tortuosity. Sci
- Adv 2020;6(49):eabd3574. https://doi.org/10.1126/sciadv.abd3574.
  [347] Sun L, Li J, Wang L, Li Q, He H, Li X, Li M, Wang T, Zhao C, Zhang X, Shu C. Aortic geometric alteration associated with acute type B aortic dissection: angulation, tortuosity, and arch type. Front Physiol 2021;12:708651.
- [348] Dotter CT, Roberts DJ, Steinberg I. Aortic length: angiocardiographic measurements. Circulation 1950;2(6):915–20.
- [349] Krüger T, Forkavets O, Veseli K, Lausberg H, Vöhringer L, Schneider W, Bamberg F, Schlensak C. Ascending aortic elongation and the risk of dissection. Eur J Cardiothorac Surg 2016;50(2):241–7.
- [350] Krüger T, Oikonomou A, Schibilsky D, Lescan M, Bregel K, Vöhringer L, Schneider W, Lausberg H, Blumenstock G, Bamberg F, Schlensak C. Aortic elongation and the risk for dissection: the Tübingen Aortic Pathoanatomy (TAIPAN) project. Eur J Cardiothorac Surg 2017;51(6):1119–26.

- [351] Krüger T, Boburg RS, Lescan M, Oikonomou A, Schneider W, Vöhringer L, Lausberg H, Bamberg F, Blumenstock G, Schlensak C. Aortic elongation in aortic aneurysm and dissection: the Tübingen Aortic Pathoanatomy (TAIPAN) project. Eur J Cardiothorac Surg 2018;54(1):26–33.
- [352] Lescan M, Veseli K, Oikonomou A, Walker T, Lausberg H, Blumenstock G, Bamberg F, Schlensak C, Krüger T. Aortic elongation and Stanford B dissection: the Tübingen Aortic Pathoanatomy (TAIPAN) project. Eur J Vasc Endovasc Surg 2017;54(2):164–9.
- [353] Adriaans BP, Heuts S, Gerretsen S, Cheriex EC, Vos R, Natour E, Maessen JG, Nia PS, Crijns HJGM, Wildberger JE, Schalla S. Aortic elongation part I: the normal aortic ageing process. Heart 2018;104:1772–7.
- [354] Heuts S, Adriaans BP, Gerretsen S, Natour E, Vos R, Cheriex EC, Crijns HJGM, Wildberger JE, Maessen JG, Schalla S, Sardari Nia P. Aortic elongation part II: the risk of acute type A aortic dissection. Heart 2018;104(21):1778–1782. https://doi.org/10.1136/heartjnl-2017-312867.
- [355] Wu J, Zafar MA, Li Y, Saeyeldin A, Huang Y, Zhao R, Qiu J, Tanweer M, Abdelbaky M, Gryaznov A, Buntin J, Ziganshin BA, Mukherjee SK, Rizzo JA, Yu C, Elefteriades JA. Ascending aortic length and risk of aortic adverse events: the neglected dimension. J Am Coll Cardiol 2019;74(15):1883–94.
- [356] Eliathamby D, Gutierrez M, Liu A, Ouzounian M, Forbes TL, Tan KT, Chung J. Ascending aortic length and its association with type A aortic dissection. J Am Heart Assoc 2021;10(13):e020140.
- [357] Sun L, Li X, Wang G, Sun J, Zhang X, Chi H, Cao H, Ma W, Yan Z, Liu G. Relationship between length and curvature of ascending aorta and type A dissection. Front Cardiovasc Med 2022;9:927105. https://doi.org/10.3389/fcvm.2022.927105.
- [358] Cao L, Lu W, Ge Y, Wang X, He Y, Sun G, Liu J, Liu X, Jia X, Xiong J, Ma X, Zhang H, Wang L, Guo W. Altered aortic arch geometry in patients with type B aortic dissection. Eur J Cardiothorac Surg 2020;58(4):714–21.
- [359] Shirali AS, Bischoff MS, Lin H-M, Öyfe I, Lookstein R, Griepp RB, Di Luozzo G. Predicting the risk for acute type B aortic dissection in hypertensive patients using anatomic variables. JACC Cardiovasc Imaging 2013;6(3):349–57.
- [360] Gode S, Akinci O, Ustunisik CT, Sen O, Kadirogullari E, Aksu T, Ersoy B, Gurbak I, Duman ZM, Erentug V. The role of the angle of the ascending aortic
- curvature on the development of type A aortic dissection: ascending aortic angulation and dissection. Interact Cardiovasc Thorac Surg 2019;29(4):615–20.
   [361] Franken R, El Morabit A, de Waard V, Timmermans J, Scholte AJ, van den Berg MP, Marquering H, Planken NR, Zwinderman AH, Mulder BJ, Groenink M. Increased aortic tortuosity indicates a more severe aortic phenotype in adults with Marfan syndrome. Int J Cardiol 2015;194:7–12.
- [362] Alhafez BA, Truong VTT, Ocazionez D, Sohrabi S, Sandhu H, Estrera A, Safi HJ, Evangelista A, Hurtado LD, Guala A, Prakash SK. Aortic arch tortuosity, a novel biomarker for thoracic aortic disease, is increased in adults with bicuspid aortic valve. Int J Cardiol 2019;284:84–9.
- [363] Chen J, Chen S, Zhang B, Liu J. Correlations between aortic tortuosity, diameter and presence of acute type A aortic dissection. J Cardiovasc Surg (Torino) 2021;62(6):632–8.
- [364] Rylski B, Blanke P, Beyersdorf F, Desai ND, Milewski RK, Siepe M, Kari FA, Czerny M, Carrel T, Schlensak C, Kruger T, Mack MJ, Brinkman WT, Mohr FW, Etz CD, Luehr M, Bavaria JE. How does the ascending aorta geometry change when it dissects? J Am Coll Cardiol 2014;63(13):1311–9.
- [365] Morris SA. Arterial tortuosity in genetic arteriopathies. Curr Opin Cardiol 2015;30(6):587–93.
- [366] O'Rourke M, Farnsworth A, O'Rourke J. Aortic dimensions and stiffness in normal adults. JACC Cardiovasc Imaging 2008;1(6):749-51.
- [367] Czerny M. Re: ascending aorta elongation and the risk of dissection. Eur J Cardiothorac Surg 2016;50(2):148.
- [368] Akin I, Nienaber CA. Age-dependent aortic elongation: a new predictor for type A aortic dissection? Heart 2018;104(21):1729–30.
- [369] Erdheim J. Medionecrosis aortae idiopathica. Virch Arch Pathol Anat 1929;273:454–79.
- [370] Erdheim J. Medionecrosis aortae idiopathica cystica. Virch Arch Pathol Anat 1930;276:187–229.
- [371] Babes V, Mironescu T. Über dissezierende Arteriitis und Aneurysma dissecans. Ziegler's Beitr z path Anat u z allg Path 1910;48:221.
- [372] Gsell O. Wandnekrosen der Aorta als selbständige Erkrankung und ihre Beziehung zur Spontanruptur. Virchows Arch path Anat 1928;270:1-36.
- [373] Hirst AE, Gore I. Editorial: is cystic medionecrosis the cause of dissecting aortic aneurysm? Circulation 1976;53(6):915-6.
- [374] Schlatmann TJM, Becker AE. Histologic changes in the normal aging aorta: implications for dissecting aortic aneurysm. Am J Cardiol 1977;39(1):13–20.
- [375] Schlatmann TJM, Becker AE. Pathogenesis of dissecting aneurysm of aorta: comparative histopathologic study of significance of medial changes. Am J Cardiol 1977;39(1):21–6.
- [376] Jain D, Dietz HC, Oswald GL, Maleszewski JJ, Halushka MK. Causes and histopathology of ascending aortic disease in childen and young adults. Cardiovasc Pathol 2011;20(1):15–25.
- [377] Homme J, Aubry M-C, Edwards W, Bagniewski S, Pankratz V, Kral C, Tazelaar H. Surgical pathology of the ascending aorta: a clinicopathologic study of 513 cases. Am J Surg Pathol 2006;30(9):1159–68.
- [378] Merendino KA, Winterscheid LC, Dillard DH. Cystic medial necrosis with and without Marfan's syndrome. Surgical experience with 20 patients and a note about the modified bicuspidization operation. Surg Clin North Am 1967;47(6):1403–18. https://doi.org/10.1016/s0039-6109(16)38390-6.
- [379] Nataatmadja M, West M, West J, Summers K, Walker P, Nagata M, Watanabe T. Abnormal extracellular matrix protein transport associated with increased apoptosis of vascular smooth muscle cells in Marfan syndrome and bicuspid aortic valve thoracic aortic aneurysm. Circulation 2003;108(Suppl 1):II329–34. https://doi.org/10.1161/01.cir.0000087660.82721.15.
- [380] Parfitt J, Chalmers RT, Wolfe JH. Visceral aneurysms in Ehlers-Danlos syndrome: case report and review of the literature. J Vasc Surg 2000;31(6):1248–51.
   [381] Fujiyama A, Nishiue T, Masaki M, Kurihara H, Iba O, Komatsu K, Maeda A, Senzaki H, Kawaguchi H, Imamura H, Iwasaka T. Aortic valvular vegetation in
- annuloaortic ectasia due to idiopathic cystic media necrosis: a case report. J Cardiol 1999;32(5):331–6.
- [382] McKusick VA, Logue RB, Bahnson HT. Association of aortic valvular disease and cystic medial necrosis of the ascending aorta; report of four instances. Circulation 1957;16(2):188–94.
- [383] Marsalese DL, Moodie DS, Lytle BW, Cosgrove DM, Ratliff NB, Goormastic M, Kovacs A. Cystic medial necrosis of the aorta in patients without Marfan's syndrome: surgical outcome and long-term follow-up. J Am Coll Cardiol 1990;16(1):67–73.
- [384] Isner JM, Donaldson RF, Fulton D, Bhan I, Payne DD, Cleveland RJ. Cystic medial necrosis in coarctation of the aorta: a potential factor contributing to adverse consequences observed after percutaneous balloon angioplasty of coarctation sites. Circulation 1987;75(4):689–95.
- [385] Lindsay Jr J. Coarctation of the aorta, bicuspid aortic valve and abnormal ascending aortic wall. Am J Cardiol 1988;61(1):182-4.
- [386] Pachulski RT, Weinberg AL, Chan KL. Aortic aneurysm in patients with functionally normal or minimally stenotic bicuspid aortic valve. Am J Cardiol 1991;67 (8):781–2.
- [387] Agozzino L, Santè P, Ferraraccio F, Accardo M, de Feo M, de Santo LS, Nappi G, Agozzino M, Esposito S. Ascending aorta dilatation in aortic valve disease: morphological analysis of medial changes. Heart Vessels 2006;21(4):213–20.
- [388] De Sa M, Moshkovitz Y, Butany J, David TE. Histologic abnormalities of the ascending aorta and pulmonary trunk in patients with bicuspid aortic valve disease: clinical relevance to the ross procedure. J Thorac Cardiovasc Surg 1999;118(4):588–94.
- [389] Guntheroth WG. A critical review of the American College of Cardiology/American Heart Association practice guidelines on bicuspid aortic valve with dilated ascending aorta. Am J Cardiol 2008;102(1):107–10.
- [390] Yuan S-M, Jing H. Cystic medial necrosis: pathological findings and clinical implications. Rev Bras Cir Cardiovasc 2011;26(1):107–15.
- [391] Sariola H, Viljanen T, Luosto R. Histological pattern and changes in extracellular matrix in aortic dissections. J Clin Pathol 1986;39(10):1074-81.
- [392] Nakashima Y, Kurozumi T, Sueishi K, Tanaka K. Dissecting aneurysm: a clinicopathologic and histopathologic study of 111 autopsied cases. Hum Pathol 1990; 21(3):291–6.
- [393] Nakashima Y, Shiokawa Y, Sueishi K. Alterations of elastic architecture in human aortic dissecting aneurysm. Lab Invest 1990;62(6):751-60.
- [394] Nakashima Y, Sueishi K. Alteration of elastic architecture in the lathyritic rat aorta implies the pathogenesis of aortic dissecting aneurysm. Am J Pathol 1992; 140(4):959–69.
- [395] Cattell MA, Hasleton PS, Anderson JC. Increased elastin content and decreased elastin concentration may be predisposing factors in dissecting aneurysms of human thoracic aorta. Cardiovasc Res 1993;27(2):176–81.
- [396] Watanabe M, Sawai T. Alteration of cross-linking amino acids of elastin in human aorta in association with dissecting aneurysm: analysis using high performance liquid chromatography. Tohoku J Exp Med 1999;187(4):291–303.

- [397] Nakashima Y. Pathophysiology of aortic dissection and abdominal aortic aneurysm. In: Kawada S, Ueda T, Shimizu H, editors. Cardio-aortic and Aortic Surgery. Tokyo: Springer Japan; 2001. p. 4–11.
- [398] Koullias GJ, Ravichandran P, Korkolis DP, Rimm DL, Elefteriades JA. Increased tissue microarray matrix metalloproteinase expression favors proteolysis in thoracic aortic aneurysms and dissections. Ann Thorac Surg 2004;78(6):2106–10. discussion 2110–1.
- [399] Wang X, LeMaire SA, Chen L, Carter SA, Shen YH, Gan Y, Bartsch H, Wilks JA, Utama B, Ou H, Thompson RW, Coselli JS, Wang XL. Decreased expression of fibulin-5 correlates with reduced elastin in thoracic aortic dissection. Surgery 2005;138(2):352–9.
- [400] Nakashima Y. Pathogenesis of aortic dissection: elastic fiber abnormalities and aortic medial weakness. Ann Vasc Dis 2010;3(1):28-36.
- [401] Mimler T, Nebert C, Eichmair E, Winter B, Aschacher T, Stelzmueller ME, Andreas M, Ehrlich M, Laufer G, Messner B. Extracellular matrix in ascending aortic aneurysms and dissections – what we learn from decellularization and scanning electron microscopy. PloS One 2019;14(3):e0213794.
- [402] Barsky SH, Rosen S. Aortic infarction following dissecting aortic aneurysm. Circulation 1978;58(5):876–81.
- [403] Wilson SK, Hutchins GM. Aortic dissecting aneurysms: causative factors in 204 subjects. Arch Pathol Lab Med 1982;106(4):175-80.
- [404] Larson EW, Edwards WD. Risk factors for aortic dissection: a necropsy study of 161 cases. Am J Cardiol 1984;53(6):849-55.
- [405] Rottino A. Medial degeneration of the aorta: a study of 210 routine autopsy specimens by a serial block method. Arch Pathol Lab Med 1939;28:377-85.
- [406] Coady MA, Rizzo JA, Goldstein LJ, Elefteriades JA. Natural history, pathogenesis, and etiology of thoracic aortic aneurysms and dissections. Cardiol Clin 1999; 17(4):615–35;vii. https://doi.org/10.1016/s0733-8651(05)70105-3.
- [407] Wang X, LeMaire SA, Chen L, Shen YH, Gan Y, Bartsch H, Carter SA, Utama B, Ou H, Coselli JS, Wang XL. Increased collagen deposition and elevated expression of connective tissue growth factor in human thoracic aortic dissection. Circulation 2006;114(1 Suppl):1200–5.
- [408] de Figueiredo Borges L, Jaldin RAG, Dias RR, Stolf NAG, Michel J-B, Gutierrez PS. Collagen is reduced and disrupted in human aneurysms and dissections of ascending aorta. Hum Pathol 2008;39(3):437–43.
- [409] Cheuk BLY, Cheng SWK. Differential expression of elastin assembly genes in patients with Stanford type A aortic dissection using microarray analysis. J Vasc Surg 2011;53(4):1071–1078.e2.
- [410] Wang Y, Ning J, Johnson J, Sutton M, Lessner S. Development of a quantitative mechanical test of atherosclerotic plaque stability. J Biomech 2011;44: 2439–45.
- [411] de Figueiredo Borges L, Blini JPF, Dias RR, Gutierrez PS. Why do aortas cleave or dilate? Clues from an electronic scanning microscopy study in human ascending aortas. J Vasc Res 2014;51(1):50–7.
- [412] Cattell MA, Hasleton PS, Anderson JC. Glycosaminoglycan content is increased in dissecting aneurysms of human thoracic aorta. Clin Chim Acta 1994;226(1): 29–46.
- [413] Fedak PW, Verma S, David TE, Leask RL, Weisel RD, Butany J. Clinical and pathophysiological implications of a bicuspid aortic valve. Circulation 2002;106(8): 900–4.
- [414] Nataatmadja M, West J, West M. Overexpression of transforming growth factor-beta is associated with increased hyaluronan content and impairment of repair in Marfan syndrome aortic aneurysm. Circulation 2006;114(1 Suppl):I371–7.
- [415] Collins MJ, Dev V, Strauss BH, Fedak PWM, Butany J. Variation in the histopathological features of patients with ascending aortic aneurysms: a study of 111 surgically excised cases. J Clin Pathol 2008;61(4):519–23.
- [416] Milewicz DM, Guo D-C, Tran-Fadulu V, Lafont AL, Papke CL, Inamoto S, Kwartler CS, Pannu H. Genetic basis of thoracic aortic aneurysms and dissections: focus on smooth muscle cell contractile dysfunction. Annu Rev Genomics Hum Genet 2008;9:283–302.
- [417] de Figueiredo Borges L, Touat Z, Leclercq A, Al Haj Zen A, Jondeau' G, Franc B, Philippe M, Meilhac O, Gutierrez PS, Michel J-B. Tissue diffusion and retention of metalloproteinases in ascending aortic aneurysms and dissections. Hum Pathol 2009;40(3):306–13.
- [418] Maleszewski JJ, Miller DV, Lu J, Dietz HC, Halushka MK. Histopathologic findings in ascending aortas from individuals with Loeys-Dietz syndrome (LDS). Am J Surg Pathol 2009;33(2):194–201.
- [419] Grond-Ginsbach C, Pjontek R, Aksay SS, Hyhlik-Dürr A, Böckler D, Gross-Weissmann M-L. Spontaneous arterial dissection: phenotype and molecular pathogenesis. Cell Mol Life Sci 2010;67(11):1799–815.
- [420] Guo D-C, Regalado ES, Minn C, Tran-Fadulu V, Coney J, Cao J, Wang M, Yu RK, Estrera AL, Safi HJ, Shete SS, Milewicz DM. Familial thoracic aortic aneurysms and dissections. Circ Cardiovasc Genet 2011;4(1):36–42.
- [421] Bode-Jänisch S, Schmidt A, Günther D, Stuhrmann M, Fieguth A. Aortic dissecting aneurysms-histopathological findings. Forensic Sci Int 2012;214(1–3):13–7.
- [422] Guzzardi DG, Barker AJ, van Ooij P, Malaisrie SC, Puthumana JJ, Belke DD, Mewhort HE, Svystonyuk DA, Kang S, Verma S, Collins J, Carr J, Bonow RO, Markl M, Thomas JD, McCarthy PM, Fedak PW. Valve-related hemodynamics mediate human bicuspid aortopathy: insights from wall shear stress mapping. J Am Coll Cardiol 2015;66(8):892–900.
- [423] Levine RA, Hagège AA, Judge DP, Padala M, Dal-Bianco JP, Aikawa E, Beaudoin J, Bischoff J, Bouatia-Naji N, Bruneval P, Butcher JT, Carpentier A, Chaput M, Chester AH, Clusel C, Delling FN, Dietz HC, Dina C, Durst R, Fernandez-Friera L, Handschumacher MD, Jensen MO, Jeunemaitre XP, Le Marce H, Le Tourneau T, Markwald RR, Mérot J, Messas E, Milan DP, Neri T, Norris RA, Peal D, Perrocheau M, Probst V, Pucéat M, Rosenthal N, Solis J, Schott J-J, Schwammenthal E, Slaugenhaupt SA, Song J-K, Yacoub MH, Leducq Mitral Transatlantic Network. Mitral valve disease-morphology and mechanisms. Nat Rev Cardiol 2015;12 (12):689–710. https://doi.org/10.1038/nrcardio.2015.161.
- [424] Bian L, Angione SL, Ng KW, Lima EG, Williams DY, Mao DQ, Ateshian GA, Hung CT. Influence of decreasing nutrient path length on the development of engineered cartilage. Osteoarthritis Cartilage 2009;17(5):677–85.
- [425] Maroudas A. Balance between swelling pressure and collagen tension in normal and degenerate cartilage. Nature 1976;260(5554):808-9.
- [426] Chahine NO, Chen FH, Hung CT, Ateshian GA. Direct measurement of osmotic pressure of glycosaminoglycan solutions by membrane osmometry at room temperature. Biophys J 2005;89(3):1543–50.
- [427] Hosoda Y, Iri H. Angiolathyrism. 1. A histological and histochemical study on successive changes of the lathyritic rat aorta. Acta Pathol Jpn 1966;16(3): 239–52.
- [428] Evanko SP, Johnson PY, Braun KR, Underhill CB, Dudhia J, Wight TN. Platelet-derived growth factor stimulates the formation of versican–hyaluronan aggregates and pericellular matrix expansion in arterial smooth muscle cells. Arch Biochem Biophys 2001;394(1):29–38.
- [429] Bunton TE, Biery NJ, Myers L, Gayraud B, Ramirez F, Dietz HC. Phenotypic alteration of vascular smooth muscle cells precedes elastolysis in a mouse model of Marfan syndrome. Circ Res 2001;88(1):37–43.
- [430] Hornebeck W, Emonard H. The cell-elastin-elastase(s) interacting triade directs elastolysis. Front Biosci (Landmark Ed) 2011;16(2):707–22.
  [431] Rai P, Robinson L, Davies HA, Akhtar R, Field M, Madine J. Is there enough evidence to support the role of glycosaminoglycans and proteoglycans in thoracic
- aortic aneurysm and dissection? A systematic review. Int J Mol Sci 2022;23(16):9200.
- [432] van Baardwijk C, Roach MR. Factors in the propagation of aortic dissections in canine thoracic aortas. J Biomech 1987;20(1):67-73.
- [433] Sommer G, Gasser TC, Regitnig P, Auer M, Holzapfel GA. Dissection properties of the human aortic media: an experimental study. J Biomech Eng 2008;130(2): 021007.
- [434] Boyle EC, Sedding DG, Haverich A. Targeting vasa vasorum dysfunction to prevent atherosclerosis. Vascul Pharmacol 2017;96–98:5–10.
- [435] Haverich A, Boyle EC. Incriminating evidence for the role of the microvasculature in atherosclerosis. In: Atherosclerosis Pathogenesis and Microvascular Dysfunction. Cham, Switzerland: Springer Nature; 2019. p. 55–74.
- [436] Stefanadis C, Vlachopoulos C, Karayannacos P, Boudoulas H, Stratos C, Filippides T, Agapitos M, Toutouzas P. Effect of vasa vasorum flow on structure and function of the aorta in experimental animals. Circulation 1995;91(10):1669–78.
- [437] Angouras DC, Sokolis DP, Dosios T, Kostomitsopoulos N, Boudoulas H, Skalkeas G, Karayannacos PE. Effect of impaired vasa vasorum flow on the structure and mechanics of the thoracic aorta: implications for the pathogenesis of aortic dissection. Eur J Cardiothorac Surg 2000;17(4):468–73.
- [438] Haverich A, Boyle EC. Aortic dissection is a disease of the vasa vasorum. JTCVS Open 2021;5:30–2.
- [439] Osada H, Kyogoku M, Ishidou M, Morishima M, Nakajima H. Aortic dissection in the outer third of the media: what is the role of the vasa vasorum in the triggering process? Eur J Cardiothorac Surg 2013;43(3):e82–8.

- [440] Gornik HL, Creager MA. Aortitis. Circulation 2009;117(23):3039–51.
- [441] Sörelius K, Budtz-Lilly J, Mani K, Wanhainen A. Systematic review of the management of mycotic aortic aneurysms. Eur J Vasc Endovasc Surg 2019;58(3): 426–35.
- [442] Adlam D, Alfonso F, Maas A, Vrints C. Writing Committee, European Society of Cardiology, acute cardiovascular care association, SCAD study group: a position paper on spontaneous coronary artery dissection. Eur Heart J 2018;39(36):3353–68.
- [443] Hayes SN, Tweet MS, Adlam D, Kim ESH, Gulati R, Price JE, Rose CH. Spontaneous coronary artery dissection: JACC state-of-the-art review. J Am Coll Cardiol 2020;76(8):961–84.
- [444] Schwartz SM, deBlois D, O'Brien ER. The intima. Soil for atherosclerosis and restenosis. Circ Res 1995;77(3):445-65.
- [445] Minami H, Sugimoto T, Okada M. Evaluation of acute aortic dissection by cine-MRI. Kobe J Med Sci 1999;45(1):1-11.
- [446] Karmonik C, Müller-Eschner M, Partovi S, Geisbüsch P, Ganten MK, Bismuth J, Davies MG, Böckler D, Loebe M, Lumsden AB, von Tengg-Kobligk H.
- Computational fluid dynamics investigation of chronic aortic dissection hemodynamics versus normal aorta. Vasc Endovascular Surg 2013;47(8):625–31.
   [447] Karmonik C, Duran C, Shah DJ, Anaya-Ayala JE, Davies MG, Lumsden AB, Bismuth J. Preliminary findings in quantification of changes in septal motion during follow-up of type B aortic dissections. J Vasc Surg 2012;55(5):1419–26.
- [448] Bäumler K, Vedula V, Sailer AM, Seo J, Chiu P, Mistelbauer G, Chan FP, Fischbein MP, Marsden AL, Fleischmann D. Fluid-structure interaction simulations of patient-specific aortic dissection. Biomech Model Mechanobiol 2020;19:1607–28.
- [449] Panpho P, Yang Y, Davies HA, Nawaytou O, Harky A, Torella F, Field M, Madine J, Akhtar R. Time-dependent mechanical behaviour of the aortic chronic dissection flap. Interact Cardiovasc Thorac Surg 2022;34:892–901.
- [450] Rolf-Pissarczyk M, Wollner MP, Pacheco DRQ, Holzapfel GA. Efficient computational modelling of smooth muscle orientation and function in the aorta. Proc R Soc A 2021;477:20210592.
- [451] Franchini G, Breslavsky ID, Giovanniello F, Kassab A, Holzapfel GA, Amabili M. Role of smooth muscle activation in the static and dynamic mechanical characterization of human aortas. Proc Natl Acad Sci USA 2022;119(3). e2117232119.
- [452] Yamada T, Sakata N, Wada H, Tashiro T, Tayama E. Age-related distensibility and histology of the ascending aorta in elderly patients with acute aortic dissection. J Biomech 2015;48(12):3267–73.
- [453] Babu AR, Byju AG, Gundiah N. Biomechanical properties of human ascending thoracic aortic dissections. J Biomech Eng 2015;137(8):081013.
- [454] Sommer G, Sherifova S, Oberwalder PJ, Dapunt OE, Ursomanno PA, DeAnda A, Griffith BE, Holzapfel GA. Mechanical strength of aneurysmatic and dissected human thoracic aortas at different shear loading modes. J Biomech 2016;49(12):2374–82.
- [455] Ahuja A, Noblet JN, Trudnowski T, Patel B, Krieger JF, Chambers S, Kassab GS. Biomechanical material characterization of Stanford type-B dissected porcine aortas. Front Physiol 2018;9:1317.
- [456] Manopoulos C, Karathanasis I, Kouerinis I, Angouras DC, Lazaris A, Tsangaris S, Sokolis DP. Identification of regional/layer differences in failure properties and thickness as important biomechanical factors responsible for the initiation of aortic dissections. J Biomech 2018;80:102–10.
- [457] Deplano V, Boufi M, Gariboldi V, Loundou A, D'Journo XB, Cautela J, Djemli A, Alimi YS. Mechanical characterisation of human ascending aorta dissection. J Biomech 2019;94:138–46.
- [458] Kan X, Ma T, Lin J, Wand L, Dong Z, Xu XY. Patient-specific simulation of stent-graft deployment in type B aortic dissection: model development and validation. Biomech Model Mechanobiol 2021;20:2247–58.
- [459] Li Z, Luo T, Wang S, Jia H, Gong Q, Liu X, Sutcliffe MP, Zhu H, Liu Q, Chen D, Xiong J, Teng Z. Mechanical and histological characteristics of aortic dissection tissues. Acta Biomater 2022;146:284–94.
- [460] Guo Y, Yu H, Wang L, Zhai Y, Li J, Tang D, Sun H. Layer-specific biomechanical and histological properties of normal and dissected human ascending aortas. Heliyon 2024;10(14):e34646. https://doi.org/10.1016/j.heliyon.2024.e34646.
- [461] Brunet J, Pierrat B, Maire E, Adrien J, Badel P. A combined experimental-numerical lamellar-scale approach of tensile rupture in arterial medial tissue using Xray tomography. J Mech Behav Biomed Mater 2019;95:116–23.
- [462] Bhat SK, Yamada H. Mechanical characterization of dissected and dilated human ascending aorta using Fung-type hyperelastic models with pre-identified initial tangent moduli for low-stress distensibility. J Mech Behav Biomed Mater 2022;125:104959.
- [463] Trachet B, Fraga-Silva RA, Piersigilli A, Tedgui A, Sordet-Dessimoz J, Astolfo A, Van der Donckt C, Modregger P, Stampanoni MFM, Segers P, Stergiopulos N. Dissecting abdominal aortic aneurysm in ang II-infused mice: suprarenal branch ruptures and apparent luminal dilatation. Cardiovasc Res 2014;105(2): 213–22.
- [464] Trachet B, Fraga-Silva RA, Jacquet PA, Stergiopulos N, Segers P. Incidence, severity, mortality, and confounding factors for dissecting AAA detection in angiotensin II-infused mice: a meta-analysis. Cardiovasc Res 2015;108(1):159–70.
- [465] Trachet B, Fraga-Silva R, Piersigilli A, Segers P, Stergiopulos N. Dissecting abdominal aortic aneurysm in angiotensin II-infused mice: the importance of imaging. Curr Pharm Des 2015;21(28):4049–60.
- [466] Trachet B, Piersigilli A, Fraga-Silva RA, Aslanidou L, Sordet-Dessimoz J, Astolfo A, Stampanoni MFM, Segers P, Stergiopulos N. Ascending aortic aneurysm in angiotensin II-infused mice: formation, progression, and the role of focal dissections. Arterioscler Thromb Vasc Biol 2016;36(4):673–81. https://doi.org/10. 1161/ATVBAHA.116.307211.
- [467] Trachet B, Aslanidou L, Piersigilli A, Fraga-Silva RA, Sordet-Dessimoz J, Villanueva-Perez P, Stampanoni MF, Stergiopulos N, Segers P. Angiotensin II infusion into ApoE<sup>-/-</sup> mice: a model for aortic dissection rather than abdominal aortic aneurysm? Cardiovasc Res 2017;113(10):1230–42.
- [468] Bersi MR, Bellini C, Di Achille P, Humphrey JD, Genovese K, Avril S. Novel methodology for characterizing regional variations in the material properties of murine aortas. J Biomech Eng 2016;138(7):0710051–07100515.
- [469] Phillips EH, Di Achille P, Bersi MR, Humphrey JD, Goergen CJ. Multi-modality imaging enables detailed hemodynamic simulations in dissecting aneurysms in mice. IEEE Trans Med Imaging 2017;36(6):1297–305.
- [470] Bersi MR, Acosta Santamaría V, Marback K, Di Achille P, Phillips EH, Goergen CJ, Humphrey JD, Avril S. Multimodality imaging-based characterization of regional material properties in a murine model of aortic dissection. Sci Rep 2020;10:9244.
- [471] Ito S, Lu HS, Daugherty A, Sawada H. Imaging techniques for aortic aneurysms and dissections in mice: comparisons of ex vivo, in situ, and ultrasound approaches. Biomolecules 2022;12(2):339.
- [472] Weiss D, Long AS, Tellides G, Avril S, Humphrey JD, Bersi MR. Evolving mural defects, dilatation, and biomechanical dysfunction in angiotensin II-induced thoracic aortopathies. Arterioscler Thromb Vasc Biol 2022;42(8):973–86.
- [473] Bellini C, Bersi MR, Caulk AW, Ferruzzi J, Milewicz DM, Ramirez F, Rifkin DB, Tellides G, Yanagisawa H, Humphrey JD. Comparison of 10 murine models reveals a distinct biomechanical phenotype in thoracic aortic aneurysms. J R Soc Interface 2017;14(14):20161036.
- [474] Ferruzzi J, Murtada S-I, Li G, Jiao Y, Uman S, Ting MYL, Tellides G, Humphrey JD. Pharmacologically improved contractility protects against aortic dissection in mice with disrupted transforming growth factor-β signaling despite compromised extracellular matrix properties. Arterioscler Thromb Vasc Biol 2016;36(5): 919–27.
- [475] Fuster V, Narula J, Vaishnava P, Leon MB, Callans DJ, Rumsfeld J, Poppas A, editors. Fuster and Hurst's The Heart. 15th Edition. McGraw Hill; 2022.
- [476] Leone O, Corsini A, Pacini D, Corti B, Lorenzini M, Laus V, Foà A, Bacchi Reggiani ML, Di Marco L, Rapezzi C. The complex interplay among atherosclerosis, inflammation, and degeneration in ascending thoracic aortic aneurysms. J Thorac Cardiovasc Surg 2020;160(6):1434–43.
- [477] Hirose H, Tilson MD. Abdominal aortic aneurysm as an autoimmune disease. Ann N Y Acad Sci 2001;947:416–8.
- [478] Golledge AL, Walker P, Norman PE, Golledge J. A systematic review of studies examining inflammation associated cytokines in human abdominal aortic aneurysm samples. Dis Markers 2009;26(4):181–8.
- [479] Jalalzadeh H, Indrakusuma R, Planken RN, Legemate DA, Koelemay MJ, Balm R. Inflammation as a predictor of abdominal aortic aneurysm growth and rupture: a systematic review of imaging biomarkers. Eur J Vasc Endovasc Surg 2016;52(3):333–42.
- [480] Pisano C, Balistreri CR, Ricasoli A, Ruvolo G. Cardiovascular disease in ageing: an overview on thoracic aortic aneurysm as an emerging inflammatory disease. Mediators Inflamm 2017;2017:1274034.

- [481] Tang PC, Yakimov AO, Teesdale MA, Coady MA, Dardik A, Elefteriades JA, Tellides G. Transmural inflammation by interferon-gamma-producing T cells correlates with outward vascular remodeling and intimal expansion of ascending thoracic aortic aneurysms. FASER J 2005;19(11):1528–30.
- [482] Guo D-C, Papke CL, He R, Milewicz DM. Pathogenesis of thoracic and abdominal aortic aneurysms. Ann N Y Acad Sci 2006;1085:339–52.
- [483] He R, Guo D-C, Estrera AL, Safi HJ, Huynh TT, Yin Z, Cao S-N, Lin J, Kurian T, Buja LM, Geng Y-J, Milewicz DM. Characterization of the inflammatory and apoptotic cells in the aortas of patients with ascending thoracic aortic aneurysms and dissections. J Thorac Cardiovasc Surg 2006;131(3):671–8.
- [484] Wang Y, Barbacioru CC, Shiffman D, Balasubramanian S, Iakoubova O, Tranquilli M, Albornoz G, Blake J, Mehmet NN, Ngadimo D, Poulter K, Chan F, Samaha RR, Elefteriades JA. Gene expression signature in peripheral blood detects thoracic aortic aneurysm. PLoS One 2007;2(10):e1050.
- [485] Postnov A, Suslov A, Sobenin I, Chairkin I, Sukhorukov V, Ekta MB, Khotina V, Afanasiev M, Chumachenko P, Orekhov A. Thoracic aortic aneurysm: blood pressure and inflammation as key factors in the development of aneurysm dissection. Curr Pharm Des 2021;27(28):3122–7.
- [486] Spronck B, Latorre M, Wang M, Mehta S, Caulk AW, Ren P, Ramachandra AB, Murtada SI, Rojas A, He CS, Jiang B, Bersi MR, Tellides G, Humphrey JD. Excessive adventitial stress drives inflammation-mediated fibrosis in hypertensive aortic remodelling in mice. J R Soc Interface 2021;18(180):20210336.
- [487] Liu H, Li H, Han L, Zhang Y, Wu Y, Hong L, Yang J, Zhong J, Wang Y, Wu D, Fan G, Chen J, Zhang S, Peng X, Zeng Z, Tang Z, Lu Z, Sun L, Qian S, Shao Y, Zhang H. Additive Anti-inflammatory Action for Aortopathy & Arteriopathy, Inflammatory risk stratification individualizes anti-inflammatory pharmacotherapy for acute type A aortic dissection. Innovation (Camb) 2023;4(4):100448.
- [488] Menichini C, Cheng Z, Gibbs RGJ, Xu XY. Predicting false lumen thrombosis in patient-specific models of aortic dissection. J R Soc Interface 2016;13(124). https://doi.org/10.1098/rsif.2016.0759.
- [489] Yazdani A, Li H, Bersi M, Achille P, Insley J, Humphrey JD, Karniadakis G. Data-driven modeling of hemodynamics and its role on thrombus size and shape in aortic dissections. Sci Rep 2018;8(1):2515.
- [490] Michel J-B, Jondeau G, Milewicz DM. From genetics to response to injury: vascular smooth muscle cells in aneurysms and dissections of the ascending aorta. Cardiovasc Res 2018;114(4):578–89.
- [491] Raffetto JD, Khalil RA. Matrix metalloproteinases and their inhibitors in vascular remodeling and vascular disease. Biochem Pharmacol 2008;75(2):346–59.
- [492] Barbour JR, Spinale FG, Ikonomidis JS. Proteinase systems and thoracic aortic aneurysm progression. J Surg Res 2007;139(2):292–307.
- [493] Zhang X, Shen Y, LeMaire S. Thoracic aortic dissection: are matrix metalloproteinases involved? Vascular 2009;17(3):147-57.
- [494] Verstraeten A, Luyckx I, Loeys B. Aetiology and management of hereditary aortopathy. Nat Rev Cardiol 2017;14(4):197–208.
- [495] Stanton H, Melrose J, Little CB, Fosang AJ. Proteoglycan degradation by the ADAMTS family of proteinases. Biochim Biophys Acta 2011;1812(12):1616–29.
- [496] Ren P, Zhang L, Xu G, Palmero LC, Albini PT, Coselli JS, Shen YH, LeMaire SA. ADAMTS-1 and ADAMTS-4 levels are elevated in thoracic aortic aneurysms and dissections. Ann Thorac Surg 2013;95(2):570–7.
- [497] Kemberi M, Salmasi Y, Santamaria S. The role of ADAMTS proteoglycanases in thoracic aortic disease. Int J Mol Sci 2023;24(15):12135.
- [498] Humphrey JD, Schwartz MA. Vascular mechanobiology: homeostasis, adaptation, and disease. Annu Rev Biomed Eng 2021;23:1–27.
- [499] Chang JM, Friese K, Caputo GR, Kondo C, Higgins CB. MR measurement of blood flow in the true and false channel in chronic aortic dissection. J Comput Assist Tomogr 1991;15(3):418–23.
- [500] Adriaans BP, Wildberger JE, Westenberg JJM, Lamb HJ, Schalla S. Predictive imaging for thoracic aortic dissection and rupture: moving beyond diameters. Eur Radiol 2019;29(21):6396–404.
- [501] Mohri M, Nagata Y, Hisano R, Koyanagi S, Hirata T, Nakamura M. Detection of different blood flow patterns in the true and false lumina with aortic root dissection by pulsed Doppler echocardiography. Clin Cardiol 1985;8(4):225–7.
- [502] Bogren HG, Underwood SR, Firmin DN, Mohiaddin RH, Klipstein RH, Rees RS, Longmore DB. Magnetic resonance velocity mapping in aortic dissection. Br J Radiol 1988;61(726):456–62.
- [503] Mitchell L, Jenkins JP, Brownlee WC, Isherwood I. Aortic dissection: morphology and differential flow velocity patterns demonstrated by magnetic resonance imaging. Clin Radiol 1988;39(6):458–61.
- [504] Mohr-Kahaly S, Erbel R, Rennollet H, Wittlich N, Drexler M, Oelert H, Meyer J. Ambulatory follow-up of aortic dissection by transesophageal two-dimensional and color-coded Doppler echocardiography. Circulation 1989;80(1):24–33.
- [505] Strotzer M, Aebert H, Lenhart M, Nitz W, Wild T, Manke C, Völk M, Feuerbach S. Morphology and hemodynamics in dissection of the descending aorta. Assessment with MR imaging, Acta Radiol 2000;41(6):594–600.
- [506] Markl M, Draney MT, Hope MD, Levin JM, Chan FP, Alley MT, Pelc NJ, Herfkens RJ. Time-resolved 3-dimensional velocity mapping in the thoracic aorta: visualization of 3-directional blood flow patterns in healthy volunteers and patients. J Comput Assist Tomogr 2004;28(4):459–68.
- [507] Sherrah AG, Callaghan FM, Puranik R, Jeremy RW, Bannon PG, Vallely MP, Grieve SM. Multi-velocity encoding four-dimensional flow magnetic resonance imaging in the assessment of chronic aortic dissection. Aorta (Stamford) 2017;5(3):80–90.
- [508] Liu D, Fan Z, Li Y, Zhang N, Sun Z, An J, Stalder AF, Greiser A, Liu J. Quantitative study of abdominal blood flow patterns in patients with aortic dissection by 4-dimensional flow MRI. Sci Rep 2018;8(1):9111.
- [509] Allen BD, Aouad PJ, Burris NS, Rahsepar AA, Jarvis KB, François CJ, Barker AJ, Malaisrie SC, Carr JC, Collins JD, Markl M. Detection and hemodynamic evaluation of flap fenestrations in type B aortic dissection with 4D flow MRI: comparison with conventional MRI and CTA. Radiol Cardiothorac Imaging 2019;1 (1):e180009.
- [510] Takei Y, Itatani K, Miyazaki S, Shibasaki I, Fukuda H. Four-dimensional flow magnetic resonance imaging analysis before and after thoracic endovascular aortic repair of chronic type B aortic dissection. Interact Cardiovasc Thorac Surg 2019;28(3):413–20.
- [511] Cosset B, Boussel L, Davila Serrano E, Millon A, Douek P, Farhat F, Sigovan M. Hemodynamic changes before and after endovascular treatment of type B aortic dissection by 4D flow MRI. Front Cardiovasc Med 2022;9:873144. https://doi.org/10.3389/fcvm.2022.873144.
- [512] Inoue T, Watanabe S, Masuda Y, Yoshida K, Imai H, Imazeki Y, Ishizuya Y, Koga M, Kato H, Ikehira H, Tateno Y. Evaluation of blood flow patterns of true and false lumens in dissecting aneurysms using MR phase-contrast techniques. Clin Imaging 1996;20(4):262–8.
- [513] Inoue T, Watanabe S, Sakurada H, Ono K, Urano M, Hijikata Y, Saito I, Masuda Y. Evaluation of flow volume and flow patterns in the patent false lumen of chronic aortic dissections using velocity-encoded cine magnetic resonance imaging. Jpn Circ J 2000;64(10):760–4.
- [514] Burris NS, Patel HJ, Hope MD. Retrograde flow in the false lumen: marker of a false lumen under stress? J Thorac Cardiovasc Surg 2019;157(488-491):2.
- [515] Tsai TT, Evangelista A, Nienaber CA, Myrmel T, Meinhardt G, Cooper JV, Smith DE, Suzuki T, Fattori R, Llovet A, Froehlich J, Hutchison S, Distante A, Sundt T, Beckman J, Januzzi JL, Isselbacher EM, Eagle KA. International Registry of Acute Aortic Dissection. Partial thrombosis of the false lumen in patients with acute type B aortic dissection, N Engl J Med 2007;357(4):349–59.
- [516] Rudenick PA, Segers P, Pineda V, Cuellar H, García-Dorado D, Evangelista A, Bijnens BH. False lumen flow patterns and their relation with morphological and biomechanical characteristics of chronic aortic dissections. Computational model compared with magnetic resonance imaging measurements. PloS One 2017; 12(1):e0170888. https://doi.org/10.1371/journal.pone.0170888.
- [517] Marlevi D, Sotelo JA, Grogan-Kaylor R, Ahmed Y, Uribe S, Patel HJ, Edelman HR, Nordsletten DA, Burris NS. False lumen pressure estimation in type B aortic dissection using 4D flow cardiovascular magnetic resonance: comparisons with aortic growth, J Cardiovasc Magn Reson 2021;23(1):51. https://doi.org/10. 1038/s41598-018-37714-0.
- [518] Marlevi D, Ruijsink B, Balmus M, Dillon-Murphy D, Fovargue D, Pushparajah K, Bertoglio C, Colarieti-Tosti M, Larsson M, Lamata P, Figueroa CA, Razavi R, Nordsletten DA. Estimation of cardiovascular relative pressure using virtual work-energy. Sci Rep 2019;9(1):1375.
- [519] Chu S, Kilinc O, Pradella M, Weiss E, Baraboo J, Maroun A, Jarvis K, Mehta CK, Malaisrie SC, Hoel AW, Carr JC, Markl M, Allen BD. Baseline 4D flow-derived in vivo hemodynamic parameters stratify descending aortic dissection patients with enlarging aortas. Front Cardiovasc Med 2022;9:905718.
- [520] Ruiz-Muñoz A, Guala A, Dux-Santoy L, Teixidó-Turà G, Valente F, Garrido-Oliver J, Galian-Gay L, Gutiérrez L, Fernandez-Galera R, Casas-Masnou G, González-Alujas T, Cuéllar-Calabria H, Carrasco-Poves A, Morales-Galán A, Johnson KM, Wieben O, Ferreira-González I, Evangelista A, Rodriguez-Palomares J. False lumen hemodynamics and partial thrombosis in chronic aortic dissection of the descending aorta. Eur Radiol 2024;34(8):5190–200.
- [521] Amano Y, Sekine T, Suzuki Y, Tanaka K, Takagi R, Kumita S. Time-resolved three-dimensional magnetic resonance velocity mapping of chronic thoracic aortic dissection: a preliminary investigation. Magn Reson Med Sci 2011;10(2):93–9.
- [522] Müller-Eschner M, Rengier F, Partovi S, Unterhinninghofen R, Böckler D, Ley S, von Tengg-Kobligk H. Tridirectional phase-contrast magnetic resonance velocity mapping depicts severe hemodynamic alterations in a patient with aortic dissection type Stanford B. J Vasc Surg 2011;54(2):559–62.
- [523] Guo B, Pirola S, Guo D, Dong Z, Xu XY, Fu W. Hemodynamic evaluation using four-dimensional flow magnetic resonance imaging for a patient with multichanneled aortic dissection. J Vasc Surg Cases Innov Tech 2018;4(1):67–71. https://doi.org/10.1016/j.jvscit.2017.11.005.
- [524] Chen CW, Tseng YH, Lin CC, Kao CC, Wong MY, Ting H, Huang YK. Aortic dissection assessment by 4D phase-contrast MRI with hemodynamic parameters: the impact of stent type. Quant Imaging Med Surg 2021;11(2):490–501.
- [525] Fang K, Zhao J, Luo M, Xue Y, Wang H, Ye L, Zhang X, Zheng L, Shu C. Quantitative analysis of renal blood flow during thoracic endovascular aortic repair in type B aortic dissection using syngo iFlow. Quant Imaging Med Surg 2021;11(8):3726–34.
- [526] Takahashi K, Sekine T, Miyagi Y, Shirai S, Otsuka T, Kumita S, Ishii Y. Four-dimensional flow analysis reveals mechanism and impact of turbulent flow in the dissected aorta. Eur J Cardiothorac Surg 2021;60(5):1064–72.
- [527] Urmeneta Ulloa J, Martínez de Vega V, Álvarez Vázquez A, Rivas Oyarzabal J, Forteza Gil A, Cabrera JÁ. 4D-flow cardiac magnetic resonance imaging in chronic aortic dissection assessment: a powerful qualiquantitative diagnostic tool. JACC Case Rep 2022;4(21):1399–1403. https://doi.org/10.1016/j.jaccas. 2022.07.006.
- [528] Ruiz-Muñoz A, Guala A, Dux-Santoy L, Teixidó-Turà G, Servato ML, Valente F, Garrido-Oliver J, Galian-Gay L, Gutiérrez L, Fernandez-Galera R, Casas G, González-Alujas T, Cuéllar-Calabria H, Johnson KM, Wieben O, Ferreira-Gonzalez I, Evangelista A, Rodriguez-Palomares J. False lumen rotational flow and aortic stiffness are associated with aortic growth rate in patients with chronic aortic dissection of the descending aorta: a 4D flow cardiovascular magnetic resonance study. J Cardiovasc Magn Reson 2022;4(21):1399–403.
- [529] Liu SZ, Engel JS, Bisen JB, Kilinc O, Quinn S, Hoel AW, Mehta CK, Markl M, Allen BD. Entry tear hemodynamics using 4D flow MRI in a patient with acute type B aortic dissection. Radiol Case Rep 2023;18(3):1037–40.
- [530] Kilinc O, Chu S, Baraboo J, Weiss EK, Engel J, Maroun A, Giese D, Jin N, Chow K, Bi X, Davids R, Mehta C, Malaisrie SC, Hoel A, Carr J, Markl M, Allen BD. Hemodynamic evaluation of type B aortic dissection using compressed sensing accelerated 4D flow MRI. J Magn Reson Imaging 2023;57(6):1752–63.
- [531] Bellala AP, Valakkada J, Ayappan A, Kannath S, Shivanesan P. False lumen regurgitation fraction and energy loss in the aorta measured using four-dimensional flow MRI to predict expansion of acute uncomplicated type B aortic dissection: a prospective study. Clin Radiol 2024;79(4):e616–23.
- [532] Nayler GL, Firmin DN, Longmore DB. Blood flow imaging by cine magnetic resonance. J Comput Assist Tomogr 1986;10(5):715–22.
- [533] Menichini C. Mathematical, Modelling of Thrombus Formation in Type B Aortic Dissection, Ph.D. thesis. Imperial College London, Department of Chemical Engineering, 2018.
- [534] Hathcock JJ. Flow effects on coagulation and thrombosis. Arterioscler Thromb Vasc Biol 2006;26(8):1729–37.
- [535] Fogelson AL, Neeves KB. Fluid mechanics of blood clot formation. Annu Rev Fluid Mech 2015;47:377–403.
- [536] Smith SA, Travers RJ, Morrissey JH. How it all starts: initiation of the clotting cascade. Crit Rev Biochem Mol Biol 2015;50(4):326–36.
- [537] Tomaiuolo M, Brass LF, Stalker TJ. Regulation of platelet activation and coagulation and its role in vascular injury and arterial thrombosis. Interv Cardiol Clin 2018;6(1):1–12.
- [538] Malek AM, Alper SL, Izumo S. Hemodynamic shear stress and its role in atherosclerosis. JAMA 1999;282(21):2035-42.
- [539] Rana A, Westein E, Niego B, Hagemeyer CE. Shear-dependent platelet aggregation: mechanisms and therapeutic opportunities. Front Cardiovasc Med 2019;6: 141.
- [540] Chernysh IN, Nagaswami C, Kosolapova S, Peshkova AD, Cuker A, Cines DB, Cambor CL, Litvinov RI, Weisel JW. The distinctive structure and composition of arterial and venous thrombi and pulmonary emboli. Sci Rep 2020;10(1):5112.
- [541] Boodt N, Snouckaert van Schauburg PRW, Hund HM, Fereidoonnezhad B, McGarry JP, Akyildiz AC, van Es ACGM, De Meyer SF, Dippel DWJ, Lingsma HF, van Beusekom HMM, van der Lugt A, Gijsen FJH. Mechanical characterization of thrombi retrieved with endovascular thrombectomy in patients with acute ischemic stroke. Stroke 2021;52(8):2510–7.
- [542] Stein PD, Sabbah HN, Pitha JV. Continuing disease process of calcific aortic stenosis. Role of microthrombi and turbulent flow. Am J Cardiol 1977;39(2): 159–63.
- [543] Folie BJ, McIntire LV. Mathematical analysis of mural thrombogenesis. Concentration profiles of platelet-activating agents and effects of viscous shear flow. Biophys J 1989;56(6):1121–41.
- [544] Bluestein D, Niu L, Schoephoerster RT, Dewanjee MK. Fluid mechanics of arterial stenosis: relationship to the development of mural thrombus. Ann of Biomed Eng 1997;25(2):344–56.
- [545] Woolf N. Pathology: Basic and Systemic. Saunders; 1998.
- [546] Colman RW, Marder VJ, Clowes AW, George JN, Goldhaber SZ. Hemostasis and Thrombosis: Basic Principles and Clinical Practice, Lippincott Williams & Wilkin, 2001. https://doi.org/10.1373/49.2.345/-a.
- [547] Helms CC, Marvel M, Zhao W, Stahle M, Vest R, Kato GJ, Lee JS, Christ G, Gladwin MT, Hantgan RR, Kim-Shapiro DB. Mechanisms of hemolysis-associated platelet activation. J Thromb Haemost 2013;11(12):2148–54.
- [548] Hellums JD. 1993 Whitaker lecture: biorheology in thrombosis research. Ann Biomed Eng 1994;22(5):445–55.
- [549] Sutera SP, Nowak MD, Joist JH, Zeffren DJ, Bauman JE. A programmable, computer-controlled cone-plate viscometer for the application of pulsatile shear stress to platelet suspensions. Biorheology 1988;25(3):449–59.
- [550] Nesbitt W, Westein E, Tovar-Lopez F, Tolouei E, Mitchell A, Fu J, Carberry J, Fouras A, Jackson S. A shear gradient-dependent platelet aggregation mechanism drives thrombus formation. Nat Med 2009;15(6):665–73.
- [551] Sheriff J, Bluestein D, Girdhar G, Jesty J. High-shear stress sensitizes platelets to subsequent low-shear conditions. Ann Biomed Eng 2010;38(4):1442–50.
   [552] Yin W, Shanmugavelayudam SK, Rubenstein DA. The effect of physiologically relevant dynamic shear stress on platelet and endothelial cell activation. Thromb
- Res 2011;127(3):235-41. [553] Tanaka A, Sakakibara M, Ishii H, Hayashida R, Jinno Y, Okumura S, Okada K, Murohara T. Influence of the false lumen status on clinical outcomes in patients
- [553] Tanaka A, Sakakibara M, Ishii H, Hayashida R, Jinno Y, Okumura S, Okada K, Muronara T. Influence of the faise fumen status on clinical outcomes in patients with acute type B aortic dissection. J Vasc Surg 2014;59(2):321–6.
- [554] Trimarchi S, Nienaber CA, Rampoldi V, Myrmel T, Suzuki T, Bossone E, Tolva V, Deeb MG, Upchurch GR, Cooper Jr JV, Fang J, Isselbacher EM, Sundt 3rd TM Eagle KA. Role and results of surgery in acute type B aortic dissection: insights from the International Registry of Acute Aortic Dissection (IRAD). Circulation 2006;114:1357–64. https://doi.org/10.1161/CIRCULATIONAHA.105.000620.
- [555] Song S-W, Chang B-C, Cho B-K, Yi G, Youn Y-N, Lee S, Yoo K-J. Effects of partial thrombosis on distal aorta after repair of acute DeBakey type I aortic dissection. J Thorac Cardiovasc Surg 2010;139(4):841–847.e1;discussion 847. https://doi.org/10.1016/j.jtcvs.2009.12.007.
- [556] Higashigaito K, Sailer AM, van Kuijk SMJ, Willemink MJ, Hahn LD, Hastie TJ, Miller C, Fischbein MP, Fleischmann D. Aortic growth and development of partial false lumen thrombosis are associated with late adverse events in type B aortic dissection. J Thorac Cardiovasc Surg 2021;161(4):1184–1190.e2.
- [557] Larsen M, Bartnes K, Tsai TT, Eagle KA, Evangelista A, Nienaber CA, Suzuki T, Fattori R, Froehlich JB, Hutchison S, Sundt TM, Januzzi JL, Isselbacher EM, Montgomery DG, Myrmel T. Extent of preoperative false lumen thrombosis does not influence long-term survival in patients with acute type A aortic dissection. J Am Heart Assoc 2013;2(4):e000112.
- [558] Tsai MT, Wu HY, Roan JN, Tsai YS, Hsieh PC, Yang YJ, Luo CY. Effect of false lumen partial thrombosis on repaired acute type A aortic dissection. J Thorac Cardiovasc Surg 2014;148(5):2140–2146.e3.
- [559] Fattori R, Bacchi-Reggiani L, Bertaccini P, Napoli G, Fusco F, Longo M, Pierangeli A, Gavelli G. Evolution of aortic dissection after surgical repair. Am J Cardiol 2000;86(8):868–72.
- [560] Bernard Y, Zimmermann H, Chocron S, Litzler J-F, Kastler B, Etievent J-P, Meneveau N, Schiele F, Bassand J-P. False lumen patency as a predictor of late outcome in aortic dissection. Am J Cardiol 2001;87(12):1378–82.
- [561] Akutsu K, Nejima J, Kiuchi K, Sasaki K, Ochi M, Tanaka K, Takano T. Effects of the patent false lumen on the long-term outcome of type B acute aortic dissection. Eur J Cardiothorac Surg 2004;26(2):359–66.

- [562] Fattouch K, Sampognaro R, Navarra E, Caruso M, Pisano C, Coppola G, Speziale G, Ruvolo G. Long-term results after repair of type A acute aortic dissection according to false lumen patency. Ann Thorac Surg 2009;88(4):1244–50.
- [563] Kim JB, Lee C-H, Lee TY, Jung S-H, Choo SJ, Lee JW. Descending aortic aneurysmal changes following surgery for acute DeBakey type I aortic dissection. Eur J Cardiothorac Surg 2012;42(5):851–6.
- [564] De León Ayala IA, Yang Y-H, Chien T-M, Yu C-P, Chen C-W, Chen C-C, Chen Y-F. Partially patent false lumen does not exhibit the highest growth rate. Int J Cardiol 2014;175(2):385–388. https://doi.org/10.1016/j.ijcard.2014.05.039.

[565] Satta J, Läärä E, Juvonen T. Intraluminal thrombus predicts rupture of an abdominal aortic aneurysm. J Vasc Surg 1996;23(4):737-9.

- [566] Vorp DA, Lee PC, Wang DHJ, Makaroun MS, Nemoto EM, Ogawa S, Webster MW. Association of intraluminal thrombus in abdominal aortic aneurysm with local hypoxia and wall weakening. J Vasc Surg 2001;34(2):291–9.
- [567] Nienaber CA, Rousseau H, Eggebrecht H, Kische S, Fattori R, Rehders TC, Kundt G, Scheinert D, Czerny M, Kleinfeldt T, Zipfel B, Labrousse L, Ince H. Randomized comparison of strategies for type B aortic dissection: the INvestigation of STEnt grafts in Aortic Dissection (INSTEAD) trial. Circulation 2009;120 (25):2519–28.
- [568] Roberts JT. Medionecrosis aortae idiopathica cystica: report of a case, with "healed" dissecting aneurysm. Am Heart J 1939;18(2):188-200.
- [569] Halstead JC, Meier M, Etz C, Spielvogel D, Bodian C, Wurm M, Shahani R, Griepp RB. The fate of the distal aorta after repair of acute type A aortic dissection. J Thorac Cardiovasc Surg 2007;133(1):127–35.
- [570] Agrafiotis E, Zimpfer D, Mächler H, Holzapfel GA. Review of systemic mock circulation loops for evaluation of implantable cardiovascular devices and biological tissues. J Endovasc Ther 2024:1526602824123876.
- [571] Blanton FSJ, Muller WHJ, Warren WD. Experimental production of dissecting aneurysms of the aorta. Surgery 1959;45(1):81-90.
- [572] Razavi MK, Nishimura E, Slonim S, Zeigler W, Kee S, Witherall HL, Semba CP, Dake MD. Percutaneous creation of acute type-B aortic dissection: an experimental model for endoluminal therapy. J Vasc Interv Radiol 1998;9(4):626–32.
- [573] Fujii H, Tanigawa N, Okuda Y, Komemushi A, Sawada S, Imamura H. Creation of aortic dissection model in swine. Jpn Circ J 2000;64(9):736-7.
- [574] Terai H, Tamura N, Yuasa S, Nakamura T, Shimizu Y, Komeda M. An experimental model of Stanford type B aortic dissection. J Vasc Interv Radiol 2005;16(4): 515–9.
- [575] Cui JS, Zhuang SJ, Zhang J, Mei ZJ, Jing ZP, Liao MF. Two-end intimal flap suturing method for establishing Stanford B type aortic dissection in a canine model. Eur J Vasc Endovasc Surg 2009;38(5):603–7.
- [576] Tang J, Wang Y, Hang W, Fu W, Jing Z. Controllable and uncontrollable Stanford type B aortic dissection in canine models. Eur Surg Res 2010;44(3–4):179–84.
   [577] Okuno T, Yamaguchi M, Okada T, Takahashi T, Sakamoto N, Ueshima E, Sugimura K, Sugimoto K. Endovascular creation of aortic dissection in a swine model
- [577] Okuno I, Yamaguchi M, Okada I, Takanashi I, Sakamoto N, Oesnima E, Sugimura K, Su
- [578] Wu C-A, Squelch A, Sun Z. Investigation of three-dimensional printing materials for printing aorta model replicating type B aortic dissection. Curr Med Imaging 2021;17(7):843–9.
- [579] Liu Z, Bi J, Niu F, Liang H, Fan J, Li J, Wang D, Dai X. Construction of a canine model with acute type B aortic dissection using a self-made pressure water flow device. Exp Anim 2023;72(2):253–61.
- [580] Morales DL, Quin JA, Braxton JH, Hammond GL, Gusberg RJ, Elefteriades JA. Experimental confirmation of effectiveness of fenestration in acute aortic dissection. Ann Thorac Surg 1998;66(5):1679–83.
- [581] El Batti S, Abdallah IB, Dufetelle E, Julia P, Menasche P, Alsac J-M. Experimental evaluation of endovascular fenestration scissors in an ovine model of aortic dissection. Eur J Vasc Endovasc Surg 2018;56(3):373–80.
- [582] Trent M, Parsonnet V, Shoenfeld R, Brener B, Eisenbud D, Novick A, Campbell A, Ferrara-Ryan M, Villanueva A. A balloon-expandable intravascular stent for obliterating experimental aortic dissection. J Vasc Surg 1990;11(5):707–17.
- [583] Moon MR, Dake MD, Pelc LR, Liddell R, Castro LJ, Mitchell RS, Miller DC. Intravascular stenting of acute experimental type B dissections. J Surg Res 1993;54 (4):381–8.
- [584] Kato N, Hirano T, Takeda K, Nakagawa T, Mizumoto T, Yuasa H. Treatment of acute aortic dissections with expandable metallic stents: experimental study. J Vasc Interv Radiol 1994;5(3):417–23.
- [585] Kato M, Matsuda T, Kaneko M, Ueda T, Kuratani T, Yoshioka Y, Ohnishi K. Experimental assessment of newly devised transcatheter stent-graft for aortic dissection. Ann Thorac Surg 1995;59(4):908–14;discussion 914–5. https://doi.org/10.1016/0003-4975(94)00958-a.
- [586] Marty-Ané CH, SerreCousiné O, Laborde JC, Costes V, Mary H, Senac JP. Use of a balloon-expandable intravascular graft in the management of type B aortic dissection in an animal model. J Vasc Interv Radiol 1995;6(1):97–103.
- [587] Faure EM, Canaud L, Cathala P, Serres I, Marty-Ané C, Alric P. Assessment of abdominal branch vessel patency after bare-metal stenting of the thoracoabdominal aorta in a human ex vivo model of acute type B aortic dissection. J Vasc Surg 2015;61(5):1299–305.
- [588] Mohl L, Karl R, Hagedorn MN, Runz A, Skornitzke S, Toelle M, Bergt CS, Hatzl J, Uhl C, Böckler D, Meisenbacher D, Engelhardt S. Simulation of thoracic endovascular aortic repair in a perfused patient-specific model of type B aortic dissection. Int J Comput Assist Radiol Surg 2024. https://doi.org/10.1007/ s11548-024-03190-3.
- [589] Zannoli R, Corazza I, Cremonesi A, Branzi A. A mechanical device for aortic compliance modulation: in vitro simulation of aortic dissection treatment. J Biomech 2007;40(14):3089–95.
- [590] Moore MJ, Malaxos L, Doyle BJ. Development of a shear-thinning biomaterial as an endovascular embolic agent for the treatment of type B aortic dissection. J Mech Behav Biomed Mater 2019;99:66–77.
- [591] Halliday JH, Robertson JS. Dissecting aneurysm of the aorta. Proc R Australas Coll Physicians 1946;1(1):3–15.
- [592] Robertson JS, Viner Smith K. An analysis of certain factors associated with the production of experimental dissection of the aortic media, in relation to the pathogenesis of dissecting aneurysm. J Pathol Bacteriol 1948;60(1):43–49. https://doi.org/10.1002/path.1700600105.
- [593] Hirst Jr AE, Johns Jr VJ. Experimental dissection of media of aorta by pressure. Circ Res 1962;10(6):897–903.
- [594] Milazzo S. Dissecting aortic aneurysm. Med J Aust 1952;1(24):807–13.
- [595] Carson MW, Roach MR. The strength of the aortic media and its role in the propagation of aortic dissection. J Biomech 1990;23(6):579-88.
- [596] Roach MR, Song S. Variations in strength of the porcine aorta as a function of location. Clin Invest Med 1994;17(4):308-18.
- [597] Song S, Roach MR. Quantitative changes in the size of fenestrations of the elastic laminae of sheep thoracic aorta studied with SEM. Blood vessels 1983;20(3): 145–53.
- [598] Song S, Roach MR. Comparison of fenestrations in internal elastic laminae of thoracic and abdominal aortae of dog, sheep, and rabbit. Blood vessels 1984;21 (2):90–7.
- [599] Song SH, Roach MR. A morphological comparison of aortic elastin from five species as seen with electron microscopy. Acta Anat 1985;123(1):45–50.
- [600] Roach MR, He JC, Kratky RG. Tear propagation in isolated, pressurized porcine thoracic aortas. Can J Cardiol 1999;15(5):569–75.
- [601] Prokop EK, Palmer RF, Wheat Jr MW. Hydrodynamic forces in dissecting aneurysms. In-vitro studies in a Tygon model and in dog aortas, Circ Res 1970;27(1): 121–7.
- [602] Carney WIJ, Rheinlander HF, Cleveland RJ. Control of acute aortic dissection. Surgery 1975;78(1):114-20.
- [603] Senoo M, Teramoto S, Uchida H. Experimental study of ascending aortic dissection: hemodynamics and pathologic aspects. Vasc Endovasc Surg 1989;23(5): 381–97.
- [604] Mitsui H, Uchida H, Teramoto S. Correlation between the layer of an intimal tear and the progression of aortic dissection. Acta Med Okayama 1994;48(2):93–9.

[605] Tam AS, Sapp MC, Roach MR. The effect of tear depth on the propagation of aortic dissections in isolated porcine thoracic aorta. J Biomech 1998;31:673–6.
[606] Peelukhana SV, Wang Y, Berwick Z, Kratzberg J, Krieger J, Roeder B, Cloughs RE, Hsiao S, Chambers A, Kassab GS. Role of pulse pressure and geometry of primary entry tear in acute type B dissection propagation. Ann Biomed Eng 2017;45(3):592–603.

[607] Guo B, Dong Z, Pirola S, Liu Y, Menichini C, Xu XY, Guo D, Fu W. Dissection level within aortic wall layers is associated with propagation of type B aortic dissection: a swine model study. Eur J Vasc Endovasc Surg 2019;58(3):415–25.

- [608] Chi Q-Z, Ge Y-Y, Cao Z, Long L-L, Mu L-Z, He Y, Luan Y. Experimental study of the propagation process of dissection using an aortic silicone phantom. J Funct Biomater 2022;13(4):290.
- [609] Brunet J, Pierrat B, Adrien J, Maire E, Lane BA, Curt N, Bravin A, Laroche N, Badel P. In situ visualization of aortic dissection propagation in notched rabbit aorta using synchrotron X-ray tomography. Acta Biomater 2023;155:449–60.
- [610] Tsai TT, Schlicht MS, Khanafer K, Bull JL, Valassis DT, Williams DM, Berguer R, Eagle KA. Tear size and location impacts false lumen pressure in an ex vivo model of chronic type B aortic dissection. J Vasc Surg 2008;47(4):844–51.
- [611] Liang S, Jia H, Dong W, Li Z, Zhou G, Zhang Y, Duanduan C, Xiong J. Hemodynamic study of Stanford type B aortic dissection based on ex vivo porcine aorta models. J Endovasc Ther 2022;30(3). 15266028221081089.
- [612] Zimmermann J, Bäumler K, Loecher M, Cork TE, Marsden AL, Fleischmann D, Ennis DB. Hemodynamic effects of tear size in aortic dissections: comparing in vitro flow magnetic resonance imaging with fluid-structure interaction simulations. Sci Rep 2023;13:22557.
- [613] Ahuja A, Guo X, Noblet JN, Krieger JF, Roeder B, Haulon S, Chambers S, Kassab GS. Validated computational model to compute re-apposition pressures for treating type-B aortic dissections. Front Physiol 2018;9:513.
- [614] Iwai F, Sostman HD, Evans AJ, Nadel SN, Hedlund LW, Beam CA, Charles HC, Spritzer CE. Cine phase-contrast magnetic resonance imaging for analysis of flow phenomena in experimental aortic dissection. Invest Radiol 1991;26(12):1071–8.
- [615] Chung J, Elkins C, Sakai T, Kato N, Vestring T, Semba C, Slonim S, Dake M. True-lumen collapse in aortic dissection: part I. Evaluation of causative factors in phantoms with pulsatile flow, Radiology 2000;214(1):87–98.
- [616] Chung JW, Elkins C, Sakai T, Kato N, Vestring T, Semba CP, Slonim SM, Dake MD. True-lumen collapse in aortic dissection: part II. Evaluation of treatment methods in phantoms with pulsatile flow. Radiology 2000;214(1):99–106.
- [617] Qing KX, Chan YC, Lau SF, Yiu WK, Ting AC, Cheng SW. Ex-vivo haemodynamic models for the study of Stanford type B aortic dissection in isolated porcine aorta. Eur J Vasc Endovasc Surg 2012;44(4):399–405.
- [618] Rudenick PA, Bijnens BH, García-Dorado D, Evangelista A. An in vitro phantom study on the influence of tear size and configuration on the hemodynamics of the lumina in chronic type B aortic dissections. J Vasc Surg 2013;57(2):464–474.e5.
- [619] Faure EM, Canaud L, Cathala P, Serres I, Marty-Ané C, Alric P. Human ex-vivo model of Stanford type B aortic dissection. J Vasc Surg 2014;60(3):767-75.
- [620] Veger HTC, Westenberg JJM, Visser MJT. The role of branch vessels in aortic type B dissection: an in vitro study. Eur J Vasc Endovasc Surg 2015;49(4):375–81.[621] Veger HTC, Pasveer EH, Visser MJT. Where to fenestrate in aortic dissection type B? An ex vivo study. Ann Vasc Surg 2017;43:296–301.
- [622] Veger HTC, Pasveer EH, Westenberg JJM, Wever JJ, van Eps RGS. The influence of aortic wall elasticity on the false lumen in aortic dissection: an in vitro study. Vasc Endovascular Surg 2020;54(7):592–7.
- [623] Veger HTC, Pasveer EH, Westenberg JJM, Wever JJ, van Eps RGS. Wall shear stress assessment of the false lumen in acute type B aortic dissection visualized by 4-dimensional flow magnetic resonance imaging: an ex-vivo study. Vasc Endovascular Surg 2021;55(7):696–701.
- [624] Birjiniuk J, Ruddy JM, Iffrig E, Henry TS, Leshnower BG, Oshinski JN, Ku DN, Veeraswamy RK. Development and testing of a silicone in vitro model of descending aortic dissection. J Surg Res 2015;198(2):502–7.
- [625] Birjiniuk J, Timmins LH, Young M, Leshnower BG, Oshinski JN, Ku DN, Veeraswamy RK. Pulsatile flow leads to intimal flap motion and flow reversal in an in vitro model of type B aortic dissection. Cardiovasc Eng Technol 2017;8(3):378–89.
- [626] Birjiniuk J, Veeraswamy RK, Oshinski JN, Ku DN. Intermediate fenestrations reduce flow reversal in a silicone model of Stanford type B aortic dissection. J Biomech 2019;93:101–10.
- [627] Marconi S, Lanzarone E, De Beaufort H, Conti M, Trimarchi S, Auricchio F. A novel insight into the role of entry tears in type B aortic dissection: pressure measurements in an in vitro model. Int J Artif Organs 2017;40(10):563–74.
- [628] Canchi S, Guo X, Phillips M, Berwick Z, Kratzberg J, Krieger J, Roeder B, Haulon S, Chambers S, Kassab GS. Role of re-entry tears on the dynamics of type B dissection flap. Ann Biomed Eng 2018;46(1):186–96.
- [629] Salameh MJ, Saade C, Oweis GF. Experimental insight into the hemodynamics and perfusion of radiological contrast in patent and non-patent aortic dissection models. Cardiovasc Eng Technol 2019;10(2):314–28.
- [630] Zadrazil I, Corzo C, Voulgaropoulos V, Markides CN, Xu XY. A combined experimental and computational study of the flow characteristics in a type B aortic dissection: effect of primary and secondary tear size. Chem Eng Res Des 2020;160:240–53.
- [631] Chen D, Liang S, Li Z, Mei Y, Dong H, Ma Y, Zhao J, Xu S, Zheng J, Xiong J. A mock circulation loop for in vitro hemodynamic evaluation of aorta: application in aortic dissection. J Endovasc Ther 2022;29(1):132–42.
- [632] Zimmermann J, Loecher M, Kolawole F, Bäumler K, Gifford K, Dual SA, Levenston M, Marsden AL, Ennis DB. On the impact of vessel wall stiffness on quantitative flow dynamics in a synthetic model of the thoracic aorta. Sci Rep 2021;11:6703.
- [633] Bollache E, van Ooij P, Powell A, Carr J, Markl M, Barker AJ. Comparison of 4D flow and 2D velocity-encoded phase contrast MRI sequences for the evaluation of aortic hemodynamics. Int J Cardiovasc Imaging 2016;32(10):1529–41.
- [634] Franzetti G, Díaz-Zuccarini V, Balabani S. Design of an in vitro mock circulatory loop to reproduce patient-specific vascular conditions: toward precision medicine. ASME JESMDT 2019;2(4):041004.
- [635] Bonfanti M, Franzetti G, Homer-Vanniasinkam S, Díaz-Zuccarini V, Balabani S. A combined in vivo, in vitro, in silico approach for patient-specific haemodynamic studies of aortic dissection. Ann Biomed Eng 2020;48(12):2950–64.
- [636] Franzetti G, Bonfanti M, Homer-Vanniasinkam S, Díaz-Zuccarini V, Balabani S. Experimental evaluation of the patient-specific haemodynamics of an aortic dissection model using particle image velocimetry. J Biomech 2022;134:110963.
- [637] Chatpattanasiri C, Franzetti G, Bonfanti M, Díaz-Zuccarini V, Balabani S. Towards reduced order models via robust proper orthogonal decomposition to capture personalised aortic haemodynamics. J Biomech 2023;158:111759.
- [638] Aghilinejad A, Wei H, Bilgi C, Paredes A, DiBartolomeo A, Pahlevan NM. Framework development for patient-specific compliant aortic dissection phantom model fabrication: magnetic resonance imaging validation and deep-learning segmentation. J Biomech Eng 2023;145(9):091010.
- [639] Wang S, Jia H, Xi Y, Yuan P, Wu M, Guo W, Chen D, Xiong J. The role of location, length, and thickness of the intimal flap in the propagation of Stanford type B aortic dissection based on ex vivo porcine aorta models. J Endovasc Ther 2023:15266028231199930. https://doi.org/10.1177/15266028231199930.
- [640] Birjiniuk J, Oshinski JN, Ku DN, Veeraswamy RK. Endograft exclusion of the false lumen restores local hemodynamics in a model of type B aortic dissection. J Vasc Surg 2020;71(6):2108–18.
- [641] Dalbosco M, Haspinger C, Li K, Murtada S-I, Pukaluk A, Rolf-Pissarczyk M, Sherifova S, Sommer G. Multiscale experimental characterization and computational modeling of the human aorta. In: Sommer G, Li K, Haspinger DC, Ogden RW, editors. Solid (Bio)mechanics: Challenges of the Next Decade: A Book Dedicated to Professor Gerhard A. Holzapfel: Springer International Publishing, Cham; 2022. p. 3–52.
- [642] Balzani D, Brinkhues S, Holzapfel GA. Constitutive framework for the modeling of damage in collagenous soft tissues with application to arterial walls. Comput Methods Appl Mech Eng 2012;213–216:139–51.
- [643] Holzapfel GA, Fereidoonnezhad B. Modeling of damage in soft biological tissues. Hyperelastic constitutive laws for finite element modeling. In: Payan Y, Ohayon J, editors. Biomechanics of Living Organs. Oxford: Academic Press; 2017. p. 101–23.
- [644] Fries T-P, Belytschko T. The extended/generalized finite element method: an overview of the method and its applications. Int J Numer Meth Engng 2010;84(3): 253–304.
- [645] Park K, Paulino GH. Cohesive zone models: a critical review of traction-separation relationships across fracture surfaces. Appl Mech Rev 2011;64(6):060802.
- [646] Piska R, Sivadas K, Boyina K, Vuppuluri A, Chaurasia A, Parimi C, Rabczuk T. Recent trends in computational damage models: an overview. Theor Appl Fract Mech 2024;132:104494.
- [647] Tiessen IM, Roach MR. Factors in the initiation and propagation of aortic dissections in human autopsy aortas. J Biomech Eng 1993;115(1):123–5.
- [648] Gasser TC, Holzapfel GA. Physical and numerical modeling of dissection propagation in arteries caused by balloon angioplasty. In: Hamza MH, editor. Proceedings of the 3rd IASTED International Conference on Biomechanics, Anaheim, 2005, p. 229–33.

- [649] Holzapfel GA, Schulze-Bauer CAJ, Stadler M. Mechanics of angioplasty: wall, balloon and stent. In: Casey J, Bao G, editors. Mechanics in Biology, Vol. AMD-Vol. 242/BED-Vol. 46. New York: The American Society of Mechanical Engineers (ASME); 2000. p. 141–56.
- [650] Gasser TC, Holzapfel GA. Modeling the propagation of arterial dissection. Eur J Mech A-Solid 2006;25(4):617-33.
- [651] Gasser TC, Holzapfel GA. 3D crack propagation in unreinforced concrete.: a two-step algorithm for tracking 3D crack paths. Comput Methods Appl Mech Eng 2006;195(37):5198–219.
- [652] Gasser TC, Ogden RW, Holzapfel GA. Hyperelastic modelling of arterial layers with distributed collagen fibre orientations. J R Soc Interface 2006;3(6):15–35. [653] Ferrara A, Pandolfi A. A numerical study of arterial media dissection processes. Int J Fracture 2010;166:21–33.
- [654] Ferrara A, Pandolfi A. Numerical modelling of fracture in human arteries. Comput Methods Biomech Biomed Engin 2008;11(5):553-67.
- [655] Tsamis A, Phillippi JA, Koch RG, Pasta S, D'Amore A, Watkins SC, Wagner WR, Gleason TG, Vorp DA. Fiber micro-architecture in the longitudinal-radial and circumferential-radial planes of ascending thoracic aortic aneurysm media. J Biomech 2013;46(16):2787–94.
- [656] Pal S, Tsamis A, Pasta S, D'Amore A, Gleason TG, Vorp DA, Maiti S. A mechanistic model on the role of "radially-running" collagen fibers on dissection properties of human ascending thoracic aorta. J Biomech 2014;47(5):981–988. https://doi.org/10.1016/j.jbiomech.2014.01.005.
- [657] Vorp DA, Schiro BJ, Ehrlich MP, Juvonen TS, Ergin MA, Griffith BP. Effect of aneurysm on the tensile strength and biomechanical behavior of the ascending thoracic aorta. Ann Thorac Surg 2003;75(4):1210–4.
- [658] Pasta S, Phillippi JA, Gleason TG, Vorp DA. Effect of aneurysm on the mechanical dissection properties of the human ascending thoracic aorta. J Thorac Cardiovasc Surg 2012;143(2):460–7.
- [659] Shah SB, Witzenburg C, Hadi MF, Wagner HP, Goodrich JM, Alford P, Barocas VH. Prefailure and failure mechanics of the porcine ascending thoracic aorta: experiments and a muliscale model. J Biomech Eng 2014;136(2):021028.
- [660] Witzenburg CM, Dhume RY, Shah SB, Korenzuk C, Wagner HP, Alford P, Barocas VH. Failure of the porcine ascending aorta: multidirectional experiments and a unifying microstructural model. J Biomech Eng 2017;139(3):031005. https://doi.org/10.1115/1.4035264.
- [661] Leng X, Chen X, Deng X, Sutton M, Lessner SM. Modeling of experimental atherosclerotic plaque delamination. Ann Biomed Eng 2015;43(12):2838–51.
   [662] Leng X, Zhou B, Deng X, Davis L, Lessner S, Sutton M, Shazly T. Experimental and numerical studies of two arterial wall delamination modes. J Mech Behav Biomed Mater 2018;77:321–30.
- [663] Thunes JR, Pal S, Fortunato RN, Phillippi JA, Gleason TG, Vorp DA, Maiti S. A structural finite element model for lamellar unit of aortic media indicates heterogeneous stress field after collagen recruitment. J Biomech 2016;49(9):1562–9.
- [664] Pasta S, Rinaudo A, Luca A, Pilato M, Scardulla C, Gleason TG, Vorp DA. Difference in hemodynamic and wall stress of ascending thoracic aortic aneurysms with bicuspid and tricuspid aortic valve. J Biomech 2013;46(10):1729–38.
- [665] Thunes JR, Phillippi JA, Gleason TG, Vorp DA, Maiti S. Structural modeling reveals microstructure-strength relationship for human ascending thoracic aorta. J Biomech 2018;71:84–93.
- [666] Noble C, van der Sluis O, Voncken RMJ, Burke O, Franklin SE, Lewis R, Taylor ZA. Simulation of arterial dissection by a penetrating external body using cohesive zone modelling. J Mech Behav Biomed Mater 2017;71:95–105.
- [667] Holzapfel GA. Nonlinear Solid Mechanics. A Continuum Approach for Engineering. Chichester: John Wiley & Sons; 2000.
- [668] Tong J, Sommer G, Regitnig P, Holzapfel GA. Dissection properties and mechanical strength of tissue components in human carotid bifurcations. Ann Biomed Eng 2011;39:1703–19.
- [669] Noble C, Smulders N, Lewis R, Carré MJ, Franklin SE, MacNeil S, Taylor ZA. Controlled peel testing of a model tissue for diseased aorta. J Biomech 2016;49 (15):3667–75.
- [670] Helfenstein-Didier C, Taïnoff D, Viville J, Adrien J, Maire É, Badel P. Tensile rupture of medial arterial tissue studied by X-ray micro-tomography on stained samples. J Mech Behav Biomed Mater 2018;78:362–8.
- [671] Yu X, Suki B, Zhang Y. Avalanches and power law behavior in aortic dissection propagation. Sci Adv 2020;6(21):eaaz1173.
- [672] Hadi MF, Sander EA, Barocas VH. Multiscale model predicts tissue-level failure from collagen fiber-level damage. J Biomech Eng 2012;134:091005.
- [673] Miyazaki H, Hayashi K. Tensile tests of collagen fibers obtained from the rabbit patellar tendon. Biomed Microdevices 1999;2:151-7.
- [674] Wang R, Yu X, Zhang Y. Mechanical and structural contributions of elastin and collagen fibers to interlamellar bonding in the arterial wall. Biomech Model Mechanobiol 2021;20(1):93–106.
- [675] Wang L, Hill NA, Roper SM, Luo X. Modelling of tear propagation and arrest in fibre-reinforced soft tissue subject to internal pressure. J Eng Math 2015;95(1): 249–65.
- [676] Petrivý Z, Horný L, Tichý P. Traction-separation law parameters for the description of age-related changes in the delamination strength of the human descending thoracic aorta. Biomech Model Mechanobiol 2024.
- [677] Wang L, Hill NA, Roper SM, Luo X. Propagation of dissection in a residually-stressed artery model. Biomech Model Mechanobiol 2017;16(1):139–49.
  [678] Holzapfel GA, Ogden RW. Modelling the layer-specific three-dimensional residual stresses in arteries, with an application to the human aorta. J R Soc Interface 2010;7(46):787–99.
- [679] Wang L, Hill NA, Roper SM, Luo X. Modelling peeling- and pressure-driven propagation of arterial dissection. J Eng Math 2018;109(1):227-38.
- [680] Raina A, Miehe C. A phase-field model for fracture in biological tissues. Biomech Model Mechanobiol 2016;15:479–96.
- [681] Gültekin O, Dal H, Holzapfel GA. A phase-field approach to model fracture of arterial walls: theory and finite element analysis. Comput Methods Appl Mech Eng 2016;312:542–66.
- [682] Gültekin O, Dal H, Holzapfel GA. Numerical aspects of anisotropic failure in soft biological tissues favor energy-based criteria: a rate-dependent anisotropic crack phase-field model. Comput Method Appl M 2018;331:23–52.
- [683] Gültekin O, Hager SP, Dal H, Holzapfel GA. Computational modeling of progressive damage and rupture in fibrous biological tissues: application to aortic dissection. Biomech Model Mechanobiol 2019;18(6):1607–28.
- [684] Mousavi SJ, Farzaneh S, Avril S. Computational predictions of damage propagation preceding dissection of ascending thoracic aortic aneurysms. Int J Numer Method Biomed Eng 2018;34(4):e2944.
- [685] Rolf-Pissarczyk M, Li K, Fleischmann D, Holzapfel GA. A discrete approach for modeling degraded elastic fibers in aortic dissection. Comput Methods Appl Mech Eng 2021;373:113511.
- [686] Li K, Ogden RW, Holzapfel GA. A discrete fibre dispersion method for excluding fibres under compression in the modelling of fibrous tissues. J R Soc Interface 2018;15(138):20170766.
- [687] Tanishita K, Yamamoto K, editors. Vascular Engineering: New Prospects of Vascular Medicine and Biology with a Multidiscipline Approach. Springer Japan; 2016.
- [688] Enea D, Henson F, Kew S, Wardale J, Getgood A, Brooks R, Rushton N. Extruded collagen fibres for tissue engineering applications: effect of crosslinking method on mechanical and biological properties. J Mater Sci Mater Med 2011;22:1569–78.
- [689] Schriefl AJ, Zeindlinger G, Pierce D, Regitnig P, Holzapfel GA. Determination of the layer-specific distributed collagen fibre orientations in human thoracic and abdominal aortas and common iliac arteries. J R Soc Interface 2012;9(71):1275–86.
- [690] Amabili M, Balasubramanian P, Bozzo I, Breslavsky ID, Ferrari G. Layer-specific hyperelastic and viscoelastic characterization of human descending thoracic aortas. J Mech Behav Biomed Mater 2019;99:27–46.
- [691] Brunet J, Pierrat B, Badel P. A parametric study on factors influencing the onset and propagation of aortic dissection using the extending finite element method. IEEE Trans Biomed Eng 2021;68(10):2918–29.
- [692] Angouras DC, Eleftherios PK, Sokolis DP. Regional distribution of delamination strength in ascending thoracic aortic aneurysms. J Mech Behav Biomed Mater 2019;98:58–70.
- [693] Ban E, Cavinato C, Humphrey JD. Differential propensity of dissection along the aorta. Biomech Model Mechanobiol 2021;20:895-907.
- [694] Vande Geest JP, Sacks MS, Vorp DA. Age dependency of the biaxial biomechanical behavior of human abdominal aorta. J Biomech Eng 2004;126(6):815-22.

- [695] Haskett D, Johnson G, Zhou A, Utzinger U, Vande Geest JP. Microstructural and biomechanical alterations of the human aorta as a function of age and location. Biomech Model Mechanobiol 2010;9(6):725–36.
- [696] Ban E, Cavinato C, Humphrey JD. Critical pressure of intramural delamination in aortic dissection. Ann Biomed Eng 2022;50:183-94.
- [697] Yin M, Ban E, Rego BV, Zhang E, Cavinato C, Humphrey JD, Karniadakis GE. Simulating progressive intramural damage leading to aortic dissection using DeepONet: an operator-regression neural network. J R Soc Interface 2022;19:20210670.
- [698] Lu L, Jin P, Pang G, Zhang Z, Karniadakis GE. Learning nonlinear operators via DeepONet based on the universal approximation theorem of operators. Nat Mach Intell 2021;3:218-29.
- [699] García-Herrera CM, Celentano DJ, Cruchaga MA, Rojo FJ, Atienza JM, Guinea GV, Goicolea JM. Mechanical characterisation of the human thoracic descending aorta: experiments and modelling. Comput Methods Biomech Biomed Eng Imaging Vis 2012;15(2):185–93.
- [700] Roccabianca S, Ateshian GA, Humphrey JD. Biomechanical roles of medial pooling of glycosaminoglycans in thoracic aortic dissection. Biomech Model Mechanobiol 2014;13(1):13–25.
- [701] FitzGibbon B, Fereidoonezhad B, McGarry P. Cohesive zone model analysis, development, and application in mixed-mode arterial dissection. In: Sommer G, Li K, Haspinger DC, Ogden RW, editors. Solid (Bio)mechanics: Challenges of the Next Decade: A Book Dedicated to Professor Gerhard A. Holzapfel: Springer International Publishing, Cham; 2022. p. 3–52.
- [702] Fereidoonnezhad B, O'Connor C, McGarry JP. A new anisotropic soft tissue model for elimination of unphysical auxetic behaviour. J Biomech 2020;111: 110006.
- [703] FitzGibbon B, McGarry P. Development of a test method to investigate mode II fracture and dissection of arteries. Acta Biomater 2021;121:444-60.
- [704] Li B, Roper SM, Wang L, Luo X, Hill NA. An incremental deformation model of arterial dissection. J Math Biol 2019;78(5):1277-98.
- [705] Zhang M, Liu H, Cai Z, Sun C, Sun W. A novel framework for quantifying the subject-specific three-dimensional residual stress field in the aortic wall. J Mech Behav Biomed Mater 2022;125:1049006.
- [706] Holzapfel GA, Sommer G, Auer M, Regitnig P, Ogden RW. Layer-specific 3D residual deformations of human aortas with non-atherosclerotic intimal thickening. Ann Biomed Eng 2007;35(4):530–45.
- [707] Liu H, Zhang M, Liu M, Martin C, Cai Z, Sun W. Finite element simulation of three dimensional residual stress in the aortic wall using an anisotropic tissue growth model. J Mech Behav Biomed Mater 2019;92:188–96.
- [708] Han H, Guo B, Gao P, Yang F, Sun C, Hill NA, Liu H. Finite-element simulation of in-plane tear propagation in the dissected aorta: implications for the propagation mechanism. Int J Numer Meth Biomed Engng 2023;39(9):e3743.
- [709] Gheysen L, Maes L, Caenen A, Segers P, Peirlinck M, Famaey N. Uncertainty quantification of the wall thickness and stiffness in an idealized dissected aorta. J Mech Behav Biomed Mater 2024;151:106370.
- [710] McComb BL, Munden RF, Duan F, Jain AA, Tuite C, Chiles C. Normative reference values of thoracic aortic diameter in American College of Radiology Imaging Network (ACRIN 6654) arm of National Lung Screening Trial. Clin Imaging 2016;40(5):936–43.
- [711] Rylski B, Muñoz C, Beyersdorf F, Siepe M, Reser D, Carrel T, Schoenhoff F, Schlensak C, Lescan M, Eckstein HH, Reutersberg B, Erbel R, Janosi RA, Czerny M. How does the descending aorta geometry change when it dissects? Eur J Cardiothorac Surg 2018;53(4):815–21.
- [712] Yamauchi T, Masai T, Takano H, Shirakawa Y, Toda K, Sawa Y. Osaka Cardiovascular Surgery Research Group (OSCAR), Equations for estimating the predissected diameter of the descending aorta from computed tomographic images at the onset of aortic dissection. J Am Heart Assoc 2018;7:e009196.
- [713] Logghe G, Trachet B, Segers P, De Backer J, Mulorz J, Dueppers P, Vermassen F, Schelzig H, Van Herzeele I, Wagenhäuser MU. Outflow through aortic side branches drives false lumen patency in type B aortic dissection. Front Cardiovasc Med 2021;8:710603.
- [714] Gheysen L, Maes L, Famaey N, Segers P. Growth and remodeling of the dissected membrane in an idealized dissected aorta model. Biomech Model Mechanobiol 2023;23(2):413–31.
- [715] Weisbecker H, Pierce DM, Regitnig P, Holzapfel GA. Layer-specific damage experiments and modeling of human thoracic and abdominal aortas with nonatherosclerotic intimal thickening. J Mech Behav Biomed Mater 2012;12:93–106.
- [716] Famaey N, Vastmans J, Fehervary H, Maes L, Vanderveken E, Rega F, Mousavi SJ, Avril S. Numerical simulation of arterial remodeling in pulmonary autografts. Z Angew Math Mech 2018;98(12):2239–57.
- [717] Ranftl S, Rolf-Pissarczyk M, Wolkerstorfer G, Pepe A, Egger J, von der Linden W, Holzapfel GA. Stochastic modeling of inhomogeneities in the aortic wall and uncertainty quantification using a Bayesian encoder-decoder surrogate. Comput Methods Appl Mech Eng 2022;401:115594.
- [718] Rausch MK, Karniadakis GE, Humphrey JD. Modeling soft tissue damage and failure using a combined particle/continuum approach. Biomech Model Mechanobiol 2017;16(1):249–61.
- [719] Ahmadzadeh H, Rausch MK, Humphrey JD. Particle-based computational modelling of arterial disease. J R Soc Interface 2018;15(149):20180616.
- [720] Bersi MR, Bellini C, Wu J, Montaniel KRC, Harrison DG, Humphrey JD. Excessive adventitial remodeling leads to early aortic maladaptation in angiotensininduced hypertension. Hypertension 2016;67(5):890–6.
- [721] Latorre M, Humphrey JD. Modeling mechano-driven and immuno-mediated aortic maladaptation in hypertension. Biomech Model Mechanobiol 2018;17(5): 1497–511.
- [722] Ahmadzadeh H, Rausch M, Humphrey JD. Modeling lamellar disruption within the aortic wall using a particle-based approach. Sci Rep 2019;9(1):15320.
   [723] Murtada S-I, Lewin S, Arner A, Holzapfel GA. Adaptation of active tone in the mouse descending thoracic aorta under acute changes in loading. Biomech Model Mechanobiol 2016;15:579–92.
- [724] Liu X, Ilseng A, Prot V, Skallerud BH, Holzapfel GA. Swelling of interlamellar GAGs/PGs as an initiation mechanism for aortic dissection: constitutive modeling and numerical simulations. Mech Mater 2022;4:5.
- [725] Iliopoulos DC, Kritharis EP, Giagini AT, Papadodima SA, Sokolis DP. Ascending thoracic aortic aneurysms are associated with compositional remodeling and vessel stiffening but not weakening in age-matched subjects. J Thorac Cardiovasc Surg 2009;137(1):101–9.
- [726] Niestrawska JA, Viertler C, Regitnig P, Cohnert TU, Sommer G, Holzapfel GA. Microstructure and mechanics of healthy and aneurysmatic abdominal aortas: experimental analysis and modelling. J R Soc Interface 2016;13(124):20160620.
- [727] Stracuzzi A, Mazza E, Ehret AE. Chemomechanical models for soft tissues based on the reconciliation of porous media and swelling polymer theories. Z Angew Math Mech 2018;98(12):2135–54.
- [728] Soleimani M, Deo R, Hudobivnik B, Poyanmehr R, Haverich A, Wriggers P. Mathematical modeling and numerical simulation of arteria dissection based on a novel surgeon's view. Biomech Model Mechanobiol 2023;22(6):2097–116.
- [729] Soleimani M, Haverich A, Wriggers P. Mathematical modeling and numerical simulation of atherosclerosis based on a novel surgeon's view. Arch Comput Methods Eng 2021;28:4263–82.
- [730] Biehler J, Gee MW, Wall WA. Towards efficient uncertainty quantification in complex and large-scale biomechanical problems based on a Bayesian multifidelity scheme. Biomech Model Mechanobiol 2015;14(3):489–513.
- [731] Fanni BM, Antonuccio MN, Pizzuto A, Berti S, Santoro G, Celi S. Uncertainty quantification in the in vivo image-based estimation of local elastic properties of vascular walls. J Cardiovasc Dev Dis 2023;10(3):109.
- [732] Cyron CJ, Aydin RC, Humphrey JD. A homogenized constrained mixture (and mechanical analog) model for growth and remodeling of soft tissue. Biomech Model Mechanobiol 2016;15(6):1389–403.
- [733] Humphrey JD, Rajagopal KR. A constrained mixture model for growth and remodeling of soft tissues. Math Mod Meth Appl S 2002;12(03):407-30.
- [734] Zhang S, Laubrie JD, Mousavi SJ, Ben Ahmed S, Avril S. Patient-specific finite element modeling of aneurysmal dilatation after chronic type B aortic dissection. In: Nielsen PMF, Nash MP, Li X, Miller K, Wittek A, editors. Computational Biomechanics for Medicine: Towards Translation and Better Patient Outcomes. Springer; 2022. p. 15–38.
- [735] Braeu FA, Seitz A, Aydin RC, Cyron CJ. Homogenized constrained mixture models for anisotropic volumetric growth and remodeling. Biomech Model Mechanobiol 2017;16(3):889–906.

- [736] Mousavi SJ, Farzaneh S, Avril S. Patient-specific predictions of aneurysm growth and remodeling in the ascending thoracic aorta using the homogenized constrained mixture model. Biomech Model Mechanobiol 2019;18:1895–913.
- [737] Mousavi SJ, Jayendiran R, Farzaneh S, Campisi S, Viallon M, Croisille P, Avril S. Coupling hemodynamics with mechanobiology in patient-specific computational models of ascending thoracic aortic aneurysms. Comput Methods Programs Biomed 2021;205:106107.
- [738] Luo F, Zhou X-L, Li J-J, Hui R-T. Inflammatory response is associated with aortic dissection. Ageing Res Rev 2009;8(1):31-5.
- [739] Wang X, Zhang H, Cao L, He Y, Ma A, Guo W. The role of macrophages in aortic dissection. Front Physiol 2020;11:54.
- [740] Gheysen L, Maes L, Famaey N, Segers P. Pulse wave velocity: a clinical measure to aid material parameter estimation in computational arterial biomechanics. J Biomech 2023;149:111482.
- [741] Yeerella RH, Cai S. Fracture mechanics modeling of aortic dissection. Biomech Model Mechanobiol 2024;23(4):1377-91.
- [742] Poya R, Gil AJ, Ortigosa R. A high performance data parallel tensor contraction framework: application to coupled electro-mechanics. Comput Phys Commun 2017;216:35–52.
- [743] Poya R, Gil AJ, Ortigosa R, Sevilla R, Bonet J, Wall WA. A curvilinear high order finite element framework for electromechanics: from linearised electroelasticity to massively deformable dielectric elastomers. Comput Methods Appl Mech Eng 2018;329:75–117.
- [744] Formaggia L, Quarteroni A, Veneziani A. Cardiovascular Mathematics: Modeling and Simulation of the Circulatory System. 1st Edition. Springer; 2009.
- [745] Quarteroni A, Manzoni A, Vergara C. The cardiovascular system: mathematical modelling, numerical algorithms and clinical applications. Acta Numerica 2017;26:365–590.
- [746] Fumagalli I, Pagani S, Vergara C, Dede L, Adebo DA, Del Greco M, Frontera A, Luciani GB, Pontone G, Scrofani R, Quarteroni A. The role of computational methods in cardiovascular medicine: a narrative review. Transl Pediatr 2024;13(1):146–63.
- [747] Soudah E, Rudenick P, Bordone M, Bijnens B, García-Dorado D, Evangelista A, Oñate E. Validation of numerical flow simulations against in vitro phantom measurements in different type B aortic dissection scenarios. Comput Methods Biomech Biomed Engin 2015;18(8):805–15.
- [748] Rudenick PA, Bijnens BH, Segers P, García-Dorado D, Evangelista A. Assessment of wall elasticity variations on intraluminal haemodynamics in descending aortic dissections using a lumped-parameter model. PLoS One 2015;10(4):e0124011.
- [749] Fan Y, Cheng S, Qing K, Chow K. Endovascular repair of type B aortic dissection: a study by computational fluid dynamics. J Biomed Sci Eng 2010;3(9):900–7.
   [750] Guan J, Chu B, Chang Y, Qiao A. Two-dimensional computational simulation of bypassed aortic dissection with a blind false lumen. In: 22nd International Conference on Biomedical Engineering and Informatics, 2009, p. 1–4. https://doi.org/10.1109/BMEI.2009.5305016.
- [751] Chitsaz S, Azadani AN, Matthews PB, Chuter TA, Tseng EE, Ge L. Hemodynamic determinants of aortic dissection propagation by 2D computational modeling: implications for endovascular stent-grafting. J Cardiovasc Surg (Torino) 2012;53:631–40.
- [752] Tang AYS, Fan Y, Cheng SWK, Chow KW. Biomechanical factors influencing type B thoracic aortic dissection: computational fluid dynamics study. Eng Appl Comput Fluid Mech 2012;6(4):622–32.
- [753] Ben Ahmed S, Dillon-Murphy D, Figueroa CA. Computational study of anatomical risk factors in idealized models of type B aortic dissection. Eur J Vasc Endovasc Surg 2016;52(6):736–45.
- [754] Zorrilla R, Soudah E, Rossi R. Computational modeling of the fluid flow in type B aortic dissection using a modified finite element embedded formulation. Biomech Model Mechanobiol 2020;19:1565–83.
- [755] Peng Y, Zhang X, Li J, Zhang X, He H, Li X, Fang K, Zheng L, Shu C. Enlarged lumen volume of proximal aortic segment and acute type B aortic dissection: a computer fluid dynamics study of ideal aortic models. Int J Gen Med 2022;15:535–43.
- [756] Qiao Y, Mao L, Wang Y, Luan J, Chen Y, Zhu T, Luo K, Fan J. Hemodynamic effects of stent-graft introducer sheath during thoracic endovascular aortic repair. Biomech Model Mechanobiol 2022;21(2):419–31.
- [757] Li D, Zheng T, Liu Z, Li Y, Yuan D, Fan Y. Influence of distal re-entry tears on false lumen thrombosis after thoracic endovascular aortic repair in type B aortic dissection patients: a computational fluid dynamics simulation. Cardiovasc Eng Technol 2021;12(4):426–36.
- [758] Li D, Yuan D, Peng L, Zheng T, Fan Y. The characteristics of distal tears affect false lumen thrombosis rate after thoracic endovascular aortic repair for acute type B dissection. Interdiscip Cardiovasc Thorac Surg 2021;33:755–62.
- [759] Cheng Z, Tan FPP, Riga CV, Bicknell CD, Hamady MS, Gibbs RGJ, Wood NB, Xu XY. Analysis of flow patterns in a patient-specific aortic dissection model. J Biomech Eng 2010;132(5):051007.
- [760] Cheng Z, Juli C, Wood NB, Gibbs RGJ, Xu XY. Predicting flow in aortic dissection: comparison of computational model with PC-MRI velocity measurements. Med Eng Phys 2014;36(9):1176–84.
- [761] Cheng Z, Wood N, Gibbs R, Xu XY. Geometric and flow features of type B aortic dissection: initial findings and comparison of medically treated and stented cases. Ann Biomed Eng 2015;43(1):177–89.
- [762] Djorovic S, Filipovic N, Stojic V, Velicki L. Computational simulation of blood flow in a DeBakey type I aortic dissection. In: IEEE 15th International Conference on Bioinformatics and Bioengineering (BIBE), 2015, p. 1–5. https://doi.org/10.1109/BIBE.2015.7367656.
- [763] Shang EK, Nathan DP, Fairman RM, Bavaria JE, Gorman RC, Gorman JH, Jackson BM. Use of computational fluid dynamics studies in predicting aneurysmal degeneration of acute type B aortic dissections. J Vasc Surg 2015;62(2):279–84.
- [764] Abazari MA, Rafieianzab D, Soltani M, Alimohammadi M. The effect of beta-blockers on hemodynamic parameters in patient-specific blood flow simulations of type-B aortic dissection: a virtual study. Sci Rep 2021;11:16058.
- [765] Alimohammadi M, Agu O, Balabani S, Díaz-Zuccarini V. Development of a patient-specific simulation tool to analyse aortic dissections: assessment of mixed patient-specific flow and pressure boundary conditions. Med Eng Phys 2014;36(3):275–84.
- [766] Alimohammadi M, Bhattacharya-Ghosh B, Seshadhri S, Penrose J, Agu O, Balabani S, Díaz-Zuccarini V. Evaluation of the hemodynamic effectiveness of aortic dissection treatments via virtual stenting. Int J Artif Organs 2014;37(10):753–62.
- [767] Chen D, Müller-Eschner M, von Tengg-Kobligk H, Barber D, Böckler D, Hose R, Ventikos Y. A patient-specific study of type-B aortic dissection: evaluation of true-false lumen blood exchange. Biomed Eng Online 2013;12:65.
- [768] Bonfanti M, Balabani S, Greenwood JP, Puppala S, Homer-Vanniasinkam S, Díaz-Zuccarini V. Computational tools for clinical support: a multi-scale compliant model for haemodynamic simulations in an aortic dissection based on multi-modal imaging data. J R Soc Interface 2017;14(136):20170632.
- [769] Bonfanti M, Balabani S, Alimohammadi M, Agu O, Homer-Vanniasinkam S, Díaz-Zuccarini V. A simplified method to account for wall motion in patientspecific blood flow simulations of aortic dissection: comparison with fluid–structure interaction. Med Eng Phys 2018;S1350–4533(18). 30074-2.
- [770] Bonfanti M, Franzetti G, Maritati G, Homer-Vanniasinkam S, Balabani S, Díaz-Zuccarini V. Patient-specific haemodynamic simulations of complex aortic dissections informed by commonly available clinical datasets. Med Eng Phys 2019;71:45–55.
- [771] Pant S, Fabrèges B, Gerbeau J-F, Vignon-Clementel IE. A methodological paradigm for patient-specific multi-scale CFD simulations: from clinical measurements to parameter estimates for individual analysis. Int J Numer Method Biomed Eng 2014;30(12):1614–48.
- [772] Wang Q, Guo X, Stäb D, Jin N, Poon EKW, Lim RP, Ooi A. Computational fluid dynamic simulations informed by CT and 4D flow MRI for post-surgery aortic dissection – a case study. Int J Heat Fluid Flow 2022;96:108986.
- [773] Armour CH, Guo B, Saitta S, Pirola S, Liu Y, Dong Z, Xu XY. Evaluation and verification of patient-specific modelling of type B aortic dissection. Comput Biol Med 2022;140:105053.
- [774] Xu H, Zhenfeng L, Huiwu D, Yilun Z, Jianyong W, Watton PN, Wei G, Duanduan C, Jiang X. Hemodynamic parameters that may predict false-lumen growth in type-B aortic dissection after endovascular repair: a preliminary study on long-term multiple follow-ups. Med Eng Phys 2017;50:12–21.
- [775] Kimura N, Nakamura M, Takagi R, Mieno MN, Yamaguchi A, Czerny M, Beyersdorf F, Kari FA, Rylski B. False lumen/true lumen wall pressure ratio is increased in acute non-A non-B aortic dissection. Interact Cardiovasc Thorac Surg 2022;35(3):ivac138. https://doi.org/10.1093/icvts/ivac138.
- [776] Moretti S, Tauro F, Orrico M, Mangialardi N, Facci AL. Comparative analysis of patient-specific aortic dissections through computational fluid dynamics suggests increased likelihood of degeneration in partially thrombosed false lumen. Bioengineering 2023;10:316.
- [777] Stokes C, Ahmed D, Lind N, Haupt F, Becker D, Hamilton J, Muthurangu V, von Tengg-Kobligk H, Papadakis G, Balabani S, Díaz-Zuccarini V. Aneurysmal growth in type-B aortic dissection: assessing the impact of patient-specific inlet conditions on key haemodynamic indices. J R Soc Interface 2023;20:20230281.

## M. Rolf-Pissarczyk et al.

- [778] Tomaiuolo G, Carciati A, Caserta A, Guido S. Blood linear viscoelasticity by small amplitude oscillatory flow. Rheol Acta 2016;55:485–95.
- [779] Black SM, Maclean C, Hall Barrientos P, Ritos K, McQueen A, Kazakidi A. Calibration of patient-specific boundary conditions for coupled CFD models of the aorta derived from 4D flow-MRI. Front Bioeng Biotechnol 2023;11:1178483.
- [780] Karmonik C, Bismuth J, Shah DJ, Davies MG, Purdy D, Lumsden AB. Computational study of haemodynamic effects of entry- and exit-tear coverage in a DeBakey type III aortic dissection: technical report. Eur J Vasc Endovasc Surg 2011;42(2):172–7.
- [781] Chen D, Müller-Eschner M, Kotelis D, Böckler D, Ventikos Y, Von Tengg-Kobligk H. A longitudinal study of type-B aortic dissection and endovascular repair scenarios: computational analyses. Med Eng Phys 2013;35(9):1321–30.
- [782] Wan Ab Naim WN, Ganesan P, Sun Z, Osman K, Einly L. The impact of the number of tears in patient-specific Stanford type B aortic dissection aneurysm: CFD simulation. J Mech Med Biol 2014;14(2):20.
- [783] Dillon-Murphy D, Noorani A, Nordsletten D, Figueroa CA. Multi-modality image-based computational analysis of haemodynamics in aortic dissection. Biomech Model Mechanobiol 2016;15:857–76.
- [784] Shi Y, Minjia Z, Chang Y, Qiao H, Liu Y. The risk of Stanford type-A aortic dissection with different tear size and location: a numerical study. Biomed Eng Online 2016;15(Suppl 2):128.
- [785] Yu SCH, Liu W, Wong RH, Underwood M, Wang D. The potential of computational fluid dynamics simulation on serial monitoring of hemodynamic change in type B aortic dissection. Cardiovasc Intervent Radiol 2016;39(8):1090–8.
- [786] Dai Y, Luo G, Dai X, Liu H. Hemodynamic effect of multiple overlapping uncovered stents on aortic dissection: surgical strategies and implications for false lumen thrombosis. Cardiovasc Eng Technol 2020;11(1):24–35.
- [787] Xiong Z, Yang P, Li D, Qiu Y, Zheng T, Hu J. A computational fluid dynamics analysis of a patient with acute non-A-non-B aortic dissection after type I hybrid arch repair. Med Eng Phys 2020;77:43–52.
- [788] Li Z, Xu H, Armour CH, Guo Y, Xiong J, Xu XY, Chen D. The necessity to seal the re-entry tears of aortic dissection after TEVAR: a hemodynamic indicator. Front Bioeng Biotechnol 2022;10:831903.
- [789] Li D, Wang J, Zeng W, Zeng X, Liu Z, Cao H, Yuan D, Zheng T. The loss of helical flow in the thoracic aorta might be an identifying marker for the risk of acute type B aortic dissection. Comput Methods Programs Biomed 2023;230:107331.
- [790] Wen J, Huang H, Su Z, Jiang L, Gao Q, Chen X, Yan T, Peng L. Predicting the risk of type B aortic dissection using hemodynamic parameters in aortic arches: a comparative study between healthy and repaired aortas. Comput Methods Programs Biomed 2023;230:107326.
- [791] Mohamied Y, Sherwin SJ, Weinberg PD. Understanding the fluid mechanics behind transverse wall shear stress. J Biomech 2017;50:102-9.
- [792] Mazzi V, Gallo D, Calò K, Najafi M, Khan MO, De Nisco G, Steinman DA, Morbiducci U. A Eulerian method to analyze wall shear stress fixed points and manifolds in cardiovascular flows. Biomech Model Mechanobiol 2020;19(5):1403–23.
- [793] Mazzi V, Morbiducci U, Calò K, De Nisco G, Rizzini ML, Torta E, Caridi GCA, Chiastra C, Gallo D. Wall shear stress topological skeleton analysis in cardiovascular flows: methods and applications. Mathematics 2021;9(7):720.
- [794] Tasso P, Raptis A, Matsagkas M, Rizzini ML, Gallo D, Xenos M, Morbiducci U. Abdominal aortic aneurysm endovascular repair: profiling postimplantation morphometry and hemodynamics with image-based computational fluid dynamics. J Biomech Eng 2018;140(11):111003.
- [795] Prashantha B, Anish S. Computational investigations on the hemodynamic performance of a new swirl generator in bifurcated arteries. Comput Methods Biomech Biomed Engin 2019;22(4):364–75.
- [796] Ku DN, Giddens DP, Zarins CK, Glagov S. Pulsatile flow and atherosclerosis in the human carotid bifurcation. Positive correlation between plaque location and low oscillating shear stress, Arteriosclerosis 1985;5(3):293–302.
- [797] Karmonik C, Bismuth J, Davies MG, Shah DJ, Younes HK, Lumsden AB. A computational fluid dynamics study pre- and post-stent graft placement in an acute type B aortic dissection. Vasc Endovascular Surg 2011;45(2):157–64.
- [798] Karmonik C, Partovi S, Müller-Eschner M, Bismuth J, Davies M, Shah DJ, Loebe M, Böckler D, Lumsden AB, von Tengg-Kobligk H. Longitudinal computational fluid dynamics study of aneurysmal dilatation in a chronic DeBakey type III aortic dissection. J Vasc Surg 2012;56(1). 260–3.e1.
- [799] Cheng Z, Riga C, Chan J, Hamady M, Wood NB, Cheshire NJW, Xu XY, Gibbs RGJ. Initial findings and potential applicability of computational simulation of the aorta in acute type B dissection. J Vasc Surg 2013;57(2 Suppl):35S-43S.
- [800] Costache VS, Yeung KK, Solomon C, Popa R, Melnic T, Sandu M, Bucurenciu C, Candea G, Santa A, Costache A. Aortic remodeling after total endovascular aortic repair with multilayer stents: computational fluid dynamics analysis of aortic remodeling over 3 years of follow-up. J Endovasc Ther 2018;25(6):760–4.
- [801] Costache VS, Meekel JP, Costache A, Melnic T, Solomon C, Chitic AM, Bucurenciu C, Moldovan H, Antoniac I, Candea G, Yeung KK. Geometric analysis of type B aortic dissections shows aortic remodeling after intervention using multilayer stents. Materials (Basel) 2020;13(10):2274.
- [802] Costache V, Meekel JP, Costache A, Melnic T, Bucurenciu C, Chitic A, Candea G, Solomon C, Yeung KK. One-year single-center results of the multilayer flow modulator stents for the treatment of type B aortic dissection. J Endovasc Ther 2021;2021(28):20–31.
- [803] Pirola S, Guo B, Menichini C, Saitta S, Fu W, Dong Z, Xu XY. 4D flow MRI-based computational analysis of blood flow in patient-specific aortic dissection. IEEE Trans Biomed Eng 2019;66(12):3411–9.
- [804] Xu H, Xiong J, Han X, Mei Y, Shi Y, Wang D, Zhang M, Chen D. Computed tomography-based hemodynamic index for aortic dissection. J Thorac Cardiovasc Surg 2021;162(2):e165–76.
- [805] Armour CH, Guo B, Pirola S, Saitta S, Liu Y, Dong Z, Xu XY. The influence of inlet velocity profile on predicted flow in type B aortic dissection. Biomech Model Mechanobiol 2021;20. 481–480.
- [806] Zhu Y, Mirsadraee S, Asimakopoulos G, Gambaro A, Rosendahl U, Pepper J, Xu XY. Association of hemodynamic factors and progressive aortic dilatation following type A aortic dissection surgical repair. Sci Rep 2021;11:11521.
- [807] Zhu Y, Xu XY, Rosendahl U, Pepper J, Mirsadraee S. Prediction of aortic dilatation in surgically repaired type A dissection: a longitudinal study using computational fluid dynamics. JTCVS Open 2022;9:11–27.
- [808] Parker LP, Reutersberg B, Syed MBJ, Munshi B, Richards S, Kelsey LJ, Sakalihasan N, Eckstein HH, Norman PE, Doyle BJ. Proximal false lumen thrombosis is associated with low false lumen pressure and fewer complications in type B aortic dissection. J Vasc Surg 2022;75(4):1181–90.
- [809] Armour C, Guo B, Saitta S, Guo D, Liu Y, Fu W, Dong Z, Xu XY. The role of multiple re-entry tears in type B aortic dissection progression: a longitudinal study using a controlled swine model. J Endovasc Ther 2022. 15266028221111295.
- [810] Fatma K, Carine G-C, Marine G, Philippe P, Deplano V. Numerical modeling of residual type B aortic dissection: longitudinal analysis of favorable and unfavorable evolution. Med Biol Eng Comput 2022;60:769–83.
- [811] Osswald A, Weymann A, Tsagakis K, Zubarevich A, Thielmann M, Schmack B, Ruhparwar A, Karmonik C. First insights into the role of wall shear stress in the development of a distal stent graft induced new entry through computational fluid dynamics simulations. J Thorac Dis 2023;15(2):281–90.
- [812] Karmonik C, Bismuth JX, Davies MG, Lumsden AB. Computational hemodynamics in the human aorta: a computational fluid dynamics study of three cases with patient-specific geometries and inflow rates. Technol Health Care 2008;16(5):343–54.
- [813] D'Ancona G, Lee JJ, Pasta S, Pilato G, Rinaudo A, Follis F, Pilato M. Computational analysis to predict false-lumen perfusion and outcome of type B aortic dissection. J Thorac Cardiovasc Surg 2014;148(4). P1756–1748.
- [814] Wan Ab Naim WN, Ganesan PB, Sun Z, Liew YM, Qian Y, Lee CJ, Jansen S, Hashim SA, Lim E. Prediction of thrombus formation using vortical structures presentation in Stanford type B aortic dissection: a preliminary study using CFD approach. Appl Math Model 2016;40(4):3115–27.
- [815] Osswald A, Karmonik C, Anderson JR, Rengier F, Karck M, Engelke J, Kallenbach K, Kotelis D, Partovi S, Böckler D, Ruhparwar A. Elevated wall shear stress in aortic type B dissection may relate to retrograde aortic type A dissection: a computational fluid dynamics pilot study. Eur J Vasc Endovasc Surg 2017;54(3): 324–30.
- [816] Long Ko JK, Liu RW, Ma D, Shi L, Ho Yu SC, Wang D. Pulsatile hemodynamics in patient-specific thoracic aortic dissection models constructed from computed tomography angiography. J Xray Sci Technol 2017;25:233–45.
- [817] Iida Y, Yoshitake A, Shimizu H. Blood flow into ulcer-like projection of a type B aortic dissection visualized with computational fluid dynamics. J Thorac Cardiovasc Surg 2017;154(4):1217–8.

## M. Rolf-Pissarczyk et al.

- [818] Chi Q, He Y, Luan Y, Qin K, Mu L. Numerical analysis of wall shear stress in ascending aorta before tearing in type A aortic dissection. Comput Biol Med 2017; 89:236–47.
- [819] Peng L, Qiu Y, Yang Z, Yuan D, Dai C, Li D, Jiang Y, Zheng T. Patient-specific computational hemodynamic analysis for interrupted aortic arch in an adult: implications for aortic dissection initiation. Sci Rep 2019;9(1):8600.
- [820] Tomasi J, Le Bars F, Shao C, Lucas A, Lederlin M, Haigron P, Verhoye JP. Patient-specific and real-time model of numerical simulation of the hemodynamics of type B aortic dissections. Med Hypotheses 2020;135:109477.
- [821] Li X, Qiao H, Shi Y, Xue J, Bai T, Liu Y, Sun L. Role of proximal and distal tear size ratio in hemodynamic change of acute type A aortic dissection. J Thorac Dis 2020;12(6):3200–10.
- [822] Qiao Y, Mao L, Ding Y, Fan J, Zhu T, Luo K. Hemodynamic consequences of TEVAR with in situ double fenestrations of left carotid artery and left subclavian artery. Med Eng Phys 2020;76:32–9.
- [823] Marrocco-Trischitta MM, Sturla F. Blood flow helical pattern in type III arch configuration as a potential risk factor for type B aortic dissection. Eur J Cardiothorac Surg 2021;61(1):132–9.
- [824] Li Z, Liang S, Xu H, Zhu M, Mei Y, Xiong J, Chen D. Flow analysis of aortic dissection: comparison of inflow boundary conditions for computational models based on 4D PCMRI and Doppler ultrasound. Comput Methods Biomech Biomech Biomed Engin 2021;24(11):1251–62.
- [825] Hohri Y, Numata S, Itatani K, Kanda K, Yamazaki S, Inoue T, Yaku H. Prediction for future occurrence of type A aortic dissection using computational fluid dynamics. Eur J Cardiothorac Surg 2021;60(2):384–91.
- [826] Wen J, Yan T, Su Z, Huang H, Gao Q, Chen X, Wong KKL, Peng L. Risk evaluation of type B aortic dissection based on WSS-based indicators distribution in different types of aortic arch. Comput Methods Programs Biomed 2022;221:106872.
- [827] Takeda R, Sato F, Yokoyama H, Sasaki K, Oshima N, Kuroda A, Takashima H, Li C, Honda S, Kamiya H. Investigations into the potential of using open source CFD to analyze the differences in hemodynamic parameters for aortic dissections [healthy versus Stanford type A and B]. Ann Vasc Surg 2022;79:310–23.
   [828] Tsai T-Y, Guo X-J, Kageyama S, Lim RP, Tanaka K, De Mey J, La Meir M, Onuma Y, Poon EKW, Serruys PW. Managing iatrogenic aortic dissection: insight from
- 3D-holographic imaging and CT computational fluid dynamic simulations. J Am Coll Cardiol 2024;S0735–1097(24). 07129–8.
- [829] Messou JCE, Yeung K, Sudbrook E, Zhang J, Toursavadkohi S, Ucuzian AA, Tubaldi E. Investigating the role of thrombosis, fenestration, and false lumen orbital orientation in the hemodynamics of type B aortic dissection. Sci Rep 2024.
- [830] Wang K, Armour CH, Guo B, Dong Z, Xu XY. A new method for scaling inlet flow waveform in hemodynamic analysis of aortic dissection, Int J Numer Method. Biomed Eng 2024:e3855.
- [831] Zorrilla R, Soudah E. An efficient procedure for the blood flow computer simulation of patient-specific aortic dissections. Comput Biol Med 2024;179:108832.
  [832] Wang K, Armour CH, Guo B, Dong Z, Xu XY. A new method for scaling inlet flow waveform in hemodynamic analysis of aortic dissection. Int J Numer Method.
- Biomed Eng 2024;40(9):e3855. https://doi.org/10.1002/cnm.3855.
  [833] Rikhtegar Nezami F, Athanasiou LS, Amrute JM, Edelman ER. Multilayer flow modulator enhances vital organ perfusion in patients with type B aortic dissection. Am J Physiol Heart Circ Physiol 2018;315(5):H1182–93.
- [834] Jiang Y, Qiu Y, Li D, Yuan D, Zheng T, Peng L. Influence of aortic branch arteries on the hemodynamics of patient-speci?c type B aortic dissection following TEVAB. Med Nov Technol 2019;4:100028.
- [835] Qiao Y, Fan J, Ding Y, Luo PK, Zhu T. A primary computational fluid dynamics study of pre- and post-TEVAR with intentional left subclavian artery coverage in a type B aortic dissection. J Biomech Eng 2019;141(11):111002.
- [836] Qiu Y, Dong S, Liu Z, Yuan D, Zheng T. Effect of geometric accuracy at the proximal landing zone on simulation results for thoracic endovascular repair patients. Cardiovasc Eng Technol 2020;11(6):679–88.
- [837] Polanczyk A, Piechota-Polanczyk A, Huk I, Neumayer C, Balcer J, Strzelecki M. Computational fluid dynamic technique for assessment of how changing character of blood flow and different value of hct influence blood hemodynamic in dissected aorta. Diagnostics (Basel) 2021;11(10):1866.
- [838] Sengupta S, Hamady M, Xu XY. Haemodynamic analysis of branched endografts for complex aortic arch repair. Bioengineering (Basel) 2022;9(2):45.
- [839] Li D, Zeng X, Wang J, Yuan D, Zheng T. Effects of different bypass surgeries on LSA revascularization in patients with left subclavian occlusion. Int J Numer Meth Biomed Engng 2022;38:e3636.
- [840] Stokes C, Haupt F, Becker D, Muthurangu V, von Tengg-Kobligk H, Balabani S, Díaz-Zuccarini V. The influence of minor aortic branches in patient-specific flow simulations of type-B aortic dissection. Ann Biomed Eng 2023;51(7). s10439–023–03175–4.
- [841] Motoki K, Zhu Y, Mirsadraee S, Rosendahl U, Pepper J, Xu XY. A computational study of the effects of size, location, and number of tears on haemodynamics in surgically repaired type A aortic dissection. Front Cardiovasc Med 2023;10:1215720.
- [842] Zhu Y, Xu XY, Rosendahl U, Pepper J, Mirsadraee S. A simplified computational workflow for evaluation of aortic hemodynamics after frozen elephant trunk intervention in type A aortic dissection. Int J Appl Mech 2023;5(15):2350033.
- [843] Chen H, Su T, Wang Q, Zheng Z, Li H, Li J. Comparison of thrombosis risk in an abdominal aortic dissection aneurysm with a double false lumen using computational fluid dynamic simulation method. Technol Health Care 2023;31(3):1003–15.
- [844] Liu H, Zhao G, Zhang G-E, Xiong F, Hu S, Ouyang Y, Xiong F. Three-dimensional modelling and hemodynamic simulation of the closure of multiple entry tears in type B aortic dissection. Med Phys 2024;51(1):42–53.
- [845] Girardin L, Stokes C, Thet MS, Oo AY, Balabani S, Díaz-Zuccarini V. Patient-specific haemodynamic analysis of virtual grafting strategies in type-B aortic dissection: impact of complicated mismatch. Cardiovasc Eng Technol 2024;15:290–304. https://doi.org/10.1007/s13239-024-00713-6.
- [846] Tse KM, Chiu P, Lee HP, Ho P. Investigation of hemodynamics in the development of dissecting aneurysm within patient-specific dissecting aneurismal aortas using computational fluid dynamics (CFD) simulations. J Biomech 2011;44:827–36.
- [847] Mori F, Ohtake H, Watanabe G, Matsuzawa T. Numerical simulation in ulcer-like projection due to type B aortic dissection with complete thrombosis type. In: Amalinei C, editor. Aortic Aneurysm – Recent Advances. IntechOpen; 2013. p. 213–29.
- [848] Rinaudo A, D'Ancona G, Lee JJ, Pilato G, Amaducci A, Baglini R, Follis F, Pilato M, Pasta S. Predicting outcome of aortic dissection with patent false lumen by computational flow analysis. Cardiovasc Eng Technol 2014;5(2):176–88.
- [849] Stefanov F, Sultan S, Morris L, Elhelali A, Kavanagh EP, Lundon V, Sultan M, Hynes N. Computational fluid analysis of symptomatic chronic type B aortic dissections managed with the streamliner multilayer flow modulator. J Vasc Surg 2017;65(4):951–63.
- [850] Xu H, Piccinelli M, Leshnower BG, Lefieux A, Taylor WR, Veneziani A. Coupled morphological-hemodynamic computational analysis of type B aortic dissection: a longitudinal study. Ann Biomed Eng 2018;46(7):927–39.
- [851] Wan Ab Naim WN, Ganesan PB, Sun Z, Lei J, Jansen S, Hashim SA, Ho TK, Lim E. Flow pattern analysis in type B aortic dissection patients after stent-grafting repair: comparison between complete and incomplete false lumen thrombosis, Int J Numer Method. Biomed Eng 2018;34(5):e2961.
- [852] Athanasiou LS, Rikhtegar FR, Edelman ER. Hemodynamic consequences of a multilayer flow modulator in aortic dissection. Med Biol Eng Comput 2019;57(9): 1861–74.
- [853] Wang Z, Tian S, Yu L, Ma X, Xing Y, Ma X. The role of the aortic area in type A aortic dissection. Biomed Signal Process Control 2021;68:102642.
- [854] Shad R, Kong S, Fong R, Quach N, Kasinpila P, Bowles C, Lee A, Hiesinger W. Computational fluid dynamics simulations to predict false lumen enlargement after surgical repair of type-A aortic dissection. Semin Thorac Cardiovasc Surg 2022;34(2):443–8.
- [855] He X, Peng L, Cheng C, Zhu F, Huang W, Yu H. To explore the effect of preoperative hemodynamic factors on the outcome of pseudolumen after Stanford type B aortic dissection TEVAR based on computer fluid dynamics. Heart Surg Forum 2022;25(4):E483–8.
- [856] Liu D, Wang X, Zhao D, Sun Z, Biekan J, Wen Z, Xu L, Liu J. Influence of MRI-based boundary conditions on type B aortic dissection simulations in false lumen with or without abdominal aorta involvement. Front Physiol 2022;13:977275.
- [857] Zhang X, Chen D, Wu M, Dong H, Wan Z, Jia H, Liang S, Shao J, Zheng J, Xu S, Xiong J, Guo W. Functional evaluation of embedded modular single-branched stent graft: application to type B aortic dissection with aberrant right subclavian artery. Front Cardiovasc Med 2022;9:869505.
- [858] Wang C, Zou R, Lv L, Wang G, Fan X. Computational fluid dynamics-assisted analysis for residual aortic dissection after thoracic endovascular aortic repair, Eur Heart J Cardiovasc. Imaging 2023; jead211.

- [859] Yoon WJ, Mani K, Wanhainen A. Computational fluid dynamics analysis of a new dissection specific stent graft with the aim to prevent distal stent graft induced new entry (dSINE). Eur J Vasc Endovasc Surg 2023;66(1):146–7.
- [860] Pan W, Kulisa P, Bou-Saïd B, El Hajem M, Simoëns S, Sigovan M. A proposal of risk indicators for pathological development from hemodynamic simulation: application to aortic dissection. J Cardiol Cardiovasc Med 2023;8:029–38.
- [861] Baumler K, Phillips EH, Gutferrez NG, Fleischmann D, Marsden AL, Goergen CJ. Longitudinal investigation of aortic dissection in mice with computational fluid dynamics. Comput Methods Biomech Biomed Engin 2023:1–14.
- [862] Wang K, Armour CH, Ma T, Dong Z, Xu XY. Hemodynamic parameters impact the stability of distal stent graft-induced new entry. Sci Rep 2023;13(1):12123.[863] Jiang X, Xiang G, Du G, Li X, Wu P, Du X. A hemodynamic analysis of fenestrated physician-modified endograft repair for complicated aortic dissections
- involving the visceral arteries. Comput Methods Programs Biomed 2023;242:107785.
   [864] Ritter JC, Munshi B, Letizia N, Parker LP, Kelsey LJ, Gilfillan M, Vo UG, Doyle BJ. Haemodynamic implications of STABILISE technique for aortic dissection repair. Ann Vasc Surg 2024;98:155–63.
- [865] Kimura N, Imada S, Hori D, Nakamura M. Thoracic endovascular aortic repair for acute aortic dissection complicated by mesenteric malperfusion: an evaluation by computational fluid dynamics. Interdiscip Cardiovasc Thorac Surg 2024;38(3):ivae047.
- [866] Forneris A, Hassanabad AF, Appoo JJ, Di Martino ES. Predicting aneurysmal degeneration in uncomplicated residual type B aortic dissection. Bioeng 2024;11 (7):690.
- [867] Mei J, Yuan Y, Yan H, Zhao X, Xue T, Su H, Jia Z. Factors associated with false lumen changes in patients with superior mesenteric artery dissection. Vasc Med 2024;29(3):274–85.
- [868] Alimohammadi M, Sherwood J, Karimpour M, Agu O, Balabani S, Díaz-Zuccarini V. Aortic dissection simulation models for clinical support: fluid-structure interaction vs. rigid wall models. Biomed Eng Online 2015;14:34.
- [869] Qiao Y, Zeng Y, Ding Y, Fan J, Luo K, Zhu T. Numerical simulation of two-phase non-Newtonian blood flow with fluid-structure interaction in aortic dissection. Comput Methods Biomech Biomed Engin 2019;22(6):620–30.
- [870] Taylor C, Figueroa CA. Patient-specific modeling of cardiovascular mechanics. Annu Rev Biomed Eng 2009;11(1):109–34.
- [871] Taylor CA, Steinman DA. Image-based modeling of blood flow and vessel wall dynamics: applications, methods and future directions. Ann Biomed Eng 2010;38 (3):1188–203.
- [872] Qiao A, Yin W, Chu B. Numerical simulation of fluid-structure interaction in bypassed DeBakey III aortic dissection. Comput Methods Biomech Biomed Engin 2015;18(11):1173–80.
- [873] Chen HY, Peelukhana S, Berwick Z, Kratzberg J, Krieger JF, Roeder B, Chambers S, Kassab GS. Fluid-structure interaction simulations of aortic dissection with bench validation. Eur J Vasc Endovasc Surg 2016;52(5):589–95.
- [874] Jayendiran R, Nour BM, Ruimi A. Dacron graft as replacement to dissected aorta: a three-dimensional fluid-structure-interaction analysis. J Mech Behav Biomed Mater 2018;78:329-41.
- [875] Ryzhakov P, Soudah E, Dialami N. Computational modeling of the fluid flow and the flexible intimal flap in type B aortic dissection via a monolithic arbitrary Lagrangian/Eulerian fluid-structure interaction model. Int J Numer Method Biomed Eng 2019;35(11):e3239.
- [876] Keramati H, Birgersson E, Ho JP, Kim S, Chua KJ, Leo HL. The effect of the entry and re-entry size in the aortic dissection: a two-way fluid-structure interaction simulation. Biomech Model Mechanobiol 2020;19(6):2643–56.
- [877] Keramati H, Birgersson E, Kim S, Ho JP, Leo HL. Using a reduced-order model to investigate the effect of the heart rate on the aortic dissection. Int J Numer Meth Biomed Engng 2022;38:e3596.
- [878] Keramati H, Birgersson E, Kim S, Leo HL. A Monte Carlo sensitivity analysis for a dimensionally reduced-order model of the aortic dissection. Cardiovasc Eng Tech 2024;15(3):333–45.
- [879] Saveljic IB, Filipovic ND. Parametric modeling of blood flow and wall interaction in aortic dissection. In: Filipovic ND, editor. Computational Modeling and Simulation: Examples in Bioengineering. Hoboken, New Jersey: John Wiley & Sons Inc; 2021. p. 175–218.
- [880] Qiao Y, Luo K, Fan J. Component quantification of aortic blood flow energy loss using computational fluid-structure interaction hemodynamics. Comput Methods Programs Biomed 2022;221:106826.
- [881] Aghilinejad A, Wei H, Magee GA, Pahlevan NM. Model-based fluid-structure interaction approach for evaluation of thoracic endovascular aortic repair endograft length in type B aortic dissection. Front Bioeng Biotechnol 2022;10:825015.
- [882] Kim T, van Bakel PAJ, Nama N, Burris N, Patel HJ, Williams DM, Figueroa CA. A computational study of dynamic obstruction in type B aortic dissection. J Biomech Eng 2023;145(5):031008.
- [883] Lee G-H, Heo W, Kim T-H, Huh H, Song SW, Ha H. Fluid-structure interaction simulation of visceral perfusion and impact of different cannulation methods on aortic dissection. Sci Rep 2023;13:1116.
- [884] Chong MY, Gu B, Chan BT, Ong ZC, Xu XY, Lim E. Effect of intimal flap motion on flow in acute type B aortic dissection by using fluid-structure interaction. Int J Numer Method Biomed Eng 2020;36(12):e3399.
- [885] Chong MY, Gu B, Armour CH, Dokos S, Ong ZC, Xu XY, Lim E. An integrated fluid-structure interaction and thrombosis model for type B aortic dissection. Biomech Model Mechanobiol 2022;21:261–75.
- [886] Raghavan ML, Vorp DA. Toward a biomechanical tool to evaluate rupture potential of abdominal aortic aneurysm: identification of a finite strain constitutive model and evaluation of its applicability. J Biomech 2000;33(4):475–82.
- [887] Vardakis JC, Bonfanti M, Franzetti G, Guo L, Lassila T, Mitolo M, Hoz de Vila M, Greenwood JP, Maritati G, Chou D, Taylor ZA, Venneri A, Homer-Vanniasinkam S, Balabani S, Frangi AF, Ventikos Y, Díaz-Zuccarini V. Highly integrated workflows for exploring cardiovascular conditions: exemplars of precision medicine in Alzheimer's disease and aortic dissection. Morphologie 2019;103(343):148–60.
- [888] Khannous F, Guivier-Curien C, Gaudry M, Piquet P, Deplano V. Residual type B aortic dissection FSI modeling. Comput Methods Biomech Biomed Engin 2020; 23(sup1):S153–5.
- [889] Moireau P, Xiao N, Astorino M, Figueroa CA, Chapelle D, Taylor CA, Gerbeau J-F. External tissue support and fluid-structure simulation in blood flows. Biomech Model Mechanobiol 2012;11(1–2):1–18.
- [890] Hsu M-C, Bazilevs Y. Blood vessel tissue prestress modeling for vascular fluid-structure interaction simulation. Finite Elem Anal Des 2011;47(6):593-9.
- [891] Schussnig R, Pacheco DRQ, Fries TP. Efficient split-step schemes for fluid-structure interaction involving incompressible generalised Newtonian flows. Comput Struct 2022;260:106718.
- [892] Schussnig R, Pacheco DRQ, Kaltenbacher M, Fries TP. Semi-implicit fluid-structure interaction in biomedical applications. Comput Methods Appl Mech Eng 2022;400:115489.
- [893] Pacheco DRQ, Schussnig R, Fries T-P. An efficient split-step framework for non-Newtonian incompressible flow problems with consistent pressure boundary conditions. Comput Methods Appl Mech Eng 2021;382:113888.
- [894] Schussnig R. Generalised Newtonian Fluids in Cardiovascular Fluid-Structure Interaction, Ph.D. thesis, Monographic Series TU Graz in Structural Analysis (2023).
- [895] Holzapfel GA, Niestrawska JA, Ogden RW, Reinisch AR, Schriefl AJ. Modelling non-symmetric collagen fibre dispersion in arterial walls. J R Soc Interface 2015;12(106):1–14.
- [896] Schussnig R, Rolf-Pissarczyk M, Holzapfel GA, Fries T-P. Fluid-structure interaction simulations of aortic dissection, in: Proc Appl Math Mech, Vol. 20, 2020, p. e202000125. https://doi.org/10.1002/pamm.202000125.
- [897] Schussnig R, Bäumler K, Fries T-P. Multi-layered tissue models in patient-specific simulations of aortic dissection, in: Proc Appl Math Mech, Vol. 21, 2021, p. e202100090. https://doi.org/10.1002/pamm.202100090.
- [898] Fehn N. Robust and Efficient Discontinuous Galerkin Methods for Incompressible Flows. Technische Universität München; 2021. Ph.D. thesis.
- [899] Schussnig R, Rolf-Pissarczyk M, Bäumler K, Fries T-P, Holzapfel GA, Kronbichler M. On the role of tissue mechanics in fluid-structure interaction simulation of patient-specific aortic dissection. Int J Numer Methods Eng 2024;125(14):e7478.

- [900] Zhu Y, Mirsadraee S, Rosendahl U, Pepper J, Xu XY. Fluid-structure interaction simulations of repaired type A aortic dissection: a comprehensive comparison with rigid wall models. Front Physiol 2022;13:913457.
- [901] Radtke L, Larena-Avellaneda A, Debus ES, Düster A. Convergence acceleration for partitioned simulations of the fluid-structure interaction in arteries. Comput Mech 2016;57:901–20.
- [902] Wang X, Ghayesh MH, Kotousov A, Zander AC, Dawson JA, Psaltis PJ. Fluid-structure interaction study for biomechanics and risk factors in Stanford type A aortic dissection. Int J Numer Method Biomed Eng 2023;39(8):e3736.
- [903] Qiao Y, Luo K, Fan J. Heat transfer mechanism in idealized healthy and diseased aortas using fluid-structure interaction method. Biomech Model Mechanobiol 2023;22(6):1953–64.
- [904] Malvindi PG, Pasta S, Raffa GM, Livesey S. Computational fluid dynamics of the ascending aorta before the onset of type A aortic dissection. Eur J Cardiothorac Surg 2017;51(3):597–9.
- [905] Deng L, Qin H, Guan Z, Mu Q, Xia Q, Wang M, Huan W-H, Gu K. Computational numerical analysis of different cannulation methods during cardiopulmonary bypass of type A aortic dissection model based on computational fluid dynamics. Ann Transl Med 2021;9(8):667.
- [906] Qiao Y, Luan J, Mao L, Fan J, Zhu T, Luo K. Biomechanical mechanism of distal stent-graft-induced new entry deterioration after thoracic endovascular aortic repair. Phys Fluids 2022;34:101902.
- [907] Luan J, Qiao Y, Mao L, Fan J, Zhu T, Luo K. The role of aorta distal to stent in the occurrence of distal stent graft-induced new entry tear: a computational fluid dynamics and morphological study. Comput Biol Med 2023;166:107554.
- [908] Deplano V, Guivier-Curien C. Fluid-structure interaction in aortic dissections. In: Gasser TC, Avril S, Elefteriades JA, editors. Biomechanics of the Aorta: Modeling for Patient Care. Biomechanics of Living Organs: Academic Press; 2024. p. 547–68.
- [909] Girardin L, Stokes C, Thet MS, Oo AY, Balabani S, Díaz-Zuccarini V. Patient-specific haemodynamic analysis of virtual grafting strategies in type-B aortic dissection: impact of complicated mismatch. Cardiovasc Eng Technol 2024;15(3):290–304.
- [910] Beller CJ, Labrosse MR, Thubrikar MJ, Robicsek F. Role of aortic root motion in the pathogenesis of aortic dissection. Circulation 2004;109:763-9.
- [911] Doyle BJ, Hoskins PR, Miller K, Newby DE, Dweck MR. Biomechanical analyses of the thoracic aorta: could wall stress and 3D geometry help identify patients at risk of acute aortic dissection? Computational & Mathematical Biomedical Engineering (CMBE) 2013;1:148–51.
- [912] Plonek T, Zak M, Rylski B, Berezowski M, Czerny M, Beyersdorf F, Jasinski M, Filipiak J. Wall stress correlates with intimal entry tear localization in type A aortic dissection. Interact Cardiovasc Thorac Surg 2018;27(6):797–801.
- [913] Plonek T, Zak M, Burzynska K, Rylski B, Gozdzik A, Kustrzycki W, Beyersdorf F, Jasinski M, Filipiak J. The combined impact of mechanical factors on the wall stress of the human ascending aorta a finite elements study. BMC Cardiovasc Disord 2017;17(1):297.
- [914] Menichini C, Pirola S, Guo B, Fu W, Dong Z, Xu XY. High wall stress may predict the formation of stent-graft-induced new entries after thoracic endovascular aortic repair. J Endovasc Ther 2018;25(5):571–7.
- [915] Subramaniam DR, Gutmark E, Andersen N, Nielsen D, Mortensen K, Gravholt C, Backeljauw P, Gutmark-Little I. Influence of material model and aortic root motion in finite element analysis of two exemplary cases of proximal aortic dissection. J Biomech Eng 2021;143(1):014504.
- [916] Hu H, Liu Z, Chen G, Yuan D, Zheng T. Analysis of aortic wall stress and morphology in patients with type B aortic dissection. Med Nov Technol Devices 2021; 11:100081.
- [917] Pasta S, Cho J-S, Dur O, Pekkan K, Vorp DA. Computer modeling for the prediction of thoracic aortic stent graft collapse. J Vasc Surg 2013;57(5):1353-61.
- [918] Prasad A, Xiao N, Gong X-Y, Zarins CK, Figueroa CA. A computational framework for investigating the positional stability of aortic endografts. Biomech Model Mechanobiol 2013;12:869–87.
- [919] Derycke L, Perrin D, Cochennec F, Albertini JN, Avril S. Predictive numerical simulations of double branch stent-graft deployment in an aortic arch aneurysm. Ann Biomed Eng 2019;47(4):1051–62.
- [920] Ma T, Dong ZH, Wang S, Meng ZY, Chen YY, Fu WG. Computational investigation of interaction between stent graft and aorta in retrograde type A dissection after thoracic endovascular aortic repair for type B aortic dissection. J Vasc Surg 2018;68(6S). 14S–21S–e2.
- [921] Meng Z, Ma T, Cai Y, Liu X, Wang S, Dong Z, Fu W. Numerical modeling and simulations of type B aortic dissection treated by stent-grafts with different oversizing ratios. Artif Organs 2020;44(11):1202–10.
- [922] Yuan X, Kan X, Yun X, Nienaber C. Finite element modeling to predict procedural success of thoracic endovascular aortic repair in type A aortic dissection. JTCVS Tech 2020;4:40–7.
- [923] Votta E, Presicce M, Della Corte A, Dellegrottaglie S, Bancone C, Sturla F, Redaelli A. A novel approach to the quantification of aortic root in vivo structural mechanics, Int J Numer Method. Biomed Eng 2017;33(9):e2849.
- [924] Kan X, Ma T, Dong Z, Xu XY. Patient-specific virtual stent-graft deployment for type B aortic dissection: a pilot study of the impact of stent-graft length. Front Physiol 2021;12:718140.
- [925] Kan X, Ma T, Jiang X, Holzapfel GA, Dong Z, Xu XY. Towards biomechanics-based pre-procedural planning for thoracic endovascular aortic repair of aortic dissection. Comput Methods Programs Biomed 2024;244:107994.
- [926] Dong ZH, Fu WF, Wang YQ, Guo DQ, Xu XY, Ji Y, Chen B, Jiang JH, Yang J, Shi ZY, Zhu T, Shi Y. Retrograde type A aortic dissection after endovascular stent graft placement for treatment of type B dissection. Circulation 2009;119(5):735–41.
- [927] Dong Z, Fu W, Wang Y, Wang C, Yan Z, Guo D, Xu XY, Chen B. Stent graft-induced new entry after endovascular repair for Stanford type B aortic dissection. J Vasc Surg 2010;52(6):1450–7.
- [928] Avril S, Gee MW, Hemmler A, Rugonyi S. Patient-specific computational modeling of endovascular aneurysm repair: state of the art and future directions. Int J Numer Method Biomed Eng 2021;37(12):e3529.
- [929] Mandigers TJ, Ramella A, Bissacco D, Domanin M, van Herwaarden JA, Heijmen R, Luraghi G, Migliavacca F, Trimarchi S. Thoracic stent graft numerical models to virtually simulate thoracic endovascular aortic repair: a scoping review. Eur J Vasc Endovasc Surg 2023;S1078–5884(23):00452–5.
- [930] Zhou X, Yand F, Gong X, Zhao M, Zheng Y, Sun Z. Development of new endovascular stent-graft system for type B thoracic aortic dissection with finite element analysis and experimental verification. J Mater Sci Technol 2019;35:2682–92.
- [931] Armour CH, Menichini C, Hanna L, Gibbs RGJ, Xu XY. Computational modeling of flow and thrombus formation in type B aortic dissection: the influence of false lumen perfused side branches. In: Sommer G, Li K, Haspinger DC, Ogden RW, editors. Solid (Bio)mechanics: Challenges of the Next Decade: A Book Dedicated to Professor Gerhard A. Holzapfel: Springer International Publishing, Cham; 2022. p. 53–72.
- [932] Menichini C, Xu XY. Mathematical modeling of thrombus formation in idealized models of aortic dissection: initial findings and potential applications. J Math Biol 2016;73(3):1205–26.
- [933] Menichini C, Cheng Z, Gibbs R, Xu XY. A computational model for false lumen thrombosis in type B aortic dissection following thoracic endovascular repair. J Biomech 2017;66:36–43.
- [934] Armour CH, Menichini C, Milinis K, Gibbs RGJ, Xu XY. Location of reentry tears affects false lumen thrombosis in aortic dissection following TEVAR. J Endovasc Ther 2020;27(3):396–404.
- [935] Brunkwall J, Kasprzak P, Verhoeven E, Heijmen R, Taylor P, Trialists ADSORB, Alric P, Canaud L, Janotta M, Raithel D, Malina W, Resch T, Eckstein HH, Ockert S, Larzon T, Carlsson F, Schumacher H, Classen S, Schaub P, Lammer J, Lönn L, Clough RE, Rampoldi V, Trimarchi S, Fabiani JN, Böckler D, Kotelis D, Böckler D, Kotelis D, von Tenng-Kobligk H, Mangialardi N, Ronchey S, Dialetto G, Matoussevitch V. Endovascular repair of acute uncomplicated aortic type B dissection promotes aortic remodelling: 1 year results of the ADSORB trial. Eur J Vasc Endovasc Surg 2014;48(3):285–91.
- [936] Jafarinia A, Müller TS, Windberger U, Brenn G, Hochrainer T. Blood rheology influence on false lumen thrombosis in type B aortic dissection. J Biosci Bioeng 2020;7:13–24.
- [937] Jafarinia A, Melito GM, Müller TS, Rolf-Pissarczyk M, Holzapfel GA, Brenn B, Ellermann K, Hochrainer T. Morphological parameters affecting false lumen thrombosis following type B aortic dissection: a systematic study based on simulations of idealized models. Biomech Model Mechanobiol 2023;22:885–904.
- [938] Jafarinia A, Armour CG, Gibbs RGJ, Xu XY, Hochrainer T. Shear-driven modelling of thrombus formation in type B aortic dissection. Front Bioeng Biotechnol 2022;10:831903.

- [939] Melito GM, Jafarinia A, Hochrainer T, Ellermann K. Sensitivity analysis of a phenomenological thrombosis model and growth rate characterisation. J Biosci Bioeng 2020;7:31–40.
- [940] Wang K, Armour CH, Gibbs RGJ, Xu XY. A numerical study of the effect of thrombus breakdown on predicted thrombus formation and growth. Biomech Model Mechanobiol 2024;23(1):61–71. https://doi.org/10.1007/s10237-023-01757-8.
- [941] Yang L, Neuberger T, Manning KB. In vitro real-time magnetic resonance imaging for quantification of thrombosis. MAGMA 2021;34(2):285–95.
- [942] Yazdani A, Li H, Humphrey JD, Karniadakis GE. A general shear-dependent model for thrombus formation. PLoS Comput Biol 2017;13(1):1–24.
- [943] Karniadakis GE, Sherwin S. Spectral/hp Element Methods for Computational Fluid Dynamics. 3rd Edition. Oxford University Press; 2013.
- [944] Pivkin IV, Richardson PD, Karniadakis GE. Blood flow velocity effects and role of activation delay time on growth and form of platelet thrombi. Proc Natl Acad Sci U S A 2006;103(46):17164–9.
- [945] Wang Y, Luo K, Qiao Y, Fan J. An integrated fluid-chemical model toward modeling the thrombus formation in an idealized model of aortic dissection. Comput Biol Med 2021;136:104709.
- [946] Bodnár T, Sequeira A. Numerical simulation of the coagulation dynamics of blood. Comput Math Methods Med 2008;9(2):83-104.
- [947] Anand M, Rajagopal K, Rajagopal KR. A model incorporating some of the mechanical and biochemical factors underlying clot formation and dissolution in flowing blood. J Theor Med 2003;5(3–4):183–218.
- [948] Anand M, Rajagopal K, Rajagopal KR. A model for the formation and lysis of blood clots. Pathophysiol Haemost Thromb 2005;34(2-3):109-20.
- [949] Jiang X, Li D, Wu P, Li X, Zheng T. A two-fluid blood stasis model for false lumen thrombosis after type B dissection repair. Comput Methods Biomech Biomed Engin 2022;25(13):1499–508.
- [950] Gupta I, Schanz M. Modelling growth and formation of thrombi: a multiphasic approach based on the theory of porous media. Arch Appl Mech 2023;93: 4107–23.
- [951] Li M, Ma T, Cai Y, Li J, Meng Z, Dong Z, Wang S. Numerical simulation of the distal stent graft-induced new entry after TEVAR. Int J Numer Method Biomed Eng 2024;40(5):e3819.
- [952] Wang Y, Luan J, Luo K, Zhu T, Fan J. Multi-constituent simulation of thrombosis in aortic dissection. Int J Eng Sci 2023;184:103817.
- [953] Pei Y, Song P, Zhang K, Dia M, He G, Wen J. Assessing the impact of tear direction in coronary artery dissection on thrombosis development: a hemodynamic computational study. Comput Methods Programs Biomed 2024;249:108144.
- [954] Schwarz EL, Pfaller MR, Szafron JM, Latorre M, Lindsey SE, Breuer CK, Humphrey AL, Marsden JD. A fluid-solid-growth solver for cardiovascular modeling. Comput Methods Appl Mech Eng 2023:116312.
- [955] Castro MA, Putman CM, Sheridan MJ, Cebral JR. Hemodynamic patterns of anterior communicating artery aneurysms: a possible association with rupture. AJNR Am J Neuroradiol 2009;30(2):297–302.
- [956] Kiema M, Sarin JK, Kauhanen SP, Torniainen J, Matikka H, Luoto ES, Jaakkola P, Saari P, Liimatainen T, Vanninen R, Ylä-Herttuala S, Hedman M, Laakkonen JP. Wall shear stress predicts media degeneration and biomechanical changes in thoracic aorta. Front Physiol 2022;13:934941.
- [957] Salmasi MY, Pirola S, Sasidharan S, Fisichella SM, Redaelli A, Jarral OA, O'Regan DP, Oo AY, Moore Jr JE, Xu XY, Athanasiou T. High wall shear stress can predict wall degradation in ascending aortic aneurysms: an integrated biomechanics study. Front Bioeng Biotechnol 2021;9:750656. https://doi.org/10.3389/ fbioe.2021.750656.
- [958] Minderhoud SCS, Arrouby A, van den Hoven AT, Bons LR, Chelu RG, Kardys I, Rizopoulos D, Korteland SA, van den Bosch AE, Budde RPJ, Roos-Hesselink JW, Wentzel JJ, Hirsch A. Regional aortic wall shear stress increases over time in patients with a bicuspid aortic valve. J Cardiovasc Magn Reson 2024;26(2): 101070. https://doi.org/10.1016/j.jocmr.2024.101070.
- [959] Nightingale M, Scott MB, Sigaeva T, Guzzardi D, Garcia J, Malaisrie SC, McCarthy P, Markl M, Fedak PWM, Di Martino ES, Barker AJ. Magnetic resonance imaging-based hemodynamic wall shear stress alters aortic wall tissue biomechanics in bicuspid aortic valve patient. J Thorac Cardiovasc Surg 2023;168(2): 465–476.e5.
- [960] Trenti C, Fedak PWM, White JA, Garcia J, Dyverfeldt P. Oscillatory shear stress is elevated in patients with bicuspid aortic valve and aortic regurgitation: a 4D flow cardiovascular magnetic resonance cross-sectional study. Eur Heart J Cardiovasc Imaging 2024;25(3):404–12.
- [961] Zolfaghari H, Andiapen M, Baumbach A, Mathur A, Kerswell RR. Wall shear stress and pressure patterns in aortic stenosis patients with and without aortic dilation captured by high-performance image-based computational fluid dynamics. PLoS Comput Biol 2023;19(10):e1011479.
- [962] Boussel L, Rayz V, McCulloch C, Martin A, Acevedo-Bolton G, Lawton M, Higashida R, Smith WS, Young WL, Saloner D. Aneurysm growth occurs at region of low wall shear stress: patient-specific correlation of hemodynamics and growth in a longitudinal study. Stroke 2008;39:2997–3002.
- [963] Meng H, Tutino VM, Xiang J, Siddiqui A. High WSS or low WSS? Complex interactions of hemodynamics with intracranial aneurysm initiation, growth, and rupture: toward a unifying hypothesis, AJNR Am J Neuroradiol 2014;35(7):1254–62.
- [964] Wang Y, Johnson JA, Spinale FG, Sutton MA, Lessner SM. Quantitative measurement of dissection resistance in intimal and medial layers of human coronary arteries. Exp Mech 2014;54(4):677–83.
- [965] Cebral JR, Detmer F, Chung BJ, Choque-Velasquez J, Rezai B, Lehto H, Tulamo R, Hernesniemi J, Niemela M, Yu A, Williamson R, Aziz K, Shakur S, Amin-Hanjani S, Charbel F, Tobe Y, Robertson A, Frösen J. Local hemodynamic conditions associated with focal changes in the intracranial aneurysm wall. AJNR Am J Neuroradiol 2019;40(3):510–6.
- [966] Frösen J, Cebral J, Robertson AM, Aoki T. Flow-induced, inflammation-mediated arterial wall remodeling in the formation and progression of intracranial aneurysms. Neurosurg Focus 2019;47(1):E21.
- [967] Ciurică S, Lopez-Sublet M, Loeys BL, Radhouani I, Natarajan N, Vikkula M, Maas AHEM, Adlam D, Persu A. Arterial tortuosity. Hypertension 2019;73(5): 951–60.
- [968] Poullis MP, Warwick R, Oo A, Poole RJ. Ascending aortic curvature as an independent risk factor for type A dissection, and ascending aortic aneurysm formation: a mathematical model. Eur J Cardiothorac Surg 2008;33(6):995–1001.
- [969] Sahni S, Bhatia SK. Ascending aortic curvature as an independent risk factor for aortic dissection: the mathematical model and underlying equations. Eur J Cardio-Thorac Surg 2012;42(4):755.
- [970] Arcari L, Limite LR, Cacciotti L, Alonzo A, Musumeci MB, Passaseo I, Marazzi G, Berni A, Ansalone G, Volpe M, Autore C. Tortuosity, recurrent segments, and bridging of the epicardial coronary arteries in patients with the Takotsubo syndrome. Am J Cardiol 2017;119(2):243–8.
- [971] Saba L, Sanfilippo R, Suri JS, Cademartiri F, Corrias G, Mannelli L, Zucca S, Senis I, Montisci R, Wintermark M. Does carotid artery tortuosity play a tole in stroke? Can Assoc Radiol J 2021;72(4):789–96.
- [972] Welby JP, Kim ST, Carr CM, Lehman VT, Rydberg CH, Wald JT, Luetmer PH, Nasr DM, Brinjikji W. Carotid artery tortuosity is associated with connective tissue diseases. AJNR Am J Neuroradiol 2019;40(10):1738–43.
- [973] Li J, Cao X, Xu L, Qi L. Finite element analysis for the effects of the descending aorta tortuosity on aortic hemodynamics. Procedia Comput Sci 2022;209: 148–56.
- [974] Lane B, Sherifova S, Santamaría VA, Molimard J, Holzapfel GA, Avril S. Novel experimental methods to characterize the mechanical properties of the aorta. In: Gasser TC, Avril S, Elefteriades JA, editors. Biomechanics of the Aorta: Modeling for Patient Care. Biomechanics of Living Organs: Academic Press; 2024. p. 91–108.
- [975] Geith MA, Nothdurfter L, Heiml M, Agrafiotis E, Gruber M, Sommer G, Schratzenstaller TG, Holzapfel GA. Quantifying vascular damage by investigating stenttriggered mechanical and morphological alterations in coronary arteries. Acta Biomater 2020;116:285–301.
- [976] Agrafiotis E, Mayer C, Grabenwöger M, Zimpfer D, Regitnig P, Mächler H, Holzapfel GA. Global and local stiffening of ex vivo-perfused stented human thoracic aortas: a mock circulation study. Acta Biomater 2023;161:170–83.
- [977] Suh G-Y, Beygui RE, Fleischmann D, Cheng CP. Aortic arch vessel geometries and deformations in patients with thoracic aortic aneurysms and dissections. J Vasc Interv Radiol 2014;25(12):1903–11.
- [978] Suh G-Y, Fleischmann D, Beygui RE, Cheng CP. Quantification of motion of the thoracic aorta after ascending aortic repair of type-A dissection. Int J Comput Assist Radiol Surg 2017;12(5):811–9.

- [979] Talvitie M, Lindquist Liljeqvist M, Siika A, Hultgren R, Roy J. Localized hyperattenuations in the intraluminal thrombus may predict rupture of abdominal aortic aneurysms. J Vasc Interv Radiol 2018;29(1):144–5.
- [980] Sakalihasan N, Michel JB, Katsargyris A, Kuivaniemi H, Defraigne JO, Nchimi A, Powell JT, Yoshimura K, Hultgren R. Abdominal aortic aneurysm. Nat Rev Dis Primers 2018;4(1):34.
- [981] Trenti C, Ziegler M, Bjarnegard N, Ebbers T, Lindenberger M, Dyverfeldt P. Wall shear stress and relative residence time as potential risk factors for abdominal aortic aneurysms in males: a 4D flow cardiovascular magnetic resonance case-control study. J Cardiovasc Magn Reson 2024;24(1):18.
- [982] Virag L, Horvat N. I. Karay, Bio-chemo-mechanical role of intraluminal thrombus deposition on arterial tissue growth and remodeling. In: Sommer G, Li K, Haspinger DC, Ogden RW, editors. Solid (Bio)mechanics: Challenges of the Next Decade: A Book Dedicated to Professor Gerhard A. Holzapfel: Springer International Publishing, Cham; 2022. p. 119–41.
- [983] Morrison TM, Pathmanathan P, Adwan M, Margerrison E. Advancing regulatory science with computational modeling for medical devices at the FDA's office of science and engineering laboratories. Front Med (Lausanne) 2018;5:241.
- [984] Viceconti M, Emili L, editors. Towards Good Simulation Practice: Best Practice for the Use of Computational Modelling and Simulation in the Regulatory Process of Biomedical Products. Springer Cham; 2024.
- [985] Viceconti M, Henney A, Morley-Fletcher E. In silico clinical trials: how computer simulation will transform the biomedical industry. Int J Clin Trials 2016;3(2): 37–46.
- [986] Viceconti M, Pappalardo F, Rodriguez B, Horner M, Bischoff J, Tshinanu FM. In silico trials: verification, validation and uncertainty quantification of predictive models used in the regulatory evaluation of biomedical products. Methods 2021;185:120–7.
- [987] Viceconti M, Emili L, Afshari P, Courcelles E, Curreli C, Famaey N, Geris L, Horner M, Jori MC, Kulesza A, Loewe A, Neidlin M, Reiterer M, Rousseau CF, Russo G, Sonntag SJ, Voisin EM, Pappalardo F. Possible contexts of use for in silico trials methodologies: a consensus-based review. IEEE J Biomed Health Inform 2021;25:3977–82.
- [988] Pappalardo F, Wilkinson J, Busquet F, Bril A, Palmer M, Walker B, Curreli C, Russo G, Marchal T, Toschi E, Alessandrello R, Costignola V, Klingmann I, Contin M, Staumont B, Woiczinski M, Kaddick C, Salvatore VD, Aldieri A, Geris L, Viceconti M. Toward a regulatory pathway for the use of in silico trials in the CE marking of medical devices. IEEE J Biomed Health Inform 2022;26(11):5282–6.
- [989] Favre P, Bischoff J. Identifying the patient harms to include in an in silico clinical trial. Comput Methods Programs Biomed 2023;241:107735.
- [990] Brüning J, Yevtushenko P, Schlief A, Jochum T, van Gijzen L, Meine S, Romberg J, Kuehne T, Arndt A, Goubergrits L. In-silico enhanced animal study of pulmonary artery pressure sensors: assessing hemodynamics using computational fluid dynamics. Front Cardiovasc Med 2023;10:1193209.
- [991] Corral-Acero J, Margara F, Marciniak M, Rodero C, Loncaric F, Feng Y, Gilbert A, Fernandes JF, Bukhari HA, Wajdan A, Martinez MV, Santos MS, Shamohammdi M, Luo H, Westphal P, Leeson P, DiAchille P, Gurev V, Mayr M, Geris L, Pathmanathan P, Morrison T, Cornelussen R, Prinzen F, Delhaas T, Doltra A, Sitges M, Vigmond EJ, Zacur E, Grau V, Rodriguez B, Remme EW, Niederer S, Mortier P, McLeod K, Potse M, Pueyo E, Bueno-Orovio A, Lamata P. The 'Digital Twin' to enable the vision of precision cardiology. Eur Heart J 2020;41(48):4556–64.
- [992] Viceconti M, De Vos M, Mellone S, Geris L. Position paper from the digital twins in healthcare to the Virtual Human Twin: a moon-shot project for digital health research, IEEE J Biomed Health. Inform 2023;11:PP.
- [993] Chase JG, Preiser JC, Dickson JL, Pironet A, Chiew YS, Pretty CG, Shaw GM, Benyo B, Moeller K, Safaei S, Tawhai M, Hunter P, Desaive T. Next-generation, personalised, model-based critical care medicine: a state-of-the art review of in silico virtual patient models, methods, and cohorts, and how to validation them. Biomed Eng Online 2018;17(1):24.
- [994] Romero P, Lozano M, Martínez-Gil F, Serra D, Sebastián R, Lamata P, García-Fernández I. Clinically-driven virtual patient cohorts generation: an application to aorta. Front Physiol 2021;12:713118.
- [995] Gonzales A, Guruswamy G, Smith SR. Synthetic data in health care: a narrative review. PLOS Digit Health 2023;2(1):e0000082.
- [996] Verstraeten S, Hoeijmakers M, Tonino P, Brüning J, Capelli C, van de Vosse F, Huberts W. Generation of synthetic aortic valve stenosis geometries for in silico trials. Int J Numer Method Biomed Eng 2024;40(1):e3778.
- [997] Ostendorf K, Bäumler K, Mastrodicasa, Fleischmann D, Preim B, Mistelbauer G. Synthetic surface mesh generation of aortic dissections using statistical shape modeling. Comput Graph 2024;124:104070. https://doi.org/10.1016/j.cag.2024.104070.
- [998] American Society of Mechanical Engineers. Assessing Credibility of Computational Modeling through Verification and Validation: Application to Medical Devices, ASME V&V40-2018, 2018.
- [999] Food and Drug Administration (FDA) Center for Devices and Radiological Health, Assessing the Credibility of Computational Modeling and Simulation in Medical Device Submissions, 2021. https://www.fda.gov/media/154985/download.
- [1000] Lesagne R, van Oudheusden M, Schievano S, van Hoyweghen I, Geris L, Capelli C. Mapping the use of computational modelling and simulation in clinics: a survey. Front Med Technol 2023;5:1125524.
- [1001] Ranftl S, Müller TS, Windberger U, Brenn G, von der Linden W. A Bayesian approach to blood rheological uncertainties in aortic hemodynamics. Int J Numer Meth Biomed Engng 2021;39:e3576.
- [1002] Moireau P, Bertoglio C, Xiao N, Figueroa CA, Taylor CA, Chapelle D, Gerbeau JF. Sequential identification of boundary support parameters in a fluid-structure vascular model using patient image data. Biomech Model Mechanobiol 2013;12(3):475–96.
- [1003] Bracco MI, Broda M, Lorenzen US, Florkow MC, Somphone O, Avril S, Biancolini ME, Rouet L. Fast strain mapping in abdominal aortic aneurysm wall reveals heterogeneous patterns. Front Physiol 2023;14:1163204.
- [1004] Tezduyar TE, Sathe S, Schwaab M, Conklin BS. Arterial fluid mechanics modeling with the stabilized space-time fluid-structure interaction technique. Int J Numer Meth Fluids 2008;57:601-29.
- [1005] Marx L, Niestrawska JA, Gsell MAF, Caforio F, Plank G, Augustin CM. Robust and efficient fixed-point algorithm for the inverse elastostatic problem to identify myocardial passive material parameters and the unloaded reference configuration. J Comput Phys 2022;463:111266.
- [1006] Gee MW, Reeps C, Eckstein HH, Wall WA. Prestressing in finite deformation abdominal aortic aneurysm simulation. J Biomech 2009;42(11):1732–9. [1007] Gee MW, Förster C, Wall WA. A computational strategy for prestressing patient-specific biomechanical problems under finite deformation. Int J Numer Meth
- Biomed Engng 2010;26:52–72.
- [1008] Hang S, Sadrehaghighi I. TetGen, a Delaunay-based quality tetrahedral mesh generator. ACM Trans Math Softw 2015;41(2):1-36.
- [1009] Ray N, Sokolov D, Reberol M, Ledoux F, Lévy B. Hex-dominant meshing: mind the gap! Comput Aided Des 2018;102:94–103.
   [1010] De Santis G, De Beule M, Van Canneyt K, Segers P, Verdonck P, Verhegghe B. Full-hexahedral structured meshing for image-based computational vascular modeling. Med Eng Phys 2011;10(10):1318–25.
- [1011] Armstrong CG, Fogg HJ, Tierney CM, Robinson TT. Common themes in multi-block structured quad/hex mesh generation. Procedia Eng 2015;124:70-82.
- [1012] Vukicevic AM, Çimen S, Jagic N, Jovicic G, Frangi AF, Filipovic N. Three-dimensional reconstruction and NURBS-based structured meshing of coronary arteries from the conventional X-ray angiography projection images. Sci Rep 2018;8(1):1711.
- [1013] Bošnjak D, Pepe A, Schussnig R, Schmalstieg R, Fries T-P. Higher-order block-structured hex meshing of tubular structures. Eng Comput 2024;40:931–51. [1014] Bošnjak D, Pepe A, Schussnig R, Egger J, Fries T-P. A semi-automatic method for block-structured hexahedral meshing of aortic dissections. Int J Numer
- Method Biomed Eng 2024:e3860. [1015] Bayer JD, Blake RC, Plank G, Trayanova NA. A novel rule-based algorithm for assigning myocardial fiber orientation to computational heart models. Ann
- Biomed Eng 2012;40:2243–54.
  [1016] Berhane H, Scott M, Elbaz M, Jarvis K, McCarthy P, Carr J, Malaisrie C, Avery AJ, Barker R, Robinson CK, Rigsby JD, Markl M. Fully automated 3D aortic segmentation of 4D flow MRI for hemodynamic analysis using deep learning. Magn Reson Med 2020;84(4):2204–18.
- [1017] Liang L, Mao W, Sun W. A feasibility study of deep learning for predicting hemodynamics of human thoracic aorta. J Biomech 2020;99:109544.
- [1018] Hahn LD, Mistelbauer G, Higashigaito K, Koci M, Willemink MJ, Sailer AM, Fischbein M, Fleischmann D. CT-based true- and false-lumen segmentation in type B aortic dissection using machine learning. Radiol Cardiothorac Imaging 2020;2(3):e190179.

- [1019] Yu Y, Gao Y, Wei J, Liao F, Xiao Q, Zhang J, Yin W, Lu B. A three-dimensional deep convolutional neural network for automatic segmentation and diameter measurement of type B aortic dissection. Korean J Radiol 2021;22(2):168–78.
- [1020] Chen X, Liu J, Pang L, Chi Y, Gong C. Developing a new mesh quality evaluation method based on convolutional neural network. Eng Appl Comput Fluid Mech 2019;14(1):391–400.
- [1021] Chen X, Liu J, Gong C, Li S, Pang Y, Chen B. MVE-Net: an automatic 3-D structured mesh validity evaluation framework using deep neural networks. Comput Des 2021;141:103104.
- [1022] Zhang Z, Jimack PK, Wang H. MeshingNet3D: efficient generation of adapted tetrahedral meshes for computational mechanics. Adv Eng Softw 2021;157–158 (2):103021.
- [1023] Du P, Zhu X, Wang J-X. Deep learning-based surrogate model for 3-dimensional patient-specific computational fluid dynamics. Phys Fluids 2022;34(8): 081906
- [1024] Zhu Y, Xu XY, Rosendahl U, Pepper J, Mirsadraee S. Advanced risk prediction for aortic dissection patients using imaging-based computational flow analysis. Clin Radiol 2023;78(3):e155–65.
- [1025] Romarowski RM, Faggiano E, Conti M, Reali A, Morganti S, Auricchio F. A novel computational framework to predict patient-specific hemodynamics after TEVAR: integration of structural and fluid-dynamics analysis by image elaboration. Comput Fluids 2019;179:806–19.
- [1026] Chinesta F, Cueto E, Payan Y, Ohayon J, editors. Reduced order models for the biomechancis of living organs. Elsevier; 2023.